Disaster Resilience and Green Growth Series Editors: Anil Kumar Gupta · SVRK Prabhakar · Akhilesh Surjan

Pallavi Saxena Anuradha Shukla Anil Kumar Gupta *Editors*

Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation



Disaster Resilience and Green Growth

Series Editors

Anil Kumar Gupta, National Institute of Disaster Management, New Delhi, Delhi, India

SVRK Prabhakar, Climate Change Adaptation, Institute of Global Environment Strategies, Kanagawa, Japan

Akhilesh Surjan, College of Indigenous Futures, Arts and Society, Charles Darwin University, Darwin, Australia

Over the years, the relationship between environment and disasters has received significant attention. This is largely due to the emerging recognition that environmental changes - climate change, land-use and natural resource degradation make communities more vulnerable to disaster impacts. There is a need to break this nexus through environment based and sustainability inclusive interventions. Science – technology and economic measures for disaster risk management, hence, need to adapt more integrated approaches for infrastructure and social resilience. Environmental and anthropogenic factors are key contributors to hazard, risk, and vulnerability and, therefore, should be an important part of determining risk-management solutions.

Green growth approaches have been developed by emphasizing sustainability inclusion and utilizing the benefits of science-technology interventions along policypractice linkages with circular economy and resource efficiency. Such approaches recognize the perils of traditional material-oriented economy growth models that tend to exploit natural resources, contribute to climate change, and exacerbate disaster vulnerabilities, Green growth integrated approaches are rapidly becoming as preferred investment avenue for mitigating climate change and disaster risks and for enhancing resilience. This includes ecosystem-based and nature-based solutions with potential to contribute to the resilience of infrastructure, urban, rural and periurban systems, livelihoods, water, and health. They can lead to food security and can further promote people-centric approaches.

Some of the synergistic outcomes of green growth approaches include disaster risk reduction, climate change mitigation and adaptation, resilient livelihoods, cities, businesses and industry. The disaster risk reduction and resilience outcome of green growth approaches deserve special attention, both for the academic and policy communities. Scholars and professionals across the domains of DRR, CCA, and green growth are in need of publications that fulfill their knowledge needs concerning the disaster resilience outcomes of green growth approaches. Keeping the above background in view, the book series offers comprehensive coverage combining the domains of environment, natural resources, engineering, management and policy studies for addressing disaster risk and resilience in the green growth context in an integrated and holistic manner. The book series covers a range of themes that highlight the synergistic outcomes of green growth approaches.

The book series aims to bring out the latest research, approaches, and perspectives for disaster risk reduction along with highlighting the outcomes of green growth approaches and including Science-technology-research-policy-practice interface, from both developed and developing parts of the world under one umbrella. The series aims to involve renowned experts and academicians as volume-editors and authors from all the regions of the world. It is curated and developed by authoritative institutions and experts to serve global readership on this theme. Pallavi Saxena • Anuradha Shukla • Anil Kumar Gupta Editors

Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation



Editors Pallavi Saxena Hindu College University of Delhi New Delhi, Delhi, India

Anuradha Shukla Transport Planning and Environment Division Central Road Research Institute New Delhi, Delhi, India

Anil Kumar Gupta Environment, Climate and Disaster Risk Management National Institute of Disaster Management New Delhi, Delhi, India

ISSN 2662-4885 ISSN 2662-4893 (electronic) Disaster Resilience and Green Growth ISBN 978-981-16-7726-7 ISBN 978-981-16-7727-4 (eBook) https://doi.org/10.1007/978-981-16-7727-4

© The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors, and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Singapore Pte Ltd. The registered company address is: 152 Beach Road, #21-01/04 Gateway East, Singapore 189721, Singapore

Foreword

Human society can be very tolerant of activities and phenomena which are highly undesirable if they develop gradually. One example is road traffic accidents which reap a huge toll of death and injury internationally, but are tolerated as part of everyday living, and are one of the risks of life which has to be accepted. One only has to look to air travel, which is far more highly regulated to ensure passenger safety, to see that things could be different and that such a toll is not inevitable. Outdoor air pollution causes around 30,000 premature deaths annually in my own country (the UK), and according to the Global Burden of Disease Study around four million worldwide, with a similar number of deaths arising from exposure to indoor sources. This is another risk of life which is tolerated because it has been there for longer than anyone can remember, and in the UK, documentation of urban air pollution problems extends back for many centuries although it was the nuisance, rather than the health risks which led to the earliest legislation. The current downward trend in air pollution in developed countries stems largely from a few extreme events. Probably, the best documented event is the 1952 London smog which killed 4000 (and most probably more) people over a period of a week of stagnant weather. Similar events in the Meuse Valley of Belgium and Donora, Pennsylvania, USA, were also key to raising awareness and stimulating change.

We now know that air pollution has effects far beyond those on human health, and extreme events serve to highlight the risks, although often the consequences are the result of long-term lower level exposures. Extreme air pollution events have been very important in the past and will continue to be in the future until all countries have achieved acceptable air quality, which will take many decades, especially as research advances continue to show adverse effects at ever lower concentrations.

Roy M. Harrison

I very much welcome the initiative by three talented scientists to develop this book and believe that a deep understanding of extreme events is one of the keys to safer air for us all.

Queen Elizabeth II Birmingham Centenary Professor of Environmental Health, School of Geography, Earth and Environmental Sciences, University of Birmingham Birmingham, UK July 2021

Preface

Extremes in atmospheric processes and phenomena often lead to change in air quality and can have harmful impacts on climatic processes as well as human lives. Atmospheric extreme events are generally very intense and may lead to short-lived as well as long-lived changes in the environment. In the recent years, extremes in atmosphere and air pollution-related emergencies like smog, biomass burning, dust storms, heat waves, etc. have increased and, therefore, there is an urgent need to focus the attention of governments, policy makers and scientists on such issues and to come up with sustainable solutions for present and future.

Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation encompasses a holistic comprehension on fundamental concepts of atmospheric processes, mechanisms, phenomena, air pollution episodes and extreme events, impacts and mitigation strategies. Chapters provide a holistic overview of atmospheric extreme events like dust storms, fog, smog, biomass burning, volcanic eruptions, heat waves etc. and also describe their impacts on environment, challenges and possible strategies to mitigate and control them.

This book covers a special approach towards disaster risk management methods, policies and governance to manage and mitigate atmospheric extreme events which can help reduce the havoc and damaging consequences from them. This book is expected to serve as valuable knowledge support system for scientists, policy makers, academicians, consultants and research scholars in the area of Environmental Science, Biological Science, Medical Science, Policy Planning, Disaster Management and Agriculture. It will also be a unique document for air quality professionals like meteorologists, atmospheric chemists, modellers and disaster management experts.

New Delhi, India

Pallavi Saxena Anuradha Shukla Anil Kumar Gupta

Acknowledgement

Editors gratefully acknowledge all the authors who contributed chapters on important attributes of book to help make it comprehensive and holistic reference publication. We also wish to pay our sincere thanks to Ms. Aakanksha Tyagi, Associate Editor, Springer, for her cooperation and help in completion of this assignment.

Dr. Pallavi Saxena pays her sincere thanks to Prof. Anju Srivastava, Principal, Hindu College, University of Delhi, for the encouragement and support in research field. Dr. Saxena is highly thankful to iLEAPS community for receiving the exposure at international level in the field of atmospheric sciences that have driven the motivation for the idea and framework of this manuscript. She also acknowledges her parents, Dr. Akhilesh Saxena and Dr. Neelima Priyadarshini for their constant support, encouragement and act as inspiration to work for science and society. Dr. Anuradha Shukla, co-editor conveys her sincere gratitude to her husband Dr. Dilip Shukla and daughters Ankeeta and Akanksha for their support and motivation. Prof. Anil K. Gupta expresses gratitude to the encouragements from Dr. Akhilesh Gupta, Senior Advisor, Head of Climate Change Programme at DST-GOI and to his team at the Centre for Excellence on Climate Resilience/ ECDRM Division NIDM for support and cooperation in all the strategic and academic pursuits.

Contents

1	An Introduction to Extremes in Atmospheric Processes and Phenomena: Assessment, Impacts and Mitigation Pallavi Saxena, Anuradha Shukla, and Anil K. Gupta	1
2	Atmospheric Phenomena: Origin, Mechanism, and Impacts Vanisa Surapipith and Pornpan Uttamang	9
3	Air Pollution and Its Associated Impacts on Atmosphere and Biota Health	29
4	South Asian Monsoon Extremes and Climate Change	59
5	Contribution of Fog in Changing Air Quality: Extremities and Risks to Environment and Society Saraswati	87
6	Nature of Sand and Dust Storm in South Asian Region:Extremities and Environmental ImpactsSanjoy Maji and Saurabh Sonwani	113
7	Assessment of Heat and Cold Waves Phenomena and Impacts on Environment	141
8	Intense Biomass Burning Over Northern India and Its Impact on Air Quality, Chemistry and Climate	169
9	Rising Extreme Event of Smog in Northern India: Problemsand ChallengesAshima Sharma and Renu Masiwal	205
10	Volcanic Emissions: Causes, Impacts, and Its Extremities Rahul Kant Yadav, Debasish Mahapatra, and Chinmay Mallik	237

11	Assessment of Extreme Firework Episode in a Coastal City of Southern India: Kannur as a Case Study	267
12	Air Pollution Episodes: Brief History, Mechanisms and Outlook Narendra Ojha, Meghna Soni, Manish Kumar, Imran Girach, Som Kumar Sharma, and Sachin S. Gunthe	283
13	Increasing Atmospheric Extreme Events and Role of Disaster Risk Management: Dimensions and Approaches Madhavi Jain	303
14	Disaster Preparedness and Emergency Response for Air Pollution and Related Health Extremes	329
15	Cost-Effective Technologies for Control of Air Pollution and Atmospheric-Related Extremes	349
16	Ecological and Natural-Based Solutions as Green Growth Strategies for Disaster and Emergency Management of Air Pollution Extremes	369
17	Policy Implications and Mitigation Strategies for Air Pollution and Atmospheric Extreme Events Disha Sharma	397
18	Participatory and Collaborative Governance Approach for Management of Atmospheric Processes-Related Extremities Kamna Sachdeva	411
Glossary		

About the Editors

Pallavi Saxena is an Assistant Professor, Environmental Science, Hindu College, University of Delhi, India. She has been awarded DST Fast Track Young Scientist Award at JNU, New Delhi, India. She has completed her Post Doc from PRL, Ahmedabad, India, and Ph.D. in Environmental Studies from DU in the field of atmospheric sciences. Her area of interest is air pollution and environmental health. She is serving as SSC member and earlier, served as Chair of South Asia and Middle East Region of ECSN in iLEAPS, UK, and co-author of TOAR, USA. She has more than 45 research publications.

Anuradha Shukla is an Ex-Chief Scientist and Advisor of R&D at Transport Planning and Environment Division, CSIR-CRRI, New Delhi, India. She has completed her M.Sc. and M.Tech. from IIT and awarded Ph.D. in Environmental Science and Engineering from JMI, Delhi. Her area of expertise is air pollution due to road transport. She was a recipient of Fulbright-Nehru Environmental Leadership Program, Visiting Fellow at the University of Birmingham, UK, and International Scientist Exchange Programmer under Indo-Hungarian Programme. She has also published more than 100 research papers and served as Chairperson, SWM Sectional Committee, BIS Govt. of India, Member, EAC for projects.

Anil Kumar Gupta is Head, Environment, Climate and DRM Division, NIDM, India. He is Programme Director of CECR and implementing projects, CAP-RES, NADMP, DMP and Nodal Officer for UNCCD, UNFCCC. He is Expert Team Member of WMO Climate and South Asia Member of IUCN-CEM. He has to credit over 150 publications. He is principal editor of Springer-Nature Book series on Disaster Resilience and Green Growth. He was recipient of National Science Day Young Scientist Award, Excellence Award by the SEOH and IDRC Canada's Thank Tank Initiative Senior Fellowship. He was Head and Director (Acting) of IEDS, Bundelkhand University.

Abbreviations

AD	Aerodynamic diameter
AERMOD	Atmospheric dispersion module
AMSL	Above mean sea level
AOD	Aerosol optical depth
APEC	Asia-Pacific Economic Cooperation
APTI	Air pollution tolerance index
AQI	Air quality index
BC	Black carbon
BOB	Bay of Bengal
BSOA	Biogenic secondary organic aerosol
BTEX	Benzene toluene ethylbenzene xylene
CAP	Comprehensive action plan
CC	Carbonate carbon
CCAC	Climate and clean air coalition
CCN	Cloud condensation nuclei
CEC	Cation exchange capacity
CEPI	Comprehensive environmental pollution index
CFCs	Chlorofluorocarbons
CLRTAP	Convention on long-range transboundary air pollution
CMIP	Coupled model intercomparison project
CNG	Compressed natural gas
CO	Carbon monoxide
CONUS	Continental United State
COPD	Chronic obstructive pulmonary disease
CPCB	Central Pollution Control Board
CRB	Crop residue burning
CWWG	Crop Weather Watch Group
DGR	Diameter growth rate
DI	Discomfort INDEX
DMI	Dipole moment index
DOC	Dissolved organic carbon
DS	Dust storm
DU	Dobson unit

EC	Elemental carbon
ECMWF	European Centre for Medium-Range Weather Forecasts
EDGAR	Emissions database for global atmospheric research
EIA	Environmental impact assessment
ENSO	El Nino Southern oscillation
ESP	Electrostatic precipitator
EU	European Union
FHWA	Federal Highway Administration
GCMS	Gas chromatography mass spectrometry
GDP	Gross Domestic product
GFED	Global fire emissions database
GHGs	Greenhouse gases
GIS	Geographical information system
GRAP	Graded response action plan
GTP	Global temperature change potential
GWP	Global warming potential
H_2SO_4	Sulphuric acid
HGR	Height growth rate
HHWSs	Heat-health warning systems
HNO ₃	Nitric acid
HW	Heat waves
IARI	Indian Agricultural Research Institute
IAV	Inter annual variability
IAVCEI	International Association of Volcanology and Chemistry of the Earth's
	Interior
ICAO	International Civil Aviation Organization
ICMR	Indian Council of Medical Research
IGP	Indo gangetic plain
IHME	Institute for Health Metrics and Evaluation
IMD	Indian Meteorological Department
IOD	Indian Ocean dipole
IPCC	Intergovernmental panel for climate change
IPM	Inhalable particulate matter
IPO	Interdecadal pacific oscillation
ISMR	Indian summer monsoon rainfall
ITCZ	Intertropical convergence zone
LIPs	Large igneous provinces
LLJ	Somali low-level jet
LPS	Low pressure system
LRTAP	Long-range transport of air pollutants
MH	Mascarene high
MIC	Methyl isocyanate
MJC	Madden Julian oscillation
MODIS	Moderate resolution imaging spectroradiometer
MoES	Ministry of earth sciences

NAAQS	National ambient air quality standards
NCAP	National clean air programme
NCDs	Non-communicable diseases
NCP	Northern China plain
NDMA	National Disaster Management Authority
NFHS	National family health survey
NHAP	National health adaptation plan
NIDM	National Institute of Disaster Management
NMVOCs	Non-methane volatile organic compounds
NOAA	National Oceanic Atmospheric Administration
NOx	Nitrogen oxides
O ₃	Ozone
OC	Organic carbon
ОН	Hydroxyl radical
OLR	Outgoing longwave radiation
OP	Oxidative potential
PAHs	Polycyclic aromatic hydrocarbons
PBL	Planetary boundary layer
PM	Particulate matter
PUC	Pollution under control
RCP	Representative concentration pathway
RH	Relative humidity
ROT	Runway occupancy time
RYL	Relative yield loss
SAM	South Asian monsoon
SDS	Sand and dust storms
SGVP	Smithsonian global volcanism program
SIA	Secondary inorganic aerosols
SLCPs	Short-lived climate pollutants
SO_2	Sulphur dioxide
SOĀ	Secondary organic aerosol
SPI	Standardized precipitation index
SSH	Sea surface height
SST	Sea surface temperature
TCO	Total columnar ozone
TEA	Triethylamine
TEJ	Tropical easterly jet
TMA	Trimethylamine
TRAP	Traffic-related air pollution
TRMM	Tropical rainfall measuring mission
UAVs	Unmanned aerial vehicles
UFP	Ultrafine particles
UHI	Urban Heat Island
UKMO	United Kingdom Met Office
UKWO	Office Kingdom Met Office

UNCCD	United Nations convention to combat desertification
UNECE	United Nations Economic Commission for Europe
UNFCCC	United Nations framework convention on climate change
USEPA	United States Environmental Protection Agency
VEI	Volcanic explosivity index
VFR	Visual flight rules
VHR	Very heavy rainfall
VOCs	Volatile organic compounds
WD	Western disturbance
WHO	World Health Organization
WMO	World Meteorological Organization
WSN	Wireless sensor network
WSOC	Water-soluble organic carbon



An Introduction to Extremes in Atmospheric Processes and Phenomena: Assessment, Impacts and Mitigation

Pallavi Saxena, Anuradha Shukla, and Anil K. Gupta

Abstract

Extremes in atmospheric phenomena and processes can strongly affect ambient air quality that has harmful impacts on human and climate health. High urbanization and industrialization are highly responsible for causing disturbances in different spheres of environment like atmosphere and its processes and phenomena. This will further impact human and climate health. Various episodic extreme events are reported in urban areas causing health impacts like respiratory disorders, cardiovascular diseases and damage to central nervous system that often lead to death. This chapter focuses on factors governing in atmosphere that resulted in air pollution, fundamental understanding about recent and historic air pollution episodes, emergencies and extremes and their impacts on living organisms. Moreover, possible mitigation strategies have also been discussed to reduce the occurrence of air pollution episodes and minimize the disturbance in atmospheric processes. It also summarizes the highlights of different chapters signifying various aspects of atmospheric processes, mechanisms, extremities, impacts and disaster risk mitigation strategies.

P. Saxena (🖂)

A. Shukla

A. K. Gupta

Department of Environmental Sciences, Hindu College, University of Delhi, Delhi, India e-mail: pallavisaxena@hinducollege.ac.in

Department of Transport Planning and Environment Division, CSIR-Central Road Research Institute, Delhi, India

Department of Environment, Climate Change and Disaster Risk Management, National Institute of Disaster Management, New Delhi, India

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_1

Keywords

 $Atmosphere \cdot Air \ quality \cdot Extreme \ events \cdot Mitigation \cdot Disaster \ risk \\ management$

1.1 General Introduction

Extreme events in atmospheric processes and phenomena are majorly responsible for affecting the air quality and pose deleterious impact on plant, animal, human and material health. Extreme events can be heat waves, extreme precipitation, volcanic eruptions, smog, biomass burning, fireworks, dust storms and many more which are significantly important to impair the air quality and can show impacts in the long run on environment (Jacob and Winner 2009; Saxena and Sonwani 2019; Sonwani and Saxena, 2022). Extremities in atmosphere can occur due to drastic change in weather processes, disturbance in atmospheric processes and climate change that ultimately affect air quality. Pattern changes in duration, frequency and spatial level in weather extremes have been discussed by Intergovernmental Panel on Climate Change (IPCC 2007; Blanken et al. 2022). For instance, extremities in heat waves can cause high temperature; increased biogenic emissions and rapid chemical reactions can give rise to high production of ozone and fine particulate matter concentrations. Goel et al. (2021) reported that fine particulate matter-related health hazards like premature mortality, respiratory ailments and cardiovascular diseases may increase mostly in dense populated areas and can become more worsen if not controlled. Climate prediction models, emission inventories and population dynamics model are greatly acknowledged for assessing the possibility of health hazards in a particular area due to extremes in air pollution and atmospheric processes. Such assessment practices can be further improved if the relationship between air pollution events, atmospheric processes and phenomena is presented in a more understandable and efficient way (Saxena and Srivastava 2020; Sonwani et al. 2021).

In the recent era, extremes in atmosphere and air pollution-related emergencies (fog, smog, health-related, heat and cold wave) has become a major environmental hazard and a big reason to affect human health (Pant and Harrison 2013; Saxena et al. 2020; Tiwari and Saxena 2021). Outdoor air pollution has become one of the major threats to human health and was estimated to cause 3.9 million premature deaths each year (WHO Fact Sheet 2018). Bell et al. (2004) reported that 0.52% high-risk daily mortality was caused by ozone in megacities of the United States over the period of 1987–2000. Satellite-based observations have shown that worldwide population-weighted fine particulate matter (PM_{2.5}) has increased by 0.55 μ gm³, mostly covering an exponential increase in East and South Asia (3.2 and 2.9% per year). Moreover, projection models also estimate that contribution of megacities to worldwide mortality from outdoor air pollution will rise from 60 to 65% of the total from 2010 to 2050 (Lelieveld et al. 2015; Saxena and Sonwani 2019). The effects caused by air pollution can lead to various health hazards like chronic obstructive pulmonary disease, acute respiratory infections, bronchitis, etc (Sonwani and Saxena



Fig. 1.1 Comparison of healthy and infected respiratory system due to inhalation of particulate matter in atmosphere (Source: www.blf.org.uk)

2016). Figure 1.1 shows the comparison of respiratory system of healthy and infected person due to inhalation of particulate matter in atmosphere. Populations living in developing countries and underdeveloped countries suffer from both indoor as well as outdoor air pollution (Pope III et al. 2009; Saxena et al. 2012).

Atmospheric processes and extremes like fog episodes also give rise to health hazards and huge impact on environment. Rise in aerosol concentrations in urban areas due to high urbanization and industrialization and high vehicular density catalyses fog events. Increase in relative humidity and aerosols are major factors responsible for poor visibility and causing rail, road and flight accidents (Saxena et al. 2020; Saxena and Kulshrestha 2015). In later run, particularly during winter season, extensively increased concentrations of NOx, SOx, oxidants and particulate matter along with favourable meteorological conditions like low boundary layer, low temperature, temperature inversions and low wind speed give rise to the condition "Smog" (Sonwani and Saxena 2021). Smog is very much prevalent and recorded since historic times like famous episodes of "London Smog" and "Los Angeles Smog". London Smog is one of the most deadly smog recorded so far that killed around 12,000 people due to emissions from thermal power plants (Tiwari and Saxena 2021). Presently, India witnesses smog events every winter almost every year and in 2016, Delhi's smog was declared as "The Great Delhi Smog". Schools were closed and people were advised to stay indoors during high pollution days (Bhalla et al. 2019). Smoke from rural kitchens, vehicular emission, industries, more numbers of vehicles, jeopardization of non-polluting public transports, dense population, low investment in public transport and insufficiency of public infrastructure, extensive construction activities, and crop residue burning issues from adjacent states such as Punjab, Haryana and Uttar Pradesh contribute together. During the months of post-monsoon and winter, approximately 500 million tons of crop residues are burned in Indo-Gangetic plains leading to mass emissions from biomass burning (Sharma et al. 2010; Saxena et al. 2021). According to another study, 72% of the total air pollution load in Delhi is from vehicular emissions as calculated using emission factor and activity-based approach suggested by Intergovernmental Panel on Climate Change (IPCC) (Sindhwani and Goyal 2014). Furthermore, during winters calm winds and low temperatures entrap dust particles and pollutants near the ground setting up smog layer. And then the festival of Diwali, due to fireworks, gives an incremental effect to the air pollution of Delhi. Another atmospheric extreme event is volcanic eruption. On the basis of type and region of eruption, volcanic events can become extreme events in particular regions of the Earth. Emission of tropospheric aerosols from quiescent volcano will have a lifetime of weeks; however, an explosive eruption injecting the aerosols in the stratosphere can scale up the lifetime from weeks to years. The resulting impact on radiative forcing will hence be much more significant. The extremity of an eruption impact is not only visible in the amount of material that is ejected but also by the distance the impact propagates. Even smaller eruptions can have a long-range impact under favourable meteorological conditions. Recently in 2018, Manaro Voui volcano on Ambae island of South Pacific injected a record amount of 400,000 tons of SO₂ in the UTLS region, which was thrice the amount of SO₂ released from all volcanoes in the previous year (NASA Report 2018). The impacts ranged from destruction of homes. burying of crops, blackening of the sky by volcanic ash, acidifying the drinking water and forcing the entire population of the island to evacuate. Likewise, there are various extreme events which occurred due to disturbance in seasonal cycles and atmospheric phenomena that ultimately causes ecological imbalance and severe changes in air quality.

Atmospheric extremes and air pollution episodes are considered as a form of disaster and therefore disaster risk management plan is very important to mitigate such events. Disaster management authorities that can strategize pre-disaster, immediate post-disaster and recovery steps taken during a disaster are a necessity. Disasters involving extreme air pollution are causing an immediate public health crisis leading to an increased pressure on stakeholders and agencies to provide solutions to protect impacted communities during any such emergencies (Chandrappa and Kulshrestha 2016). During any disaster, the air quality management is dependent on the local community preparedness and external aid.

1.2 Summary of Chapters

A number of points discussed above have clearly depicted that extremities in atmospheric processes, phenomena and resulted degradation in air quality are important issues that need to be given attention and should be taken into serious consideration. This book "Extremes in Atmospheric Processes and Phenomena: Assessment, Impacts and Mitigation" covers a general overview of factors governing in atmosphere to lead to air pollution, description about recent and hazardous air pollution episodes, emergencies and extremes in atmospheric sciences, impact studies on living organisms and atmosphere related to emergencies and possible remedies/mitigation strategies which may also include green growth strategies for management. Chapter 2 provides a compact overview of atmospheric phenomena and their mechanism; and how they will play a significant role as both positive and negative effects on human livelihoods and human well-being. Moreover, it also provides the effects of human activities on the mechanism of

atmospheric phenomena. Chapter 3 deals with the major atmospheric pollutants (sulphur dioxide, oxides of nitrogen, ground level ozone, suspended particulate matter, fluorides and peroxyacyl nitrates), their impacts on atmosphere (global warming, stratospheric ozone depletion, acid rain, etc.) and threats to the biota health. This chapter has given a fundamental understanding about air pollution, air pollutants, reasons of their impacts on atmosphere and living organisms health. Chapter 4 focuses on South Asian monsoon extremes, intra-seasonal variations and modulation by remote forcing and their future trends in global warming scenario. Moreover, it also summarizes that how good quality observational data sets can help future climate management strategies for land-based water resources, groundwater, agricultural best practices, food security, disaster early warning and preparedness. Chapter 5 presents a brief review on the contribution of fog in changing air quality, its extremities and risks to environment and society. It also focuses on fog types and distribution, physio-chemical characterization, factors for its formation and evolution and extremities of this fog pollution phenomenon on environment and society. Chapter 6 deals with the concept to understand nature and phenomena of dust and sandstorms (SDS) processes, current and past trends of SDS impacts, as well as understanding impact of SDS hazard on environment and human health as well as extremities and their correlation with climate change. Chapter 7 focuses primarily on heat waves (HW) addressing with details their definition, generation, future projection and synergy with the urban heat island (UHI) effect and air pollution in cities, as well as their harsh impacts on several sectors and the approaches for mitigating these effects. A part of this chapter is also devoted to cold waves, while a case study concerning the inter-play between extreme temperatures, thermal comfort and air pollution during a significant HW episode that occurred in Greece is presented. Chapter 8 deals with the status of excessive biomass burning emissions over the Indo-Gangetic Plain (IGP), and their impacts. It also describes the sources of biomass burning with depiction of intense biomass burning, illustrates extreme biomass burning episodes and provides a description of the elevated levels of trace gases and aerosols generated as a result of biomass burning, followed by their impacts on environment. Chapter 9 gives an overview of the challenges behind the increasing trend of smog events across Northern India. It also seeks to deliver an understanding of the question of why the northern part of our country is becoming the pollution centre of the world and intends to link the plausible causes and impacts for the rising extreme events of smog. Chapter 10 focuses on different types of volcanoes and their atmospheric impacts. It also aims to simplify and summarize the information available regarding volcanic eruptions from an atmospheric and climate perspective and ends with the case study of the most famous volcano the Mt. Pinatubo of 1991. Chapter 11 deals with the extremities caused due to emission of high concentrations of air pollutants like trace gases and particulate matter during fireworks and how they affect the air quality by taking the support of a case study of Vishu festival celebrated in Kannur, a Coastal city of Southern India. Chapter 12 focuses on an overview of the history of air pollution episodes and their impacts, key driving mechanisms, followed by an outlook including some perspectives on the control measures. Chapter 13 outlines various disaster mitigation and adaptation strategies. Importance of science and decision support systems, data availability and sharing, early warning systems, timely information dissemination and post-disaster management are also highlighted. Chapter 14 aims to examine and assess the various components of air pollution, their effects on human health and way forward to deal with air pollution as a disaster and public health emergency with existing policies and programmes. The identified framework will be useful to all stakeholders and decision makers as a roadmap while designing the future air pollution mitigation policies from a disaster management point of view. Chapter 15 aims to discuss common technologies for air pollution control and measurement. The discussion about commonly used technologies for air pollution management and their costeffectiveness has been highlighted in this chapter. Chapter 16 talks about air pollution and its extremes, how anthropogenic actions have lead air pollution to a disaster and how the air pollution and its extremes can be mitigated through ecological and natural-based solution as green growth strategies. The aim of the chapter is to use the approach of nature-based solutions that uses ecological and environment-friendly methods in urban planning and management to mitigating air pollution. Chapter 17 reviews all the existing international and national policies that outline measures to curtail and abate ambient air pollution in chronic and episodic events as well as targeting natural and anthropogenic sources. It also gives details of the international policies that have been implemented to address the transboundary pollution with global implications and also the policies adopted in India in response to abatement of deteriorating air quality at the state and national level. This book ends up with Chapter 18 that discusses different approaches with national and international examples to understand the effectiveness of participatory model in relation to atmospheric processes and extremities. Although participation of different stakeholders in decision making process and in managing the extremities is not straightforward process. It is discussed that role of training and decision on appropriate level of involvement of people is also important for further improvement and governance of extremities in atmospheric processes and phenomena.

1.3 Conclusions

It is important to build significant relationship between extremes in atmospheric processes, phenomena and resulted air quality degradation. This is a vital step toward analysing the pattern of air quality extremes in changing climate. For instance, high concentrations of ozone and fine particulate matter are common air pollutants responsible for causing extremes in atmosphere and major factors behind air pollution episodes. Therefore, more systematic studies should be conducted to understand the factors behind the disturbances held in various atmospheric processes and that ultimately lead to extremes in weather and climate. Such extremities are further responsible for health hazards and often leads to epidemic outbreaks too. Moreover, the extreme events like heat waves, cold waves, dust storms, smog, biomass burning, etc. can cause deleterious impacts on plants, animals, human and even material health. High mortality and premature deaths are reported every year on

worldwide basis and stringent disaster risk management methods and green technologies can help to reduce and mitigate extreme events in atmosphere and related air pollution episodes. Thus, the different chapters in this book explain about factors governing in atmosphere to lead to air pollution, description about recent and hazardous air pollution episodes, emergencies and extremes in atmospheric sciences, impact studies on living organisms and atmosphere related to emergencies and possible remedies/mitigation strategies which may also include green growth strategies for management.

References

- Bell ML, McDermott A, Zeger SL, Samet JM, Dominici F (2004) Ozone and short-term mortality in 95 US urban communities, 1987-2000. J Am Med Assoc 292:2372–2378
- Bhalla N, O'Boyle J, Haun D (2019) Who is responsible for Delhi air pollution? Indian newspapers' framing of causes and solutions. Int J Commun 13:24
- Blanken PD, Brunet D, Dominguez C, Goursaud S, Hussain S, Jain M, Koren G, Mu Y, Ray P, Saxena P, Sonwani S (2022) Atmospheric sciences perspectives on Integrated, Coordinated, Open, Networked (ICON) science. Earth Space Sci 9:1–7
- Chandrappa R, Kulshrestha UC (2016) Air pollution and disasters. In: Sustainable air pollution management. Springer, Cham, pp 325–343
- Goel A, Saxena P, Sonwani S, Rathi S, Srivastava A, Bharti AK, Jain S, Singh S, Shukla A, Srivastava A (2021) Health benefits due to reduction in respirable particulates during COVID-19 lockdown in India. Aerosol Air Qual Res 21
- IPCC (2007) Climate change 2007: impacts, adaptation and vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, 976 p
- Jacob DJ, Winner DA (2009) Effect of climate change on air quality. Atmos Environ 43:51-63
- Lelieveld J, Evans JS, Fnais M, Giannadaki D, Pozzer A (2015) The contribution of outdoor air pollution sources to premature mortality on a global scale. Nature 525(7569):367–371
- NASA Report (2018). https://www.nasa.gov/feature/goddard/2019/2018-s-biggest-volcanic-erup tion-of-sulfur-dioxide. Accessed 8 May 2021
- Pant P, Harrison RM (2013) Estimation of the contribution of road traffic emissions to particulate matter concentrations from field measurements: a review. Atmos Environ 77:78–97
- Pope CA III, Ezzati M, Dockery DW (2009) Fine-particulate air pollution and life expectancy in the United States. N Engl J Med 360(4):376–386
- Saxena P, Kulshrestha UC (2015) Scavenging of urban air emissions by Fog at Delhi, India. In AGU Fall Meeting Abstracts, vol. 2015, pp A33H-0274, December
- Saxena P, Sonwani S (2019) Criteria air pollutants and their impact on environmental health. Springer, Singapore, 157 p
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Bhardwaj R, Ghosh C (2012) Status of air pollutants after implementation of CNG in Delhi. Curr World Environ 7(1):109
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8): 6639–6656
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973

- Sharma AR, Kharol SK, Badarinath KV, Singh D (2010) Impact of agriculture crop residue burning on atmospheric aerosol loading—a study over Punjab State, India. Ann Geophys 28(2):367–379
- Sindhwani R, Goyal P (2014) Assessment of traffic-generated gaseous and particulate matter emissions and trends over Delhi (2000–2010). Atmos Pollut Res 5(3):438–446
- Sonwani S, Saxena P (2016) Identifying the sources of primary air pollutants and their impact on environmental health: a review. IJETR 6(2):111–130
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sonwani S, Saxena P (eds) (2022) Greenhouse gases: sources, sinks and mitigation. Springer-Nature, Singapore, pp 1–255
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3(690444):1–20
- Tiwari S, Saxena P (2021) Air pollution and its complications. Springer-Nature, Singapore, pp 1–178
- WHO Fact Sheet (2018) Ambient (outdoor) air pollution, key facts. https://www.who.int/en/news-room/fact-sheets/detail/ambient-(outdoor)-air-quality-and-health



2

Atmospheric Phenomena: Origin, Mechanism, and Impacts

Vanisa Surapipith and Pornpan Uttamang

Abstract

Haze, fogs, wind-blown dust, and tornadoes are well-known examples of atmospheric phenomena. These atmospheric occurrences are normally observed around the world; however, some phenomena may be frequently observed over specific regions than others. For example, the United States (the U.S.) has the most experience in tornado strikes than any nation worldwide. The U.S. receives approximately 1200 tornadoes per year while this phenomenon annually occurs over New Zealand with about 20 tornadoes (the National Severe Storms Laboratory (NSSL), Severe weather 101-Tornadoes, 2021). When the strengths and intensities of these phenomena are raised, they become more and more severe; eventually, they turn to be natural hazards that threaten properties, economy, environment, and lives. In May 2011, Joplin, a destructive tornado, hit the city of Joplin, Missouri, USA that caused 161 fatalities and more than 1000 injuries; about 553 business structures and 7500 residential structures were destroyed. Economic loss of the city of Joplin after the devastating Joplin tornado stroke was estimated to be \$2.8 billion (Houston et al., PLoS Curr 7:ecurrents. dis.18ca227647291525ce3415bec1406aa5, 2015; the National Institute of Standards and Technology (NIST), Joplin Missouri Tornado 2011, 2021).

The aim of this chapter is to demonstrate important atmospheric phenomena, in particular, hydrometeor and lithometeor, and their mechanism, since these atmospheric phenomena play a significant role in human livelihoods and human well-being with both positive and negative effects. On the other hand,

V. Surapipith

National Astronomical Research Institute (NARIT), Chiang Mai, Thailand

P. Uttamang (🖂)

Environmental Technology Program, Faculty of Science, Maejo University, Chiang Mai, Thailand

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_2

human activities such as urbanization, industrialization, and intensive agriculture may accelerate the formation of these natural events and strengthen their intensities that may ultimately result in an increase in more frequent and more severe natural hazards. Therefore, in this chapter, effects of human activities on the mechanism of atmospheric phenomena are addressed and discussed. In the last session, we focus on the impacts of hydrometeors and lithometeor on human society in order to emphasize human-environment interactions, to raise awareness of environmental protection to societies which atmospheric phenomena should be integrated into environmental management. After all, the chapter illustrates how the environmental mechanisms enhanced by human activities are having tremendous impact on the human society ourselves.

Keywords

 $Atmospheric \ phenomena \ \cdot \ Atmospheric \ phenomena \ mechanism \ \cdot \ Hydrometeor \ \cdot \ Lithometeor \ \cdot \ Effects \ of \ atmospheric \ phenomena$

2.1 Introduction

"Atmospheric Phenomenon" is an observable physical occurrence in the atmosphere which opposed to dynamic or synoptic phenomena (American meteorological society (AMS) 2012a). Dynamic meteorology is usually referred to a meteorology study that focuses on fluid mechanism and thermodynamics of atmosphere since the atmosphere is considered as a continuous fluid medium (Holton 2004). Synoptic phenomena are atmospheric occurrences that occur in a range scale of 500–10,000 km such as cyclones (Jacobson 2005; AMS 2012b). Therefore, the atmospheric phenomena include all hydrometeors, lithometeors, igneous meteors (for example, lightning), and luminous meteors or optical meteors (for example, rainbow); however, local-scale or large-scale occurrences of winds, pressure, temperature, and clouds are excluded (AMS 2012a, c, d).

Among the atmospheric phenomena mentioned above hydrometeors and lithometeors are two major atmospheric phenomena with so far greatest impacts on human society. Hydrometeors are various forms of water in the atmosphere that produced through condensation and deposition (AMS 2012c); while lithometeors are small particles that can be wet or dry particles suspended in the atmosphere (AMS 2012d).

Significantly at present, intensity of the atmospheric phenomena is changing, and this is affecting the society. Learning about the origin including mechanism of the formation and process is therefore highly important. Understanding on the transformation of the hydrometeors and lithometeors till they reach the receptors, whether human or the environment, will provide the possibility to prepare and adapt to any negative impacts of the phenomena. As found through the literature reviews, for example, vast area from Eastern Asia to Middle-central Europe recently experienced the impacts of carbonaceous lithometeors, so-called radiation-absorbing aerosols, on enhancing the melt of snow cover and ice sheet, and cause the retreat of glaciers, as observed by decadal space-based active and passive measurements (Wang et al. 2020a, b; Sonwani and Saxena 2021). The study showed that the Pan Third Pole was mainly affected by natural dust, polluted dust, and elevated smoke causing detrimental environmental and climate change. Moreover, in Indo Gangetic Plain (IGP) annually there occur persistent fog events during the winter time (Saxena et al. 2020; Saxena and Kulshrestha 2016). The fog-affected area extends throughout the IGP ranging from Pakistan, in the west, to Bangladesh, in the east, covering about 3000 km in length and 400 km wide taking the plain area of Northern India and southern Nepal (Gautam et al. 2007; Saxena and Kulshrestha 2015). Studies based on meteorological data over two decades indicated the increased number of foggy days during winter (Jenamani 2007; Shrestha et al. 2018; Syed et al. 2012), while the IGP is one of the world's densely populated regions with the significant fertile land for agricultural activities and numerous operational industries which generate pollutants emitted into the atmosphere mixing with the fog. It was found by Izhar et al. (2019) that the mechanism of fog formation involved both the long-range and regional contributions, indicated by the presence of inorganic species like nitrate and sulfate versus dissolved organic carbon (DOC) as contents in the fog water droplets. The observed sulfate concentration in the IGP fog water highlighted the aqueous processing of gaseous SO_2 into fog water droplets. The clustering analysis results showed that locally driven air masses were enriched with the DOC, which is essentially influenced by biogenic emissions from the intensive agricultural land, cooking and heating activities, brick kilns, and resuspension of unpaved roads in the region.

In this chapter, we are focusing on the formation of hydrometeors and lithometeors in the atmosphere. The effects of biomass burning, urbanization, and industrialization on the hydrometeors and lithometeors formations are discussed to point out the impacts of human activities on enhancing frequencies and severities of these atmospheric phenomena. Finally, hydrometeors and lithometeor are discussed as natural hazard events that threaten human society, economy, and health in order to emphasize why it is important to care about the environment and natural phenomena.

2.2 Atmospheric Phenomena and Mechanism

As mentioned earlier, atmospheric phenomenon is an observable physical natural event including hydrometeors, lithometeors, igneous meteors, and luminous meteors. Hydrometeors basically can be classified into many types, but well-known hydrometeors are clouds, fog, rain, snow, hail, dew, rime, glaze, blowing snow, and blowing spray. The term hydrometeors encompasses water and ice particles suspended, or falling, in the air as well as those formed at the surface, such as dew. A precipitation particle is a type of hydrometeor. More details of their formation mechanism are explained in this section.

2.2.1 Hydrometeor

"Hydrometeors" refers to water in liquid or solid phases suspended or falling through the atmosphere (World Meteorological Organization (WMO) 2017a). Water particles that are blown from the Earth's surface are also included in the term hydrometeors (AMS 2012c). Hydrometeors can be classified into five categories including:

- Suspended particles: liquid or solid suspended water particles in the atmosphere.
- Precipitation: liquid or solid falling water particles.
- Wind-blown particles from the Earth's surfaces: sea sprays.
- Deposits of particles: dews, frozen dews, and frost.
- Spouts: rotating columns of air that contain water droplets.

In general, snow or water on the ground are not included as hydrometeor (WMO 2017a); therefore, in the Hydrometeors section, we put more focus on the formation of suspended particles, precipitation, and spouts.

2.2.1.1 Suspended Particle

"Fog" and "mist" are well-known examples of suspended particle. They are water particles that can be in liquid phase or solid phase (ice crystal particles) or both that suspend in the atmospheric boundary layer. These suspended particles are the results of condensation process (Gultepe et al. 2007; Ahrens and Henson 2014) which is the process that atmospheric water vapor is changed to liquid or solid water particles. Except relative humidity, condensation nuclei are one of the most important factors that drive condensation processes in the atmosphere (Pui et al. 2014). Condensation nuclei are extreme light and tiny particles in the atmosphere where their surfaces serve as water vapor condensation surfaces. Condensation nuclei can be dust, smoke, sea salt particles, and sulfate particles with various particle sizes (range from less than 0.1 μ m to greater than 1 μ m in radius) and number concentrations (range from 10,000 to less than 1 particle per cubic centimeter air) (Ackerman and Knox 2003; Ahrens and Henson 2014). The most common condensation nuclei mode considered by number of particles per cubic centimeter air is "Aitkin condensation nuclei". Aitkin condensation nuclei are particles with radius less than 0.1 µm and the number concentration of this particle mode typically ranges from 1000 to 10,000 particles per cubic centimeter air. Particles within the radius range of 0.1-1.0 µm are considered as "large condensation nuclei" whose number concentrations normally range from 1 to 1000 particles a cubic centimeter of air. "Giant condensation nuclei" are referred to particles with mean radius larger than 1.0 µm (Ahrens and Henson 2014).

The condensation nuclei can be characterized into two groups (Fig. 2.1). Hygroscopic particles or water-seeking particles are particles that allowed water vapor to condense on their surfaces; even the atmospheric humidity (RH) is less than 100%, a condensation process can take place. Hydrophobic particles or water-repelling, on the other hand, are particles that resist water vapor condensation on the particle surfaces. Without hygroscopic particles, very high atmospheric relative humidity



Fig. 2.1 Hygroscopic nuclei and hydrophobic nuclei (modified from: University of Arizona 2018a)



Fig. 2.2 The formation of radiation fog (modified from: Midwestern Regional Climate Center 2021)

(may greater than 100%) may be required in order to induce condensation (Ahrens and Henson 2014).

When the hygroscopic nuclei have water condensation on their surfaces and generate small droplets, fog and mist occur. Even fog and mist have a similar formation process, the major difference between them is horizontal visibility reductions. In general, fog reduces horizontal visibility in the range of less than 1 km, but mist has horizontal visibility range from 1 to 5 km (WMO 1966). "*Heavy fog*" might be used when the horizontal visibility is less than about 200 m (Van Oldenborgh et al. 2010).

Fog can be characterized into several types depending on fog formation processes and place of occurrences; however, in this chapter, we are more focused on types of fog based on the formation processes.

"*Radiation fogs*" are formed through the radiation cooling process of the Earth's surface (Fig. 2.2). This type of fog usually is formed during nighttime with cloudless. This condition promotes the escape of longwave radiation emitted by earth to space (Aguado and Burt 2007). Due to the nighttime radiational cooling process, moist air masses become saturated and radiation fogs are formed. Since radiation fogs are generated near the ground, it is also known as "ground fog" (Ahrens and Henson 2014).



Fig. 2.3 The formation of advection fog (modified from: Midwestern Regional Climate Center 2021)



Fig. 2.4 The formation of upslope fog (modified from: Northern Vermont University n.d.)

"Advection fogs" are formed when moist air travels over a cooler surface. The moist air is being cooled due to the heat transfer process from the moist air to the surface. When the moist air temperature reaches its dew point, advection fog is formed (Aguado and Burt 2007; Ahrens and Henson 2014). Advection fogs can be found near coastal areas when warmer air over the sea moves toward the cooler land (Bergot and Guedalia 1994; Ahrens and Henson 2014). Figure 2.3 illustrates schematics diagrams of radiation fog and advection fog formations.

"Upslope fog" is a type of fog that is formed through the adiabatic cooling process of air. Upslope fog is formed when warm and moist air travels along an elevated slope such as hill or mountain. While the air is rising, its temperature drops owing to its expansion, then upslope fog occurs (Fig. 2.4) (Ackerman and Knox 2003; Aguado and Burt 2007; Ahrens and Henson 2014).

By now, one should be apparent that the formation of suspended particles (i.e., fog and mist) is closely related to relative humidity conditions in which suspended particles can grow faster under high relative humidity levels and number concentration of condensation nuclei. Another important factor is windspeed (Luan et al. 2018; Deng et al. 2008; Wang et al. 2019). Light windspeeds (about 5 km h⁻¹) promote condensation and fog formation, but stronger windspeeds (greater than 5 km h⁻¹) stir warm air downward to the surface, which resist cooling processes in the atmosphere (Aguado and Burt 2007; Ahrens and Henson 2014).

Human activities also accelerate the formation of suspended particles and their lifetime. Increases in local emissions and regional transport lead to increases in the suspended particulate levels (Wang et al. 2019; Sonwani et al. 2021). Changes in the characteristic of urban and rural areas can show direct and indirect impacts on suspended particles lifetime (Saxena and Sonwani 2019a). Generally, the lifetime of suspended particles is strongly influenced by canopy layer, vegetation, land use, and soil (Saxena and Sonwani 2019b). For example, fog is unlikely to occur in urban areas because the air has low relative humidity and high temperature owing to the high proportion of impermeable objects that prevent evaporation. However, the number concentration of aerosols over polluted areas are higher than these over rural areas. These aerosols can bind more water vapor, results in thicker and longer life fog once it forms in urbans (Tiwari et al. 2011; Stolaki et al. 2015; Van Oldenborgh et al. 2010). In China, increases in energy consumption, industrialization, and urbanization are considered as the major sources of increases in suspended particles, haze, and PM_{25} concentrations, resulting in the visibility reduction in China with the rate of 2.1 km per decade (Che et al. 2009). Recent studies showed that amines emitted from vehicles, waste treatment plants, agriculture, animal farming, ocean, biomass burning, vegetation, and soil play an important role in secondary organic aerosol (SOA) formation in the atmosphere (Malloy et al. 2009; Murphy et al. 2007; Price et al. 2014, 2016; Tang et al. 2013; Yu et al. 2012). Amines can form salt particles via chemical reaction with a presence of gaseous nitric acid (HNO₃) and gaseous sulfuric acid (H_2SO_4) (Murphy et al. 2007). Besides salt particles, tertiary amines with no hydrogen atoms (H) on the nitrogen atom (N), such as trimethylamine (TMA) and triethylamine (TEA), have significantly contributed to non-salt SOA formation; which in general, they are more stable salt particles (Price et al. 2014, 2016; Tang et al. 2013; Murphy et al. 2007).

2.2.1.2 Precipitation

Precipitation is liquid or solid water particles that are falling from the atmosphere or cloud bases to the ground. "*Rain*", "*freezing rain*", "*sleet*", and "*snow*" are examples of precipitation. Rain is defined as liquid water falling drops that have diameter equal to or greater than 0.5 mm. Smaller drops are considered as drizzle (Ahrens and Henson 2014). If water droplets contact with thin cold air layer over the surface, water droplets become supercooled and freezing rain occurs. Similar to freezing rain with a thicker cold air layer over the surface, water droplets become frozen and turn back into tiny ice pellets which are called sleet. If the atmospheric temperature is below freezing from the cloud base to the surface, snow falls (Ackerman and Knox 2003; Ahrens and Henson 2014; NOAA 2013). Figure 2.5 shows photos of freezing rain, sleet, and snow, and Fig. 2.6 illustrates the formation of rain, freezing rain, sleet, and snow.

There are two important processes involved in the precipitation formation including droplet growth processes and precipitation forms (University of California, Irvine (UCI) 2020). Since water droplets have to be large enough to overcome the updraft and fall to the Earth's surface, growth processes are necessary. Droplets can



Fig. 2.5 Photo of freezing rain, sleet, and snow (NOAA 2015)



Fig. 2.6 Rain, freezing rain, sleet, and snow formation (modified from: NOAA 2013)

enhance their sizes through several processes, such as condensation, collision-coalescences, and ice-crystal (Bergeron) processes.

Among these growth processes, the collision-coalescences process plays an important role in increasing the size of water droplets in warm clouds in tropical regions (Ackerman and Knox 2003; Ahrens and Henson 2014). By droplet-droplet collision, collision-coalescences process can generate large water droplets with fast speed (Ackerman and Knox 2003). Condensation process is a slow process that takes time to create large water droplets. Ice-crystal process only occurs in cold clouds with ice particles that serve as ice-forming nuclei for water vapor deposition (Ackerman and Knox 2003; Ahrens and Henson 2014).

Effects of human activity on precipitation formations have been reported in several studies. Over polluted areas, precipitation may be suppressed. A study of Givati and Rosenfeld (2004) revealed a strong evidence of precipitation downwind of pollution sources was suppressed by 15–25% resulted in water losses in those areas ranging from 15 to 25% compared with the annual average precipitation in California, the USA, and in Israel. A similar effect of air pollution on precipitation formation was shown by Jirak and Cotton (2006). Their study showed that upslope precipitation over western Denver and Colorado Springs, Colorado, USA had a decreasing trend (about 30% compared to upwind urban sites) during the twentieth

century; while the concentration of pollutants increased. Air pollution can suppress precipitation formation by creating a narrow spectrum of small droplet size that limits the collision-coalescence process. A study of Borys et al. (2003) showed that anthropogenic aerosols created tiny cloud droplets that retarded snow particle growth process, resulting in a decrease in snow fall over the northern Rocky Mountains of Colorado, Colorado, USA.

2.2.1.3 Spout

Spout is identified as a phenomenon consisting of rotating columns of air extending from the base of a cumuliform cloud. The rotating air columns consist of water droplets from the sea or dust particles over land (AMS 2012e). A *"Tornadoes"* over the sea (Fig. 2.7) is a good example of a waterspout. Generally, tornadoes form from thunderstorms with an unstable atmospheric condition or form with supercell thunderstorm (called supercell tornadoes). Tornadogenesis or tornado formation can be classified into three steps (Danielson et al. 2003):

- Mesocyclone formation
- Wall cloud formation
- · Tornado formation

A mesocyclone is a large air vortex that occurs from a slow and horizontal rotation of a large segment of the cloud (Aguado and Burt 2007). The rotational speed of the mesocyclone increases when it is reshaped to be slim and extended vertically toward the Earth's surface. During this transitional stage, a wall cloud may be observed below the cloud base of the mesocyclone. Since the mesocyclone is a low-pressure area, then air moves rapidly toward the mesocyclone and increases the acceleration. If the circulation of the mesocyclone reaches the Earth's surface, a tornado is formed (Ahrens and Henson 2014; Danielson et al. 2003).



Fig. 2.7 Waterspouts in the Florida Keys (NOAA 2018). Photographer: Dr. Joseph Golden



Fig. 2.8 Temporal distribution of Tornado count over the U.S. (a) from 1990 to 2019, (b) monthly average, and (c) daily average (NOAA 2019)

Tornado may occur more frequently over some Earth's regions, such as the U.S., than other regions. Regarding NOAA tornado's report, the number of tornadoes that occurred over the U.S. was increasing from 1995 to 2019 (Fig. 2.8a) with the annual average tornado count from 1950 to 2010 being about 1251 tornadoes (NOAA 2019). Tornadoes can occur at any month of the year and any time of the day, but most of them usually occur during May and June (Fig. 2.8b) and they can be found in the afternoon and evening more than in the morning (Fig. 2.8c).

Possible linkages between urbanization and tornadoes were reported in several studies. Molina and Allen (2020) revealed that, in regards to climate change, severe thunderstorms might occur more frequently in the future over Continental United States (CONUS). The study showed that moisture content and moisture flux had been increasing continuously since 1980s. The increases in moisture factors would lead to lower convective bases and larger vertical instability and eventually resulted

in high frequency of severe thunderstorms. Saide et al. (2015) studied the effects of biomass burning smoke on the increases in tornado severity over the U.S. The study showed that by accompanying biomass burning smoke, more severe tornadoes were more likely developed over the Southeast and central U.S. During the smoke events, lower cloud bases and stronger low-level wind shear were conducted by smoke and soot resulted in stronger tornado intensity.

2.2.2 Lithometeor

"Lithometeors" are similar to hydrometeors, but lithometeors are suspended of dry particles and non-aqueous particles (AMS 2012d; WMO 2017b). Well-known examples of lithometeors are haze and lifting or blowing dust (Fig. 2.9) (WMO 2017b).

2.2.2.1 Haze

"Haze" refers to a layer of tiny dust or salt particles with diameter of about or less than 0.1 μ m that are suspended in the atmosphere and it occurs when relative humidity is low. Horizontal visibility can be reduced by haze to less than 10 km through the light scattering process (Zhu and Wan 2019).

Haze can be categorized into two categories including dry haze and wet haze. Particles of dry haze are extremely small; thus, they can scatter short wavelengths of light (AMS 2012f). Due to their light-scattering ability, haze reveals bluish colors when one views against a dark background. When haze is viewed against a lighter background, it appears yellowish tint colors (AMS 2012e; Ahrens and Henson 2014). Wet haze usually occurs during nighttime to early morning due to an increase in relative humidity. When the atmosphere relative humidity reaches about 75%, water vapors collect on the dry haze particles, the size of the particles increases, and



Fig. 2.9 A hazy morning during a $PM_{2.5}$ episode across Mae Salong Mountain, Chiang Rai, Thailand and a dust devil at Arizona desert, Arizona, USA (NASA 2007)



Fig. 2.10 The formation of dry haze with low relative humidity and wet haze with higher relative humidity (modified from: University of Arizona 2018b)

wet haze forms. Since the size of wet haze particles is larger than the dry haze, light is more scattered by wet haze (University of Arizona (UA) 2018). Figure 2.10 illustrates the formation of dry haze and wet haze.

Effects of human activities on haze formation are reported in many studies. Zhu and Wan (2019) reported that serious haze events that occurred in Harbin were associated with the local straw burning that generated a large amount of particles. Increasing fuel consumption also enhances the severity of haze events. Li et al. (2017) reported that more severe haze phenomena that occurred in China were a result of increases in national fossil fuel consumption that led to enhance $PM_{2.5}$ sulfate levels. Ye et al. (2011) investigated the effects of ammonia on haze formation in Shanghai, China which the study showed that ammonia from the agriculture sector might be a major source of $PM_{2.5}$ nitrate particles ((NH₄)₂SO₄) in Shanghai.

More pronounced impacts of haze are found in South and Southeast Asia. Very recently, Saxena et al. (2021) confirm that the haze event in conjunction with crop residue burning (CRB) over northern India has been a major air quality and human health issue. They found that CRB activities in Haryana had been affecting the air quality of Delhi, the capital region of India, with a considerable increase in pollutant concentrations during the transition from pre-burning to burning period. PM10 and PM2.5concentrations exceeded National Ambient Air Quality Standard (NAAQS) limits by 2–3 times. Clearly meteorological conditions influence pollutant concentrations during both seasons, while the conditions were exacerbated by dust storms (rabi) in the upwind city called Dusshera, as well as the firework during Diwali celebrations in kharif season.

In the Southeast Asian region, Chantara et al. (2012) describe how haze has also become an important environmental issue in the past decade, due to its impacts on both the economy and human health. They found through the daily samples of PM2.5 collected during the dry season (March–April) in the years 2017 and 2018 in Chiang Mai and Nan Provinces in Northern Thailand that the forest fires during the dry season, and agricultural residue burning regularly impacted the levels of PM2.5 concentrations. Comparisons made during La Niña years (which are usually less affected by open burning), i.e., in 2012 indicated a drop in PM2.5 values could be influenced by the implementation of the zero-burning policy that was enforced for a period of about 60 days in an attempt to control open burning practices in upper
Northern Thailand. Interestingly, nevertheless, the zero-burning policy has also contributed to prolonging the smoke haze situation from a 2-month period (mid February till mid April) to a 3-month-long period (mid February till mid May). Moreover, there are health risks from the inhalation of PM2.5 bound PAHs found in the study. Therefore, not only fine particulate matters are concerned, but their chemical compositions, particularly the levels of carcinogenic compounds that are bound to the PMs, are also posing significantly adverse health effects to the exposed population (Saxena and Sonwani 2019c).

2.2.2.2 Blowing Dust

"Dust devil" is a small column of air in a cylindrical shape that contains dust or sand. The size of dust devils ranges from 1 to 1000 m tall and less than 50 m in horizontal diameter at ground level. In general, dust devils are considered as a small-scale phenomenon (Luan et al. 2017). Dust devils usually occur under weak wind and sunny conditions when the hot air mass over ground level rises and generates a thermal convection whirlwind with a low-pressure core. This whirlwind has to have sufficient strength to lift and carry dust, sand, or particles (Fig. 2.11) (Luan et al. 2017; Onishchenko et al. 2019; Rafkin et al. 2016). Since dust devils are small-scale with short lifetime, normally, they are not dangerous.

2.3 Impacts of Hydrometeors and Lithometeors on Human Society

Fogs, mists, precipitation, tornadoes, and dust devils are natural phenomena; however, they may have negative impacts on human society. From the previous section, it should be apparent that visibility reduction is a major effect of suspended particles and haze on environment.

This visibility reduction influences on transportation, travel business that eventually lead to economic losses (Gultepe et al. 2007). Anaman and Looi (2000) estimated economic loss due to haze episodes that occurred over Brunei Darussalam during 1997–1998. The study showed that, by using ordinary least squares, the number of tourists to Brunei Darussalam decreased about 3.75% (compared with the tourist number in 1999) and direct economic loss was about B\$1 million due to haze



Fig. 2.11 The formation of dust devils (modified from: Gabbert 2019)

pollutions. More recent study showed a similar negative impact of haze events on tourism industry. A study of Wang et al. (2020a, b) revealed that the number of tourist arrivals to Beijing, China decreased about 5.22 million tourists owing to moderate to severe haze events, resulting in 8.95 billion yuan in Beijing's economic loss during 2016–2018. Even the number of foreign tourists decreased owing to haze events, the haze events have no significant impact on domestic tourists (Sun et al. 2019).

Negative impacts of suspended particle and haze on human health have been reported in many studies. Adverse health effects of fine particulate matters are lung irritation, respiratory diseases, cardiovascular diseases, cardiopulmonary disorders, morbidity, and mortality (Saxena and Srivastava 2020). Human respiratory system can be susceptible from fine particle's components such as free radicals, metal, and organic compounds, resulting in an inflammation of respiratory system, cell and DNA damages, impairment of human immune system, cardiac autonomic nervous system, and antioxidant system (Feng et al. 2016; Xing et al. 2016). Hanafi et al. (2019) investigated the negative effects of haze event on human health and economic losses in Pasir Gudang and Larkin, Malaysia based on observations during 2014–2016. Their study revealed that the number of morbidity cases for children (age < 12 years old) and adults (age > 13 years old) increased during haze episodes compared with those during non-haze events, resulting in economic losses of roughly RM83,233 and RM107,486 per year in Pasir Gudang and Lakin, respectively. In China, particulate matter is suspected as a major cause of respiratory diseases, lung disease, and premature death (Hou et al. 2019; Matus et al. 2012; Zhu and Wan 2019). A study of Matus et al. (2012) showed that high haze pollution was a substantial China's economic burden due to health damages which decreased gross domestic product in 1995 US\$64 billion.

Fine particulate matter also shows negative impacts on agricultural sectors. Very high concentrations of fine particulate matter reduce wheat and corn yields. Corn yield decreases by 0.5% per unit area while PM_{2.5} level increases about 1% (Zhou et al. 2017).

Thick haze plays an important role in controlling climate and pollution dispersion. Incoming solar radiation is attenuated due to the thick haze, resulting in lower surface temperature, lower heat flux, and shallow boundary layer that limit pollution dispersion. Conversely, precipitation may increase owing to more nuclei under hazy condition that can bind more water vapor (Kokkonen et al. 2019).

Tornado threats over the U.S. are obvious examples of the effects of spout on human society. Tornadoes are one of the major natural disasters in the U.S. that cause the third highest number of deaths (after floods and lightning) and it is the top three causes of economic losses owning to natural damage (after floods and hurricanes) (Boruff et al. 2003). During April 25–28, 2011, an extreme severe tornado event occurred over the U.S. About 300 tornadoes hit the U.S. from the Midwest through the Mid-Atlantic that caused about \$9 billion economic loss and nearly 360 fatalities during the event (NOAA 2021; Simmons et al. 2012). Patterns of neighborhood change and relocation are associated with natural hazards. Raker (2020) studied demographic changes after natural disasters (i.e., severe thunderstorms and tornadoes) across the U.S. The study implied that after natural

disasters, financially advantaged residences reconstructed the built environment rather than relocated which financially disadvantaged residences did (Raker 2020).

2.4 Conclusion

Learning about origin, mechanism, and impact of atmospheric phenomena is the first key to know how to make prediction and anticipate the effect of the phenomena. The learned society may be able to avoid loss in lives and negative consequences of experiencing the phenomena. Hydrometeors including suspended particle, precipitation and spout, and lithometeors such as haze and lifting or blowing dust are atmospheric phenomena that their formation mechanisms are significantly affected by human activities. There are strong evidence that the increases in pollutant emissions due to fossil fuel consumption and biomass burning accelerate the formation of these atmospheric phenomena and strengthen their severity; eventually; they will show negative impacts on human society such as property damages, economic losses, and the toll of death and injuries. Research on finding the origin of the mechanism and impact of the atmospheric phenomena required more support, particularly in the many parts of Asia, where less resources are available. Joint research collaboration and knowledge transfer for building the capacity of the lessdeveloped country to monitor and collect data are crucial for the analysis on trends of the atmospheric changes. Without such effort, the society may not survive as a whole, despite the advance in technology serving other conveniences. In this respect, systematic study on the issue is highly encouraged.

References

- Ackerman AS, Knox AJ (2003) Meteorology. Brooks/Cole-Thomson Learning
- Aguado E, Burt JE (2007) Weather & climate, 4th edn. Pearson Education
- Ahrens CD, Henson R (2014) Meteorology today, 11th edn. Cengage Learning
- AMS (2012a) Atmospheric phenomenon. https://glossary.ametsoc.org/wiki/Atmospheric_phenom enon. Accessed Mar 2021
- AMS (2012b) Synoptic. https://glossary.ametsoc.org/wiki/Synoptic. Accessed Mar 2021
- AMS (2012c) Hydrometeor. https://glossary.ametsoc.org/wiki/Hydrometeor. Accessed Mar 2021
- AMS (2012d) Lithometeor. https://glossary.ametsoc.org/wiki/Lithometeor. Accessed Mar 2021
- AMS (2012e) Spout. https://cloudatlas.wmo.int/spout.html. Accessed Mar 2021
- AMS (2012f) Haze. https://glossary.ametsoc.org/wiki/Haze. Accessed Mar 2021
- Anaman KA, Looi CN (2000) Economic impact of haze-related air pollution on the tourism industry in Brunei Darussalam. Econ Anal Policy 30(2):133–143
- Bergot T, Guedalia D (1994) Numerical forecasting of radiation fog. Part I: numerical model and sensitivity tests. Mon Weather Rev 122(6):1218–1230
- Boruff BJ, Easoz JA, Jones SD, Landry HR, Mitchem JD, Cutter SL (2003) Tornado hazards in the United States. Clim Res 24:103–117
- Borys RD, Lowenthal DH, Cohn SA, Brown WOJ (2003) Mountaintop and radar measurements of anthropogenic aerosol effects on snow growth and snowfall rate. Geophys Res Lett 30(10): 1538. https://doi.org/10.1029/2002GL016855

- Chantara S, Sillapapiromsuk S, Wiriya W (2012) Atmospheric pollutants in Chiang Mai (Thailand) over a five-year period (2005–2009), their possible sources and relation to air mass movement. Atmos Environ 60:88–98. https://doi.org/10.1016/j.atmosenv.2012.06.044
- Che H, Zhang X, Li Y, Zhou Z, Qu JJ, Hao X (2009) Haze trends over the capital cities of 31 provinces in China, 1981-2005. Theor Appl Climatol 97:235–242. https://doi.org/10.1007/s00704-008-0059-8
- Danielson EW, Levin J, Abrams E (2003) Meteorology, 2nd edn. McGraw-Hill
- Deng X, Tie X, Wu D, Zhou X, Bi X, Tan H, Li F, Jiang C (2008) Long-term trend of visibility and its characterizations in the Pearl River Delta (PRD) region, China. Atmos Environ 42:1424– 1435. https://doi.org/10.1016/j.atmosenv.2007.11.025
- Feng S, Gao D, Liao F, Zhou F, Wang X (2016) The health effects of ambient PM2.5 and potential mechanisms. Ecotoxicol Environ Saf 128:67–74. https://doi.org/10.1016/j.ecoenv.2016.01.030
- Gabbert B (2019) How a dust devil forms. https://wildfiretoday.com/2019/07/30/how-a-dust-devilforms/. Accessed Apr 2021
- Gautam R, Hsu NC, Kafatos M, Tsay S-C (2007) Influences of winter haze on fog/low cloud over the Indo-Gangetic plains. J Geophys Res 112:D05207. https://doi.org/10.1029/2005JD007036
- Givati A, Rosenfeld D (2004) Quantifying precipitation suppression due to air pollution. J Appl Meteorol 43:1038–1056
- Gultepe I, Tardif R, Michaelides S, Cermak J, Bott A, Bendix J, Muller MD, Pagowski M, Hansen B, Ellrod G, Jacobs W, Toth G, Cober SG (2007) Fog research: a review of past achievements and future perspectives. Pure Appl Geophys 164:1121–1159
- Hanafi NH, Hassim MH, Noor ZZ, Ten JY, Aris NM, Jalil AA (2019) Economic losses due to health hazards caused by haze event in Johor Bahru. In: E3S Web of Conferences, vol 90, p 01009. https://doi.org/10.1051/e3sconf/20199001009
- Holton JR (2004) An introduction to dynamic meteorology, 4th edn. Elsevier Academic Press
- Hou J, An Y, Song H, Chen J (2019) The impact of haze pollution on regional eco economic treatment efficiency in China: an environmental regulation perspective. Int J Environ Res Public Health 16:4059. https://doi.org/10.3390/ijerph16214059
- Izhar S, Gupta T, Minz AP, Senapati S, Panday AK (2019) Influence of regional and long range transport air masses on fog water composition, contribution and toxicological response at Indo Gangetic Plain. Atmos Environ 214:116888. https://doi.org/10.1016/j.atmosenv.2019.116888
- Jacobson MZ (2005) Fundamentals of atmospheric modeling. Cambridge university press
- Jenamani RK (2007) Alarming rise in fog and pollution causing a fall in maximum temperature over Delhi. Curr Sci 93(3):314–322. https://doi.org/10.2307/24099461
- Jirak IL, Cotton WR (2006) Effect of air pollution on precipitation along the front range of the Rocky Mountains. J Appl Meteorol Clim 45:236–245
- Kokkonen TV, Grimmond S, Murto S, Liu H, Sundström A-M, Järvi L (2019) Simulation of the radiative effect of haze on the urban hydrological cycle using reanalysis data in Beijing. Atmos Chem Phys 19:7001–7017. https://doi.org/10.5194/acp-19-7001-2019
- Li G, Bei N, Cao J, Huang R, Wu J, Feng T, Wang Y, Liu S, Zhang Q, Tei X, Molina LT (2017) A possible pathway for rapid growth of sulfate during haze days in China. Atmos Chem Phys 17: 3301–3316. https://doi.org/10.5194/acp-17-3301-2017
- Luan Z, Han Y, Liu F, Zhao T, Liu C, Rood MJ, Yang X, He Q, Lu H (2017) Dust opacities inside dust devil column in the Taklimakan Desert. Atmos Meas Tech 10:273–279. https://doi.org/10. 5194/amt-10-273-2017
- Luan T, Guo X, Guo L, Zhang T (2018) Quantifying the relationship between PM2.5 concentration, visibility and planetary boundary layer height for long-lasting haze and fog-haze mixed events in Beijing. Atmos Chem Phys 18:203–225. https://doi.org/10.5194/acp-18-203-2018
- Malloy QGJ, Li Q, Warren B, Cocker DR, Erupe ME, Silva PJ (2009) Secondary organic aerosol formation from primary aliphatic amines with NO₃ radical. Atmos Chem Phys 9:2051–2060
- Matus K, Nam K-M, Selin NE, Lamsal LN, Reilly JM, Paltsev S (2012) Health damages from air pollution in China. Glob Environ Chang 22(1):55–66. https://doi.org/10.1016/j.gloenvcha. 2011.08.006

- Midwestern Regional Climate Center (2021) Fog. https://mrcc.illinois.edu/living_wx/fog/index. html. Accessed Apr 2021
- Molina MJ, Allen JT (2020) Regionally-stratified tornadoes: moisture source physical reasoning and climate trends. Weather Clim Extrem 28:100244
- Murphy SM, Sorooshian A, Kroll JH, Ng NL, Chhabra P, Tong C, Surratt JD, Knipping E, Flagan RC, Seinfeld JH (2007) Secondary aerosol formation from atmospheric reactions of aliphatic amines. Atmos Chem Phys 7:2313–2337
- NASA (2007) Phantoms from the sand: tracking dust devils across earth and mars. https://www. nasa.gov/vision/universe/solarsystem/2005_dust_devil.html. Accessed Mar 2021
- NOAA (2013) What is the difference between sleet, freezing rain, and snow? https://www.weather. gov/iwx/sleetvsfreezingrain. Accessed Mar 2021
- NOAA (2015) NSSL research: winter weather. https://www.nssl.noaa.gov/research/winter/. Accessed Mar 2021
- NOAA (2018) NOAA photo library: waterspouts. https://photolib.noaa.gov/Collections/National-Weather-Service/Meteorological-Monsters/Waterspouts/. Accessed Mar 2021
- NOAA (2019) Tornadoes—Annual 2019. https://www.ncdc.noaa.gov/sotc/tornadoes/201913. Accessed Mar 2021
- NOAA (2021) Tornadoes—April 2011. https://www.ncdc.noaa.gov/sotc/tornadoes/201104#:~: text=The%20April%2026%2D28%2C%202011,people%20were%20killed%20by%20 tornadoes. Accessed Mar 2021
- Northern Vermont University (n.d.) Upslope fog. https://apollo.nvu.vsc.edu/classes/met130/notes/ chapter5/upslope_fog3.html. Accessed Apr 2021
- Onishchenko O, Fedun V, Horton W, Pokhotelov O, Verth G (2019) Dust devils: structural features, dynamics and climate impact. Climate 7:12. https://doi.org/10.3390/cli7010012
- Price DJ, Clark CH, Tang X, Cocker DR, Purvis-Roberts KL, Silva PJ (2014) Proposed chemical mechanisms leading to secondary organic aerosol in the reactions of aliphatic amines with hydroxyl and nitrate radicals. Atmos Environ 96:135–144
- Price DJ, Kacarab M, Cocker DR, Purvis-Roberts KL, Silva PJ (2016) Effects of temperature on the formation of secondary organic aerosol from amine precursors. Aerosol Sci Technol 50:1216– 1226
- Pui DYH, Chen S-C, Zuo Z (2014) PM2.5 in China: measurements, sources, visibility and health effects, and mitigation. Particuology 13:1–26
- Rafkin S, Jemmett-Smith B, Fenton L, Lorenz R, Takemi T, Ito J, Tyler D (2016) Dust devil formation. Space Sci Rev 203:183–207. https://doi.org/10.1007/s11214-016-0307-7
- Raker EJ (2020) Natural hazards, disasters, and demographic change: the case of severe tornadoes in the United States, 1980–2010. Demography 57:653–674. https://doi.org/10.1007/s13524-020-00862-y
- Saide PE, Spak SN, Peirce RB, Otkin JA, Schaack TK, Heidinger AK, da Silva AM, Kacenelenbogen M, Redemann J, Carmichael GR (2015) Central American biomass burning smoke increase tornado severity in the U.S. Geophys Res Lett 42:956–965. https://doi.org/10. 1002/2014GL062826
- Saxena P, Kulshrestha UC (2015) Scavenging of urban air emissions by Fog at Delhi, India. In AGU Fall Meeting Abstracts, vol. 2015, pp A33H-0274, December
- Saxena P, Kulshrestha UC (2016) Contribution of carbonaceous species in SOA formation during fog and non-fog period. J Chem Bio Phys Sci 7:31–48
- Saxena P, Sonwani S (2019a) Primary criteria air pollutants: environmental health effects. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 49–82
- Saxena P, Sonwani S (2019b) Criteria air pollutants: chemistry, sources and sinks. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 7–48
- Saxena P, Sonwani S (2019c) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253

- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8):6639– 6656
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7:e06973. https://doi.org/10.1016/j.heliyon.2021.e06973
- Shrestha S, Moore GA, Peel MC (2018) Trends in winter fog events in the Terai region of Nepal. Agric For Meteorol 259:118–130. https://doi.org/10.1016/J.AGRFORMET.2018.04.018
- Simmons KM, Sutter D, Pielke R (2012) Blown away: monetary and human impacts of the 2011 U.S. tornadoes. In: Courbage C, Stahel WR (eds) Extreme events and insurance: 2011 annus horribilis. The Geneva Association Rep., pp 107–120. www.genevaassociation.org/media/200 995/GA-2012-Geneva_report%5B5%5D.pdf
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sonwani S, Saxena P, Shukla A (2021) Carbonaceous aerosol characterization and their relationship with meteorological parameters during summer monsoon and winter monsoon at an industrial region in Delhi, India. Earth Space Sci 8(4):e2020EA001303
- Stolaki S, Haeffelin M, Lac C, Dupont J-C, Elias T, Masson V (2015) Influence of aerosols on the life cycle of a radiation fog event. A numerical and observational study. Atmos Res 151:146– 161
- Sun J, Zhang J-H, Wang C, Duan X, Wang Y (2019) Escape or stay? Effects of haze pollution on domestic travel: comparative analysis of different regions in China. Sci Total Environ 690:151– 157. https://doi.org/10.1016/j.scitotenv.2019.06.415
- Syed FS, Körnich H, Tjernström M (2012) On the fog variability over south Asia. Clim Dyn 39: 2993–3005. https://doi.org/10.1007/s00382-012-1414-0
- Tang XC, Price D, Praske E, Lee SA, Shattuck MA, Purvis-Roberts K, Silva PJ, Asa-Awuku A, Cocker DR (2013) NO₃ radical, OH radical and O₃-initiated secondary aerosol formation from aliphatic amines. Atmos Environ 72:105–112
- Tiwari S, Payra S, Mohan M, Verma S, Bisht DS (2011) Visibility degradation during foggy period due to anthropogenic urban aerosol at Delhi, India. Atmos Pollut Res 2:116–120. https://doi.org/ 10.5094/APR.2011.014
- UA (2018) Condensation nuclei and the formation of dew, frost, haze, fog, and clouds. http://www. atmo.arizona.edu/students/courselinks/spring08/atmo336s1/courses/spring18/atmo170a1s1/lec ture_notes/mar27.html. Accessed Mar 2021
- UCI (2020) Chapter 7: precipitation processes. https://www.ess.uci.edu/~yu/class/ess5/Chapter.7. precipitation.all.pdf. Accessed Mar 2021
- University of Arizona (2018a) NATS 101 Section 13: lecture 10 condensation, cloud formation, and fog. http://www.atmo.arizona.edu/students/courselinks/fall10/nats101s13/Lectures/Lec ture10_webpost.pdf. Accessed Apr 2021
- University of Arizona (2018b) Condensation, cloud formation, and fog. http://www.atmo.arizona. edu/students/courselinks/spring08/atmo336s1/courses/spring18/atmo170a1s1/lecture_notes/ mar27.html. Accessed Apr 2021
- Van Oldenborgh GJ, Yiou P, Vautard R (2010) On the roles of circulation and aerosols in the decline of mist and dense fog in Europe over the last 30 years. Atmos Chem Phys 10:4597–4609
- Wang X, Zhang R, Yu W (2019) The effects of PM 2.5 concentrations and relative humidity on atmospheric visibility in Beijing. J Geophys Res Atmos 124:2235–2259. https://doi.org/10. 1029/2018JD029269
- Wang L, Zhou X, Lu M, Cui Z (2020a) Impacts of haze weather on tourist arrivals and destination preference: analysis based on Baidu Index of 73 scenic spots in Beijing, China. J Clean Prod 273:122887. https://doi.org/10.1016/j.jclepro.2020.122887
- Wang T, Chen Y, Gan Z, Han Y, Li J, Huang J (2020b) Assessment of dominating aerosol properties and their long-term trend in the Pan-Third Pole region: a study with 10-year multi-

sensor measurements. Atmos Environ 239:117738. https://doi.org/10.1016/j.atmosenv.2020. 117738

- WMO (1966) International meteorological vocabulary. World Meteorological Organization
- WMO (2017a) Hydrometeors. https://cloudatlas.wmo.int/en/general-classification-of-meteorshydrometeors.html#:~:text=Hydrometeors%20consist%20of%20liquid%20or,convention%2C %20not%20considered%20a%20hydrometeor. Accessed Mar 2021
- WMO (2017b) Lithometeors: classification of lithometeors. https://cloudatlas.wmo.int/en/ lithometeors.html. Accessed Mar 2021
- Xing Y-F, Xu Y-H, Shi M-H, Lian Y-X (2016) The impact of PM2.5 on the human respiratory system. J Thorac Dis 8(1):E69–E74
- Ye X, Ma Z, Zhang J, Du H, Chen J, Chen H, Yang X, Gao W, Geng F (2011) Important role of ammonia on haze formation in Shanghai. Environ Res Lett 6:024019. https://doi.org/10.1088/ 1748-9326/6/2/024019
- Yu HL, Chien LC, Yang CH (2012) Asian dust storm elevates children's respiratory health risks: a spatiotemporal analysis of children's clinic visits across Taipei (Taiwan). PLoS One 7:e41317
- Zhou L, Chen X, Tian X (2017) The impact of fine particulate matter (PM2.5) on China's agricultural production 2.5 from 2001 to 2010. J Clean Prod 178:133–141. https://doi.org/10. 1016/j.jclepro.2017.12.204
- Zhu D, Wan L (2019) Characteristics and formation mechanism of a serious haze event in autumn 2017 in Harbin, China. Air Qual Atmos Health 12:1169–1179. https://doi.org/10.1007/s11869-019-00726-4



3

Air Pollution and Its Associated Impacts on Atmosphere and Biota Health

Shishir Singh and Rakesh Kumar

Abstract

Unregulated humans' activities have over-exploited the earth's natural resources, disturbed its terrestrial and aquatic ecosystems, and released vast amounts of pollutants into the environment. Air pollution is typically the presence of one or more contaminants in the atmosphere and in such quantity that is injurious to animals, plants, humans, and all other man-made and natural resources. World Health Organization (WHO) estimates that every 9 out of 10 people are exposed to high levels of air pollutants. World over air pollution is responsible for the death of about seven million people annually. A large proportion of this affected population (about 91%) resides in the developing economies in the South-East Asia and Western Pacific regions. The low- to middle-income groups are the ones who are most affected by air pollutants. Scientific studies have reported that a marginal increase of 10 μ g/m³ of ambient PM₁₀ levels in Asian cities could lead to a surge in the mortality rate by 0.6%, which is significantly higher than the trends estimated for Western cities. The atmospheric CO₂ concentrations have already crossed the 410 ppm mark, which has never been recorded in human history. Air pollution operates on variable time scales and at various levels, i.e., local, regional, and global; and also produce impacts such as greenhouse effect and global warming; stratospheric; ozone depletion; atmospheric deposition, and acid rain; suppression of rainfall; atmospheric visibility reduction, etc. The severity of the impacts also varies from one organism to another and from one ecosystem to another ecosystem. Impacts on biota health are generally measured

S. Singh

National Physical Laboratory, New Delhi, India

R. Kumar (🖂) Department of Environmental Sciences, University of Jammu, Jammu, India

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_3

in terms of their influence on the morphological, biochemical, and physiological status and the responses generated thereof. Plants exposed to air pollutants may exhibit (1) direct or visible effects, which are often linked with short-term exposures to high concentrations of air pollutants or (2) indirect or invisible from long-term exposure to pollutants. Several national and international regulatory frameworks have been established to control and manage air pollutants. Despite intergovernmental and local efforts, air pollution has continued to rise across the major urban centers of the world, especially in developing economies like India and China. Scientific understanding and timely policy interventions are crucial in combatting the problem of air pollution emissions from the developed and developing world and consequently reducing their impacts on the living and non-living world. In this chapter, we have attempted to deliberate upon the major atmospheric pollutants (sulfur dioxide, oxides of nitrogen, ground level ozone, suspended particulate matter, fluorides, and peroxyacyl nitrates), their impacts on the atmosphere (global warming, stratospheric ozone depletion, acid rain, etc.), and threats to the biota health.

Keywords

Air pollution · Plants · Human health · Air quality · Atmosphere

3.1 Introduction

Earth and its atmosphere have been incessantly evolving since its inception. The primitive Earth had a toxic atmosphere with no traces of molecular oxygen and was too hostile to support any organic life. Early photosynthetic organisms were stromatolites that looked much similar to modern-day cyanobacteria that emerged around 3.2–3.5 billion years ago (Blankenship 2010). The gradual emergence and expansion of photosynthetic organisms over this Earth's surface paced the evolution of its atmosphere from an oxygen deficit to the present-day levels of around 21% (Allègre and Schneider 1994). The present-day atmosphere is an outcome of the earth's evolutionary history of 4.5 billion years; living organisms have played a crucial role in shaping the Earth and its atmosphere (Knoll 2015). Historical records suggest that over the ages various lifeforms have dwelled in a unique synergy with the Earth's atmosphere until the large-scale human interventions made since the dawn of the industrial revolution. Intensified and unregulated human actions have not only looted the Earth of its natural resources but also resulted in widespread landuse changes, emission of pollutants, and large-scale disturbances in the terrestrial and aquatic ecosystems. The atmospheric CO₂ concentrations have already touched 410 parts per million (ppm) marks at Mauna Loa Observatory, Hawaii (NOAA) (https://www.co2.earth/). Last time, the Earth's atmosphere possessed such high concentrations of CO_2 which was around three million years ago, when human had not even existed on this planet, and the average earth' surface temperature was as high as 2–3 °C relative to pre-industrial levels, while the average sea level (AMSL) was 15–25 m higher than at present (Lindsey 2020; Sonwani and Saxena, 2022).

Typically, air pollution is defined as the presence of one or more contaminants (chemicals, particulate matter, or biological materials) in the atmosphere at such concentrations and for such duration, that it is injurious, or tends to be injurious, to natural flora and fauna including humans (Katulski et al. 2011; Saxena and Sonwani, 2019a; Saxena and Srivastava, 2020; Saxena et al. 2021a); results in deterioration of surface or groundwater quality, soil productivity; damages terrestrial or aquatic ecosystems, and or materials (Manisalidis et al. 2020). In the last few decades, the urban and industrial clusters in developing economies like India and China have emerged as the new epicenters of atmospheric pollution, firstly from unplanned and unregulated economic development and secondly due to poor implementation of the regulatory framework (Chung et al. 2011). Besides, in the rural regions, the burning of traditional biomass-based fuel for heating and cooking has remained the primary source of pollution in indoor environments. According to WHO, world over around 90% of people were exposed to high levels of atmospheric pollutants in 2016. The air pollution alone kills an alarming 7 million people annually, out of which a whopping 4.2 million premature deaths are caused by outdoor air pollutants (Sonwani et al. 2021a). A large proportion of this burden, estimated to be around 91%, is shared by third world countries mainly in South-East Asia and Western Pacific regions (WHO 2020). Chung et al. (2011) reported that a marginal increase of 10 μ g/m³ of ambient PM₁₀ levels in some Asian cities would increase the mortality rates by 0.6%, which is much higher relative to the Western cities. In financial terms, premature deaths due to air pollution cost about \$5 trillion in welfare losses worldwide (The World Bank 2016; Sonwani et al. 2021b).

Air pollutants have been in existence for centuries. The first such complaint of air pollution from coal-burning was recorded in the thirteenth century in London (Katulski et al. 2011). Two major types of air pollution episodes were identified in the twentieth century: (1) London smog or "Reducing smog" and (2) the photochemical smog also called a "Los Angeles Type Smog". The former is formed as a consequence of the combustion of coal under strong inversion conditions, while the latter is formed from photochemical interactions of hydrocarbons and oxides of nitrogen (emitted from sources like automobiles, incinerators, municipal dumps, etc.) in the presence of ultraviolet light from the sun (Jacobson and Jacobson 2002). The worst pollution episode of London type smog occurred in 1952, which resulted in an estimated 4000 excess deaths than the normal (Brimblecombe 1987a, b). Asian region in recent decades has emerged as a new pollution epicenter of the twenty-first century. India alone accounts for 14 of the 20 worst polluted cities in the world in terms of atmospheric PM_{2.5} levels (IQAIR 2019). Recently, India has made a drastic shift in its policy toward regulating air pollution. India has committed to shifting its 40% of electricity generation to non-fossil fuels under the Paris agreement (2015). Besides, in 2019, India launched its first National Clean Air Programme (NCAP), which aims to reduce 20-30% PM_{2.5} and PM₁₀ levels across 102 cities in India by 2024 relative to 2017 levels (Government of India 2019). Lately, the Covid-19 shutdown in the country brought a short-term drop in air pollution levels. But as the lockdown restrictions have eased, the clear skies have disappeared again. The cities like New Delhi have already crept into the category of "severe" level of air pollution. Burning of agricultural residue is one of the major reasons for air pollution in Northern India and several other regions of South Asia too. There is a sharp dip in the air quality in the National capital New Delhi during winter. The latest reports suggest that this year, the share of crop stubble burning to New Delhi's $PM_{2.5}$ load has gone beyond 36% (SAFAR-India 2020). Huge economic losses are incurred by the airline companies due to dense foggy conditions prevailing over New Delhi in winter every year. In parts of Indonesia and Malaysia, burning of peatland and forests for land clearing for lucrative oil palm plantations also resulted in deterioration of air quality causing a choking smog which blankets not only parts of Indonesia but also spreads to Malaysia, Singapore, and southern Thailand.

3.2 Air Pollutants: Types and Sources

The majority of the air pollutants in the atmosphere are short-lived and quickly removed by wet or dry deposition generally in the vicinity of their sources. Some of them are having longer residence times (half-life of a pollutant) in the atmosphere. Many of these pollutants undergo transformation or "aging" forming new secondary species and transported via local as well as long-transport pathways. More than 3000 air pollutants have been identified which are otherwise not present in the natural environments (Gheorghe and Ion 2011). The places from where these pollutants originate are called "Sources", and the places they disappear are called "Sinks". There are a wide variety of sinks such as forests and other vegetation, soil, material structures, water bodies, etc. The removal mechanism of a pollutant from the atmosphere into the sink is called "Scavenging", while the time of its stay in the atmosphere is called its half-life. The half-life of a pollutant is defined as the time it takes for the disappearance or scavenging of half of a pollutant emitted into the atmosphere from its source. Atmospheric conditions and topography to a greater extent influence the local as well as the regional concentrations of a pollutant (Watson et al. 1988). For example, urban zones located in the valleys where atmospheric inversions take place generally have high pollutants owing to lower atmospheric mixing heights. Such places are more likely to experience a high frequency of severe smog episodes. Several gases like CO₂, CFCs, and N₂O have longer half-life times and tend to accumulate in the atmosphere over the years reaching concentrations high enough to substantially alter the natural composition of the air. The chlorofluorocarbons (CFCs) are chemically very inert and don't undergo any chemical degradation in the troposphere. In the stratosphere, high energetic ultraviolet (UV radiation) photolyze the CFC molecules, thereof releasing free chlorine atoms. Excess of free chlorine atoms disturbs the steady-state of ozone concentration in the stratosphere, triggering the destruction of stratospheric ozone, ultimately permitting harmful UV to penetrate the earth's surface (Vallero 2014; Saxena and Sonwani, 2019b).

Air pollutants can be distinguished by their chemical composition, persistence, transformation reactions, sources and sinks, and transportation mechanisms and effects on plants and animals including humans (Kampa and Castanas 2008).

Pollutants emitted directly from their sources are categorized as the primary pollutants, whereas those formed by transformation reactions with atmospheric gases or by other pollutants are termed as secondary air pollutants (Bernstein et al. 2004). The concentrations of a secondary atmospheric pollutant may decrease if a reduction in the emissions of precursors could be achieved (Gaba and Iordache 2011). Air pollutants are also classified into (1) criteria pollutants and (2) hazardous air pollutants. Both the categories of pollutants are hazardous. Regulatory authorities conduct regular monitoring for the criteria pollutants to maintain the state of ambient air quality of a region (Katulski et al. 2011). The majority of the countries including India have devised their National Ambient Air Quality Standards (NAAQS) for air pollutants of utmost concern. Under the Clean Air Act (Environmental Protection Agency 2008), United States Environmental Protection has devised NAAQS for six pollutant criteria air pollutants namely: SO₂, NO₂, CO, tropospheric O₃, suspended particulate matter, and lead (Saxena and Sonwani, 2019c).

3.3 Air Pollution and Its Impacts on the Atmosphere

Air pollutants operate at local, regional, and global levels and produce different impacts on each level (Manisalidis et al. 2020). For example, the presence of tropospheric ozone is known to augment Earths' greenhouse effect, while ozone in the stratosphere produces overall cooling effects. Some of the impacts of air pollutants on the atmosphere are discussed as:

3.3.1 Greenhouse Effect and Global Warming

Earths' atmosphere is dynamic and continually in motion. The vertical and horizontal mass movements in the atmosphere are driven by differential vertical and horizontal heating of Earth by incoming solar radiations. The state of the atmosphere measured in terms of variations in temperature, precipitation, wind direction and speed, relative humidity, the extent of cloud cover, and solar flux measures in hours, days, or weeks defines its weather (Tinsley and Yu 2003). These parameters, when observed within the geographical context and over longer durations (years, decades, centuries, thousands of years), define climate (Paruelo et al. 1998). In general, the climate is defined as the average weather conditions over a minimum period of 30 years. Air pollutants to a larger extent impact the global climate, likewise, the changing climate can impact air quality. The black carbon, a particulate product of incomplete combustion, is a strong radiative agent. This contributes to the warming of the Earth, while the particulate sulfates reflect the incoming solar radiation and exhibit a cooling effect on the Earth's atmosphere (Pöschl 2005).

The atmosphere allows the visible short-wave radiation from the sun to pass through it while inhibits the flow of long-wave infrared radiation back into space, a phenomenon which is known as the "Greenhouse Effect". The greenhouse effect helps the earth to maintain its average surface temperature at $\sim +15$ °C (Schneider

1989). Without the natural greenhouse effect, the average temperature on Earth is estimated to have been colder by approximately 30 °C, cold enough to sustain any of the present-day ecosystems. CO_2 and water vapors are the major gases that regulate the greenhouse effect on Earth. Other important greenhouse gases (GHGs) present in trace amounts in the atmosphere are CH₄, N₂O, tropospheric O₃, CFCs, PFCs, and SF6 (Lee 1973). Accumulation of GHGs since the onset of the industrial revolution and change in earth's albedo from land-use activities have significantly raised the global average temperatures and consequently bringing in the anthropogenicinduced climate change (Ramanathan et al. 2001: Schmidt et al. 2006). As of 1750, the GHGs like CO₂, CH₄, and N₂O have shown a respective increase of 40%, 150%, and 20% as observed till 2011. United Nations Framework Convention on Climate Change (UNFCCC) refers to climate change to a change of climate that is attributed directly or indirectly to human activity that alters the composition of the global atmosphere and that is in addition to natural climate variability observed over comparable periods. The alteration in the energy budget of the earth by atmospheric pollutants is measured in terms of radiative or climate forcing (Pachauri and Reisinger 2007) defined as any variation forced on the energy balance of the Earth (Murphy et al. 2009) and expressed in terms of watts per square meter (Wm^{-2}) . The causal substances responsible for radiative forcing are known as radiative forcing agents. Earth is warmed by positive forcing agents and cooled by negative forcing agents. Greenhouse gases are positive radiative forcing agents whose presence in the atmosphere reduces the earth's ability as a whole to re-radiate sunlight into space, as a consequence of which the earth absorbs more thermal energy and warms (Zhao et al. 2019). Climate change has posed danger for all countries, however, the nations with limited adaptive capability are at the greater risks of climate-induced extreme events. Many regions in India due to their fragility are highly vulnerable to climatic impacts. Several kinds of research have been carried in India to make future climatic projections as well as to assess the vulnerability of different regions of the country toward climate change and other associated risks. In their study on Karnataka, Kumar et al. (2016a, b) have estimated that around 70% of the cultivated area of the State has been facing extreme to a high level of vulnerability from the climateassociated impacts. In another study from the Himalayan state of Arunachal Pradesh, Maiti et al. (2017) have highlighted that 7 out of 12 district studies show high social vulnerability toward changing climate. Sam et al. (2020) also emphasized that the formulation of local adaptive measures in India could potentially reduce the risk of climate-induced extreme events. Researchers have associated the probability and severity of heatwaves in India with global warming (van Oldenborgh et al. 2018). Uprety and Saxena, 2021 have predicted that the average temperatures in India would increase by 2.5-4.4 °C by end of the century w.r.t. to the current scenario, whereas the annual rainfall has been projected to show an average increase by 15–24%. Gupta and Jain (2018) have estimated that in the future the droughts in the Indian sub-continent would get more severe and frequent. Likewise, Chevuturi et al. (2018) have reported that the Leh region of India had shown a warming trend and reduced precipitation which could be associated with the reducing number of days with higher precipitation amounts over the region.

Worldwide, anthropogenic activities have produced an increase of 45% in the atmospheric concentrations of CO₂ from 280 ppm to present-day levels of 411 ppm (Shieber 2019). Such high concentrations of CO_2 have never been witnessed in human history; last time, the earth's atmosphere had these much high CO_2 concentrations which were around three million years ago, when the average earth' surface was as high as 2-3 °C relative to pre-industrial levels, and average sea level was 15–25 m higher than at present. The cumulative impact of human activities on the global average temperature rise has been approximated to around 1.0 °C relative to the pre-industrial levels of 1750. At the present pace, the GHG emissions would result in warming up to 1.5 °C between 2030 and 2052 (IPCC 2018). United Nations Framework Convention on Climate Change (UNFCCC) has taken steps from time to time to develop a global consensus to check the GHGs emissions by various countries. Under UNFCCC's Paris agreement (2015), keen efforts have been made by the global community to restrengthen the commitment for the regulation of GHGs emissions, limit the global temperature rise in this century well below 2 °C above pre-industrial levels, and to pursue efforts to limit the temperature increase even further to 1.5 °C (UNFCCC 2015). IPCC which provides important scientific inputs for framing international policies toward climate change recently has also released a special report in 2018 to evaluate the global response concerning advancement made toward limiting the temperature rise in this century up to 1.5 °C above pre-industrial levels (IPCC 2018).

3.3.2 Stratospheric Ozone Depletion

The stratospheric ozone layer or "Good" ozone is a region of the atmosphere between 10 and 50 km where 90% of the atmospheric ozone is contained. Stratospheric ozone (O_3) absorbs harmful ultraviolet (UV) radiations from the solar flux, which could otherwise harm life on the earth's surface. The total atmospheric O_3 mass is estimated to be around three billion metric tons, which constitutes only 0.00006% of the atmospheric mass of all gases combined. The highest O_3 concentrations nearly equal to ~15 ppm is found at roughly 32 km above the earth's surface. Total columnar ozone (TCO) is represented as a Dobson Unit (DU). One Dobson Unit is defined as the number of molecules of O_3 required to create a pure O_3 layer of 0.01 mm thickness at a temperature of 0 °C and sea level (1 atm pressure) (Godish et al. 2014). Natural O_3 formation and degradation begin by high-intensity, shortwave solar ultraviolet radiation (<242 nm) in the stratosphere, as seen in the equations below:

$$O_2 + h\upsilon \to O + O \tag{3.1}$$

$$O + O_2 + M \to O_3 + M \tag{3.2}$$

$$\mathbf{O} + \mathbf{O}_3 \to 2\mathbf{O}_2 \tag{3.3}$$

$$O_3 + h\upsilon \rightarrow O_2 + O(^1D) \tag{3.4}$$

where hv is a UV photon and M is a molecule that absorbs energy. These reactions are known as the Chapman Cycle. In the stratosphere, ozone production and degradation involve molecular-oxygen (O_2), atomic oxygen (O), and O_3 reactions (Rowland 2009).

Ultraviolet radiations coming from the sun are classified into UV-A, UV-B, and UV-C radiations. Stratospheric ozone absorbs almost 99% of UV wavelengths below 320 nm. UV-C light (100-290 nm) has a relatively smaller wavelength and is extremely harmful to the living biota on the earth's surface. Most of it is absorbed by O₃ and O₂ molecules in the stratosphere. The UV-B (290–320 nm) are highly energetic UV radiations and harmful for living biota, roughly 95% of all UV-B is absorbed by the stratospheric ozone. UV-A (320-400 nm) has the longest wavelength and is the least harmful UV radiations (Rowland 2009). Earth's surface receives a high influx of UV-A radiation and low to moderate influx of UV-B radiations depending upon the state of stratospheric O₃ (Cutchis 1974). Scientific deliberations on stratospheric O_3 depletion started in the late 1960s. In 1974, Sherwood Rowland and Mario Molinas suggested that chlorofluorocarbons (CFCs) in the stratosphere undergo photolytic break down releasing Cl atoms, which in turn chemically degrade the stratospheric O_3 . Later evidence suggested that O_3 layers could also be affected by the increased release of nitrous oxide (N_2O) from the denitrification of ammonium and nitrate fertilizer (Rowland 2009). The stratospheric ozone hole was first discovered by a team of scientists, Farman, Gardiner, and Shanklin working in Antarctica; their findings were published in the "Nature" journal in 1985. The decline in the polar ozone was far more than the anticipated figures (Farman et al. 1985). More information from satellite data suggested that the ozone depletion extended beyond the experimental sites, covering most of the Antarctic region. Technically, the ozone hole is defined as a geographical region having an ozone concentration of the air column lower than 220 DU (Newman et al. 2004). In the 1980s, several deliberations between scientists and policymakers resulted in the framing of landmark international regulatory provisions aimed at phasing out and prohibiting the use of long-lived Cl and Br halocarbons (Godish et al. 2014). Montreal Protocol of 1987, and subsequent amendments made therein, placed an international ban on the manufacture and distribution of CFCs, halons, and other ozone-depleting substances (ODS) with effect from 1989. It is expected that stratospheric O₃ would eventually recover to its pre-industrial levels, though it would take years for its complete recovery since CFCs and halons have a longer residence time in the atmosphere (Wilmouth et al. 2018).

In the lower troposphere, surface O_3 has a crucial role in increasing the oxidizing ability of the atmosphere but also has detrimental effects on plant and animal health including humans (Saxena et al. 2019). In India, surface ozone observations were carried out by various groups in different regions and their findings were highly useful to explain the regional air quality. South-East Asia is an anthropogenically active region and model simulations predicted major environmental consequences due to increased ozone values only in India, China, etc.

3.3.3 Atmospheric Deposition and Acid Rain

Atmospheric gases and particulates emitted from different sources are transported from air to water and soil through atmospheric deposition. This transfer process may be harmful or beneficial in terms of the impact of deposition on various sources. For instance, ozone deposition has detrimental effects on vegetation, while the deposition of nitrogenous compounds may be beneficial or detrimental. Nitrogen deposition can bring in additional nutrients for the plants, though the excessive deposition may cause eutrophication of water bodies (Smith et al. 1999). The acidification of precipitation, also termed as "Acid Rain", is also an example of the detrimental effects of atmospheric deposition. During their atmospheric transport and deposition, gases and particles undergo different physical and chemical transformations. These transformations affect significantly the fate and behavior of different chemicals undergoing deposition (Pacyna 2008). Besides, atmospheric deposition has potential consequences of increasing glacier melting by the darkening of the surfaces of snow and ice; decreases the photosynthesis process on the surface of the leaves; acidification of water bodies; changes in the soil's properties, etc. (Menon et al. 2007).

Atmospheric depositions on the terrestrial and aquatic ecosystems may happen through (1) dry deposition and (2) wet deposition. Primary and secondary particulates are removed by the gravitational force from the atmosphere over time without precipitation, this process is termed as "dry deposition" (Lovett and Kinsman 1990). Dry deposition primarily depends on the size, concentration, and reactivity of the particle deposited and the surface being affected (Lindberg et al. 1982). Particles with a size of >0.8 mm are considered to be easy to settle whereas those with a diameter of <0.08 mm may remain suspended for a longer duration in the atmosphere and therefore transported over to longer distances (Ruijrok et al. 1995). Dry deposition primarily delivers particulates of crustal origin predominantly rich in Ca^{2+} , Mg^{2+} , K^+ , and total phosphorous to the earth's surface. On the other hand, wet deposition is assisted by wet weather events (e.g., rain, snow, and hail) and dominantly deposits Na⁺, total nitrogen, NO₃⁻, and fine particulates to the earth' surface (Morales-Baquero et al. 2013). In countries like India, where dry conditions prevail in a larger part of the year, dry deposition is an important process of removal of atmospheric pollutants (Sharma et al. 2008). Wet deposition predominates in the higher precipitation regime zones (Balestrini et al. 2000). Dry deposition is an important feature in the local removal of micropollutants in the industrialized regions too (Gambaro et al. 2009). Studies have shown that dust load deposited from dry deposition has profound effects that impact on the lake, marine sediments, leaf surfaces, forest canopy, crop productivity, etc. (Gao et al. 2019). The aerosol

deposition over snow or ice affects its reflective efficiency and increases the heat absorption and melting of Himalayan glaciers and polar ice (Hansen and Nazarenko 2004; Yasunari et al. 2013). Wet deposition, on the contrary, is completely related to wet weather events (e.g., rain, snow, and hail). Wet deposition is responsible for the removal of the majority of aerosols from the atmosphere. Wet deposition can be classified as (1) in the cloud scavenging, where condensation occurs around one or more aerosols which are used as a core (nucleation) and (2) scavenging beneath the cloud, where the particles are impacted in the air column beneath the cloud. Both of these may occur individually or together, depending on the aerosol concentration in and below the cloud when precipitation is taking place (Roux et al. 2016).

Acid rain refers to low pH precipitation events. Natural rain is slightly acidic, having a pH of 5.7 due to the formation of carbonic acid formed from the dissolution of atmospheric CO₂ following Henry's law governing the dissolution of gases. The acidity of rain is caused by the dissolution of atmospheric SO₂ and NO₂, which undergo reactions with water molecules in the atmosphere to produce acidic compounds, i.e., H_2SO_4 and HNO_3 (Magaino 1997). Acid rain is highly harmful to terrestrial plants, aquatic biota, crops, and materials. Such precipitation events are threats to the terrestrial and aquatic ecosystems, the local and regional geology is important in neutralizing the acidity. Soils and surface water having lower amounts of Ca and Mg carbonates exhibit lower buffering capacities and are most prone to damages from acid deposition. Acidity lowers the cation exchange capacity of the fertile soils, it also results in leaching away of an important nutrient, thus restricting their availability to the plants. On the other hand, the availability of toxic ions like Al^{3+} and Fe^{2+} increases in the soil solution (Aerts and Bobbink 1999). The acidic deposition has resulted in the chronic acidification of many lakes and streams in North America and the Scandinavian peninsular region (Godish et al. 2014). Beier et al. (2017) reported that the chronic acidification of the Adirondack region's (USA) forests, lakes, and streams affected the potential economic and cultural benefits they provide to society. Liu et al. (2018) observed that the Chinese fir sapling height growth rate (HGR) and basal diameter growth rate (DGR) decreased as acid rain pH decreased. In a similar study, Du et al. (2017) observed that acid rain-reduced leaf chlorophyll content on average by 6.7% per pH unit across the recorded plant species. Sosa-Echeverría et al. (2019) also emphasized that external emission sources play a decisive role in acid rain within Mexico City. In India, Roy et al. (2016) reported that 55 and 65% of the rainwater samples during monsoon seasons of 2013 and 2014 over Falta and Darjeeling, West Bengal (India) were acidic. In similar observations, Garg et al. (2000) reported that acidic rain events occurred throughout the year, with a frequency of 64% and 87% during monsoon and non-monsoon seasons, respectively, in Guwahati during June 2016–June 2017.

3.3.4 Suppression of Rainfall

Air pollutants produce diminishing effects on the intensity of rainfall. This could be a serious public issue especially for the water-scarce and densely populated regions of the world. Long-term precipitation data of 50 years collected from several mountain ranges in the western United States have suggested a gradual decrease in the rainfall by around 10–25% (Schiermeier 2007) during the observation period. Cloud is formed of tiny cloud droplets typically of a 10 µm radius. These droplets coalesce together to form precipitation-sized raindrops of radius 1000 µm (= 1 mm), thus the mass of typical raindrops is equivalent to one million cloud droplets.

The coagulation of tiny cloud droplets into raindrops is governed by their size, at least some of the cloud droplets in a cloud mass should have a size of radius $>15 \,\mu m$ for the effective formation of precipitation in that cloud (Rosenfeld 1999, 2000). Hydrophilic aerosol particles are good cloud condensation nuclei (CCN). Water vapor easily condenses on the surface of such particles resulting in the formation of clouds under supersaturated conditions (Che et al. 2016). Whereas, the pollution aerosols act as small CCN, which produces a large number of smaller-sized cloud droplets. This leads to the suppression in the formation of precipitation-sized rain droplets within the cloud mass and prolongs the clouds' life, warm rain process, and ice precipitation (Rosenfeld 2000; Borys et al. 2003; Rosenfeld et al. 2008). Besides, the sulfate aerosols are also known to suppress the precipitation from the clouds. Borys et al. (2003) have reported that an addition of 1 μ g m⁻³ of sulfate aerosols leads to the reduction of snowfall rate by 50% in the Colorado Rocky Mountains (USA). Thus, there is enough experimental evidence to suggest that anthropogenic aerosols can disturb the hydrological balance over a landmass by interfering with the cloud formation and precipitation.

3.3.5 Visibility Reduction

Visibility is considered as a primary air quality index. Visibility reduction results from the absorption and dispersal of sunlight by suspended particulate matter as well as from the atmospheric gases (Hyslop 2009). Visibility is also limited by natural factors such as rain, snow, fog, aeolian dust, sea spray, hazes, etc. The first case of Arctic haze was reported in the 1950s. Industrial pollution of SO₂, elementary carbon (EC), HCs, metals, other gases, and particulate matter are the main cause of the Arctic haze. As the occurrence of Arctic haze was identified and researched over decades, observation of the Asian brown clouds was made in 1999 (Graff Zivin and Neidell 2013). The brown haze of about 3 km high covers the majority of the tropical Indian Ocean, South East, and East Asia. Studies have revealed high concentrations of sulfates, nitrates, organics, carbon (C), fly ash, and other pollutants as the major components of such haze (Chen et al. 2012).

Indo-Gangetic plains (IGP), in particular, are the hotspots of Asian brown haze or clouds. Large-scale agricultural, industrial, and other commercial activities in this

region generate a wide variety of atmospheric pollutants resulting in the formation of brown haze. The winter in North India is frequented by low visibility from fog formation. Indian capital city, New Delhi, and nearby regions, experience a high frequency of smog episodes during winter due to pollutants from the burning of crop stubble in the surrounding states of Punjab, Haryana, and Uttar Pradesh during the winter season (Kumar et al. 2016a, b) along with high relative humidity (RH), sufficient sunlight, and reduced vertical mixing (Chen et al. 2018). Smog episodes have been associated with serious health ailments, besides low visibility often disrupts railways, flights, and road transport services yearning economic losses. Visibility reduction is not only detrimental to ambient air quality (Gu et al. 2017) but can also diminish solar flux reaching the earth's surface. Volcanic emissions from the eruption of Mount Pinatubo, the Philippines in June 1991 released an estimated 20 million sulfur dioxide in the atmosphere. Sulfate aerosol emissions of this magnitude reduce the solar flux reaching the earth's surface lowering the tropospheric temperatures.

3.4 Impacts on Biota Health

Atmospheric pollution significantly affects human health, terrestrial and aquatic ecosystems, local and regional climate, and consequently society and economy (Correia et al. 2013; Fang et al. 2013; Rao et al. 2017). The Geneva Convention held in 1979 on Long-range Transboundary Air Pollution emphasized the protection of humans and the natural ecosystem from the deleterious effects of atmospheric pollutants. The convention has endorsed the reduction of eight air pollutants: SO_{x} , NO_x, persistent organic pollutants, non-methane volatile organic compounds, tropospheric ozone, heavy metals, particulate matters (PM₁₀, PM_{2.5}), and black carbon (De Marco et al. 2019). The effects of air pollutants on lower and higher organisms are measured in terms of their influence on the morphological, biochemical, and physiological status and responses thereof. In plants, the penetrations of pollutants mainly occur via leaves (Fig. 3.1). Small penetration of pollutants may happen through the stems and trunk (Seyyednejad et al. 2011). Air pollution effects may be classified into (1) direct or visible and (2) indirect or invisible injury (Gheorghe and Ion 2011). Visible injury is often linked with short-term exposures to high concentrations of air pollutants. The damaged portions of the leaf surface can offer easy entry for plant pathogens. Besides the visible injury in the form of discoloration, scabs, etc. can reduce the market values especially of the crops, for instance, visible injury can reduce the market value for tobacco, spinach, and horticultural crops. Invisible injury results from long-term exposure to pollutants. It impacts various physiological or biochemical processes in plants that can ultimately damage their growth and development, reduce nutritional quality, and yield (Liu and Ding 2008).



Fig. 3.1 Schematic representation of the environment of leaf surfaces. [adapted from Garrec 2020]

3.4.1 Sulphur Dioxide (SO₂)

Sulfur dioxide (SO₂) is a common atmospheric pollutant that affects plants, animals, men, and materials. SO₂ has a solubility of 228 g/L in the water at 0 °C. SO₂ is an important precursor for the formation of secondary atmospheric pollutants and acid rain. As of 2019, India contributes to the largest anthropogenic emissions of SO₂ in the world, which roughly equals to >15% of global emissions of SO₂ from anthropogenic sources (Kapil 2019). Under high humidity conditions, SO₂ is oxidized to SO₃ which forms sulfuric acid (H₂SO₄) when hydrated. SO₂ is taken up by the plants predominantly through the stomata. The uptake is dependent both on the stomatal conductance (Mudd 2012) as well as the prevailing environmental conditions such as the amount of light, temperature, relative humidity, wind speed, etc. Under the low light and water deficit condition, stomata remain closed and exhibit a lower degree of susceptibility to the SO₂ injury.

Nonvascular plants like lichens and bryophytes having low chlorophyll contents and no cuticle present are relatively more susceptible to SO_2 injury under high exposures. SO_2 has been the major culprit responsible for the dramatic decline in several bryophytes and lichens species in urban and industrialized regions of the world. Under high SO_2 polluted atmospheres, lichens become compact and sterile with a much smaller colony size. Among the plants, the most common type of acute injury due to SO_2 pollution is interveinal chlorosis. Some common species susceptible to SO₂ injury are oats (Avena sativa), cucumber (Cucumis sativus), and spinach (Spinacia oleracea), while celery (Apium graveolens), maize (Zea may), and citrus species (Mudd 1975) are known to exhibit resistance toward SO₂ injury. India is the largest emitter of sulfur dioxide in the world. Several studies have been conducted in India to evaluate its effects on plants and animals. Farooq et al. (1985) reported the depletion of starch and accumulation of free sugars in tissues in 2-month-old chilbil Holoptelea integrifolia exposed to SO₂ in experimental chambers. Similarly, Agrawal and Verma (1997) have also observed that SO₂ exposure leads to a reduction in total chlorophyll, ascorbic acid, starch, and protein contents in the 30-day-old two cultivars of wheat (Malviya 206 and Malviya 234). In another study, Verma and Agrawal (1996) observed that SO₂ fumigation to soybean (Glycine max) plants resulted in severe effects on photosynthetic pigments, ascorbic acid and protein contents, biomass, and productivity. A study by Dwivedi and Tripathi (2008) revealed a positive correlation between ambient air sulfur dioxide and sulfate contents in leaves of Ficus religiosa. The nonvascular plants show the acute direct impacts of SO_2 at much lower concentration relative to vascular plants, thus bryophytes and lichens often determine critical level set for ambient atmospheric SO₂ concentration in the natural environment. Experiments by Majernik and Mansfield (1970) revealed that exposure of SO₂ to fully hydrated 5-week-old broad bean (V. faba) leaves at 40% relative humidity and 18 °C, promoted stomatal opening, thereby promoting greater diffusion of sulfur dioxide into the leaves. Similar experiments by Unsworth et al. (1972) using ¹⁴C technique in the lichen species Cladonia alpestris, C. deformis, Stereocaulon paschle, and Umbilocaria muhlenbergii suggested that exposure to SO₂ at low pH resulted in a reduction of carbon fixation by lichen species. Similarly, studies have shown that exposure to SO₂ increased the rate of respiration in lichens and bryophytes.

3.4.2 Oxides of Nitrogen

Nitrogen in molecular form (N_2) is the most abundant gas in the atmosphere. Lesser amounts of nitrogen exist as nitrous oxide (N_2O) , nitrogen dioxide NO_2 , nitric oxide (NO), ammonia (NH_3) , and several other oxides of nitrogen. Nitrogen dioxide acute injury signs appears in plants when the latter is exposed to a threshold concentration of 2 ppm of NO_2 for a duration of 4-h. This rarely happens at its natural ambient background concentrations of NO_2 . Natural sources of oxides of nitrogen include atmospheric lighting, the decay of organic matter, and forest fires. While human activities such as the burning of fossil fuels in factories, automotive, and other hightemperature processes are its primary sources, which often emits more than its natural sources. The usage of catalytic converters in the locomotive exhaust can prevent emissions of nitrogen oxides to a larger extent (Forcetto and Daemme 2017). Many oxides of nitrogen are important precursors resulting in the formation of secondary pollutants including smog, acid rain, ground-level ozone, and formation of particulates (Brown et al. 2006; Yung et al. 2007; Ghazali et al. 2010).

43

Exposure of NO₂ in plants beyond tolerable threshold levels triggers complex physiological responses and changes in antioxidant enzyme activity. The use of tolerant species which could catabolize the atmospheric NO_2 is considered a viable ecological method to reduce its air concentrations. A recent experiment conducted by Sheng and Zhu (2019) revealed that NO₂ exposure in garden plants (6 μ L/L NO₂ in glass chambers and subsequently transferred to a natural environment for a 30-day recovery) affected the leaf chlorophyll content and induced biochemical defense responses. Mansfield et al. (1982) also suggested that SO₂ interferes with the induction of the nitrite reductase, an enzyme responsible for the reduction of NO_2 to NH₃, which possibly the main mechanism by which NO₂ toxicity is regulated by the plant cells. Besides, atmospheric nitrogen deposition results in soil degradation thus indirectly affecting the soil biota. Acidic deposition of oxidized and reduced nitrogenous compounds is characterized by loss of buffer capacity of the soils and reduces their cation exchange capacity (CEC). Lowering of the CEC of the soils leads to the increased loss of the Ca^{2+} and Mg^{2+} and other micronutrients to the soil solution, and increased availability of harmful metals like Al³⁺ and Fe²⁺ (Aerts and Bobbink 1999). Besides, the lower pH triggers the inhibition of nitrification, resulting in an increase of ammonium to nitrate ratios. Because of their high sensitivity to atmospheric NO₂, lichens can be used as potential cost-effective bioindicators of NO_2 pollution in the air (Conti 2008).

Lichen distribution, colony size, and concentration of elemental nitrogen in thalli have been used for biomonitoring nitrogen distribution and deposition patterns (Johansson et al. 2012) in many parts of the world. Owing to their unique biology, and lack of a cuticle layer to regulate the atmospheric absorption and the leaching of nutrients, lichens dynamically concentrate such nutrients in proportion to their atmospheric abundances. Increased nitrogen deposition favors the growth and abundances of eutrophic nitrogen-loving lichen species over oligotrophic species, growing in nutrient-deficient environments (Root et al. 2015). Bignal et al. (2007) reported the ecological impacts of nitrogen oxides from vehicular sources on vegetation at two experimental sites in England. Their findings suggested that NO_2 emissions from vehicular emissions exerted a strong influence on vegetation within the first 50–100 m from the road. In India, Pandey and Agrawal (1994) reported that exposure of SO₂ and NO₂ (0.1 ppm SO₂ + 0.2 ppm NO₂ for 4 h daily for 50 days) to 45-day-old Tomato plants (*Lycopersicon esculentum*) produced reduced root growth and low root:shoot ratio.

3.4.3 Ground Level Ozone

Ground level or tropospheric ozone is a secondary air pollutant unlike good ozone in the stratosphere. It is produced in the troposphere by a series of complex photochemical reactions from its precursor gases, i.e., NO_x and VOCs (Brown et al. 2006; Ghazali et al. 2010). The emissions of precursors of O_3 have decreased or remained constant in the last few decades especially in developed countries. But in the developing countries China and India, due to rapid industrial development and



Fig. 3.2 Effects of ozone on vegetation: from plant cells to ecosystems. [Source: ① J.P. Garrec]

urbanization, emissions of such gases have been continuously increasing (IPCC 2007; Ohara et al. 2007). Other than precursors, meteorological parameters like solar irradiations and temperature influence atmospheric ozone concentration.

Ozone is highly phytotoxic, its adverse effects on plants were first identified in the 1950s (Ashmore 2005), which can affect many plant species growing in the wild as well as the cultivated ones (Runeckles et al. 1992; Saxena et al. 2021b). Common symptoms of ozone injury include a decrease of photosynthetic pigments, visible leaf injury, and decline of photosynthetic activity in plants exposed to higher doses of atmospheric ozone (Krupa and Manning 1988; Soldatini et al. 1998; Guidi et al. 1998; Saitanis and Karandinos 2001). The acute visible injury in plants (Fig. 3.2) could decrease the economic value of species such as horticulture crops whose market value highly depends on their visible appearance (Fumagalli et al. 2001). For instance, in North Athens, an ozone episode of 1998 caused such severe reddening and necrosis on Chicorium endivum and Lactuca sativa devalued the market price of that local crops (Velissariou 1999). Many examples of ozone injury to the crops have been reported from India (Bambawale 1986), Taiwan (Sheu and Liu 2003). Proietti et al. (2016) reported a decrease in Gross Primary Production (GPP) ranging from 0.4 to 30%, in European forest from ozone exposure during 2000–2010. Heck and Dunning (1967) have also observed that plants are grown under shorter photoperiods and lower light intensities are more sensitive to ozone stress relative to those grown under longer photoperiods and lower light intensities or shorter photoperiods and higher light intensities. Further work by Hoshika et al.

(2013) suggests that ozone exposure and water stress together delayed the time of the closing signal in snap bean. Gas chamber experiments by Alonso et al. (2001) also suggested that cumulative effects of ozone and drought decreased antioxidant enzyme activities in *Pinus halepensis* exposed to 40 nL L^{-1} concentrations of ozone. Studies have revealed that ozone exposure resulted in reduced chlorophyll contents of leaves, many workers have used plant's chlorophyll content as a reliable indicator of ozone injury (Agrawal et al. 1993; Welfare et al. 1996; Della Torre et al. 1998). In India, Sarkar and Agrawal (2012) observed that the growth parameters such as pollen viability and grain yield were reduced under elevated O_3 stress exposure to two high-yielding rice cultivars (Oryza sativa L. cvs Malviva dhan 36 and *Shivani*). Bhatia et al. (2011) reported that ozone exposure to rice (Oryza sativa L.) has been associated with a decrease in methane and nitrous oxide (GHGs) emissions. In similar observations, Singh et al. (2018) reported that elevated ozone exposure of 30 ppb above ambient levels to 14 widely grown Indian cultivars exposed to 30 ppb elevated O_3 resulted in foliar injury and severely affected grain yield. Mina et al. (2010) have also observed the effects of O_3 exposure on different growth stages of Potato (Solanum tuberosum).

3.4.4 Suspended Particulate Matter

Suspended particulate matter includes the solid and liquid particles suspended into the air. These particulates vary in their shape, size, and composition and originate from diverse natural and anthropogenic sources (El Morabet 2019). Based on their size which is measured in terms of their aerodynamic diameter (dp), suspended particulate matters are classified into three commonly used size categories as (1) coarse particulates (dp = $2.5-10 \mu$ m), (2) fine particulates (dp $\leq 2.5 \mu$ m), and (3) ultrafine particulates (dp $\leq 0.1 \mu$ m) (Seinfeld and Pandis 2016). The aerodynamic diameter of a particle is defined as the diameter of a unit density spherical particle, having the same settling velocity as that of the particle under consideration. Coarser particulates are generally of crustal in origin, i.e., derived from aeolian action on dry soil, road dust sediments, combustion particulates are originated directly from the combustion processes, the nucleation of the small particulates, and/or gas to particle conversion mechanisms.

Many studies have revealed that in the urban environments the PM_{10} and $PM_{2.5}$ atmospheric concentrations are closely linked to vehicular emissions and other combustion processes (Prajapati and Tripathi 2007; Kumar and Yadav 2016; Kumar et al. 2016a, b). World over, countries including India have established the legal framework to regulate the ambient concentration of suspended particulates especially $PM_{2.5}$ (dp $\leq 2.5 \mu$ m) and PM_{10} (dp $\leq 10 \mu$ m). Ultrafine particles can penetrate the systemic circulation and exhibit high levels of toxicity relative to coarse and fine mode particles. Moreover, the larger surface area of the fine particles increases their potential to adsorb high content of hazardous substances including heavy metals (Bernstein et al. 2004). Deposition of particulates on the leaf surface

may obstruct the amount of sunlight, result in the plant by increasing leaf surface temperature (Rahul and Jain 2014), and, the effect on plant physiology and biochemistry (Rai 2016). Particulates with diameter lesser than the stomatal aperture can directly enter the sub-stomatal cavity, may result in stomata clogging (Vijaywargiya and Pandey 2003), which in turn could decrease the rate of carbon dioxide exchange (Darley 1966), carbon assimilation, transpiration (Singh and Rao 1982), and therefore net photosynthesis (Singh and Sthapak 1999). Studies have reported an increase in the number of epidermal cells and stomata, necrotic lesions, and death of epidermal cells in plant species exposed to particulate pollution (Jafri et al. 1979; Borka 1981; Shanmughavel 1997; Pal et al. 2000).

Plant leaves provide a large surface area for impingement, absorption, and accumulation of air pollutants. Vegetation cover can be used as an important means for filtering out suspended particulates in the air, especially in the high traffic density locations and industrial zones (Freer-Smith et al. 1997) for enhancement of urban air quality (Beckett et al. 1998; Freer-Smith et al. 2005). The capacity to filter atmospheric pollutant may vary from one species to other (Liu and Ding 2008). Sett (2017) suggested that cuticular and epidermal traits of the leaves in response to the dust pollution could be used as the bioindicator of the pollution levels.

3.4.5 Fluorides

Fluoride compounds have severe impacts on the animals and terrestrial and aquatic ecosystems (Weinstein and Davison 2004; Divan et al. 2008; Walna et al. 2013). Natural sources of fluoride emissions into the atmosphere are rock weathering, volcanic eruptions, or the marine environment (Barnard and Nordstrom 1982), while the principal anthropogenic sources are fertilizer production and application, household detergents, aluminum smelters, cement industries, ceramic and pottery works and glass manufacture (Cape et al. 2003). Common examples of fluoride pollutants are SiF₆, HF, CF₄, and F₂ in the gaseous phase and CaF₂, NH₃F, Ca₃AlF₆, AlF₆, NaF, and Na₂SiF₆ in the particulate phase. Hydrogen fluoride (HF), is the most hazardous fluoride pollutant. Fluorine emitted from anthropogenic sources in the gaseous phase is hydrolyzed under moist conditions to form HF. Hydrogen fluoride undergoes a rapid chemical reaction with silicates to form volatile SiF₆. HF also reacts with many materials forming nonvolatile, stable fluoride compounds (Kirk and Lester 1986). The distribution and deposition of its compounds are primarily governed by the magnitude of their emissions from the source, chemical reactivity of the fluoride compounds (Hara et al. 1998), size of the particulates, and meteorological conditions (Gasic et al. 2010; Scheringer 2009).

Damage to plants from fluoride compounds can happen at concentrations about 1000 times lower than those causing any visible effects on human health (Manins et al. 2001). Atmospheric HF can result in visible injury in plants at concentrations as low as $0.3 \ \mu m^{-3}$ (Cape et al. 2003). Fluoride contamination affects photosynthesis and other physiological processes in plants. It enters into the transpiration stream of plants through stomata or roots (Feng et al. 2003) and accumulates in leaf margins.

Symptoms of fluorine damage includes tip necrosis in confers that spreads to the base, while in broadleaf plants include marginal and tip necrosis that spread inward (Krupa 2001; Sharma and Kaur 2018). The vegetation contaminated by fluoride deposition if eaten by domestic animals may produce serious effects since these animals are typically vulnerable to fluoride (Shupe 1969). Studies have shown that roughly 96% of the ingested fluoride accumulates as mineral crystals in the teeth and bones in the bodies of animals (Zipkin 1972). Irrigation from fluoride contaminated water has been linked to a gradual decrease in the germination pattern of *Oryza sativa* L. Mondal (2017). In India, groundwater contamination of fluoride is a major environmental issue. Many studies have been conducted in India to document the effects of fluoride on plants (Mondal 2017; Sharma and Kaur 2018; Khound and Bharali 2018; Usham et al. 2018).

3.4.6 Peroxyacyl Nitrates

Peroxyacetic nitric anhydride ($CH_3COO_2NO_2$), also known as peroxyacetyl nitrate, is a powerful respiratory and eye irritant. Peroxyacetyl nitrate (PAN) is an important reservoir species of NO_x in the atmosphere, consequently leads to the formation of ozone and hydroxyl radicals in the lower troposphere (Singh and Hanst 1981). PAN has a longer residence time in the atmosphere and can be further transported and distributed to remote locations away from their source of origin (Mudd 1975; Teklemariam and Sparks 2004; Fischer et al. 2014). The presence of non-methane volatile organic compounds (NMVOCs) and NOx is a prerequisite for the formation of PAN, both these precursors can be made available from natural as well as anthropogenic sources (Roberts 2007; Fischer et al. 2014). The burning of fossil fuels is the major source of NOx production, with additional inputs from biomass burning, natural lighting, and soils.

In plants like bean leaves (*Phaseolis vulgaris* L.), PAN could induce the oxidization of sulfhydryl groups on enzymes required for the photophosphorylation, thus producing inhibitory effects on the cyclic photophosphorylation of chloroplasts (Koukol et al. 1967). Plants are susceptible to PAN pollution and show visible injury at much lower concentrations. Studies by Temple and Taylor (1983) have shown that PAN exposure of 14 nmol mol⁻¹ to petunia and tomato plants for 4-h duration initiated acute injury symptoms (Taylor 1969). In the susceptible plant species like lettuce and beet, the foliar injury was initiated at a threshold concentration of 15 nmol mol⁻¹ and for 4-h exposure, while severe injury resulted from 25–30 nmol mol⁻¹ concentration of PAN for 4-h exposure.

Besides major air pollutants, there are many minor pollutants such as hydrogen sulfide (H_2S), carbon monoxide (CO), vapors of metals like mercury, arsenic, cadmium, etc.; heavy metals bound in particulates; bromine (Br_2) and iodine (I_2); soot dust; chlorides of sodium, potassium and calcium, sodium sulfate dust; pesticides, insecticides, and herbicides which were known to significantly affect the biota health.

3.5 Impacts of Air Pollutants on Human Health

Humans are exposed to air pollutants directly via inhalation, ingestions, and through dermal contact or indirectly from ingestion of contaminated water, soil, or through the food chain. According to WHO, world over around 90% of the people were exposed to high levels of atmospheric pollutants in 2016. The air pollution alone kills an alarming 7 million people annually, out of which a whopping 4.2 million premature deaths are caused by outdoor air pollutants. A large proportion of this burden, estimated to be around 91%, is shared by third world countries mainly in South-East Asia and Western Pacific regions (WHO 2020). Short-term exposure to air pollutants exposure to humans may induce symptoms ranging from acute to severe breathing problems, wheezing, eyes burning, skin irritation, higher rates of hospitalization, whereas, the long-term exposure is closely associated with more complicated health issues leading to cardiovascular diseases, chronic asthma, effects on the nervous system, reduced activity of the immune system, cardiovascular mortality, carcinogenicity, etc. (Manisalidis et al. 2020). Traffic-related air pollution (TRAP) is the predominant source of air pollution in urban areas throughout the world. In rural regions, especially those in the developing countries, indoor domestic emissions from coal and biomass burning are the major pollution source (Laumbach and Kipen 2012). The occurrence of sporadic events like the Bhopal gas tragedy in 1984 in the Indian state of Madhya Pradesh that claimed around 2000 from the release of methyl isocyanate (MIC) from pesticide plant is another serious concern of air pollution exposure to humans, especially in the occupational environments. Many urban regions of India and China in recent decades have emerged as the new hotspots of the world, where a large proportion of the world's population lives and is exposed to the dangers of air pollution.

The majority of the air pollution-related health issues include respiratory and allergic diseases such as asthma, chronic obstructive pulmonary disease (COPD), pneumonia, skin irritation, and tuberculosis (Laumbach and Kipen 2012). Nations across the world including India have devised their National Ambient Air Quality Standards (NAAQS) to regulate the ambient concentrations for air pollutants of utmost concern. USEPA has identified six pollutant criteria air pollutants, i.e., SO₂, NO₂, CO, tropospheric O₃, suspended particulate matter, and lead for regular monitoring to maintain the ambient air quality of different regions. NO₂ is a respiratory irritant, as it penetrates deep in the lung. Its inhalation may induce respiratory diseases, coughing, wheezing, dyspnea, bronchospasm, and even pulmonary edema at higher ambient concentrations. It may affect the T-lymphocytes in the body as their concentrations exceed 2.0 ppm. Likewise, the SO₂ is a sensory irritant and penetrates deep into the lung converted into bisulfite and interacting with sensory receptors, causing bronchoconstriction. Its inhalation, particularly in the industrialized and high traffic density locations, induces respiratory irritation, bronchitis, mucus production, and bronchospasm, (Manisalidis et al. 2020). Air pollutants act as unspecific irritants and immunomodulators; studies by revealed that high pollution levels especially NO_x and PM₁₀ are positively correlated with increased risk of asthma among asthmatics in Taiwan. Chronic exposure to such

pollutants could further enhance the risk of lung cancer-sensitive individuals. In India, Singh et al. (2000) have observed severe respiratory symptoms and reduced forced expiratory volume among the Traffic Police personals of Jaipur City, Rajasthan due to high exposure of air pollutants. Ambient concentrations of $PM_{2.5}$ and PM_{10} are reported to exhibit adverse health effects on humans such as heart disease, stroke, blood pressure, and cardiovascular diseases (Brook et al. 2010; Lee et al. 2014). Bourdrel et al. (2017) in their review work on cardiovascular effects of air pollution found that a 10 µg/m³ increase in long-term exposure to $PM_{2.5}$ was associated with an 11% increase in cardiovascular mortality. Exposure to carbon monoxide affects the oxygen-carrying capacity of the blood resulting in impaired concentration, slow reflexes, confusion, and in extreme cases deaths. In India, Suresh and Das, 2003 reported that the levels of various antioxidants in the RBC lysate such as catalase, superoxide dismutase, and glutathione peroxidase were found to be low in a traffic policeman in the city of Hyderabad.

Besides, exposure to air pollutants in particular heavy metals and dioxins adversely affects the central nervous system, and is closely associated with cognitive dysfunction, Alzheimer's disease and Parkinson's disease, structural brain effects, and white matter injury (Genc et al. 2012; Babadjouni et al. 2017). Dioxins are known to decrease nerve conduction velocity and impair the mental development of children (Thömke et al. 1999; Walkowiak et al. 2001). Exposure of As, Pb, and Hg has been linked to symptoms such as memory disturbances, anger, fatigue, sleep disorders, blurred vision, slurred speech, and hand tremors (Ewan and Pamphlett 1996; Ratnaike 2003). Maternal exposure to air pollutants affects the developing fetus (Schell et al. 2006). Maternal exposure to lead increases the risks of spontaneous abortion and reduced fetal growth, congenital malformations (Bellinger 2005). Dioxins can be transferred to the developing fetus from the mother via the placenta, they act as endocrine disruptors and affect the growth and development of the central nervous system of the fetus (Wang et al. 2004). Oxygen supply to the fetus is maintained from the mother's blood. Breathing polluted air during pregnancy can result in premature birth, low birth weight impaired lung development, increased later respiratory morbidity, early alterations in immune development, and can increase the baby's risk of contracting illnesses (Proietti et al. 2013; Korten et al. 2017; La Marca and Gava 2018). Besides, the prenatal exposure of PAHs affects the child's behavior and neurodevelopment, a decrease of IQ, increase of attentiondeficit hyperactivity disorder, a decrease of brain-derived neurotrophic factor, and reduction of left hemisphere white matter (Sram et al. 2017). Furthermore, epidemiological evidence suggests that periods of elevated air pollution affect the semen quality and damage sperm DNA (Dejmek et al. 2000; Rubes et al. 2005).

Besides the above-mentioned health implications, many of the air pollutants are known to be potential carcinogens that can either initiate and/or promote carcinogenesis by a diverse number of mechanisms such as DNA adducts and DNA protein cross-links formation, epigenetic alterations, regenerative cell proliferation, oxidative stress, inhibition of DNA repair, and altered cell cycle regulation. Common air carcinogens are formaldehyde, benzene, PAHs, ETS (environmental tobacco smoke), etc.

3.6 Conclusions

Effects of air pollution on plant and animal biota and natural and man-made materials are manifested directly as well as indirectly. The severity of the impacts of pollutants is variable and differs from one ecosystem to another and one organism to another. Emissions of such pollutants also strongly interfere with the atmospheric processes at local, regional, or global scales. Common atmospheric impacts of air pollutants are greenhouse effect and global warming; stratospheric; ozone depletion; atmospheric deposition, and acid rain; suppression of rainfall; atmospheric visibility reduction, etc. Developing economies are at higher risks of their owing to higher abatement costs associated with it. Even a marginal increase of $10 \,\mu\text{g/m}^3$ of ambient PM_{10} levels in Asian cities could lead to a surge in the mortality rate by 0.6%, which is significantly higher than the trends estimated for Western cities. Plants are highly susceptible to direct damages caused by contacting the air pollutants and indirectly from the changing state of the atmosphere, e.g., global warming, acid rain, stratospheric ozone depletion, etc. At the intergovernmental level several efforts have been made to regulate the air pollutants, e.g., the Geneva Convention on Long-range transboundary air pollution (1979) to regulate SO_x , NO_x , persistent organic pollutants, non-methane volatile organic compounds, tropospheric ozone, heavy metals, particulate matters, and black carbon; Montreal protocol to regulate ozonedepleting substances; the Paris agreement for control of GHGs, etc. Despite these efforts, control of air pollutants is still a far-off dream for policymakers, especially in developing economies like India and China, which are now the new pollution epicenters of the world. Therefore, timely policy interventions based on scientific facts must be made to regulate the air pollutants world over and minimize their impacts on living biota and atmospheric processes.

References

- Aerts R, Bobbink R (1999) The impact of atmospheric nitrogen deposition on vegetation processes in terrestrial, non-forest ecosystems. In: The impact of nitrogen deposition on natural and seminatural ecosystems. Springer, Dordrecht, pp 85–122
- Agrawal M, Verma M (1997) Amelioration of sulphur dioxide phytotoxicity in wheat cultivars by modifying NPK nutrients. J Environ Manag 49(2):231–244
- Agrawal M, Krizek DT, Agrawal SB, Kramer GF, Lee EH, Mirecki RM, Rowland RA (1993) Influence of inverse day/night temperature on ozone sensitivity and selected morphological and physiological responses of cucumber. J Am Soc Hortic Sci 118(5):649–654

Allègre CJ, Schneider SH (1994) The evolution of the Earth. Sci Am 271(4):66-75

- Alonso R, Elvira S, Castillo FJ, Gimeno BS (2001) Interactive effects of ozone and drought stress on pigments and activities of antioxidative enzymes in Pinus halepensis. Plant Cell Environ 24(9):905–916
- Ashmore MR (2005) Assessing the future global impacts of ozone on vegetation. Plant Cell Environ 28(8):949–964
- Babadjouni RM, Hodis DM, Radwanski R, Durazo R, Patel A, Liu Q, Mack WJ (2017) Clinical effects of air pollution on the central nervous system; a review. J Clin Neurosci 43:16–24
- Balestrini R, Galli L, Tartari G (2000) Wet and dry atmospheric deposition at prealpine and alpine sites in northern Italy. Atmos Environ 34(9):1455–1470

- Bambawale OM (1986) Evidence of ozone injury to a crop plant in India. Atmos Environ 20(7): 1501–1503
- Barnard WR, Nordstrom DK (1982) Flouride in precipitation—II. Implications for the geochemical cycling of fluorine. Atmos Environ (1967) 16(1):105–111
- Beckett KP, Freer-Smith PH, Taylor G (1998) Urban woodlands: their role in reducing the effects of particulate pollution. Environ Pollut 99(3):347–360
- Beier CM, Caputo J, Lawrence GB, Sullivan TJ (2017) Loss of ecosystem services due to chronic pollution of forests and surface waters in the Adirondack region (USA). J Environ Manag 191: 19–27
- Bellinger DC (2005) Teratogen update: lead and pregnancy. Birth Defects Res A Clin Mol Teratol 73:409
- Bernstein JA, Alexis N, Barnes C, Bernstein IL, Nel A, Peden D et al (2004) Health effects of air pollution. J Allergy Clin Immunol 114(5):1116–1123
- Bhatia A, Ghosh A, Kumar V, Tomer R, Singh SD, Pathak H (2011) Effect of elevated tropospheric ozone on methane and nitrous oxide emission from rice soil in north India. Agric Ecosyst Environ 144(1):21–28
- Bignal KL, Ashmore MR, Headley AD, Stewart K, Weigert K (2007) Ecological impacts of air pollution from road transport on local vegetation. Appl Geochem 22(6):1265–1271
- Blankenship RE (2010) Early evolution of photosynthesis. Plant Physiol 154(2):434-438
- Borka G (1981) Effect of cement-kiln dust on the maize plant. Acta Agron Acad Sci Hung Budapest 30(3–4):289–295
- Borys RD, Lowenthal DH, Cohn SA, Brown WO (2003) Mountaintop and radar measurements of anthropogenic aerosol effects on snow growth and snowfall rate. Geophys Res Lett 30(10)
- Bourdrel T, Bind MA, Béjot Y, Morel O, Argacha JF (2017) Cardiovascular effects of air pollution. Arch Cardiovasc Dis 110(11):634–642
- Brimblecombe P (1987a) The big smoke. Methuen, London
- Brimblecombe P (1987b) The big smoke: a history of air pollution in London since medieval times. Methuen, London
- Brook RD, Rajagopalan S, Pope CA III, Brook JR, Bhatnagar A, Diez-Roux AV et al (2010) Particulate matter air pollution and cardiovascular disease: an update to the scientific statement from the American Heart Association. Circulation 121(21):2331–2378
- Brown SG, Hyslop NP, Roberts PT, McCarthy MC, Lurmann FW (2006) Wintertime vertical variations in particulate matter (PM) and precursor concentrations in the San Joaquin Valley during the California Regional coarse PM/fine PM Air Quality Study. J Air Waste Manage Assoc 56(9):1267–1277
- Cape JN, Fowler D, Davison A (2003) Ecological effects of sulfur dioxide, fluorides, and minor air pollutants: recent trends and research needs. Environ Int 29(2–3):201–211
- Che HC et al (2016) Characterization and parameterization of aerosol cloud condensation nuclei activation under different pollution conditions. Sci Rep 6:24497. https://doi.org/10.1038/srep24497
- Chen J, Zhao CS, Ma N, Liu PF, Göbel T, Hallbauer E et al (2012) A parameterization of low visibilities for hazy days in the North China Plain. Atmos Chem Phys 12(11):4935–4950
- Chen X, Li X, Yuan X, Zeng G, Liang J, Li X et al (2018) Effects of human activities and climate change on the reduction of visibility in Beijing over the past 36 years. Environ Int 116:92–100
- Chevuturi A, Dimri AP, Thayyen RJ (2018) Climate change over Leh (Ladakh), India. Theor Appl Climatol 131(1–2):531–545
- Chung KF, Zhang J, Zhong N (2011) Outdoor air pollution and respiratory health in Asia. Respirology 16(7):1023–1026
- Conti ME (2008) Lichens as bioindicators of air pollution. WIT Trans State Art Sci Eng 30
- Correia AW, Pope CA III, Dockery DW, Wang Y, Ezzati M, Dominici F (2013) The effect of air pollution control on life expectancy in the United States: an analysis of 545 US counties for the period 2000 to 2007. Epidemiology (Cambridge, MA) 24(1):23
- Cutchis P (1974) Stratospheric ozone depletion and solar ultraviolet radiation on earth. Science 184(4132):13–19
- Darley EF (1966) Studies on the effect of cement-kiln dust on vegetation. J Air Pollut Control Assoc 16(3):145–150

- De Marco A, Proietti C, Anav A, Ciancarella L, D'Elia I, Fares S et al (2019) Impacts of air pollution on human and ecosystem health, and implications for the National Emission Ceilings Directive: insights from Italy. Environ Int 125:320–333
- Dejmek J, Jelínek R, Solansky' I, Benes I, Srám RJ (2000) Fecundability and parental exposure to ambient sulfur dioxide. Environ Health Perspect 108(7):647–654
- Della Torre G, Ferranti F, Lupattelli M, Pocceschi N, Figoli A, Nali CRISTINA, Lorenzini GIACOMO (1998) Effects of ozone on morpho-anatomy and physiology of Hedera helix. Chemosphere 36(4–5):651–656
- Divan AM, Oliva MA, Ferreira FA (2008) Dispersal pattern of airborne emissions from an aluminium smelter in Ouro Preto, Brazil, as expressed by foliar fluoride accumulation in eight plant species. Ecol Indic 8(5):454–461
- Du E, Dong D, Zeng X, Sun Z, Jiang X, de Vries W (2017) Direct effect of acid rain on leaf chlorophyll content of terrestrial plants in China. Sci Total Environ 605:764–769
- Dwivedi AK, Tripathi BD (2008) Effect of ambient air sulphur dioxide on sulphate accumulation in plants. J Environ Biol 29(3):377–379
- El Morabet R (2019) Effects of outdoor air pollution on human health. In: Reference module in earth systems and environmental sciences. https://doi.org/10.1016/B978-0-12-409548-9.11012-7
- Environmental Protection Agency (2008) Clean Air Act. In: U.S. Code, 03.2011. http://www.epa. gov/air/caa/. Accessed 10 Sept 2020
- Ewan KB, Pamphlett R (1996) Increased inorganic mercury in spinal motor neurons following chelating agents. Neurotoxicology 17(2):343–349
- Fang Y, Naik V, Horowitz LW, Mauzerall DL (2013) Air pollution and associated human mortality: the role of air pollutant emissions, climate change and methane concentration increases from the preindustrial period to present. Atmos Chem Phys 13(3):1377–1394
- Farman JC, Gardiner BG, Shanklin JD (1985) Large losses of total ozone in Antarctica reveal seasonal ClOx/NOx interaction. Nature 315(6016):207–210
- Farooq M, Masood A, Beg MU (1985) Effect of acute exposure of sulphur dioxide on the metabolism of Holoptelea integrifolia plants. Environ Pollut Ser A Ecol Biol 39(3):197–205
- Feng YW, Ogura N, Feng ZW, Zhang FZ, Shimizu H (2003) The concentrations and sources of fluoride in atmospheric depositions in Beijing, China. Water Air Soil Pollut 145(1–4):95–107
- Fischer EV, Jacob DJ, Yantosca RM, Sulprizio MP, Millet DB, Mao J et al (2014) Atmospheric peroxyacetyl nitrate (PAN): a global budget and source attribution. Atmos Chem Phys 14(5): 2679–2698
- Forcetto ALS, Daemme LC (2017) Environmental impact of motorcycle replacement exhaust systems in Sao Paulo City. WIT Trans Ecol Environ 210:641–652
- Freer-Smith PH, Holloway S, Goodman A (1997) The uptake of particulates by an urban woodland: site description and particulate composition. Environ Pollut 95(1):27–35
- Freer-Smith PH, Beckett KP, Taylor G (2005) Deposition velocities to Sorbus aria, Acer campestre, Populus deltoides × trichocarpa 'Beaupré', Pinus nigra and × Cupressocyparis leylandii for coarse, fine and ultra-fine particles in the urban environment. Environ Pollut 133(1):157–167
- Fumagalli I, Gimeno BS, Velissariou D, De Temmerman L, Mills G (2001) Evidence of ozoneinduced adverse effects on crops in the Mediterranean region. Atmos Environ 35(14):2583–2587
- Gaba A, Iordache SF (2011) Reduction of air pollution by combustion processes. In: The impact of air pollution on health, economy, environment and agricultural sources. p 119
- Gambaro A, Radaelli M, Piazza R, Stortini AM, Contini D, Belosi F et al (2009) Organic micropollutants in wet and dry depositions in the Venice Lagoon. Chemosphere 76(8):1017–1022
- Gao Y, Marsay CM, Yu S, Fan S, Mukherjee P, Buck CS, Landing WM (2019) Particle-size variability of aerosol iron and impact on iron solubility and dry deposition fluxes to the Arctic ocean. Sci Rep 9(1):1–11
- Garg SS, Kumar N, Das G (2000) Effects of the Bansal Ramraj mill dust on vegetation and health at Jaitwara, district Satna. Indian J Environ Prot 20(5):326–328
- Garrec J-P (2020) What is the impact of air pollutants on vegetation. https://www.encyclopedieenvironnement.org/en/life/impact-air-pollutants-on-vegetation/. Accessed 18 Sept 2020
- Gasic B, MacLeod M, Scheringer M, Hungerbuhler K (2010) Assessing the impact of weather events at mid-latitudes on the atmospheric transport of chemical pollutants using a 2-dimensional multimedia meteorological model. Atmos Environ 44(35):4489–4496

- Genc S, Zadeoglulari Z, Fuss SH, Genc K (2012) The adverse effects of air pollution on the nervous system. J Toxicol 2012:782462
- Ghazali NA, Ramli NA, Yahaya AS, Yusof NFFM, Sansuddin N, Al Madhoun WA (2010) Transformation of nitrogen dioxide into ozone and prediction of ozone concentrations using multiple linear regression techniques. Environ Monit Assess 165(1–4):475–489
- Gheorghe IF, Ion B (2011) The effects of air pollutants on vegetation and the role of vegetation in reducing atmospheric pollution. In: Khallaf M (ed) The impact of air pollution on health, economy, environment and agricultural sources, pp 241–280

Godish T, Davis WT, Fu JS (2014) Air quality. CRC Press

- Government of India, Ministry of Environment, Forest and Climate Change (2019) National Clean Air Programme (NCAP). http://moef.gov.in/wed-2019-2/national-clean-air-programme-ncap/. Accessed 30 Oct 2020
- Graff Zivin J, Neidell M (2013) Environment, health, and human capital. J Econ Lit 51(3):689-730
- Gu S, Yang J, Woodward A, Li M, He T, Wang A et al (2017) The short-term effects of visibility and haze on Mortality in a coastal city of China: a time-series study. Int J Environ Res Public Health 14(11):1419
- Guidi LUCIA, Nali CRISTINA, Lorenzini GIACOMO, Soldatini GF (1998) Photosynthetic response to ozone of two poplar clones showing different sensitivity. Chemosphere 36(4–5):657–662
- Gupta V, Jain MK (2018) Investigation of multi-model spatiotemporal mesoscale drought projections over India under climate change scenario. J Hydrol 567:489–509
- Hansen J, Nazarenko L (2004) Soot climate forcing via snow and ice albedos. Proc Natl Acad Sci 101(2):423–428
- Hara H, Yabuuchi K, Higashida M, Ogawa M (1998) Determination of free and total fluoride in rain water using a continuous-flow system equipped with a fluoride ion-selective electrode detector. Anal Chim Acta 364(1–3):117–123
- Heck WW, Dunning JA (1967) The effects of ozone on tobacco and pinto bean as conditioned by several ecological factors. J Air Pollut Control Assoc 17(2):112–114
- Hoshika Y, Omasa K, Paoletti E (2013) Both ozone exposure and soil water stress are able to induce stomatal sluggishness. Environ Exp Bot 88:19–23
- Hyslop NP (2009) Impaired visibility: the air pollution people see. Atmos Environ 43(1):182-195
- IPCC (2007) Climate change 2007: impacts, adaptation and vulnerability. In: Parry ML, Canziani OF, Palutikof JP, van der Linden PJ, Hanson CE (eds) Contribution of Working Group II to the Fourth Assessment. Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge
- IPCC (2018) Summary for Policymakers. In: Global warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. https://www.ipcc.ch/ sr15/. Accessed 16 Sept 2020
- IQAIR (2019) World's most polluted cities 2019 (PM2.5) https://www.iqair.com/world-mostpolluted-cities. Accessed 30 Oct 2020
- Jacobson MZ, Jacobson MZ (2002) Atmospheric pollution: history, science, and regulation. Cambridge University Press
- Jafri S, Srivastava K, Ahmad KJ (1979) Environmental pollution and epidermal structure in Syzygium cumini L. Skeel. Indian J Air pollut Control 2:74–77
- Johansson O, Palmqvist K, Olofsson J (2012) Nitrogen deposition drives lichen community changes through differential species responses. Glob Chang Biol 18(8):2626–2635
- Kampa M, Castanas E (2008) Human health effects of air pollution. Environ Pollut 151(2):362-367
- Kapil S (2019) India emits the most sulphur dioxide in the world. https://www.downtoearth.org.in/ news/air/india-emits-the-most-sulphur-dioxide-in-the-world-66230. Accessed 10 Sept 2020
- Katulski RJ, Namieśnik J, Sadowski J, Stefański J, Wardencki W (2011) Monitoring of gaseous air pollution. In: Khallaf M (ed) The impact of air pollution on health, economy, environment and agricultural sources, pp 241–280
- Khound NJ, Bharali RK (2018) Biosorption of fluoride from aqueous medium by Indian sandalwood (Santalum Album) leaf powder. J Environ Chem Eng 6(2):1726–1735

- Kirk PWW, Lester JN (1986) Halogen compounds. In: Harrison RM, Perry R (eds) Handbook of air pollution analysis, 2nd edn. Chapman & Hall, London, pp 425–462
- Knoll AH (2015) Life on a young planet: the first three billion years of evolution on Earth-updated edition. Princeton University Press
- Korten I, Ramsey K, Latzin P (2017) Air pollution during pregnancy and lung development in the child. Paediatr Respir Rev 21:38–46
- Koukol J, Dugger WM, Palmer RL (1967) Inhibitory effect of peroxyacetyl nitrate on cyclic photophosphorylation by chloroplasts from black valentine bean leaves. Plant Physiol 42(10): 1419–1422
- Krupa S (2001) Fluorine. In: Maloy OC, Murray TD (eds) Encyclopedia of plant pathology, p 465
- Krupa SV, Manning WJ (1988) Atmospheric ozone: formation and effects on vegetation. Environ Pollut 50(1–2):101–137
- Kumar P, Yadav S (2016) Seasonal variations in water soluble inorganic ions, OC and EC in PM10 and PM> 10 aerosols over Delhi: influence of sources and meteorological factors. Aerosol Air Qual Res 16(5):1165–1178
- Kumar P, Kumar R, Yadav S (2016a) Water-soluble ions and carbon content of size-segregated aerosols in New Delhi, India: direct and indirect influences of firework displays. Environ Sci Pollut Res 23(20):20749–20760
- Kumar S, Raizada A, Biswas H, Srinivas S, Mondal B (2016b) Application of indicators for identifying climate change vulnerable areas in semi-arid regions of India. Ecol Indic 70:507–517
- La Marca L, Gava G (2018) Air pollution effects in pregnancy. In: Capello F, Gaddi A (eds) Clinical handbook of air pollution-related diseases. Springer, Cham. https://doi.org/10.1007/ 978-3-319-62731-1_26
- Laumbach RJ, Kipen HM (2012) Respiratory health effects of air pollution: update on biomass smoke and traffic pollution. J Allergy Clin Immunol 129(1):3–11
- Lee ACL (1973) A study of the continuum absorption within the 8–13 µm atmospheric window. Q J R Meteorol Soc 99(421):490–505
- Lee BJ, Kim B, Lee K (2014) Air pollution exposure and cardiovascular disease. Toxicol Res 30(2): 71–75
- Lindberg SE, Harriss RC, Turner RR (1982) Atmospheric deposition of metals to forest vegetation. Science 215(4540):1609–1611
- Lindsey R (2020) Climate change: atmospheric carbon dioxide. https://www.climate.gov/news-features/ understanding-climate/climate-change-atmospheric-carbon-dioxide. Accessed 14 Sept 2020
- Liu YJ, Ding HUI (2008) Variation in air pollution tolerance index of plants near a steel factory: implication for landscape-plant species selection for industrial areas. WSEAS Trans Environ Dev 4(1):24–32
- Liu X, Fu Z, Zhang B, Zhai L, Meng M, Lin J et al (2018) Effects of sulfuric, nitric, and mixed acid rain on Chinese fir sapling growth in Southern China. Ecotoxicol Environ Saf 160:154–161
- Lovett GM, Kinsman JD (1990) Atmospheric pollutant deposition to high-elevation ecosystems. Atmos Environ Part A Gen Top 24(11):2767–2786
- Magaino S (1997) Corrosion rate of copper rotating-disk-electrode in simulated acid rain. Electrochim Acta 42(3):377–382
- Maiti S, Jha SK, Garai S, Nag A, Bera AK, Paul V et al (2017) An assessment of social vulnerability to climate change among the districts of Arunachal Pradesh, India. Ecol Indic 77:105–113
- Majernik O, Mansfield TA (1970) Direct effect of SO 2 pollution on the degree of opening of stomata. Nature 227(5256):377–378
- Manins P, Allan R, Beer T, Fraser P, Holper P, Suppiah R, Walsh K (2001) Atmosphere, Australia State of the Environment Report 2001 (Theme Report). CSIRO Publishing on Behalf of the Department of the Environment and Heritage, Canberra
- Manisalidis I, Stavropoulou E, Stavropoulos A, Bezirtzoglou E (2020) Environmental and health impacts of air pollution: a review. Front Public Health 8:14. https://doi.org/10.3389/fpubh.2020. 00014
- Mansfield TA, Whitmore ME, Law RM (1982) Effects of nitrogen oxides on plants: two case studies. In: Air pollution by nitrogen oxides, vol 21. Elsevier Scientific Publishing Company. https://doi.org/10.1016/B978-0-444-42127-2.50051-8

- Menon S, Denman KL, Brasseur G, Chidthaisong A, Ciais P, Cox PM et al (2007) Couplings between changes in the climate system and biogeochemistry (No. LBNL-464E). Lawrence Berkeley National Lab. (LBNL), Berkeley, CA (United States)
- Mina U, Kumar P, Varshney CK (2010) Effect of ozone stress on different growth stages of potato (Solanum tuberosum). Phyton Horn 49(2):253–266
- Mondal NK (2017) Effect of fluoride on photosynthesis, growth and accumulation of four widely cultivated rice (Oryza sativa L.) varieties in India. Ecotoxicol Environ Saf 144:36–44
- Morales-Baquero R, Pulido-Villena E, Reche I (2013) Chemical signature of Saharan dust on dry and wet atmospheric deposition in the south-western Mediterranean region. Tellus B Chem Phys Meteorol 65(1):18720
- Mudd JB (1975) Sulphur dioxide. In: Mudd JB, Kozlowski TT (eds) Responses of plants to air pollution. Academic, London, pp 9–22
- Mudd JB (ed) (2012) Responses of plants to air pollution. Elsevier
- Murphy DM, Solomon S, Portmann RW, Rosenlof KH, Forster PM, Wong T (2009) An observationally based energy balance for the Earth since 1950. J Geophys Res Atmos 114(D17)
- Newman PA, Kawa SR, Nash ER (2004) On the size of the Antarctic ozone hole. Geophys Res Lett 31(21)
- Ohara TAHK, Akimoto H, Kurokawa JI, Horii N, Yamaji K, Yan X, Hayasaka T (2007) An Asian emission inventory of anthropogenic emission sources for the period 1980? 2020
- Pachauri RK, Reisinger A (2007) IPCC fourth assessment report. IPCC, Geneva
- Pacyna JM (2008) Atmospheric deposition. In: Jørgensen SE, Fath BD (eds) Encyclopedia of ecology. Academic Press, pp 275–285. https://doi.org/10.1016/B978-008045405-4.00258-5
- Pal A, Kulshreshtha K, Ahmad KJ et al (2000) Changes in leaf surface structures of two avenue tree species caused by auto exhaust pollution. J Environ Biol 21(1):15–21
- Pandey J, Agrawal M (1994) Growth responses of tomato plants to low concentrations of sulphur dioxide and nitrogen dioxide. Sci Hortic 58(1–2):67–76
- Paruelo JM, Beltrán A, Jobbágy E, Sala OE, Golluscio RA (1998) The climate of Patagonia: general patterns and controls on biotic processes. Ecol Austral 8(02):085–101
- Pöschl U (2005) Atmospheric aerosols: composition, transformation, climate and health effects. Angew Chem Int Ed 44(46):7520–7540
- Prajapati SK, Tripathi BD (2007) Biomonitoring trace-element levels in PM10 released from vehicles using leaves of Saraca indica and Lantana camara. AMBIO 36(8):704–705
- Proietti E, Röösli M, Frey U, Latzin P (2013) Air pollution during pregnancy and neonatal outcome: a review. J Aerosol Med Pulm Drug Deliv 26(1):9–23
- Proietti C, Anav A, De Marco A, Sicard P, Vitale M (2016) A multi-sites analysis on the ozone effects on Gross Primary Production of European forests. Sci Total Environ 556:1–11
- Rahul J, Jain MK (2014) An investigation in to the impact of particulate matter on vegetation along the national highway: a review. Res J Environ Sci 8:356–372
- Rai PK (2016) Impacts of particulate matter pollution on plants: implications for environmental biomonitoring. Ecotoxicol Environ Saf 129:120–136
- Ramanathan VCPJ, Crutzen PJ, Kiehl JT, Rosenfeld D (2001) Aerosols, climate, and the hydrological cycle. Science 294(5549):2119–2124
- Rao MP, Cook BI, Cook ER, D'Arrigo RD, Krusic PJ, Anchukaitis KJ et al (2017) European and Mediterranean hydroclimate responses to tropical volcanic forcing over the last millennium. Geophys Res Lett 44(10):5104–5112
- Ratnaike RN (2003) Acute and chronic arsenic toxicity. Postgrad Med J 79(933):391-396
- Roberts JM (2007) PAN and related compounds. In: Volatile organic compounds in the atmosphere. Blackwell Publishing Ltd, pp 221–268
- Root HT, Geiser LH, Jovan S, Neitlich P (2015) Epiphytic macrolichen indication of air quality and climate in interior forested mountains of the Pacific Northwest, USA. Ecol Indic 53:95–105
- Rosenfeld D (1999) TRMM observed first direct evidence of smoke from forest fires inhibiting rainfall. Geophys Res Lett 26(20):3105–3108
- Rosenfeld D (2000) Suppression of rain and snow by urban and industrial air pollution. Science 287(5459):1793–1796

- Roux G, Hansson SV, Claustres A (2016) Inorganic Chemistry in the Mountain Critical Zone: are the mountain water towers of contemporary society under threat by trace contaminants? In: Developments in earth surface processes, vol 21. Elsevier, pp 131–154
- Rowland FS (2009) Stratospheric ozone depletion. In Twenty years of ozone decline. Springer, Dordrecht, pp 23–66
- Roy A, Chatterjee A, Tiwari S, Sarkar C, Das SK, Ghosh SK, Raha S (2016) Precipitation chemistry over urban, rural and high altitude Himalayan stations in eastern India. Atmos Res 181:44–53
- Rubes J, Selevan SG, Evenson DP, Zudova D, Vozdova M, Zudova Z, Robbins WA, Perreault SD (2005) Episodic air pollution is associated with increased DNA fragmentation in human sperm without other changes in semen quality. Hum Reprod 20(10):2776–2783
- Ruijrok W, Davidson CI, Nicholson KW (1995) Dry deposition of particles. Tellus B 47(5):587-601
- Runeckles VC, Chevone BI, Lefohn A (1992) Surface level ozone exposures and their effects on vegetation. In: Lefohn AS (ed) Crop responses to ozone. Lewis Publishers, Inc., Chelsea, pp 189–270
- SAFAR-India (2020) System of Air Quality and Weather Forecasting And Research. http://safar. tropmet.res.in/. Accessed 30 Oct 2020
- Saitanis CJ, Karandinos MG (2001) Effects of ozone on chlorophyll and quantum yield of tobacco (Nicotiana tabacum L.) varieties. Chemosphere 42:945–953
- Sam AS, Padmaja SS, Kächele H, Kumar R, Müller K (2020) Climate change, drought and rural communities: understanding people's perceptions and adaptations in rural eastern India. Int J Disaster Risk Reduct 44:101436
- Sarkar A, Agrawal SB (2012) Evaluating the response of two high yielding Indian rice cultivars against ambient and elevated levels of ozone by using open top chambers. J Environ Manag 95:S19–S24
- Saxena P, Sonwani S (2019a) Primary criteria air pollutants: environmental health effects. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 49–82
- Saxena P, Sonwani S (2019b) Criteria air pollutants: chemistry, sources and sinks. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 7–48
- Saxena P, Sonwani S (2019c) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021a) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Saxena P, Sonwani S, Gupta AK (2021b) Assessment of Air Pollution Tolerance Index (APTI) for selected ornamental plants in urban metropolis of Northern India. Asian J Chem 33(9):2150– 2156
- Saxena P, Srivastava A, Tyagi M, Kaur S (2019) Impact of tropospheric ozone on plant metabolism-a review. Pollut Res 38(1):175–180
- Schell LM, Gallo MV, Denham M, Ravenscroft J (2006) Effects of pollution on human growth and development: an introduction. J Physiol Anthropol 25:103
- Scheringer M (2009) Long-range transport of organic chemicals in the environment. Environ Toxicol Chem 28(4):677–690
- Schiermeier Q (2007) Pollution decreases rainfall. Nature. https://doi.org/10.1038/news070305-11
- Schmidt GA, Ruedy R, Hansen JE, Aleinov I, Bell N, Bauer M et al (2006) Present-day atmospheric simulations using GISS ModelE: comparison to in situ, satellite, and reanalysis data. J Clim 19(2):153–192
- Schneider SH (1989) The greenhouse effect: science and policy. Science 243(4892):771-781
- Seinfeld JH, Pandis SN (2016) Atmospheric chemistry and physics: from air pollution to climate change. Wiley
- Sett R (2017) Responses in plants exposed to dust pollution. Hortic Int J [Internet] 1(2):53-56
- Seyyednejad SM, Niknejad M, Koochak H (2011) A review of some different effects of air pollution on plants. Res J Environ Sci 5(4):302
- Shanmughavel P (1997) Effect of air pollution of tobacco (*Nicotiana tobacum*) in agricultural fields. Ecol Environ Conserv 2:173–174

- Sharma R, Kaur R (2018) Insights into fluoride-induced oxidative stress and antioxidant defences in plants. Acta Physiol Plant 40(10):181
- Sharma RK, Agrawal M, Marshall FM (2008) Atmospheric deposition of heavy metals (Cu, Zn, Cd and Pb) in Varanasi city, India. Environ Monit Assess 142(1–3):269–278
- Sheng Q, Zhu Z (2019) Effects of nitrogen dioxide on biochemical responses in 41 garden plants. Plants 8(2):45
- Sheu BH, Liu CP (2003) Air pollution impacts on vegetation in Taiwan. In: Air pollution impacts on crops and forests–a global assessment, pp 145–163
- Shieber (2019) CO2 in the atmosphere just exceeded 415 parts per million for the first time in human history. https://techcrunch.com/2019/05/12/co2-in-the-atmosphere-just-exceeded-415parts-per-million-for-the-first-time-in-human-history/. Accessed 31 Aug 2019
- Shupe JL (1969) Fluorosis of livestock, air quality monograph 69-4 69 4. American Petroleum Institute, New York
- Singh HB, Hanst PL (1981) Peroxyacetyl nitrate (PAN) in the unpolluted atmosphere: an important reservoir for nitrogen oxides. Geophys Res Lett 8:941–944. https://doi.org/10.1029/ GL008i008p00941
- Singh M, Rao DN (1982) The influence of ozone and sulphur dioxide on *Cicer arietinum L*. J Indian Bot Soc 61:51–58
- Singh P, Sthapak J (1999) Reduction in protein contents in a few plants as indicators of air pollution. Pollut Res 18(3):281–283
- Singh PK, Schaefer AL, Parsek MR, Moninger TO, Welsh MJ, Greenberg EP (2000) Quorumsensing signals indicate that cystic fibrosis lungs are infected with bacterial biofilms. Nature 407 (6805):762–764
- Singh AA, Fatima A, Mishra AK, Chaudhary N, Mukherjee A, Agrawal M, Agrawal SB (2018) Assessment of ozone toxicity among 14 Indian wheat cultivars under field conditions: growth and productivity. Environ Monit Assess 190(4):190
- Smith VH, Tilman GD, Nekola JC (1999) Eutrophication: impacts of excess nutrient inputs on freshwater, marine, and terrestrial ecosystems. Environ Pollut 100(1–3):179–196
- Soldatini GF, Nali C, Guidi L, Lorenzini G (1998) Photosynthesis of Hedera canariensis var. azorica variegated leaves as affected by ozone. Photosynthetica 35(2):247–253
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021a) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3:690444, 1–20
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021b) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3:690444, 1–20
- Sonwani S, Saxena P (eds) (2022) Greenhouse gases: sources, sinks and mitigation. Springer-Nature, Singapore, pp 1–255
- Sosa-Echeverría R, Alarcón-Jimémez AL, Torres-Barrera MDC, Jaimes-Palomera M, Retama-Hernández A, Sánchez-Álvarez P et al (2019) Spatial and temporal variation of acid rain in the Mexico City Metropolitan Zone. Atmósfera 32(1):55–69
- Sram RJ, Veleminsky M, Veleminsky M, Stejskalová J (2017) The impact of air pollution to central nervous system in children and adults. Neuroendocrinol Lett 38(6):389–396
- Suresh Y, Das UN (2003) Long-chain polyunsaturated fatty acids and chemically induced diabetes mellitus: effect of ω-3 fatty acids. Nutrition 19(3):213–228
- Taylor OC (1969) Importance of peroxyacetyl nitrate (PAN) as a phytotoxic air pollutant. J Air Pollut Control Assoc 19:347–351
- Teklemariam TA, Sparks JP (2004) Gaseous fluxes of peroxyacetyl nitrate (PAN) into plant leaves. Plant Cell Environ 27(9):1149–1158
- Temple PJ, Taylor OC (1983) World-wide ambient measurements of peroxyacetyl nitrate (PAN) and implications for plant injury. Atmos Environ 17:1583–1587
- The World Bank (2016, 8 Sept) Air pollution deaths cost global economy us\$225 billion. https:// www.worldbank.org/en/news/press-release/2016/09/08/air-pollutiondeaths-cost-global-econ omy-225-billion. Accessed 17 Feb 2020
- Thömke F, Jung D, Besser R, Röder R, Konietzko J, Hopf HC (1999) Increased risk of sensory neuropathy in workers with chloracne after exposure to 2, 3, 7, 8-polychlorinated dioxins and furans. Acta Neurol Scand 100(1):1–5
- Tinsley BA, Yu F (2003) Atmospheric ionization and clouds as links between solar activity and climate. In: Geophysical monograph, vol 141. American Geophysical Union, pp 321–340
- UNFCCC (2015) The Paris Agreement. https://unfccc.int/process-and-meetings/the-paris-agree ment/the-paris-agreement. Accessed 5 Sept 2020
- Unsworth MH, Biscoe PV, Pinckney HR (1972) Stomatal responses to Sulphur-dioxide. Nature 239:458–459
- Uprety DC, Saxena P (2021) Technologies for green house gas assessment in crop studies. Springer, Singapore, pp 1–134
- Usham AL, Dubey CS, Shukla DP, Mishra BK, Bhartiya GP (2018) Sources of fluoride contamination in Singrauli with special reference to Rihand reservoir and its surrounding. J Geol Soc India 91(4):441–448
- Vallero DA (2014) Fundamentals of air pollution. Academic
- van Oldenborgh GJ, Philip S, Kew S, van Weele M, Uhe P, Otto F, Singh R, Pai I, Cullen H, AchutaRao K (2018) Extreme heat in India and anthropogenic climate change. Nat Hazards Earth Syst Sci 18:365–381. https://doi.org/10.5194/nhess-18-365-2018
- Velissariou D (1999) Toxic effects and losses of commercial value of lettuce and other vegetables due to photochemical air pollution in agricultural areas of Attica, Greece. Critical levels for ozone level II. Bern, Switzerland: Swiss Agency for Environment, Forest and Landscape, pp 253–256
- Verma M, Agrawal M (1996) Alleviation of injurious effects of sulphur dioxide on soybean by modifying NPK nutrients. Agric Ecosyst Environ 57(1):49–55
- Vijaywargiya A, Pandey GP (2003) Effect of cement dust on soybean, Glycine max (L.) Merr. and maize, Zea mays linn.: fluorescence studies. Geobios 30(4):209–212
- Walkowiak J, Wiener JA, Fastabend A, Heinzow B, Krämer U, Schmidt E et al (2001) Environmental exposure to polychlorinated biphenyls and quality of the home environment: effects on psychodevelopment in early childhood. Lancet 358(9293):1602–1607
- Walna B, Kurzyca I, Bednorz E, Kolendowicz L (2013) Fluoride pollution of atmospheric precipitation and its relationship with air circulation and weather patterns (Wielkopolski National Park, Poland). Environ Monit Assess 185(7):5497–5514
- Wang SL, Lin CY, Guo YL, Lin LY, Chou WL, Chang LW (2004) Infant exposure to polychlorinated dibenzo-p-dioxins, dibenzofurans and biphenyls (PCDD/Fs, PCBs)—correlation between prenatal and postnatal exposure. Chemosphere 54(10):1459–1473
- Watson AY, Bates RR, Kennedy D (1988) Atmospheric transport and dispersion of air pollutants associated with vehicular emissions. In: Air pollution, the automobile, and public health. National Academies Press (US)
- Weinstein LH, Davison A (2004) Fluorides in the environment: effects on plants and animals. Cabi. https://doi.org/10.1079/9780851996837.0000
- Welfare K, Flowers TJ, Taylor G, Yeo AR (1996) Additive and antagonistic effects of ozone and salinity on the growth, ion contents and gas exchange of five varieties of rice (Oryza sativa L.). Environ Pollut 92(3):257–266
- WHO (2020) Air pollution. https://www.who.int/health-topics/air-pollution#tab=tab_3. Accessed 10 Sept 2020
- Wilmouth DM, Salawitch RJ, Canty TP (2018) Stratospheric ozone depletion and recovery. In: Green chemistry. Elsevier, pp 177–209
- Yasunari TJ, Tan Q, Lau KM, Bonasoni P, Marinoni A, Laj P et al (2013) Estimated range of black carbon dry deposition and the related snow albedo reduction over Himalayan glaciers during dry pre-monsoon periods. Atmos Environ 78:259–267
- Yung MM, Holmgreen EM, Ozkan US (2007) Cobalt-based catalysts supported on titania and zirconia for the oxidation of nitric oxide to nitrogen dioxide. J Catal 247(2):356–367
- Zhao A, Stevenson DS, Bollasina MA (2019) Climate forcing and response to greenhouse gases, aerosols, and ozone in CESM1. J Geophys Res Atmos 124(24):13876–13894
- Zipkin I (1972) Air pollutants affecting the performance of domestic animals, agricultural handbook 380U.S. Dept. of Agriculture, revised, 109 p



South Asian Monsoon Extremes and Climate Change

Mamta Yadav

Abstract

South Asian monsoon (SAM) is an important segment of the tropical monsoon system that mainly lies in the range of the seasonal oscillations of the intertropical convergence zone (ITCZ). About 70% of total rainfall in South Asia occurs during monsoon months each year, between June and September. Even within season, rainfall pattern is highly variable. The South Asian developing countries are mainly agro-based economies and livelihood of rural population is mainly dependent on natural resources which revive through the monsoon. The South Asian monsoon has influence on global atmospheric circulation and its future behavioural patterns will have serious implications for the entire world. Monsoon seasonal rainfall has temporal and spatial variability. There is no significant trend in Monsoon rainfall over past 100 years over the entire South Asian region. Ocean-atmosphere coupling in terms of ENSO and IOD has significant correlation with performance of monsoon over Indian subcontinent. IOD and ENSO are playing complementary to each other and maintaining good monsoon during recent decades. Along with year-to-year variability there exists intra-seasonal variations which are more markedly noticeable nowadays. These are in the form of prolong dry days with no rain or little rain and prolong wet days with rather good to very heavy rain forming a cycle of active and break spells. Extremes occur during active and break spells of monsoon. The extreme rainfall events mainly occur during synoptic scale weather systems like lows and depressions. Extreme events like flood, drought and more landslides are high impacting events that need proper strategies to overcome the adversities. The present study focuses on monsoon extremes, intra-seasonal variations and modulation by remote

M. Yadav (🖂)

Meteorological Centre, India Meteorological Department, Bhopal, India

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_4

forcing and their future trends in global warming scenario. The global oceanatmosphere coupling and its linkage with South Asian monsoon is on chance and how it will be progressing in future.

Keywords

South Asian monsoon · Rainfall · Extreme event · Atmosphere · Natural hazards

4.1 Introduction

South Asian monsoon is basically summer monsoon on Indian subcontinent and is large-scale manifestation of ocean-atmosphere coupling (Hernandez et al. 2015; Ramaswamy 1965). The monsoon rainfall is mainly driving agricultural production supporting food security and steering economy of South Asia. The word monsoon is basically having its origin from Arabic language. Monsoon is a season during which reversal of wind direction takes place over a large geographic area. South Asia, Africa, Australia and Central America's Pacific coast are the main monsoonal regions of the world and mainly characterized by wind reversal during winter to summer or summer to winter transition mainly (Buckley et al. 2014). These are extended on tropical region 25°S to 35°N, 30°W to 170°E. Indian monsoon region is main central region of the South Asia. Wind reversal during seasonal transition can be understood by global pressure and wind patterns in January and July. The Siberian High of northern Asia and polar highs are main features of the winter (January) circulation in northern hemisphere with two intense low-pressure centres (the Aleutian low and the Icelandic low) situated over the North Pacific and North Atlantic, respectively, while in the summer months, the sub-tropical high-pressure areas of Northern Hemisphere shift westward and have more intensity than winter months (Sonwani and Saxena 2021). These strong high-pressure areas regulate the summer circulation over the oceanic regions and supply warm moist air on to the continents that are situated to the west of these highs leading to precipitation over parts of eastern North America and Southeast Asia. The changes in pressure patterns reverse wind from north-easterly in winter to south-westerly in summer (Fig. 4.1).

Monsoon can also be considered as atmospheric response of variable sun radiation governed by seasonal movement of sun. The movement of sun between Tropic of Cancer in Northern Hemisphere and Tropic of Capricorn leads to differential heating on different land and oceanic masses and generates significant temperature gradient which is opposite in direction compared to winter season (Lau and Li 1984; Webster 1987; Meehl 1992). This temperature gradient is basically responsible for the generation of low-pressure areas and faster wind movement towards these low-pressure areas through wind convergence along with vertical lifting of air masses and formation of clouds.

Moreover, if we look at general circulation patterns of atmosphere, we find alternate low- and high-pressure belts in both hemispheres. The equator is low-pressure belt where low-level surface north-easterly winds converge over a



Fig. 4.1 Average surface pressure and associated global circulation (Source: The Atmosphere-An introduction to Meteorology, Book, Eleventh Edition, Authors: F. K. Lutgens, E.J. Tarbuck Illustrated by Dennis Tasa) source : The Atmosphere, 8th edition, Lutgens and Tarbuck, 8th edition, 2001

broad area forming inter-tropical convergence zone ITCZ. In monsoon season, it travels across lower latitudes of the Northern Hemisphere and in winter season shifts towards Southern Hemisphere low latitudes main (Gadgil 2018). The average position of ITCZ is a few degrees north of the equator due to extensive land masses of the Northern Hemisphere as compared to Southern Hemisphere creating

maximum heating zone in Northern Hemisphere and equatorial trough remains in this maximum heating zone. The position and movement of this convergence zone is linked with movement of sun and formation of new low-pressure areas further north of its position. Monsoon can also be defined as gradual movement of ITCZ towards north and south with movement of sun. The movement of ITCZ is a characteristic feature of South Asian monsoon (Turner and Annamalai 2012).

South Asian developing countries maintain increase in growth rate from past few years except covid 19 pandemic crises. These countries have large domestic market and agro-based economy well maintained by regular robust monsoon. The growth of agriculture sector, work-based requirement and demand in domestic market are directly linked with performance of monsoon. The monsoon rainfall and its intraseasonal variations are deeply affecting life of 1.89 billion people. The well-noticed variations in dry and wet spells and high impact extreme events like floods, droughts, severe thunderstorms and landslides during monsoon season carry atmospheric response to anthropogenic activities which can be quantified through meteorological and other related observations needed to understand climate change and associated risk. The prediction of onset and withdrawal of monsoon and also dry and wet periods is of key importance. The good quality observational data sets can help future climate management strategies for land-based water resources, ground water, agricultural best practices, food security, disaster early warning and preparedness, and infrastructural needs. The synoptic meteorology seems very important in this regard. Assessment of large-scale semi-permanent features based on observed data and how the synoptic features have modulated under various ocean-atmosphere forcing in recent climate change scenario are also addressed in this chapter.

4.2 Semi-permanent Features of Monsoon

4.2.1 Heat Low

The low-pressure area develops over Pakistan and North West India, also called "heat low", indicating thermal heating conducive for its formation. The Hindukush mountains regionally play important role in its formation. Its location and intensity are significant for monsoon advancement process. Intensity of heat low is correlated to rainfall activity (Ramage 1971). The negative pressure departure in heat low region and positive pressure departure in peninsula are favourable for monsoon progress. Heat low formation starts from May and get established by June near to its normal position. Some variations occur in spatial position and strength of heat low during monsoon season but mostly it remains at its normal position and gradual weakening starts from September and vanish in first fortnight of October. The observed minimum values of the heat low were 992 hpa to 994 hpa during June and July and 996 hpa to 1000 hpa in August and September for the year 2016.

4.2.2 Monsoon Trough

The seasonal trough line, starting from heat low over Pakistan and adjoining areas, extending in southeast direction and passing through West Bengal's Gangetic region and reaching up to eastern part of West Bengal and Orissa known as Head Bay. The low-pressure line generally runs from Ganganagar to eastern West Bengal coming through Allahabad, having westerly to south-westerly winds in the south of the low-pressure line and easterlies to the north of the low-pressure line. The position of the low-pressure line oscillates on day-to-day basis and affects the monsoon rains. When the trough is near to the foothills of Himalayas termed as "break in monsoon" as there is decrease in rainfall activity over most parts of the country, though the Himalayan mountain region get good and heavy falls leading to occasional overflow of rivers creating floods. The migration towards south direction of trough from its mean position, i.e. the central parts of the country, under the influence of monsoon depressions forms in the northern Bay of Bengal and generally moves in westerly and west north-westerly direction across Indian land mass, giving widespread rainfall with some heavy to very heavy rain spells over the central India and some parts of peninsula. The oscillations of monsoon trough about its mean position govern active and break phases of monsoon. The total number of days in these positions is crucial for seasonal rainfall. In September month, western end of monsoon trough shifts near to foothills of Himalayas and eastern end remains south of its normal position and during last 15 days it almost vanishes.

4.2.3 Tibetan Anticyclone

Elevated Tibetan plateau by regular sun heating starts transferring its heat to upper winds acting as a heat source and plays significant role in global circulation and South Asian monsoon or Indian monsoon. The heating of the plateau just reverses the meridional temperature gradient throughout the troposphere. This reversal is the main factor which brings changes in seasonal circulation over East Asia, with poleward shift of the subtropical jet. At 500 hPa it is situated to the east of Long 80° E with axis near about 28° N. It is more intense at 300 hPa and extents between70 E and 110 E. The rainfall and its distribution are linked with east west well-oriented Tibetan anticyclone. The location of anticyclone changes during the monsoon and corresponding changes occurs in rainfall also. It has been observed that when it shifts towards west from its mean location rainfall activities get enhanced over western Himalayan region. Shifting of anticyclone towards west or southwest of its mean position enhances rainfall in northern India and shift towards north or northeast of mean position decreases rainfall over northern regions.

4.2.4 Tropical Easterly Jet (TEJ)

Strong concentrated easterly winds in the form of jet are seen in upper air wind charts located south of subtropical ridge over Asia. It is having centre over Chennai city latitude and prominent at upper tropospheric level, i.e. 100 hpa, mainly during the month of July. The Tropical Easterly Jet (TEJ) starts from eastern coastal areas of Vietnam and runs up African west coast. The jet gains strength over the region of South China Sea and southern Indian region, and after crossing this it starts weakening. A broad latitudinal spread of jet winds during monsoon months has been observed through wind charts and maximum wind velocity of 126 knots at upper tropospheric level around 154 hPa pressure level was recorded at Guwahati in radiosonde data. These easterlies of jet provide steering to LPSs formed in Bay of Bengal for westward movement. The jet is likely to weaken resulting in reduced wind shear favourable for more cyclonic storms (Brahmananda Rao et al. 2008).

4.2.5 Mascarene High (MH)

The strength of cross-equatorial flow is maintained by intensity and location of this high. It is a part of high-pressure belt located at subtropical latitudes in southern hemispheric Indian Ocean centred at 30° S/50° E, termed as Mascarene High. This high pressure generally forms close to the Madagascar island off the coast. The waves coming from extra tropics cause small duration fluctuations in its intensity (Sikka and Gray 1981). The increase in intensity of Mascarene High accelerates the moist wind current crossing equator at lower levels and forms a jet stream type of flow also known as low-level jet and identified as wind field associated with monsoon strength over regions of Arabian Sea. On similar lines decrease in intensity of this high weaken the cross-equatorial flow and monsoon current. The parameters like temperature over sea surface (SST), heat energy content of ocean and sea surface height (SSH) are on increasing trend over its region as evident from observations of recent 20 years. The warming of sea surface aids in curtailing the pressure at sea level resulting in feeble or weak gradient of pressure between the Mascarene high and the pressure over land and oceanic regions of Northern Hemisphere (Vidya et al. 2020).

4.2.6 Somali Low-Level Jet (LLJ)

Inter-hemispheric lower-level strong wind flow as a jet stream having location of core near 850 hpa appears flowing from Mauritius across Kenya, also called Find later jet or low-level jet. The lower-level strong winds with highest speed around 40–60 kts identified at core of the wind field over SE Arabian Sea and adjoining Peninsular India is a part of this Somalian or Find later jet. The frequent westerly waves of extra tropical origin modulates its intensity. Strong cross-equatorial low-level jet (LLJ) increases rainfall activity around entire west coast of India. The

wind speed at core, depth of westerlies, zonal water vapour flux, horizontal wind shear and cyclonic vorticity show an increase near to the heavy rainfall days and a decrease afterwards (Xavier et al. 2017). The fluctuations of LLJ or Somali jet can be assessed through Somali jet speed index (Boos and Emanuel 2009). The LLJ intensity changes within 20- to 90-day period, in relation with the northward-advancement of active and break phases of the monsoon (Swathi et al. 2020).

4.3 Phases of Monsoon

4.3.1 Onset of Monsoon

The monsoon over south Asian land reaches initially at Andaman and Nicobar Islands around 20 May normally and starts moving towards Indian land; strong wind flow from Southern Hemisphere to Northern Hemisphere mainly cross the equator as strong concentrated wind flow, termed as low-level jet (LLJ) or Somali jet or Find later jet, and establishment of tropical easterly jet at upper level (100 hpa) is the main feature of wind charts during the onset phase. Onset date of monsoon over Andaman and Nicobar Island is based on actual observations, model forecast and convection in satellite imageries. The Zonal winds of 8 mps at 925 hpa and westerly wind up to 600 hpa and Outgoing Longwave Radiation (OLR) equal to or less than 210 watt/m^2 with fairly widespread or widespread rain over islands are realized in actual observation. Monitoring all such parameters for few days the onset date over Andaman and Nicobar can be declared (Debnath et al. 2019). The India Meteorological Department every year announces onset of monsoon over Kerala state considering three parameters: precipitation, wind speed and wind velocity, and vertical depth forming, i.e. wind field and outgoing longwave radiation mainly. The temperature gradient in the upper atmosphere, i.e. 700 hpa and 500 hpa, get reversed during onset (Ananthakrishnan and Thiruvengadathan 1966). The kinetic energy at low level, total kinetic energy generation in vertical direction and total available tropospheric moisture over Arabian Sea may also be used as possible predictors for onset date over Kerala (Raju et al. 2005). Normal date of monsoon arrival over Kerala is considered to be 1 June with a deviation of 7–8 days prior or later. Sea surface temperature gradient (SST) over Indian and Pacific Ocean have linear correlation with onset date on decadal time scale (Preenu et al. 2017).

4.3.2 Monsoon Advance

The progress of monsoon varies from year to year. Monsoon advances by two branches, one is Arabian Sea branch and the other is Bay of Bengal (BOB) branch. The Arabian Sea branch soon after onset over Kerala moves northward along the west coast and covers Gujarat region near June 15. Whole Peninsula and Central India get monsoon rainfall through this branch. The BoB branch moves northward and enters regions of Bangladesh, Assam and adjacent north eastern states. The



Fig. 4.2 (a) Monsoon onset and progress 2019; (b) Monsoon onset and progress 2013 (Fast Advancement) Source: India Meteorological Department; https://www.imdpune.gov.in/Weather/weatherforecast.html)

height and orientation of the hills of Assam, Meghalaya and Myanmar create obstruction, resulting in deflection of BoB branch in west direction under the Himalayas range. The BoB branch reaches over the Gangetic plains of north India in the form of south-easterly/easterly winds. The branches meet over north India around longitude 80° E. Monsoon takes 15–35 days to cover the entire country as observed during 2010–2020. Monsoon progressed fast in 2013 and taken only 16 days to cover the whole country.

Formation of low-pressure systems (LPS) in Indian Ocean region may facilitate timely onset of monsoon over Kerala or it may cause delay in monsoon advancement over Arabian Sea. In this context, the location of formation of system, its intensity and track are of much importance. The LPSs in BOB and Arabian Sea disturb the south-westerly wind flow coming from Southern Hemisphere crossing the equator. Sometimes the flow gets intensified favouring fast monsoon advancement over country and sometimes flow get decreased giving a temporary halt to the advancement of monsoon. The advancement of monsoon is not a continuous process during some periods northern limit of monsoon remain stagnant for 5–6 days or even up to 10–12 days and termed as hiatus in monsoon. The normal dates of monsoon progress (onset and withdrawal) have been revised by India Meteorological Department based on 1961–2019 data. The progress of the monsoon is sometimes fast and sometimes slow (Fig. 4.2).

4.3.3 Withdrawal of SW Monsoon

The withdrawal of monsoon starts from north western part of country in the month of September. During past decade 2010–2020, the withdrawal generally commenced from third week of September and took 15–30 days or even more for the entire country. The important meteorological aspects to be looked into for the withdrawal;



Fig. 4.3 (a) Monsoon withdrawal 2014; (b) Monsoon withdrawal (Fast) (Source: India Meteorological Department; https://www.imdpune.gov.in/Weather/weatherforecast.html)

no rainfall for consecutive 5 days and development of anti-cyclonic circulation which can be seen at 925 hPa or 850 hPa levels in upper air charts, i.e. reversal of vorticity field in the lower troposphere and dominance of northerly or north-westerly winds in over northwest India, shift of ITCZ in south and rapid reduction in available moisture in middle and upper atmosphere may be observed through satellite water vapour imageries and radio-sonde data. During past decade, most of the monsoon withdrawal commenced in late October month, so monsoon withdrawal dates has been revised based on operation dates data 1971–2019 (Fig. 4.3).

4.4 Intra-seasonal Variations/Active and Break Spells

Indian monsoonal system is a segment of large-scale circulation patterns associated with monsoon over Asian countries and having different temporal variability on daily, seasonal, annual and decadal time scale. In general term, active spells mean period of good rainfall days and weak/break spells are days of little rain or no rain. The long period of active spell or a greater number of days of good rainfall impacts monsoon rainfall of the country and with widespread spatial coverage fulfil agricultural needs and water requirements of the country. Similarly, long period or more days with no rain, i.e. breaks, impacts the total precipitation in monsoon (Gadgil and Joseph 2003) and adversely affect crop growth and management practices. The timing of occurrence of break spells is also very much crucial for agriculture and if coincides with period of crop maturity it will affect crop yield. Long and frequent break spells may lead to drought situation.

A break situation arises by shifting of low-pressure trough near to Himalayan foothills as defined by (Ramamurthy 1969). In this situation more rainfall in Himalayan region and rainfall activities get reduced in other parts of country. The position of monsoon trough is very critical for determining active and break spells.

When monsoon trough migrates towards south of its mean position good rainfall received over central and other coastal states of the country making monsoon active.

Earlier surface pressure and wind patterns were used for identification of breaks over the Indian region. A break could be identified by disappearance of easterly winds from sea level and 850 hpa charts at least for 2 days (Blanford 1886; Goswami and Mohan 2001) zonal winds at 850 hPa over the Bay of Bengal, used for "active" and "break" situation. The high frequency of active spells leads to strong monsoon and more number of break spells leads to weak monsoon. The variations with in the season ultimately decide the seasonal mean. The intra-seasonal variations are dominated by small time scale processes very chaotic in nature.

The positive OLR anomalies over western portion or segment of monsoon trough, i.e. northwest and central India is an indicative of break and the average OLR should exceed 10 Wm⁻², (Krishnan et al. 2000). Daily rainfall anomalies over Indian core region 73–82°E, 18–28°N were also used to define the active and break (Mandke et al. 2007). Low-pressure systems from and move over Indian subcontinent during active phase of monsoon and these systems generally follow the monsoon trough and clustering of clouds takes place around that trough line (Goswami et al. 2003). The gridded data sets prepared and used for classifying active and break spells using standard rainfall anomaly over a region (Rajeevan et al. 2006, 2010).

Indian Summer monsoon rainfall (ISMR) have significant negative correlation with break days and significant positive correlation with active days but the relationship of ISMR with break days is stronger than those of active days. Active and breaks operate on 10–12 days' time scale.

Revival of active spell after break spell is mainly governed by oscillations of monsoon trough. When a weather system of low-pressure develops in Bay of Bengal (BoB) under influence of cyclonic upper air winds, it pulls down the monsoon trough towards south. Monsoon trough again migrates to its mean position or further south of its normal position giving lot of rains in various states. The spatial coverage, intensity and period of rain depends on intensity of low-pressure system, its track and strength of moist winds and support from geographic features like mountain range or big water bodies. The LPS move Northward initially or North-westward along the monsoon trough (Goswami 1987; Sikka 1977). Some LPS formed in Bay of Bengal (BOB) entered Arabian Sea water after crossing Indian land while some LPS of BOB move Northward and then recurve to North-eastward towards Bangladesh and Myanmar. These systems following different tracks give wide-spread rain and make monsoon active again.

The second way of revival is through northward propagation of tropical convergence one. The deep convective clouds get oriented in east west direction in Indian subcontinent region linked with tropical convergence zone (TCZ) during active spell (Sikka and Gadgil 1980). The life period of active spell is small as compared to break spells, so after the dissipation of deep convective clouds associated with TCZ, there should be northward propagation of freshly formed convective clouds from the equatorial Indian Ocean, so TCZ has two arms—one on Indian land mass and the other on ocean. When land arm is active, the oceanic arm is in formative or beginning stage, and when land arm gets dissipated, the oceanic arm gets developed with convective clouds and starts propagating towards northern latitudes with winds and starts rain leading to revival from a break phase.

4.5 Extremities in South Asian Monsoon

The rainfall extremes are high magnitude rain particularly above threshold value for a day or more number of days leading to large runoff on ground. South Asian nations receive rainfall via southwest monsoon. In Asian continent, monsoon extreme event frequency has been increased (Yao et al. 2008; Shrestha et al. 2017). The change and rise in of extreme also assisting probability of the event to an increase noticed over various regions of countries situated in South Asia (Christensen et al. 2007). Some opposite trends of decreasing precipitation over a time period of 30 years from 1976 to 2005 and large regional variations of high intensity rainfall events have been observed addressing impact of regional variability over Pakistan (Salma et al. 2012).

The monsoon season extreme rainfall events (rainfall >124.4 mm in a day) over Indian region have spatial variability over selected regions like entire west coast, some parts of central and northeast India. The significant rising trend of frequency of extreme rainfall over the Indian monsoon region season and significant increasing trend during June and July months during period 1951–2005. The increased fraction of extreme rainfall events to the total seasonal is compensated by a significant decreasing trend in rainfall (rainfall R < 64.4 mm in a day) events (Pattanaik and Rajeevan 2010). Most prominent extremes are caused by synoptic scale weather systems, mainly the low-pressure systems, leading to floods and landslide. The low-pressure system may intensify to depression or occasionally to a cyclone surrounded by convective clouds mass and give heavy rains. The rainfall activities in south west sector of the LPS (Godbole 1977; Sikka 1977). These systems remain active for 3-5 days and cover area more than 1000 km². A cyclonic system of the intensity characterized for low has associated wind speed not more than 8.5 ms⁻¹ and next stage of intensity, i.e. depression wind speed, ranges from 8.5 to 16.5 ms^{-1} . The low-level meridional wind shear linked with the ITCZ assists the processes involved in genesis of lows. Whole area of North Indian Ocean is known for formation of lows but comparatively more number of lows form in Bay of Bengal (BoB) because of high SST of oceanic water and the effective mixed layer not very deep and trough also passes through various sectors of BOB during monsoon season. The low-pressure systems generally follow monsoon trough (Goswami 1987) and their movement towards West and North West direction explained through change of Coriolis force across latitudes on movement track, i.e. beta-effect. Research studies (Dash et al. 2004; Pattanaik 2007; Prajeesh et al. 2013) are indicating decrease in frequency of depression from past few years although the total number of LPSs remained constant. The observed decrease in number of depression stage cyclonic systems in regions below 20°N latitude in the Bay of Bengal starting from 1950s is associated with declining trend of available moisture in the middle atmosphere, 600–300 hPa mainly over the Indian latitudes (Prajeesh et al. 2013). Pollution is also claimed as inhibitor of deep convective clouds and interaction among

clouds (Krishnamurti et al. 2013). Western Ghats experience 60% of extreme rain events synoptic disturbances (Francis and Gadgil 2006). Sometimes meso-scale convective systems getting moisture from local water bodies and thermodynamic support from heating may lead to local extreme rainfall. The increasing temperature in over Indian subcontinent enhance evaporation leading to more moist airmasses in the atmosphere which is a positive forcing for strong convection triggered extreme precipitation events (Kothawale et al. 2010).

The landslide increases in Nepal, India, Bangladesh, Bhutan and north Pakistan during the monsoon season. India and Nepal have high rainfall triggered landslide events which is about 21% of global rainfall triggered landslides mainly during monsoons (Froude and Petley 2018). There exists inter-annual variability in landslide incidences occurring in globe. The years with more landslide incidences found associated with regional well distributed rainfall patterns like positive rainfall anomaly but there is no simple relationship with ocean–atmosphere coupling like ENSO. The improved seasonal rainfall forecast which can capture temporal and spatial variation in active and break spells may be helpful for probable landslide zone identification and impact reduction particularly for major affected areas such as India, China and Nepal. Climate change is casually linked with landslides (Gariano and Guzzetti 2016).

4.6 Regional Trends in Monsoon

Walker (1910, 1914) found no trend in monsoon rainfall during 1841–1908 periods over Indian region. Kumar et al. (1992) have done a study on spatial and intraseasonal nature of climatic trends of rainfall. West coast, North Andhra Pradesh and North West India are the areas where monsoon season rainfall trends were increasing, whereas in East MP and adjoining areas, North East India and some parts of Gujrat, the trends were decreasing. August rainfall contribution is more prominent to monsoon seasonal rainfall mainly over coastal areas of West and central parts of India. Parthasarathy and Dhar (1976) observed increasing trend for annual rainfall over East MP and West MP for 1901–1960 periods. Naidu et al. (1999) studied annual rainfall trends for 1871–1994 period; the negative trends are recognized in geographic region of west central India, middle parts of north India and north-eastern region of India. The enhancing trend observed over north-west India spanning states like Haryana, Punjab, Rajasthan and West MP, an isolated area in the east and the peninsular region.

Guhathakurta and Rajeevan (2008) observed no significant trend for monsoon and monsoon months for the country as a whole with large variations in different regions. The July rainfall is decreasing for most parts of the central and peninsular India but increasing significantly in the north-eastern parts of the country. August months' monthly rainfall has increased significantly (at 95% significance level) for the subdivisions Konkan and Goa, Marathwada, Madhya Maharashtra subdivision. Kerala and Tamil Nadu, dry days is on increase resulting in reducing trends of mean yearly rainfall. Rajeevan et al. (2008) examined variations in long period climatic trends of extreme precipitation events over the middle India using 104 years (1901–2004) high resolution daily gridded rainfall data and found that SST fluctuations over Tropical Indian Ocean are linked with year-to-year and decade-to-decade variations of rainfall extremes. The strong connection between the SST over equatorial Indian Ocean and very heavy rainfall (VHR) events over the central meteorological subdivisions of Indian region will lead to more and more VHR events and further supported by the atmospheric global warming condition the floods occurrence will increase over central India.

Annual normal rainy days are nearly 10 over extreme west Rajasthan to 130 days over north eastern states of the country. The north eastern areas of the country, sub-Himalayan West Bengal and extreme western coast region receive approximately 100 rainy days in a year. The frequency of rainy days and heavy rainfall days are decreasing significantly over central, north and some areas of eastern parts of India. Wet days frequency is increasing over peninsular India, mainly Karnataka, Andhra Pradesh and some parts of eastern India. Positive trends in the frequency of heavy rainfall days are observed in parts of Konkan and Goa and adjoining coast and some areas of eastern India. Single-day extreme rain events are on rising over costal districts of Andhra Pradesh Orissa, West Bengal, some northeast regions and on west coast of India parts of Saurashtra and Kutch and east Rajasthan. The extreme induced flood situation and associated risk has increased manifold during past decades mainly over coastal states (Guhathakurta et al. 2011). Extreme rainfall frequency is on decreasing line over some areas of states like Chattisgarh, Jharkhand and some areas of north India.

Goswami et al. (2006) using daily realized rainfall data set found that amount and occurrence of extreme precipitation events are significantly rising and moderate amount events and their frequency are decreasing significantly over central India in monsoon season based on data from 1951 to 2000. Kumar et al. (2010) observed no significant trend to address the variations for yearly and monthly rainfall for India as a whole.

Singh and Mal (2014) studied rainfall trends and variability over western Himalaya and found declining trend in annual and monsoon rainfall over high altitudes and increasing over low altitude. Agarwal (1952) investigated the rainfall variations over middle geographic land areas of India and found a steady change which was not significant during 1908–1940. Temperature and precipitation data of Ganga Basin analysed by Kothyari et al. (1997) shows that rainy days are decreasing, resulting in decrease of total monsoonal rain, and the annual record of daily maximum temperature is showing rise. These variations became noticeable after the year 1965, highlighting climate change signatures on regional scale.

Mullicka et al. (2019) analysed the trend of mean annual rainfall and reported increasing trend of 5.675 mm/y for Bangladesh during 1966–2015 period. The 1 day and consecutive 3-day maximum rainfall trends are increasing in few places and decreasing at other places with no significant change.

4.7 ENSO, IOD and MJO Links with Monsoon

Rainfall during monsoon season and associated planetary and synoptic scale circulation are mainly controlled by global parameters like ENSO and IOD, which are indicative of quantification of ocean-atmosphere coupling in which sea surface temperature also plays influential role. Mooley and Parthasarathy (1983) investigated the relationships of indices of dryness and wetness over India with SOI and SST anomalies are expected to be useful in understanding implications of large-scale anomalies in performance of Indian summer monsoon. Rasmusson and Carpenter (1983) and Shukla (1987) studied the link between the inter-annual variability (IAV) in the Indian summer monsoon rainfall (ISMR) and the El Nino-Southern Oscillation (ENSO) phenomenon. Yaday (2009) used surface temperature. pressure mean sea level (MSLP), geopotential height in the troposphere and global flow patterns of wind to study rainfall variations occurring over years for 1949–2005 period and found the El Nino-Southern Oscillation (ENSO) linkage with strength of monsoon has been deteriorated whereas it has become strong with SST over northwest (NW) of North Atlantic. Yaday (2008) established link between 2mST over tropical southwest Pacific and intensification of subtropical westerly jet over Indian Ocean and enhancement of monsoon circulation during negative phase of ENSO. While during non-ENSO year positive relationship of ISMR with 2mST anomaly over northwest of North Atlantic Ocean. Srivastava et al. (2020) observed ENSO impact on July and August rainfall and found significant changes at multidecadal time scale. The influence of ENSO was significantly strong in August and significantly week in July during 1948-1980 and post 1980s, ENSO monsoon relationship less strong in August than in July. During 2017 monsoon three regions Central India, West Central India and North East states experienced extreme events in active phase and northeast region of Himalayan range experienced during break phase. Most of the extreme events happened during July with significantly high seasonal frequency favoured by weak La Niña and positive Indian Ocean dipole (Suthinkumar et al. 2019). ENSO (El-Niño Southern Oscillation) term is used for ocean-atmosphere coupling. It is a combination of southern oscillation in pressure field and temperature oscillations Pacific Ocean. Warm Phase of ENSO means El Niño; above average SSTs over eastern Pacific and cool phase of ENSO means La Niña, SSTs in the eastern Pacific below average. ENSO have opposite phase relationship or the inverse relationship with Indian monsoon, i.e. warmer ENSO event weaker is the monsoon. Indian monsoon rainfall has strong correlation with anomalies of SST over Nino 3.4 region mainly. The negative SST anomalies over crossing threshold $(-0.5 \,^{\circ}\text{C})$ lead to La Niña event and above-normal or excess rainfall over the country. The La Niña events have positive influence on monsoon and its seasonal variation. The positive SST anomalies corresponding to El Niño over Nino regions [Niño1 + 2 (0° -10°S, 90°W–80°W), Niño 3 (5°N–5°S, 150°W– 90°W), Niño-3.4 (5°N-5°S, 170°W120°W), Niño-4 (150° W-160° E and 5° N-5° S)] lead to deficient or below-normal rainfall and weak monsoon over the country. This SST–Monsoon correlation is also changing with decadal–epochal oscillations. The seasonal rainfall time-series have epochs of more than and less than the mean rainfall. El Niño with below-normal rainfall Epochs will be more severe creating rainfall deficiency and similarly La Niña with above-normal rainfall Epochs will give surplus excess rains. The major rain related extremes events (massive flooding/ acute droughts) are because of the reason that external forcing, i.e. El Niño and La Niña, variability and epochal variability are in same (Krishna et al. 1999).

The ENSO-monsoon relationship seems weakening (Kumar et al. 2010) in recent decades due to shifting of walker circulation anomalies south-eastward causing reduction in sinking motion over Indian region and favouring the monsoon. The persistent higher surface temperatures in winter and spring over Eurasia, the mid latitude warming enhancing land ocean temperature contrast required for monsoon.

The effect of ENSO on monsoon rainfall and its intra-seasonal variations are also influenced by Indian Ocean Dipole (IOD). The oscillations in SSTs over the eastern and western sides of Indian Ocean form the dipole. The dipole moment index (DMI) is used to quantify the Indian Ocean warming influence on monsoon rainfall. Positive IOD is favourable for monsoon rainfall and negative IOD is unfavourable for the monsoon. During the period 1958–1997 the El Niño/Southern Oscillation (ENSO) and IOD behaved in complementary manner with each other and overall balancing of monsoon rainfall during the last four decades claimed by research studies (Ashok et al. 2001). The temperature dipole in Indian Ocean (IOD) is acting as a modulator of rainfall over the Indian subcontinent and redefining the correlation between monsoon rainfall and ENSO.

Another important factor causing intra-seasonal variations in monsoon rainfall is Madden Julian Oscillation (MJO). Pai et al. (2009) analysed the linkage of various phases of MJO on day-to-day variation of monsoon rainfall. As MJO moves eastwards, northward shift of rainfall belt from peninsular India takes place. The above-normal rainfall band is observed along monsoon trough region in 5 and 6 phases of MJO. When MJO moves in 7, 8 phases, there is decrease in rainfall over various parts of country. MJO phase 1 and 2 are unfavourable for monsoon rainfall activity. MJO plays crucial role for onset of active rainfall spells and break spells and their duration. The amplitude of MJO is significant in this connection but phase of MJO is more important as related to a region.

4.8 Climate Change and Global Warming Impacts

Warming temperature trends, high magnitude extremes, are prominent over the interior land masses of various parts of Asia. From 1900 to 2005, precipitation declined in parts of southern Asia (IPCC 2007). The global warming will be enhancing and climate models are forecasting increase in annual and monsoon rainfall over Indian region. There is high probability of precipitation extremes to increase during monsoon season over various parts of southeast sector of Asian continent. All available climate-related models with the maximum possible emission regimes all are projecting rise of mean and extreme precipitation in Southern Asia (SAS). The different phases of Coupled Model Inter-comparison Project, mainly phase 3 (CMIP3) and Phase 5 (CMIP5), are forecasting weakening of large-scale monsoon circulation in future (Kripalani et al. 2007; Sooraj et al. 2015; Krishnan

et al. 2012). The changes in location, intensity of large-scale systems like monsoon trough over Indian region, the Mascarene High, the western North Pacific (NP) subtropical high, and the upper tropospheric South Asian high are projected (Preethi et al. 2017). There is no clear consensus in models about genesis and track modification in future warming scenarios but more extreme rainfall events are projected over the coastal areas where cyclone landfall is expected to occur. The recent Coupled Model Inter-comparison Project Phase 5 (CMIP5) simulations with Representative Concentration Pathway (RCP) framework are forecasting positive dilation of mean temperature greater than 2 degrees change in mean annual temperature above the reference temperature of late twentieth century over the most of the countries across the globe in the middle of twenty-first century according to RCP8.5 and are above 3 °C for countries situated in Southeast sector of Asia having high population density and big coastal boundary and of the order of 6 °C or even more over higher latitudes in the second half of twenty-first century. Model projections have uncertainties posing challenging situation for policy makers. Looking for more reliable forecast for South Asian monsoon (SAM) the uncertainty/over projection should be identified scientifically. The internal variability is mixed with forced rainfall trends, leading to over-projection, and inter-decadal Pacific Oscillation (IPO) is identified as cause of more uncertainties (Huang et al. 2020). The improved IPO predictions can considerably remove near-term over-projections of SAM.

Worldwide mean temperatures will be on rising trend during span of twenty-first century if greenhouse gas (GHG) emission remains on similar pattern of increase. The global temperatures averaged over the period 2081–2100 are projected to likely exceed 1.5 °C above 1850–1900 average temperature threshold. Heavy rainfall frequency will increase over many parts of South Asia. Increased count of warm days and less count of cold days are likely to remain on same track as the warming is likely to enhance in future century also. The problem of water scarcity will be amplified by increasing population and high sub-regional variability of precipitation. The well-known most fertile plains of Ganga may have 50% decrease of crop yield wheat due to expected double CO_2 emissions. Rise in sea level will inundate lower areas and Mangroves, salt marshes, and sea grass beds and rice growing regions may decline. Over Indian region localized heavy to very heavy rainfall events reported an increase during past 50 years. Land use pattern and urbanization and geographic location are the factors enforcing localized events. The year-to-year variations of monsoon will be more pronounced during twenty-first century. Heavy rainfall events and localized intense rainfall events will be enhancing flood risk as per Ministry of Earth Science (MoES) report 2020. The westerly wind over north Arabian sea brings strong moisture carrying spells and gives rise to exceptionally heavy rainfall events over districts of Gujarat and landlocked areas of Central Indian region. The main reason behind it is warming of north Arabian sea due to carbon emission making winds more variable. In ongoing climate change and warming scenario, pole ward shift of low-level jet is projected leading to more dry days and variation in precipitation over west coast adversely affecting biodiversity of Western Ghats (Sandeep and Ajayamohan 2015).

Floods are the result of monsoon extremes causing loss of human and animal life and property. Major flood events with largest loss of human lives and property occurred over the central India in 1989 and 2000, Mumbai in 2005, Uttarakhand floods 2013, Kerala floods 2018, etc., which resulted from 3 to 4 days continuous raining supported by westerly flow from the Arabian Sea. In 2016, due to heavy rainfall, widespread flooding damaged life and properties over west coast, Mumbai. The Kaziranga National Park in Assam submerged due to extreme rain. Drought is another climate extreme that arises due to rainfall deficiency. Although monsoon total rain remains normal, certain districts remain deficient in rainfall and suffer from drought. Drought is not instant but cumulative deficiency of rainfall for months and years. Drought declaration and management is state subject in India. Almost every year each state has some districts under drought conditions in spite of normal rainfall because of high spatial variability in rainfall. In 2016, 46 districts were drought affected in M.P. and in 2019 again 36 suffered acute drought conditions. In 2018, 18 districts of Jharkhand, in 2019, due to weak northeast monsoon, 23 districts of Karnataka suffered drought. Similarly, other parts of the country are also facing drought conditions mainly before and after the onset of monsoon and sometimes even in monsoon months. Delay of monsoon rain intensifies use of ground water resources, high heat leading to fast evaporation of water from surface water bodies and drying them fast and creating water scarcity in drought-affected area. Drought becomes more severe if it occurs during consecutive years at the same place. IMD can declare drought when rainfall is deficient or less by 10% and spatially 20-40% geographic area of the entire India is encountering dry weather situation or no rain type of situation. Rainfall-related drought is termed as meteorological drought but agricultural drought is different from it and several other crop-related parameters are taken into account for agricultural drought. Mainly lack of rainfall or uneven rainfall can lead to agricultural drought-like situation. In Rajasthan in 2002, due to almost 50% less rainfall as compared to July month normal rainfall, farmers could not sow kharif crop and drought was declared. The concept and definition of drought is subjective in nature. The methodologies for assessment and parameters under consideration are changing from state to state. In order to address various issues associated drought declaration and bring some uniformity in the process government of India brought the new guidelines for drought declaration in 2016. The monitoring of drought is done by Crop Weather Watch Group (CWWG) central and state level having responsibility of regular interactions, analysis of data coming from various institutes and government bodies and assessing its impacts on crop. The new guidelines brought by Indian government drought declaration given first priority to Meteorological information (a) the rainfall measured by the percentage deviation in rainfall from average or the Standardized Precipitation Index (SPI) along with the information of dry spell. Looking for future changes in critical meteorological parameter, standardized precipitation index (SPI) new research study (Guhathakurta et al. 2017) projected that out of 32 meteorological subdivisions seven are serious track of spatial increase of drought hit districts mainly covering states like Uttar Pradesh Uttarakhand, Jharkhand, Chhattisgarh, Kerala and seven sister states of India. Although many areas of India located on eastern side and northeast region have increasing percentage of areas hit by water shortage condition and drought, but still the SPI-based trends are not significant (SPI values with moderate dry, severely



Fig. 4.4 Probability of occurrences of (**a**) moderate drought, (**b**) severe drought, (**c**) extreme drought, (**d**) combination of all drought types: Reproduced from Guhathakurta, P., Menon, P., Inkane, P., Krishnan, U. and Sable, S., 2017. Trends and variability of meteorological drought over the districts of India using standardized precipitation index. Journal of Earth System Science, 126(8)

dry or extremely severe dry conditions). Significantly decreasing trends are no where observed in Indian region. The probability stating that drought will occur in every district under three main categories of drought, i.e. moderate, severe or extremely severe and sum of all, over India using 1901–2015 data are shown in Fig. 4.4. Almost all states have some districts vulnerable to extreme drought conditions.

The evidence of observed climate change data during twentieth century highlighted the food grain availability issue and region-specific climate modifications have shown their signatures to noticeable extent to the scientific community. Nowadays, various models duplicating the various crop processes known as simulation models are being used to understand the influence of climate change on production and management practices of crops. The output generated by simulation models with some value addition can be shared with farmers, which can guide them to opt better management skill and technology and suggest them appropriate alternatives for their farming, thus making agriculture economically stable. In India, substantial research and analysis work has been done focusing on building an understanding of nature and amount of change in yield of various crops under the conditions of forecasted climate change (Aggarwal 2003; Uprety and Saxena 2021). The repeated droughts and massive floods seriously imperiled the working and earning conditions of billions of people depending on land and other natural resources for most of their needs (Hamdani Rizwana and Pathan 2016). The failure of monsoon increases pressure on available forest resources like trees, grasslands and water ponds and over-exploitation accelerates drought during dry years, hitting hard to soil moisture content and increasing dryness in soil making it loose and vulnerable for erosion.

The agricultural response to climate change will be the major deciding factor about the future food security in South Asia. Agricultural sector is very much sensitive for climate variations but it is one of the critical drivers for climate change. The modification of agricultural management practices according to weather changes and yield optimization are challenges for agricultural sector. The regional variation of weather parameters like temperature, rainfall, evapotranspiration, sun shine hours, crops and cropping patterns brings uncertainty in agricultural response to climate change. The crop damage/losses may be increasing if the forecasted climate change crosses the climate variability limit.

Scientific analysis and research done by Indian Agricultural Research Institute (IARI) and various other institutions estimated more loss/damage for Rabi season crops. Corresponding to each 1 °C increase in temperature, wheat production gets reduced about 4–5 million tonnes. Even small variations in temperature and precipitation have considerable impact on the quality and quality of various fruits, green vegetables, tea, coffee, aromatic plants and medicinal plants and basmati rice. The population of pathogens and insects have direct link with temperature and humidity of air. The alterations in weather condition play a dominant role in regulation of insects and pathogen population. The other diverse impacts of climate change on allied agricultural sectors include low productivity from dairy cattle and slow rate of fish breeding, migration of agri-employed population and poor harvests. Agricultural efficiency/output is reactive to two main types of broad climate-associated effects: primary effects by variation of temperature, rainfall and carbon dioxide concentrations and secondary effects from soil moisture and diverse spread and occurrence of pest attack and linked diseases. The productivity of main food crops rise and wheat estimated to decrease by considerable quantity with forecasted climate changes (IPCC 2007).

The climate change affects physiological response of the plant enhancing vulnerability of agricultural production system. The estimated net revenue loss at the farm level is around 9% to 25% for 2–3.5 °C increase in temperature. Scientific community also calculated that a 2 °C increase in average temperature and a 7% enhancement in average rain would decrease total revenues by 12.3% for entire India. Agricultural activities and production in coastal land areas of Gujarat, Maharashtra, and Karnataka states are bearing most adverse impacts of climate change. Agricultural losses of various extent come to notice form the major food-grain producing parts of states like Punjab, Haryana, and western Uttar Pradesh. On another front, states like West Bengal, Orissa, and Andhra Pradesh are forecasted to gain advantage to little extent due to warming (Aggarwal et al. 2004).

The period of unusual dryness is a normal feature of the climate and weather system in semi-arid and arid regions of the tropics, which covers more than one third of the land surface and is vulnerable to drought and desertification (Nagarajan 2003). Drought is seen as an outcome of monsoon failure causing to be occurring due to monsoon failure causing crop under stress, demand for supportive management practices. Intense drought leads to complete crop damage, fast drying of natural ecosystems and lack of drinking water leading to undue hard work for the rural and urban communities (NDMD Moha. GOI 2000).

Maharashtra state has some drought-affected districts almost every year but some districts received even less than half of normal rain during two consecutive years, almost 15,000 villages and 15 districts declared drought affected. So continuous poor monsoon for 2 years caused the question of livelihood for the rural and semiurban papulation. Monsoon failure generates need of immediate help for villagers for their survival (Hamdani Rizwana and Pathan 2016; Warde 2016).

4.8.1 Remedial Measures

- 1. Effective monitoring and regulation of available sources of water in areas encountering problem of drinking water.
- 2. Arrangement of green fodder for cattle and other animals in drought-affected villages and districts.
- 3. Provide alternative means for livelihood and assured supply of eatable items and food grains to people through effective management of public distribution system (PDS).
- Farmer awareness for adoption of drought-resistant varieties of crops and weather-linked new cropping practices and technologies.

4.8.2 Impact of South Asia Monsoon on Society

The high variability of South Asian monsoon and climate change are hitting hard densely populated South Asian region. The impacts of climate extremes are closely tied to population distribution, regional geography and development activities. Monsoon in Indian subcontinent demonstrate large-scale atmospheric processes and important component of global climate system. Flood, drought, landslide and thunderstorms are the associated extremes affecting the population. Floods are natural disaster that account for 50% of the mortalities occurring because of extreme weather events (EWEs; Ray et al. 2021). Flood-associated mortalities reduced about 3-5% per decade since 1980 to 2009 and 20% reduction is seen in the recent decade, even with rise in flood frequency in India. The average frequency of cyclones per year has increased a little (2.0) in comparison to years of previous decades (1.7). The associated mortality rates have declined by 94% during last two decades (Ray et al. 2021) mainly due to improved model forecast, accurate track prediction and good management by Government of India (GoI). Loss of human life, livestock, crops and property, damage to kuttchha houses, roads, and railway tracks, breaking of bridges and dams, inundation of low lying areas, disruption of electric lines and local connecting routes, overflow of rivers and deterioration of water quality due to mixing of water coming from various sources are the immediate impacts of flood event. Rural employment, food security, migration of population for livelihood and health issues like waterborne diseases and with time lag vector-borne diseases spread or outbreak in affected population. Some well-known waterborne diseases are typhoid fever, cholera, leptospirosis and hepatitis A and vector-borne diseases are malaria, dengue and dengue haemorrhagic fever, yellow fever, West Nile Fever, etc. These are commonly observed during and after flood situation. Flood are known for associated diarrhoeal diseases causing significant morbidity and mortality in flood prone areas of the world. According to World Health Organization (WHO), flood of 1988 in Bangladesh, diarrhoeal disease accounted for 35% of all types of illnesses associated to floods and 27% of 154 deaths associated to flood out of total 45,000 patients in rural areas of Bangladesh. Large number of cholera cases were reported during 1998 floods in West Bengal. The coastal populations in South Asia are becoming more vulnerable for cholera outbreaks with increased SST and algal blooms (Huq and Reid 2007). Thunderstorms and lightening events are common in monsoon season and severe thunderstorm/lightening associated strong wind damage crops, trees and structures and even claim life of people. Lightning events and associated mortality rates are showing significantly increasing trend over the past four decades (Ray et al. 2021). The farmer community mostly hit hard by flood and excess rain. Excess rain damages the crop and reduces productivity. The problem of soil erosion is quite common. Specially in coastal areas where cyclone brings salty water to land during flood situation creates problem of soil fertility. The cities have grown with fast pace during past decades and generated problem of urban floods. The urban floods are more impactful as the urban areas are the main controller of economy. Most of the urban floods are not due to weather but due to human unplanned development, rapid urbanization and lack of proper drainage system and also effective application of technological tools during disaster management process.

Drought is climatic extreme which is subjective in nature and shows its signature almost every year in various regions of Indian subcontinent. Drought causes crop failure, depletion of surface water resources, more exploitation of ground water and creates food scarcity for the population leading to migration of population from one place to another in search of livelihood. Climatic extremes like flood and drought threat food security, food items become inaccessible for some fraction of population and social fragmentation becomes more evident in population. Looking at the past eighteenth century which was famous for long droughts witnessed great political of turmoil across Southeast Asia, during extended periods of drought of the past centuries. Coming to recent decades, water availability and crop-linked four kinds of droughts, i.e. meteorological, hydrological, soil moisture also known as soil water and greenery or vegetation drought, have high seasonal and spatial variation. The hydrological and greenery/vegetation droughts cause more impact on growth, quality and quantity of crop than meteorological and soil moisture droughts (Xiang et al. 2017).

Extreme climatic events as visually generate many adversities in social life and act as natural cleaning agents of natural and social atmosphere. During extremes various faults in management and technological practices and social gaps become more evident and thus opens the path for the science and society to identify the need and accordingly guiding the scientific community too.

4.9 Summary

The chapter deals with various kinds of monsoonal extremes such as floods and droughts resulting from accumulative variation in monsoon season and landslides and flash floods like weather extremes caused by lows and depressions and climate change aspects over South Asia. Along with extremes large-scale circulation features of monsoon, i.e. semi-permanent synoptic systems and probable implication of climate change, have been discussed.

The ocean-atmosphere coupling indicator like ENSO and El Nino, La Niña conditions have been linked with floods and droughts in Indian subcontinent but all the droughts are not due to El Nino and all the flood extremes are due to La Niña. The connection of ENSO with inter-annual variability (IAV) of the Indian monsoon rainfall has weakened while relationship with the northwest (NW) of North Atlantic sea surface temperature (SST) has become strong during recent decades. The ENSO monsoon relationship is changing due to shifting of walker circulation anomalies south-eastward and mid latitude warming. Also the IOD behaved complementary to ENSO during past few decades. The seasonal rainfall time series epochs of belownormal/above-normal rainfall coincide El Niño/La Niña resulted in major climatic extremes like flood and drought. Overall seasonal extremes are governed by external factors like ENSO, IOD, NIO, etc. and through natural dynamics. Monsoon intraseasonal oscillations (MISO) are observed through low-pressure systems (LPSs) forming in Indian Ocean region and moving North, North Northwest, North West Wards give extreme rainfall over large geographic area covering many cities. More LPSs form in Bay of Bengal (BOB) than Arabian Sea. Recent years depressions are less as compared to lows. Why lows are not intensifying to depression or less depressions in monsoon season is still not clear. Reduction of relative humidity

between 600 hPa and 300 hPa levels of atmosphere over the Indian latitudes and pollution claimed for less depressions during monsoon.

The precipitation extremes linked with monsoon are very likely to increase over East, South, and Southeast Asia. All climate models project are projecting an increase in both the mean and extreme precipitation in Southern Asia (SAS) during late twenty-second century.

The frequency and probability of extreme rainfall events increased over South Asian regions and likely to be on same track under climate change impacts but contrasting trends also observed in various regions on India, Bangladesh and Pakistan.

The significant increasing trend of frequency of rainfall extremes (rain above 124.4 mm during 24 h) mainly at districts of west coast and some land areas of central and northeast region of India. The increased number of rainfall extremes is compensated by a considerable decrease in number of moderate and heavy precipitation events (≤ 64 mm in a day) and balancing overall monsoon robustness. There is no statistically significant trends for annual, seasonal, and monthly rainfall over Indian region whereas mean annual rainfall increasing trend of 5.675 mm/y. for Bangladesh during (1966–2015). The frequency of rainy days is decreasing significantly over central, north and some areas of eastern parts of India and wet days frequency is increasing over peninsular India. Single-day extreme rain events has increased over coastal states and parts of northeast India and some parts of Bangladesh. The core monsoon region, i.e. central India, extreme associated to heavy rain events are influenced by variations of SST over Indian Ocean and number of extreme rainfall events estimated to rise in coming future under warming and associated floods will also increase. Floods are frequent in South Asia. Nepal, Bangladesh and Indian states encounter flood during monsoon. The flood water carries silt load and deposits in plains; the course of river also changes sometimes, like river Kosi in Bihar. Floods submerge hectares of farmland and destroy crops, loss of life, livestock and property and cause landslides in hilly regions. India and Nepal have 21% of global rainfall triggered landslides mainly during monsoons. The problem of safe shelter to the people and water quality arises concerns mainly during floods. Bangladesh is one of the world's most densely populated countries and prone to cyclones and flooding due to its low-lying position, so it is more climate-sensitive economy. Major floods are quite often in phase with active spell and drought is accumulative picture of dry days. The drought is subjective in nature and methodologies for assessment and parameters under consideration vary from region to region. The number of dry days, dry spell information, Standardized Precipitation Index (SPI) values along with rainfall departure from normal have been included in the list of important parameters for drought declaration in India. Area under droughtaffected districts is showing significant increasing trends or no trend in India. Drought situations occur every year in many states of India and some coastal states like Tamil Nadu, Kerala and Karnataka. Flood and drought both lead to question of food security and health issues due to diseases outbreak linked with post flood weather conditions.

Coming to climate change and warming aspects, South Asian regions observed decline in precipitation from 1900 to 2005. The monsoon extremes very likely to increase over East, South, and Southeast Asia. The recent Coupled Model Intercomparison Project Phase 5 (CMIP5) simulations with Representative Concentration Pathway (RCP) scenarios are projecting greater than 2 degree rise in average mean yearly temperature with respect to reference level temperature of period 1970–1999 twentieth century estimated by ensemble of climate over huge and extensive land mass of the world during middle of twenty-first century under RCP8.5, and above 3 °C over continental land areas of Southeast sector of Asia in the late twenty-first century. Increased frequency of heavy rainfall with non-uniform spatial and temporal distribution, more number of warm days, and less cold days will be increasing and create acute water scarcity problem amplified by rising population and its demands, huge pressure on natural resources and ecosystems, extinction of more species, and loss of biodiversity.

The large-scale circulations associated with SAM will be modulating under the warming atmosphere. Weakening of circulations projected by models but extremes will increase. The future modulation of six semi-permanent synoptic circulation will lead to more extreme events. The future poleward shift of LLJ. The position of Tibetan anticyclone and associated wind strength determining reversal of temperature gradient in troposphere will be playing crucial role for monsoon advancement in South Asia. Large-scale circulations inducing/favouring regional circulation modifying regional weather and governing internal variability of monsoon will be more pronounced under changing climate. The projections of climate models with probable emission ranges have deviations and uncertainties. The reduction in range of uncertainty is required for planning technology driven solutions and sustainable practices for better adaptability of changing climate.

References

- Agarwal KS (1952) Fluctuations of annual rainfall in central India 1872-1947. Indian J Meteorol Geophys 3:229–230
- Aggarwal PK (2003) Impact of climate change on Indian agriculture. J Plant Biol 30(2):189-198
- Aggarwal PK, Joshi PK, Ingram JSI, Gupta RK (2004) Adapting food systems of the Indo-Gangetic plains to global environmental change: key information needs to improve policy formulation. Environ Sci Pol 7:487–498
- Ananthakrishnan R, Thiruvengadathan A (1966) Biennial oscillation in the equatorial troposphere. Nature 212(5069):1443–1443

Ashok K, Guan Z, Yamagata T (2001) Impact of the Indian Ocean dipole on the relationship between the Indian monsoon rainfall and ENSO. Geophys Res Lett 28(23):4499–4502

Blanford HF (1886) Rainfall of India. Mem Ind Met Dept 2:217-448

- Boos W, Emanuel K (2009) Annual intensification of the Somali jet in a quasi-equilibrium framework: observational composites. Q J R Meteorol Soc 135(639):319–335
- Brahmananda Rao V, Ferreira C, Franchito SH, Ramakrishna SSVS (2008) In a changing climate weakening tropical easterly jet induces more violent tropical storms over the north Indian Ocean. Geophys Res Lett 35(15):15710. https://doi.org/10.1029/2008GL034729
- Buckley BM, Fletcher R, Wang S-YS, Zottoli B, Pottier C (2014) Monsoon extremes and society over the past millennium on mainland Southeast Asia. Quat Sci Rev 95:1–19

- Christensen JH, Hewitson B, Busuioc A, Chen A, Gao X, Held I, Jones R, Kolli RK, Kwon WT, Laprise R, Magaña Rueda V, Mearns L, Menéndez CG, Räisänen J, Rinke A, Sarr A, Whetton P (2007) Regional climate projections. In: Climate change 2007: the physical science basis. Contribution of working group I to the fourth assessment report of the intergovernmental panel on climate change. Cambridge University Press, Cambridge and New York
- Dash SK, Rajendra Kumar J, Shekhar MS (2004) On the decreasing frequency of monsoon depressions over the Indian region. Curr Sci 86(10):1404–1411
- Debnath GC, Das GK, Sunithadevi S, Singh C (2019) Summer monsoon onset over Andaman & Nicobar Islands: objective criteria for operational forecaster. J MAUSAM 70:121–132
- Francis PA, Gadgil S (2006) Intense rainfall events over the west coast of India. Meteorog Atmos Phys 94:27–42. https://doi.org/10.1007/s00703-005-0167-2
- Froude MJ, Petley DN (2018) Global fatal landslide occurrence from 2004 to 2016. Nat Hazards Earth Syst Sci 18:2161–2181
- Gadgil S (2018) The monsoon system: land-sea breeze or the ITCZ? J Earth Syst Sci 127(1). https:// doi.org/10.1007/s12040-017-0916-x
- Gadgil S, Joseph P (2003) On breaks of the Indian monsoon. J Earth Syst Sci 112(4):529-558
- Gariano SL, Guzzetti F (2016) Landslides in a changing climate. Earth-Sci Rev 162:227–252. https://doi.org/10.1016/j.earscirev.2016.08.011
- Godbole RV (1977) The composite structure of the monsoon depression. Tellus 29(1):25-40
- Goswami BN (1987) A mechanism for the West Northwest movement of the monsoon depressions. Nature 326:370–376
- Goswami B, Mohan R (2001) Intraseasonal oscillations and interannual variability of the Indian summer monsoon. J Clim 14(6):1180–1198
- Goswami BN, Ajayamohan RS, Xavier PK, Sengupta D (2003) Clustering of low pressure systems during the Indian summer monsoon by intraseasonal oscillations. Geophys Res Lett 30(8):1431. https://doi.org/10.1029/2002GL016734
- Goswami B, Venugopal V, Sengupta D, Madhusoodanan M, Xavier P (2006) Increasing trend of extreme rain events over India in a warming environment. Science 314(5804):1442–1445
- Guhathakurta P, Rajeevan M (2008) Trends in the rainfall pattern over India. Int J Climatol 28(11): 1453–1469
- Guhathakurta P, Sreejith O, Menon P (2011) Impact of climate change on extreme rainfall events and flood risk in India. J Earth Syst Sci 120(3):359–373
- Guhathakurta P, Menon P, Inkane P, Krishnan U, Sable S (2017) Trends and variability of meteorological drought over the districts of India using standardized precipitation index. J Earth Syst Sci 126(8)
- Hamdani Rizwana MJ, Pathan MG (2016) Drought causes and its impact. In: National seminar on problem of drought & its remedies. ISBN 978-93-83672-23-3
- Hernandez M, Ummenhofer CC, Anchukaitis KJ (2015) Multi-scale drought and ocean–atmosphere variability in monsoon Asia. Environ Res Lett 10:1748–9326
- Huang X, Zhou T, Dai A, Li H, Li C, Chen X, Lu J, Von Storch J-S, Wu B (2020) South Asian summer monsoon projections constrained by the interdecadal Pacific oscillation. Sci Adv 6(11): eaay6546
- Huq SR, Reid H (2007) Community-based adaptation: a vital approach to the threat climate change poses to the poor. IIED Briefing, International Institute for Environment and Development (IIED), London, p 2
- IPCC (2007) Climate change: the physical science basis. Extracts from the IV Assessment Report. Survey of the Environment, The Hindu, pp 147–155
- Kothawale DR, Revadekar JV, Rupa Kumar K (2010) Recent trends in pre-monsoon daily temperature extremes over India. J Earth Syst Sci 119(1):51–65
- Kothyari U, Singh V, Aravamuthan V (1997) Water resources management. 11(1):17-34
- Kripalani RH, Oh JH, Kulkarni A, Sabade SS, Chaudhari HS (2007) South Asian summer monsoon precipitation variability: coupled climate model simulations and projections under IPCC AR4. Theor Appl Climatol 90:133–159

- Krishna KK, Rajagopalan B, Cane MA (1999) On the weakening relationship between the Indian monsoon and ENSO. Science 284:2156–2159
- Krishnamurti TN, Martin A, Krishnamurti R, Simon A, Thomas A, Kumar V (2013) Impacts of enhanced CCN on the organization of convection and recent reduced counts of monsoon depressions. Clim Dyn 41:117–134. https://doi.org/10.1007/s00382-012-1638-z
- Krishnan R, Zhang C, Sugi M (2000) Dynamics of breaks in the Indian summer monsoon. J Atmos Sci 57(9):1354–1372
- Krishnan R, Sabin TP, Ayantika DC, Kitoh A, Sugi M, Murakami H, Turner AG, Slingo JM, Rajendran K (2012) Will the South Asian monsoon overturning circulation stabilize any further? Clim Dyn 40:187–211
- Kumar KR, Pant GB, Parthasarathy B, Sontakke NA (1992) Spatial and subseasonal patterns of the long-term trends of Indian summer monsoon rainfall. Int J Climatol 12:257–268. https://doi.org/ 10.1002/joc.3370120303
- Kumar V, Jain S, Singh Y (2010) Analysis of long-term rainfall trends in India. Hydrol Sci J 55(4): 484–496
- Lau W, Li MT (1984) The monsoon of East Asia and its global associations—a survey. Bull Am Meteorol Soc 65:114–125
- Mandke S, Sahai A, Shinde M, Joseph S, Chattopadhyay R (2007) Simulated changes in active/ break spells during the Indian summer monsoon due to enhanced CO2 concentrations: assessment from selected coupled atmosphere–ocean global climate models. Int J Climatol 27(7): 837–859
- Meehl GA (1992) Effect of tropical topography on global climate. Annu Rev Earth Planet Sci 20: 85–112
- Mooley D, Parthasarathy B (1983) Variability of the Indian summer monsoon and tropical circulation features. Mon Weather Rev 111(5):967–978
- Mullicka MRA, Mohammed NR, Alam MJ, Islam KMA (2019) Observed trends in temperature and rainfall in Bangladesh using pre-whitening approach. Glob Planet Chang 172:104–113
- Nagarajan R (2003) Drought assessment monitoring management and resource conservation. Capital Publishing Company, New Delhi, p 312
- Naidu C, Srinivasarao B, BhaskarRao D (1999) Climatic trends and periodicities of annual rainfall over India. Meteorol Appl 6(4):395–404
- NDMD Moha. GOI (2000) Contingency plan-drought 2000, Department of Agriculture and co-operative. MoAg GOI, New Delhi
- Pai D, Bhate J, Sreejith O, Hatwar H (2009) Impact of MJO on the intraseasonal variation of summer monsoon rainfall over India. Clim Dyn 36(1–2):41–55
- Parthasarathy B, Dhar ON (1976) A study of trends and periodicities in the seasonal and annual rainfall of India. Indian J Meteorol Hydrol Geophys 27:23–28
- Pattanaik DR (2007) Analysis of rainfall over different homogeneous regions of India in relation to variability in westward movement frequency of monsoon depressions. Nat Hazards 40:635–646. https://doi.org/10.1007/s11069-006-9014-0
- Pattanaik D, Rajeevan M (2010) Variability of extreme rainfall events over India during southwest monsoon season. Meteorol Appl 17:88–104
- Prajeesh AG, Ashok K, Bhaskarrao DV (2013) Falling monsoon depression frequency: a Gray-Sikka conditions perspective. Sci Rep 3:2989. https://doi.org/10.1038/srep02989
- Preenu P, Joseph P, Dineshkumar P (2017) Variability of the date of monsoon onset over Kerala (India) of the period 1870–2014 and its relation to sea surface temperature. J Earth Syst Sci 126(5)
- Preethi B, Mujumdar M, Prabhu A, Kripalani RH (2017) Variability and teleconnections of South and East Asian summer monsoons in present and future projections of CMIP5 climate models. Asia-Pac J Atmos Sci 53:305–325
- Rajeevan M, Bhate J, Kale JD, Lal B (2006) High resolution daily gridded rainfall data for the Indian region: analysis of break and active monsoon spells. Curr Sci 91(3):296–306

- Rajeevan M, Bhate J, Jaswal A (2008) Analysis of variability and trends of extreme rainfall events over India using 104 years of gridded daily rainfall data. Geophys Res Lett 35(18)
- Rajeevan M, Gadgil S, Bhate J (2010) Active and break spells of the Indian summer monsoon. J Earth Syst Sci 119(3):229–247
- Raju P, Mohanty U, Bhatla R (2005) Onset characteristics of the southwest monsoon over India. Int J Climatol 25(2):167–182
- Ramage CS (1971) Monsoon meteorology. Academic, New York and London, p 296
- Ramamurthy K (1969) Monsoon of India: some aspects of the 'break' in the Indian southwest monsoon during July and August. Forecasting manual 1–57 No. IV 18.3, India Met. Dept., Poona, India
- Ramaswamy C (1965) On synoptic methods of forecasting the vageries of southwest monsoon over India and neighbouring countries. In: Proc. Symp. Meteor. Results IIOE, Bombay, pp 317–349
- Rasmusson E, Carpenter T (1983) The relationship between eastern equatorial Pacific Sea surface temperatures and rainfall over India and Sri Lanka. Mon Weather Rev 111(3):517–528
- Ray K, Giri RK, Ray SS, Dimri AP, Rajeevan M (2021) An assessment of long-term changes in mortalities due to extreme weather events in India: a study of 50 years' data, 1970–2019. Weather Clim Extremes 32:100315
- Salma S, Rehman S, Shah MA (2012) Rainfall trends in different climate zones of Pakistan. Pak J Meteorol 9:37–47
- Sandeep S, Ajayamohan RS (2015) Poleward shift in Indian summer monsoon low level jetstream under global warming. Clim Dyn 45:337–351. https://doi.org/10.1007/s00382-014-2261-y
- Shrestha AB, Bajracharya SR, Sharma AR, Duo C, Kulkarni A (2017) Observed trends and changes in daily temperature and precipitation extremes over the Koshi river basin 1975–2010. Int J Climatol 37(2):1066–1083
- Shukla J (1987) Interannual variability of monsoons. Monsoons: 399-464
- Sikka DR (1977) Some aspects of the life history, structure and movements of monsoon depressions. Pure Appl Geophys 115:1501–1529
- Sikka D, Gadgil S (1980) On the maximum cloud zone and the ITCZ over Indian, longitudes during the southwest monsoon. Mon Weather Rev 108(11):1840–1853
- Sikka DR, Gray WM (1981) Cross-hemispheric actions and the onset of summer monsoon over India. In: International conference on Sci. Results of monsoon experiments, Bali, Indonesia, 26–30 Oct, pp 3–74 to 3–78
- Singh R, Mal S (2014) Trends and variability of monsoon and other rainfall seasons in Western Himalaya, India. Atmos Sci Lett 15(3):218–226
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sooraj KP, Terray P, Mujumdar M (2015) Global warming and the weakening of the Asian summer monsoon circulation: assessments from the CMIP5 models. Clim Dyn 45:233–252
- Srivastava G, Chakraborty A, Nanjundiah R (2020) Multidecadal variations in ENSO-Indian summer monsoon relationship at sub-seasonal timescales. Theor Appl Climatol 140(3–4): 1299–1314
- Suthinkumar P, Babu C, Varikoden H (2019) Spatial distribution of extreme rainfall events during 2017 southwest monsoon over Indian subcontinent. Pure Appl Geophys 176(12):5431–5443
- Swathi M, Izumo T, Lengaigne M, Vialard J, Kumar M (2020) Remote influences on the Indian monsoon low-level jet intraseasonal variations. Clim Dyn 54(3–4):2221–2236
- Turner AG, Annamalai H (2012) Climate change and the South Asian summer monsoon. Nat Clim Chang 2:587–595
- Uprety DC, Saxena P (2021) Technologies for green house gas assessment in crop studies. Springer, Singapore, pp 1–134
- Vidya PJ, Ravichandran M, Subeesh M, Chatterjee S (2020) Global warming hiatus contributed weakening of the Mascarene High in the Southern Indian Ocean. Sci Rep 10(1)

- Walker GT (1910) Correlation in seasonal variations of weather. II. Memoirs of the Indian Meteorological Department 21(Part 2):22–45
- Walker GT (1914) Correlation in seasonal variations of weather. III. On the criterion for the reality of relationships or periodicities. Memoirs of the Indian Meteorological Department 21(Part 9): 13–15
- Warde VA (2016) A conceptional study of causes and impact of drought on Indian Economy. National seminar on problem of drought & its remedies. ISBN 978-93-83672-23-3
- Webster PJ (1987) The elementary monsoon. In: Fein J, Stephens P (eds) Monsoons. Wiley-Interscience
- Xavier A, Kottayil A, Mohanakumar K, Xavier P (2017) The role of monsoon low-level jet in modulating heavy rainfall events. Int J Climatol 38:e569–e576
- Xiang Z, Obringer R, Wei C, Chen N, Niyogi D (2017) Droughts in India from 1981 to 2013 and implications to wheat production. Sci Rep 7(1). https://doi.org/10.1038/srep44552
- Yadav RK (2008) Role of equatorial Central Pacific and northwest of North Atlantic 2-metre surface temperatures in modulating Indian summer monsoon variability. Clim Dyn 32(4): 549–563. https://doi.org/10.1007/s00382-008-0410
- Yadav RK (2009) Changes in the large-scale features associated with the Indian Summer Monsoon in the recent decades. Int J Climatol 29(1):117–133. https://doi.org/10.1002/joc.1698
- Yao C, Yang S, Qian W, Lin Z, Wen M (2008) Regional summer precipitation events in Asia and their changes in the past decades. J Geophys Res 113:D17107. https://doi.org/10.1029/ 2007JD009603



5

Contribution of Fog in Changing Air Quality: Extremities and Risks to Environment and Society

Saraswati

Abstract

This chapter presents a brief review on the contribution of fog in changing air quality, its extremities and risks to environment and society. Fog is one of the extreme environmental phenomena which cause significant societal and economic problems specially as a great devastation to road and air traffic. As the hazardous effect of fog events is very high, their formation, spatial extent and evolution are needed to be investigated in detail. The detailed characterization of the formation and evolutionary mechanisms of fog pollution is necessary. The purpose of this chapter is to provide a brief outline on fog formation, development, distribution, characterization as well as its extremities and impacts on environment and society. In this chapter, we have started our study with the description of fog, its types and distribution worldwide, specifically in northern India. The microphysical structure, chemical composition and interactions which determine the behaviour of fog is also explained. The physicochemical characterization is described in detail which is necessary to understand the fog formation and its impact on environment and society. The chapter also describes the different factors responsible for fog formation. The different meteorological conditions and role of aerosol were explained which were amenable for fog formation. In the last section, the fog as an extreme event is described. The causes and their extremities on different aspects of environment and society are also provided in detail.

Keywords

 $Fog \cdot Microphysical \cdot Meteorological \cdot Formation \cdot Evolution$

Saraswati (🖂)

Environmental and Water Resource Engineering Division, IIT Madras, Chennai, India

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_5

5.1 Introduction

Fog is one of the important climatic phenomena which set up crucial societal and economic problems specifically as a notable disruption to air and road traffic. The many regions of the world are affected by fog (Gautam and Singh 2018; Saxena and Kulshrestha 2015) as it results in low visibility, has effects on air quality, disrupts transportation, increases economic losses and intensifies health risks. The different parts of the world including Po valley in Europe, the central valley of California, Indo-Gangetic Plains (IGP) and North China Plains (NCP) in south Asia are distressed with dense fog cover (Baldocchi and Waller 2014; Gautam et al. 2007; Ghude et al. 2017; Giulianelli et al. 2014; Hameed et al. 2000; Herckes et al. 2015; Quan et al. 2011). The fog is mostly observed in plain areas adjacent to continuous mountain ranges and extensively spread over hundreds of kilometres during the winter season. The steady boundary layer structure, calm winds, high relative humidity (RH), lower temperature and local radiative cooling are some of the characteristics of these regions (Duynkerke 1991). These prevalent factors result in the cooling of air near the earth's surface, provide moisture and are responsible for the formation of widespread fog in these regions. The build-up of fog is further intensified due to the availability of high aerosol concentrations and their accumulation in lower planetary boundary layer (PBL) in these polluted regions (Duynkerke 1991; Quan et al. 2011; Wang et al. 2016; Saxena et al. 2020). Hence, the height of the PBL plays a crucial role in restraining severe air pollution episodes and weather phenomena. The dispersion of particles is also affected by PBL in the concerned regions (Petäjä et al. 2016). The insight of characterization of hygroscopic particles present in the boundary layer and their effect is necessary, as visibility is reduced by these particles. The life cycle of fog is modified by atmospheric aerosol as they can act as cloud condensation nuclei (CCN) and impact the development of fog (Safai et al. 2019).

There is a tremendous challenge in relation to air quality, as frequency and severity of haze and extremities of fog event around the world have enhanced in recent years as a result of rapid urbanization and industrialization, which has caused great concerns worldwide (Chan and Yao 2008). The fog and haze events were observed and segregated into different phases like haze, wet haze, mist and fog in relation to relative humidity and visibility by Yang et al. (2010) in the winter of Nanjing (Yang et al. 2010). The haze, mist and fog extremities are more prevalent in the winter season and result in visibility reduction often even up to zero. In the winter season, the northern Indian cities are more affected by the visibility reduction as increasing pollution results in radical deterioration of air quality in these regions which further reduced the atmospheric visibility adequately (Goyal et al. 2014). The fog and haze particles present in the stable boundary layer attenuated the visible light (Pinnick et al. 1978) and result in lower visibility which greatly impact aviation. The travellers are badly distressed due to delays, diversions and cancellations of flights as a result of the existence of dense fog extremities. The airlines also suffered huge economic losses due to disruption in flight operations at airports (Kulkarni et al. 2019). The effect of fog and poor visibility on land, marine and air transportation results in huge financial and human losses which can be compared with other weather events like hurricanes, winter storms and tornadoes (Gultepe et al. 2007).

The extreme fog event becomes a topic of research for public, government and atmospheric scientists in the world to understand the mechanism behind its formation, as it greatly affects the human being. The multiple car accidents each year are the result of unsafe road conditions caused by fog. Most people have heard of the horrific chain reaction accidents which occur in fog. It is hardly surprising, given the effects of fog on perception. Fog obviously produces accidents because a driver cannot see as far ahead. The droplets in a fog vary in concentration, which can cause partial to zero visibility. Other methods of transport such as ships and planes are also badly affected by the poor visibility associated with fog (Saxena and Kulshrestha 2016). The aviation industry is particularly prone to the effects of fog, as limited visibility can make flying unsafe, especially during take-off and landing. The human health is also affected by fog episode in addition to reduction in visibility which severely affected transportation. Renhe et al. (2014) observed increase in cases of respiratory and cardiovascular diseases during these extreme fog events (Renhe et al. 2014). The discerning of physical and chemical characteristics of fog, meteorological factors responsible for its genesis, sustenance and intensity is necessary for the detailed characterization of the fog pollution and its extremities over various space and time scales (Gultepe et al. 2007). The initiation, persistence and enhancement of this severe and extreme fog event are contributed by large-scale circulation patterns, atmospheric circulations and boundary layer processes at all scales. This extreme fog event has a tremendous effect on air quality, visibility, human health and transportation.

Thus, the insight of complicated chemistry of different atmospheric activity accountable for the fog formation is necessary and the chemical makeup of aerosols is also required to estimate the influence of aerosols in the atmosphere which result in evaluation and improvement of existing fog forecasting systems. The comprehensive understanding of the mechanisms controlling fog formation and evolution is needed to provide much-needed knowledge benefiting public health, efficiency of transport as well as areas of environmental research and development. The persistence and enhancement mechanism of extreme fog event was described in detail in this chapter. This chapter reviews the fog types and distribution, physicochemical characterization, factors for its formation and evolution and extremities of this fog pollution phenomenon on environment and society.

5.2 Fog

Meteorological fog can be defined as thin cloud consisting of microscopically small water droplets which are kept in suspension in the air near the ground surface and reduce horizontal visibility (Canada and Toth 2010). The fog is a natural, climatic condition in the form of clouds coming close to ground surface, formed by warm air currents touching cold surfaces or when cold air currents blow on warm surfaces. Fog is formed when the moist air (with relative humidity above 97%) becomes saturated, reaches its dew point and further cools so that water vapour is condensed

International number	Fog	Maximum distance to see clearly
0	Opaque fog	50
1	Heavy fog	200
2	Show?? is very poor	500
3	Show is poor	1
4	Rusk??	2
5	Show weak	4
6	Show moderate	10
7	Opaque fog	20
8	Heavy fog	50
9	Show is very poor	More than 50

Table 5.1 Visibility through the various types of fog denoting with international number (0–9) (Source: Al-Mutairi 2017)

around dust particles, smokes etc. to form tiny water droplets which being very light are suspended in the air. It is a form of condensation, containing minute water drops with the diameter of each is no more than 100 μ m (Al-Mutairi 2017). The horizontal visibility is used as a criterion to define fog and haze by World Meteorological Organization (WMO). WMO reported that the fog can be defined with the visibility below 1 km and light fog or haze with the visibility in the range of 1 km and 5 km. The fog and haze cannot be identified separately when they are coexisted under the condition of pollution. The fog and haze can also be pinpointed on the basis of relative humidity (RH) as according to UK Met Office (UKMO). It represents fog when RH is 100%, mist, when RH is in the range of 95–100% and it represents haze when RH is less than 95% (Guo et al. 2015). The increasing haze aerosol could intensify the formation of fog and its build up as reported in some studies (Jia and Guo 2012).

Fog dissipates at sunrise when air temperature begins to increase. It is thicker in cities than in rural areas due to dust and grime pollution where dust and grime cling to the water drops. As a rule, upon formation of fog, visibility becomes low which leads to accidents, particularly in the event of thick fogs. Owing to the risks involved in fogs, scientists have developed an international fog measure (from 0 to 9) relying on the naked eye for observation (Table 5.1) (Al-Mutairi 2017).

5.3 Types of Fog

Fog is physically no different than a cloud. However, what differentiates one from the other is the location and mechanism of formation (Lutgens and Tarbuck 2004). Both fog and clouds form when air becomes saturated and the water vapour condenses. Fog may also form from cloud base lowering to the surface in relation to temperature inversions (Lewis 2004). Peter Nolan (2010) classifies fog according to the manner it is formed: it can occur as a result of the air being saturated with water vapour or through cooling the air until it has reached saturation point. So, the agents

that caused air saturation with vapour, the speed of evaporation and the factors affecting the cool air that is close to ground surface are the determinants for classifying fog types.

The approach of recognition and categorization of fog events into a particular category which emulate the primary mechanism accountable for their formation is deemed more insightful as it stipulates particular information about the disposition of formation mechanisms of fog (Croft and Burton 2006). The fog formation is the result of boundary layer structure, different meteorological conditions, synoptic causes and geographical conditions (Li et al. 2011b). Irrespective of the mechanism, fog has generally been observed to be associated with a stable atmosphere and calm wind conditions (Holton et al. 2003). For fog, there are two main mechanisms that can take place, namely: the addition of water vapour close to the surface and cooling. Both are further explained below.

5.3.1 Fog Formed Through the Addition of Water Vapour

Steam fog and frontal fog are examples of fog formed by this mechanism.

5.3.1.1 Steam Fog

Steam fog is a fog resulting from intense evaporation from water to relatively cool air, so the state turns from saturation to condensation to fog. Steam fog is formed when cold air travels over relatively warm water (Gultepe et al. 2007). The evaporation of surface warm water supplies the cold air above with water vapour. The colder temperature causes the water vapour to cool and condense, causing steam fog at the air-water interface (Gultepe et al. 2007). Steam fogs are produced with the moving of cold air from land to oceanic surface and there is evaporation of large quantity of moisture from water surface to saturate the overlying cold air. It is fascinating to notice that in such situation overlying cold air is warmed from below and thus water vapour due to evaporation of warm water surface rises upward and condenses after meeting cold air above to form fog. The vapour particles above the water surface look as if steam is coming up from the water. This is why such fogs are called steam fogs. They are also called evaporation fogs (Okland and Gotass 1995). Evaporation fog can be noticed over water bodies and in tropical areas. Sometimes it occurs in warm areas or wet lands, or immediately after rains as is the case in Northern Borders.

5.3.1.2 Frontal Fog

Frontal fog forms with the lifting of a warm air mass over a colder air mass and supplies it with moisture or is cooled to dew point (Gultepe et al. 2007). If the cold air is near dew point, a slight addition of moisture can lead to frontal fog. As the name suggests, this is formed along the edges of two masses or fronts of air: warm rain evaporates, falling from a higher layer, through a drier air mass, in a lower layer, thus bringing about a state of water-vapour-saturated air, followed by a state of condensation of colder layers thus forming frontal fogs (Stewart et al. 1995). Fronts

are formed when two contrasting air masses (warm and cold air masses) converge along a line. Warm air is pushed upward by cold air and hence overlying warm air is cooled from below due to underlying cold air and fogs originate after condensation (Guo et al. 2015). Such fogs are formed in temperate regions. In fact, the saturation and condensation of overlying warm and moist air causes rainfall which saturates the underlying cold surface air and thus fog is produced.

5.3.2 Fog Formed by Cooling

This mechanism involves the earth or ice surface to cool the overlying air. When air with a high relative humidity is cooled to below its dew point, condensation takes place and fog forms. The warm and moist air when blows over cold surface (either land surface or sea surface) is cooled from below so that dew point is reached and condensation of water vapour occurs around hygroscopic nuclei and thus microscopic water droplets are formed to create fogs. There are four main types of fog that are formed through this mechanism, namely radiation fog, advection fog, upslope fog and mixing fog.

5.3.2.1 Radiation Fog

Radiation fog is formed after nightfall when the ground has released its heat through emission of longwave radiation, and the ground is cooler than the overlying air (Gultepe et al. 2007). Due to this situation overlying warm and moist air cools and thus dew point is reached, with the result condensation of water vapour around hygroscopic nuclei (dust particles and smokes) forms numerous tiny water droplets and thus fog is originated. It is particularly formed when the wet air touching ground is fully quiet. As a result of cooling, saturation occurs, followed by condensation and then fog is formed. This takes place as a result of radiation at night time, when air is fully quiet, dew or frost is formed. At the slightest air motion, fog becomes deeper. If air motion further increases, fog disperses and clears up. As a rule, radiation fog occurs on flat ground (Bergot and Guedalia 1994). Under this scenario, fog formation has been shown to be sensitive to the coupled dynamical and thermodynamical structure of the evolving boundary layer over land (Duynkerke 1999; Roach 1995). The occurrence of radiation fog requires certain conditions, e.g., long and cool winter nights, cloudless sky, sufficient amount of moisture in the air, very weak air motion (light wind having speed of 3–5 km per hour) and ground inversion of temperature. Fog is formed at the ground surface during nights and thickness upward. It disappears in the morning with the sunrise because the water droplets are evaporated due to the rise in air temperature.

Advection-Radiation Fog

The radiative cooling of moist air in the coastal area results in advection-radiation fog. The advection of moist air takes place over land which is coming from the ocean or from any large water body during the previous daylight hours. The moisture availability and radiative cooling potential result in inland radiation fog in the fall season. The occurrence of radiation fog in the coastal plains in the spring season is due to the influence of advection inland of moist marine air during the previous afternoon (Ryznar 1977). The fog formed as a result of mixing of warm moist air and cold air due to the arrival of moist and warm air over the cold ground surface is called advection-radiation fog. There are certain required conditions for the origin of advection radiation fog, e.g. (1) horizontal movement of air, (2) greater contrast between air temperature and the temperature of the ground surface, (3) moist air or say high relative humidity in the air, and (4) stable stratification in the atmosphere, etc.

High Inversion Fog

The strong inversion over a deep moist layer in valleys results in the formation of high inversion fog (Holets and Swanson 1981). The prolonged subsidence and radiative cooling in association with a persistent anticyclone results in the inversion. The Central Valley of California is afflicted with inversion fog commonly during the winter season (Underwood et al. 2004). Radiation fog is more common over large cities and surrounding areas because of the abundance of hygroscopic nuclei. When fog is combined with sulphur dioxide it becomes poisonous and causes human deaths. Such fog is called urban smog. Sometimes, such fogs also occur at higher elevation due to upper air inversion of temperature.

5.3.2.2 Advection Fog

The formation of advection fog occurred when warm humid air travels over a cold surface such as sea ice. The cold surface cools the overlying warm moist air to dew point, condensation takes place, and advection fog is formed. It is formed when wet air passes over a cold surface, thus effecting saturation and dew drops formed on the ground. Further cooling of the air layer touching the ground results in high saturation rates. With the light air movement, air mixes up, leading to condensation on higher layers above ground level and so fog covers the whole area. Advection fogs are generally originated during winters on land surfaces and during summers on sea surfaces because lands are relatively colder than seas and oceans during winters (Gultepe et al. 2007).

Advection fog shows up mostly in places where warm, tropical air meets cooler ocean water. The Pacific coast of the United States, from Washington to California, is often covered in advection fog. The cold California Current, which runs along the western coast of North America, is much cooler than the warm air along the coast. Dense advection fogs are also formed where cold and warm ocean currents coverage (Guo et al. 2015). For example, dense fogs are formed near Newfound land due to the convergence of cool Labrador and warm Gulf Stream ocean currents. Similarly, dense fogs are formed near the Japanese coast due to the convergence of cool Kurile and warm Kuroshio currents (Duanyang et al. 2010; Pu et al. 2008). The different types of advection fog are explained below:

Sea fog is formed over the ocean with advection of relatively warm marine air over the cold sea ice in the summer during sea ice break-up (Lewis et al. 2003). The fogs occurring over sea surfaces are called sea fogs which are generally formed near
the coastal areas frequented by cold ocean currents. For example, cool California current (along the Californian coast), cool Peru current (along the Peruvian coast), cool Benguela current (along the western coast of South Africa) and cool Western Australian current cause sea fogs (Lewis 2004).

Land and sea breeze fogs are formed with the transportation of warm moist air from land to cool coastal ocean. The sea-breeze circulation results in the moving of this fog over the land during the following afternoon hours. The formation of land and sea breeze fogs is particularly a coastal occurrence.

Tropical air fog is also a kind of advection fog. The air mass is gradually cooled due to ocean temperature gradient and long-range transport of tropical air towards the pole (Friedlein 2004) during this phenomenon.

5.3.2.3 Upslope Fog

Upslope fog, on the other hand, is formed when relatively humid air is cooled by moving up a sloping terrain. Upslope or hill fogs originate when continental warm and moist air rises upslope along the hillslopes because the rising air is saturated due to cooling and condensation of moisture around hygroscopic nuclei and forms fogs which cover the lower segments of hillslopes (Gultepe et al. 2007; Lutgens and Tarbuck 2004). It may be mentioned that such fogs are formed due to adiabatic expansion and cooling of moist air. Such fogs are very common on the hillslopes in temperate regions. They may occur in any season (Nolan 2010).

5.3.2.4 Mixing Fog

The formation of mixing fog occurs when a wet, warm air mass meets a wet, cold one. This mixture results in a temperature drop enough to effect saturation and condensation, yielding mixed fog. An example of mixing fog is when a person exhales in very cold days in a mist that he can see before his eyes. The fog lowers visibility on highways and open spaces, endangering lives. The reason for these in most of the cases is the drop in the temperature of the air that is close to ground so much so that it gets to zero or below, accompanied with a considerable increase in the rates of relative humidity during night time, especially in winter following rains. The Northern Borders witness annually a number of fog formation cases which has negative impacts on people's health and life since it limits visibility on roads leading to accidents (Nolan 2010).

5.4 World Distribution of Fog

If the world distribution of fogs is considered it becomes apparent that they are more widespread over oceans. The oceanic fogs are generally advectional in origin because relatively warm air coming from over the land surfaces produces fog over relatively colder oceanic surfaces. Fog areas are positively related to cool ocean currents. The eastern portions of the oceans and western continental margins are characterized by most fogs because these areas are frequented by cool ocean currents (e.g. California current, Peru current, Benguela current, Canary current, Western

Australia current, etc.). Frontal fogs are produced in the high latitudes due to convergence of cold polar air mass and warm westerlies mainly in the eastern North Atlantic Ocean and western coasts of north-west Europe and in the eastern North Pacific Ocean near Aleutian Islands and Alaska coast (Guo et al. 2015). Radiation fogs are produced in low latitudes on continental areas during the winter season. Dense fogs are formed at the places where cold and warm Ocean currents converge, e.g., near Newfoundland and Grand Banks (due to convergence of cold Labrador current and warm Gulf Stream) and near Japanese coasts (due to convergence of cold Kurile current and warm Kuroshio Current). Important foggy areas of the USA are California coast, New England outer coast, Northern Pacific coastline, Appalachian valleys, Pacific coast valleys, Middle Atlantic coast, Great Lakes, Southern Atlantic and Gulf coastal waters, Ohio, Missouri, Great Plains and upper Mississippi valley.

5.4.1 Northern India Fog

The pervasive layer of fog can usually be observed with the satellite during the winter season in the northern India, which is the remarkably fog prone areas in the world (Gautam et al. 2007). The stable atmospheric conditions, vigorous temperature inversions, probable amount of moisture as a result of western disturbance (WD) (Dimri et al. 2015) and prevalent terrain characteristics of Indo-Gangetic plains of Northern India lead to fog formation and its subsistence during the winter season in this region which result in lower visibility. Compared to other areas of the world, fog events observed over the northern region of India form rapidly, with largest coverage area and longest duration. The current research on fog in India during the past 10–15 years have raised significant concerns as a result of enhancement in frequency, persistence and severity of fog occurrence over the northern India (Jenamani 2007; Syed et al. 2012). The fog days were gradually increasing for the last three decades, starting from the 1980s. The days having visibility <200 m are represented as dense fog days which enhanced from 10 days in 1980s to 20 days in recent years over Delhi as recorded by some studies (Jenamani and Kalsi 2012: Ghude et al. 2017). The 60% days having fog events and 16% days with dense fog events are indicated by the research on instances of fog in Delhi (Jaswal et al. 2013; Safai et al. 2019). Maximum fog and dense fog events are mostly seen in December and January. A similar trend is also seen in eastern-central China and linked to increased urban aerosol loading (Niu et al. 2010). The results can be interpreted as an indication of the role of increasing air pollution and humidity (Jenamani 2007; Syed et al. 2012) to increase fog events over this region. The atmospheric aerosol nuclei emitted from various human activities along with favourable meteorological condition result in the formation of haze and fog in the winter season in Indo-Gangetic plain (IGP) region of India (Badarinath et al. 2007). The interaction amid particulate matter (PM) gaseous pollutants and atmospheric water vapour affect the chemical composition of fog and also reduce visibility on the basis of concentrations, size and chemical characteristics of the particles. The condensation nuclei are particularly

formed due to small particles such as accumulation mode particles which are also very effective in lowering visibility (Goyal et al. 2014). The annual morning poor visibility days (PVDs < 4 km) in India have enhanced from 6.7% days in 1961 to 27.3% days in 2008 (Jaswal et al. 2013).

The rapid urbanization and industrialization result in lower visibility during prolonged fog phenomena in India which caused disruption in trains and flight operations and large vehicular accidents. These severe fog events even affect human health, disrupt daily life and sometimes fully shut down airports services for hours as a result of zero visibility induced by dense fog extremities in the winter season (Badarinath et al. 2009; Chandra et al. 2018). There is serious compulsion from air, road and rail transport authorities to know whether a fog will occur during the next 24 h, and if so, what could be its intensity, onset time and dissipation rate. The understanding of fog physical characteristics, factors responsible for fog genesis, intensity and expeditious thickening of the fog layer over IGP is necessary.

5.5 Characterization of Fog

The understanding of microphysical structure, chemical composition and interactions is required for determining the behaviour of fog. The in-depth exploration of microphysical and chemical properties of fog is needed for understanding its formation and development and also for discerning its aftermath on transportation and flight operation. Thus, physicochemical characterization is necessary to discern the generation of fog and its impact on environment and society.

5.5.1 Microphysical Structure of Fog

Once fog is formed, it is largely the atmospheric stability but also the fog's microphysical structure that determines its occurrence, intensity, duration, horizontal and vertical extent. The microphysical structure is determined by the type of condensation nuclei, local nuclei populations, drop size distribution and atmospheric chemistry. An essential process in fog formation is nucleation. Condensation nuclei are agents that facilitate droplet formation due to their properties such as surface curvature, electric charge, ionic charge and hygroscopy (water seeking), that make them effective sites for condensation (Gueye 2014; Willett 1928). The surface with fog formation plays a crucial role in determining the type of microphysical structure.

The accumulation of aerosol, transition process of mixing of aerosol and fog and dissipation are the three main stages of the process of fog formation. The increasing Aitkin and accumulation mode nuclei number concentration of aerosol is the characterization of aerosol accumulation stage. In the mixing stage of haze and fog, the enhancement in the activation process of CCN and increase in the growth of small size aerosol to larger ones is the result of Brownian coagulation process, which is aggravated by higher aerosol number concentration and latent heat production from condensation process of fog droplet. This stage results in hysterical development of

the fog. The increased heating due to solar radiation or passing cold frontal system results in the decrease and dissipation of the fog (Guo et al. 2015).

5.5.2 Haze to Fog Transition

The transformation of haze aerosols into fog droplets takes place in the presence of some particular condition. The size of haze aerosol plays a crucial role in the transition of these particular aerosols to fog droplets. The formation of fog droplets takes place due to the activation of haze aerosol having size larger than 1 μ m. The haze particle having size between 0.3 and 1.0 µm does not change and remains almost consistent during the entire life cycle of fog. Eldridge (1969) reported the transformation of haze into fog with the presence of visibility in the range of 500–1000 m (Eldridge 1969). The sporadic and unusual transformation occurs in the relationship of aerosol number concentration and visibility was observed by Meyer et al. (1980) during the transition process of haze to fog, as visibility reduced from 2.1 to 1.4 km (Meyer et al. 1980). Generally, the clear, haze and fog days were categorized on the basis of relative humidity (RH) and visibility as done in some studies (Ramachandran et al. 2006). The days were represented as clear days with visibility of 10 km, haze days with visibility between 1 and 2 km and fog days with visibility of 0.2–0.5 km (Ramachandran et al. 2006). In concern with relative humidity, when RH was less than 70%, the days are represented as clear days. In comparison, when the RH was in the range of 80–100% represents hazy and foggy days. The classification of weather into fog, haze and clear days was also done by Elias et al. (2009) during the investigation of haze and fog events in Paris. The researcher done this classification on the basis of attenuation contributions of fog, haze and clear days and observed the transformation of haze into fog with the visibility reduction from 0.88 to 0.4 km (Elias et al. 2009).

5.5.3 Chemical Composition of Fog

The conversion capacity of hygroscopic aerosols into fog droplets with the increase in moisture can be discovered by analysing the chemical composition of the fog (Fahey et al. 2005). Data from the aerosol and fog water chemistry so far indicate a highly polluted environment in which the fog developed. The concentrations of NH_4^+ , SO_4^{2-} , CI^- , Ca^{2+} and NO_3^- were found to be very high, indicating that fog water is highly polluted. The ionic composition was found in the order of $NH_4^+ > SO_4^{2-} > CI^- > Ca^{2+} > NO_3^- > Na^+ > Mg^{2+} > K^+ > F^- > HCO^{3-}$. Among the anions, the concentration of SO_4^{2-} (3763 µeq/L) was maximum, whereas NH_4^+ (4162 µeq/L) showed maximum concentration among the cations (Ghude et al. 2017). The sulphate (SO_4^{2-}), ammonium (NH_4^+), nitrate (NO_3^-) and chloride (CI^-) were the major constituents of fog droplets as also implied by chemical analysis of fog droplets in some studies (Li et al. 2011a). The highest values of chemical species were observed during stable conditions especially during

night hours on foggy days (~11% higher). Secondary inorganic aerosols (sulphate and nitrate) were the dominant ions (38%) in the chemical constituents of the fine particle and were higher during the fog events. The chemical partitioning of fog water samples suggests that NH_4^+ (28%) and SO_4^{2-} (26%) dominate the chemical composition (Ghude et al. 2017). Ammonium and sulphate are in the form of ammonium sulphate ((NH₄)₂SO₄). Even though the acidic ions SO_4^{2-} and NO_3^{-} are very high, fog water is not acidic because of the very high concentration of neutralizing ions NH_4^+ and Ca^{2+} in the samples. The major neutralization was due to NH₄⁺ followed by Ca²⁺. The alkaline nature of fog water was also reported by Lakhani et al. (2007) during chemical probe of fog water at Agra (Lakhani et al. 2007). The ammonium (NH_4^+) was showing higher concentration followed by SO_4^{2-} , NO_3^{-} , Ca^{2+} , Mg^{2+} , Na^+ and K^+ in this study. The SO_4^{2-} was present in higher amount than NO_3^- in fog water as indicated by SO_4^{2-}/NO_3^- equivalent ratio of 1.4 in fog water (Singh and Gupta 2014). The acidic effects of SO_4^{2-} and NO_3^{-} could also be neutralized by Ca²⁺ as there is presence of higher concentration of soiloriented aerosols (Ca²⁺, Mg²⁺, K⁺ and Na⁺) with dominance of Ca²⁺in Delhi as reported by Kapoor et al. (1993) in this particular region which also justified the alkaline nature of the fog (Kapoor et al. 1993).

5.6 Factor for Fog Formation

Fog formation is mainly reliant on complicated interactions amid microphysics, radiation, turbulence, availability of moisture and surface processes (Gultepe et al. 2007). The fog formation was triggered by the physical mechanisms representing different meteorological conditions in a particular area (Chaouch et al. 2017). The pollution in a particular area greatly influenced fog formation and the heterogeneous nucleation of pollution particle led to fog particle formations (Gultepe et al. 2001). The increasing concentration of atmospheric aerosol (condensation nuclei) and enhanced condensation of water vapour on higher available condensation nuclei result in the formation of long-lasting and denser fog (Ram et al. 2012).

5.6.1 Meteorological Condition

Fog occurrence is correlated with the relative humidity, temperature, wind velocity, cloud cover and weather thermal stability. The dew point occurs when air temperature is saturated with water vapour. This takes place when air or the water drops clinging thereto takes in more humidity (Nolan 2010). When air temperature nears the dew point, a little humidity condenses on the particles suspended in the air, such as dust and snow, and turns into liquid state matter that results in the fog. Therefore, air saturation with vapour and the condensation of minute water particles are two major conditions for fog formation (Nolan 2010). The scenarios that most commonly come to mind when considering fog are associated with calm clear nights over land or fog observed at sea over cold waters. But the fact is that fog presents itself under

various scenarios, including some associated with large-scale mid-latitude cyclonic disturbances. The various scenarios are often described through a classification according to fog types. Due to the numerous factors involved, it is not surprising that fog forms under a wide range of scenarios. The fog formation and subsequent evolution is driven by a plethora of mechanisms and interactions that span several time and space scales.

5.6.2 Role of Aerosol

Mensbrugghe (1892) acknowledged the role of aerosols in the formation of fog and declared that the solid particles present in the air result in aqueous vapour condensation (Mensbrugghe 1892). The turbulent mixing, growth of droplet due to radiative cooling, differential gravitational settling of different size drops and activation growth of droplets are some of the crucial exercises that shape the microstructure of fog. There is the presence of sophisticated relationship amid characteristics of fog and aerosols, as the aerosol drives the activation growth of droplets. The area with higher aerosol concentration results in larger fog as compared to other (Knott 1923). The dependence of microstructure of fog on condensation nuclei was entrenched by many researchers in the past by studying the relationship between aerosol characteristics and droplet nucleation under different meteorological conditions (Eldridge 1971; Neiburger and Wurtele 1949).

The formation of fog droplet is the result of water vapour condensation on the accumulation mode particles which are acting as cloud condensation nuclei (CCN) during the foggy conditions (Seinfeld and Pandis 1998). The mass of accumulation mode particles increased due to aqueous reactions in the fog droplet (Kaul et al. 2011). The presence of meteorological conditions like lower temperature and higher relative humidity and along with higher concentration of aerosol which particularly acts as CCN in a particular area results in the appearance of fog. The saturation of water vapour during the generation and fog evolution mainly occurs in the presence of higher relative humidity (RH), sometimes at 100%. When the condensation of water vapour takes place on the available surface area of aerosol, dew point reached and growing of droplet results in perpetual fog (Singh et al. 2011). The formation of secondary organic aerosol (SOA) and other inorganic species also increased in presence of fog which further provide more surface for condensation (Kaul et al. 2011; Singh et al. 2011). Some researchers in India over the IGP investigated the role of local meteorological conditions and primary and secondary aerosol in the modulation of fog and haze formation in the wintertime. Where, low ambient temperatures, weak winds, and stable boundary layer height result in pollutant accumulation (Nair et al. 2007; Ram et al. 2010). The researcher worldwide has also observed the fog formation as a result of higher aerosol concentration emitted from human activities along with secondary aerosols formations via gas-to-particle conversion under favourable meteorological conditions (Fang et al. 2011; Husain et al. 2004; Reilly et al. 2001; Sun et al. 2006; Zhang et al. 2010; Sonwani and Saxena 2021).

5.7 Fog as an Extreme Event: Causes and Impacts

5.7.1 Fog as an Extreme Event

Today, there are many tools for identifying and predicting severe and high-impact meteorological events. However, short-lived weather phenomena like fog does not lend themselves as easily to identification and prediction. Fog creates low visibility conditions, which increased a greater crash risk. According to the Federal Highway Administration (FHWA), every year, over 38,700 vehicle crashes occur in fog, killing more than 600 people and injuring more than 16,300 people annually. And vehicles are not the only mode of transportation affected by fog; there can be significant impact to aviation and marine operations as well. On all, fog is an extreme event which causes more deaths (from traffic accidents) than heat, tornadoes, floods and lightning.

5.7.2 Causes

The changes in meteorological conditions and concentration of aerosol are correlated with the formation and intensification of the fog. The advection and radiation effects in combination effects the strength and intensity of extreme fog event. The maintenance and fog deterioration are mainly the result of alternative temperature advection. The strengthening and weakening of the extreme fog event were also affected by boundary layer height. The dynamic factors like weak vertical wind shear persistence in middle-lower layer and weak vertical motion were favourable for the formation of extreme fog event. The multiple factors like atmospheric stratification, the atmospheric circulation background, vertical wind shear structure, temperature advection, aerosol pollution characteristics and the meteorological conditions play a crucial role in the conversion and enhancement of fog events. The crucial condition for the formation of fog is the inversion stratification. The advection and radiation cooling were liable for the strengthening and intensification of dense fog event in the area of wetter and thicker ground layer where the middle high layer with colder and drier characteristics results in more formation and strengthening of fog. The worst air quality in urban area further increase the occurrence of fog weather and impacts human being in many ways.

5.7.3 Impacts

The high occurrence of fog has numerous and diverse impacts on human activities, ecosystems and regional climate which caused deep concern in the scientific community recently. The fog event has wider impact on environment, as it plays an important role in the degradation of air quality by increasing pollutant concentration which further reduced the visibility. The people's daily life is badly affected due to prolonged fog episodes and the reduced visibility further affected the road, marine and air transportation. The number of roads, rails and air accidents also increased due to poor visibility in the presence of fog events (Gultepe et al. 2007). The low visibility results in delay, diversion and cancellation of flight which lead to huge economic losses. Studies also provide evidence that the occurrence of fog in polluted environments leads to adverse health effects in the human population (Villeneuve et al. 2005). Among these, low visibility and degradation of air quality is of main concern (Zhang et al. 2010) as they are causes of all the fatalities occurring related to fog.

5.7.4 Effect on Air Quality

The occurrence of fog events in a particular region has great effect on the air quality of that region. The increasing concentration of fine particulates ($PM_{2.5}$) during the events of fog results in many environmental problems by reducing visibility, affecting human health and climate (Bi et al. 2014; Zhang et al. 2015). The concentration of secondary aerosol increased due to increase in aqueous phase oxidation reaction in presence of fog, which also has substantial impact on visibility, atmospheric chemistry and air quality in these fog events. The fog episode in a region represents the reduction of visibility below 2 km, lowering of PBL below 1 km, reduction in wind speeds below 1 m/s and increase in concentration of different pollutants in that region (Guo et al. 2015).

5.7.4.1 Effect on Pollutant Concentration

The increase in mass concentration of particulates matter (PM) has been reported by Tiwari et al. (2011) due to the event of fog (Tiwari et al. 2011). The increase in total suspended particulates concentration during the foggy period was also reported at Hisar because of higher relative humidity (RH) in the combination of fog. The increase in particulate mass loading and reduction in visibility was observed by Shen et al. (2015) in Yangtze River Delta, China during the fog event of January 2013 (Shen et al. 2015). The foggy days in Kanpur also observed increase in mass loading of PM_1 as reported by Gupta and Mandariya (2013) (Gupta and Mandariya 2013). The increase of hygroscopic growth of increased aerosol particles in the presence of higher relative humidity results in enhancement in the light scattering efficiency during the foggy periods as observed by Malm and Day (2001) (Malm and Day 2001). The study in Kanpur, India (Kaul et al. 2011) and in London (Dall'Osto et al. 2009), both recorded increase in production of secondary organic aerosol during fog events. The fog processing results in increase in ratio of organic carbon to elemental carbon (OC/EC) and water-soluble organic carbon to organic carbon (WSOC/OC) as observed by Rajput et al. (2018) during the winter season and also observed increase in organic matter (OM) from non-foggy to foggy periods (Rajput et al. 2018).

During the severe foggy days, all the chemical components of particulates matter specifically SO_4^{2-} , NH_4^+ , BC and NO_3^- represent increase in their concentration. The mass concentration of all ions was observed higher at day and night-time both

during the time of fog. Ghude et al. (2017) reported aqueous-gas phase partitioning of ammonia due to increased ammonium ions concentration in fog water which is recorded as 4162 μ M in the fog water. The gas to particle conversion increased due to the presence of aqueous phase, i.e. clouds or fogs in the atmosphere which acts as a physicochemical processor for gases and particles which result in higher concentration of ammonium, sulphate and nitrate during fog events (Kaul et al. 2011; Gupta and Mandariya 2013). Several chemical reactions take place at the medium provided by the aqueous phase during the fog processing.

5.7.4.2 Visibility

The level of atmospheric quality can be indicated by visibility, as it is an eminent index for assessing the air quality of urban environment. The increasing air pollution leads to increase in optical extinction coefficient of gas and particulates which result in poor visibility (Chang et al. 2009; Deng et al. 2011). The studies on air pollution and climatology included visibility for representing the air quality (Chen and Wang 2015; Shi et al. 2014). The atmospheric aerosol results in low visibility with the process of light extinction, as the increased concentration of particulates in atmosphere degrades visibility through scattering light (Deng et al. 2011). The reduction in visibility was mainly occurred due to the presence of fine particulates (PM_{2.5}) in the atmosphere as reported by several studies (Yang et al. 2012; Zhang et al. 2015). The smaller size $(0.1-1.0 \ \mu\text{m})$ particulate matter (PM) has higher extinction efficiency as compared to larger ones as observed by Shi et al. (2014).

5.7.5 Effect on Transport System

The lower visibility as a result of extreme fog event exemplifies an imperilment to road, marine and air transportation. Fogs effectively hinder sea navigation, land and air transport systems. Dense fogs cause severe accidents on high-speed highways involving collision of trucks, buses, cars, etc. Dense fogs are navigational hazards in the seas as ships and huge super tankers carrying oil sometimes collide resulting in spilling of huge quantity of oil causing enormous oil slicks on seawater which cause ecological disaster. Vehicular traffic comes to grinding halt in the event of dense fogs. Landing and take-off of aircrafts are delayed by dense fogs causing economic loss to airlines and inconvenience to stranded passengers (Kulkarni et al. 2019).

5.7.5.1 Traffic Accidents

Fog is formed when water vapour is condensed on the surface of the atmosphere in the form of small drops; the diameter of each is no more than 100 microns. Because of the smallness of these drops, they remain suspended in the air, the thing that limits visibility hence causing traffic accidents. In fog days, there are more incidents of traffic accidents which aggravate the economic situation since they necessitate exceptional measures and continuous lighting of streets, to say nothing of the cost of repairs due to damages incurred. Visibility, on fog formation, is correlated with the rate of water vapour in the air, the higher the relative humidity, the less the visibility. The major meteorological criteria directly affecting the users of roads and their safety are: snow, rains, ice, winds and fog. As compared with other phenomena, fog is relatively rare. Fog occurrence is not easy to predict since fog varies in density and occurrence location. It moves horizontally and vertically from a place to another, covering small and large spaces, and it can clear up quickly. Fog accidents have recently become a topic of interest in different parts of the world where this problem featured in the areas witnessing cases of dense fogs. The fog blocks visibility, which leads to numerous traffic accidents since the suspended water drops work to disperse the light, which sometimes causes visibility to be as low as a few metres, thus arresting traffic and causing accidents. Therefore, this fog phenomenon has become the concern of traffic and transport system planners in most of the world.

With the proliferation of motorized vehicles in the early part of the twentieth century, occurrences of fog began to be reported as a cause of road and air accidents (Case 1916; McAdie 1929). Ingenious minds have tried to minimize the risk of accidents by devising methods of detecting nearby objects in poor visibility (Bell 1885). In spite of these efforts, the most infamous maritime disaster occurred when the Titanic collided with an iceberg shrouded in fog. Nowadays, continued technological advances have reduced but not eliminated the risk of accidents in adverse weather conditions. Accidents in which the presence of fog played a role still occur. Technical problems, human error and poor visibility in fog contributed to the worst aviation accident in history in which two Boeing 747 s collided on the runway at the Los Rodeos Airport on the island of Tenerife, Canary Islands in March of 1977. More recent accidents have also occurred (Pike 1998), underlining the persisting danger linked with the presence of fog in spite of our technological advances. The presence of fog also creates more perilous driving circumstances on the roads by reducing the visibility of road signs (Munehiro et al. 2005) and reducing the ability of drivers to react in a timely manner to changing driving conditions as manifested with the large number of multi-vehicle accidents which occurred under lower visibility conditions (Whiffen et al. 2004). Fatalities are often an unfortunate outcome of accidents occurring under foggy conditions, as evidenced by 6000 fatalities reported over a 9-year period as a result of accidents on foggy highways in the United States (Shepard 1996).

5.7.6 Economic Impact

The occurrence of dense fog affected the flight operations by increasing delay, diversion and cancellation of flights (Fig. 5.1) which result in huge financial losses to the aviation industry. The take-off and landing of airlines are not allowed with visibility lower than 500 m under visual flight rules (VFR), as according to International Civil Aviation Organization (ICAO) guidelines. At the time of occurrence of extreme fog event, the slowing down of air traffic, increase in runway occupancy time (ROT) and increase in spacing amid landing and take-off of each aircraft result in the reduction of airport capacity. Due to safety concerns, fog occurrences continue to be associated with reduced airport capacity and as a consequence important



Fig. 5.1 Schematic showing the possible effects of fog on flight operations. (Source: Modified from Kulkarni et al. 2019)

disruption in air traffic and associated cost. The dense fog can cause a huge loss of approximately US\$200,000 to an airline in a single day at a single airport as reported in one study (Robinson 1989). The extreme dense event also caused huge disruption to flight operation at UK airport in December 2007 as discussed in another study (Milmo 2007). Gadher and Baird (2006) reported the loss of £25 million to British Airways alone due to dense fog (Gadher and Baird 2006). The total estimated economic cost of 3.9 million USD as a result of the occurrence of extreme fog event at IGI Airport through 5 years period from 2011 to 2016 was also reported by Kulkarni et al. (2019).

5.7.7 Health Effect

The composition of fog particles is different in clean air condition as compared to polluted air conditions. It is purely made of water drops or ice crystals in clean air, while the coexistence of fog and haze particles in polluted environment can be considerably pernicious to health of human being and ecological environment besides transportation (Guo et al. 2015). When fogs are polluted through sulphur coming out of the chimneys of the mills, they become poisonous and health hazards. For example, killer poisonous smog was formed due to mixing of dense fog with smokes and sulphide fumes coming out from the chimneys in the Donora Valley of Pennsylvania (USA) on 26 October 1948. This toxic fog (smog) caused 20 human

deaths and 43% inhabitants fell ill due to cough and respiratory problem. Similar poisonous fog was produced due to mixing of sulphur dioxide coming out of the factories of zinc smelters and sulphuric acid in the Meuse Valley of Belgium in the month of December 1930. This killer fog claimed 63 human lives due to obstruction in respiratory system. The poisonous fog of December 1952 claimed 4000 human lives in London which make this extreme fog event as of worldwide concern.

5.7.7.1 Fog and Chest Infection

Fogs pose a threat to people's health as a consequence of the diseases resultant from change of weather conditions, drop of temperatures, and fogs that carry viruses and bacteria. Due to high humidity, there are more and more cases of infections related to respiratory system as well as allergic complaints. This is but common in the light of weather changes particularly in winter where fog may lead to inflammation of the upper respiratory system resulting in breathing problems as well as causing of fatigue, laboured breathing and buzz in the ears. In the event of infection getting worse, things may go even further and develop into low blood pressure, imperfect sight and hearing, sense of congestion and irritation of mucus membranes, sore throat and inflammation of the trachea (Al-Mutairi 2017).

The researcher argues that the spread of fog phenomenon in Northern Borders area adversely affects human health, especially those suffering from asthma, because relative humidity in the air gets higher than 40%, this being accompanied by an acute drop in temperatures, which aggravates asthmatics health condition. It further leads to the formation of mould and rot inside houses, which worsens the health condition of asthmatics (Al-Mutairi 2017).

5.7.8 Wireless Communication

A more recent concern, albeit from a less hazardous perspective, has emerged and is related to the effect fog has on the reliability of wireless communications (Fischer et al. 2004). The presence of fog in itself and its microphysical characteristics affect the capacity, availability and reliability of free space optics and laser telecommunication systems (Harris et al. 2006; Kedar and Arnon 2003).

5.7.9 Impact on Vegetation

Another important consideration is the influx of pollutants to ecosystems by the deposition of fog water on surface elements. The deposition of fog water by gravitational settling of fog droplets or downward turbulent fluxes of fog water can contribute a significant amount of pollutants to vegetation (Lange et al. 2003), possibly leading to an adverse impact on its health (Jagels et al. 2002).

5.8 Conclusion

Data from the aerosol and fog water chemistry so far indicate a highly polluted environment in which the fog developed. Secondary inorganic aerosols (sulphate and nitrate) were the dominant ions (38%) in the chemical constituents of fine particles and were higher during the fog events. The role of aerosols as CCN and fog droplet formation needs to be understood for accurate representation of fog droplet activation and growth. The systematic studies with detailed observations for several fog episodes are necessary to make progress in forecasting these events, their duration and dissipation, etc. The fog formation act as an extreme event which results in delaying and cancelling of flights, also disrupts train schedules lead to huge economic loss and also effects human health in the northern part of India and other parts of the world. The different studies worldwide find that fog phenomenon extremities occasioned several traffic problems, owing to blocked visibility, such as car accidents and traffic jams. Aviation forecasters in northern India as well as worldwide require exact information about dense fog formation for the safe operation of flights. The increased forecasting ability of fog in the affected region is of central interest to the agencies responsible for public safety and the transportation industry as accurate and timely forecasts would provide a basis for increased safety, as well as leading to economic benefits through better management of disruptions in air and surface transportation activities.

The improved forecasting of fog can be obtained through an increased discerning of different physical mechanisms concerned with the formation and subsequent evolution of extreme fog event. The understanding of different parameters like kind of near-surface thermodynamical conditions which lead to the initiation of a fog event, the high aerosol number concentration which is prerequisite for fog formation, different physical processes which are responsible for extended-period fog extreme event, microphysical aspects which are decisive for the fog lifecycle and boundary layer dynamics which control fog microphysics is necessary. These key parameters are needed for physical parameterization of fog extreme event and to improve fog prediction.

Overall, the detailed observations undertaken will be used to better understand the fog and improve its prediction, by tuning the physical parameterizations and also possibly finding the chains of associations for fog droplet formation. The ultimate goal of this study is thus to identify and describe the physical processes and interactions that influence the occurrence of fog events in the different complex regions of the world.

References

Al-Mutairi MK (2017) Fog: its causes, types, and dangers in Saudi Northern Borders (a climatic study). J Geogr 5:64–76

- Badarinath K, Latha KM, Chand TK, Reddy R, Gopal KR, Reddy LSS, Narasimhulu K, Kumar KR (2007) Black carbon aerosols and gaseous pollutants in an urban area in North India during a fog period. Atmos Res 85:209–216
- Badarinath K, Kharol SK, Sharma AR, Roy P (2009) Fog over Indo-Gangetic plains—a study using multisatellite data and ground observations. IEEE J Select Top Appl Earth Observ Remote Sens 2:185–195
- Baldocchi D, Waller E (2014) Winter fog is decreasing in the fruit growing region of the Central Valley of California. Geophys Res Lett 41:3251–3256
- Bell AG (1885) Preventing collisions with icebergs in a fog. Science 5(122):460-461
- Bergot T, Guedalia D (1994) Numerical forecasting of radiation fog. Part I: numerical model and sensitivity tests. Mon Weather Rev 122:1218–1230
- Bi J, Huang J, Hu Z, Holben B, Guo Z (2014) Investigating the aerosol optical and radiative characteristics of heavy haze episodes in Beijing during January of 2013. J Geophys Res Atmos 119:9884–9900
- Canada CE, Toth G (2010) The environment Canada handbook on fog and fog forecasting. Environment Canada
- Case ME (1916) Highway accidents in New York City during 1915. Q Publ Am Stat Assoc 15:318–323
- Chan CK, Yao X (2008) Air pollution in mega cities in China. Atmos Environ 42:1-42
- Chandra B, Sinha V, Hakkim H, Kumar A, Pawar H, Mishra A, Sharma G, Garg S, Ghude SD, Chate D (2018) Odd—even traffic rule implementation during winter 2016 in Delhi did not reduce traffic emissions of VOCs, carbon dioxide, methane and carbon monoxide. Curr Sci 00113891:114
- Chang D, Song Y, Liu B (2009) Visibility trends in six megacities in China 1973–2007. Atmos Res 94:161–167
- Chaouch N, Temimi M, Weston M, Ghedira H (2017) Sensitivity of the meteorological model WRF-ARW to planetary boundary layer schemes during fog conditions in a coastal arid region. Atmos Res 187:106–127
- Chen H, Wang H (2015) Haze days in North China and the associated atmospheric circulations based on daily visibility data from 1960 to 2012. J Geophys Res Atmos 120:5895–5909
- Croft PJ, Burton AN (2006) Fog during the 2004–2005 winter season in the northern mid-Atlantic states: spatial characteristics and behaviors as a function of synoptic weather types. In: Proc. 12th Conf. on Aviation, Range and Aerospace Meteorology
- Dall'Osto M, Harrison R, Coe H, Williams P (2009) Real-time secondary aerosol formation during a fog event in London. Atmos Chem Phys 9:2459–2469
- Deng J, Wang T, Jiang Z, Xie M, Zhang R, Huang X, Zhu J (2011) Characterization of visibility and its affecting factors over Nanjing, China. Atmos Res 101:681–691
- Dimri A, Niyogi D, Barros A, Ridley J, Mohanty U, Yasunari T, Sikka D (2015) Western disturbances: a review. Rev Geophys 53:225–246
- Duanyang L, Meijuan P, Jun Y, Guozheng Z, Wenlian Y, Zihua L (2010) Microphysical structure and evolution of a four-day persistent fog event in Nanjing in December 2006. J Meteorol Res 24:104–115
- Duynkerke PG (1991) Radiation fog: a comparison of model simulation with detailed observations. Mon Weather Rev 119:324–341
- Duynkerke PG (1999) Turbulence, radiation and fog in Dutch stable boundary layers. Bound-Layer Meteorol 90:447–477
- Eldridge RG (1969) Mist-the transition from haze to fog. Bull Am Meteorol Soc 50:422-427
- Eldridge RG (1971) The relationship between visibility and liquid water content in fog. J Atmos Sci 28:1183–1186
- Elias T, Haeffelin M, Drobinski P, Gomes L, Rangognio J, Bergot T, Chazette P, Raut J-C, Colomb M (2009) Particulate contribution to extinction of visible radiation: pollution, haze, and fog. Atmos Res 92:443–454

- Fahey K, Pandis S, Collett J Jr, Herckes P (2005) The influence of size-dependent droplet composition on pollutant processing by fogs. Atmos Environ 39:4561–4574
- Fang G-C, Lin S-C, Chang S-Y, Lin C-Y, Chou C-C, Wu Y-J, Chen Y-C, Chen W-T, Wu T-L (2011) Characteristics of major secondary ions in typical polluted atmospheric aerosols during autumn in central Taiwan. J Environ Manag 92:1520–1527
- Fischer KW, Witiw MR, Baars JA, Oke TR (2004) Atmospheric laser communication: new challenges for applied meteorology. Bull Am Meteorol Soc 85:725–732
- Friedlein MT (2004) Dense fog climatology: Chicago O'Hare International Airport July 1996– April 2002. Bull Am Meteorol Soc 85:515–517
- Gadher D, Baird T (2006) Cited 2007: airport dash as the fog lifts. The Sunday Times
- Gautam R, Singh MK (2018) Urban heat island over Delhi punches holes in widespread fog in the Indo-Gangetic Plains. Geophys Res Lett 45:1114–1121
- Gautam R, Hsu NC, Kafatos M, Tsay SC (2007) Influences of winter haze on fog/low cloud over the Indo-Gangetic plains. J Geophys Res Atmos 112
- Ghude SD, Bhat G, Prabhakaran T, Jenamani R, Chate D, Safai P, Karipot A, Konwar M, Pithani P, Sinha V (2017) Winter fog experiment over the Indo-Gangetic plains of India. Curr Sci 00113891:112
- Giulianelli L, Gilardoni S, Tarozzi L, Rinaldi M, Decesari S, Carbone C, Facchini M, Fuzzi S (2014) Fog occurrence and chemical composition in the Po valley over the last twenty years. Atmos Environ 98:394–401
- Goyal P, Budhiraja S, Kumar A (2014) Impact of air pollutants on atmospheric visibility in Delhi. Int J Geol Agric Environ Sci 2
- Gueye S (2014) Frequency, timing and temporal patterns of regional coastal Arctic fog in East Greenland. MSc Research Thesis, University of Amsterdam (Netherlands)
- Gultepe I, Isaac G, Cober S (2001) Ice crystal number concentration versus temperature for climate studies. Int J Climatol 21:1281–1302
- Gultepe I, Tardif R, Michaelides S, Cermak J, Bott A, Bendix J, Müller MD, Pagowski M, Hansen B, Ellrod G (2007) Fog research: a review of past achievements and future perspectives. Pure Appl Geophys 164:1121–1159
- Guo L, Guo X, Fang C, Zhu S (2015) Observation analysis on characteristics of formation, evolution and transition of a long-lasting severe fog and haze episode in North China. Sci China Earth Sci 58:329–344
- Gupta T, Mandariya A (2013) Sources of submicron aerosol during fog-dominated wintertime at Kanpur. Environ Sci Pollut Res 20:5615–5629
- Hameed S, Mirza MI, Ghauri B, Siddiqui Z, Javed R, Khan A, Rattigan O, Qureshi S, Husain L (2000) On the widespread winter fog in northeastern Pakistan and India. Geophys Res Lett 27: 1891–1894
- Harris A, Sluss JJ, Refai HH, LoPresti PG (2006) Free-space optical wavelength diversity scheme for fog mitigation in a ground-to-unmanned-aerial-vehicle communications link. Opt Eng 45: 086001
- Herckes P, Marcotte A, Wang Y, Collett J Jr (2015) Fog composition in the Central Valley of California over three decades. Atmos Res 151:20–30
- Holets S, Swanson RN (1981) High-inversion fog episodes in central California. J Appl Meteorol 20:890–899
- Holton JR, Curry JA, Pyle JA (2003) Encyclopedia of atmospheric sciences. Academic
- Husain L, Ghauri B, Yang K, Khan AR, Rattigan O (2004) Application of the SO42–/Se tracer technique to study SO2 oxidation in cloud and fog on a time scale of minutes. Chemosphere 54: 177–183
- Jagels R, Jiang M, Marden S, Carlisle J (2002) Red spruce canopy response to acid fog exposure. Atmos Res 64:169–178
- Jaswal AK, Kumar N, Prasad AK, Kafatos M (2013) Decline in horizontal surface visibility over India (1961–2008) and its association with meteorological variables. Nat Hazards 68:929–954

- Jenamani RK (2007) Alarming rise in fog and pollution causing a fall in maximum temperature over Delhi. Curr Sci 93(3):314–322
- Jenamani RK, Kalsi A (2012) Micro-climatic study and trend analysis of fog characteristics at IGI airport New Delhi using hourly data (1981–2005). Mausam 63:203–218
- Jia X, Guo X-L (2012) Impacts of anthropogenic atmospheric pollutant on formation and development of a winter heavy fog event. Chin J Atmos Sci 36:995–1008
- Kapoor R, Tiwari S, Ali K, Singh G (1993) Chemical analysis of fogwater at Delhi, North India. Atmos Environ Part A Gen Top 27:2453–2455
- Kaul D, Gupta T, Tripathi S, Tare V, Collett J Jr (2011) Secondary organic aerosol: a comparison between foggy and nonfoggy days. Environ Sci Technol 45:7307–7313
- Kedar D, Arnon S (2003) Optical wireless communication through fog in the presence of pointing errors. Appl Opt 42:4946–4954
- Knott CG (1923) Collected scientific papers of John Aitken, LL. D., FRS. University Press
- Kulkarni R, Jenamani RK, Pithani P, Konwar M, Nigam N, Ghude SD (2019) Loss to aviation economy due to winter fog in New Delhi during the winter of 2011–2016. Atmosphere 10:198
- Lakhani A, Parmar RS, Satsangi GS, Prakash S (2007) Chemistry of fogs at Agra, India: influence of soil particulates and atmospheric gases. Environ Monit Assess 133:435–445
- Lange CA, Matschullat J, Zimmermann F, Sterzik G, Wienhaus O (2003) Fog frequency and chemical composition of fog water—a relevant contribution to atmospheric deposition in the eastern Erzgebirge, Germany. Atmos Environ 37:3731–3739
- Lewis DM (2004) Forecasting advective sea fog with the use of classification and regression tree analyses for Kunsan Air Base. In: AIR FORCE INST OF TECH WRIGHT-PATTERSON AFB OH SCHOOL OF ENGINEERING AND ...
- Lewis J, Koracin D, Rabin R, Businger J (2003) Sea fog off the California coast: viewed in the context of transient weather systems. J Geophys Res Atmos 108
- Li Z, Liu D, Feng Y (2011a) Recent progress in the studies of the fog-water chemical characteristics in China. Acta Meteor Sin 69:544–554
- Li Z, Liu D, Yang J (2011b) The microphysical processes and macroscopic conditions of the radiation fog droplet spectrum broadening. Chin J Atmos Sci (In Chinese) 35
- Lutgens F, Tarbuck E (2004) The atmosphere: an introduction to meteorology. Pearson Education, Inc., New Jersey, 508 p
- Malm WC, Day DE (2001) Estimates of aerosol species scattering characteristics as a function of relative humidity. Atmos Environ 35:2845–2860
- McAdie A (1929) Weather hazards in aviation. Sci Mon 29:66-71
- Mensbrugghe V (1892) The formation of fog and of clouds, translated from Ciel et Terre. Symons's Mon Meteor Mag 27:40–41
- Meyer MB, Jiusto JE, Lala GG (1980) Measurements of visual range and radiation-fog (haze) microphysics. J Atmos Sci 37:622–629
- Milmo D (2007) BAA counts the cost of December fog. Guardian Unlimited
- Munehiro K, Tokunaga RA, Asano M, Hagiwara T (2005) Effect of time and foggy conditions on subjective visibility: evaluation of retroreflective traffic control devices. Transp Res Rec 1911: 85–104
- Nair VS, Moorthy KK, Alappattu DP, Kunhikrishnan P, George S, Nair PR, Babu SS, Abish B, Satheesh S, Tripathi SN (2007) Wintertime aerosol characteristics over the Indo-Gangetic Plain (IGP): impacts of local boundary layer processes and long-range transport. J Geophys Res Atmos 112
- Nolan PJ (2010) Meteorology the physics of the atmosphere, An introduction to the weather. Farmingdale State College State University of New York
- Neiburger M, Wurtele M (1949) On the nature and size of particles in haze, fog, and stratus of the Los Angeles region. Chem Rev 44:321–335
- Niu S, Lu C, Liu Y, Zhao L, Lü J, Yang J (2010) Analysis of the microphysical structure of heavy fog using a droplet spectrometer: a case study. Adv Atmos Sci 27:1259–1275

- Okland H, Gotaas Y (1995) Modelling and prediction of steam fog. Beitrage zur Physik der Atmosphare-Contributions to Atmospheric Physics 68:121–132
- Petäjä T, Järvi L, Kerminen V-M, Ding A, Sun J, Nie W, Kujansuu J, Virkkula A, Yang X, Fu C (2016) Enhanced air pollution via aerosol-boundary layer feedback in China. Sci Rep 6:18998
- Pike WS (1998) The accident to a Boeing 737, 7T-VEE, on approach to Coventry Airport on 21 December 1994. Meteorol Appl 5:25–31
- Pinnick RG, Hoihjelle D, Fernandez G, Stenmark E, Lindberg J, Hoidale G, Jennings S (1978) Vertical structure in atmospheric fog and haze and its effects on visible and infrared extinction. J Atmos Sci 35:2020–2032
- Pu M, Zhang G, Yan W, Li Z (2008) Features of a rare advection-radiation fog event. Sci China Ser D Earth Sci 51:1044–1052
- Quan J, Zhang Q, He H, Liu J, Huang M, Jin H (2011) Analysis of the formation of fog and haze in North China Plain (NCP). Atmos Chem Phys Discuss 11
- Rajput P, Singh DK, Singh AK, Gupta T (2018) Chemical composition and source-apportionment of sub-micron particles during wintertime over Northern India: new insights on influence of fog-processing. Environ Pollut 233:81–91
- Ram K, Sarin M, Tripathi S (2010) A 1 year record of carbonaceous aerosols from an urban site in the Indo-Gangetic Plain: characterization, sources, and temporal variability. J Geophys Res Atmos 115
- Ram K, Sarin M, Sudheer A, Rengarajan R (2012) Carbonaceous and secondary inorganic aerosols during wintertime fog and haze over urban sites in the Indo-Gangetic Plain. Aerosol Air Qual Res 12:359–370
- Ramachandran S, Rengarajan R, Jayaraman A, Sarin M, Das SK (2006) Aerosol radiative forcing during clear, hazy, and foggy conditions over a continental polluted location in north India. J Geophys Res Atmos 111
- Reilly JE, Rattigan OV, Moore KF, Judd C, Sherman DE, Dutkiewicz VA, Kreidenweis SM, Husain L, Collett JL Jr (2001) Drop size-dependent S (IV) oxidation in chemically heterogeneous radiation fogs. Atmos Environ 35:5717–5728
- Renhe Z, Li Q, Zhang R (2014) Meteorological conditions for the persistent severe fog and haze event over eastern China in January 2013. Sci China Earth Sci 57:26–35
- Roach W (1995) Back to basics: fog: part 2—the formation and dissipation of land fog. Weather 50: 7–11
- Robinson PJ (1989) The influence of weather on flight operations at the Atlanta Hartsfield International Airport. Weather Forecast 4:461–468

Ryznar E (1977) Advection-radiation fog near Lake Michigan. Atmos Environ 1967(11):427-430

- Safai PD, Ghude S, Pithani P, Varpe S, Kulkarni R, Todekar K, Tiwari S, Chate DM, Prabhakaran T, Jenamani RK (2019) Two-way relationship between aerosols and fog: a case study at IGI airport, New Delhi. Aerosol Air Qual Res 19:71–79
- Saxena P, Kulshrestha UC (2015) Scavenging of urban air emissions by Fog at Delhi, India. In AGU Fall Meeting Abstracts, vol. 2015, pp A33H-0274, December
- Saxena P, Kulshrestha UC (2016) Contribution of carbonaceous species in SOA formation during fog and non-fog period. J Chem Bio Phys Sci 7:31–48
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8):6639– 6656
- Seinfeld J, Pandis S (1998) Atmospheric chemistry and physics. Wiley, New York
- Shen X, Sun J, Zhang X, Zhang Y, Zhang L, Che H, Ma Q, Yu X, Yue Y, Zhang Y (2015) Characterization of submicron aerosols and effect on visibility during a severe haze-fog episode in Yangtze River Delta, China. Atmos Environ 120:307–316
- Shepard FD (1996) Reduced visibility due to fog on the highway. Transportation Research Board
- Shi Y, Chen J, Hu D, Wang L, Yang X, Wang X (2014) Airborne submicron particulate (PM1) pollution in Shanghai, China: chemical variability, formation/dissociation of associated semivolatile components and the impacts on visibility. Sci Total Environ 473:199–206

- Singh DK, Gupta T (2014) Field performance evaluation during fog-dominated wintertime of a newly developed denuder-equipped PM 1 sampler. Environ Sci Pollut Res 21:4551–4564
- Singh VP, Gupta T, Tripathi SN, Jariwala C, Das U (2011) Experimental study of the effects of environmental and fog condensation nuclei parameters on the rate of fog formation and dissipation using a new laboratory scale fog generation facility. Aerosol Air Qual Res 11: 140–154
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Stewart RE, Yiu DT, Chung KK, Hudak DR, Lozowski EP, Oleskiw M, Sheppard BE, Szeto KK (1995) Weather conditions associated with the passage of precipitation type transition regions over eastern Newfoundland. Atmosphere-Ocean 33:25–53
- Sun Y, Zhuang G, Tang A, Wang Y, An Z (2006) Chemical characteristics of PM2. 5 and PM10 in haze—fog episodes in Beijing. Environ Sci Technol 40:3148–3155
- Syed FS, Körnich H, Tjernström M (2012) On the fog variability over south Asia. Clim Dyn 39: 2993–3005
- Tiwari S, Payra S, Mohan M, Verma S, Bisht DS (2011) Visibility degradation during foggy period due to anthropogenic urban aerosol at Delhi, India. Atmos Pollut Res 2:116–120
- Underwood SJ, Ellrod GP, Kuhnert AL (2004) A multiple-case analysis of nocturnal radiation-fog development in the central valley of California utilizing the GOES nighttime fog product. J Appl Meteorol 43:297–311
- Villeneuve PJ, Leech J, Bourque D (2005) Frequency of emergency room visits for childhood asthma in Ottawa, Canada: the role of weather. Int J Biometeorol 50:48–56
- Wang G, Zhang R, Gomez ME, Yang L, Zamora ML, Hu M, Lin Y, Peng J, Guo S, Meng J (2016) Persistent sulfate formation from London Fog to Chinese haze. Proc Natl Acad Sci 113:13630– 13635
- Whiffen B, Delannoy P, Siok S (2004) Fog: impact on road transportation and mitigation options. In: National Highway Visibility Conference, Madison, Wisconsin, USA, pp 1–12
- Willett HC (1928) Fog and haze, their causes, distribution, and forecasting. Mon Weather Rev 56(11):435-468
- Yang J, Niu Z, Shi C, Liu D, Li Z (2010) Microphysics of atmospheric aerosols during winter haze/ fog events in Nanjing. Huan Jing Ke Xue 31:1425–1431
- Yang F, Chen H, Du J, Yang X, Gao S, Chen J, Geng F (2012) Evolution of the mixing state of fine aerosols during haze events in Shanghai. Atmos Res 104:193–201
- Zhang Q, Zhang J, Xue H (2010) The challenge of improving visibility in Beijing. Atmos Chem Phys 10:7821–7827
- Zhang Q, Yan R, Fan J, Yu S, Yang W, Li P, Wang S, Chen B, Liu W, Zhang X (2015) A heavy haze episode in Shanghai in December of 2013: characteristics, origins and implications. Aerosol Air Qual Res 15:1881–1893



Nature of Sand and Dust Storm in South Asian Region: Extremities and Environmental Impacts

Sanjoy Maji and Saurabh Sonwani

Abstract

Sand and dust storms (SDS) are very important atmospheric extreme events that occurred under the influence of turbulent winds with dust particles in any arid region. These are the lower atmospheric meteorological events and their environmental implications are being recognized in recent years due to their growing significance with the climate, human health, and socio-economy. The SDS involves a reduction of visibility to less than 1000 m. The identification of SDS events and their frequency is important to assess their role as climatological and geomorphological agents. The SDS events pose a challenge to the goals of sustainable development. Generally, SDS is a common atmospheric phenomenon in arid, semi-arid, and dry sub-humid areas, though it can travel thousands of kilometers across the countries and oceans, and depends on the wind speed and particle size distribution. Sometimes, SDS accumulate other pollutants on their way and transport them from one place. On the routes of transportation SDS also affects the regional biogeochemical cycle. SDS have a several beneficial and adverse impact on the environment. The primary impacts of SDS events include atmospheric radiation balance, regional precipitation, and hurricane activity. The SDS events are also responsible for elevated levels of fine particulates in the atmosphere that are associated with premature mortality and cardiovascular problems, respiratory problems, lung carcinoma, and severe respiratory tract infections. Such inhaled fine particle is not only composed of fine mineral

S. Maji

Faculty of Engineering and Technology, Jamia Millia Islamia, Delhi, India

S. Sonwani (🖂)

Department of Environmental Studies, Zakir Husain Delhi College, University of Delhi, Delhi, India

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_6

particles but also contains a hazardous mixture of chemicals, spores, microorganisms, fungi, and harmful allergens. Apart from their adverse effects on human health, they also pose several economic impacts such as disturbance of infrastructures, transportation, and supply chain set-ups. SDS events are also responsible for the elimination of the top fertile layer of the soil resulting in the decrease in the mineral content in the soil that ultimately affects the plant health and productivity of the landmass. The United Nations General Assembly (UNGA) adopted resolutions entitled "Combatting sand and dust storms" in 2015 and mentioned that SDS signify a major obstruction to the sustainable development that affect developing nations and their peoples. The objective of the current paper is to understand the nature and phenomena of SDS processes, current and past trends of SDS impacts, as well as to understand the impact of SDS hazards on environment and human health as well as extremities and their correlation with climate change.

Keywords

Sand and dust storm \cdot Distribution \cdot Transportation \cdot Climate \cdot Environment \cdot Human health

6.1 Introduction

The dust storm events, though a common atmospheric phenomenon in arid and semi-arid areas during summer months, have gained great interest in recent times throughout the world due to the environmental implication and geomorphological importance of events of extreme nature. In the last few decades, there are almost a significant rise in the number of scientific contributions on understanding the nature, mechanism, extremities, and environmental impact of dust storm events (Ramanathan et al. 2001; Stout et al. 2009).

SDS event is a consequence of wind erosion releasing participles from the top soil profile. Sand storms happen very near to the ground, but fine particles may be raised up to several kilometers into the atmosphere, where high-speed winds convey them long distances. Bare grounds where vegetation cover is limited are most susceptible to sediment entrainment. Dust storms may travel up to thousands of kilometers from their source location and transport suspended materials over land and ocean, other available surfaces. It was also reported that the dust storm trajectories are part of natural ecological systems and the rate of aeolian deposition may be of same extent to rates of fluvial and aeolian erosion (Goudie and Middleton 2006; Muhs et al. 2014).

However, as of today, there is not a firm delimitation in the explanation of dust storms versus sandstorms. The World Meteorological Organization (WMO 1975) defines dust storms and sandstorms as "An ensemble of particles of dust and sand energetically lifted to great heights by a strong and turbulent wind". As a convention, winds raise a large quantity of dust into the atmosphere and reduce the visibility to less than 1000 m, considered as an SDS event (McTainsh and Pitblado 1987).



Fig. 6.1 Dust potential map of the World (Source: UNEP, WMO, UNCCD 2016)

Silica (SiO₂), iron, magnesium, calcium, aluminum, and sodium are some of the important components of the dust particles that make up SDS. Other chemical species commonly found include traces of arsenic, chromium, zinc, strontium, lead, nickel, cobalt, cadmium, manganese, vanadium, and copper. However, dust particles may also carry organic matter including pathogenic microorganisms like fungi, bacteria, and viruses. The relative composition of different components depends on the source regions as well as transport pathways (Sun et al. 2005).

Worldwide 45 countries are identified as SDS source-areas, whereas more than 151 countries are identified as direct suffering by SDS events (Middleton and Kang 2017). The major source-areas of SDS are in North Africa, the Middle East, and Central Asia (Herman et al. 1997). About 32 million km² of the global land area is susceptible to wind erosion. Stretch extending all the way from the tropical and subtropical deserts of the Sahara through the Middle East to the Great Indian Desert and the mid-latitude deserts of Central Asia, China, and Mongolia is considered as the high-risk belt of SDS. In the Northern Hemisphere, the area extended from the west coast of North Africa, over the Middle East, Central and South Asia, to China is referred to as "dust belt". The Asia-Pacific region is considered as the second largest contributor of dust emissions, accountable for more than half a billion tone of dust to the atmosphere every year. Impacted areas include East, North-East, South, South-West and Central Asia and the Pacific (Fig. 6.1).

6.1.1 Dust Storms: a Global Phenomenon and a Transboundary Hazard

The sites of the main dust sources are: (1) North Africa (Tunisia and Northeast Algeria, Eastern Libyan Desert, Egypt, Sudan, Mauritania, and Western Sahara),

(2) Middle East Asia (Arabian Peninsula, Oman), (3) Central and South Asia (Afghanistan, Iran and Pakistan Basins, Indian subcontinent), (4) North East Asia (Mongolia, China), (5) North America (Mexico), (6) South America (Bolivia, Western Argentina), and (7) Australia (Lake Eyre and the Great Artesian Basin). Dust storm events are considered as a transboundary hazard and their dispersion across the continents depends on wind speed and particle size distribution. SDS can accumulate other pollutants on their way and is the cause of transport of pollutants from one place to another on the routes of transportation impacting the regional biogeochemical cycle. Estimates put figure of about 2 billion tons of dust are released into the air every year (Goudie and Middleton 2006). However, it is mentioned that a maximum (75%) of current emissions originate naturally, not anthropogenically.

Dust is usually created from a small terrestrial area and is spared to cover transboundary regions. The length of SDS can differ from a few hours to several days depending on atmospheric pressure gradients. When SDS originate, initially, they tend to drift near to the earth's surface. However, with the convective currents, fine particles are raised high into the atmosphere and travel thousands of kilometers. The transportation also depends on particles size and wind speed that pass beyond regional-political boundaries and even across continents. The average atmospheric lifetime of dust particles differs from a few hours for particles with changing aerodynamic diameter (particle > 10 μ m), to more than several days (for the ultrafine particles) and the relative polarity of the particle (Van Der Does et al. 2018). Coarser particulates are dumped within short distances from their origin and particles of size below 10 µm are transported to very distant locations. The Saharan dust which is considered as the largest single source of wind-blown sediments is capable of transportation from their sources to Amazonia, North America, Europe, and China. Dust from Central Asia and China reaches Korea, Japan, the Pacific Islands, and North America. Dust from China has been reported as reaching the European Alps, a distance of over 20,000 km (Grousset et al. 2003). It has been observed that Saharan dust reaches up to Amazon rainforest and carries micronutrients which is the source of natural fertilizer (phosphorus) inputs for the rainforest (Nogueira et al. 2021). Likewise, Hawaiian rain forests obtain nutrients from Central Asian dust. Simultaneously, dust from Africa and Asia is believed to damage Caribbean coral reefs (Garrison et al. 2003).

Many countries located outside to the drylands receive dust due to the long-range transport from different sources located inside the dryland, such as Japan receive dust deposits from Chinese and Mongolian deserts (Hong et al. 2010), and Panama which receives Saharan dust (Gross et al. 2016). Nevertheless, some nations experience desert dust while also undergoing wind erosion on their own lands. Northeast Asian deserts are other important sources of dust that frequently carried over the Korean Peninsula and Japan (Kashima et al. 2016), across the Pacific Ocean to the North American continent (McKendry et al. 2001), Europe (Grousset et al. 2003) and on occasion even further (Uno et al. 2009). Antarctica receives deposits of desert dust from Patagonia and Australia (Li et al. 2008) and material from farmland in Ukraine has been traced to Central Europe (Birmili et al. 2008).

SDS involve in various atmospheric processes and are responsible for several positive and negative impacts on the environment (Knippertz 2014). Dust not only affects the atmospheric process but also involves in changing marine, biotic and terrestrial processes (Nickovic et al. 2012). Thus, it disturbs the strengths of tropical storm and cyclone (Evan et al. 2006) and the earth's radiative balance, which can intensify the drought (Han et al. 2008; Highwood and Ryder 2014). It is also a wellknown fact that the fine dust particles act as cloud condensation nuclei (CCN) and participate in precipitation (Nenes et al. 2014). Suspended dust particle causes human health issues (Sonwani and Kulshrestha 2016), like atmospheric fine particles are associated with cardio-pulmonary problems and severe respiratory infections that are responsible for premature death (Sonwani et al. 2021c). Inhaled fine dust particles have been reported with the harmful combination pollutants, spores, bacteria, fungi, and potential allergens with mineral dusts (Kellogg et al. 2004; Goudie (2014)). SDS also increase incidences of road accidents and aviation hazards due to the degradation of atmospheric visibility. Apart from the human health impacts, they also pose adverse impact on economy by disrupting communications, transportation, and supply chain set-ups (Goudie and Middleton 2006). SDS remove the top layer of soils and damage crops, kill livestock, thus increase the cost of food production and threaten sustainable food production (Middleton and Sternberg 2013).

However, SDS have some positive impacts also on the ecosystem. Air blown dust forms loess soils which are considered as one of the most fertile soils for agriculture. Mineral dust deposition provides micro-nutrients to terrestrial and marine ecosystems, boosting primary productivity. SDS also plays a significant role in changing biogeochemical cycles and sustaining both oceans and forests (Goudie 2009).

With the growing focus of bearing of dust storm events on the climate, human health, socio-economy, and the environment, dust storms pose a formidable challenge to the goal of sustainable development, particularly in arid and semi-arid regions. Thus, the impacts of SDS also play an important role in achieving sustainable development at the local, regional, and global levels. Over the past few decades, the social interest to enhance the understanding of the mechanism and nature of SDS processes, and to prevent their undesired impacts on the socio-economic sector is increasing rapidly. With the global collaboration, two resolutions have been approved by the United Nations Global Assembly (A/RES/71/219in 2015 and A/RES/70/195 in 2016) to promote a coordinated approach to combat SDS globally.

This article provides information about the dust cycle and discusses its interaction with weather, the climate system, and terrestrial and marine ecosystems. It also summarizes about the adverse impact of SDS on health and economic sectors. The article also makes an attempt to understand the nature and phenomena of SDS processes in relation to current and past trends of SDS event, as well as understanding the interaction of SDS events with the climate change.

6.1.2 How SDS Act as Extreme Event?

SDS events happen when turbulent winds combine with soil/sand particles in any arid area and transport them along with their movement. Mostly, SDS originated through natural causes that contribute around 75% of total current worldwide emission. Climatic factors affect the dust storm while anthropogenic factors can alter that climatic factors. Human-induced sources, like agricultural activities, deforestation, changes in land use, water diversion contribute rest of 25%. SDS happens as a result of a series of interconnected direct and indirect drivers acting at different levels. The duration of SDS events ranges from a few hours to many days. The intensity of SDS is usually revealed in terms of the particle's concentration and visibility reduction in atmosphere. The common and intense SDS are usually linked with particular synoptical meteorological conditions like: (1) steep atmospheric pressure gradients around subtropical anticyclones, (2) surface cyclones and their associated fronts, (3) monsoonal airflows, (4) local winds associated with strong gradients in relief, and (5) dust devils and convective plumes as a result of daytime turbulence in the planetary boundary layer. Sandstorms arise comparatively adjacent to the soil surface, but finer dust particles may be lifted kilometers high into the atmosphere and transported up to several kilometers through strong winds. Bare land is most susceptible to SDS events and usually happen in semi-arid and arid areas (the area with less or no vegetation cover). Their impacts, however, are frequently felt far beyond those dry lands due to the propensity of dust to be transported over long distances. Thus, due to high-speed violent winds along with sand and dust, SDS events cause significant negative impacts on society, economy, and environment at local, regional, and global scale.

6.2 Sand and Dust Storm Processes

In arid and semi-arid regions, high-speed winds carry huge quantities of dust with them resulting in visibility reduction to below 1000 m. The various sources of dust in the atmosphere may come from natural as well as anthropogenic sources. Natural sources of dust are due to entrainment and re-entrainment of loose dust particles from the ground surface to the atmosphere by the action of high wind speed. Entrainment of dust particles happens when the force exerted by the high wind speed on the dust particle overcomes the retention force. Secondary fine particulates like sulfate and nitrate particles are derived in the atmosphere by the photo-chemical conversion of primary sulfur and nitrogen oxide emissions. As per the WMO (Cuevas Agulló 2013), SDS events are classified on the basis of visibility and segregated into the category of:

- 1. Suspended dust: dust in suspension that is not elevated at or near the station at the time of observation; visibility is less than 10 km.
- 2. Blowing dust: such type of SDS particles that raised at the time of observation, visibility is between 1 and 10 km.

- 3. Dust storm: strong winds push up a huge quantity of dust particles in atmosphere, visibility ranges between 200 and 1000 m.
- 4. Severe SDS: very strong winds push up a huge quantity of SDS particles in atmosphere, visibility usually ranged below 200 m.

SDS event frequently happen under the influence of high-pressure gradient associated with cyclones that surges wind speed in atmosphere above the wide area. In this process, strong winds lift and transport a huge quantity of dust and sand from bare, dry area to hundreds to thousands of kilometers. Particles with huge size, density or with the presence of more water content deposit in the soil easily under the gravitational pull. Dust storm happens when the value of threshold loose particles increases that can be lifted up from ground due to rise in the wind's force. Floral community acts as a protective covering to the earth's surface from the wind erosion. In the absence of vegetation, drought also contributes to the appearance of dust storms due to poor farming/grazing practices/inadequate water management and by exposing dry and loos soil to wind. About 40% of atmospheric particles are dust particles as a result of wind erosion. After their release from soil surface, dust particles are elevated to higher in the troposphere through turbulent mixing and convective updrafts. The transportation of suspended dust depends on their size, wind speed, and meteorological conditions while gravitational force is also an important constrain that decides the deposition of dust particles back down to the ground (Saxena et al. 2021). Such dry deposition is a result of impaction and turbulent diffusion, where coarser particles are deposited more frequently than finer particles. Atmospheric dust load can also decrease through wet deposition through various precipitation phenomena like rain, fog, and hail (Sonwani and Kulshrestha 2019; Saxena et al. 2020; Sonwani and Saxena 2021).

There are three different physical processes which cause dust particles to be emitted into the atmosphere: (1) direct aerodynamic lifting, (2) ejection of dust aerosols from soil aggregates by impacting saltating particles, and (3) ejection of dust aerosols from soil aggregates that are participating in saltation. Aerodynamic lifting of dust particles to the atmosphere happens when the aerodynamic force exerted on the dust particle dislodge particles from the surface. The dust entrainment to the atmosphere from the ground is related with the saltation process. Saltation refers to a layer of soil moving with the wind just above the surface. The vertical mass flux carries lifted particles which undergo ballistic trajectories and can eject new particles into the stream.

Wind causes trigger the processes of creep, saltation, and suspension to transport the dust particles. Larger particles (>500 μ m diameter) usually creep on the soil surface. In the process of saltation, wind transports particles of 60–500 μ m diameter size at a height less than 1.5 m above ground level. Moreover, suspension refers to longer range transport of smaller particles (<60 μ m diameter). The transport of soil particles by wind can thus be crudely separated into several physical regimes: long-term suspension (< ~20 μ m diameter), short-term suspension (~20–60 μ m), saltation (~60–500 μ m), and creep (> ~500 μ m).

Particles of size greater than 100 μ m are generally called settleable particles and settle down under the influence of gravity. Particles of size below 100 μ m are called suspended particles and can remain suspended in the air for several days. The aerodynamic behavior of dust is generally governed by shape, size, density, chemicals composition, and electromagnetic properties of the dust particles. Normally, coarser particles usually return to the ground soon after being entrained whereas smaller and lighter particles are carried to a long distance (Gillett and Morales 1979).

Long-term suspended dust can remain in the atmosphere up to several weeks and can thus be transported thousands of kilometers from source regions (Miller et al. 2006). However, particles with diameters <10 microns (PM10) and those with diameter <2.5 microns (PM2.5) are the focal area of research from the health impact point of view (Sonwani et al. 2016; Goel et al. 2021). Following the SDS event, atmospheric PM10 dust concentrations can reach up to 15,000 μ g/m³ whereas hourly PM2.5 concentration touches about 1000 μ g/m³ (Jugder et al. 2011).

The most important climatic factors which influence the frequency and severity of SDS events depends on (1) local winds, (2) atmospheric turbulence, (3) weather fronts, and (4) anticyclonic condition. Wind is the result of differential heating of earth's surface producing differential pressure and temperature which causes local wind. The profile of local wind varies widely from region to region based on topography and other meteorological factors. Depending on the velocity, local winds can cause severe SDS events. Atmospheric turbulence results in spatially dispersing the dust particles from the source-areas. Higher the turbulence, higher will be the spatial spread of the dust particles. Weather fronts play an important role for the regional SDS events. The weather fronts are created when two different air masses with different air temperature, humidity, density pressure, and direction come face to face. The consequent weather becomes active and aggressive. Anticyclonic weather condition is the main causal factor for dust emission and large-scale circulation of dust. An anticyclone is defined as a large-scale circulation of winds around a central region of high atmospheric pressure, clockwise in the Northern Hemisphere and counterclockwise in the Southern Hemisphere. Anticyclonic condition often creates severe SDS events.

6.3 Drivers of Sand and Dust Storm

About 40% of particulate in the lower atmosphere is a result of dust particles released from wind erosion. The main regions for the origin of these mineral dusts are the arid regions of Northern Africa, the Arabian Peninsula, Central Asia, and China. Comparatively, Australia, America, and South Africa make minor contributions. Global estimates of dust emissions vary between one and three Gigatons per year. SDS happen as a consequence of a sequence of interwoven direct and indirect and independent causes. Land degradation, desertification, and changing climate are some of the major drivers of SDS that are aggravated by careless land use planning, strong winds, frequent and severe drought condition for longer period of time. There

Factors	Soil/sediment	Vegetation	Landform
Wind speed (+)	Soil/sediment type	Туре	Surface roughness $(+/-)$
Wind direction	Dust composition	Coverage (-)	Slop (–)
Turbulence (+)	Soil/sediment structure	Density	Ridge
Precipitation (-)	Organic matter (-)	Distribution (+/-)	
Evaporation (+)	Carbonate (-)		
Air temperature	Bulk density		
Air pressure (+)	Degree of aggregation (-)		

Table 6.1 Factor affecting wind erosion

Note: (+) indicates that the factor promotes wind erosion; (-) indicates that the factor has a shielding effect, reducing wind erosion; (+/-) indicates the effect can be positive or negative

are few important factors that are identified for the SDS events in the South Asian region. The first is changing wind pattern and the second is expanding sand and dust source areas. The changing wind pattern can be classified into (1) surface winds exceeding the erosion threshold, (2) incidence of extreme weather events, and (3) variable wind direction. Moreover, the expansion of the dust storm and sand storm source area can be identified on the basis of (1) increasing frequency of drought and dry arid region; (2) sparse vegetation; (3) degradation of land use pattern; and (4) interference with defensive earth surface crusts. There is a necessity to identify drivers (natural and anthropogenic both) of SDS that contributes most of the global dust emissions. Wind erosion acts as a key driver of SDS emissions in every system. Basically, erodibility of surface material along with dryness is a natural governing factor of SDS.

6.3.1 Natural Drivers

Topographic depressions in the arid region contribute 75% of the current global dust emissions (ESCAP 2018; UNEP, WMO, UNCCD 2016). Identification of the natural drivers is one of the important tasks to recognize the global dust emission. Strong winds (frequently originated by thunderstorms) are main drivers of SDS event in every region of the world. In arid and semi-arid regions, strong winds pull dust from the ground up in the atmosphere and create sand and dust storm. When the wind force surpasses the threshold limit for the loose soil particles to be lifted off the ground, dust storm occurs. After released, dust particles are elevated to high in the troposphere by turbulent mixing and convective updrafts. Then these particles are transported by winds and their transportation depends on their size and meteorological variables such as air temperature, humidity, and height of the boundary layer (Knippertz 2014). Distribution of vegetation across the geographical area is also responsible for the occurrence of sand and dust storm in that area. The sand and dust storms are frequent in plain area with sparse vegetation, and these features allow winds to build momentum to produce stronger winds that drive more dust into the atmosphere.

6.3.2 Anthropogenic Drivers

There are a variety of manmade sources behind the SDS that are one-third of the total natural sources; however, from the past few decades, land use has been changing thus contributing to increase in the anthropogenic drivers contributing SDS. Manmade sand/dust emission can be classified into their direct and indirect emission sources. Anthropogenic drivers, such as changes in land use pattern, agriculture activities, diversion of water bodies, and deforestation, contribute up to 25% of the present global dust generation (ESCAP 2018; UNEP, WMO, UNCCD 2016). Inadequate water management, poor farming, and grazing practices are some of the drivers cause dry conditions resulting in the appearance of SDS. The changing climatic conditions are also responsible for the generation of SDS. Several authors estimated the average contribution of the anthropogenic drivers ranging from 10 to 50% in global dust emission (Tegen et al. 2004, 2006; Mahowald and Luo 2003). The major anthropogenic factors of SDS may be highlighted as follows:

- 1. Overgrazing and thumping/stampede by livestock
- 2. Overuse of agricultural field in cultivation, prolonged farming without crop rotation, improper water impoundment, over-drafting of ground water, increased fire frequency, etc.
- 3. Deforestation to clear land for urbanization and agricultural practices
- 4. Over-exploitation of natural vegetation for domestic uses like fuel, construction material, etc.
- 5. Bio-industrialization

Even though there is large uncertainty about the impact of human activities on dust emission from disturbed soil, deforestation, disturbance in the hydrological cycle, and changing climate, several studies mentioned that the anthropogenic contribution to the dust emission is not negligible at the world scale (Guan et al. 2016; Huang et al. 2015; Tegen and Fung 1995). Figure 6.2 shows the percentage contribution of the different SDS emission sources.

6.4 State and Trends of SDS

Dust storms are significant climatic events for the people living in the "Dust Belt". However, there are very few studies available on the state and trend of SDS events globally. A large temporal variation in dust emissions has also been observed. Dust events generally happen during late spring and summer season; however, there is also substantial transportation of dust particles into the South America during the winter period.



Fig. 6.2 Causes of SDS in Asia

Though it is postulated that there is substantial temporal and spatial variation in the SDS events, but there is no consistent evidence to substantiate that SDS occurrence and intensity has raised globally in recent years. Ground-level monitoring records, satellite data, and the results of dust model are used to identify the distribution and occurrence of SDS events. Though it has been observed that SDS events have increased in few parts mainly due to deforestation activities caused by the urbanization and industrialization but in few places, there are record of lesser number of SDS events due to improvement in vegetative cover. A recent study has observed that within the last few decades there are noticeable changes in the SDS pattern in Australia, Central Asia, and in the US high plains; while SDS events over Northern Africa, the Middle East, and South America have continued with similar activity. Few stations like the Sahel show a strong rising trend of great disaster, others indicate a descending trend (e.g., Mexico City) (Goudie and Middleton 2006). Several studies have conclusively established a decreasing trend of SDS activities in Northern China; however, investigation of storms in the Tengger Desert in China has revealed that the frequency of dust storms has increased significantly during 2000-2007 in comparison to 1980s (Guan et al. 2017). In the Eastern Asian countries like Mongolian People's Republic, no conclusive pattern is observed, rather different parts of the country observe differential patterns of annual duststorm frequency. Long-term SDS frequency data for Australia show an increasing trend for the continent. Dust storm frequency in Sahelian Africa has been observed to increase sharply in the early 1970s. During the 1960s, fewer than two dust storms observed in a year that was rose through the 1970s to peak at 85 dust storm days in 1983 that continued high through the 1980s. In South Asia, Hamoun Basin which is considered as one of the dustiest areas of south Asia is been reported to experiencing an increased level of dust storm frequency over the period 2000–2004. The scientists have observed a strong correlation with current level of global dust generation and man-made dust emissions. The expansion of livestock grazing areas is considered to have contributed in the annual dust concentration by 500% in the western United States (Neff et al. 2008). Even today, crop land and rangelands are one of the major origins of SDS in North America (Lee and Gill 2015). Similarly, increasing industrial production especially mining and related operations create dust sources.

The comparative involvement of anthropogenic activity to the present levels of dust production varies from 10 to 50% (Tegen et al. 2004; Mahowald and Luo 2003). However, there are wide spatial variations in releases between natural and man-made sources. As North Africa release about 20% of its release by man-made activities but contribute about 55% of the world total release, while 75% of Australia's dust is man-made but contribute about 13% of the world man-made release. In reference to the notable contributions over the twentieth century, the data mentioned a doubling of world dust abundance in the air due to man-made activities and/or climate change. Stanelle et al. (2014) mentioned the comparative significance of changing climate and man-made land cover change for dust generation. Thus, the world yearly dust generation has raised by 25% as compared to the late nineteenth century. Approximately 56% of this variation contribute to change in climate and 40% to anthropogenic land cover change.

6.5 Geography of Dust Storms in South Asia

SDS events are common in every part of South Asia. SDS critically impacts the South Asian subregion. Since past several decades, the climatology of the Asian SDS was well documented through synoptic records and satellite data (Sun et al. 2001; Qian et al. 2002; Shao et al. 2003; Rezazadeh et al. 2013; Zhao et al. 2008). The SDS in the South Asian region are originated by pre-frontal and post-frontal winds in winter, and strong "Shamal" winds in summer. Moreover, SDS of low scale, time duration and strength are triggered by downwind regimes, sea breeze, and convection cell in topographic regions (Zarrin et al. 2011; Choobari et al. 2012). It is postulated that the increased SDS activity over the South Asian region during the El-Nino years linked with strong westerlies (Abish and Mohanakumar 2013); however, La-Nina phases are observed to be linked with the disturbance in precipitation over Indian Ocean, central and south west Asia (Barlow et al. 2002).

The bigger SDS incidences happen in the Sistan Basin in south-eastern of Iran and south-western Afghanistan, areas of southeastern of Iran, Baluchistan in Pakistan, the Thar Desert of Rajasthan in western India, the plains of Afghan Turkestan, and the Registan area of Uzbekistan. Dust originated from such areas is transported north to Central Asia, south over the Arabian Sea, and east over South-East Asia. The distribution of SDS belt in south Asia is shown in Fig. 6.3.



Fig. 6.3 Distribution of the SDS belts in south Asia (Adapted from: UNEP, WMO, UNCCD 2016)

6.5.1 India

Dust storms are most common in north-western India, in the arid and semi-arid states of Rajasthan, Haryana, and Punjab, but they also occur on the Ganges plain and in the northern cold arid areas of Jammu and Kashmir. In India the SDS events are regulated by farming activities and desert dust transportation from western India (Prasad et al. 2007). Maximum SDS occurrence is noticed in the Rajasthan with detected frequency beyond 70% along the seasonal Ghaggar River. The Rajasthan Desert is an important source of dust in southwest Asia. Human-induced desertification has been caused due to wind erosion and deposition of sediments in saline lake basins and agricultural areas in the Indian Plateau (Pandithurai et al. 2008). Indian SDS events are mostly seasonal in their incidence, and occur mostly in the pre-monsoon months (from April, May to June). In Rajasthan maximum dust storm frequency occurs at Ganganagar, which, with 17.0 days per annum (highest

in India) and frequencies decrease south and westward, Bikaner records 8.0 per annum and Jodhpur 6.0. (Krishnan 1978) has put forward evidence indicating an increasing trend of dust storms in the Jodhpur region in association with a decreasing trend of monsoonal rainfall over a 22-year period.

6.5.2 Pakistan

Dust storms occur in all parts of Pakistan, largely during the hot season that prevails from April to June. Kureshy and Ahmad (1977) divide the country into four climatic regions that provide a good framework for an analysis of dust-raising activity in Pakistan. The subtropical coastal strip (arid marine) includes the stations of Pasni and Ormara and the Indus delta south-east of Hyderabad. The area receives less than 175 mm rainfall per annum, and dust storms occur largely from April to July. The subtropical continental plateau (extreme climate, very arid) receives less than 125 mm rainfall a year. Snead (1968) reports that, for southern Pakistan as a whole, dust storms are most frequent in the afternoon, although larger, less local storms associated with an approaching eastern depression occur at any hour of the day and sometimes at night. The Makran coast is a hyper-arid area and the important universal dust sources (Goudie and Middleton 2001).

6.5.3 Afghanistan

As per the ground-based data collection, the maximum occurrence of dust storms happens at the junction of the borders of Iran, Pakistan, and Afghanistan (Middleton 1986). The bed of Lake Hamun and the large deltaic fan of the Helmand River are identified with the dust storms as a predominant source, perhaps due to the diversion of upstream water for irrigation purpose and by risky droughts in recent years. In Afghanistan, there is a widespread man-made source in the Hindu Kush, with several rivers nourishing into the Helmand River that support agriculture activity. There is also natural source from seasonal lakes within the Sistan Basin.

6.6 SDS Hazards and Their Impacts

The impact of SDS on the environment is far reaching. One of the primary impacts of SDS on the atmosphere is its impact on local atmospheric energy balance (Singh et al. 2004). Dust particles in the atmosphere disturb the local energy balance by impacting atmospheric radiation and absorption profile of solar energy in the atmosphere. Change in energy balance disturbs local climatic condition aggravating climatic extremities (Highwood and Ryder 2014) which ultimately impact regional bio-geochemical cycle. Current research suggests that SDS also play a role in extreme weather events like altering tropical storm, escalation of cyclonic activities, or frequent drought events.

Dust storms also play a role in impacting local rainfall pattern (Creamean et al. 2013). Dust particles in the atmosphere act like a condensation nuclei for the formation of atmospheric cloud. Atmospheric dust particles disturb radiation balance which effects temperature profile of the atmosphere (Maley 1982). One of far-reaching effects of dust storm events on global climate change impact is that that the mineral dust on glaciers lowers surface albedo and faster ice melting rate (Oerlemans et al. 2009). Some researcher has observed impact of dust particles on global Ozone balance by reducing photolysis rates of ozone production (Martin et al. 2003).

SDS are also known for their impacts on the south Asian monsoon, radiative forcing, climate, and hydrological cycles (Sonwani and Saxena 2021; Kaskaoutis et al. 2017, 2018; Pandey et al. 2017). Several other studies identified the impact of SDS on aerosol physical, optical, and radiative properties over the South Asian region (Kedia et al. 2018; Bhattacharjee et al. 2007; Prasad and Singh 2007).

Apart from the several negative impacts of SDS on environment, it has some positive global impacts in terms of their transboundary movement, impact on global climate, and their impacts on biogeochemical cycling (World Bank 2019; Prospero and Mayol-Bracero 2013). Some of the major impacts of SDS events are discussed below:

6.6.1 Impacts on Ocean and Its Productivity

SDS delivers a main source of several nutrients and trace elements to the oceans (Mahowald et al. 2018). Such elements are important for all life forms and can control over marine primary production (Jickells and Moore 2015). The impact on ocean primary productivity happens mainly through the effects of dust-borne nutrients on phytoplankton. Thus, SDS is responsible for boosting primary productivity of ocean but having adverse effects on coral reefs. It is also responsible for increasing primary productivity at surface and sea bed by increasing the growth of phytoplankton (Rosenberg et al. 1983). The dust deposition on marine environment has been reported to increase the primary productivity of ocean by 6% (Jickells and Moore 2015). Similarly, trace metals, nutrients, organic constituents, and microorganism are also some other important factors deposited along with the dust and sand storm and responsible for significant changes in the coral reefs and other ecosystems globally (Hallock 2001). Macronutrients like phosphorus and nitrogen, and the micronutrient like iron play an important role in marine primary production (Jickells and Moore 2015). It was also mentioned that most of the atmospheric nitrogen (organic and inorganic) deposition to ocean comes from SDS appear to be both soluble and bioavailable. In contrary, phosphorus and iron carried by SDS are not immediately soluble in water, and therefore are not bioavailable (Jickells et al. 2005; Tagliabue et al. 2017; Okin et al. 2011).

6.6.2 Impact of SDS on Air Quality

SDS activities enhance the overall loading of the atmospheric particles that affect visibility and air quality (Verma et al. 2013; Sikka 1997). Desert dust contributes about 50% of the naturally occurring tropospheric aerosols. SDS events increase the concentration of PM2.5 and PM10 in the atmosphere which ultimately impacts general public health (Saxena et al. 2020; Sonwani et al. 2021a; Sonwani and Kulshreshtha 2016; Maji et al. 2015, 2020). Poor air quality has been linked to be responsible for 1.4% of global mortality and 2% of all cardiopulmonary disease (WHO 2006). Exposure to elevated levels of dust concentration for a prolonged period of time has been linked to premature mortality due to different cardiorespiratory diseases. Inhalation of fine dust particles exposes individuals not only to hazardous fine mineral particulates, but different bio-allergen like spores, bacteria, fungi, and potential allergens are also carried with dusts (Kellogg et al. 2004; Sonwani et al. 2022). The Global Burden of Disease Comparative Risk Assessments study published by the World Health Organization (WHO) designates indoor air pollution and ambient air pollution as the world's second- and third-largest environmental health risks factors, respectively (WHO 2014). Impacts of SDS events on air quality over the north-western parts of India and Indo Gangetic Plain (IGP) have been extensively studied (Sonwani and Kulshrestha 2019; Sonwani et al. 2021a, b).

6.6.3 Human Health Impacts

SDS events lead to increase in particulate matter concentration in the atmosphere. The elevated levels of particulate matter concentration in the atmosphere are linked to so many cardio-respiratory diseases ranging from allergic symptoms to exacerbation of asthma occurrence, surges in bronchitis cases, whereas long-term exposures have been reported to be responsible for chronic health effects like cardiovascular changes like chronic obstructive pulmonary diseases (COPD) as well as congestive heart failures (Guarnieri and Balmes 2014; Maji et al. 2017, 2018; Chen et al. 2012; Mostofsky et al. 2012; Saxena and Sonwani 2019a).

Not only concentration, the origin of dust particles plays a major role in the mechanism of health impacts. Size, shape as well as composition of the particulate are all governing factors for the mechanism of toxicity. The dust particles can vary widely in size, shape, and composition depending on the nature of parent soils, emissions as well as transport processes. Acute exposure of elevated levels of particulate matter is linked to effects like bronchoconstriction and increased asthma symptoms (Kanatani et al. 2010); whereas scientific evidences have also linked long-term exposure to diseases like cancers (Arden Pope III and Dockery 2006), and cognitive impairments (Chen et al. 2009). Accounting the health significance of ambient air pollution, the International Agency for Research on Cancer (IARC) has designated outdoor air pollution as carcinogenic to humans (Bae and Hong 2018).

6.6.3.1 Cardio-respiratory Diseases

The normal human lung consumes about 5–6 mL oxygen per minute at an esophageal temperature of 28 °C (Loer et al. 1997). Therefore, any contaminant in ambient air enters the human cardio-respiratory system easily during respiration. Particulates, once inhaled, enter deep into the respiratory system depending on size fraction. The most common respiratory diseases affected by air pollution include upper respiratory symptom (URS), lower respiratory system (LRS), bronchitis, asthma, and chronic obstructive pulmonary disease (COPD), and even cancer due to prolonged damaging effects (Saxena and Sonwani 2019b).

Particulates of size below 2.5 μ m can reach up to alveoli of lungs and are capable of absorbing by blood and transported to heart. A large number of literatures have reported cardiovascular events such as stroke, myocardial infarction, arrhythmias, and sudden cardiac death due to chronic exposure to elevated levels of dust particles (Arden Pope III et al. 2004; Brook et al. 2010; Saxena and Naik 2018). Cardiorespiratory impacts of particulate matter have been reported from the literature throughout the world. Studies conducted in Europe reported an increase in mortality rate of 0.69% for 10 µg/m³ increase in PM10 concentration (Zanobetti et al. 2003). Studies conducted in the United States reported an increase in cardiopulmonary mortality of 0.31% for 10 µg/m³ increase in PM10 concentration (Samet et al. 2000). Studies conducted across Asia have reported an increase in mortality rate of 6% for 10 µg/m³ increase in PM10 concentration (HEI 2010). Plausible evidences show exacerbation of cardiovascular disease with increase in particulate matter concentration. A recent global meta-analysis of vital statistics has observed an increase in cardiovascular diseases by 41% during 1990 and 2013 (Roth et al. 2015).

Literatures have established clear link of increase in hospital admission with respiratory, cardiovascular disease and increase in acute bronchitis after dust events (Saxena et al. 2021; Nogueira 2009). Studies conducted in Asian cities have observed a strong correlation of increase in emergency visit and daily hospital admissions rate with cardio-respiratory diseases after dust storms events (Otani et al. 2012, 2014). Similar observations have been reported from Italy (Sajani et al. 2011) and Spain (Jiménez et al. 2010) which have observed increase in respiratory mortality rate among the elders during Saharan dust events. Exacerbation in asthma attacks after episodic increase in particulate matter concentration has been conclusively established in so many literatures (Atkinson et al. 1999; Dockery 2001; WHO 2003). Asthma is one of the world's leading noncommunicable diseases. As per the estimates made by the World Health Organization, asthma affected an estimated 262 million people in 2019 and caused 461,000 deaths.

6.6.3.2 Valley Fever

Valley fever, also called coccidioidomycosis, is an infection caused by the fungus Coccidioides. Two Coccidioides fungi species (C. immitis and C. posadasii) cause valley fever (Williams et al. 1979). These fungi are commonly found in soil and are almost endemic in the southwestern United States and parts of Mexico and Central and South America. The valley fever is one of the fast-rising infectious diseases in the Southwestern United States. A positive correlation between dust storm events and increase in valley fever incidents has been observed by several researchers (Tong et al. 2017).

6.6.3.3 Eye and Skin Infections

Peoples leaving in arid areas often experience eye and skin infections after dust storm events. High concentration of particulate matter in the ambient air reported to cause acute effects like conjunctivitis, itchy eye, and skin infections. However, these effects are not fatal, and effects are allergic response to dust particles only (Saxena and Srivastava 2020).

6.6.4 Economic Impacts

Apart from its effects on air quality and human health impact, SDS events are also related to have both immediate and in longer term economic impacts. The direct effect of SDS event is observed on agriculture productivity due to soil erosion, loss of nutrient-rich topsoil, and crop damages due to sandblasting and burial of seedlings associated with dust storm events. Direct economic losses due to dust storm events also come from destruction of infrastructure services like transport and communication systems, buildings, roads, etc. Long-term indirect economic impacts of dust storm events are accounted for the economic loss due to chronic health problems and the disruption of global climate regulation services. It is reported that millions of tracts of agricultural land have become unfertile due to loss of top soil as well as soil pollution due to deposition of pollutants.

Continued incidence of dust storms can also result in the migration of people from the place. Some of the major SDS events and the estimated economic loss due to dust storm event in the past few decades may be noted as follows:

- US\$262 million: A single massive dust storm called, Red Dawn, in Australia on 22–23 September 2009
- US\$964 million: Dust events in northern China from 2010 to 2013
- US\$125 million: Dust events in Sistan region, Iran from 2000 to 2005
- US\$0.5 million: A dust storm in eastern Mongolia in May 2008

6.7 Climate Change and SDS Events

SDS aerosol attracts gaining attention in recent times due to their interrelation with environment, weather, and climate (Shao et al. 2011). SDS aerosol can alter global thermal balance by absorbing and scattering the incoming shortwave and outgoing longwave solar radiation (Wu et al. 2018). Mineral dust particles in the atmosphere impact weather as well as global and regional climate (Boucher et al. 2013; Nickovic et al. 2012). SDS particles act as condensation nuclei for cloud formation; clouds absorb solar radiation, which indirectly affects the energy reaching the earth's surface (Boucher et al. 2013). SDS particles also influence the growth of cloud
droplets and ice crystals, thus affecting the amount and location of precipitation (Mahowald and Kiehl 2003). SDS also affects precipitation processes through connective activities due to variable temperature (Maley 1982; Tiwari and Saxena 2021).

6.7.1 Positive and Negative Forcing and SDS

Aerosol particles emitted during SDS events can be different in the size, shape, and chemical composition that determine their optical properties. Thus, the aerosol particles originated during SDS events can have positive (warming) and negative (cooling) radiative forcing of climate system (Sokolik and Toon 1996; Dey et al. 2004; Sonwani et al. 2021b). Such forcing (positive/negative) depends on extent of dust layer and local surface albedo (Tanré et al. 2003). At global level, the net annual radiative forcing for desert dust at the top of the atmosphere is negative (cooling), whereas at the regional level, it is identified with positive value (warming) that occurs over snow-covered or desert areas (Bonan 1997; Miller and Tegen 1998). Due to the dust, darker surface (ocean) absorbs more sunlight whereas brighter surfaces like snow/ice surfaces. Such SDS events play an important role in changing radiative forcing that affects global dust cycle that might play an important role in increasing past climate change (Abbot and Halevy 2010).

6.7.2 Connection Between SDS and Climate Change

High temperature, low rainfall, and strong wind speeds are important variables for the SDS origin (Guan et al. 2017). Different climatic simulation models have confirmed that the surface wind speed and precipitation are the main meteorological factors to influence the SDS formation and have strong impact on the Asian dust formation and occurrences (Zhang et al. 2003). SDS during the winter and spring season are influenced by cold air interference from north that is triggered by the Siberian high, while during summer, SDS occurrence in the Asian region is generally controlled by high pressure over the Caspian Sea (Indoitu et al. 2012). The pressure gradient between Caspian Sea High and Hindu Kush Low acts as the main governing factor affecting the SDS action over southwest Asia and the Karakum desert (Kaskaoutis et al. 2016).

6.8 Conclusion

SDS is a natural extreme atmospheric event of arid and semi-arid regions that occurs beneath the earth's surface. It is a result of strong turbulent winds in arid zone that carry dust particles to distant locations and impact air quality and atmospheric visibility. SDS events are responsible for the presence of airborne dust particles. The occurrence of SDS event depends on wind speed and susceptibility of surface materials for erosion. Thus, it transports a huge quantity of particulate matter to thousands of kilometers away from its source location. Globally, more than 150 nations are affected directly by such events, and 45 nations are identified as SDS source-areas. The areas like deserts, beaches, and other dry lake beds are most susceptible for wind erosion, hence important sources for SDS. Climatic factors directly cause dust storms, while several human-induced factors influence the impacts and frequency of SDS events. Land degradation, desertification, indiscriminate exploitation of land, and water resources are important drivers of SDS events.

SDS have significant socio-economic impacts on human health, agriculture, industry, transportation, water, and air quality. The direct impact of dust storm events is on agriculture due to loss of top fertile soil, soil erosion, and crop damages. Following the dust storm events, disruption of communication and transport services are some of the immediate effects. Dust particles in the atmosphere impact radiative balance impacting local and regional climatic balance. Several studies have also observed the impact of dust storm events with climatic change impacts. Drought intensification as well as enhanced precipitation are some of the long-term climate change impacts which are correlated with dust storm events. SDS events are also linked with health concerns, airborne dust aggravates respiratory symptoms. Inhalation of fine dust particles exposes individuals not only to hazardous fine mineral particulates, but also to harmful combinations of pollutants, spores, bacteria, fungi, and potential allergens carried along with mineral dusts. Globally, 334 million people and 14% of world's children experience asthmatic symptoms. In addition, dust storms can transport pathogens such as meningitis and valley fever (Abdussalam et al. 2013).

Thus, SDS is identified as one of the important extreme events with global importance due to their huge impacts on the economy, human health, and the environment. SDS has large-scale impact and affects a range of Sustainable Development Goals (SDGs) related to human health, productivity, agriculture, and infrastructure (transport). With increasing urbanization and land use/land cover changes and an increase in surface temperature, the frequency of similar SDS events is expected to increase in the future. To reduce the harmful impacts of SDS is identified as a priority area for achieving SDGs. With such wide transboundary impacts, managing the SDS risks requires both global and regional cooperation.

References

Abbot DS, Halevy I (2010) Dust aerosol important for Snowball Earth deglaciation. J Clim 23(15): 4121–4132

Abdussalam A, Monaghan A, Dukic V, Hayden M, Hopson T, Leckebusch G (2013) Meteorological influences on the interannual variability of meningitis incidence in northwest Nigeria. In EGU General Assembly Conference Abstracts, p EGU2013-7600

Abish B, Mohanakumar K (2013) Absorbing aerosol variability over the Indian subcontinent and its increasing dependence on ENSO. Glob Planet Chang 106:13–19

Arden Pope C III, Dockery DW (2006) Health effects of fine particulate air pollution: lines that connect. J Air Waste Manag Assoc 56(6):709–742

- Arden Pope C III, Burnett RT, Thurston GD, Thun MJ, Calle EE, Krewski D, Godleski JJ (2004) Cardiovascular mortality and long-term exposure to particulate air pollution: epidemiological evidence of general pathophysiological pathways of disease. Circulation 109(1):71–77
- Atkinson RW, Bremner SA, Ross Anderson H, Strachan DP, Martin Bland J, Ponce de Leon A (1999) Short-term associations between emergency hospital admissions for respiratory and cardiovascular disease and outdoor air pollution in London. Arch Environ Health 54(6): 398–411

Bae S, Hong YC (2018) Health effects of particulate matter. J Korean Med Assoc 61(12):749–755

- Barlow M, Cullen H, Lyon B (2002) Drought in central and southwest Asia: La Niña, the warm pool, and Indian Ocean precipitation. J Clim 15(7):697–700
- Bhattacharjee PS, Prasad AK, Kafatos M, Singh RP (2007) Influence of a dust storm on carbon monoxide and water vapor over the Indo-Gangetic Plains. J Geophys Res Atmos 112(D18)
- Birmili W, Kerstin Schepanski A, Ansmann G, Spindler I, Tegen B, Wehner AN et al (2008) A case of extreme particulate matter concentrations over Central Europe caused by dust emitted over the southern Ukraine. Atmos Chem Phys 8(4):997–1016
- Bonan GB (1997) Effects of land use on the climate of the United States. Clim Chang 37(3): 449–486
- Boucher O, Randall D, Artaxo P, Bretherton C, Feingold G, Forster P, Kerminen VM, Kondo Y, Liao H, Lohmann U, Rasch P (2013) Clouds and aerosols. In: Climate change 2013: the physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, pp 571–657
- Brook RD, Rajagopalan S, Arden Pope C III, Brook JR, Bhatnagar A, Diez-Roux AV, Holguin F et al (2010) Particulate matter air pollution and cardiovascular disease: an update to the scientific statement from the American Heart Association. Circulation 121(21):2331–2378
- Chen R, Kan H, Chen B, Huang W, Bai Z, Song G, Pan G (2012) Association of particulate air pollution with daily mortality: the China Air Pollution and Health Effects Study. Am J Epidemiol 175(11):1173–1181
- Chen WQ, Zhao XL, Hou Y, Li ST, Hong Y, Wang DL, Cheng YY (2009) Protective effects of green tea polyphenols on cognitive impairments induced by psychological stress in rats. Behav Brain Res 202(1):71–76
- Choobari OA, Zawar-Reza P, Sturman A (2012) Feedback between windblown dust and planetary boundary-layer characteristics: sensitivity to boundary and surface layer parameterizations. Atmos Environ 61:294–304
- Creamean JM, Suski KJ, Rosenfeld D, Cazorla A, DeMott PJ, Sullivan RC, White AB et al (2013) Dust and biological aerosols from the Sahara and Asia influence precipitation in the western US. Science 339(6127):1572–1578
- Cuevas Agulló E (2013) Establishing a WMO sand and dust storm warning advisory and assessment system regional node for West Asia: current capabilities and needs: technical report. WMO, UNEP, Collection(s) and Series: WMO- No. 1122
- Dey S, Tripathi SN, Singh RP, Holben BN (2004) Influence of dust storms on the aerosol optical properties over the Indo-Gangetic basin. J Geophys Res Atmos 109(D20)
- Dockery DW (2001) Epidemiologic evidence of cardiovascular effects of particulate air pollution. Environ Health Perspect 109(Suppl 4):483–486
- Evan AT, Dunion J, Foley JA, Heidinger AK, Velden CS (2006) New evidence for a relationship between Atlantic tropical cyclone activity and African dust outbreaks. Geophys Res Lett 33(19)
- Garrison VH, Shinn EA, Foreman WT, Griffin DW, Holmes CW, Kellogg CA, Majewski MS, Richardson LL, Ritchie KB, Smith GW (2003) African and Asian dust: from desert soils to coral reefs. Bioscience 53(5):469–480
- Gillett D, Morales C (1979) Environmental factors affecting dust emission by wind erosion. Saharan Dust 71–94
- Goel A, Saxena P, Sonwani S, Rathi S, Srivastava A, Bharti AK, Jain S, Singh S, Shukla A, Srivastava A (2021) Health benefits due to reduction in respirable particulates during COVID-19 lockdown in India. Aerosol Air Qual Res 21:200460

Goudie AS (2009) Dust storms: recent developments. J Environ Manag 90(1):89-94

- Goudie AS (2014) Desert dust and human health disorders. Environ Int 63:101-113
- Goudie AS, Middleton NJ (2001) Saharan dust storms: nature and consequences. Earth Sci Rev 56(1-4):179–204
- Goudie AS, Middleton NJ (2006) Desert dust in the global system. Springer Science & Business Media
- Gross A, Turner BL, Goren T, Berry A, Angert A (2016) Tracing the sources of atmospheric phosphorus deposition to a tropical rain forest in Panama using stable oxygen isotopes. Environ Sci Technol 50(3):1147–1156
- Grousset FE, Ginoux P, Bory A, Biscaye PE (2003) Case study of a Chinese dust plume reaching the French Alps. Geophys Res Lett 30(6)
- Guan X, Huang J, Zhang Y, Xie Y, Liu J (2016) The relationship between anthropogenic dust and population over global semi-arid regions. Atmos Chem Phys 16(8):5159–5169
- Guan Q, Sun X, Yang J, Pan B, Zhao S, Wang L (2017) Dust storms in northern China: long-term spatiotemporal characteristics and climate controls. J Clim 30(17):6683–6700
- Guarnieri M, Balmes JR (2014) Outdoor air pollution and asthma. Lancet 383(9928):1581–1592
- Hallock P (2001) Coral reefs, carbonate sediments, nutrients, and global change. In: The history and sedimentology of ancient reef systems. Springer, Boston, pp 387–427
- Han Y, Dai X, Fang X, Chen Y, Kang F (2008) Dust aerosols: a possible accelerant for an increasingly arid climate in North China. J Arid Environ 72(8):1476–1489
- HEI, 2010. Outdoor Air Pollution and Health in the Developing Countries of Asia: A Comprehensive Review, Special Report 18, Health Effects Institute, Boston, USA
- Herman JR, Bhartia PK, Torres O, Hsu C, Seftor C, Celarier E (1997) Global distribution of UV-absorbing aerosols from Nimbus 7/TOMS data. J Geophys Res Atmos 102(D14): 16911–16922
- Highwood EJ, Ryder CL (2014) Radiative effects of dust mineral dust: a key player in the earth system. In: Knippertz P, Stuut JB (eds). pp 267–283
- Hong Y-C, Pan X-C, Kim S-Y, Park K, Park E-J, Jin X, Yi S-M et al (2010) Asian dust storm and pulmonary function of school children in Seoul. Sci Total Environ 408(4):754–759
- Huang JP, Liu JJ, Chen B, Nasiri SL (2015) Detection of anthropogenic dust using CALIPSO lidar measurements. Atmos Chem Phys 15(20):11653–11665
- Indoitu R, Orlovsky L, Orlovsky N (2012) Dust storms in Central Asia: spatial and temporal variations. J Arid Environ 85:62–70
- Jickells T, Moore CM (2015) The importance of atmospheric deposition for ocean productivity. Annu Rev Ecol Evol Syst 46:481–501
- Jickells TD, An ZS, Andersen KK, Baker AR, Bergametti G, Brooks N, Cao JJ, Boyd PW, Duce RA, Hunter KA, Kawahata H, laRoche J, Liss PS, Mahowald N, Prospero JM, Ridgwell AJ, Tegen I, Torre R (2005) Global iron connections between desert dust, ocean biogeochemistry, and climate. Science 308:67–71
- Jiménez E, Linares C, Martínez D, Díaz J (2010) Role of Saharan dust in the relationship between particulate matter and short-term daily mortality among the elderly in Madrid (Spain). Sci Total Environ 408(23):5729–5736
- Jugder D, Shinoda M, Sugimoto N, Matsui I, Nishikawa M, Park S-U, Chun Y-S, Park M-S (2011) Spatial and temporal variations of dust concentrations in the Gobi Desert of Mongolia. Glob Planet Chang 78(1–2):14–22
- Kanatani KT, Ito I, Al-Delaimy WK, Adachi Y, Mathews WC, Ramsdell JW (2010) Desert dust exposure is associated with increased risk of asthma hospitalization in children. Am J Respir Crit Care Med 182(12):1475–1481
- Kashima S, Yorifuji T, Bae S, Honda Y, Lim Y-H, Hong Y-C (2016) Asian dust effect on causespecific mortality in five cities across South Korea and Japan. Atmos Environ 128:20–27
- Kaskaoutis DG, Houssos EE, Rashki A, Francois P, Legrand M, Goto D, Bartzokas A, Kambezidis HD, Takemura T (2016) The Caspian Sea–Hindu Kush Index (CasHKI): a regulatory factor for dust activity over southwest Asia. Glob Planet Chang 137:10–23

- Kaskaoutis DG, Rashki A, Houssos EE, Legrand M, Francois P, Bartzokas A, Kambezidis HD, Dumka UC, Goto D, Takemura T (2017) Assessment of changes in atmospheric dynamics and dust activity over southwest Asia using the Caspian Sea–Hindu Kush Index. Int J Climatol 37: 1013–1034
- Kaskaoutis DG, Houssos EE, Solmon F, Legrand M, Rashki A, Dumka UC, Francois P, Gautam R, Singh RP (2018) Impact of atmospheric circulation types on southwest Asian dust and Indian summer monsoon rainfall. Atmos Res 201:189–205
- Kedia S, Kumar R, Islam S, Sathe Y, Kaginalkar A (2018) Radiative impact of a heavy dust storm over India and surrounding oceanic regions. Atmos Environ 185:109–120
- Kellogg CA, Griffin DW, Garrison VH, Kealy Peak K, Royall N, Smith RR, Shinn EA (2004) Characterization of aerosolized bacteria and fungi from desert dust events in Mali, West Africa. Aerobiologia 20(2):99–110
- Knippertz P (2014) Meteorological aspects of dust storms. In: Mineral dust. Springer, Dordrecht, pp 121–147
- Krishnan A (1978) Climatic changes relating to desertification in the arid zone of north West India. Annals of Arid Zone (India) 16(3):302–309
- Kureshy KU, Ahmad KSU (1977) A geography of Pakistan. Oxford University Press
- Lee JA, Gill TE (2015) Multiple causes of wind erosion in the Dust Bowl. Aeolian Res 19:15–36
- Li F, Ginoux P, Ramaswamy V (2008) Distribution, transport, and deposition of mineral dust in the Southern Ocean and Antarctica: contribution of major sources. J Geophys Res Atmos 113(D10)
- Loer SA, Scheeren TW, Tarnow J (1997) How much oxygen does the human lung consume? J Am Soc Anesthesiol 86(3):532–537
- Mahowald NM, Kiehl LM (2003) Mineral aerosol and cloud interactions. Geophys Res Lett 30(9): 1475
- Mahowald NM, Luo C (2003) A less dusty future? Geophys Res Lett 30(17)
- Mahowald NM, Hamilton DS, Mackey KR, Moore JK, Baker AR, Scanza RA, Zhang Y (2018) Aerosol trace metal leaching and impacts on marine microorganisms. Nat Commun 9(1):1–15
- Maji S, Ahmed S, Siddiqui WA (2015) Air quality assessment and its relation to potential health impacts in Delhi, India. Curr Sci 109(5):902–909
- Maji S, Ahmed S, Siddiqui WA, Ghosh S (2017) Short term effects of criteria air pollutants on daily mortality in Delhi, India. Atmos Environ 150:210–219
- Maji S, Ghosh S, Ahmed S (2018) Association of air quality with respiratory and cardiovascular morbidity rate in Delhi, India. Int J Environ Health Res 28(5):471–490
- Maji S, Ahmed S, Ghosh S, Garg SK (2020) Evaluation of air quality index for air quality data interpretation in Delhi, India. Curr Sci 119(6):1019–1026
- Maley J (1982) Dust, clouds, rain types, and climatic variations in tropical North Africa. Quat Res 18(1):1–16
- Martin RV, Jacob DJ, Yantosca RM, Chin M, Ginoux P (2003) Global and regional decreases in tropospheric oxidants from photochemical effects of aerosols. J Geophys Res Atmos 108(D3)
- McKendry IG, Hacker JP, Stull R, Sakiyama S, Mignacca D, Reid K (2001) Long-range transport of Asian dust to the lower Fraser Valley, British Columbia, Canada. J Geophys Res Atmos 106 (D16):18361–18370
- McTainsh GH, Pitblado JR (1987) Dust storms and related phenomena measured from meteorological records in Australia. Earth Surf Process Landf 12(4):415–424
- Middleton NJ (1986) A geography of dust storms in South-west Asia. J Climatol 6(2):183-196
- Middleton N, Kang U (2017) Sand and dust storms: impact mitigation. Sustainability 9(6):1053
- Middleton NJ, Sternberg T (2013) Climate hazards in drylands: a review. Earth Sci Rev 126:48-57
- Miller RL, Tegen I (1998) Climate response to soil dust aerosols. J Clim 11(12):3247-3267
- Miller RL, Cakmur RV, Perlwitz J, Geogdzhayev IV, Ginoux P, Koch D, Kohfeld KE et al (2006) Mineral dust aerosols in the NASA Goddard Institute for Space Sciences ModelE atmospheric general circulation model. J Geophys Res Atmos 111(D6)

- Mostofsky E, Schwartz J, Coull BA, Koutrakis P, Wellenius GA, Suh HH et al (2012) Modeling the association between particle constituents of air pollution and health outcomes. Am J Epidemiol 176(4):317–326
- Muhs DR, Prospero JM, Baddock MC, Gill TE (2014) Identifying sources of aeolian mineral dust: present and past. In: Mineral dust. Springer, Dordrecht, pp 51–74
- Neff JC, Ballantyne AP, Farmer GL, Mahowald NM, Conroy JL, Landry CC, Overpeck JT, Painter TH, Lawrence CR, Reynolds RL (2008) Increasing eolian dust deposition in the western United States linked to human activity. Nat Geosci 1(3):189–195
- Nenes A, Murray B, Bougiatioti A (2014) Mineral dust and its microphysical interactions with clouds. In: Mineral dust. Springer, Dordrecht, pp 287–325
- Nickovic S, Vukovic A, Vujadinovic M, Djurdjevic V, Pejanovic G (2012) High-resolution mineralogical database of dust-productive soils for atmospheric dust modeling. Atmos Chem Phys 12(2):845–855
- Nogueira JB (2009) Air pollution and cardiovascular disease. Port J Cardiol 28(6):715-733
- Nogueira J, Evangelista H, de Morisson Valeriano C, Sifeddine A, Neto C, Vaz G, Moreira LS et al (2021) Dust arriving in the Amazon basin over the past 7,500 years came from diverse sources. Commun Earth Environ 2(1):1–11
- Oerlemans J, Giesen RH, Van den Broeke MR (2009) Retreating alpine glaciers: increased melt rates due to accumulation of dust (Vadret da Morteratsch, Switzerland). J Glaciol 55(192): 729–736
- Okin GS, Baker AR, Tegen I, Mahowald NM, Dentener FJ, Duce RA, Galloway JN, Hunter K, Kanakidou M, Kubilay N, Prospero JM (2011) Impacts of atmospheric nutrient deposition on marine productivity: roles of nitrogen, phosphorus, and iron. Glob Biogeochem Cycles 25(2)
- Otani S, Onishi K, Mu H, Yokoyama Y, Hosoda T, Okamoto M, Kurozawa Y (2012) The relationship between skin symptoms and allergic reactions to Asian dust. Int J Environ Res Public Health 9(12):4606–4614
- Otani S, Onishi K, Mu H, Hosoda T, Kurozawa Y, Ikeguchi M (2014) Associations between subjective symptoms and serum immunoglobulin E levels during Asian dust events. Int J Environ Res Public Health 11(8):7636–7641
- Pandey SK, Vinoj V, Landu K, Babu SS (2017) Declining pre-monsoon dust loading over South Asia: signature of a changing regional climate. Sci Rep 7(1):1–10
- Pandithurai G, Dipu S, Dani KK, Tiwari S, Bisht DS, Devara PCS, Pinker RT (2008) Aerosol radiative forcing during dust events over New Delhi, India. J Geophys Res Atmos 113(D13)
- Prasad AK, Singh RP (2007) Changes in aerosol parameters during major dust storm events (2001–2005) over the Indo-Gangetic Plains using AERONET and MODIS data. J Geophys Res Atmos 112(D9)
- Prasad AK, Singh S, Chauhan SS, Srivastava MK, Singh RP, Singh R (2007) Aerosol radiative forcing over the Indo-Gangetic plains during major dust storms. Atmos Environ 41(29): 6289–6301
- Prospero JM, Mayol-Bracero OL (2013) Understanding the transport and impact of African dust on the Caribbean basin. Bull Am Meteorol Soc 94(9):1329–1337
- Qian W, Quan L, Shi S (2002) Variations of the dust storm in China and its climatic control. J Clim 15(10):1216–1229
- Ramanathan VCPJ, Crutzen PJ, Kiehl JT, Rosenfeld D (2001) Aerosols, climate, and the hydrological cycle. Science 294(5549):2119–2124
- Rezazadeh M, Irannejad P, Shao Y (2013) Climatology of the Middle East dust events. Aeolian Res 10:103–109
- Rosenberg NJ, Blad BL, Verma SB (1983) Microclimate: the biological environment. Wiley
- Roth GA, Forouzanfar MH, Moran AE, Barber R, Nguyen G, Feigin VL et al (2015) Demographic and epidemiologic drivers of global cardiovascular mortality. N Engl J Med 372(14):1333–1341
- Sajani SZ, Hänninen O, Marchesi S, Lauriola P (2011) Comparison of different exposure settings in a case–crossover study on air pollution and daily mortality: counterintuitive results. J Expo Sci Environ Epidemiol 21(4):385–394

- Samet JM, Dominici F, Curriero F, Coursac I, Zeger SL (2000) Particulate air pollution and mortality in 20 US cities: 1987–1994. New Engl J Med 343:1742–1757
- Saxena P, Naik V (eds) (2018) Air pollution: sources, impacts and controls. Cabi, Wallingford, pp 1–217
- Saxena P, Sonwani S (2019a) Criteria air pollutants: chemistry, sources and sinks. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 7–48
- Saxena P, Sonwani S (2019b) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8): 6639–6656
- Saxena P, Srivastava A, Sonwani S (2021) Impact of dust storms on air quality and human health in Delhi. In: Health adaptation and resilience to climate change and related disasters - a compendium of case studies. National Institute of Disaster Management, New Delhi, pp 127–147
- Shao Y, Yang Y, Wang J, Song Z, Leslie LM, Dong C, Zhang Z, Lin Z, Kanai Y, Yabuki S, Chun Y (2003) Northeast Asian dust storms: real-time numerical prediction and validation. J Geophys Res Atmos 108(D22)
- Shao Y, Wyrwoll KH, Chappell A, Huang J, Lin Z, McTainsh GH, Mikami M, Tanaka TY, Wang X, Yoon S (2011) Dust cycle: an emerging core theme in Earth system science. Aeolian Res 2(4):181–204
- Sikka DR (1997) Desert climate and its dynamics. Curr Sci 72:35-46
- Singh RP, Dey S, Tripathi SN, Tare V, Holben B (2004) Variability of aerosol parameters over Kanpur, northern India. J Geophys Res Atmos 109(D23)
- Snead RE (1968) Weather patterns in southern West Pakistan. Arch Meteorol Geophys Bioklimatol Ser B 16(4):316–346
- Sokolik IN, Toon OB (1996) Direct radiative forcing by anthropogenic airborne mineral aerosols. Nature 381(6584):681–683
- Sonwani S, Kulshreshtha U (2016) Particulate matter levels and it's associated health risks in East Delhi. In: Proceedings of Indian aerosol science and technology association conference on aerosol and climate change: insight and challenges. IASTA Bull, vol 22, no 1–2
- Sonwani S, Kulshrestha UC (2019) PM 10 carbonaceous aerosols and their real-time wet scavenging during monsoon and non-monsoon seasons at Delhi, India. J Atmos Chem 76(3):171–200
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res Int 28(38):53058–53073
- Sonwani S, Saxena P, Kulshrestha U (2016) Role of global warming and plant signaling in BVOC emissions. In: Plant responses to air pollution. Springer, Singapore, pp 45–57
- Sonwani S, Saxena P, Shukla A (2021a) Carbonaceous aerosol characterization and their relationship with meteorological parameters during summer monsoon and winter monsoon at an industrial region in Delhi, India. Earth and Space. Science 8(4):e2020EA001303
- Sonwani S, Yadav A, Saxena P (2021b) Atmospheric brown carbon: a global emerging concern for climate and environmental health. Management of Contaminants of Emerging Concern (CEC). Environment 1:225–247
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021c) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3(690444):1–20
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1–2):39–50
- Stanelle T, Bey I, Raddatz T, Reick C, Tegen I (2014) Anthropogenically induced changes in twentieth century mineral dust burden and the associated impact on radiative forcing. J Geophys Res Atmos 119(23):13–526

- Stout JE, Warren A, Gill TE (2009) Publication trends in aeolian research: an analysis of the Bibliography of Aeolian Research. Geomorphology 105(1–2):6–17
- Sun J, Zhang M, Liu T (2001) Spatial and temporal characteristics of dust storms in China and its surrounding regions, 1960–1999: relations to source area and climate. J Geophys Res Atmos 106(D10):10325–10333
- Sun Y, Zhuang G, Wang Y, Zhao X, Li J, Wang Z, An Z (2005) Chemical composition of dust storms in Beijing and implications for the mixing of mineral aerosol with pollution aerosol on the pathway. J Geophys Res Atmos 110(D24)
- Tagliabue A, Bowie AR, Boyd PW, Buck KN, Johnson KS, Saito MA (2017) The integral role of iron in ocean biogeochemistry. Nature 543(7643):51–59
- Tanré D, Haywood J, Pelon J, Léon JF, Chatenet B, Formenti P, Francis P, Goloub P, Highwood EJ, Myhre G (2003) Measurement and modeling of the Saharan dust radiative impact: overview of the Saharan Dust Experiment (SHADE). J Geophys Res Atmos 108(D18)
- Tegen I, Fung I (1995) Contribution to the atmospheric mineral aerosol load from land surface modification. J Geophys Res Atmos 100(D9):18707–18726
- Tegen I, Werner M, Harrison SP, Kohfeld KE (2004) Relative importance of climate and land use in determining present and future global soil dust emission. Geophys Res Lett 31(5)
- Tegen I, Heinold B, Todd M, Helmert J, Washington R, Dubovik O (2006) Modelling soil dust aerosol in the Bodélé depression during the BoDEx campaign. Atmos Chem Phys 6(12): 4345–4359
- Tiwari S, Saxena P (2021) Air pollution and its complications. Springer-Nature, Singapore, pp 1– 178
- Tong DQ, Wang JXL, Gill TE, Lei H, Wang B (2017) Intensified duststorm activity and valley fever infection in the southwestern United States. Geophys Res Lett 44:4304–4312
- UNEP, WMO, UNCCD (2016) Global assessment of sand and dust storms. United Nations Environment Programme, Nairobi. Editor: Gemma Shepherd, UNEP, Published by the United Nations Environment Programme (UNEP), available through UNEP Live (uneplive.unep.org), UNEP website (http://www.unep.org/publications)
- United Nations, Economic and Social Commission for Asia and the Pacific (ESCAP) (2018) Sand and dust storms in Asia and the Pacific: opportunities for regional cooperation and action. United Nations Publication 2018, Bangkok
- Uno I, Eguchi K, Yumimoto K, Takemura T, Shimizu A, Uematsu M, Liu Z, Wang Z, Hara Y, Sugimoto N (2009) Asian dust transported one full circuit around the globe. Nat Geosci 2(8): 557–560
- Van Der Does M, Knippertz P, Zschenderlein P, Harrison RG, Stuut J-BW (2018) The mysterious long-range transport of giant mineral dust particles. Sci Adv 4(12):eaau2768
- Verma S, Payra S, Gautam R, Prakash D, Soni M, Holben B, Bell S (2013) Dust events and their influence on aerosol optical properties over Jaipur in Northwestern India. Environ Monitor Assess 185(9):7327–7342
- Williams PL, Sable DL, Mendez P, Smyth LT (1979) Symptomatic coccidioidomycosis following a severe natural dust storm: an outbreak at the Naval Air Station, Lemoore, Calif. Chest 76(5): 566–570
- WMO World Meteorological Organization (1975) International Cloud Atlas: manual on the observation of clouds and other meteors. WMO-No. 407
- World Bank (2019) Sand and dust storms in the Middle East and North Africa region: sources, costs, and solutions. World Bank. https://doi.org/10.1596/33036
- World Health Organization (2003) Health aspects of air pollution with particulate matter, ozone and nitrogen dioxide: report on a WHO working group, Bonn, Germany 13–15 January 2003. No. EUR/03/5042688. WHO Regional Office for Europe, Copenhagen
- World Health Organization (2006) Air quality guidelines for particulate matter, ozone, nitrogen dioxide and sulfur dioxide: global update. World Health Organization, Geneva. http://www. euro.who.int/__data/assets/pdf_file/0005/78638E90038

- World Health Organization (2014) Burden of disease from ambient air pollution. Glob. Health Obs. Data
- Wu X, Liu J, Wu Y, Wang X, Yu X, Shi J, Bi J, Huang Z, Zhou T, Zhang R (2018) Aerosol optical absorption coefficients at a rural site in Northwest China: the great contribution of dust particles. Atmos Environ 189:145–152
- Zanobetti A, Schwartz J, Samoli E, Gryparis A, Touloumi G, Peacock J, Anderson RH et al (2003) The temporal pattern of respiratory and heart disease mortality in response to air pollution. Environ Health Perspect 111(9):1188–1193
- Zarrin A, Ghaemi H, Azadi M, Mofidi A, Mirzaei E (2011) The effect of the Zagros Mountains on the formation and maintenance of the Iran Anticyclone using RegCM4. Meteorol Atmos Phys 112(3):91–100
- Zhang XY, Gong SL, Zhao TL, Arimoto R, Wang YQ, Zhou ZJ (2003) Sources of Asian dust and role of climate change versus desertification in Asian dust emission. Geophys Res Lett 30(24)
- Zhao TL, Gong SL, Zhang XY, Jaffe DA (2008) Asian dust storm influence on North American ambient PM levels: observational evidence and controlling factors. Atmos Chem Phys 8(10): 2717–2728



Assessment of Heat and Cold Waves Phenomena and Impacts on Environment

Daphne Parliari, Christos Giannaros, and Stavros Keppas

Abstract

Extreme temperatures, either elevated or low, are known to cause adverse impacts on the natural and human environment. Increased hospital admissions and mortality rates during heat and cold waves demonstrate the severity of the problem concerning human health. Heat waves (HWs) are expected to become more severe in terms of frequency, intensity, and duration due to climate change, whereas cold events will become rarer and milder. Along with rapid urbanization and deterioration of air quality due to human activities, heat waves pose a major threat for people's well-being and quality living conditions, affecting especially the urban population. Thus, appropriate counter-measures and mitigation strategies are necessary worldwide in order to combat these effects in timely intervention. The current chapter focuses primarily on heat waves addressing in detail their definition, generation, future projection, and synergy with the urban heat island (UHI) effect and air pollution in cities, as well as their harsh impacts on several sectors and the approaches for mitigating these effects. A part of this chapter is also devoted to cold waves, while a case study concerning the interplay between extreme temperatures, thermal comfort, and air pollution during a significant HW episode that occurred in Greece is presented.

D. Parliari (🖂) · S. Keppas

Laboratory of Atmospheric Physics, School of Physics, Aristotle University of Thessaloniki, Thessaloniki, Greece e-mail: dparliar@physics.auth.gr

C. Giannaros

Frederick Research Center, Nicosia, Cyprus

Institute for Environmental Research and Sustainable Development, National Observatory of Athens, Athens, Greece

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_7

Keywords

Heat waves \cdot Cold waves \cdot Air quality \cdot Human health \cdot Urban heat island \cdot Climate change

7.1 Introduction

From ancient until current times, weather and climate directly affect people's lives on a daily basis. Fringe events particularly, such as floods, droughts, tornadoes, and heat waves, may induce disproportional impact on human societies, as infrastructure can be significantly devastated, economy and health are affected, and loss of life is frequent (World Meteorological Organization-WMO 2016). Even though an extreme episode would normally be as rare as the 10th or 90th percentile of a probability density function estimated from observations (Cubasch et al. 2013), in recent years, natural and climate hazards are becoming more frequent, as well as more severe under the influence of climate change (IPCC 2014). This is particularly true for heat waves, as research findings show a rise in the frequency, intensity, and duration of extreme temperature events (Fischer and Schär 2010), which will also be evident in the future. In contrast, occasional but milder cold wave events are expected to occur in future years (IPCC 2014). HWs can result in increased mortality and health-related risks, agricultural losses, wild fires, and power shortage (WMO 2016; Jain et al. 2021). Rapid urbanization, urban heat island effect, and poor air quality work synergistically with high temperatures prevailing during a heat wave, thus exacerbating the adverse impacts on urban residents (World Health Organization—WHO 2016).

Considering the above, it becomes evident that it is essential to better understand the physical characteristics of HWs and their association with the UHI effect and air pollution in cities. Most importantly, it is urgent to develop counter-measures and strategies that alleviate these combined effects and improve living conditions in urban agglomerations (Voogt 2004), but also integrated heat action plans that can provide guidelines for the management of heat as a hazard (McGregor et al. 2015). The aforementioned exigent necessity is reflected on the present chapter, which primarily aims to present key points on heat waves, including their definition and generation mechanisms, and highlight their severe impacts on environment and human health. Particular attention is given to the synergy between heat waves, air pollution and UHI effect, as well as to the future projections of this detrimental combination, along with their adverse repercussions. Mitigation and adaptation actions are also presented in the current chapter, focusing on heat-health warning systems, implemented by many countries worldwide. Moreover, cold waves and their impacts on environment are investigated. The present chapter attempts to summarize the existing scientific knowledge concerning major environmental problems arising from heat and cold waves, and present a concise overview of the most important aspects, in order to better understand the importance of effective adaptation for the human well-being, especially in urban areas.

7.2 Heat Waves

7.2.1 Defining Heat Waves

Although heat waves are among the most damaging climate extremes, there is no global, standard definition of them (Giannaros 2018). Generally, a HW can be described as a period of abnormally hot weather over a region that usually lasts from few days to few weeks (WMO 2016). However, the standards for determining the extreme warming at a place depend primarily on the local climate features. This fact is reflected on the diversity of the definitions in different countries or regions. The National Meteorological Services (NMSs) around the world have adopted their own standards for issuing HW advisories. These standards mainly involve the application of absolute temperature thresholds for a given number of days. In Greece, for example, a heat wave is defined when a daily maximum temperature greater than 38 °C is recorded for 3 days in sequence (Giannaros 2018). Other approaches for determining a HW may involve the application of different measures of heat and the use of relative exceedance thresholds. For instance, the National Weather Service (NWS) in the United States (US) apply the daytime heat index (incorporating temperature and humidity) to issue early warnings (Barcena-Martin et al. 2019), while the 95th percentile of summer maximum temperature is used in Belgium (WMO 2016). In overall, characterizing quantitatively a heat wave event in terms of meteorological conditions should consider four factors: severity, geographical extent, magnitude, and duration (WMO 2016), while percentile-based thresholds are necessary for the statistical robustness of any HW definition (Kiktev et al. 2003). A recent attempt to integrate these elements into a HW definition was realized by Giannaros et al. (2019), who utilized the IPCC's TX90p index for extreme temperature (Hartmann et al. 2013). Based on this index, a HW day is defined when the daily maximum temperature is higher than the 90th percentile of the reference period that spans from 1961 to 1990. Giannaros et al. (2019) applied the TX90p index to the temperature data over three meteorological stations across Greece and defined a HW episode when at least three successive HW days were identified at the same time over all three stations.

Beyond the meteorological perspective of heat waves, another aspect that cannot be disregarded lies on the impacts of HWs on human health. Matzarakis and Nastos (2011) point out that, although a heat wave is a meteorological event, its assessment should include reference to the impacts on humans. Therefore, the definition and analysis of HWs should always evaluate the human sensation of heat, contributing to the decision-making for the prevention and protection of people. In this direction, determining the links between extreme hot weather conditions and morbidity and mortality is necessary, especially when targeting vulnerable groups of people. Heatrelated epidemiological studies assist the incorporation of the public health risk into HW definition indices (Metzger et al. 2010; Williams et al. 2012) and the development of heat-health warning systems (Michelozzi et al. 2010). Nori-Sarma et al. (2019) indicated the crucial importance of various heat wave definitions versus health effects for the provision of necessary tools to the policymakers, in order to implement effective heat wave warnings.

7.2.2 Generation of Heat Waves

7.2.2.1 Atmospheric Characteristics

Large-scale atmospheric circulation is considered as the factor that primarily affects surface temperature. Heat waves and warm spells have been related to the development of anticyclonic anomalies in various regions across the globe (Meehl and Tebaldi 2004; Maheras et al. 2006; Chen and Lu 2015; Lu and Chen 2016; Luo and Lau 2017). Surface air temperature may increase due to various reasons, such as the adiabatic heating and increasing solar radiation due to subsidence in an anticyclonic regime (Zaitchik et al. 2006; Loikith and Broccoli 2012; Chen and Lu 2015), the horizontal airflow via temperature advection (Ziv et al. 2004) or through the mechanism of katabatic winds (Ágústsson et al. 2007; Fig. 7.1).

Depending on the region, sometimes more than one particular atmospheric circulation patterns can cause a heat wave (e.g., Harpaz et al. 2014), while even regions within a few hundreds of kilometers may be affected by a heat wave under significantly different circulations (e.g., in Iberia; García-Herrera et al. 2005). In addition, in some cases, typical synoptic conditions for a region during summertime may be determinant for the occurrence of a HW. As an example (Harpaz et al. 2014), in southeast Mediterranean Sea, heat waves over the eastern Mediterranean region can be mostly affected and regulated by the mechanism of temperature advection rather than the prevailing subsidence due to dominant northeasterly winds (etesians; e.g., Poupkou et al. 2011; Dafka et al. 2020), which usually blow during summertime.

According to Stefanon et al. (2012), there are six dominant circulation patterns, which can lead to the occurrence of a HW over different regions in Europe. A ridge centered in west Russia can cause heat waves in this region (e.g., Russian heatwave of summer 2010; Grumm 2011). Within this case, a secondary maximum of



Fig. 7.1 Schematic representation of the mechanisms of (a) subsidence in an anticyclone, (b) horizontal warm air advection, and (c) katabatic wind, which may lead to and/or enhance a heat wave

geopotential heights over the eastern Atlantic Ocean affects western Iberia with high temperatures. If a ridge is located over Poland, then a heat wave occurs in regions of Eastern Europe and around the Baltic and Black Sea. In contrast, a region of enhanced geopotential heights over France is associated with HWs in central Europe and Iberia (e.g., August 2003; Fink et al. 2004). In case of an enhanced ridge over the North Sea and British Isles, high temperatures occur over northwest Europe and southern Scandinavia (e.g., summer 1976), while if this ridge is located over Scandinavia, northern Europe is affected. Finally, there is another common circulation pattern, which is a case of a wide, but weak, positive anomaly of geopotential heights over central-east Europe. This pattern has been associated with high temperatures in Iberia and Turkey. Under this pattern, warm air is advected via southerlies caused by the low-pressure system in the Atlantic Ocean.

Although the previously mentioned patterns are fairly typical for the European continent, according to Harpaz et al. (2014) and García-Herrera et al. (2005), there are various circulation patterns that may lead to a HW occurrence in eastern and western Mediterranean Sea. In particular, according to Harpaz et al. (2014), there are three main circulation patterns associated with heat waves in eastern Mediterranean. During the tropical pattern, tropical air mass is transported from south. The subtropical pattern occurs when the subtropical high intensifies from North Africa toward the east Mediterranean. Last, the baroclinic pattern is associated with an enhanced mid-latitude disturbance, while the east Mediterranean is affected by a ridgeinducing subsidence. Moving to the western of the Mediterranean Sea, García-Herrera et al. (2005) suggest that extreme temperatures are usually observed in Iberia when the high-pressure system of Azores enhances and extends further to the west linking with the system of high pressure over west Mediterranean. The example of Iberia is an illustration of the variation of circulation patterns in a limited region that may lead to a heat wave. Although extreme heat may reach Madrid when anticyclone of Azores is further to the west over the Atlantic, Lisbon may experience a HW when the center of the same anticyclonic system is located in northeastern regions.

Hu et al. (2019) detected two circulation patterns that lead to heat waves in south and north side of central Africa. HWs in southern parts have been associated with a positive anomaly in vertical velocities at 500 mb in both the south side of central Africa and northeast Africa/Saudi Arabia regions, which implies an anticyclonic regime there. Both subsidence and warm advection from the north seem to be the mechanisms contributing to a heat wave over the area. In contrast, HWs in northern parts of central Africa are related with the occurrence of a depression in North Africa and central/east Mediterranean causing southerlies, which advect warm air from the Sahara.

Moving to North America, Loikith and Broccoli (2012) found that the warmest 5% of days (for a period of ~50 years) in south-central regions are associated with an extensive region of positive anomalies of geopotential heights at the level of 500 mb over the south-eastern states. Regarding South America, Cerne and Vera (2011) showed that 73% of the heat waves in subtropical South America develop when South Atlantic Convergence Zone (Cerne et al. 2007) is active. A ridge over

Argentine and Chile is associated with an active convergence zone and a HW in the wider area of northern Argentina. An extreme heat episode under similar conditions occurred in February 2003.

In Australia, there are two key atmospheric circulations, each of them relating to a heat wave over different areas (Pezza et al. 2012). In particular, an anticyclone centered at an area between southeast Australia and New Zealand favors a HW in the eastern parts of the continent, while an elongated anticyclone centered over the south Australian coastline allows warm air advection toward the west. In both cases, the anticyclone causes a flow from the inland to peripheral regions.

In eastern Asia, Lu and Chen (2016) distinguished between days with high maximum temperatures (called Extreme Heat—EH) and nights with high minimum temperatures (called Tropical Nights—TN). Each of these types occurs under different synoptic conditions. EH cases occur when a cyclonic system over northeast China allows warm air to be advected from the inland to the coast, while TN cases are present when an anticyclonic system located at a similar position blocks any cyclonic system to pass through, but advects moisture from the eastern maritime area. In addition, Chen and Lu (2015) conducted a more comprehensive study about the heat waves in eastern China highlighting, between others, the impact of westerlies blowing from the mountains to the east (foehn; Speirs et al. 2010; Takane and Kusaka 2011) increasing the temperature further.

Areas of the north and central India located far from the coastline are usually affected by a HW when a ridge at 500 mb is present over northern India and Pakistan. The main mechanism that causes a heat wave in this case is the subsidence and adiabatic heating over the region. According to Ratnam et al. (2016), this type of circulation is associated with an anomalous blocking system in northern Atlantic Ocean resulting in a cyclonic anomaly in northwest Africa. Next, this generates Rossby waves close by the entrance of the African Jet Stream, which maintains the heat wave in India. Another atmospheric pattern that affects mostly eastern parts of this country is linked with an extensive cyclonic system in eastern Asia and western Pacific Ocean.

Finally, focusing on central Asia, heat waves in July may be favored by a positive anomaly over this region at 500 mb. Again, in this massive continental region, the dominant mechanism of the temperature increase is the air descent in the lower troposphere, which leads to clear skies, increased solar radiation, and adiabatic heating. Such a circulation pattern can explain more than 40% of warm spells/heat waves in Eastern Europe and northwest Asia.

Besides the synoptic atmospheric conditions, several regional land-atmosphere interactions and characteristics, such as soil moisture, have a governing role in the generation and development of HWs. Various studies have shown that extreme HWs in summer are preceded by droughts in spring, which are associated with rainfall deficiencies during the winter (e.g., Diffenbaugh et al. 2005; Fischer et al. 2007; Zampieri et al. 2009). The spring soil moisture deficit propagates during the summer strengthening the sensible heat fluxes due to limited evapotranspiration. As a result, heat wave temperatures are enhanced. Moreover, the increased sensible heating assist the growth of the planetary boundary layer (PBL), leading to higher heat

entrainment from above the PBL (Miralles et al. 2014). Furthermore, the development of clouds is inhibited by the dry and hot conditions, resulting in rainfall deficiency and, consequently, in a greater demand for evapotranspiration, which further dries out the soil and, in turn, enhances the temperatures increase (Vautard et al. 2007). Other important factors contributing to the shaping of HWs include the atmospheric water vapor content and anomalies in sea surface temperature (Giannaros 2018).

7.2.3 Climate Change and Heat Waves

There is a significant uncertainty about the characteristics of future climate. One of the reasons is the uncertainty on the human activities in the future. Therefore, there are various scenarios considering different volumes of greenhouse gases emissions until 2100. Four different scenarios, which are called Representative Construction Pathways (RCPs), are used. The RCPs are labeled according to the possible radiative forcing that they represent in the year 2100 (2.6, 4.5, 6, and 8.5 W m⁻² corresponding to 490, 650, 850, and >1370 ppm) (e.g., Moss et al. 2008; Taylor et al. 2012; Jubb et al. 2013; Rogelj et al. 2012).

In general, climate change is expressed by increase in temperature around the most regions of the world. It seems that more frequent extreme events, such as heat waves, will be part of this change in the future. In a "business as usual" scenario (RCP8.5) until 2100, 3-day HWs are anticipated to be significantly strengthened in intensity (by 3 °C in daily minimum temperature), duration (by more than 5 days), and frequency (by 25–30%) in parts of north America and Europe (Meehl and Tebaldi 2004). Such a change of the climatology of summer may impact on people losses, failure of crops, and significantly decreased discharge in many rivers (Beniston 2004).

The 2003 HW in Europe may provide an insight of the course of summers in the late twenty-first century (Beniston 2004). Summer of 2003 had been the hottest since 1540. This conclusion comes up not only from the meteorological observations (e.g., the average maximum temperature of the summer 2003 in Basel of Switzerland exhibited a positive anomaly of 6 °C), but also from the noticeable changes in rivers and crops in Europe by the end of that summer. A summer in Europe in the latter period of the current century is expected to present increased average of maximum temperatures by 4–6 °C comparing to the present, with 40–60 more days of temperatures >30 °C in an annual basis and the occurrence of 3–4 heat waves per year on average based on the RCP8.5 (Beniston 2004). Under the RCP4.5 scenario, the impact of the climate change on heat waves is estimated to be roughly halved (Lhotka et al. 2018).

Looking at the heat waves in a global scale, it seems that they will increase in frequency, duration, and intensity. Under the "business as usual" scenario and the trend of increasing temperature, extreme events will increase in frequency and heat waves, which are considered as extreme today (e.g., extreme heat waves in USA (1980; Livezey 1980; Applegate et al. 1981; Karl and Quayle 1981; Jones et al.

1982), Europe (2003; Beniston 2004; Stott et al. 2004; Fouillet et al. 2006), and Russia (2010; Dole et al. 2011; Otto et al. 2012)), will be the normality in the future. It is noticeable though that the frequency of HWs will rise in a much more significant rate during the later decades of the twenty-first century (Russo et al. 2014). According to the IPCC report (Stocker et al. 2013), under high-emissions scenarios the current rate of 1 extreme heat wave event per 20 years will increase in case to 1 event per 2 years by the end of the twenty-first century. Additionally, it seems that the maximum increase of $T_{\rm max}$ of an extreme event will occur in central Europe and northern America exhibiting, by the end of the century, values 7–9 °C larger than the $T_{\rm max}$ of an extreme heat wave at the present.

7.2.4 Urban-Scale Aspects of Heat Waves

The urban micrometeorology is significantly complicated, as the presence of buildings of various heights and different densities, various structure materials, and green areas locally differentiate the near-surface meteorological conditions. An urban area usually presents higher temperatures than the surrounding areas due to the materials that the buildings are made of and the anthropogenic heat (Oke 1967; Kim 1992; Shahmohamadi and Etessam 2011). An extra factor that is involved in the microclimate of a city is the pollution produced by it. Both the complicated structure and the pollution can add extra charge under a heat wave condition.

7.2.4.1 Heat Waves and Urban Heat Island Effect

The adverse impacts of HWs are more evident in cities due to the urban heat island (UHI) phenomenon and the greater population density. UHI denotes the higher temperatures observed in urban environments compared to their rural surroundings (Giannaros et al. 2018). Today, more that 50% of the world's population is living in urban areas, while this proportion is expected to reach up to 68% by 2050, with the most urbanized regions being the Northern America, Latin America and the Caribbean, Europe and Oceania (United Nations 2018). The combined effects of climate change, fast urban development, and UHIs induce an important health concern for the urban residents.

UHI effect is considered to be the most illustrative environmental problem emerging from local climate changes induced by humans (Giannaros et al. 2018). Its causation is primarily associated with changes in the surface characteristics due to replacement of the natural environment by artificial materials (e.g., asphalt and concrete). These changes alter the surface energy budget (SEB) due to the artificial facets having high heat capacity storage and the low vegetation coverage resulting in limited evaporative cooling. SEB is also modified by the enhanced production of anthropogenic heat (Founda and Santamouris 2017), while the complicated geometry of streets of buildings introduce modifications in the near-surface air flow (Laaidi et al. 2012).

UHI is a local phenomenon demonstrating different characteristics depending on time and place (Fig. 7.2) (Giannaros and Melas 2012) found that UHI in



Fig. 7.2 U.S. Environmental Protection Agency (2019). Illustration of the impact of different land uses to the air and surface temperatures of a city. www.usgs.gov/media/images/urban-heat-islands

Thessaloniki, Greece, is primarily a nighttime effect, with maximum values ranging between 2–4 °C and 1–3 °C during the warm and cold period of the year, respectively. The city of Athens in Greece is also characterized by a strong UHI effect, with intensity up to 4 °C during the night (Giannaros 2013). The summertime UHI intensity in Beijing, China, ranges from 4 to 6 °C between the urban core of the city and the suburbs and from 8 to 10 °C between the city center and outer suburban areas. In Johannesburg, South Africa, the temperature in the city center was found to be higher than 11 °C compared to the suburban valleys during the dry season. The London UHI is primarily evident during the night hours, with its intensity being equal to around 0.7–0.8 °C at noon and rising to 2.5–3.0 °C after sunset. In Melbourne, Australia, UHI intensity can range from a mean of 2–4 °C, with daily peaks as high as 7 °C (Tzavali et al. 2015).

Extreme heat in urban areas is intensified by the interaction of UHIs and heat waves. Founda and Santamouris (2017) investigated five heat waves events in Athens, Greece, during the summer of 2012. They showed that the mean UHI intensity increased by up to 3.5 °C during the studied HWs compared to the summertime normal conditions. Furthermore, Zhao et al. (2018) reported higher UHI intensities (0.4–2.8 K) during HWs than during normal days, after investigating interactions between UHIs and HWs for 50 cities in the United States. A case study conducted for Szeged, Hungary, showed that nocturnal UHI was 2–3 times stronger in the HW period compared to normal summer periods (Unger et al. 2020), while

positive synergy between UHI and HW events were also reported for Washington in US (Li and Bou-Zeid 2013).

The elevated temperatures resulting from interaction between UHIs and HWs have been linked to increased mortality rates. Tan et al. (2010) showed that UHI in Shanghai, China, enhances the intensity of heat waves. In turn, the increased exposure to extreme heat results in adverse effects on the human health of urban residents. More specifically, the heat-related excess mortality was significantly higher in the urban areas of the city compared to the surrounding suburbs. This outcome was particularly evident for the two extreme heat waves in 1998 and 2003 (Tan et al. 2010). Moreover, Heaviside et al. (2016) concluded that UHI in the West Midlands in the United Kingdom (UK) contributed 50% approximately to the total heat-related mortality during the 2003 European HW. Laaidi et al. (2012) also studied the heat wave that occurred in August 2003 in Europe by assessing the impact of UHI on mortality in Paris. They highlighted that the exposure to high nighttime temperatures over several days during the HW increased the probability of death for elderly people living in urban areas.

7.2.4.2 Heat Waves and Air Quality

Air pollution, both ambient (outdoor) and household (indoor), has become one of the major concerns affecting life quality in modern societies (Saxena and Srivastava 2020; Saxena and Sonwani 2019). Especially, outdoor air pollution is considered the biggest environmental risk to health, as it is responsible for around three million deaths each year, mainly from non-communicable diseases (WHO 2016). It is characteristic that for 2017, accumulative air pollution (indoor and outdoor) was the fourth most important risk factor for the total annual number of deaths globally, as illustrated in Fig. 7.3.

Pollutants observed in urban agglomerations are in higher concentrations than in rural areas due to the emissions caused by human activities (Williams et al. 2008; Kalisa et al. 2018; Saxena and Naik 2018). During heat waves, the air masses remain stagnated, trapping pollutants at layers of the troposphere near the surface (Tressol et al. 2008). In addition, tall buildings in cities may trap pollutants as they stop/ decelerate air movement. Several studies have shown that the already enhanced concentrations of NO₂, O₃, and PM₁₀ within urban areas increase under high temperature, increased solar radiation, and the absence of significant winds. O_3 is a secondary pollutant in the cities, which means that it is not directly emitted, but is formed on hot days by the influence of solar radiation and the presence of NO_x (Yin et al. 2019; Saxena et al. 2019; Archibald et al. 2020). A recent study conducted in Birmingham, UK, showed that the levels of O_3 increase by more than 50% under elevated temperatures during heat waves (Kalisa et al. 2018). Pollutants such as NO₂ and PM_{10} are produced in cities by vehicle emissions and road transport-related emissions and other human activities. However, secondary formation is again favored by increased temperatures (e.g., Vardoulakis and Kassomenos 2008; Papanastasiou et al. 2015). Particularly for NO₂, its concentration increases by 120% for temperatures higher than 35 °C. Also, it is worth mentioning that emissions of pollutants on hot days may be further increased by behavioral changes (e.g., enhanced air-conditioning and use of cars).



Fig. 7.3 Total global annual number of deaths by risk factor, measured across all age groups and both sexes (Data published by Global Burden of Disease Collaborative Network. Global Burden of Disease Study 2017 (GBD 2017) Results. Seattle, United States: Institute for Health Metrics and Evaluation (IHME), 2018. http://ghdx.healthdata.org/gbd-results-tool)

Considering the increasing concern about the health impacts induced by the climate change, especially in cities, the study of the synergistic effects of air pollution and extreme heat conditions on health is of great importance. An early study by Katsouyanni et al. (1993), which focused on the deadly heat wave of July 1987 in Greece, showed that the greater mortality recorded in Athens compared to other urban and rural areas of the country was probably associated with the increased air pollution that characterizes the city. Concerning the more recent heat wave of 2003 in Europe, Filleul et al. (2006) demonstrated that the excess risk of deaths linked to the combined effects of ozone and temperature ranged in France from 10.6 to 174.7%, in Le Havre and Paris, respectively. This outcome highlights the enhanced risk in highly urbanized areas during heat waves due to the synergy between poor air quality and elevated temperatures. Further, Analitis et al. (2014) investigated the relationship among daily mortality, meteorological, and air quality data from 1990 to 2004 in nine European cities. They found a 54% increase in mortality for people aged 75-84 years during HW episodes characterized by high ozone concentrations, compared to low ozone days. Similar findings were demonstrated during the HW episodes with high PM10 days, when daily deaths were increased 106% in the 85+ year age group. Chen et al. (2018) studied the impact of urban air pollutants and temperatures on the total and cardiovascular mortality in eight European cities and showed overall positive feedbacks between air pollution and mortality, which were generally stronger during days characterized by high temperatures. In particular, an increase of 10,000 particles/cm³ in particle number concentration during days with elevated temperatures corresponded to a 2.51% increase in cardiovascular mortality. Similarly, Shaposhnikov et al. (2014) suggest that at temperature ≤ 18 °C mortality rate is +0.43% increasing to +1.44% at the temperature near 30 °C in the city of Moscow.

7.2.5 Impacts of Heat Waves and Mitigation Strategies

Heat waves affect severely the natural and human environment, including human health, energy demand, infrastructure, and forests (Zuo et al. 2015).

7.2.5.1 Human Health

HWs deteriorate the human thermal comfort conditions. This fact may result in a significant increase in morbidity and mortality because of diseases associated with the cardiovascular and respiratory systems (Giannaros 2018). Several studies across the world have clearly connected elevated temperatures with excess in mortality rates. A characteristic example is the severe heat wave of 2003 in Europe, when over 70,000 people died (Robine et al. 2008). Concerning non-fatal health effects of heat waves, Semenza et al. (1999) showed that during the July 1995 HW in Chicago, hospital admissions were higher by 11% than average and 35% more than expected for patients aged 65+ years. Dehydration, heat stroke, and heat exhaustion were among the major treatments (59%), especially for people with underlying medical conditions. Moreover, according to Theoharatos et al. (2010) more than 140 emergency department visits related to heat exhaustion and heatstroke were reported by the National Health's Operational Center in Attica, Greece, during June 2007 HW that affected severely the capital of Greece.

According to Giannaros et al. (2018), enhanced susceptibility to heat-health impacts is evident for particular vulnerable groups, such as children, elderly, and people having pre-existing cardiovascular or respiratory diseases or other medical conditions (e.g., diabetes). Urban residents are also susceptible to heat-related health issues. Several studies highlight that death rates during heat waves are often much higher in cities encumbered to the rural areas (Conti et al. 2005; Gabriel and Endlicher 2011; Heaviside et al. 2016; Matzarakis et al. 2009). This is because of the interplay between HWs and the urban heat island effect, as already analyzed in the previous Sect. 7.2.4, exacerbating the thermal load on people living in urban areas. Especially, the nighttime UHI leading to increased nocturnal temperatures in the urban cores has a critical damaging effect on human health, as the urban residents are deprived of the sense of thermal relief (Giannaros 2013).

7.2.5.2 Energy Sector and Infrastructure

Higher energy demand is induced during HW events due to the increased need for cooling. This fact can lead to power shortages and even outages, while it also results in a greater release of greenhouse gases, since the major sources of the electricity production are the fossil-fueled and nuclear power plants. This environmental side effect, apart from leading to the deterioration of air quality and to consequent worsening of the associated health problems, has a positive interaction with the elevated temperatures that introduces further increase for energy consumption in order to meet the cooling needs. High temperatures during a HW can also have significant impacts on the energy generation, transmission, and distribution (Giannaros 2018).

Given the complexity of urban infrastructure and the interaction between its elements, increased temperatures especially during prolonged periods of time cause problems and malfunctions (e.g., damages on rail tracks and roads). Additionally, extremely high temperatures may affect significantly artificial materials in buildings. Some characteristic examples are the expansion stress (e.g., in concrete), issues with protective finishes, stress in large sections of insulated glass units, and damage to durability of coatings due to ultraviolet radiation (Zuo et al. 2015). According to the Institute for Sustainable Resources (2010), HWs may also induce major effects on the electricity infrastructure, while air transportation and telecommunications are less vulnerable to extreme heat.

7.2.5.3 Other Aspects

According to Unal et al. (2013) and Westerling and Cayan (2006) HWs are highly associated with wildfires. In particular, during dry heat waves, the significant declination of relative humidity increases the development probability of disastrous wildfires (Pezza et al. 2012), like the devastating bushfires in Australia that burned more than 12.35 million acres of land and caused 25 deaths in 2019. Water services are also significantly affected by heat waves, as water usage is enhanced during HW periods, while the water infrastructure problems may arise due to extreme heat (Zuo et al. 2015). Moreover, the elevated air temperatures contribute to the deterioration of water quality. Eutrophication, increased concentrations of major elements, and warmer temperatures in water are evident during HW events (van Vliet and Zwolsman 2008). Agricultural and livestock productivity are also decreased in the course of heat waves (Smoyer-tomic et al. 2003). Potential impacts of HWs also affect the natural environment, forestry agriculture, and tourism (McGregor et al. 2007). Last, but not least, HWs affect the mental health of humans. The heat stress induced by extreme high temperatures exacerbates pre-existing mental disorders and affects sleeping leading to fatigue and lack of concentration (Zuo et al. 2015), while there is evidence that hot weather is associated with crime, aggressive behavior, and domestic violence (McGregor et al. 2007).

7.2.5.4 Mitigation Strategies

Due to the growing severity of the elevated temperatures and the consequent adverse impacts on human health and environment, several strategies and measures have been established in order to attenuate these effects. Nevertheless, proposed solutions should take into consideration that heat waves act synergistically with UHIs, air quality, rapid urbanization, and the growing percentage of elderly people. People living in urban environments are exposed to higher risks associated with heat waves, as UHI effect makes the cities warmer than their surrounding non-urban areas.

According to Akbari and Kolokotsa (2016), the mitigation technologies to counteract the urban warming effects follow major approaches. The first approach aims at increasing solar reflectance (e.g., cool roofs and pavements). It contains measures aiming to keep urban surfaces cool by decreasing the absorption of the solar radiation with the use of materials characterized by high solar reflectance and high thermal emittance. In this framework, Morini et al. (2016) investigated the effect of albedo change in urban temperatures for the area of Terni, Italy, Their results demonstrated that when the albedo was increased from 0.2 to 0.8 for roofs, walls, and roads in the whole urban area, a decrease of the daytime and night-time urban temperature by up to 2.5 °C followed. The second approach aims at increasing evapotranspiration, using technologies that favor urban greenery and water availability (e.g., green walls). For instance, Li (2014) highlighted that green roofs had a substantial effect in reducing the UHI intensity during a HW episode occurred in Baltimore-Washington in June 2008. Other urban heat mitigation actions may include improved ventilation in urban areas by assisting cold air drainage flows, especially at night (Gabriel and Endlicher 2011).

A comprehensive study concerning the effectiveness of heat mitigation measures was presented by Silva et al. (2010). They investigated the impacts of increasing the overall emissivity, percentage of vegetated area, thermal conductivity, and albedo in the urban area of Phoenix, Arizona, USA, to human health vulnerability associated with extreme heat-related events from 2002 to 2006. According to their findings, the albedo strategy had the strongest influence on temperature followed by increasing thermal conductivity and vegetated area, whereas increasing emissivity was the least effective. The most important outcome of their study was that the combination of the studied mitigation approaches led to a 48% reduction in annual heat-related emergency service calls.

Another way to categorize extreme heat mitigation measures is according to their time frame. Short-term measures include actions in a more immediate scale (e.g., heat-health warning advisory), while long-term strategies require more time to be introduced and implemented (e.g., construction of green parks). Heat-Health Warning Systems (HHWSs) are extremely important for the short-term mitigation measures. Since the severe heat wave that stroke Europe in 2003, causing more than 70,000 deaths, interest on HHWSs has raised significantly and most of concerned countries have established a type of heat-health action plan to protect their population. A HHWS aims at reducing heat-related impacts on human health, especially during hot conditions, by combining meteorological forecasts and public health actions (Koppe et al. 2004). Through HHWSs, decision-makers and the general public can be warned about extreme hot weather and obtain advice on how to avoid associated adverse health outcomes. Although HHWS can vary significantly as they are based on area-specific information depending on local

populations, systems, human and technical resources and heat–health associations, there are some universal characteristics for such application. Among these is the consideration of local meteorology, demographics and urban structure, thresholds associated with heat-health outcomes and nomenclature that is easily understood by the public, local stakeholders, and decision-makers (McGregor et al. 2015).

As there is no global definition of HWs (see Sect. 7.2.1), there is also a variety of methods to define the weather-health relation which acts as "action trigger" for issuance of warnings. According to McGregor et al. (2015), current HHWSs incorporate three broad categories of heat situations that affect human health. The most common of them is the single-or few-parameter method, where the exceedance of one or more variables above a certain threshold triggers a respective warning. Different countries use different metrics and methods to achieve the most effective result on national or regional levels. For example, Belgium, France, Greece, and UK use Temperature, Australia and Italy apply Apparent Temperature, while Humidex is implemented by USA and Airmass by Italy and China. In any case, the first step for building a NHWSS is selecting the proxies which best depict heat-health relation and setting the thresholds that are either correlated to the consequences or depend on climatological fringe values. Then, thresholds initiate a system of scaled warnings that are often related to specific recommendations. The dissemination of the alerts addresses both general public and vulnerable groups, informing them about the severity of the event and providing advice on health impacts (Casanueva et al. 2019).

Italy was one of the first countries to establish HHWS that is covering the vast majority of urban agglomerations and it is operational for almost 17 years. The Italian system combines mortality data and meteorological variables to detect conditions connected to an increase in daily mortality. The heat-health-related risk is classified from low (<20% excess in mortality) to high (\geq 20% excess in mortality). This HHWS is based on two pillars: the air mass-based approach, and the relationship between mortality and maximum Apparent Temperature. A series of conditions, based on the two approaches mentioned earlier, must be combined interactively in order for Level 1 and Level 2 warnings to be issued while Level 3 warning corresponds to three continuous days of Level 2 (Michelozzi et al. 2010).

A recent example of HHWS is implemented in the framework of the project LIFE ASTI "Implementation of a forecasting system for urban heat island effect for the development of urban adaptation strategies" (https://lifeasti.eu/). Based on the approach of the Italian system, urban-specific relationships between maximum apparent temperature and daily mortality of vulnerable population groups were extracted for the involved cities (Thessaloniki and Rome). In the second stage, HHWS was developed and implemented to issue differential alerts at high spatial and temporal resolution during the warm period of the year. The results from this effort will conclude to the improvement of the decision-making of local policy-making authorities in the integration of heat prevention plans.

Last, but not least, improved education and public awareness also play an important role when adapting to heat, especially regarding sensitive groups. According to the US Environmental Protection Agency, local policy-making authorities should disseminate information about conceivable risk factors, symptoms

of overdone heat exposure, and recommend advice on remedy. Particular attention must be given to vulnerable groups, namely infants, outdoor workers, pregnant women, elderly people, as they must be specifically taken into consideration in information campaigns utilizing more active approaches, such as a tandem system, home visits, and regular phone calls (Matthies et al. 2008).

7.3 Cold Waves

According to WMO (2016), a cold wave is defined as "unusual cold weather characterized by significant temperature decrease that remains below certain thresholds for at least two consecutive days during the cold season". A cold wave is usually linked to an incursion of very cold air that is displaced to lower latitudes, or it is enforced by long radiative cooling under clear sky conditions (WMO 2016).

Cold-related mortality is a public health concern that affects many areas of the world and cannot be ignored. A study by Analitis et al. (2008) on the impact of low temperatures on mortality using data from 1990 to 2000 for 15 European cities revealed increase of the daily amount of deaths from natural causes with 1 °C decrease in temperature. The respective increase for cardiovascular, respiratory, and cerebrovascular deaths was found to be 1.25% (Analitis et al. 2008). In a different aspect of the association of cold weather and its impacts on human health, Hajat et al. (2004) associated low temperatures and an increase in consultations about lower respiratory tract infections made by elderly people (65+ years) over a 10-year period, 1992–2001, for 16 selected locations. Furthermore, a study conducted by Keatinge and Donaldson (1997) linked winter temperatures with mortality resulting from ischemic heart, cerebrovascular, and respiratory disease. In particular, increases in all types of mortality were depicted with 1 °C fall in temperature. Fu et al. (2018) also investigated the mortality attributed to cold ambient temperatures in India using national data for 2001–2013 and proved that medium low temperatures showed higher attributable risk than extremely low.

As regards connection of winter-related mortality and air pollution, several studies have been published. Low temperatures significantly enhanced the acute mortality effects of ozone in Suzhou, China. According to Chen et al. (2013), the connection of ozone and daily mortality in winter was much stronger than during summer, both for total and for cardiovascular mortality. Interestingly, O_3 was more distinctly associated with cardiovascular mortality than total mortality.

Results published by Cheng and Kan (2012) revealed significant association between PM10 and extreme low temperatures, not only for non-accidental mortality but for cause-specific also. On cold days, an increase of $10-\mu g/m^3$ in PM10 corresponded to a 0.40% increase in total mortality, while similar behavior was demonstrated by O₃. As indicated in Peterson et al. (2014), pollutants can be ensnared by inversions related to cold air near the surface and warmer air in higher altitude. Therefore, it is evident that low temperatures are directly linked to increased air pollution and compromised air quality, consequently. In a wider context, cold spells can induce severe impact on several aspects of environment and life. In a research conducted by Morignat et al. (2018) it was found that prolonged exposure to cold temperatures showed a 23% increase in female cattle mortality, and the mortality risk increased with the duration and the severity of the exposure.

Añel et al. (2017) investigated a possible connection between cold waves and power production and consumption. The energy network is exceptionally stressed and power blackouts are very likely to happen during winter; the accumulation of snow, freezing rain, and ice can cause significant damage to airlines, control towers, and power transmission lines.

In the cold wave that stroke the United States and Canada in January 1998 (Chang et al. 2007), 120,000 km of power lines were destroyed and four million people experienced days with no heating and electricity. 100,000 inhabitants were forced to evacuate their residences, several were injured, and 23 lost their lives. Bridges, highways, railroads, train services, and subway systems were completely destroyed. People living in the affected areas had to boil water for up to 48 h as the water supply was damaged; dairy farms, maple syrup producers, and millions of trees were ruined. The total economic cost reached US\$1 billion in the United States and US\$1.5 billion in Canada (Jendrizky 1998).

Future projections of cold waves show a negative trend. For example, in Europe, current cold extremes tend to gradually disappear. As a consequence of the diminishing occurrence of moderate cold waves in many parts of the world, the number of attributed fatalities will decline strongly. Feyen et al. (2020) demonstrated that European population exposed annually to extreme cold is expected to lower from ten million in the baseline scenario to five million at 1.5 °C global warming scenario. Similarly, Forzieri et al. (2017) showed that European people exposed to cold waves and deaths because of cold waves will rapidly decrease from the reference period to 2011–2040 (-6% and -16% respectively), 2041–2070 (-64% and -76% respectively), and 2071–2100 (-97% and -97% respectively).

7.4 Case Study of Heat Wave

Air quality is directly affected by elevated temperatures during a heat wave period, as discussed in detail in Sect. 7.2.4.2. The current study presents a characteristic case for the city of Thessaloniki, Greece, located in South-East (SE) Mediterranean. In the summer of 2007, the SE Mediterranean and Balkans were affected by three heat waves, during which record-breaking temperatures were recorded in several weather stations in Greece (Founda and Giannakopoulos 2009). The present case study focuses on the July 2007 event, investigating the relationship between air temperature and pollutants. The thermal comfort conditions during the examined episode are also assessed using an appropriate index.

The study area is the city of Thessaloniki in northern Greece, the second largest city of the country accommodating 800,000 inhabitants approximately. It is built along the north-east coast of the Thermaikos Gulf and as a result, its climate is



Fig. 7.4 Topography of Thessaloniki with the locations of the weather stations and identification of the main geomorphological features. (Adopted from Giannaros 2018)

directly affected by the sea. The flat area of the city in the West is dominated by the industrial zone of Sindos, whereas Hortiatis Mountain (1200 m) is located in the East. Thessaloniki's climate can be described as Mediterranean with hot and dry summers and mild and wet winters.

The city of Thessaloniki was affected by a severe HW from July 22 to July 25, 2007, according to Giannaros et al. (2018). The approach he followed for identifying this heat wave was based on the extreme temperature index, TX90p, provided by IPCC (Hartmann et al. 2013). The observational data for the study period were collected from five ground-based weather stations (Fig. 7.4) operated by the Region of Central Macedonia (RCM). In particular, time series of air temperature and relative humidity, wind speed and direction, and pollutants (O₃, NO₂, PM₁₀ according to availability) were obtained for the period 20–28 July 2007.

Based on the temperature and humidity measurements, Thom's Discomfort Index (DI; Thom 1959) was computed in order to assess the thermal stress during the examined heat wave. Table 7.1 presents the DI classes and the corresponding effects on human thermal sensation.

Figure 7.5 shows the temporal evolution of temperature, air pollutants, and DI from July 20 00Z to July 28 00Z. A gradient rise of temperature is evident from July 22 (the starting day of the HW), while a significant decrease in temperature is observed after the 26th of July. The maximum daily values of temperatures were recorded on July 24 and 25, exceeding and reaching up to 42 °C in the highly (AUTH, Kordelio, Kalamaria; Fig. 7.5a–c) and semi-urbanized (Panorama;

Discomfort conditions
Disconfort conditions
No discomfort
Less than half of the population feels discomfort
More than half of the population feels discomfort
Most of the population feels discomfort
Everyone feels severe stress
State of medical emergency

Table 7.1 DI classes and the corresponding thermal discomfort conditions



Fig. 7.5 Temporal evolution of air temperature, pollutants, and DI from July 20 00Z to July 28 00Z over (a) AUTH, (b) Kordelio, (c) Kalamaria, (d) Panorama, and (e) Sindos in Thessaloniki, Greece

Fig. 7.5d) areas of the city, respectively. The excess warmth during these days led to maximum DI values that ranged from 29 to 32 °C in AUTH, Kordelio, and Kalamaria (Fig. 7.5a–c). Thus, all population in the urban core of the city experienced severe heat stress. In Panorama (Fig. 7.5d), DI was also increased in the course of the HW, maintaining, however, its levels below 29 °C due to the low relative humidity values in the area (not shown). Interestingly, although the lowest daily maximum temperature among the examined sites was recorded in the semi-rural station of Sindos, the highest DI maximum value was observed in this location (Fig. 7.5e). This could be attributed to the fact that Sindos area is frequently exposed to high relative humidity values. Especially during the examined episode, the analysis of the wind field in Sindos (not shown) demonstrated south winds from the sea, which were amplified by 2–3 m/s in the afternoon hours, thus, possibly carrying high levels of humidity that pushed DI values higher than other areas.



Fig. 7.6 Air temperature against O_3 , NO_2 , and PM10 concentration over Kordelio (**a**) and Sindos (**b**) during the examined HW episode. Rho values indicate the Spearman correlation coefficient with statistical significance (*p* value) being symbolized as ****p* (<0.001; 99.9% confidence interval)

Nevertheless, during the night hours, DI values in Sindos decrease significantly (≤ 24 °C) revealing that the local population experiences a vital heat relief, in contrast to the urban residents, especially in AUTH and Kordelio. This fact highlights the importance of the UHI effect and its role in deteriorating the human thermal comfort conditions.

Concerning air quality, higher concentration of NO₂ is evident in the urban sites (AUTh, Kordelio; Fig. 7.5a, b) due to emissions from traffic congestion, especially in the morning rush hours. The highest values of PM₁₀ are also observed in areas with enhanced human activities, including the city's urban core (Kordelio, Kalamaria; Fig. 7.5b, c) and suburbs (Panorama; Fig. 7.5d). O₃ concentrations follow the temperature's daily cycle in both urban (Kordelio; Fig. 7.5b) and semirural (Sindos; Fig. 7.5e) areas of the city, with maximum O₃ coinciding with the peaks in the temperature, as O_3 formation is favored by heat and sunlight. The two variables are strongly correlated, as the values (0.86 and 0.89; statistically significant at the 99.9% confidence interval) of the nonparametric Spearman coefficient (rho) indicate in Kordelio and Sindos (Fig. 7.6). PM10 concentrations also increase as the temperature elevates during the examined HW period. This is due to the formation of secondary aerosols induced by the warmer weather conditions. Figure 7.6 demonstrates a weak positive correlation (statistically significant at the 99.9% confidence interval) between PM10 and air temperature, with Spearman coefficient varying from 0.23 (Sindos; Fig. 7.6b) to 0.25 (Kordelio; Fig. 7.6a). On the contrary, the Spearman coefficient values when NO2 and temperature are examined reveal a negative correlation (statistically significant at the 99.9% confidence interval) between the two variables, which is more profound in the urban site (Kordelio; rho = -0.47; Fig. 7.6a). This finding indicates that NO₂ promotes the formation of O₃ during the mid-day, when the highest temperatures during the HW were observed.

The above outcomes are in satisfactory agreement with previous studies (e.g., Theoharatos et al. 2010; Kalisa et al. 2018), which reported that high levels of pollutants were much more prevalent during prolonged heat waves. Such findings are important to the establishment of long-term prevention measures in HW emergency plans and urban planning, considering the human thermal comfort and the synergistic impacts of extreme heat conditions and poor air quality.

7.5 Conclusions

Heat waves pose a major threat to people's well-being. Their constantly increasing frequency and duration due to climate change and their synergy with air pollution and UHI effect cause adverse impacts on environment and human health. Urban populations in areas most in danger, as in the South-East Mediterranean, are already experiencing the unpleasant consequences concerning thermal comfort conditions and the overall view of their everyday performance. Heat-related mortality is under thorough investigation in recent years in order to define ways to understand and analyze the effect, as the scientific community worldwide is eager to assist policy-making authorities who wish to protect and provide guidelines to the general public and vulnerable groups. Heat-health warning systems are a valuable tool for this endeavor. Further effort should be continuously put to enrich and sophisticate HHWS through high-quality and accurate scientific services and products.

Cold waves can also pose threat to human health and environment as low temperatures are related to increased mortality and damage in infrastructure. Nevertheless, as global temperature is rising, number and duration of cold spells are expecting to decrease even in the colder parts of the planet. Therefore, with no intention to underestimate their importance, attention has been moved to the investigation of heat wave events and their impact worldwide. Extended research and collaboration with key policy-making entities is essential for the provision of suitable guidelines, advisory, and overall protection of urban populations and vulnerable groups of citizens.

References

- Ágústsson H, Cuxart J, Mira A, Ólafsson H (2007) Observations and simulation of katabatic flows during a heatwave in Iceland. Meteorol Z 16(1):99–110
- Akbari H, Kolokotsa D (2016) Three decades of urban heat islands and mitigation technologies research. Energy Buildings 133:834–842. https://doi.org/10.1016/J.ENBUILD.2016.09.067. Elsevier
- Analitis A et al (2008) Effects of cold weather on mortality: results from 15 European cities within the PHEWE project. Am J Epidemiol 168(12):1397–1408. https://doi.org/10.1093/aje/kwn266
- Analitis A et al (2014) Effects of heat waves on mortality: effect modification and confounding by air pollutants. Epidemiology 25(1):15–22. https://doi.org/10.1097/EDE.0b013e31828ac01b

Añel JA et al (2017) Impact of cold waves and heat waves on the energy production sector. Atmosphere 8(11):1–13. https://doi.org/10.3390/atmos8110209

- Applegate WB et al (1981) Analysis of the 1980 heat wave in Memphis. J Am Geriatr Soc 29 (8):337–342
- Archibald AT, Neu JL, Elshorbany YF, Cooper OR, Young PJ, Akiyoshi H, Cox RA, Coyle M, Derwent RG, Deushi M, Finco A (2020) Tropospheric ozone assessment report: a critical review of changes in the tropospheric ozone burden and budget from 1850 to 2100. Elem Sci Anth 8 (1):1–34
- Barcena-Martin E, Molina J, Ruiz-Sinoga JD (2019) Issues and challenges in defining a heat wave: a Mediterranean case study. Int J Climatol 39(1):331–342. https://doi.org/10.1002/joc.5809
- Beniston M (2004) The 2003 heat wave in Europe: a shape of things to come? An analysis based on Swiss climatological data and model simulations. Geophys Res Lett 31(2)
- Casanueva A et al (2019) Overview of existing heat-health warning systems in Europe. Int J Environ Res Public Health 16(15):2657. https://doi.org/10.3390/ijerph16152657
- Cerne SB, Vera CS (2011) Influence of the intraseasonal variability on heat waves in subtropical South America. Clim Dyn 36(11–12):2265–2277
- Cerne SB, Vera CS, Liebmann B (2007) The nature of a heat wave in eastern Argentina occurring during SALLJEX. Mon Weather Rev 135(3):1165–1174
- Chang SE et al (2007) Infrastructure failure interdependencies in extreme events: power outage consequences in the 1998 Ice Storm. Nat Hazards 41(2):337–358. https://doi.org/10.1007/s11069-006-9039-4
- Chen R, Lu R (2015) Comparisons of the circulation anomalies associated with extreme heat in different regions of eastern China. J Clim 28(14):5830–5844
- Chen K et al (2013) Influence of temperature to the short-term effects of various ozone metrics on daily mortality in Suzhou, China. Atmos Environ 79:119–128. https://doi.org/10.1016/j. atmosenv.2013.06.004. Elsevier Ltd
- Chen K et al (2018) Two-way effect modifications of air pollution and air temperature on total natural and cardiovascular mortality in eight European urban areas. Environ Int 116:186–196. https://doi.org/10.1016/j.envint.2018.04.021
- Cheng Y, Kan H (2012) Effect of the interaction between outdoor air pollution and extreme temperature on daily mortality in Shanghai, China. J Epidemiol 22(1):28–36. https://doi.org/ 10.2188/jea.JE20110049
- Conti S et al (2005) Epidemiologic study of mortality during the Summer 2003 heat wave in Italy. Environ Res 98(3):390–399. https://doi.org/10.1016/j.envres.2004.10.009
- Cubasch U et al (2013) Introduction. In: Climate change 2013: the physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, pp xix–xx. https://doi.org/10.1016/B978-0-12-809665-9.15009-8
- Dafka S, Akritidis D, Zanis P, Pozzer A, Xoplaki E, Luterbacher J, Zerefos C (2020) On the link between the Etesian winds, tropopause folds and tropospheric ozone over the Eastern Mediterranean during summer. Atmos Res 248:105161
- Diffenbaugh NS et al (2005) Fine-scale processes regulate the response of extreme events to global climate change. Proc Natl Acad Sci U S A 102(44):15774–15778. https://doi.org/10.1073/pnas. 0506042102
- Dole R et al (2011) Was there a basis for anticipating the 2010 Russian heat wave? Geophys Res Lett 38:6
- Feyen L et al (2020) Global warming and human impacts of heat and cold extremes in the EU. https://doi.org/10.2760/47878
- Filleul L et al (2006) The relation between temperature, ozone, and mortality in nine French cities during the heat wave of 2003. Environ Health Perspect 114(9):1344–1347. https://doi.org/10. 1289/ehp.8328
- Fink AH, Brücher T, Krüger A, Leckebusch GC, Pinto JG, Ulbrich U (2004) The 2003 European summer heatwaves and drought-synoptic diagnosis and impacts. Weather 59(8):209–216
- Fischer EM, Schär C (2010) Consistent geographical patterns of changes in high-impact European heatwaves. Nat Geosci 3(6):398–403. https://doi.org/10.1038/ngeo866

- Fischer EM et al (2007) Soil moisture-atmosphere interactions during the 2003 European summer heat wave. J Clim 20(20):5081–5099. https://doi.org/10.1175/JCL14288.1
- Forzieri G et al (2017) Increasing risk over time of weather-related hazards to the European population: a data-driven prognostic study. Lancet Planet Health 1(5):e200–e208. https://doi.org/10.1016/S2542-5196(17)30082-7. The Author(s). Published by Elsevier Ltd. This is an Open Access article under the CC BY 4.0 license
- Fouillet A et al (2006) Excess mortality related to the August 2003 heat wave in France. Int Arch Occup Environ Health 80(1):16–24
- Founda D, Giannakopoulos C (2009) The exceptionally hot summer of 2007 in Athens, Greece—a typical summer in the future climate? Glob Planet Chang 67(3–4):227–236. https://doi.org/10. 1016/j.gloplacha.2009.03.013. Elsevier B.V.
- Founda D, Santamouris M (2017) Synergies between Urban Heat Island and Heat Waves in Athens (Greece), during an extremely hot summer (2012). Sci Rep 7(1):10973. https://doi.org/10.1038/ s41598-017-11407-6. Nature Publishing Group
- Fu SH et al (2018) Mortality attributable to hot and cold ambient temperatures in India: a nationally representative case-crossover study. PLoS Med 15(7):1–17. https://doi.org/10.1371/journal. pmed.1002619
- Gabriel KMA, Endlicher WR (2011) Urban and rural mortality rates during heat waves in Berlin and Brandenburg, Germany. Environ Pollut 159(8–9):2044–2050. https://doi.org/10.1016/j. envpol.2011.01.016. Elsevier Ltd
- García-Herrera R, Díaz J, Trigo RM, Hernández E (2005) Extreme summer temperatures in Iberia: health impacts and associated synoptic conditions. Ann Geophys 23:239–251. https://doi.org/ 10.5194/angeo-23-239-2005
- Giannaros TM (2013) Study of the urban heat island effect using a meso-scale atmospheric model and analyzing observational data. Aristotle University of Thessaloniki
- Giannaros C (2018) Sensitivity analysis and optimization of a mesoscale atmospheric model. Aristotle University of Thessaloniki
- Giannaros TM, Melas D (2012) Study of the urban heat island in a coastal Mediterranean City: the case study of Thessaloniki, Greece. Atmos Res 118:103–120. https://doi.org/10.1016/j. atmosres.2012.06.006. Elsevier B.V.
- Giannaros C et al (2018) A comprehensive approach for the simulation of the Urban Heat Island effect with the WRF/SLUCM modeling system: the case of Athens (Greece). Atmos Res 201: 86–101. https://doi.org/10.1016/j.atmosres.2017.10.015. Elsevier B.V.
- Giannaros C, Melas D, Giannaros TM (2019) On the short-term simulation of heat waves in the Southeast Mediterranean: sensitivity of the WRF model to various physics schemes. Atmos Res 218:99–116. https://doi.org/10.1016/J.ATMOSRES.2018.11.015. Elsevier
- Grumm RH (2011) The central European and Russian heat event of July–August 2010. Bull Am Meteorol Soc 92(10):1285–1296
- Hajat S, Bird W, Haines A (2004) Cold weather and GP consultations for respiratory conditions by elderly people in 16 locations in the UK. Eur J Epidemiol 19(10):959–968. https://doi.org/10. 1007/s10654-004-5731-4
- Harpaz T et al (2014) Extreme summer temperatures in the East Mediterranean—dynamical analysis. Int J Climatol 34(3):849–862
- Hartmann DL et al (2013) Observations: Atmosphere and Surface, Climate Change 2013 the Physical Science Basis: Working Group I Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. https://doi.org/10.1017/CBO9781107415324.008
- Heaviside C, Vardoulakis S, Cai XM (2016) Attribution of mortality to the urban heat island during heatwaves in the West Midlands, UK. Environ Health 15(Suppl 1). https://doi.org/10.1186/ s12940-016-0100-9
- Hu L, Luo JJ, Huang G, Wheeler MC (2019) Synoptic features responsible for heat waves in Central Africa, a region with strong multidecadal trends. J Clim 32(22):7951–7970
- Institute for Sustainable Resources (2010) Impacts and adaptation responses of infrastructure and communities to heatwaves: the Southern Australian Experience of 2009, 00213, p 163. http://

www.nccarf.edu.au/business/sites/www.nccarf.edu.au.business/files/attached_files_ publications/Pub13_10SouthernCitiesHeatwaves—CompleteFindings.pdf

- IPCC (2014) Climate Change 2014: Synthesis Report. Contribution of WorkingGroups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (Geneva: 2014), Managing the risks of extreme events and disasters to advance climate change adaptation: special report of the intergovernmental panel on climate change. Geneva, Switzerland. https://doi.org/10.1017/CBO9781139177245.003
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Jendrizky G (1998) Impacts of extreme and persistent temperature-cold waves and heat waves. In: WMO/UNESCO Sub-Forum on Natural Disaster Reduction, pp 43–53. http://www.eird.org/ bibliovirtual/capra/pdf/eng/doc15263/doc15263-contenido.pdf
- Jones TS et al (1982) Morbidity and mortality associated with the July 1980 heat wave in St Louis and Kansas City, Mo. JAMA 247(24):3327–3331
- Jubb I, Canadell P, Dix M (2013) Representative concentration pathways (RCPs). Australian Government, Department of the Environment
- Kalisa E et al (2018) Temperature and air pollution relationship during heatwaves in Birmingham, UK. Sustain Cities Soc 43:111–120. https://doi.org/10.1016/j.scs.2018.08.033. Elsevier Ltd
- Karl TR, Quayle RG (1981) The 1980 summer heat wave and drought in historical perspective. Mon Weather Rev 109(10):2055–2073
- Katsouyanni K et al (1993) Evidence for interaction between air pollution and high temperature in the causation of excess mortality. Arch Environ Health 48(4):235–242. https://doi.org/10.1080/ 00039896.1993.9940365
- Keatinge WR, Donaldson GC (1997) Cold exposure and winter mortality from ischaemic heart disease, cerebrovascular disease, respiratory disease, and all causes in warm and cold regions of Europe. Lancet 349(9062):1341–1346. https://doi.org/10.1016/S0140-6736(96)12338-2
- Kiktev D et al (2003) Comparison of modeled and observed trends in Indices of daily climate extremes. J Clim 16(22):3560–3571. https://doi.org/10.1175/1520-0442(2003)016<3560: COMAOT>2.0.CO;2
- Kim HH (1992) Urban heat island. International Journal of Remote Sensing 13:2319-2336
- Koppe C, Kovats S, Jendritzky G, Menne B (2004) Heat-waves: risks and responses. In: Health and Global Environmental Change Series, No. 2, World Health Organization, Geneva, vol 2, pp 3505–3345. https://doi.org/10.1007/s00484-009-0283-7
- Laaidi K et al (2012) The impact of Heat Islands on mortality in Paris during the August 2003 heat wave. Environ Health Perspect 120(2):254–259
- Lhotka O, Kyselý J, Farda A (2018) Climate change scenarios of heat waves in Central Europe and their uncertainties. Theor Appl Climatol 131(3–4):1043–1054
- Li B-Z (2014) The effectiveness of cool and green roofs as urban heat island mitigation strategies. Environ Res Lett 9. https://doi.org/10.1088/1748-9326/9/5/055002
- Li D, Bou-Zeid E (2013) Synergistic interactions between urban heat islands and heat waves: the impact in cities is larger than the sum of its parts. J Appl Meteorol Climatol 52(9):2051–2064. https://doi.org/10.1175/JAMC-D-13-02.1
- Livezey RE (1980) Weather and circulation of July 1980. Climax of a historic heat wave and drought over the United States. Mon Weather Rev 108(10):1708–1716
- Loikith PC, Broccoli AJ (2012) Characteristics of observed atmospheric circulation patterns associated with temperature extremes over North America. J Clim 25(20):7266–7281
- Lu RY, Chen RD (2016) A review of recent studies on extreme heat in China. Atmos Ocean Sci Lett 9(2):114–121
- Luo M, Lau NC (2017) Heat waves in southern China: synoptic behavior, long-term change, and urbanization effects. J Clim 30(2):703–720
- Maheras P, Flocas H, Tolika K, Anagnostopoulou C, Vafiadis M (2006) Circulation types and extreme temperature changes in Greece. Clim Res 30(2):161–174

- Matthies F et al (2008) Heat-health action plans-guidance. WHO Europe. http://www.euro.who.int/ ___data/assets/pdf_file/0006/95919/E91347.pdf
- Matzarakis A, Nastos PT (2011) Human-biometeorological assessment of heat waves in Athens. Theor Appl Climatol 105(1):99–106. https://doi.org/10.1007/s00704-010-0379-3
- Matzarakis A, Rocco M, Najjar G (2009) Thermal bioclimate in Strasbourg—the 2003 heat wave. Theor Appl Climatol 98(3–4):209–220. https://doi.org/10.1007/s00704-009-0102-4
- McGregor GR et al (2007) The social impacts of heat waves, using science to create a better place. https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/291052/scho0 807bncw-e-e.pdf
- McGregor GR et al (2015) Heatwaves and health: guidance on warning-system development. World Meteorological Organization and World Health Organization. https://doi.org/10.1109/ ICEBE.2013.49. M4—Citavi
- Meehl GA, Tebaldi C (2004) More intense, more frequent, and longer lasting heat waves in the 21st century. Science 305(5686):994–997
- Metzger KB, Ito K, Matte TD (2010) Summer heat and mortality in New York City: how hot is too hot? Environ Health Perspect 118(1):80–86. https://doi.org/10.1289/ehp.0900906
- Michelozzi P et al (2010) Surveillance of summer mortality and preparedness to reduce the health impact of heat waves in Italy. Int J Environ Res Public Health 7(5):2256–2273. https://doi.org/ 10.3390/ijerph7052256
- Miralles DG et al (2014) Mega-heatwave temperatures due to combined soil desiccation and atmospheric heat accumulation. Nat Geosci 7(5):345–349. https://doi.org/10.1038/ngeo2141
- Morignat E et al (2018) Impact of heat and cold waves on female cattle mortality beyond the effect of extreme temperatures. J Therm Biol 78:374–380. https://doi.org/10.1016/j.jtherbio.2018.11. 001. Elsevier Ltd
- Morini E et al (2016) The impact of albedo increase to mitigate the urban heat island in Terni (Italy) using the WRF model. Sustainability (Switzerland) 8(10):1–14. https://doi.org/10.3390/ su8100999
- Moss R et al (2008) Towards new scenarios for the analysis of emissions: climate change, impacts and response strategies. Intergovernmental Panel on Climate Change Secretariat (IPCC)
- Nori-Sarma A et al (2019) Advancing our understanding of heat wave criteria and associated health impacts to improve heat wave alerts in developing country settings. Int J Environ Res Public Health 16(12):2089. https://doi.org/10.3390/ijerph16122089. MDPI AG
- Oke TR (1967) City size and the urban heat island. Atmos Environ 7(8):769-779
- Otto FEL et al (2012) Reconciling two approaches to attribution of the 2010 Russian heat wave. Geophys Res Lett 39:4
- Papanastasiou DK, Melas D, Kambezidis HD (2015) Air quality and thermal comfort levels under extreme hot weather. Atmos Res 152:4–13. https://doi.org/10.1016/j.atmosres.2014.06.002
- Peterson TC et al (2014) Changes in weather and climate extremes: state of knowledge relevant to air and water quality in the United States. J Air Waste Manag Assoc 64(2):184–197. https://doi.org/10.1080/10962247.2013.851044
- Pezza AB, van Rensch P, Cai W (2012) Severe heat waves in Southern Australia: synoptic climatology and large scale connections. Clim Dyn 38(1–2):209–224. https://doi.org/10.1007/ s00382-011-1016-2
- Poupkou A, Zanis P, Nastos P, Papanastasiou D, Melas D, Tourpali K, Zerefos C (2011) Present climate trend analysis of the Etesian winds in the Aegean Sea. Theor Appl Climatol 106(3–4):459–472
- Ratnam JV et al (2016) Anatomy of Indian heatwaves. Sci Rep 6(1):1-11
- Robine J-M et al (2008) Death toll exceeded 70,000 in Europe during the summer of 2003. C R Biol 331(2):171–178. https://doi.org/10.1016/j.crvi.2007.12.001
- Rogelj J, Meinshausen M, Knutti R (2012) Global warming under old and new scenarios using IPCC climate sensitivity range estimates. Nat Clim Chang 2(4):248–253
- Russo S et al (2014) Magnitude of extreme heat waves in present climate and their projection in a warming world. J Geophys Res Atmos 119(22):12–500

- Saxena P, Naik V (eds) (2018) Air pollution: sources, impacts and controls. Cabi, Wallingford, pp 1–217
- Saxena P, Sonwani S (2019) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Srivastava A, Tyagi M, Kaur S (2019) Impact of tropospheric ozone on plant metabolism—a review. Pollut Res 38(1):175–180
- Semenza JC et al (1999) Excess hospital admissions during the July 1995 heat wave in Chicago. Am J Prev Med 16:269–277
- Shahmohamadi C-A, Etessam M (2011) Healthy environment: the need to mitigate urban Heat Island effects on human health. Procedia Engineering 20:61–70. https://doi.org/10.1016/j. proeng.2011.11.139
- Silva HR, Phelan PE, Golden JS (2010) Modeling effects of urban heat island mitigation strategies on heat-related morbidity: a case study for Phoenix, Arizona, USA. Int J Biometeorol 54(1):13–22. https://doi.org/10.1007/s00484-009-0247-y
- Smoyer-tomic KE, Kuhn R, Hudson A (2003) Heat wave hazards: an overview of heat wave impacts in Canada. pp 463–485
- Speirs JC et al (2010) Foehn winds in the McMurdo dry valleys, Antarctica: the origin of extreme warming events. J Clim 23(13):3577–3598
- Stefanon M, D'Andrea F, Drobinski P (2012) Heatwave classification over Europe and the Mediterranean region. Environ Res Lett 7(1):014023
- Stocker TF et al (2013) Climate change 2013: the physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, p 1535
- Stott PA, Stone DA, Allen MR (2004) Human contribution to the European heatwave of 2003. Nature 432(7017):610–614
- Takane Y, Kusaka H (2011) Formation mechanisms of the extreme high surface air temperature of 40.9 C observed in the Tokyo metropolitan area: considerations of dynamic foehn and foehnlike wind. J Appl Meteorol Climatol 50(9):1827–1841
- Tan J et al (2010) The urban heat island and its impact on heat waves and human health in Shanghai. Int J Biometeorol 54(1):75–84. https://doi.org/10.1007/s00484-009-0256-x
- Taylor KE, Stouffer RJ, Meehl GA (2012) An overview of CMIP5 and the experiment design. Bull Am Meteorol Soc 93(4):485–498
- Theoharatos G et al (2010) Heat waves observed in 2007 in Athens, Greece: synoptic conditions, bioclimatological assessment, air quality levels and health effects. Environ Res 110(2):152–161. https://doi.org/10.1016/J.ENVRES.2009.12.002. Academic
- Thom EC (1959) The discomfort index. Weatherwise 12(2):57-61. https://doi.org/10.1080/ 00431672.1959.9926960
- Tressol M et al (2008) Air pollution during the 2003 European heat wave as seen by MOZAIC airliners. Atmos Chem Phys 8(8):2133–2150. https://doi.org/10.5194/acp-8-2133-2008
- Tzavali A et al (2015) Urban heat island intensity: a literature review. Fresenius Environ Bull 24(12B):4537–4554
- Unal YS, Tan E, Mentes SS (2013) Summer heat waves over western Turkey between 1965 and 2006. pp 339–350. https://doi.org/10.1007/s00704-012-0704-0
- Unger J et al (2020) Comparison of regional and urban outdoor thermal stress conditions in heatwave and normal summer periods: a case study. Urban Clim 32:100619. https://doi.org/ 10.1016/j.uclim.2020.100619. Elsevier

- United Nations (2018). World Urbanization Prospects: The 2018 Revision. (ST/ESA/SER.A/420). https://www.un.org/development/desa/publications/2018-revision-of-world-urbanization-prospects.html
- van Vliet MTH, Zwolsman JJG (2008) Impact of summer droughts on the water quality of the Meuse river. J Hydrol 353(1–2):1–17. https://doi.org/10.1016/j.jhydrol.2008.01.001
- Vardoulakis S, Kassomenos P (2008) Sources and factors affecting PM10 levels in two European cities: implications for local air quality management. Atmos Environ 42(17):3949–3963
- Vautard R et al (2007) Summertime European heat and drought waves induced by wintertime Mediterranean rainfall deficit. Geophys Res Lett 34(7):1–5. https://doi.org/10.1029/ 2006GL028001
- Voogt JA (2004) Urban heat islands: hotter cities. America Institute of Biological Sciences, pp 1–2. https://doi.org/10.1163/_q3_SIM_00374
- Westerling H, Cayan S (2006) Warming and earlier spring increase Western U.S. Forest wildfire activity. Science 313(5789):940–943. https://doi.org/10.1126/science.1128834
- WHO (2016) Ambient air pollution: a global assessment of exposure and burden of disease. http:// www.ghbook.ir/index.php?name=نفر منورس امعاین وین محنوی مناوین مناوین من معنی من محنوی من معاوم من www.ghbook.ir/index.php?name=218& task=readonline&book_id=13650&page=73&chkhashk=ED9C9491B4&Itemid=218& lang=fa&tmpl=component
- Williams J, Hewitt CN, Jackson AV (2008) Atmospheric science for environmental scientists. J Atmos Chem 61(3):203–204. https://doi.org/10.1007/s10874-009-9132-7
- Williams S et al (2012) Heat and health in Adelaide, South Australia: assessment of heat thresholds and temperature relationships. Sci Total Environ 414:126–133. https://doi.org/10.1016/j. scitotenv.2011.11.038. Elsevier B.V.
- WMO (2016) Guidelines on the definition and monitoring of extreme weather and climate events, task team on the definition of extreme weather and climate events. https://doi.org/10.1016/j. corsci.2014.12.017
- Yin Q, Wang J, Ren Z, Li J, Guo Y (2019) Mapping the increased minimum mortality temperatures in the context of global climate change. Nat Commun 10:4640. https://doi.org/10.1038/s41467-019-12663-y
- Zaitchik BF, Macalady AK, Bonneau LR, Smith RB (2006) Europe's 2003 heat wave: a satellite view of impacts and land–atmosphere feedbacks. Int J Climatol 26(6):743–769
- Zampieri M et al (2009) Hot European summers and the role of soil moisture in the propagation of Mediterranean drought. J Clim 22(18):4747–4758. https://doi.org/10.1175/2009JCLI2568.1
- Zhao L et al (2018) Interactions between urban heat islands and heat waves. Environ Res Lett 13(3). https://doi.org/10.1088/1748-9326/aa9f73
- Ziv B, Saaroni H, Alpert P (2004) The factors governing the summer regime of the eastern Mediterranean. Int J Climatol 24(14):1859–1871
- Zuo J et al (2015) Impacts of heat waves and corresponding measures: a review. J Clean Prod 92:1– 12. https://doi.org/10.1016/j.jclepro.2014.12.078. Elsevier Ltd


Intense Biomass Burning Over Northern India and Its Impact on Air Quality, Chemistry and Climate

Chaitri Roy, D. C. Ayantika, Imran Girach, and Chandrima Chakrabarty

Abstract

The provinces of Punjab, Haryana and western Uttar Pradesh in northwest Indo-Gangetic Plain (IGP) produce two-thirds of India's food grains, and are widely known as the "breadbasket" of India. This "breadbasket" accounts for roughly a quarter of the total annual fire events observed over India, and hence is a regional hotspot for biomass burning. The enormous amount of aerosols and trace gases, emitted largely from extensive post-harvest crop residue burning in autumn (October-November) and spring (April-May), along with events of wildfires, and domestic consumption of biofuels, unfurls its influence along the length of the IGP under the prevailing meteorological conditions. The elevated levels of pollutants, primarily generated from intense post-monsoon biomass burning over northwest IGP, under the influence of unfavourable meteorological conditions, contribute significantly to the episodes of extreme pollution over the regions downwind of northwest IGP. The downwind region of intense biomass burning experiences high values of particulate matter, including black carbon, organic carbon, ozone, volatile organic compounds and other ozone precursors, resulting in an intense photochemistry and an adverse air quality. Furthermore, excessive smoke aerosols in the atmosphere exert a strong surface cooling, but an overall atmospheric warming, thereby changing the regional radiative balance.

C. Roy $(\boxtimes) \cdot D$. C. Ayantika

Centre for Climate Change Research, Indian Institute of Tropical Meteorology, Ministry of Earth Sciences, Pune, India

e-mail: chaitri.cat@tropmet.res.in

I. Girach

Space Physics Laboratory, Vikram Sarabhai Space Centre, Thiruvananthapuram, India

C. Chakrabarty Indian Meteorological Department, Pune, India

169

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_8

Keywords

Biomass burning \cdot North India \cdot Extreme events \cdot Aerosols \cdot Trace gases \cdot Haze \cdot Radiative forcing \cdot Air pollution \cdot Air quality \cdot Tropospheric chemistry

8.1 Introduction

Biomass burning is defined as combustion of vegetation, either living or dead, and it can be both natural and anthropogenic (Levine 1991; Bond et al. 2004). While wildfires are the natural sources of biomass burning, the anthropogenic or humaninduced sources include burning of crop residue or grassland for the purpose of changes in land use. Combustion of biomass is one of the key sources of atmospheric trace gases and aerosols (Crutzen and Andreae 1990; Saxena et al. 2021). Across the globe, the events of biomass burning have demonstrated significant impacts on the environment, land cover, atmospheric chemistry, human health, radiative forcing and climate (Saxena and Srivastava 2020). Figure 8.1 gives an overview of the sources and impacts of biomass burning emissions.

A bulk of the Earth's biomass burning comes from the tropics—primarily the South American and Southeast Asian forests. It is mainly governed by various factors such as lightning, topography and weather. For instance, the tropical phenomenon of El Niño–Southern Oscillation (ENSO) is largely responsible for



Fig. 8.1 An overview of the sources and impacts of biomass burning emissions (modified from Wan et al. 2019)

imposing drought conditions over Southeast Asia and India (Siegert et al. 2001). Droughts enhance the dry conditions over a forested/vegetated land, making it conducive for the occurrence of fire. Natural fires triggered by lightning also contribute significantly to the total biomass burning (Andreae 1991). Besides natural factors, a major chunk of biomass burning in the tropics comes from changes in land use and clearing of land which are both human induced. Burning of agricultural waste to clear land for the next crop is a part of the agricultural practices in India, and in many other countries in South and Southeast Asia (Vadrevu et al. 2019). In the Eastern Ghats and north-eastern India, agricultural practice like slash-and-burn (jhum cultivation) is very common, which leads to fires and ignition of biomass (Prasad et al. 2002; Prasad et al. 2003).

In the recent past, there have been several instances where anthropogenic interventions have accelerated the pace of biomass burning emissions in the atmosphere. For instance, the episode of out-of-control biomass burning in Indonesia in 1997, for agricultural purposes, which continued for 5 months, posed an overwhelming environmental threat for itself as well as its neighbouring countries (Brauer and Hisham-Hashim 1998; Fang et al. 1999). Climate change-induced extreme hot and dry conditions led to unprecedented fires over millions of hectares of forest in Australia in Balwinder-Singh et al. 2019 (Boer et al. 2020; Yu et al. 2020; Jain et al. 2021). Also, anthropogenic climate change has been held responsible for an increase in intense forest fire events in western North America (Williams et al. 2019; Goss et al. 2020). These processes of biomass burning release an enormous amount of particulate matter (PM) in the atmosphere, including smoke particles, resulting in deterioration of air quality, along with land-cover changes worldwide (Sonwani and Saxena 2021). In north-western Brazil, an increase in premature deaths due to extreme biomass burning was reported owing to rise in fire events from 2016 to Balwinder-Singh et al. 2019 in the Amazon rainforest (Nawaz and Henze 2020). Apart from that, aerosols emitted from biomass burning have also decreased the amount of sunlight touching the Earth's surface, causing a positive radiative forcing of +0.04 \pm 0.07 W m⁻² globally (Forster et al. 2007). These changes are alarming since it directly impacts Earth's temperature and climate, which is a matter of major global concern (Seiler and Crutzen 1980; Crutzen and Andreae 1990; Levine et al. 1995).

South Asian countries of India and China witness severe seasonal episodes of biomass burning like wildfires as well as fires resulting from agricultural crop burning and forest conversion during the boreal spring (Gautam et al. 2013; Shi et al. 2014; Chen et al. 2017). Recently in 2017, it was reported that India generated 488 Mt. of total crop residue, of which ~24% was burnt in agricultural fields. This led to release of an enormous amount of particulate matter having a size of $\leq 2.5 \,\mu\text{m}$ (PM2.5; 824 Gg), elemental carbon (EC; 58 Gg) and organic carbon (OC; 239 Gg). In addition to this, 211 Tg of greenhouse gases equivalent to carbon dioxide (CO₂) was also released into the atmosphere (Ravindra et al. 2019c).

The Indo-Gangetic Plain (IGP) covers a vast geographical area in South Asia, and covers 20% of the geographical area in India. This region is a very important agroecological zone, producing 42% of the total food grains and holding almost

40% of India's total population (Tripathi et al. 2007). In terms of aerosol-induced health impacts, this region has regularly recorded poor air quality resulting from biomass burning, and is expected to be one of the most vulnerable areas (Gautam et al. 2007; Apte et al. 2015; Bikkina et al. 2019; Sonwani et al. 2021). A large part of this problem is attributed to open crop burning in the agriculture-dominated regions of Punjab and Haryana, in northwest India (Singh et al. 2014). India's capital city, Delhi, sits in the northern IGP, and is one of the world's most densely populated cities. It experiences severe winters and lies close to the states of Punjab and Haryana. The wind pattern over these regions associated with smoke generated from biomass burning extensively affects the air quality and climate (Chen et al. 2017). Severe worsening of air quality was witnessed in the Indian capital city of Delhi in 2016 and 2017, due to emissions from excessive crop burning events in North Indian states of Punjab and Haryana, together with emissions from celebration of festivals under unfavourable meteorological conditions (Singh et al. 2014; Chauhan and Singh 2017; Kanawade et al. 2020).

In this chapter, we therefore review the status of excessive biomass burning emissions over the IGP, and their impacts. Section 8.2 describes the sources of biomass burning with depiction of intense biomass burning. Section 8.3 illustrates extreme biomass burning episodes, Sect. 8.4 provides a description of the elevated levels of trace gases and aerosols generated as a result of biomass burning, followed by their impacts in Sects. 8.5, and summary in Sect. 8.6.

8.2 Major Sources of Intense Biomass Burning in Northern India

India is one of the well-known agricultural nations in the world. It accounts for 18% of the total biomass burning in Asia (Hao and Liu 1994; Streets et al. 2003; Taylor 2010). Nearly 12 million hectares of land in the IGP region is harvested for wheat and rice, with a common practice of rotation of rice and wheat crops (Badarinath et al. 2009a). The harvesting techniques generate a large volume of agricultural residues such as rice and wheat straws, in the field, which needs to be eliminated before the next crop planting. Open burning is one of the most convenient and inexpensive ways to eliminate the agricultural waste; therefore, a quarter of the generated agricultural crop residue is burnt in open fields in India every year (Ravindra et al. 2019a, b, c). The major sources of biomass burning can also be seen in Fig. 8.1.

Wildfires, caused by both natural and anthropogenic activities, are another major source of biomass burning in North India. Forest fire burns 32-61 Tg·year⁻¹ annually, while agricultural fire over India burns 116-289 Tg·year⁻¹ (Venkataraman et al. 2006). Approximately 61% of the forest area in India, which is predominantly covered by deciduous trees, is vulnerable to fires (FSI 2015). Owing to activities such as shifting cultivation practices, controlled burning, deforestation and firewood burning, these forest fires are mainly anthropogenic in origin (Bahuguna and Upadhay 2002; Chand et al. 2007). However, the spatial coverage and damage

due to these forest fires are also determined by local and regional weather patterns (air temperature, wind speed and prolonged droughts).

Unlike wildfires and crop residue burning, which are largely a seasonal phenomenon, domestic biomass burning for energy production constitutes a continuous input of trace compounds into the environment. Use of biofuel such as fuelwood, charcoal and non-woody biofuels like dung cakes, primarily for cooking, heating or lighting especially in the rural areas, contributes to emissions of particulate matter and trace gases (Venkataraman et al. 2010; Pandey and Tyagi 2012; Bhattacharya and Nanda 1992; Aggarwal et al. 2009; Metz 1991; Singh et al. 2010). However, domestic biomass burning events are very scattered, which is one of the main reasons for sparse availability of its information.

8.3 Extreme Biomass Burning

Figure 8.2a illustrates the logarithm of annual fire counts averaged over 17 years (2002–2018) over India based on MODIS (MODerate resolution Imaging Spectrometer) fire detection. MODIS on board Terra and Aqua satellites is one of the instruments popularly used for fire counts data which has a spatial resolution of $1 \times 1 \text{ km}^2$. Fire event is a form of thermal anomaly and is detected from both brightness temperature at 4 µm and difference between brightness temperatures at 4 and 11 µm (Giglio et al. 2003). The fire pixels having detection confidence >30% (confidence class: nominal and high) are used (Jethva et al. 2019) here to generate fire count map. It is observed that nearly 200–500 fire events occur every year over most of the grids ($0.25 \times 0.25^{\circ}$) in northwest India (i.e. mainly states of Haryana, Punjab and some parts of western Uttar Pradesh). The total number of counts over this small region (marked by a magenta box in Fig. 8.2a) accounts for 26% of the



Fig. 8.2 (a) Annually accumulated fire counts (in log scale) over India gridded at $0.25^{\circ} \times {}^{\circ}0.25^{\circ}$, averaged during 2002–2018 based on MODIS fire event detection. (b) Pie chart represents relative contribution of fire counts from northwest India (i.e. mainly states of Haryana, Punjab and some parts of western Uttar Pradesh, marked by a magenta box)



Fig. 8.3 Schematic of seasonal pattern of fire counts over northwest along with depiction of agricultural practice (modified from Vadrevu et al. 2011)

total fire counts over India (pie chart in Fig. 8.2b). Hence, biomass burning over northwest India is more extensive (Sharma et al. 2011; Kaskaoutis et al. 2014) as compared to the rest of India, and therefore represents the regional hotspot over northern India. It should be noted that since fire counts are obtained from the counts of pixels where fires are detected, the total number of counts over a grid or a region depends upon pixel resolution of the observing satellite.

In northwest India, the provinces of Haryana, Punjab and western Uttar Pradesh produce two-thirds of the food grains of India, and are commonly referred to as the "breadbasket" of India (Jethva et al. 2019). Wheat and rice are the two principal crops grown over this region. The sowing time of Kharif crop (rice) is July–August, while that of Rabi crop (wheat) is November–December. The residue generated from rice paddy is burned in post-monsoon (autumn) months of October and November, whereas the residue from wheat crop is burned in the pre-monsoon (spring) months of April and May (Vadrevu et al. 2011; Singh and Kaskaoutis 2014). The two agricultural seasons, and the associated crop residue burning over this region, are pictorially shown in Fig. 8.3 (modified from Vadrevu et al. 2011). The two prominent peaks in fire counts, indicating extreme biomass burning, are observed during both autumn and spring, where autumn peak is significantly higher than that of spring.

Crop residue burning events during spring and autumn for the years 2002 and 2018 are shown in Fig. 8.4. The autumn peak in fire count is clearly more intense than the springtime peak, during both 2002 and 2018. The seasonally accumulated MODIS-detected fire counts over $0.25^{\circ} \times 0.25^{\circ}$ grids are over 100 during spring, while it reaches values as high as 300 during autumn over northwest India. The crop residue burning during autumn of 2018 is also much higher than that of 2002. This is due to an increase in rice production over the years, with a linear trend of 0.18 million tonnes per year and an associated increase in crop residue burning, leading to a statistically significant trend of 500 fire counts per year (Jethva et al. 2019; Vadrevu et al. 2019). Singh et al. (2020b) have estimated 29.9 Mt. of crop residue burned in Punjab and Haryana during 2017–2018, which resulted in an emission of



Fig. 8.4 Seasonally accumulated fire counts gridded at $0.25^{\circ} \times 0.25^{\circ}$ for spring (March–May) and autumn (October–November) during 2002 and 2018 along with synoptic winds (black arrows) at 900 hPa

194.1, 235.8, 12.3, 63.4 and 0.11 Gg of PM2.5, PM10, EC, OC and polycyclic aromatic hydrocarbon (PAH), respectively. High fire counts over north as well as northwest India have also been reported from emission inventories like EDGAR (Emission Database for Global Atmospheric Research) and GFED (Global Fire Emissions Database) during the post-monsoon season (Sarkar et al. 2018). Also, a shift towards mechanized and advanced techniques of cultivation systems has contributed significantly to the increase in crop residue (Sharma and Prasad 2008; Kumar et al. 2015; Manjunatha et al. 2015). While cereal straws, woody stalks and sugarcane leaves are typically burned over western IGP (Venkataraman et al. 2006), crop residues are commonly used as fodder and domestic fuel in eastern IGP, which explains lower emissions from eastern IGP (Vadrevu and Lasko 2015). The mean synoptic winds (from the fifth ECMWF (European Centre for Medium-Range Weather Forecasts) reanalysis-ERA5) over North India are north-westerly during both the seasons with small differences (Fig. 8.4). Burning of crop residue contributes to enormous amount of trace gas emissions and aerosol loading over the IGP (Badarinath et al. 2006, 2009a; Ram et al. 2012; Jain et al. 2014; Kaskaoutis et al. 2014; Vijayakumar et al. 2016; Liu et al. 2018). The pollutants emitted from open crop residue burning can travel thousands of kilometres downwind, spanning the full IGP area from west to east (Kaskaoutis et al. 2014), as can be seen from the wind direction over the IGP (Fig. 8.4). The local emissions due to burning, coupled with meteorological conditions, worsen the air quality across the entire Indo-Gangetic basin (Ram et al. 2016).

Wildfires also contribute to the high aerosol loading over the IGP during both spring and autumn. The dominant fire seasons are the pre-monsoon and post-monsoon months, which account for more than 45% and 24% of total annual fire counts, respectively. The North Indian state of Uttarakhand has an extended forest area and typically exhibits forest fire activity during February to June, with a spike in May and June (FSI 2015). Widespread forest fires near the foothills of the Himalayas have impacts on severe air quality and climate, since it releases a substantial amount of greenhouse gases and pollutants (Crutzen and Andreae 1990; Andreae and Merlet 2001; Duncan et al. 2003). Episodes of violent forest fires in northern India have been reported in 2016 and 2018 (Jha et al. 2016; Srivastava and Kumar 2020; Yarragunta et al. 2020). Such events have been reported to degrade air quality, and also propagate its impact vertically up to a depth of ~6.5 km in the troposphere (Yarragunta et al. 2020).

8.4 Biomass Burning-Induced Elevated Levels of Aerosols and Trace Gases

The amount of trace constituents emitted from biomass burning depends upon the type of biomass, its moisture content and its burning conditions such as ambient temperature, humidity and local wind speed (Seinfeld and Pandis 1998). The incomplete combustion of biomass releases a variety of particulate matter/aerosols such as carbonaceous aerosols, and trace gases such as carbon monoxide (CO), nitrogen oxides (NO_x) , sulphur dioxide (SO_2) and non-methane hydrocarbons (NMHCs); (Andreae and Merlet 2001; Saxena and Naik 2018). It also releases CO_2 and methane (CH₄), which are the major drivers of the Earth's radiation balance. The trace gases and aerosols modulate the chemistry of the atmosphere that has impacts on air quality, at both local and regional scales. Some of the trace gases are precursors of ozone (O₃), and also aerosols, via the process of gas-toparticle conversion (Saxena et al. 2020). O₃ is an important source of OH radical, which plays a central role in atmospheric chemistry (Seinfeld and Pandis 1998). Interestingly, the influence of the emitted gases is not limited only to the place of biomass burning, but gets transported over longer distances under the influence of the prevailing meteorological conditions. Vertical transport of trace gases due to convection also changes the chemistry and composition of the upper troposphere (Roy et al. 2017; Girach et al. 2020).

Observational sites	PM2.5	Observation period	References
Delhi (28.6° N, 77.2° E)	$>200 \ \mu g \ m^{-3}$ (~3.5 times above NAAQS of daily mean)	October– November, 2012	Bisht et al. (2015)
	\sim 150–250 μg m ⁻³ (~4 times above NAAQS of daily mean)	October– November, 2011	Tiwari et al. (2013)
Patiala (30.2° N, 76.3 ° E)	$123.1 \pm 25.5 \ \mu g \ m^{-3}$ (~2 times above NAAQS of daily mean)	October– November, of 2007–2009	Agarwal et al. (2012)
Amritsar (27.16° N, 78.08° E)	$178.4 \pm 83.8 \ \mu g \ m^{-3}$ (~3 times above NAAQS of daily mean)	October– November, 2016	Ravindra et al. (2019b)
Kanpur (26.5° N, 80.3° E)	$189 \pm 82 \ \mu g \ m^{-3}$ (~3 times above NAAQS of daily mean)	October– November, 2008	Ram et al. (2012)
Chandigarh (26° N, 76° E)	191.1 μ g m ⁻³ (~3.5 times above NAAQS of daily mean)	October– November, 2015	Ravindra et al. (2020)

Table 8.1 Biomass burning-induced elevated PM2.5 concentrations at a few stations over the IGP during crop residue burning period

8.4.1 Particulate Matter (PM)

Biomass burning is one of the chief sources of fine particulate matter, PM2.5. Open biomass burning contributes to 9–13% of ambient PM2.5 as per the estimation made during 1995–2000 (Venkataraman et al. 2006). The cities of Delhi and Chandigarh, located in the northwest IGP, have reported 7–20% and 8% contribution to ambient PM2.5 from biomass combustion, respectively (Chowdhury et al. 2007). Analysis of observations obtained from a number of satellites and ground-based measurements post-2010 revealed increased episodes of crop residue burning in northern and northwestern regions of India. Simultaneously, emissions from active fire events have been reported to increase by 83–106% for Punjab compared to anthropogenic emissions based on the data from GFED4 and Fire Inventory from NCAR (FINN; Singh et al. 2020a).

The National Ambient Air Quality Standards (NAAQS) for annual and 24-h averaged PM2.5 over India are 40 μ g m⁻³ and 60 μ g m⁻³, respectively. Table 8.1 accounts for biomass burning-induced elevated levels of PM2.5 observed over a few stations located in the IGP during crop residue burning period, where the PM2.5 concentrations are at least two times above the annual NAAQS. A multi-city campaign across the IGP during post-monsoon period of 2016 revealed an average PM2.5 concentration as high as 148.2 \pm 20 μ g m⁻³. Amritsar had the highest average PM2.5 concentration of 178.4 \pm 83.8 μ g m⁻³, whereas Chandigarh recorded the lowest PM2.5 concentration of 112.3 \pm 6.9 μ g m⁻³. Rohtak and Sonipat recorded PM2.5 concentration values in between these two, 158.4 \pm 79.8 and 156.5 \pm 105.3 μ g m⁻³, respectively. The daily average values over these locations were several times higher than the standard quality of national air for 24 h (Ravindra et al. 2019b). Recently, Chandigarh reported very adverse PM2.5

levels (191.1 μ g m⁻³), which were 3.5 times higher than the NAAOS for 24 h during October-November 2015 (Ravindra et al. 2020). Increased number of fire counts and PM2.5 concentrations as severe as $>100-150 \ \mu g \ m^{-3}$ were also observed during October-November 2012 over the IGP from MODIS satellite (Kaskaoutis et al. 2014). Over Delhi, the maximum value of PM2.5 on a daily scale is highly variable, and it depends on the transportation of air mass via meteorological parameters from source to sink (Beig et al. 2020). On events of typical air pollution, fire emissions contributed as much as 50–75% (80–120 μ g m⁻³) to PM2.5 concentration in Delhi, thereby emphasizing the importance of both external transport and local emissions to PM2.5 pollution in Delhi (Kulkarni et al. 2020). During post-monsoon season, the impact of biomass burning on PM2.5 is higher over Delhi, as compared to fossil fuel burning (Tiwari et al. 2013). A recent study from both model and observations estimated that 7-78% of Delhi's maximum observed PM2.5 changes can be related to crop residue burning in the upwind states (Cusworth et al. 2018). Paddy crop burning in the post-monsoon season of 2012, along with long-range transport of smoke, elevated the aerosol loading over the length of IGP.

Episodes of extreme pollution over the region downwind of northwest IGP including Delhi result from extensive crop residue burning emissions, besides local emissions, under unfavourable meteorological conditions, such as shallow boundary layer and low ventilation coefficient (Ojha et al. 2020; Sembhi et al. 2020). In line with this, stagnant atmospheric conditions led to an extreme event of pollution over Delhi during 1-7 November 2016, with PM2.5 concentrations shooting up to 1294.5 μ g m⁻³ (Kanawade et al. 2020). PM2.5 concentration over Delhi has earlier showed a high annual mean of $122.3 \pm 90.7 \ \mu g \ m^{-3}$ in 2011, with values ranging from ~150 μ g m⁻³ to as severe as ~250 μ g m⁻³ during post-monsoon of 2011 (Tiwari et al. 2013). Agricultural burning during 2012 reported critically high PM2.5 concentration with values $>200 \ \mu g \ m^{-3}$ over Delhi. Severely high values of 500 μ g m⁻³ (>10 times NAAQS) were also reported on certain days (Bisht et al. 2015). The mean PM_{2.5} concentration over Delhi during the post-monsoon of 2016 was 571.9 μ g m⁻³, which was ~38–60% higher than that of the previous years (Kanawade et al. 2020). The regional chemistry-transport simulations estimated ~20-60% of PM2.5 over western IGP (74-82°E) due to biomass burning over the region during autumn (Ojha et al. 2020).

8.4.2 Carbonaceous Aerosols

Incomplete combustion of biomass results in emission of carbonaceous aerosols, which are short-lived (with a lifetime of maximum 2 weeks; Bond et al. 2007, 2013a, b). It constitutes a prominent fraction of PM2.5, with key components being OC and black carbon (BC), and can be transported over long distances along with the wind (Lelieveld et al. 2001). EC is similar to BC but is measured by thermal technique, in contrast to light absorption technique for BC measurement (Andreae and Gelencsér 2006). After greenhouse gases, BC is the dominant light-absorbing species and the second major contributor to global warming (Jacobson 2002; Husain

Observational sites	Post-monsoon BC concentration	Observation period	References
Patiala (30.2° N, 76.3 ° E)	$>20 \ \mu g \ m^{-3}$	October–November of 2008	Kharol et al. (2012)
	10–15 µg m ⁻³	October–November of 2009–2011	Shaik et al. (2019)
Delhi (28.6° N, 77.2° E)	~18–20 µg m ⁻³	October–November of 2012	Bisht et al. (2015)
Kanpur (26.5° N, 80.3° E)	~10–12 µg m ⁻³	October–November of 2007–2011	Kanawade et al. (2014)
Delhi (28.6° N, 77.2° E)	13.7 μg m ⁻³	October–November of 2006	Bano et al. (2011)
Gorakhpur (26.75° N, 83.38° E)	$18\pm15~\mu g~m^{-3}$	October–November of 2013–2015	Vaishya et al. (2017)

Table 8.2 Biomass burning-induced high concentrations of BC aerosols during post-monsoon over a few stations located in the IGP

et al. 2007). It contributes to 25–50% of CO₂ warming globally (Forster et al. 2007; Ramanathan and Carmichael 2008; Jacobson 2010). On the other hand, OC illustrates scattering effect, and leads to negative radiative forcing (Liousse et al. 1996). The composition of aerosols over India has a large portion of BC (Mitra 1999; Dickerson et al. 2002; Guazzotti et al. 2003). A recent study reported that BC constituted nearly 6% of the total PM2.5 mass concentration (Tiwari et al. 2013). In India, a large population of both rural and urban households use solid biofuels (Venkataraman et al. 2010) which leads to emission of a large amount of BC. Emission from residential biofuels like wood, crop residue and dung is another source of atmospheric BC (Venkataraman et al. 2005).

Biomass burning over India contributes to approximately 33% of the total BC concentration in the atmosphere (Venkataraman et al. 2006). Seasonal influxes of crop residue from Punjab and Haryana also contribute significantly to BC loading (Tiwari et al. 2009, 2015; Srivastava et al. 2012; Bisht et al. 2015; Sharma et al. 2018). During post-monsoon or early winter, biomass sources contribute $\sim 28\%$ to the aerosol loading over the central IGP, whereas during pre-monsoon or monsoon, the contribution from biomass sources lowers to $\sim 16\%$ (Vaishya et al. 2017). The BC concentration reaches high values during October-November (emission from bio- and fossil fuel combustion), while it remains low during the months of April-May over the IGP (Bano et al. 2011; Kanawade et al. 2014; Singh et al. 2014; Prabhu et al. 2020). Table 8.2 accounts for biomass burning-induced elevated BC concentrations observed during post-monsoon over a few stations located in the northwest/central IGP. The northwest IGP station of Patiala reported higher concentration of BC ranging between 7 and 11 μ g m⁻³ during post-monsoon of 2010–2012 than that during reported pre-monsoon/monsoon. This is significantly contributed by the emissions from post-monsoon agricultural biomass burning (Sharma et al. 2017). During 2009–2011, high concentrations of BC over Patiala during post-monsoon $(10-15 \ \mu g \ m^{-3})$ were observed, which also indicated the influence of crop residue

burning practices (Shaik et al. 2019). Moreover, Patiala also reported extremely high values of BC concentration during post-monsoon of 2008 owing to crop residue burning, where the BC concentration values were above 7–20 μ g m⁻³ (Kharol et al. 2012; Sharma et al. 2012). A recent study during 2013–2016 revealed that post-monsoon BC values are higher (~10 μ g m⁻³) as compared to the other seasons (~2–6 μ g m⁻³) over Patiala, which is attributed to biomass burning (Bansal et al. 2019). Monthly variation of BC over Gorakhpur during 2013–2015, in central IGP, showed a post-monsoon high and a monsoon low (Vaishya et al. 2017). Post-monsoon dominance of biomass burning was also observed over Kanpur during 2007–2011 (Kanawade et al. 2014). The elevated levels of BC were also in line with satellite-derived aerosol optical depth (AOD; Sharma et al. 2012).

The densely populated city of Delhi, in the northwest IGP, experiences almost a quarter ($\sim 22\%$) of its BC from biomass burning (Bisht et al. 2016). It was reported that large-scale post-harvest crop residue in nearby areas contributes to high loading of BC over Delhi during winter and autumn (Bikkina et al. 2019). While mean BC concentrations were 6.3 and 4.9 μ g m⁻³ during summer and monsoon, respectively, over Delhi, post-monsoon season reported a very high value of 17.4 μ g m⁻³ (Sharma et al. 2018). Daily mass concentration of BC over Delhi varied from as low as 0.9 to as high as $\sim 12 \ \mu g \ m^{-3}$ during October–November 2011 (Tiwari et al. 2013). Extreme biomass burning in the post-monsoon season of 2013 also resulted in high concentration of BC over Delhi (Liu et al. 2018). Agricultural burning during 2012 reported critically high BC concentration with values reaching $\sim 20 \ \mu g \ m^{-3}$ over Delhi (Bisht et al. 2015). Unlike open biomass burning during spring/autumn over northwest India, the wintertime small-scale biomass/biofuel burning contributed ~22% over Delhi during 2015-2016 (Bisht et al. 2016). It should be mentioned here that meteorological parameters (e.g. shallow boundary layer, stagnant conditions, synoptic winds) play a key role in the variability of atmospheric BC concentration (Safai et al. 2008; Tiwari et al. 2013). A study by Tyagi et al. (2020) showed that 2016 reported high values of BC as compared to 2017, which was partially due to variability in meteorological parameters.

OC is another key component of carbonaceous aerosols which constitutes a significant proportion of the PM2.5 mass. Recently, Andreae (2019) has compiled the emission factors, and provided a global estimation of the amount of trace gases and aerosols emitted from biomass burning. Aerosols generated from biomass burning include higher amount of OC, BC or EC and potassium (K), and these species and/or their ratios with other aerosol components are indicators of biomass burning activity (Andreae 1983). The OC/EC ratios are often used to find out the relative dominance of biomass versus fossil fuel burning. While OC/EC less than 2 indicates the dominance of fossil fuel burning emission, values greater than 5 suggest the dominance of biomass burning (Seinfeld and Pandis 1998; Bisht et al. 2016). Over the IGP, urban sites show higher OC/EC ratios ranging between 2.4 and 14.5 (mean = 7.8 ± 2.4), which is largely indicative of the contribution from biomass burning sources (wood-fuel and agriculture waste), whereas rural sites show a lower OC/EC ratio (range: 2.1–4.0, mean = 3.1 ± 0.6) indicating dominant contribution from coal-fired industries (Ram and Sarin 2010). Potassium being

higher in biomass, K/EC ratio was seen to be higher (average ratio ~0.80) during autumn over Patiala in the downwind of crop residue burning (Rastogi et al. 2016). In line with higher K/EC ratios, the higher OC/EC ratio (7.5 ± 1.3) observed during autumn at a high-altitude site, Nainital, over central Himalayas, indicates the contribution from both organic aerosols and transported aerosols which were emitted from western IGP biomass burning (Ram et al. 2010a).

The high OC and EC levels were observed during the winter season and low during the monsoon season (Singh et al. 2016). A year-long study of carbonaceous aerosols studied at an urban and rural site in the IGP showed average OC and EC concentrations of 25.6 and 13.7 μ g m⁻³, respectively, at the urban site, while at the rural site, OC and EC concentrations were 29.6 and 12.8 μ g m⁻³, respectively. But exceptionally high values of OC and EC observed during November were indicative of the impact of post-harvest large-scale open biomass burning. The concentration of OC observed over Delhi in 2011–2012 varied between 14.1 \pm 4.3 $\mu g~m^{-3}$ and values as adverse as $38.1 \pm 17.9 \,\mu\text{g m}^{-3}$ (Srivastava et al. 2014). Kanpur, located in the IGP, witnessed a three- to fourfold increase in the OC concentrations during winter predominantly due to biomass burning (wood fuel and agricultural waste) emission in 2007–2008. The relatively high OC/EC ratio supports that emissions from biomass burning are overwhelming for the particulate OC in the IGP (Ram et al. 2010b). The mass concentrations of EC ranged between 3.8 and 17.5 μ g m⁻³ during October-November (paddy residue burning emission), whereas it remained low during December-March (2.3-8.9 μ g m⁻³; emission from bio- and fossil fuel combustion) and April-May (2.0-8.8 μ g m⁻³; wheat residue burning emission; Singh et al. 2014).

8.4.3 Ozone

The air masses sampled over Delhi, Agra, Mohali, Pantnagar, Dehradun and Nainital trace back to the region of biomass burning (Haryana-Punjab) based on the analysis of air mass trajectories and synoptic winds during spring as well as autumn. Due to higher solar insolation over northern India during spring as compared to that in autumn, photochemistry is more intense during spring over this region in the presence of precursor gases emitted from burning biomass. Correspondingly, the amount of photochemical build-up of O_3 depends on the amount of precursor gases and the available solar insolation. Biomass burning-induced enhancement of noontime O_3 is observed to be in the range of 7 ppbv (10% over Mohali) to ~32 ppbv (72% over Delhi). Table 8.3 shows noontime O₃ enhancement over different sites highlighting the heterogeneity in O_3 enhancement due to the influence of northwest biomass burning. It is worthwhile to note that the enhancements in O_3 estimated over Delhi, Agra and Mohali are with respect to that in the preharvest period, whereas over other sites, O_3 enhancements are with respect to low-fire-activity period. The springtime maximum O₃ of 39.3 \pm 18.9 ppbv over Pantnagar (29.0° N, 79.5° E, 231 m amsl), a semi-urban site in the IGP, is also influenced by Haryana-Punjab crop residue burning (Ojha et al. 2012). Similarly, biomass burning influence is

Observational sites	Noontime surface O ₃ background values	Noontime surface O ₃ enhancement	Observation period of O ₃ enhancement	References
Delhi (28.4° N, 77.1° E)	~44 ppbv (preharvest period)	31.7 ppbv (~72%)	September– October 2015	Kumari et al. (2018)
Agra (27.2° N, 78.1° E)	~ 52 ppbv (preharvest period)	17.3 ppbv (~33%)	September– October 2015	Kumari et al. (2018)
Mohali (30.7°N, 76.7°E; 310 amsl)	66 and 68 ppbv preharvest period of spring and autumn, respectively	19 ppbv (29%) and 7 ppbv (10%)	Spring and autumn of 2011–2013	Kumar et al. (2016)
Dehradun (77.9° E, 30.3° N, 600 amsl)	51–59 ppbv (low-fire- activity period)	21–29 ppbv (~35–56%)	May 2018	Ojha et al. (2019)
Nainital (79.5° E, 29.4° N,	56 ± 11 ppbv (low-fire- activity period)	19 ppbv (~34%)	April–May 2007–2009	Kumar et al. (2011)
~2 km amsl)	-	4-18%	April–May 2009–2011	Sarangi et al. (2014)

Table 8.3 Biomass burning-induced noontime O_3 enhancement in the downwind regions ofHaryana–Punjab

observed to be infused with other anthropogenic emissions over Kanpur (26.46° N, 80.33° E, ~125 m amsl) leading to springtime maximum O_3 (Gaur et al. 2014). While in situ observations are limited, model simulations can provide biomass burning contribution at synoptic scale with effective delineation of other processes, such as changes in meteorological conditions. The regional chemistry transport model, Weather Research and Forecasting coupled with chemistry (WRF-Chem), simulated daytime O_3 enhancement of about 1–7 ppb (4–10%) over IGP and foothills of Himalayas during April and May (Jena et al. 2015). Note that this enhancement is corresponding to an average over daytime (9:30–17:30 local time), and hence it is lower in magnitude as compared to the noontime (peak values) enhancement is higher (over Mohali and Nainital) during spring as compared to that during autumn mainly due to higher availability of solar insolation intensifying the photochemistry.

A large number of forest fire events (April–May 2016) occurred in the foothills of north-western Himalayas (Uttarakhand state), which led to 9–11% enhancement in surface O_3 as inferred from WRF-Chem simulations (Yarragunta et al. 2020). Copernicus Atmosphere Monitoring Service (CAMS) combines a suite of observations with a state-of-the-art data assimilation and forecasting system. CAMS captures the variability of noontime O_3 with very good agreement (correlation coefficient of 0.86; Ojha et al. 2019) with springtime observations in the



Fig. 8.5 Fire event locations during high (11 May 2018) and low (3 May 2018) fire activity (**a**, **b**). The percent enhancement (i.e. ΔO_3) in CAMS-simulated noontime surface ozone (**c**) between the high and low fire activities. Synoptic wind (black arrows) at 900 hPa during May 2018 is overlaid in figure (**c**)

downwind of northwest IGP. However, there exists significant bias and hence CAMS is utilized here to show the relative enhancement in noontime O₃ on a regional basis. As a representative case, Fig. 8.5a, b shows fire event locations during high and low fire activity for 2 days (11 and 3 May 2018) over Haryana-Punjab region. To illustrate the O₃ enhancement in the downwind of biomass burning, the relative difference in noontime O_3 on the subsequent day of high and low fire activity is taken and represented in Fig. 8.5c. Synoptic wind (from ERA-5 reanalysis) is superimposed over the change in surface O_3 (i.e. ΔO_3). The O_3 enhancement is in the range of 10-65% over northern India caused by biomass burning during spring. The maximum O_3 enhancement is in the downwind of the source region of precursor gases. Significant amount of NO over the source region would titrate O_3 compensating the O_3 produced from other precursors such as NO₂. Titration reaction NO + O_3 adds up NO₂ and due to shorter lifetime of NO (a few minutes; Seinfeld and Pandis 1998), higher magnitude of NO₂/NO ratio in the downwind would lead to en route photochemical production of O_3 along the transport pathways, besides the contributions from other precursors.

The intense biomass burning over northern India influences not only the surfacelevel composition and chemistry but also the boundary layer as well as free tropospheric chemistry. Mean vertical profile of O₃ for the high- and low-fire-activity periods during spring over central Himalayas (Nainital, which is considered to be the regional representative site for northern India; Kumar et al. 2011) and Doon Valley (Dehradun; foothills of Himalayas) is shown in Fig. 8.6. The crop residue burning over northwest IGP leads to enhancement in noontime O₃ by 35–56% (21–29 ppbv) over Dehradun (Ojha et al. 2019) and ~ 30% (19.9 ± 4.6 ppbv) in 2–4 km altitude range over Nainital (Ojha et al. 2014).



8.4.4 Ozone Precursors

Whereas O_3 has been studied more extensively, the studies on precursors of O_3 (CO, NO_x, VOCs, etc.) are limited due to their sparse in situ measurements. CO showed enhancement by 187 ppbv over Agra, and up to 300 ppbv over Delhi, during biomass burning during post-harvest period, as compared to preharvest period, during autumn of 2015. CO levels were observed to be 204-1313 and 1060-2080 ppbv over Delhi and Agra, respectively (Kumari et al. 2018). However, NO_x did not show any significant variation in pre- and post-harvest periods at both the sites mainly due to shorter lifetime of NO_x as well as influence of local emissions (predominant emission of NO_x from vehicular source). Crop residue burning event (10-12 May 2018) over northwest IGP during pre-monsoon caused enhancement of CO, NO_x and O₃ by 86, 21 and 15%, respectively, over Agra. The peak values of CO, NO_x and O₃ were observed to be 1406, 115 and 91 ppbv, respectively, during this event (Kumari et al. 2020). Satellite observations showed surface CO higher than 300 ppbv during a severe air pollution episode (1-7 November 2016) over Delhi (Kanawade et al. 2020). High-altitude station over central Himalayas experienced the enhancements of CO by 26–117 ppbv (15–76%) and NO_v by 246–1375 pptv (35-51%) due to biomass burning during spring of 2009-2011. This led to CO and NO_v levels up to 1300 ppbv and 8000 pptv, respectively (Sarangi et al. 2014).

The observations of volatile organic compounds (VOCs) at Mohali showed massive increase by a factor of 3 with respect to baseline levels in the biomass burning plume during springtime (May 2012). This led to elevated levels reaching 100 ppbv for methanol, 50 ppbv for the sum of acetone and propanol, 30 ppbv for acetaldehyde, 7 ppbv for acetonitrile, 5 ppbv for isoprene and 20 ppbv for each of the aromatics, 80 ppbv NO_x, 50 ppbv SO₂ and 2000 ppbv CO (Sinha et al. 2014). Over the same location, the elevated concentrations of acetonitrile, benzene, toluene, C8 aromatics and C9 aromatics were 1.62 ± 0.18 ppb, 2.51 ± 0.28 ppb,



Fig. 8.7 Mean tropospheric columnar (a) CO and (b) NO₂ $(0.25^{\circ} \times 0.25^{\circ})$ for November 2018 as observed by TROPOMI satellite. The data with quality assurance value higher than 0.5 are considered

 3.72 ± 0.41 ppb, 2.88 ± 0.30 ppb and 1.55 ± 0.19 ppb, respectively, during the post-harvest periods, which were roughly 1.5 times higher than their annually averaged concentrations (Chandra and Sinha 2016).

Model simulations showed that the forest fire events over the foothills of Himalayas during spring season caused 34–52% enhancement in CO and 32–52% enhancement in NO_x over Dehradun (Yarragunta et al. 2020). Satellite observations for these events showed an increase in CO by 60-125 ppby in the upper troposphere, 925–700 hPa (Srivastava and Kumar 2020). Average background and fire-impacted tropospheric column NO₂ over Dehradun were observed be to $1.7 \times 10^{15} \pm 5.0 \times 10^{14}$ molecules cm⁻² and $3.0 \times 10^{15} \pm 8.5 \times 10^{14}$ molecules cm⁻². respectively (Srivastava and Kumar 2020). Figure 8.7 shows monthly mean tropospheric columnar CO and NO₂ gridded at $0.25^{\circ} \times 0.25^{\circ}$ during November 2018 (as representative of post-monsoon) as observed by Tropospheric Monitoring Instrument (TROPOMI) aboard Sentinel-5 Precursor satellite (Veefkind et al. 2012). Extensive biomass burning over the northwest IGP creates hotspots of CO and NO₂ with the elevated levels above 2.7×10^{18} and 3.5×10^{15} molecules cm⁻², respectively (Fig. 8.7). Due to shorter lifetime of NO₂, the influence of emission remains in the close vicinity of source as can be seen from the figure. However, due to relatively longer lifetime of CO, spread from the source region is larger. The biomass burning CO hotspot (northwest IGP) represents highest values as compared to other parts of the Indian region.

IGP experiences extreme pollution episodes frequently during post-monsoon and winter leading to elevated levels of trace gases and aerosols. An extremely high CO level up to 10 mg m⁻³ (~8700 ppbv) was observed during November 2017 over IGP, including the cities of Delhi, Agra, Kanpur, Lucknow, Patna and Kolkata. Dekker et al. (2019) showed that residential and commercial combustion was a much more important source of CO pollution than the post-monsoon crop burning during this episode, which is in contrast to what media suggested and some studies on aerosol emissions found. Besides the elevated emissions, unfavourable meteorological

parameters caused extreme levels of pollution over the IGP (Dekker et al. 2019). Similarly, recent studies (Kanawade et al. 2020; Ojha et al. 2020; Sembhi et al. 2020) emphasized on the role of meteorology to cause extreme pollution episodes over Delhi and IGP.

8.5 Impacts of Intense Biomass Burning

8.5.1 Air Quality

The large amounts of gaseous (CO₂, CH₄, CO, O₃, NO_x, VOCs) and particulate (PM2.5, BC, OC, EC) pollutants emitted into the atmosphere from extensive crop residue burning episodes over northwest India severely affect the local and regional air quality (Singh and Kaskaoutis 2014; Jethva et al. 2019). The highly polluted air over northern India has an adverse impact on atmosphere, crop, ecosystem and human health. The 2016 report of the World Health Organization listed Delhi and other cities in IGP amongst the highly polluted cities in the world. Every year, high levels of suspended particulate matter, contributed from the stubble burning smoke plumes, blanket the IGP causing severe haze pollution. These pollutant particles under conducive meteorological conditions also contribute to the formation of fog or smog over the northern and central India (Jenamani 2007; Ghude et al. 2017; Wang and Chen 2019). Low-intensity winds ($< 5-6 \text{ m s}^{-1}$) and shallow boundary layer favour persistent fog conditions, while prevalence of BC particles leads to smog (smoke + fog) formation.

Delhi, the national capital region (NCR), is affected by intense pollution episodes associated with agricultural fire emissions from northwest India (Cusworth et al. 2018; Liu et al. 2018; Bray et al. 2019; Ravindra et al. 2019c; Beig et al. 2020; Kanawade et al. 2020 and references therein). While NAAQS for PM2.5 is 60 μ g m⁻³ (daily average), the WHO standard for safe PM2.5 levels is 25 μ g m⁻³. During the October-November-December, over Delhi, the daily averages of PM2.5 concentrations are in the range of 100-200 μ g m⁻³, i.e. approximately an order higher than the safety standard, while occasionally exceeding 500 μ g m⁻³. The intense haze episode during October-November 2016 over IGP yielded exceptionally high concentrations of PM2.5 (440 \pm 265 µg m⁻³, 2-week average) at Delhi (Jethva et al. 2019), with the PM2.5 mass concentrations increasing to \sim 563 µg m⁻³ during 1-7 November 2016 (Kanawade et al. 2020). Severe smog during this episode reduced air visibility to about 5 km and less (Sawlani et al. 2019). Analysing contributions of upstream agricultural burning to pollution in Delhi, Liu et al. (2018) noted that for per unit rise in fire radiative power within the upwind region, PM10 increases by 16.34 μ g m⁻³ while visibility decreases by 0.155 km. Amongst several other severe air pollution episodes over Delhi, the event of 28 October-5 November 2019 is particularly noticeable since the PM2.5 concentrations exceeded 900 μ g m⁻³, even for a 6-h average period (Takigawa et al. 2020). Satellite observations showed that these severe pollution events occur at a much larger spatial scale and are attributed to the outflow of stubble burning smoke plumes aided by atmospheric winds (Kanawade et al. 2020; Takigawa et al. 2020). The percentage contribution of large-scale post-harvest open burning to severe haze pollution in Delhi has been observed to be $\sim 42 \pm 17\%$ (Bikkina et al. 2019). Furthermore since 2010, delayed timing of agricultural waste combustion activity in the Punjab area might have led to a gradual rise in the severity of smog episodes over Delhi in the recent years (Sawlani et al. 2019). Transport of crop residue burning of carbonaceous aerosols by north-westerly winds across IGP, particularly during the non-monsoon season, results in poor air quality over rural areas in Lumbini, located at east-central IGP (Chen et al. 2020a).

The NAAQS for 8-h average O_3 is 100 µg m⁻³ (~50 ppbv at normal temperature and pressure—NTP). Limited observations of 80 days in the Doon Valley show that MDA8 (maximum of 8-h running mean O_3 values) index frequently (47 days) crosses the WHO guideline value (50 ppbv) during spring (Ojha et al. 2019). It was also observed that O_3 exceeds the NAAQS standards on 40% and 53% of the days over Delhi and Agra, respectively (Kumari et al. 2018), during the post-harvest period. The site of Mohali experiences elevated O_3 crossing 8-h average NAAQS limit for 62% of days (451 days out of 2 years of observations; Kumar et al. 2016). 307 events of O₃ exceeding 180 μ g m⁻³ (i.e. 1-h O₃ NAAQS limit) were also observed during the polluted springtime over Mohali. In addition, the annual average concentrations of benzene, a class A carcinogen, exceeded the annual exposure NAAQS limit of 1.6 ppb at Mohali, whereas annual exposure to isocyanic acid was close to 1 ppb (Chandra and Sinha 2016). Unlike intense biomass burning during spring/autumn, wintertime biomass burning influences ambient VOC composition over Delhi during fog days with \geq 30% decrease in acetaldehyde, toluene, sum of C8 aromatics (e.g. xylenes) and sum of C9 aromatics (e.g. trimethyl benzenes) and 20% increase in acetonitrile and benzene. The lower temperatures during fog tended to cause an emission feedback from enhanced open biomass burning within Delhi for warming (Hakkim et al. 2019).

The deterioration of air quality by gaseous and particulate pollutants emitted from the stubble burning is a threat to public health and affects the lives of over 900 million people residing in the IGP (Singh and Kaskaoutis 2014; Jethva et al. 2019). The emitted particles travel downstream to urban areas with high population density and combine with local emissions and meteorology to produce dense haze/fog/smog episodes. This results in a drastic reduction in visibility and severely affects day-today life of the residents of the North Indian urban conglomerates and surrounding rural areas. In the recent years, implementation of policies regarding groundwater conversation has resulted in a shift in the crop residue burning period to a narrow window in the late fall with a 39% increase in maximum fire activity during unfavourable environmental conditions, further aggravating the poor air quality of the region (Balwinder-Singh et al. 2019). Exposure to hazardous levels of respirable fine particulate matter released from agricultural waste burning over northwest India is a consequential threat to public health. The adverse effects of aerosol emissions from crop residue burning are pronounced on respiratory health (Chakrabarti et al. 2019; Sonwani et al. 2022) and child health over northwest India (Singh Balwinder-Singh et al. 2019). However, elevated levels of trace gases like benzene and

isocyanic acid have the potential to enhance risks for cardiovascular diseases and cataracts (Chandra and Sinha 2016). Furthermore, biomass burning emissions after aging processes go through photochemical reactions with hydroxyl radicals to produce secondary organic aerosols, which is another concern for researchers owing to its harmful impact on health (Cheng et al. 2013; Bosch et al. 2014; Dumka et al. 2017).

8.5.2 Chemistry

The rate of change of O₃ calculated during the O₃ build-up hours (08:00–11:00 LT) is referred to as O_3 production rate here. It is reported to be 8.6, 8 and 8 ppbv h⁻¹ over Delhi, Agra and Mohali during the preharvest period of post-monsoon (Kumar et al. 2016; Kumari et al. 2018). These rates are higher as compared to other urban (Ahmedabad) and rural (Gadanki) sites likely due to enhanced biogenic emissions during these seasons. Biomass burning intensified the production rate by 3.4, 4.5 and 1.2 ppbv h^{-1} reaching to 12, 12.5 and 9.2 ppbv h^{-1} , respectively, during the postharvest period of autumn. The increase in the production rate is observed to be 39%, 56% and 15% over Delhi, Agra and Mohali, respectively, due to upwind biomass burning. Similarly, the higher production rate (~9.6 ppbv h^{-1}) was observed over Dehradun during May month as compared to that (~9.6 ppbv h^{-1}) during March 2018 (Ojha et al. 2019). It is worth noticing that this enhancement in production rate is partially due to changes in meteorological conditions (e.g. higher solar radiation and temperature during May as compared to March). Sensitivity simulations using the photochemical box model as well as global chemistry transport model showed that O_3 chemistry is NO_x limited over a semi-urban site, Pantnagar (29.0° N, 79.5° E, 231 m amsl), influenced by biomass burning over Haryana–Punjab. In contrast, box model simulations suggested VOC-limited O₃ chemistry over Delhi during November (Chen et al. 2020b). This suggests the heterogeneity of chemistry over nearby locations. While the rate of change of O₃ is studied well in the downwind of crop residue burning, observations of such changes in precursor gases are sparse over northern India. The rate of increase in surface CO was found to be 45 ppbv·day⁻¹ in the downwind of a widespread forest fire episode (April–May 2016) over Himalayas (Srivastava and Kumar 2020).

Variety of VOCs emitted from biomass burning influences the chemistry of the region which is inferred from the OH reactivity. The OH reactivity is observed to increase from 28 to 64 s⁻¹ (~120%) at Mohali in the downwind of crop residue burning over northwest IGP (Kumar et al. 2018). There is no missing OH reactivity when air masses are not influenced by biomass burning; however, missing OH reactivity increased to 40% at Mohali under the influence of crop residue burning, highlighting the presence of unmeasured compounds (Kumar et al. 2018). O₃ levels up to 100 ppbv would cause higher level of OH radicals which could oxidize the precursor gas and enhance the secondary formation of aerosols. Extensive measurements as well as model-based estimation of OH concentration and reactivity

are required over northwest India to better understand the changes in chemistry in the downwind of biomass burning.

8.5.3 Climate and Weather

Pollutants emitted from biomass burning have a prominent impact on global and regional climate. A large amount of trace gases and aerosol particles produced from forest fires, burning of fuelwood and agricultural waste influence Earth's radiative budget (IPCC 2013; Bond et al. 2013a, b), and are key contributors to changes in the hydrological cycle and climate (Robock 1991; Penner et al. 1992; Randerson et al. 2006; Keywood et al. 2013; Tosca et al. 2013; Liu et al. 2014). On a global scale, human-induced forest fires for land acquisition may account for up to 20-60% of the global warming caused by CO₂ emitted from fossil fuel combustion (Crutzen and Andreae 1990). The smoke aerosols produced directly, and as secondary products, from biomass burning alter the atmospheric radiative balance by absorbing, scattering and reflecting solar radiation (direct effect; Fadnavis et al. 2020). The particles also act as cloud condensation nuclei (CCN) and ice nuclei (IN) and impact cloud optical and microphysical properties (indirect effect; Cattani et al. 2006; Koren et al. 2008). Aerosols emitted from biomass burning have significantly different characteristics while acting as CCN. The mean CCN activation ratio (at a supersaturation of (0.46) was observed to be highest (>0.7) at high-altitude station, Nainital, during autumn which is attributed to the contribution of biomass-burning aerosols to CCN formation at high supersaturation conditions (Gogoi et al. 2015). In addition, the optical properties of smoke aerosols are constrained by the burning conditions; consequently, depending on the vertical variation of smoke plumes, they alter cloud cover, albedo and precipitation efficiency (Kaskaoutis et al. 2014). Owing to their short lifetime and heterogeneous distribution, the biomass burning aerosols primarily impact regional climate. Although the global mean radiative forcing from biomass burning is estimated to be 0.0 (-0.2 to 0.2) W m⁻² (IPCC 2013), the BC particles in smoke plumes strongly absorb incoming solar radiation resulting in heating of the lower to middle atmospheric layers influencing the vertical temperature profile and reducing surface radiation (Hansen et al. 1997; Ackerman et al. 2000; Krishnan and Ramanathan 2002; Randerson et al. 2006; Ramanathan and Carmichael 2008). Deposition of BC on snow and ice also affects climate by lowering surface albedo and melting snow/ice (IPCC 2013). Over South Asia, absorbing carbonaceous aerosols has a profound impact on the regional hydrological cycle through changes in atmospheric heating, cloud cover and rainfall, and is strongly linked to the vertical distribution of the aerosols and background environmental conditions (Ramanathan et al. 2005; Lau et al. 2006; Li et al. 2016).

8.5.3.1 Impact on Aerosol Characteristics and Radiative Forcing

The paddy residue burning during post-monsoon season (October–November) produces very high values of AOD (>1) and an increase in angstrom exponent value (>1.2), indicating elevated concentrations of fine-mode particles during the



Fig. 8.8 Agricultural fires and smoke plumes over northern India on 27 October 2020 and 5 November 2020 as observed from Suomi NPP/VIIRS (image generated using NASA worldview)

biomass burning activity over northwest India (Sharma et al. 2010, 2011, 2017; Kharol et al. 2012; Kaskaoutis et al. 2014). In the recent years (2003–2017), during the post-monsoon months there has been a 4% rise in crop burning activity over Haryana–Punjab which resulted in 8%, 9% and 11% increase in AOD, absorbing AOD and aerosol index, respectively (Shaik et al. 2019). Figure 8.8 shows a snapshot of intense agricultural crop waste burning activity (red dots) predominantly over Haryana–Punjab region. The associated dense smoke plumes (in white) are also seen to spread in the downwind region along the length of the IGP during post-harvest season of 2020.

Sharma et al. (2010) noted a 21% increase in aerosol loading over Patiala for October 2007 compared to October 2006. Also, Sharma et al. (2011) observed very high AOD of ≈ 2.65 (at 500 nm) in October 2008 compared to ≈ 1.71 in November 2009, signifying intense crop residue burning during the rice harvest season of 2008.

The study also noted that the substantial variations in the daily BC mass concentration resulted in single scattering albedo (SSA) value ranging from 0.76–0.88 during highly turbid days to 0.95–0.97 for less turbid days. Analysing the aerosol properties during the post-monsoon crop residue burning season of 2016 over several locations spread across IGP, Kaskaoutis et al. (2014) noted that the AOD increase over Patiala has a greater influence on the angstrom exponent values (increase) and fine-mode radii (decrease) compared to the fine/coarse ratio (increase), indicating an additional contribution of newly formed soot particles from rice harvest fires. The shortwave radiative forcing (RF) estimates over Patiala (Sharma et al. 2010, 2011, 2017) during the biomass burning periods of October–November show large reductions in the surface shortwave radiation, varying from -57.2 to -212.1 W m⁻² for highly turbid days. The shortwave RF at the top of the atmosphere (TOA) varied from 0.17 to -47.67 W m⁻². The considerable reduction in the surface forcing results in a strong atmospheric aerosol RF in the range of 43–164.4 W m⁻², suggesting the predominance of absorbing aerosols from paddy residue burning. The strong surface cooling

and large atmospheric warming result in an atmospheric heating rate cooling and large atmospheric warming result in an atmospheric heating rate of 1.7 ± 0.6 K day⁻¹ over Patiala during the post-monsoon season (Sharma et al. 2017). The BC-induced elevated atmospheric heating results in increased air temperatures at mid-tropospheric (700 hPa) level over the region (Kharol et al. 2012). WRF-Chem simulations for 2013 over Punjab also yield similar results, with carbonaceous aerosols from open biomass burning producing strong reduction of shortwave radiation at the surface resulting in a negative change in the surface temperatures (Singh et al. 2020a).

Transport of smoke particles from Punjab biomass burning region via low-intensity north-westerly winds strongly influences the accumulation of fine particulate matter over New Delhi (Bisht et al. 2015). The high atmospheric RF (~60–70 W m⁻²) due to BC from the post-monsoon extreme crop residue burning events contributes to the large positive RF (~18 W m⁻²) at TOA over Delhi, resulting in high atmospheric heating rates (1.9–2.0 K day⁻¹). Furthermore, transport of smoke aerosols from the biomass burning hotspots over northern India occurs in the downwind direction, across the entire IGP extending from east to west, to central peninsular India, north-eastern Arabian sea and occasionally onto Bay of Bengal (Badarinath et al. 2009a, 2009b; Mishra and Shibata 2012; Kaskaoutis et al. 2014; Das et al. 2015; Singh et al. 2018; Shaik et al. 2019). During the intense biomass burning season of 2012, satellite observations revealed thick smoke plumes in the lower troposphere (below $\sim 2-2.5$ km) spreading across the entire IGP (Kaskaoutis et al. 2014). As the vertical profiles of crop residue burning aerosols show an exponential decay with height, Mishra and Shibata (2012) obtained scale heights for smoke aerosols in the range of 1.25-1.66 km. Although predominantly biomass burning smoke plumes are trapped inside the planetary boundary layer, in some cases it extends vertically up to 2-3 km (Shaik et al. 2019). Air mass trajectory analyses from the province of Punjab (Kaskaoutis et al. 2014) showed the transport of the smoke-laden plumes by the north-westerly winds predominantly along western and eastern IGP, and sometimes across central India and northeast Arabian sea. Figure 8.9 shows 5-day forward trajectories, obtained from HYSPLIT (Hybrid



Fig. 8.9 5-Day forward air mass trajectory pathways and their travelling heights originating at a matrix, 29.5–31.5°N, 74–76°E (marked by stars), located over Punjab during (**a**) 27 October 2020 and (**b**) 5 November 2020

Single-Particle Lagrangian Integrated Trajectory) model (Draxler and Rolph 2003), originating from a matrix located over Punjab (29.5–31.5° N, 74–76° E) during the recent biomass burning events of 27 October 2020 and 5 November 2020. The altitude of the start point of the trajectories was fixed at 937 hPa (~500 m). The Global Forecast System (GFS) meteorological fields including the model vertical wind velocities were used for calculating the forward trajectories. Under the prevailing meteorological conditions on both the days, biomass burning plumes spread across north/north-western India, and also propagate vertically up to an altitude of 750 hPa. Singh et al. (2018) investigated the transport of extreme biomass burning emissions (October-December) of 2016 and observed the predominance of absorbing aerosols with a low smoke plume height (<1.5 km) across IGP. Owing to the vertical transport, significant enhancement in absorbing aerosol loading is observed in the free troposphere during spring (Nair et al. 2016; Gogoi et al. 2019). Radiative forcing estimates that the biomass burning-dominated period caused a significant increase in atmospheric RF (135 W m⁻²) and very high atmospheric heating rates (4.3 K day⁻¹) over Varanasi in comparison to the non-dominating period (56 W m⁻², 1.8 K day⁻¹; Singh et al. 2018). Long-range transport of biomass emission aided by circulation patterns and environmental conditions across central India resulted in large deficit in surface solar radiation $(-107.81 \text{ W m}^{-2})$ over south-central India (Badarinath et al. 2009a). Although northwest IGP is the hotspot for biomass burning, considering several weeks of burning activity and large-scale transport of emissions through prevailing winds, the smoke plumes have significant implications on regional climate through changes in aerosol RF and cloud modifications.

8.5.3.2 Impact over Himalayan Region

Light-absorbing aerosols get transported and lifted to the high mountains over northern India and are deposited on glaciers and snow, affecting the snow albedo over Himalayas (Fadnavis et al. 2020). Deposition of absorbing aerosols on snow decreases the surface albedo resulting in increased absorption of shortwave radiation enhancing the surface snowmelt. Crop residue burning emissions from northwest India and Himalayan forest fire emissions are known to be the sources of carbonaceous aerosols over various sites located across the Himalayan region (Kumar et al. 2011; Srivastava et al. 2012; Vadrevu et al. 2012; Sahai et al. 2013; Sarkar et al. 2015; Kant et al. 2020; Prabhu et al. 2020). Furthermore, You et al. (2016) showed that smoke plumes from intense biomass burning episode over the northern India can be lifted onto upper tropospheric levels and can get transported and deposited at high-altitude glaciers over the Tibetan Plateau. The springtime biomass burning over northern India during 2007-2009 resulted in significant enhancements in surface BC (~145%), 550 nm AOD (150%) and ozone (~34%) at Manora Peak, Nainital, located in the central Himalayan region (Kumar et al. 2011). Wavelength dependence of the BC observations shows that significant contribution of BC at Manora Peak occurs from biomass burning and is transported by air masses from the west and northwest sector onto central Himalayas (Srivastava et al. 2012). The northern Indian biomass burning-induced shortwave RF shows large reduction of surface solar radiation (-27 W m^{-2}) compared to TOA (-8 W m^{-2}) in high-altitude regions of the central Himalayas (Kumar et al. 2011). The surface cooling leads to an increase in atmospheric RF (19 W m⁻²), thus enhancing the lower atmospheric heating rate by 0.8 K day^{-1} . The radiative changes caused by the transport of biomass burning aerosols to the central Himalayan atmosphere may produce enhanced shortwave absorption above clouds and have an impact on the monsoon precipitation.

8.6 Summary

The IGP is a very important agroecological region which holds nearly ~40% of India's total population. The northwest IGP, comprising Haryana, Punjab and west Uttar Pradesh, produces two-thirds of India's food grains, and is commonly referred to as the "breadbasket" of India. The total number of annual fire counts over northwest IGP accounts for ~26% of total fire counts over India, manifesting that biomass burning over Haryana–Punjab region is more extensive as compared to the rest of India, and hence represents the regional hotspot over northern India. The most prominent source of biomass burning over this region is post-harvest crop residue burning, while other important sources of biomass burning include wildfires and biofuel combustion. The agricultural waste generated, as a result of rice and wheat harvesting, is burnt during post-monsoon/autumn (October–November) and pre-monsoon/spring (April–May) seasons, respectively. However, significantly high fire counts are observed in post-monsoon which is indicative of intense biomass burning events occurring predominantly during these months. Burning of biomass releases a variety of particulate matter/aerosols such as carbonaceous aerosols, and

trace gases such as CO, NO_x, SO₂ and NMHCs. Intense biomass burning under unfavourable meteorological conditions (e.g. shallower boundary layer and lower ventilation coefficient) contributes significantly to the extreme pollution episodes during late post-monsoon season over the downwind region of northwest IGP. Elevated levels of ambient PM2.5 (>120 μ g m⁻³), BC (>10 μ g m⁻³), OC/EC ratios (>5) and O₃ (>45 ppbv) and its precursors (NO_x, CO, VOCs, etc.) are observed mostly during post-monsoon crop residue burning season. However, in contrast, O₃ is higher during pre-monsoon biomass burning as compared to that during postmonsoon owing to higher solar insolation. Biomass burning also impacts the chemistry of region, as can be seen from the higher O₃ production rates and increased OH reactivity over a region under the influence of biomass burning.

The excessive emission of trace gases and particulate matter causes severe air pollution incidences over the province of Punjab and adjoining areas, particularly the NCR. The stubble burning smoke plumes get transported horizontally as well as vertically under the influence of the prevailing circulation patterns and environmental conditions, thereby affecting the entire IGP region. In situ measurements, supplemented with space-based observations, air mass trajectory analysis and chemistry-climate modelling over northern India, have unravelled the impacts of biomass burning on atmosphere. The biomass burning-induced trace gases and aerosol particles, coupled with regional meteorology, have a significant impact on air quality, chemistry as well as weather and climate of a region. Notably, the lightabsorbing BC and CCN forming OC aerosols from the thick smoke plumes have a profound impact on circulation and precipitation patterns, glacier melt and atmospheric heating over the South Asian region. With a projected rise of surface temperatures over the Himalayan region by the end of the twenty-first century, the climate community is striving to constrain any significant snowmelt and glacier retreat over the Himalayas, and also to limit the global temperature rise to 1.5 °C. For this, a massive reduction in carbon dioxide and short-lived climate forcers (methane, ozone and aerosols: particularly carbonaceous aerosols) is necessary, in order to bring about a sustainable climate over the South Asian domain. Since crop residue burning continues to play a crucial role in the problem of seasonal air pollution, effective alternative solutions through coordinated public and private actions need to be implemented to reduce stubble burning, along with more sustainable agricultural management practices. This will include educating the farming community and empowering them with technical and socio-economical assistance. Shared socioeconomic pathways (SSPs), as described by the climate change research community, have indicated that socio-economic factors will affect the amount of greenhouse gas emissions. It suggests that a sustainability-focussed world will lead to quick reduction of emissions beyond those agreed in the Paris Agreement.

Acknowledgement The authors acknowledge the use of MODIS fire detection data from ftp:// fuoco.geog.umd.edu/modis/C6/mcd14ml/; ERA5 wind data from ECMWF (European Centre for Medium range Weather Forecasting; https://cds.climate.copernicus.eu/cdsapp#!/dataset/reanalysisera5-pressure-levels-monthly-means?tab=form); tropospheric NO₂ and CO from TROPOMI (https://disc.gsfc.nasa.gov); and CAMS-simulated ozone from http://www.copernicus.eu/main/ atmosphere-monitoring. The smoke plume images embedded with fire events were obtained from Suomi NPP/VIIRS NASA worldview imageries (https://earthdata.nasa.gov/earth-observation-data/near-real-time/hazards-and-disasters/smoke-plumes). NOAA Air Resources Laboratory (ARL) is acknowledged for the HYSPLIT model and READY website (https://www.arl.noaa.gov/hysplit/hysplit/).

References

- Ackerman AS et al (2000) Reduction of tropical cloudiness by soot. Science 288(5468):1042–1047. https://doi.org/10.1126/science.288.5468.1042
- Agarwal R et al (2012) Effects of exposure to rice-crop residue burning smoke on pulmonary functions and Oxygen Saturation level of human beings in Patiala (India). Sci Total Environ 429:161–166. Elsevier
- Aggarwal A, Paul V, Das S (2009) Forest resources: Degradation, livelihoods, and climate change, Looking back to change track. TERI New Delhi 219:91–108
- Andreae MO (1983) Soot carbon and excess fine potassium: long-range transport of combustionderived aerosols. Science 220(4602):1148–1151. https://doi.org/10.1126/science.220.4602. 1148
- Andreae MO (1991) Biomass burning: its history, use, and distribution and its impact, global biomass burning: atmospheric, climatic, and biospheric implications. MIT Press, Cambridge, MA, pp 3–21
- Andreae MO (2019) Emission of trace gases and aerosols from biomass burning—an updated assessment. Atmos Chem Phys Discuss 19(13):8523–8546. https://doi.org/10.5194/acp-2019-303
- Andreae MO, Gelencsér A (2006) Black carbon or brown carbon? The nature of light-absorbing carbonaceous aerosols. Atmos Chem Phys 6(10):3131–3148. https://doi.org/10.5194/acp-6-3131-2006
- Andreae MO, Merlet P (2001) Emission of trace gases and aerosols from biomass burning. Glob Biogeochem Cycles 15(4):955–966. Wiley Online Library
- Apte JS et al (2015) Addressing global mortality from ambient PM2. 5. Environ Sci Technol 49(13):8057–8066. ACS Publications
- Badarinath KVS, Chand TRK, Prasad VK (2006) Agriculture crop residue burning in the Indo-Gangetic Plains–a study using IRS-P6 AWiFS satellite data. Curr Sci:1085–1089. JSTOR
- Badarinath KVS, Kharol SK, Sharma AR (2009a) Long-range transport of aerosols from agriculture crop residue burning in Indo-Gangetic Plains—a study using LIDAR, ground measurements and satellite data. J Atmos Sol Terr Phys 71(1):112–120. Elsevier
- Badarinath KVS et al (2009b) Analysis of aerosol and carbon monoxide characteristics over Arabian Sea during crop residue burning period in the Indo-Gangetic Plains using multisatellite remote sensing datasets. J Atmos Sol Terr Phys 71(12):1267–1276. Elsevier
- Bahuguna VK, Upadhay A (2002) Forest fires in India: policy initiatives for community participation. Int For Rev Commonwealth For Assoc 4(2):122–127
- Balwinder-Singh et al (2019) Trade-offs between groundwater conservation and air pollution from agricultural fires in northwest India. Nat Sustain 2(7):580–583. https://doi.org/10.1038/s41893-019-0304-4
- Bano T et al (2011) Variation in aerosol black carbon concentration and its emission estimates at the mega-city Delhi. Int J Remote Sensing 32(21):6749–6764. Taylor & Francis
- Bansal O, Singh A, Singh D (2019) Characteristics of black carbon aerosols over Patiala Northwestern part of the IGP: source apportionment using cluster and CWT analysis. Atmos Pollut Res 10(1):244–256. https://doi.org/10.1016/j.apr.2018.08.001
- Beig G et al (2020) Objective evaluation of stubble emission of North India and quantifying its impact on air quality of Delhi. Sci Total Environ 709:136126. https://doi.org/10.1016/j. scitotenv.2019.136126

- Bhattacharya B, Nanda SK (1992) Building fuelwood demand-supply scenario. J Rural Dev (Hyderabad) 11(6):773–787
- Bikkina S et al (2019) Air quality in megacity Delhi affected by countryside biomass burning. Nat Sustain 2(3):200–205. https://doi.org/10.1038/s41893-019-0219-0
- Bisht DS et al (2015) Carbonaceous aerosols and pollutants over Delhi urban environment: temporal evolution, source apportionment and radiative forcing. Sci Total Environ 521–522: 431–445. https://doi.org/10.1016/j.scitotenv.2015.03.083. Elsevier B.V.
- Bisht DS et al (2016) Tethered balloon-born and ground-based measurements of black carbon and particulate profiles within the lower troposphere during the foggy period in Delhi, India. Sci Total Environ 573:894–905. https://doi.org/10.1016/j.scitotenv.2016.08.185
- Boer MM, Resco de Dios V, Bradstock RA (2020) Unprecedented burn area of Australian mega forest fires. Nat Clim Chang 10(3):171–172. https://doi.org/10.1038/s41558-020-0716-1
- Bond, T. C. et al. (2004) A technology-based global inventory of black and organic carbon emissions from combustion, J Geophys Res Atmos 109(D14). doi: https://doi.org/10.1029/ 2003JD003697 John Wiley & Sons, Ltd
- Bond TC et al (2007) Historical emissions of black and organic carbon aerosol from energy-related combustion, 1850–2000. Glob Biogeochem Cycles 21(2) Wiley Online Library
- Bond TC et al (2013a) Bounding the role of black carbon in the climate system: a scientific assessment. J Geophys ResAtmos 118(11):5380–5552. Wiley Online Library
- Bond TC et al (2013b) Bounding the role of black carbon in the climate system: a scientific assessment. J Geophys Res Atmos 118(11):5380–5552. https://doi.org/10.1002/jgrd.50171
- Bosch C et al (2014) Source-diagnostic dual-isotope composition and optical properties of watersoluble organic carbon and elemental carbon in the South Asian outflow intercepted over the Indian Ocean. J Geophys Res Atmos 119(20):11–743. Wiley Online Library
- Brauer M, Hisham-Hashim J (1998) Peer reviewed: fires in Indonesia: crisis and reaction. Environ Sci Technol 32(17):404A–407A. ACS Publications
- Bray CD, Battye WH, Aneja VP (2019) The role of biomass burning agricultural emissions in the Indo-Gangetic Plains on the air quality in New Delhi, India. Atmos Environ 218:116983. https:// doi.org/10.1016/j.atmosenv.2019.116983
- Cattani E et al (2006) Influence of aerosol particles from biomass burning on cloud microphysical properties and radiative forcing. Atmos Res 82(1):310–327. https://doi.org/10.1016/j.atmosres. 2005.10.010
- Chakrabarti S et al (2019) Risk of acute respiratory infection from crop burning in India: estimating disease burden and economic welfare from satellite and national health survey data for 250,000 persons. Int J Epidemiol 48(4):1113–1124. https://doi.org/10.1093/ije/dyz022
- Chand TRK et al (2007) Active forest fire monitoring in Uttaranchal State, India using multitemporal DMSP-OLS and MODIS data. Int J Remote Sensing 28(10):2123–2132. https://doi. org/10.1080/01431160600810609. Taylor & Francis
- Chandra BP, Sinha V (2016) Contribution of post-harvest agricultural paddy residue fires in the N.W. Indo-Gangetic Plain to ambient carcinogenic benzenoids, toxic isocyanic acid and carbon monoxide. Environ Int 88:187–197. https://doi.org/10.1016/j.envint.2015.12.025. Elsevier Ltd
- Chauhan A, Singh RP (2017) Poor air quality and dense haze/smog during 2016 in the Indo-Gangetic plains associated with the crop residue burning and Diwali festival. In: 2017 IEEE International Geoscience and Remote Sensing Symposium (IGARSS). IEEE, pp 6048–6051
- Chen J et al (2017) A review of biomass burning: Emissions and impacts on air quality, health and climate in China. Sci Total Environ 579:1000–1034. Elsevier
- Chen P et al (2020a) Severe air pollution and characteristics of light-absorbing particles in a typical rural area of the Indo-Gangetic Plain. Environ Sci Pollut Res 27(10):10617–10628. https://doi.org/10.1007/s11356-020-07618-6
- Chen Y et al (2020b) Avoiding high ozone pollution in Delhi, India. Faraday Discuss. https://doi. org/10.1039/d0fd00079e
- Cheng Y et al (2013) Biomass burning contribution to Beijing aerosol. Atmos Chem Phys 13(15): 7765–7781

- Chowdhury Z et al (2007) Speciation of ambient fine organic carbon particles and source apportionment of PM2. 5 in Indian cities. J Geophys Res Atmos 112(D15) Wiley Online Library
- Crutzen PJ, Andreae MO (1990) Biomass burning in the tropics: impact on atmospheric chemistry and biogeochemical cycles. Sci Am Associ Advance Sci 250(4988):1669–1678
- Cusworth DH et al (2018) Quantifying the influence of agricultural fires in northwest India on urban air pollution in Delhi, India. Environ Res Lett 13(4):44018. IOP Publishing
- Das SK et al (2015) An integrated campaign for investigation of winter-time continental haze over Indo-Gangetic Basin and its radiative effects. Sci Total Environ 533:370–382. https://doi.org/ 10.1016/j.scitotenv.2015.06.085
- Dekker IN et al (2019) What caused the extreme CO concentrations during the 2017 high-pollution episode in India? Atmos Chem Phys 19(6):3433–3445. https://doi.org/10.5194/acp-19-3433-2019
- Dickerson RR et al (2002) Analysis of black carbon and carbon monoxide observed over the Indian Ocean: implications for emissions and photochemistry. J Geophys Res Atmos 107(D19): INX2-16. Wiley Online Library
- Draxler RR, Rolph GD (2003) HYSPLIT (HYbrid Single-Particle Lagrangian Integrated Trajectory) model access via NOAA ARL READY website (http://www.arl.noaa.gov/ready/hysplit4. html). NOAA Air Resources Laboratory, Silver Spring
- Dumka UC et al (2017) Assessment of PM 2.5 chemical compositions in Delhi: primary vs secondary emissions and contribution to light extinction coefficient and visibility degradation. J Atmos Chem 74(4):423–450. Springer
- Duncan BN et al (2003) Indonesian wildfires of 1997: impact on tropospheric chemistry. J Geophys Res Atmos 108(D15) Wiley Online Library
- Fadnavis S et al (2020) Atmospheric aerosols and trace gases. In: Assessment of climate change over the Indian Region: A report of the Ministry of Earth Sciences (MoES), Government of India. Springer, pp 93–116. https://doi.org/10.1007/978-981-15-4327-2_5
- Fang M et al (1999) The solvent-extractable organic compounds in the Indonesia biomass burning aerosols-characterization studies. Atmos Environ 33(5):783–795. Citeseer
- FSI (2015) Indian State of forest report, 2015, Ministry of environment, Forest and climate change, Government of India, Dehradun, p 300
- Forster P et al (2007) Chapter 2: Changes in atmospheric constituents and in radiative forcing. In: Climate Change 2007. The Physical Science Basis
- Gaur A et al (2014) Four-year measurements of trace gases (SO2, NOx, CO, and O3) at an urban location, Kanpur, in Northern India. J Atmos Chem 71(4):283–301. https://doi.org/10.1007/ s10874-014-9295-8
- Gautam R et al (2007) Influences of winter haze on fog/low cloud over the Indo-Gangetic plains. J Geophys Res Atmos 112(D5) Wiley Online Library
- Gautam R et al (2013) Characterization of aerosols over the Indochina peninsula from satellitesurface observations during biomass burning pre-monsoon season. Atmos Environ 78:51–59. Elsevier
- Ghude SD et al (2017) Winter fog experiment over the Indo-Gangetic plains of India. Curr Sci 112(4):00113891
- Giglio L et al (2003) An enhanced contextual fire detection algorithm for MODIS. Remote Sensing Environ 87(2–3, 273):–282. https://doi.org/10.1016/S0034-4257(03)00184-6. Elsevier Inc
- Girach IA et al (2020) Tropospheric carbon monoxide over the northern Indian Ocean during winter: influence of inter-continental transport. Clim Dyn 54(11–12). https://doi.org/10.1007/ s00382-020-05269-4
- Gogoi MM et al (2015) Optical properties and CCN activity of aerosols in a high-altitude Himalayan environment: results from RAWEX-GVAX. J Geophys Res Atmos 120(6):2453–2469. https://doi.org/10.1002/2014JD022966. John Wiley & Sons, Ltd
- Gogoi MM et al (2019) Seasonal contrast in the vertical profiles of aerosol number concentrations and size distributions over India: implications from RAWEX aircraft campaign. J Earth Syst Sci 128(8):225. https://doi.org/10.1007/s12040-019-1246-y

- Goss M et al (2020) Climate change is increasing the likelihood of extreme autumn wildfire conditions across California. Environ Res Lett 15(9):94016. IOP Publishing
- Guazzotti SA et al (2003) Characterization of carbonaceous aerosols outflow from India and Arabia: Biomass/biofuel burning and fossil fuel combustion. J Geophys Res Atmos 108(D15) Wiley Online Library
- Hakkim H et al (2019) Volatile organic compound measurements point to fog-induced biomass burning feedback to air quality in the megacity of Delhi. Sci Total Environ 689:295–304. Elsevier BV. https://doi.org/10.1016/j.scitotenv.2019.06.438
- Hansen J, Sato M, Ruedy R (1997) Radiative forcing and climate response Abstract. J Geophys Res 102:6831–6864
- Hao WM, Liu M (1994) Spatial and temporal distribution of tropical biomass burning. Glob Biogeochem Cycles 8(4):495–503. Wiley Online Library
- Husain L et al (2007) Characterization of carbonaceous aerosols in urban air. Atmos Environ 41(32):6872–6883. Elsevier
- IPCC (2013) Climate change 2013. www.climatechange2013.org
- Jacobson MZ (2002) Control of fossil-fuel particulate black carbon and organic matter, possibly the most effective method of slowing global warming. J Geophys Res Atmos 107(D19):ACH-16. Wiley Online Library
- Jacobson MZ (2010) Short-term effects of controlling fossil-fuel soot, biofuel soot and gases, and methane on climate, Arctic ice, and air pollution health. J Geophys Res Atmos 115(D14). https://doi.org/10.1029/2009JD013795. John Wiley & Sons, Ltd
- Jain N, Bhatia A, Pathak H (2014) Emission of air pollutants from crop residue burning in India. Aerosol Air Qual Res 14(1):422–430. Taiwan Association for Aerosol Research
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Jena C et al (2015) Influence of springtime biomass burning in South Asia on regional ozone (O3): a model based case study. Atmos Environ 100:37–47. https://doi.org/10.1016/j.atmosenv.2014. 10.027. Elsevier Ltd
- Jenamani RK (2007) Alarming rise in fog and pollution causing a fall in maximum temperature over Delhi. Curr Sci:314–322. JSTOR
- Jethva H et al (2019) Connecting crop productivity, residue fires, and air quality over northern India. Sci Rep 9(1):1–11. Nature Publishing Group
- Jha CS et al (2016) Monitoring of forest fires from space–ISRO's initiative for near real-time monitoring of the recent forest fires in Uttarakhand, India. Curr Sci 110(11):2057–2060. JSTOR
- Kanawade VP et al (2014) Sub-micron particle number size distributions characteristics at an urban location, Kanpur, in the Indo-Gangetic Plain. Atmos Res 147–148:121–132. https://doi.org/10. 1016/j.atmosres.2014.05.010
- Kanawade VP et al (2020) What caused severe air pollution episode of November 2016 in New Delhi? Atmos Environ 222. https://doi.org/10.1016/j.atmosenv.2019.117125. Elsevier Ltd
- Kant Y et al (2020) Black carbon aerosol quantification over north-west Himalayas: seasonal heterogeneity, source apportionment and radiative forcing. Environ Pollut 257:113446. https://doi.org/10.1016/j.envpol.2019.113446
- Kaskaoutis, D G et al. (2014) Effects of crop residue burning on aerosol properties, plume characteristics, and long-range transport over northern India. J Geophys Res Atmos, 119(9), 5424–5444 Wiley Online Library
- Keywood M et al (2013) Fire in the air: biomass burning impacts in a changing climate. Crit Rev Environ Sci Technol 43(1):40–83. Taylor & Francis
- Kharol SK et al (2012) Black carbon aerosol variations over Patiala city, Punjab, India—a study during agriculture crop residue burning period using ground measurements and satellite data. J Atmos Sol Terr Phys 84–85:45–51. https://doi.org/10.1016/j.jastp.2012.05.013
- Koren I et al (2008) Smoke invigoration versus inhibition of clouds over the Amazon. Science 321(5891):946–949. https://doi.org/10.1126/science.1159185

- Krishnan R, Ramanathan V (2002) Evidence of surface cooling from absorbing aerosols. Geophys Res Lett 29(9):54-1–54-4. https://doi.org/10.1029/2002gl014687
- Kulkarni SH et al (2020) How much does large-scale crop residue burning affect the air quality in Delhi? Environ Sci Technol ACS Publ 54(8):4790–4799
- Kumar R, Naja M, Satheesh SK et al (2011) Influences of the springtime northern Indian biomass burning over the central Himalayas. J Geophys Res Atmos 116(19):1–14. https://doi.org/10. 1029/2010JD015509
- Kumar P, Kumar S, Joshi L (2015) Problem of residue management due to rice wheat crop rotation in Punjab. In: Socioeconomic and environmental implications of agricultural residue burning. Springer, pp 1–12
- Kumar V, Sarkar C, Sinha V (2016) Influence of post-harvest crop residue fires on surface ozone mixing ratios in the N.W. IGP analyzed using 2 years of continuous in situ trace gas measurements. J Geophys Res Atmos 121(7):3619–3633. https://doi.org/10.1002/ 2015JD024308. John Wiley & Sons, Ltd
- Kumar V, Chandra BP, Sinha V (2018) Large unexplained suite of chemically reactive compounds present in ambient air due to biomass fires. Sci Rep 8(1):626. https://doi.org/10.1038/s41598-017-19139-3
- Kumari S et al (2018) Tropospheric ozone enhancement during post-harvest crop-residue fires at two downwind sites of the Indo-Gangetic Plain. Environ Sci Pollut Res 25(19):18879–18893. https://doi.org/10.1007/s11356-018-2034-y
- Kumari S, Lakhani A, Kumari KM (2020) Transport of aerosols and trace gases during dust and crop-residue burning events in Indo-Gangetic plain: influence on surface ozone levels over downwind region. Atmos Environ 241. https://doi.org/10.1016/j.atmosenv.2020.117829. Elsevier Ltd
- Lau KM, Kim MK, Kim KM (2006) Asian summer monsoon anomalies induced by aerosol direct forcing: the role of the Tibetan Plateau. Clim Dyn 26(7–8):855–864. https://doi.org/10.1007/ s00382-006-0114-z
- Lelieveld, J. others et al. (2001) The Indian Ocean experiment: widespread air pollution from South and Southeast Asia. Science, 291(5506), 1031–1036. American Association for the Advancement of Science
- Levine JS (1991) Global biomass burning: atmospheric, climatic, and biospheric implications. MIT Press
- Levine JS et al (1995) A driver for global change. Environ Sci Technol 29(3):120A–125A. ACS Publications
- Li Z et al (2016) Aerosol and monsoon climate interactions over Asia. Rev Geophys 54(4): 866–929. https://doi.org/10.1002/2015RG000500
- Liousse C et al (1996) A global three-dimensional model study of carbonaceous aerosols. J Geophys Res Atmos 101(D14):19411–19432. Wiley Online Library
- Liu Y, Goodrick S, Heilman W (2014) Wildland fire emissions, carbon, and climate: Wildfireclimate interactions. For Ecol Manag 317:80–96. https://doi.org/10.1016/j.foreco.2013.02.020
- Liu T et al (2018) Seasonal impact of regional outdoor biomass burning on air pollution in three Indian cities: Delhi, Bengaluru, and Pune. Atmos Environ 172:83–92. https://doi.org/10.1016/j. atmosenv.2017.10.024
- Manjunatha, K. et al. (2015) Role of Balers in agricultural crop residue management—a review. EM International
- Metz JJ (1991) A reassessment of the causes and severity of Nepal's environmental crisis. World Dev 19(7):805–820. https://doi.org/10.1016/0305-750X(91)90134-4
- Mishra AK, Shibata T (2012) Synergistic analyses of optical and microphysical properties of agricultural crop residue burning aerosols over the Indo-Gangetic Basin (IGB). Atmos Environ 57:205–218. https://doi.org/10.1016/j.atmosenv.2012.04.025
- Mitra AP (1999) INDOEX (India): introductory note. Curr Sci 76(7):886-889. JSTOR

- Nair VS et al (2016) Large-scale enhancement in aerosol absorption in the lower free troposphere over continental India during spring. Geophys Res Lett 43(21):11453–11461. https://doi.org/10. 1002/2016GL070669. John Wiley & Sons, Ltd
- Nawaz MO, Henze DK (2020) Premature deaths in Brazil associated with long-term exposure to PM2.5 from Amazon fires between 2016 and 2019. GeoHealth 4(8):e2020GH000268. https:// doi.org/10.1029/2020GH000268. John Wiley & Sons, Ltd
- Ojha N et al (2012) Variabilities in ozone at a semi-urban site in the Indo-Gangetic Plain region: association with the meteorology and regional processes. J Geophys Res Atmos 117(20):1–19. https://doi.org/10.1029/2012JD017716
- Ojha N et al (2014) On the processes influencing the vertical distribution of ozone over the central Himalayas: analysis of yearlong ozonesonde observations. Atmos Environ 88:201–211. https://doi.org/10.1016/j.atmosenv.2014.01.031. Elsevier Ltd
- Ojha N et al (2019) Surface ozone in the Doon Valley of the Himalayan foothills during spring. Environ Sci Pollut Res 26(19). https://doi.org/10.1007/s11356-019-05085-2
- Ojha N et al (2020) On the widespread enhancement in fine particulate matter across the Indo-Gangetic Plain towards winter. Sci Rep 10(1). https://doi.org/10.1038/s41598-020-62710-8
- Pandey R, Tyagi AK (2012) Particulate matter emissions from domestic biomass burning in a rural tribal location in the lower Himalayas in India: concern over climate change. Small-Scale For 11(2):185–192. https://doi.org/10.1007/s11842-011-9177-8
- Penner JE, Dickinson RE, O'Neill CA (1992) Effects of aerosol from biomass burning on the global radiation budget. Science 256(5062):1432–1434. https://doi.org/10.1126/science.256.5062. 1432
- Prabhu V et al (2020) Black carbon and biomass burning associated high pollution episodes observed at Doon valley in the foothills of the Himalayas. Atmos Res 243:105001. https://doi.org/10.1016/j.atmosres.2020.105001
- Prasad VK et al (2002) Biomass burning and related trace gas emissions from tropical dry deciduous forests of India: A study using DMSP-OLS data and ground-based measurements. Int J Remote Sensing 23(14):2837–2851. https://doi.org/10.1080/01431160110109598. Taylor & Francis
- Prasad KV, Lata M, Badarinath KVS (2003) Trace gas emissions from biomass burning from northeast region in India—estimates from satellite remote sensing data and GIS. Environmentalist 23(3):229–236. https://doi.org/10.1023/B:ENVR.0000017376.69456.b8
- Ram K, Sarin MM (2010) Spatio-temporal variability in atmospheric abundances of EC, OC and WSOC over Northern India. J Aerosol Sci 41(1):88–98. https://doi.org/10.1016/j.jaerosci.2009. 11.004
- Ram K, Sarin MM, Hegde P (2010a) Long-term record of aerosol optical properties and chemical composition from a high-altitude site (Manora Peak) in Central Himalaya. Atmos Chem Phys 10(23):11791–11803. https://doi.org/10.5194/acp-10-11791-2010
- Ram K, Sarin MM, Tripathi SN (2010b) Inter-comparison of thermal and optical methods for determination of atmospheric black carbon and attenuation coefficient from an urban location in northern India. Atmos Res 97(3):335–342. Elsevier
- Ram K, Sarin MM, Tripathi SN (2012) Temporal trends in atmospheric PM2. 5, PM10, elemental carbon, organic carbon, water-soluble organic carbon, and optical properties: impact of biomass burning emissions in the Indo-Gangetic Plain. Environ Sci Technol 46(2):686–695. ACS Publications
- Ram K et al (2016) Variability in aerosol optical properties over an urban site, Kanpur, in the Indo-Gangetic Plain: a case study of haze and dust events. Atmos Res 174:52–61. Elsevier
- Ramanathan V, Carmichael G (2008) Global and regional climate changes due to black carbon. Nat Geosci 1(4):221–227. https://doi.org/10.1038/ngeo156
- Ramanathan V et al (2005) Atmospheric brown clouds: impacts on South Asian climate and hydrological cycle. Proc Natl Acad Sci 102(15):5326–5333. https://doi.org/10.1073/pnas. 0500656102

- Randerson JT et al (2006) The impact of boreal forest fire on climate warming. Science 314(5802): 1130–1132. https://doi.org/10.1126/science.1132075
- Rastogi N et al (2016) Temporal variability of primary and secondary aerosols over northern India: impact of biomass burning emissions. Atmos Environ 125:396–403. https://doi.org/10.1016/j. atmosenv.2015.06.010
- Ravindra, K., Agarwal, N., et al. (2019a) Appraisal of thermal comfort in rural household kitchens of Punjab, India and adaptation strategies for better health. Environ Int, 124, 431–440 Elsevier
- Ravindra, K., Singh, T. and Mor, S. (2019b) Emissions of air pollutants from primary crop residue burning in India and their mitigation strategies for cleaner emissions. J Cleaner Prod, 208, 261–273 Elsevier
- Ravindra K, Singh T et al (2019c) Real-time monitoring of air pollutants in seven cities of North India during crop residue burning and their relationship with meteorology and transboundary movement of air. Sci Total Environ 690:717–729. https://doi.org/10.1016/j.scitotenv.2019. 06.216
- Ravindra K et al (2020) Appraisal of regional haze event and its relationship with PM2.5 concentration, crop residue burning and meteorology in Chandigarh, India. Chemosphere:128562. https://doi.org/10.1016/j.chemosphere.2020.128562
- Robock A (1991) Surface cooling due to forest fire smoke. J Geophys Res Atmos 96(D11): 20869–20878. https://doi.org/10.1029/91JD02043. John Wiley & Sons, Ltd.
- Roy C et al (2017) Influence of enhanced Asian NOx emissions on ozone in the upper troposphere and lower stratosphere in chemistry-climate model simulations. Atmos Chem Phys 17(2): 1297–1311. https://doi.org/10.5194/acp-17-1297-2017
- Safai PD et al (2008) Aerosol characteristics during winter fog at Agra, North India. J Atmos Chem 61(2):101–118. https://doi.org/10.1007/s10874-009-9127-4
- Sahai S et al (2013) Autumn-time post-harvest biomass burning in Punjab causing aerosol perturbation over Central Himalayas. AGU Fall meeting abstracts 2013:A11H-08
- Sarangi T et al (2014) First simultaneous measurements of ozone, CO, and NOy at a high-altitude regional representative site in the central Himalayas. J Geophys Res Atmos 119(3):1592–1611. https://doi.org/10.1002/2013JD020631. John Wiley & Sons, Ltd
- Sarkar C et al (2015) Characterization of black carbon aerosols over Darjeeling—a high altitude Himalayan Station in Eastern India. Aerosol Air Qual Res 15(2):465–478. https://doi.org/10. 4209/aaqr.2014.02.0028. Taiwan Association for Aerosol Research
- Sarkar S, Singh RP, Chauhan A (2018) Crop residue burning in northern India: increasing threat to Greater India. J Geophys Res Atmos 123(13):6920–6934. Wiley Online Library
- Sawlani R et al (2019) The severe Delhi SMOG of 2016: A case of delayed crop residue burning, coincident firecracker emissions, and atypical meteorology. Atmos Pollut Res 10(3):868–879. https://doi.org/10.1016/j.apr.2018.12.015
- Saxena P, Naik V (eds) (2018) Air pollution: sources, impacts and controls. Cabi, Wallingford, pp 1–217
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8):6639– 6656
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Seiler W, Crutzen PJ (1980) Estimates of gross and net fluxes of carbon between the biosphere and the atmosphere from biomass burning. Clim Change 2(3):207–247. Springer
- Seinfeld JH, Pandis SN (1998) Atmospheric chemistry and physics: from air pollution to climate change. Phys Today 10(1063/1):882420

- Sembhi H et al (2020) Post-monsoon air quality degradation across Northern India: assessing the impact of policy-related shifts in timing and amount of crop residue burnt. Environ Res Lett 15(10):104067. https://doi.org/10.1088/1748-9326/aba714
- Shaik DS et al (2019) Impact of biomass burning on regional aerosol optical properties: a case study over northern India. J Environ Manag 244:328–343. https://doi.org/10.1016/j.jenvman.2019. 04.025
- Sharma SN, Prasad R (2008) Effect of crop-residue management on the production and agronomic nitrogen efficiency in a rice–wheat cropping system. J Plant Nutr Soil Sci 171(2):295–302. Wiley Online Library
- Sharma AR et al (2010) Impact of agriculture crop residue burning on atmospheric aerosol loading—a study over Punjab State, India. Ann Geophys 28(2):367–379. https://doi.org/10. 5194/angeo-28-367-2010
- Sharma D, Singh M, Singh D (2011) Impact of post-harvest biomass burning on aerosol characteristics and radiative forcing over Patiala, north-west region of India. J Inst Eng 8(3): 11–24
- Sharma D, Singh D, Sumit D (2012) Case studies of black carbon aerosol characteristics during agriculture crop residue burning period over Patiala, Punjab, India using the synergy of ground based and satellite observations. Int J Geol Earth Environ Sci 2(2):315–326
- Sharma D et al (2017) Temporal variability in aerosol characteristics and its radiative properties over Patiala, northwestern part of India: impact of agricultural biomass burning emissions. Environ Pollut 231:1030–1041. https://doi.org/10.1016/j.envpol.2017.08.052
- Sharma MC et al (2018) Seasonal characteristics of black carbon aerosol mass concentrations and influence of meteorology, New Delhi (India). Urban Clim 24:968–981. https://doi.org/10.1016/ j.uclim.2017.12.002
- Shi, Y., Sasai, T. and Yamaguchi, Y. (2014) Spatio-temporal evaluation of carbon emissions from biomass burning in Southeast Asia during the period 2001–2010. Ecol Model, 272, 98–115 Elsevier
- Siegert F et al (2001) Increased damage from fires in logged forests during droughts caused by El Niño. Nature 414(6862):437–440. https://doi.org/10.1038/35106547
- Singh RP, Kaskaoutis DG (2014) Crop residue burning: a threat to South Asian air quality. EOS Trans Am Geophys Union 95(37):333–334. Wiley Online Library
- Singh G, Rawat GS, Verma D (2010) Comparative study of fuelwood consumption by villagers and seasonal "Dhaba owners" in the tourist affected regions of Garhwal Himalaya, India. Energy Policy 38(4):1895–1899. https://doi.org/10.1016/j.enpol.2009.11.069
- Singh A et al (2014) Black carbon and elemental carbon from postharvest agricultural-waste burning emissions in the Indo-Gangetic Plain. In: Alados-Arboledas L (ed) Advances in meteorology. Hindawi Publishing Corporation, p 179301. https://doi.org/10.1155/2014/179301
- Singh R et al (2016) Impact of anthropogenic emissions and open biomass burning on carbonaceous aerosols in urban and rural environments of Indo-Gangetic Plain. Air Qual Atmos Health 9(7): 809–822. https://doi.org/10.1007/s11869-015-0377-9
- Singh N et al (2018) Aerosol chemistry, transport, and climatic implications during extreme biomass burning emissions over the Indo-Gangetic Plain. Atmos Chem Phys 18(19): 14197–14215. https://doi.org/10.5194/acp-18-14197-2018. Copernicus Publications
- Singh P, Sarawade P, Adhikary B (2020a) Carbonaceous aerosol from open burning and its impact on regional weather in South Asia. Aerosol Air Qual Res 20(3):419–431. https://doi.org/10. 4209/aaqr.2019.03.0146. Taiwan Association for Aerosol Research
- Singh T et al (2020b) A high-resolution emission inventory of air pollutants from primary crop residue burning over Northern India based on VIIRS thermal anomalies. Environ Pollut 266. https://doi.org/10.1016/j.envpol.2020.115132. Elsevier Ltd
- Sinha V, Kumar V, Sarkar C (2014) Chemical composition of pre-monsoon air in the Indo-Gangetic Plain measured using a new air quality facility and PTR-MS: high surface ozone and strong influence of biomass burning. Atmos Chem Phys 14(12):5921–5941. https://doi.org/10.5194/ acp-14-5921-2014

- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3(690444):1–20
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1-2):39–50
- Srivastava S, Kumar AS (2020) Implications of intense biomass burning over Uttarakhand in April– May 2016. Nat Hazards:1–17. Springer
- Srivastava AK et al (2012) Characteristics of black carbon over Delhi and Manora Peak—a comparative study. Atmos Sci Lett 13(3):223–230. Wiley Online Library
- Srivastava AK et al (2014) Characterization of carbonaceous aerosols over Delhi in Ganga basin: seasonal variability and possible sources. Environ Sci Pollut Res 21(14):8610–8619. Springer
- Streets DG et al (2003) Biomass burning in Asia: annual and seasonal estimates and atmospheric emissions. Glob Biogeochem Cycles 17(4). https://doi.org/10.1029/2003GB002040
- Takigawa M et al (2020) Can Delhi's pollution be affected by crop fires in the Punjab region? SOLA. https://doi.org/10.2151/sola.2020-015. Advance Online Publication
- Taylor D (2010) Biomass burning, humans and climate change in Southeast Asia. Biodivers Conserv 19(4):1025–1042. Springer
- Tiwari S et al (2009) Black carbon and chemical characteristics of PM 10 and PM 2.5 at an urban site of North India. J Atmos Chem 62(3):193–209. Springer
- Tiwari S et al (2013) Diurnal and seasonal variations of black carbon and PM2.5 over New Delhi, India: influence of meteorology. Atmos Res 125–126:50–62. https://doi.org/10.1016/j.atmosres. 2013.01.011
- Tiwari S et al (2015) Aerosol optical properties and their relationship with meteorological parameters during wintertime in Delhi, India. Atmos Res 153:465–479. Elsevier
- Tosca MG, Randerson JT, Zender CS (2013) Global impact of smoke aerosols from landscape fires on climate and the Hadley circulation. Atmos Chem Phys 13(10):5227–5241. https://doi.org/10. 5194/acp-13-5227-2013
- Tripathi SN, Pattnaik A, Dey S (2007) Aerosol indirect effect over Indo-Gangetic plain. Atmos Environ 41(33):7037–7047. https://doi.org/10.1016/j.atmosenv.2007.05.007
- Tyagi C et al (2020) Seasonal variation of black carbon emissions in Urban Delhi, India. Environ Claims J 32(2):101–111. https://doi.org/10.1080/10406026.2019.1699723. Routledge
- Vadrevu K, Lasko K (2015) Fire regimes and potential bioenergy loss from agricultural lands in the Indo-Gangetic Plains. J Environ Manage 148:10–20. Elsevier
- Vadrevu KP et al (2011) MODIS derived fire characteristics and aerosol optical depth variations during the agricultural residue burning season, north India. Environ Pollut 159(6):1560–1569. Elsevier
- Vadrevu KP et al (2012) Vegetation fires in the Himalayan region—aerosol load, black carbon emissions and smoke plume heights. Atmos Environ 47:241–251. https://doi.org/10.1016/j. atmosenv.2011.11.009
- Vadrevu KP et al (2019) Trends in vegetation fires in South and Southeast Asian Countries. Sci Rep 9(1). https://doi.org/10.1038/s41598-019-43940-x
- Vaishya A et al (2017) Aerosol black carbon quantification in the central Indo-Gangetic Plain: seasonal heterogeneity and source apportionment. Atmos Res 185:13–21. https://doi.org/10. 1016/j.atmosres.2016.10.001
- Veefkind JP et al (2012) TROPOMI on the ESA Sentinel-5 precursor: a GMES mission for global observations of the atmospheric composition for climate, air quality and ozone layer applications. Remote Sens Environ 120(2012):70–83. https://doi.org/10.1016/j.rse.2011.09. 027. Elsevier Inc.
- Venkataraman C et al (2005) Residential biofuels in South Asia: carbonaceous aerosol emissions and climate impacts. Science 307(5714):1454–1456. https://doi.org/10.1126/science.1104359

- Venkataraman C et al (2006) Emissions from open biomass burning in India: Integrating the inventory approach with high-resolution Moderate Resolution Imaging Spectroradiometer (MODIS) active-fire and land cover data. Global Biogeochem Cycles 20(2). https://doi.org/ 10.1029/2005GB002547. John Wiley & Sons, Ltd.
- Venkataraman C et al (2010) The Indian national initiative for advanced biomass cookstoves: the benefits of clean combustion. Energy Sustain Dev 14(2):63–72. Elsevier
- Vijayakumar K et al (2016) Effects of agriculture crop residue burning on aerosol properties and long-range transport over northern India: a study using satellite data and model simulations. Atmos Res 178:155–163. Elsevier
- Wan X et al (2019) Aromatic acids as biomass-burning tracers in atmospheric aerosols and ice cores: a review. Environ Pollut 247:216–228. https://doi.org/10.1016/j.envpol.2019.01.028
- Wang Y, Chen Y (2019) Significant climate impact of highly hygroscopic atmospheric aerosols in Delhi, India. Geophys Res Lett 46(10):5535–5545. https://doi.org/10.1029/2019GL082339. John Wiley & Sons, Ltd.
- Williams AP et al (2019) Observed impacts of anthropogenic climate change on wildfire in California. Earth's Future Wiley Online Lib 7(8):892–910
- Yarragunta Y et al (2020) Influence of forest fire episodes on the distribution of gaseous air pollutants over Uttarakhand, India. GISci Remote Sens 57(2):190–206. Taylor & Francis
- You C et al (2016) Levoglucosan evidence for biomass burning records over Tibetan glaciers. Environ Pollut 216:173–181. https://doi.org/10.1016/j.envpol.2016.05.074
- Yu P et al (2020) Bushfires in Australia: a serious health emergency under climate change. Lancet Planet Health 4(1):e7–e8. Elsevier


Rising Extreme Event of Smog in Northern India: Problems and Challenges

Ashima Sharma and Renu Masiwal

Abstract

What makes Air Pollution dangerous?---it causes untimely death of innocent people, oblivious of the threat, and what did they do?- they breathed! Yes, that's it. The victims are unsuspecting and at exposure of an inopportune risk. Annually, nearly 1.67 million Indians were killed due to air pollution, while so many more were endangered with some disease, allergies, health troubles, etc., due to the toxic air they breathed. Even though it is a silent killer, it is no more invisible. A noxious smog is seen hanging in the air every winter, dooming the once bright sunlit winter days, with its gloomy gravish-brown tint. Suspended particles in air scatter the sunlight away from the surface and what might appear as winter fog is actually the fatal *smog*, an intense blow to our fundamental right to breathe in a clean air. The critically impacted states include those in North India, such as Haryana, Punjab, Uttar Pradesh, and Delhi. The present chapter gives an overview of the challenges behind the increasing trend of smog events across Northern India. It seeks to deliver an understanding of the question of why the northern part of our country is becoming the pollution center of the world and intends to link the plausible causes and impacts for the rising extreme events of smog.

Keywords

Smog · Air quality · Air pollution · Northern India · Human health

A. Sharma (🖂)

School of Earth Sciences, Banasthali Vidyapith, Vanasthali, Rajasthan, India

R. Masiwal Academy of Scientific and Innovative Research (AcSIR), Ghaziabad, India, Uttar Pradesh

205

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_9

9.1 Introduction

In the very words of Theodore Roosevelt, knowing more about the past makes one better prepared for the future. On the same lines, a thorough understanding of the historical smog events, across the globe, is henceforth necessary to be able to link the sequence of their occurrence with the current reality of Indian cities.

It was in 1905 that a London physician named Harold Des Veaux coined the term *smog*. It was a combination of the two otherwise not very destructive entities, *smoke* and *fog*, and used to describe a condition wherein natural fog is contaminated by smoke. Very soon, in 1909, the term smog was for the first time used to describe a dangerous 5-week-long pollution episode in Glasgow, Scotland. The number of reported deaths was at least 2–4 times the normal number during pre-smog days. By 1925, Shaw and Owens had already published a comprehensive work on urban air pollution problems, wherein the smoke problem in big cities was discussed in detail.

In 1939, as the Second World War started, Los Angeles (LA) was about to witness what would be remembered in history as one of the world's most talked about air pollution episodes. In 1943, as a thick smog enveloped their city, the residents believed that it was some kind of a chemical attack prompted by the Japanese. No other reason was acceptable to the residents because the air in LA was extremely clean. However, this fog made their eyes burn and noses run. Only later, in the 1950s it was realized that this was a consequence of the emissions of their vehicles and factories. Investigations led by a scientist, A.J. Haagen-Smit, at the California Institute of Technology, brought about an understanding that ozone (O_3) was a major ingredient of this haze (Haagen-Smit 1952). An apparent tipoff was an unusual increase in the number of cases of cracking of rubber during this smog event, a phenomenon related to ozone attack. He studied that the gases from the exhaust fumes of vehicles and industries release volatile organic compounds (VOCs) and nitrogen oxides (NO_X) , which photochemically oxidize to generate O_3 (Fig. 9.1). LA had also seen large-scale immigration during wartime and early adoption of automobiles in the city made it a forerunner in automobile retailing. It was this incessant emission flux from vehicles and industries, plus a geography that supported trapping of air pollutants, that caused the "photochemical smog" or the "oxidizing smog." Currently, many cities in Mexico, USA, and Brazil are facing the same type of photochemical smog.

However, in the mid-twentieth century, the air pollution problem traversed continents and another most talked about smog event came into the limelight—*the Great Smog of London*. Although owing to coal burning, the polluted fog was prevalent in London since the thirteenth century, but with continued development, the situation progressed to worsen further. King Edward I had to pass a law in the fourteenth century to ban coal burning (Gaffney and Marley 2009). This is known to be the first legislation for reducing smog pollution and resultant human exposures. During the seventeenth century, soot deposition due to coal burning and its consequent effects on the buildings also attracted attention and concern of King James I (Chen et al. 2007).



Fig. 9.1 A schematic diagram explaining factors responsible for Los Angeles type photochemical smog and formation of major chemical components

In 1661, King Charles II asked John Evelyn Esquire to assess and report London's deteriorating quality of air attributable to combustion of coal. He reported horrid smoke concealing the buildings of London, fouling people's clothes, tainting the water bodies, causing acid rain, and leading to premature mortality, especially in kids (Evelyn 1772).

Industrialization in the 1700s made the air of even poorer quality. The possibility of these smoggy occurrences was attributable to the particulate release from coal burning that would attract water droplets in the atmosphere, producing dark clouds, an accompanied haze, and impaired visibility. However, nothing was done to curb the menace of these regular smog events, and in December 1952, the great smog manifested when inversion conditions (low winds and a shallow boundary layer caused by winter season) ensured none of the pollutants dispersed (Polivka 2018). Visibility conditions were extremely impaired to an extent that there were reports that people could not even see their own feet while walking on road. The 5-day-long event led to more than 12,000 fatalities. This type of smog was the *sulfurous or reducing smog*, characterized by high concentrations of sulfur dioxide (SO₂) and particulate matter (Fig. 9.2). Ultimately, the Clean Air Act was passed in 1956, which did improve public health conditions in London (Brazell 1970). At present, urban areas in China, Egypt, and India are examples of urban centers with London-type smog (Davis 2002).

Similarly, many such smog events have been observed in Meuse River Valley, Belgium (1930), St. Louis, Missouri (1939), Donora Pennsylvania (1948), New York, USA (1963), etc. (Firket 1936; Zimmer and Nemery 2017). These



Fig. 9.2 A schematic diagram explaining factors responsible for London-type sulfurous smog and formation of major chemical components

smog events served as major catalysts to create awareness and air quality regulations.

History repeats itself if lessons are not taken, and so is the case with air pollution. Increasing haphazard development, urbanization, population and several such factors are now causing a continuous decline in the air quality with similar such situations now recurring over the Indian cities. In regard to PM_{10} pollution, World Health Organization (WHO) enlisted some 100 cities around the world, out of which 37 Indian cities were included (Guttikunda et al. 2014). Sharma et al. (2019) reported that while in 1998, the national standards for ambient air were being breached in only 27% of India's districts, however, in 2010, air quality of nearly 45% of districts was exceeding the standards. They also wrote about WHO reports stating that nine cities out of top ten most polluted ones were in India, with the national capital Delhi, the most polluted in terms of PM_{10} . Majority of these cities are in Northern region of our country, and hence, for people living here, clean air is now a distant dream. Regular smog events are indicative of an impending crisis.

This chapter aims to understand the reasons behind this increasing trend of smog events across North India. It is extremely vital that we should have concrete answers to the question of why is the northern part of our country becoming the pollution capital of the world. Perhaps the answer lies in increasing population, urbanization, vehicular growth, industrial and economic expansion, or uncontrolled constructions releasing tonnes of dust. The geography of the Indo-Gangetic plain and meteorological parameters cannot be ruled out as major inherent disadvantages. This chapter intends to link these plausible causes for the rising extreme events of smog in Northern India.

9.2 North-India's Crowning Glory or Not?

The north is among the largest region in India comprising nearly 30.7% of the total national land area. Jammu and Kashmir, Himachal Pradesh, Rajasthan, Punjab, Harvana, Delhi, Uttar Pradesh, and Uttarakhand are major component states and/or union territories. More than 368 million people currently inhabit North India (Census 2011). The region's importance can be accessed from the fact that all together it contributed almost 26% to the national GDP in the financial year 2013. The installed capacity of power utilities in the North Indian states is reported in Table 9.1. The flourishing sectors of this region are IT, hydropower, and infrastructure. The economic growth of the northern region sometimes outperformed the growth in the rest of the country. The northern region is also rich with natural resources that offer opportunities for the development of several sectors such as solar power, hydropower, and mineral resources. All this offers opportunities for the development of industries in this region and scope to export and generate revenue. Continual infrastructure development in the region by major investments in the sector of rail, power, road, and technology is also a marked advantage in this part of the country (Table 9.2). Punjab, Haryana, and Uttar Pradesh have high agriculture production, which not only supports the national food demand but also supports the growth of agro-processing industries and power generation.

The extensive north-central region of the Indian subcontinent is the Indo-Gangetic Plain (IGP). It extends from the west to the east—including the Indus River Valley, the Ganga River, and the Brahmaputra River Valley. Encompassing an area of 2.5 million km², which is nearly 20% of the national geographical area, this fertile plain spreads majorly across north and east India. Silt and alluvium deposition by the numerous rivers originating from the Himalayas in the north makes the region very fertile. It makes it a vital agricultural and ecological hub, contributing nearly 42% to the total production of food grains in the country (Vadrevu and Lasko 2018). Rice and wheat are the main crops grown here, with others like sugarcane, maize, and cotton. This also makes IGP densely populated, with approximately 500 million residents. Undoubtedly, IGP is one of the most developed regions of our country, in terms of literacy, GDP, installed power capacity, telecom services, industries, agriculture, and much more. The key urban characteristics of cities of IGP are shown in Table 9.3.

However, when it comes to air pollution, the IGP region pretty much stands out as the most polluted region of the country. According to a 2018 report by the WHO, stating the list of the world's most polluted cities based on fine PM_{2.5}, 9 of the 10 cities were located in the IGP. Reasons are aplenty, several brick kilns still work with old combustion technology, numerous coal mines and large thermal power plants, landlocked cities that offer negligible pathway for the pollutants to disperse, intense postharvest crop residue burning activities that results in a high amount of air pollution load over the IGP, high population density, regional-level dust sources such as the Thar Desert, meteorology, and lack of effective dispersion owing to the Himalayas (Bray et al. 2019; Raatikainen et al. 2014). Numerous studies (Gaur et al. 2014; Ravindra et al. 2019; Singh et al. 2020) identified the following major sources of air pollution over northern India vehicles, manufacturing, industries, construction, road dust, open waste burning, fossil fuel combustion, and indoor and outdoor biomass burning.

Table 9.1 Installed capacity installed_capacity-08.pdf)	/ (in MW) of power	stations in No	thern India (as c	on 31.08.2020) (Sc	ource: http://cea.	nic.in/reports/mon	thly/installedca _l	pacity/2020/
	Thermal						RES	
States/union territories	Coal	Lignite	Gas	Total	Nuclear	Hydro	MNRE	Total
Delhi	4406.35	0	2115.41	6521	102.83	723.09	226.93	7574.62
Haryana	8660.57	0	685.61	9346.18	100.94	2312.02	537.98	12,297
Himachal Pradesh	151.69	0	62.01	213.70	28.95	2812.88	960.85	4016.38
Jammu and Kashmir	523.42	0	304.07	827.49	62.98	2321.81	204.92	3422.27
Punjab	8327.47	0	414.01	8441.48	196.85	3809.12	1604.85	14,322.27
Rajasthan	10,862.57	1580	824	13,267.47	556	1939.19	9755.6	25,519
Uttar Pradesh	18,690	0	549	19,239.78	289.48	3396.53	3350.46	26,276
Uttarakhand	461.17	0	519.66	980.83	31.24	1876.89	662.37	3551.33
Chandigarh	40.17	0	15.03	55.76	8.01	101.71	42.94	208.42
Total	53,439.79	1580	5781.26	60,801.05	1620	20,044.77	17,346	99,812.72

nly/installedcapacity/2020/	
p://cea.nic.in/reports/mont	
on 31.08.2020) (Source: htt	
ons in Northern India (as o	
y (in MW) of power stati	
= 9.1 Installed capacity	led_capacity-08.pdf)
ľ	stal

Table 9.2 Nor	th India and its G	llory (Source: h	ttps://asse	sts.kpmg/content/d	am/kpmg/pdf/20	014/10/North	iern-India-Hei	alding.pdf)		
States	Geographical area (sq. km)	Population (million)	GDP (%)	No. of administrative division	Population density per sq. km	Literacy rate (%)	Installed power capacity (MW)	Telecom (wireless subscribers) (million)	No. of airport	Sectors Industries (I) Agriculture (A) Service (S)
Chandigarh	114	1.1	1	1	9258	86	116.34			I-14% A-0.45% S-85%
Delhi	1483	16.8	15.86	11	11,297	83.6	7840.74	42.7	2	I-9% A-1% S-90%
Haryana	44,212	25.4	13.50	21	573	75.6	8288.69	21.2	9	I-27% A-15% S-28%
Himachal Pradesh	55,673	6.9	3.17	12	123	82.8	4007.22	7.2	e	I-38% A-19% S-43%
Jammu and Kashmir	222,236	12.5	3.05	22	124	68.7	2587.71	8.1	e	I-23% A-20% S-57%
Punjab	50,362	27.7	11.62	22	551	75.84	8389.14	31.3	e	I-29% A-21% S-50%
Rajasthan	342,239	68.6	16.45	33	200	67.06	15,175	52.9	9	I-25% A-29% S-46%
Uttar Pradesh	240,928	199.6	31.44	71	829	67.6	14,401.52	126.8	9	I-21% A-22% S-57%
										(continued)

tinued)
.2 (con
Table 9

										Sectors
										Industries
							Installed	Telecom		(I)
				No. of	Population		power	(wireless		Agriculture
	Geographical	Population	GDP	administrative	density per	Literacy	capacity	subscribers)	No. of	(A)
States	area (sq. km)	(million)	$(0_{0}^{\prime\prime})$	division	sq. km	rate (%)	(MM)	(million)	airport	Service (S)
Uttarakhand	53,483	10.1	4.90	13	189	78.8	2640.3	1		I-38%
										A-10%
										S-52%

In the Indian cities. Atmospheric Environment, 95, 501–510) Build- Population %	anagement
Key urban characteristics of cities of IGP (Source: Guttikunda, S. K., Goel, R., & Pant, P. (2014). Nature of air pollution, et in the Indian cities. <i>Atmospheric Environment</i> , <i>95</i> , 501–510)	ble 9.3 k anagement

Table 9.3 $K\epsilon$	sy urban c	haracteristics of	f cities of IGP (Source: Guttikunda,	S. K., Goel, R.,	& Pant, P. (2014).	Nature of air poll	ution, emission	1 sources, and
management ii	n the India	un cities. Atmosµ	pheric Environ	ment, 95, 501–510)					
	Build-		Population		%				
	dn		density	% Households	Households	% Households			
	area		(per	with a motorized	with a four	with a non-gas		SO_2 (µg/	NO_2 (µg/
City	(km^2)	Population	hectare)	two wheelers	wheelers	cookstove	$PM_{10} (\mu g/m^3)$	m^3)	m ³)
Delhi	699	16,314,838	244	39%	21%	9%6	260.1 ± 117.1	6.5 ± 4.2	51.1 ± 17.2
Jammu	123	651,826	53	48%	25%	13%	118.2 ± 37.4	8.2 ± 4.4	12.7 ± 3.4
Ujjain	33	515,215	156	40%	6%	26%	78.4 ± 42	10.9 ± 3.4	11.9 ± 3.1
Amritsar	90	1,183,705	132	50%	15%	21%	188.7 ± 24.2	14.8 ± 2.2	35.1 ± 3.1
Chandigarh	115	1,025,682	89	47%	26%	27%	79.9 ± 32.6	5.8 ± 0.5	15.4 ± 7.8
Ludhiana	167	1,613,878	97	50%	19%	19%	251.2 ± 21.9	8.4 ± 2.3	36.2 ± 7
Agra	129	1,746,467	135	48%	12%	27%	184.1 ± 95.9	6.6 ± 3.5	20.8 ± 12.1
Allahabad	71	1,216,719	171	54%	11%	26%	165.3 ± 70.7	3.6 ± 1	23.7 ± 15.9
Firozabad	21	603,797	288	25%	4%	40%	195.6 ± 78.2	21.6 ± 4.8	32.1 ± 4.9
Kanpur	150	2,920,067	195	11%	3%	42%	211.5 ± 25.3	7.5 ± 1.2	31.3 ± 4.9
Lucknow	240	2,901,474	121	52%	15%	20%	200.4 ± 28.4	8.4 ± 1	36.1 ± 2.6
Varanasi	102	1,435,113	141	40%	7%	29%	125.3 ± 8.4	17.2 ± 0.7	19.6 ± 0.7

This trend of air pollution is widely reported in the Indian capital Delhi and its surrounding National Capital Region (NCR). As India's largest urban agglomeration and second largest in the world, Delhi–NCR has approximately more than ten million registered vehicles, which are the greatest source of pollution here (Peshin et al. 2017). More than 70,000 trucks cross Delhi roads polluting the air with its diesel exhaust. In addition, poor public transport, ever-increasing population, continuous construction and demolition activities, millions of vehicles contaminating the air with their emissions, etc., are all the preliminary anthropogenic reasons for the prevalence of smog in this region. Balakrishnan et al. (2019) reported highest mean $PM_{2.5}$ over Delhi in 2017, followed by Haryana, Uttar Pradesh, Bihar, etc., all with mean values exceeding 125 μ g/m³.

9.3 Smog Events over Northern India

North India has changed a lot, with respect to demography, urbanization, economy, and environmental pollution. Major sources of air pollution here include brick kilns, industries, coal burning, electricity production, construction, vehicles, suspended road dust, biomass burning, crop stubble burning, and generators. Indoor air pollution, majorly caused by burning biomass such as wood, dung cakes, charcoal, and agricultural waste for cooking or heating purposes, also remains a challenge (Sonwani and Saxena 2022; Tiwari and Saxena 2021; Saxena and Naik 2018).

The past decade has been very hard on Northern India in terms of air pollution. The escalating number and intensity of the smog events over the north are not only influenced by the emissions but also influenced by the weather conditions, growing population, industrial activities, open waste disposal and burning, high energy demands, inherent topography of the North Indian plains, etc. (Bhaskar and Mehta 2010). For example, the northwesterly winds from Punjab and Haryana are one of the causative factors that bring in pollutants emanating from the burning croplands to Delhi–NCR (Saxena et al. 2021). Similarly, reduced wind speeds are also responsible for the surface stagnancy of air pollutants and their coalescing onto the water vapor (Fig. 9.3). Ambient air water vapor undergoes condensation when solid particles are present as they serve as condensation nuclei (Kumar and Goyal 2014). More or less, the reasons can be clubbed together as anthropogenic (urbanization and industrialization) and natural causes as is described in the chart below:



The regular occurrence of smog events is now being observed almost on an annual basis, particularly after the monsoonal months, and during winters. Table 9.4 lists few studies across different regions of Northern India showing the severity of the situation.

Delhi has been experiencing repeated smog events since the last few years, almost on an annual basis. In the year 2014, smog intensified in Northern India, especially over Delhi–NCR, due to low wind speeds, causing pollutant build-up and visibility dropping to just 1100 m. Agricultural fires in Punjab and Haryana were detected by NASA's satellite imagery, were considered the most dominant factor enhancing pollution in the region, as the winds emanating from these burning croplands, eventually reached Delhi, from the northwestern side (Sawlani et al. 2019; Takigawa et al. 2020). Even major rice-producing districts in Punjab such as Patiala and Ludhiana recorded very high air pollution levels (Rana et al. 2019; Chawala and Sandhu 2020).

In 2015, smog was back during the post-monsoon months, blanketing the city and reducing visibility down to about 200 m. An AQI of 372 was recorded over New Delhi, which indicated that the citizens breathed hazardous air. Possible reasons were local-scale coal and biomass fires lit on the street by homeless poor, traffic emissions, temperature inversions, and the lack of wind. As the most widely celebrated festival of India, Diwali, came, the pollution breached WHO recommended levels on the festival's night, by almost 40 times. The fingers were



Fig. 9.3 A schematic diagram explaining factors responsible for Delhi's Great Indian smog and formation of major chemical components

	Place of			
S. no.	study	Findings	Duration	Study
1	Mohali, Punjab	High amounts of benzenoids and other gases, linked strongly to ozone-forming potential observed in Punjab. It can be linked to high ozone concentrations in the downwind sites	2012	Sarkar et al. (2013)
2	Delhi	Severe smog episode with extremely low visibility. High pollution (PM_{10} and NO_2 almost 10 times and 7 times more than NAAQS, respectively). Under adverse meteorological conditions, regional pollution could be a stronger factor in air quality deterioration, as compared to local pollution	2012	Sati and Mohan (2014)
3	Delhi	Concentration of aerosols breaching the NAAQS, especially during winter months	2014–2015	Sawlani et al. (2020)
4	Chandigarh, Punjab	Daily average PM _{2.5} concentration nearly 3 times the NAAQS, and an approximate 2.2 times reduction in visibility. Satellite data and backward air mass trajectory analysis showed rice stubble burning on a large scale. Enhanced concentrations of organic carbon and K ⁺ ions in PM _{2.5} and acetonitrile, which are biomass burning tracers, were observed	2015	Ravindra et al. (2020)
4	North India	The plumes generated from biomass burning during post-monsoon season were transported over the Indo-Gangetic plain region due to favorable meteorology. Higher aerosol optical depth values were also observed during this time. This depletes air quality over regions of North India, especially Delhi– NCR, and is linked to intense atmospheric pollution, resulting in haze and smog over long durations	2016	Chauhan and Singh (2017)
5	Delhi	Severe smog episode— ≤ 5 km visibility, average PM _{2.5} concentration exceeding NAAQS by nearly 13 times, just after Diwali. Secondary particulates were actively being formed during the smog period due to stagnant meteorology	2016	Sawlani et al. (2019)
6	Delhi	PM _{2.5} levels 10 times more than the NAAQS; AQI on November 8, 2017, was 999	2017	Basu (2019)
7	Delhi	Conducive meteorological conditions such as regional airflow, reduced	2017	Terry et al. (2018)

Table 9.4 Reported smog events over different parts of northern India. (NAAQS-NationalAmbient Air Quality Standards, 2009)

(continued)

S. no.	Place of study	Findings	Duration	Study
		temperature, inversions, and low-speed winds led to smog build-up		
8	Delhi	Very high concentrations of air pollutants were observed indicating severe air quality—a serious matter of concern necessitating awareness among farmers to curb crop residue burning	2017–2018	Garg and Gupta (2020)

Table 9.4 (continued)

pointed on to the massive quantities of fireworks and crackers burning during the festival. During the night hours, the coarse PM_{10} concentrations rose to $2000 \ \mu gm^{-3}$, almost 20 times the NAAQS. Burning of agricultural waste residues also continued for several days, adding to the pollution load. In December 2015, many schools restricted outdoor activities owing to high levels of air pollution, and soon, Delhi became the world's most polluted city, surpassing Beijing for air pollution limits.

Yet again, the smog returned next year in November 2016. High concentrations of $PM_{2.5}$ and PM_{10} were reported making this event, as one of the worst in terms of air quality, in Delhi since 1999. Numerous investigative analyses hinted toward cold weather and calm winds, with reduced speeds, trapping the emitted pollutants. This was understood to have been caused due to an anti-cyclonic circulation, which also caused pollutants to stick to water vapor, and severely aggravate the persisting conditions (Sawlani et al. 2019). It was an unusual event characterized by active secondary particulate formation. Primary sources of smoke were the same as before road dust, burning of crop stubble and waste, etc. (Fig. 9.4). This period also coincided with Diwali, and hence, firecracker burning was among a key cause. $PM_{2.5}$ and PM_{10} levels shot up almost 17 times the recommended standards. Visibility had reduced to about 200 m with an average temperature of around 20 °C.

In 2017, the smog, now regarded as an annual event, repeated itself once again and came to be known as "the great smog of Delhi." It was very clear visually how bad the air quality had become in the national capital territory (Fig. 9.5). A toxic haze blanketed the city, and it was so severe that schools were closed, flights were delayed and in the own words of Delhi's chief minister, the city had become a gas chamber. Research studies and medical professionals stressed on the severity of the situation that could result in irreversible severe lung damage, even among healthy nonsmoking young people, particularly children (Slutsky 2017; Chetambath and Jesin Kumar 2019). The level of PM_{10} and $PM_{2.5}$ was more than 20 times the prescribed standards. It continued even until the end of December, as the weather got chillier and fog engulfed the city. Delhi continued to face the most polluted days of the year, with the air quality index at an "emergency" level.

Smog was back in 2018 as the air pollution rose to hazardous levels, particularly after the Diwali festival that once again led to firecracker burning, despite court bans.



Fig. 9.4 Dust surface mass concentration of $PM_{2.5}$ during the smog event of 2016, time average hourly 0.5 \times 0.625 deg. (MERRA-2 Model M2T 1NXAER v5.12.4) kgm^{-3}



Fig. 9.5 The great smog of Delhi (08.11.2017, 09:00 AM) (location: Patel Nagar Metro Station, New Delhi)

Residents woke up the next day after Diwali to a thick blanket of highly toxic smog. This, besides crop residue burning emissions and emissions of a mixture of primary air pollutants ensured that the smog lasted for more than 2 months, throughout the winter season.

In 2019 also, the air quality of the city dipped and for almost 2 weeks at a stretch, the air quality was in the hazardous category, suggesting air pollutants were at levels

where the entire population is getting affected. Owing to low visibility, flights were diverted and Supreme Court—the highest court of India, stated that living in Delhi is even worse than living in hell. The air quality crisis was owing to a noxious mixture of pollutants emanating from crop residue burning, celebratory firecrackers, vehicular emissions, industrial fumes, etc.

Being a rapidly developing region, North India is facing the problem of ambient and household air pollution. This tendency of repeated and a sharp increase in the smog events not only over Delhi, but entire Northern India, is a cause of concern.

9.4 Challenges: Problems and Allied Impacts

The Great North Indian Smog is in every way an emergency—even NCR's Environment Pollution (Prevention and Control) Authority declared it as a public health emergency. Every year, several cities of North India turn dystopian, and this continues for months, till the winter ends. It is a burgeoning event, a product of numerous problems that continue to strengthen more and more with time. The ensuing smog manifests at an exponential scale, as a direct function of an indifferent attitude and lack of any stringent action. The chief concerns have been listed in the chart below, and associated key challenges are discussed in detail later:



9.4.1 Industrial and Vehicular Emissions

Industrial clusters are a key cause of air pollution. Lack of emissions data that are accurate and easily available is a major hindrance to warrant compliance to the air quality standards. Comprehensive Environmental Pollution Index (CEPI) was developed by CPCB in 2010, to assess environmental pollution at the industrial clusters in India, and 43 clusters were identified as critically polluted—most being in North India such as Faridabad, Ghaziabad, Ludhiana, Kanpur, and Agra.

Guo et al. (2017) quantified how much the different source types contributed to fine particulate matter (PM_{2.5}) and its components, in Delhi, Chandigarh, Lucknow, and Jaipur, and reported that industrial and residential activities govern primary particulate matter (PPM) mass by more than 60%. High secondary inorganic aerosol (SIA) concentrations (~25 mg/m³) were mainly ascribed to energy, industry, and residential sectors. Industrial sector was second, after residential sector for its contribution to total PM_{2.5} (~70 mg/m³) in North India. In Delhi, Jain et al. (2017) estimated 13.4% and 6.2% contribution of industrial emission in the mass concentration of PM_{2.5} using UNMIX and positive matrix factorization (PMF) models, respectively, while about 7.2% of the PM₁₀ mass were contributed by industrial emissions (Sharma et al. 2014). Sharma et al. (2016a, b) quantified industrial emissions and fossil fuel burning as major source of PM₁₀ in Delhi, contributing almost 7.3% and 15.5%, respectively, to the PM mass.

Vehicles are another key cause of air pollution. Several studies have reported the percentage contributions of vehicular emissions to the ambient air pollution. Guttikunda and Goel (2013) conducted a study at six residential and industrial zones of Delhi, and modeled contribution to PM2.5 by different emission sources. It was reported that vehicular exhaust contributed nearly 16–34%, while industries contributed approximately 14-21%. The remaining 20-27% contribution was from diffused sources, 3–16% from diesel generator sets, and 4–17% for brick kilns. Similarly, Goyal et al. (2013) used the international vehicle emission model to estimate vehicle-generated emissions of criteria pollutants over Delhi. The estimated emissions of CO, NOx, and PM in 2008-2009 were found to be 509, 194, and 15 tons/day, respectively, while calculated emissions of CO and NOx from personal cars were about 34% and 50%, respectively, and CO emission due to two wheelers was 61%. Jain et al. (2017) estimated 19.7% contribution of vehicular emission in PM_{2.5}, while others (Sharma et al. 2014, 2016b) reported 16–17% contribution from vehicular emissions to the PM mass. Rajput et al. (2018) studied PM1-their abundance, temporal variability, and source apportionment. Vehicular emission and industrial activities were among the seven dominant sources of PM_1 in Kanpur.

Kandlikar and Ramachandran (2000) discussed the use of emission factors to estimate pollutant emissions from the total number of vehicles. It requires relating the quantity of ambient air pollutants to how much is the activity of different vehicles. Indian cities have larger emission factors because of absence/insufficient pollution control equipment, poor maintenance of vehicles, insufficient monitoring, and implementation of emission standards, and average traffic speeds are usually low owing to bad condition of roads, heavy vehicle density, and the quality of fuel

used. In Delhi, the total available space on roads did increase, but so did the number of vehicles per kilometer. Between 2003 and 2009, the city also witnessed a sharp increase in the annual rate for both private and commercial vehicles—7.40% and 9.15%, respectively. Delhi is also fifth worst in the world with respect to traffic jam (Goyal et al. 2013). This is a significant reason for severe transportation and environmental problems (Goyal et al. 2013).

Even though the medium and heavy commercial vehicles do not contribute much to the total number of vehicles, they have substantial contributions to the pollution. Regular inspection and monitoring are required, but are a challenging task. Another challenge is that these measures majorly are followed stringently over the big cities, while many smaller cities or rural areas over northern India do not have any support system to access these facilities; therefore, not much implementation is done. Adoption of Bharat Stage emission norms, nationwide, is therefore recommended.

Another important concern is public transport. With cities seeing an incessant growth, there is a growing need to have a basic facility for a safe and clean public transport system. Even though there is an established public transport facility in big cities, there is a constant need to augment the current fleets, but with cleaner options. For instance, the Delhi government converted public transport such as auto rickshaws and buses to CNG between 1998 and 2002. However, the situation still has not improved—neither in terms of traffic density nor in terms of air pollution. For smaller cities, the situation is even worse. Most lack an organized public transport system, relying on paratransit systems, which ply only on main corridors of the city. With the majority of cities still lacking these kinds of transport facilities, and with the growing population and economy, the burden over these cities' road is escalated, which is visible as the rising level of air pollution.

9.4.2 Crop Residue Burning

With the advent of the green revolution and process of liberalization and economic reforms in the past two decades, the manufacturing sector geared up in northern Indian states. This led to agricultural reforms and modifications in farming practices as well, which correspondingly increased the levels of air pollution over time. Biomass burning is one of the foremost sources of trace pollutant emissions. These tend to severely influence the climate and health. RSPM (respirable suspended particulate matter) and atmospheric trace gas levels have been reported to be above the maximum permissible limits by CPCB as a direct consequence of it. It has been indicated by the several studies conducted that emitted particles are transported across IGP from the burning agricultural lands (Fig. 9.6). Millions of hectares of land of the IGP are under agriculture—mostly rice–wheat rotation, and farmers preferably use combine harvesters for harvesting them. However, this technique leaves large quantities of straw stubble, standing tall (25-30 cm) in the field, which is then burnt down. This is mainly because of its inability to be consumed as fodder due to high silica content. Furthermore, there is only a short duration of just 2–3 weeks available in between which the farmer has to harvest the rice crop and sow the wheat



Fig. 9.6 MODIS aqua fire radiative product (FRP) data were used to see the number of fire events in northern India during October, November, and December. The increase in the fire events is visible by the increase in the density of the red dots signifying the number of fires (downloaded from https://firms.modaps.eosdis.nasa.gov/map/#d:2020-09-23..2020-09-24;@40.1,7.6,3z)

crop. High labor wages and resultant anxiety among farmers to harvest, collect, and market their crops are some of the multiple reasons, due to which they find it easier to burn the stubble off.

Crop residue burning, during the harvesting season, releases huge emission load, in the form of trace gases and aerosols, which disturbs the regional atmospheric chemistry and is a health hazard as well. Jain et al. (2014) reported that nearly 80% of the crop straw was being burnt in situ in Himachal Pradesh, Punjab, and Haryana. This led to emissions of air pollutants such as CO₂, N₂O, CH₄, CO, NH₃, NOx, SO₂, NMHCs, VOCs, semi-VOCs, and particulate matter, which were not only harmful to human health but also contributed toward global warming (Mittal et al. 2009). It was observed that burning of 98.4 Mt of crop residue resulted in tonnes of emissions of CO, CO₂, SO_x, NO_x, NH₃, NMVOC, NMHC, and PM during 2008–2009 (Jain et al. 2014). Similarly, Nagar et al. (2019) also confirmed that the crop residue burning emissions in Punjab, Haryana, Uttar Pradesh, get transported toward the eastern side of the Ganga basin, causing air quality degradation over Delhi and Kanpur. Sembhi et al. (2020) reported that burning of crop residues in Punjab and Haryana led to the smog event of 2016 across the IGP. It was not only hazardous to the health, but also caused biodiversity loss over the IGP and soil fertility deterioration via loss of organic carbon, nitrogen, and other nutrients (Jain et al. 2014; Lohan et al. 2018).

Hence, mostly a hard trade-off arises between environmental improvement and profitable economic opportunities (Shyamsundar et al. 2019), and despite regulations, burning continues to take place because of lack of implementation, uncertainty regarding alternative technologies, etc. However, a more coordinated effort involving public and private partnership both could be a way ahead for a transition toward a more sustainable agriculture.

9.4.3 Natural Processes

The formation of fog over northern India is a natural phenomenon, which dominates particularly during the winter season, under suitable meteorological conditions like atmospheric stability, calm winds, low temperatures, and high humidity. Low-level temperature inversions are formed, which eventually facilitate formation and persistence of fog over northern India (Fig. 9.7).

A combination of atmospheric and anthropogenic factors can be held accountable for magnified pollution levels across Northern India. In summers, vertical mixing is enhanced as the surface temperatures are high. This ensures effective dispersion of the released pollutants away from the ground, as the warm air rises upward easily. The same process is diminished during winters, when the surface air in the planetary boundary layer remains cool and hence dense. It gets trapped under the warm air above, and this is called winter inversion. This hugely reduces the dispersion of pollutants from the surface layers to the upper layers of the atmosphere.

Moreover, as the winter onset happens, the wind direction over the IGP changes from west to east, which brings in pollution load from the burning agricultural fields



Fig. 9.7 Normal atmospheric unstable conditions, with air in contact with a warm surface, and layer of cool air above while inversion conditions are when stable conditions persist. The layer of air in contact with the surface is cool with a warm layer of air sandwiched between two layers of cool air. Under such conditions, the pollutants released and formed at the ground level get trapped at the surface level. This provides ideal conditions for smog formation and persistence at the surface level. It continues to persist until the inversion layer breaks

of Punjab and Haryana. Wind speeds also slows down. The geography of IGP is truly cursed—it is like a hugely populated and polluted low-lying valley, surrounded by Himalayas in the north and Vindhyas in the south. Without any possible outlet and low wind speeds, the pollutant emissions thus collect and accumulate in this landlocked, low-lying valley, which is also known as the valley effect. Moreover, heavy emissions from industries, vehicles, biomass burning activities, etc., cause an even intense pollution, specifically, over major cities like Delhi, Kanpur, and Varanasi, especially when temperature inversions are prevalent.

Ground-level emissions of pollutants under these circumstances lead to fog transformation into smog. The aerosol formation is one of the key mechanisms of smog manifestation. The abundance of nitrogen dioxide (NO_2) in the ambient atmosphere of northern India results in its reaction with hydroxyl radical (OH) and forms nitric acid (HONO₂), which can further undergo reaction with ammonia (NH_3) to generate ammonium nitrate (NH_4NO_3), which is an important contributor to the secondary aerosol formation in urban and regional settings.

 $\begin{array}{l} OH + NO_2 \rightarrow HONO_2 \\ \\ HONO_2 + NH_3 \rightarrow NH_4NO_3 \end{array}$

Hydrogen peroxide is very soluble in water and can also react very rapidly with sulfur dioxide (SO_2) in cloud droplets or on wet aerosol surfaces to form sulfuric acid

 (H_2SO_4) . The ambient SO₂ can be transformed into sulfate, which further transforms into sulfate particles. Moreover, sulfuric acid can also undergo reaction with ammonia and create ammonium bisulfate (NH_4HSO_4) and ammonium sulfate $[(NH_4)_2SO_4]$. Under low humidity levels, these persist as solid particles, while under high levels of humidity, these continue to persist as solution droplets, henceforth causing visibility reduction.

$$H_2O_2 + SO_2 (aqueous) \rightarrow H_2SO_4$$

9.4.4 Road Dust, Construction, and Demolition

Road dust consists of solid particles that are generated by mechanical processes like crushing, grinding, impaction, detonation, and decrepitation of materials like rock or ore (Khan and Strand 2018). When vehicles ply on roads, any resulting friction of tyres moving on the roads causes the dust to become airborne which is then known as the road dust. This road dust, along with the construction and demolition (C&D) dust, are key contributors to pollution in the Indian cities (Guttikunda and Gurjar 2012; Guttikunda 2012). A contribution of nearly 131 Gg year⁻¹ has been reported from windblown road dust for PM₁₀ (Sahu et al. 2011). High concentrations of heavy metals like lead, nickel, and chromium have been reported from the road dust samples of industrial areas (Suryawanshi et al. 2016). Similarly, vehicular sources contribute high levels of zinc, copper, and barium, while industrial sources contribute high concentrations of cobalt, chromium, copper, and manganese (Rajaram et al. 2014).

Jain et al. (2020) studied source apportionment of $PM_{2.5}$ and PM_{10} using three different models and observed that road dust has significant contribution in both $PM_{2.5}$ and PM_{10} levels of Delhi, though the percentage contribution of road dust varies with season and model. Road dust emits more particles in the coarser particles range and these coarse particles then become major components of smog, reducing visibility after being suspended in the air. About 22.4% of the PM_{10} mass were contributed by soil dust, the largest contributor among all the major sources of PM_{10} mass concentration (Sharma et al. 2014).

This source largely contributes to the coarse PM_{10} pollution, and several factors like vehicle movement on roads, type of roads, silt loading on the roads, and prevailing meteorological conditions determine the quantitative contribution of road dust. It can be managed by strategic planning and measures like wet sweeping, vegetation, landscaping, road pavement, and maintenance that prevent dust resuspension (Sonwani et al. 2021; Sonwani and Saxena 2021).

For dust management, the Ministry of Environment, Forest and Climate Change issued a dust mitigation notification in January 2018. Under this, efforts were taken to reduce air pollution via compulsorily including dust mitigation measures in infrastructural projects and demolition activities nationwide. However, the compliance of these rules is still an issue, especially in smaller cities.

9.4.5 Lack of Source Identification Studies

With an incessant and continuous rise in air pollution over the Indian cities, research in this area is gradually snowballing and there has been an increase in published work in this area. However, the problem is that it is still widely scattered, with a severe lack of studies discussing the aspects of air pollution pertaining to their sources and eventual impacts.

For pollution abatement, source apportionment is a valuable tool to derive information about the sources of air pollution, and how much they contribute to the pollution levels. Receptor models apportion the measured mass of an air pollutant at the receptor site to its sources of emission. However, the results of receptor modeling approach are representative only of the area within and around the sampling location. Hence, multiple site analysis is required to be conducted to obtain a detailed data set that can give the needed information (Peshin et al. 2017). Moreover, a detailed chemical characterization of emission sources is also needed, for the completion of receptor modeling study. This makes it a comparatively expensive method with inadequate spatial coverage.

Similarly, source modeling is another useful tool to derive information about air pollution over a region and relies on the availability of data including that of vehicles, fuel consumption in the different domains, population, and meteorology. With the help of computational facilities, source dispersion modeling can be achieved.

There have been some source apportionment studies that have been conducted over North India—such as Kanpur (Rajput et al. 2018), Mirzapur (Murari et al. 2020), Varanasi (Murari et al. 2020), Delhi (Srivastava et al. 2008; Sharma et al. 2014, 2016a, b; Jain et al. 2017, 2020), and Patiala (Singh et al. 2016). However, it is vital to develop these data and use preexisting knowledge as a tool to design effective pollution strategies in India.

9.4.6 Inefficient Waste Management

Several million metric tonnes of municipal solid waste are generated annually in India, which is a cause behind the numerous environmental and health problems, and still, unsustainable practices of solid waste management continue to plague the North Indian cities. These include open burning, landfill fires, and incineration that are a common sight all over Northern India. Open disposal and burning of MSW are becoming a serious issue. The burning of waste products like plastic and rubber releases noxious fumes containing dioxins and furans in the ambient atmosphere, while burning of e-waste releases harmful toxins like cadmium, dioxin, and furans.

Landfills are the most commonly implemented practice for waste disposal in India. However, a key problem is the lack of an effective design and planning for a scientifically engineered landfill. This results in several environmental problems, expanding into severe health problems. For instance, methane emissions, groundwater contamination from the toxic leachate, and soil pollution are all commonly observed scenarios over the North Indian landfills, or more commonly mere dumpsites. There are several landfills scattered across North India dented with occurrences of hazardous methane emissions leading to frequent fire outbreaks, mudslides, smell nuisance, and diseases for the people residing there. Methane not only is a much potent greenhouse gas but also is a key precursor to ozone formation, which is a major ingredient of smog (Sharma et al. 2017).

Incinerators are another way to tackle the solid waste problem and are slowly gaining ground in India. However, since majority of the waste has high moisture content, therefore, waste burning in incinerators becomes an inefficient process, which worsens air pollution further through emissions of toxic gases and particulates, while offering little to no benefits for solid waste management. These are also precursors to ground-level ozone formation and hence contribute to smog.

9.4.7 Impacts

9.4.7.1 Health Impacts

The smog events over northern India are a health emergency. In fact, during the intense smog days, the government declared it as a public health emergency and all schools, offices, and government centers were shut down. The primary impact of such recurring smog events is on health. Poor air quality is like an invisible danger, and smog can be thought of as a "slightly visible, but highly perilous danger" significantly affecting body function. We breathe every second of our lives, taking about 20,000 breaths per day and undoubtedly, our health is strongly influenced by the air we breathe. So, many people die prematurely due to diseases of the respiratory system, such as asthma, bronchitis, and lung cancers, affecting every individual beyond the barriers of status, age, or place of habitat.

Air pollution comes consistently at the top risk factors for death and disability. Some of these effects are short term; for example, WHO reports that high pollution events can trigger asthma or lead to a spike in hospitalization related to cardiovascular and respiratory disease, while long-term exposure could lead to an increase in the potential of developing chronic heart and lung diseases and also cancers. On an average, it has been studied that air pollution can reduce life expectancy. On an average, citizens living in the IGP region did lose nearly 7 years of their lives, because of bad air quality, especially with respect to fine particulate pollution (Greenstone and Fan 2018). A special report on global exposure to air pollution and its disease burden, the State of Global Air, outlined that in the year 2017, air pollution was identified as the fifth major risk factor that causes the highest mortality in that year, with nearly 4.9 billion deaths. Due to air pollution, premature mortality and rise in disease burden could be evident, particularly higher in low- and middle-income countries.

Millions of people living in the northern region account for 40% of the Indian population residing in the country, and all of them are at varying levels of risks and impacts. Rise in pollution over this region has caused a drastic change in its health burden, patterns of diseases, their composition, the type of dominant disease,

morbidity and mortality determinants, etc. The report of the Global Burden of Disease stated that air pollution was a large risk factor for disease burden, second only to malnutrition, in 2016.

Even though air pollution continues to plague India nationwide, high concentrations of particulate matter in the north, including cities of Delhi, Gurgaon, Ghaziabad, Chandigarh, and Kanpur, are prominent. Studies have also indicated proof of irreversible health damage of children residing in polluted cities such as Delhi, especially vulnerable are those who walk to school alongside polluted, busy roads (Slutsky 2017). Restricted outdoor activities because of smog are another disadvantage for these children, being deprived of physical activity, in their growing years. Poor air quality during smog could reduce the lung capacity, trigger headaches, and lead to sore throat, coughs, fatigue, and cancers (Sonwani et al. 2022).

The annual exposure of the Indian population to ambient $PM_{2.5}$ was highest compared to the world, in the year 2017 (89.9 µg/m³) (Balakrishnan et al. 2019). The disability-adjusted life year (DALY) is another parameter used to measure the burden of disease and is an expression for the number of lost life years as a result of disability, bad health, or premature death. Balakrishnan et al. (2019) reported that DALY rate for the ambient air particulate matter was highest in the North Indian states—Uttar Pradesh, Haryana, Delhi, Punjab, and Rajasthan. Rizwan et al. (2013) identified significant associations of air pollution with adverse health outcomes and its severe effects on respiratory functions and morbidity.

All Indian states had annual population-weighted mean ambient $PM_{2.5}$ higher than the recommended levels by the WHO. In fact, states especially Uttar Pradesh, Rajasthan, etc., had some of the highest levels of both ambient and household air pollution levels, while citizen of states like Delhi, Haryana, and Punjab was exposed to some of the highest particulate matter pollution in the country.

India has been witnessing a continuous upsurge in the cases of chronic obstructive pulmonary disorder (COPD) mortality, and this is projected to be among the highest in the world (Bhome 2012). The National Family Health Survey (NCMH) estimates showed that there were around 17 million COPD patients in India, most likely to reach 22 million by the next 10 years. The Indian Council of Medical Research (ICMR) conducted a questionnaire-based study with several thousand respondents from urban and rural areas. It was revealed that nearly 4.1% of surveyed individuals (5% males to 3.2% females) were affected by COPD.

However, a much bigger challenge is resource constraints for conducting effective large-scale surveys based on spirometry. Mostly, the studies and estimations are done using questionnaire-based methodologies. The understanding of air pollution sources and events over the northern plains and its critical relationship with trends in air pollution levels, population structure, underlying disease, and economic factors will help estimate the air pollution effects. These trends will help to develop an understanding of the air pollution trend over the northern plain, which will be helpful to take requisite actions to reduce pollution in ways that have the greatest potential to benefit health.

9.4.8 Visibility Reduction

Visibility is the measure of how well an observer can view a scene. It serves as a good measure to specify a location's air quality given the fact that even laypersons can recognize when it undergoes impairment. Horizontal surface visibility range is critical not only for ease of movement of the surface transportation, but also for the aviation sector (Jaswal et al. 2013). The reduction in visibility occurs when tiny particles and gases in the atmosphere cause scattering or absorption of light. Internationally, fog is defined as a phenomenon when visibility is less than 1 km (Kumar and Goyal 2014). Smog is a dispersion of smoke within fog, and hence, visibility reduction is a part of it.

The impact of air pollutants on visibility reduction in the North Indian cities has been studied and reported (Choudhury et al. 2007; Guttikunda and Gurjar 2012; Sati and Mohan 2014; Ravindra et al. 2020). Most studies conclude that during winters, dense fog or smog is mostly widespread all through the north and northeast plains including the Indo-Gangetic Plain region, which results in visibility reduction, sometimes down to just a few meters. Jaswal et al. (2013) reported that winters coupled with fog, water vapor, and aerosol pollution led to a nationwide decrease in visibility, especially over the IGP.

Meteorological factors such as temperature, wind speed, wind direction, relative humidity, rainfall, and air pressure are key determinants for air pollution over a region. In the ambient atmosphere, the rate at which one pollutant is transformed into another depends upon these meteorological factors like availability of moisture, sunlight, and clouds, whereas the atmospheric vertical temperature determines pollutant's lifetime in the ambient atmosphere—whether they will get mixed or diluted or get trapped and accumulated. During the cleanest, clear, and windiest days, the emitted pollutants are well mixed in the ambient atmosphere. However, under stable conditions, either the pollutants will form a plume or stagnate over the site to form haze.

Low-visibility conditions are challenging for transportation and general movement and consequently lead to economic losses. Yadav and Rawal (2016) reported extremely poor visibility conditions during November 2016, down to around 200 m. Poor visibility was caused by dense smog over the IGP even before the fog season started and had thrown flights and rail schedule out of gear. In 2017, the smog event led to the delay of 69 trains and rescheduling of many others, while a lot of others were canceled. In 2019 smog event, nearly 150 flights got impacted, some diverted, and hundreds were delayed and canceled. Low visibility during this time also led to a high number of road accidents and affected the day-to-day life of people living in northern India.

The aviation sector in India is huge, and domestic and international air traffic is continuously increasing. For instance, Delhi's Indira Gandhi International (IGI) airport is ranked one of the busiest in the country and worldwide, meanwhile significantly contributing to Delhi's and national gross development product, as per a survey by the National Council for Applied Economic Research. The dense winter fog affects flight operations of IGI airport, causing interruptions, alterations in flight departures, or cancelations, trigger significant economic losses. A growing intensity and frequency of fog events over north India have resulted in significant increase in the low or poor visibility days during the last five decades. Kulkarni et al. (2019) reported that the number of fog/smog-ridden days with visibility less than 1000 m and dense fog/smog days with visibility less than 200 m over Delhi have doubled up in the last three decades. This severely affected flight operations, leading to increased diversion, delays, and cancelation of flights, ultimately causing economic losses to aviation industries, worth millions of rupees.

9.4.9 Economic Losses

The World Bank and the Institute for Health Metrics and Evaluation (IHME) at the University of Washington, Seattle, collaborated for a study and reported that governments across the world are prioritizing pollution control at the top of their agendas. However, any expenditure done for this will ultimately be competing with other budgetary priorities and policy objectives, further increasing an economic burden, particularly on developing nations. On a global scale, premature deaths due to air pollution are responsible for several trillion worths of losses annually (World Bank and IHME 2016).

Countries that have a greater proportion of younger population tend to suffer from higher pollution-related income losses because children have more years remaining in their lifetimes in which they are likely to work. If we start accounting for the mortality effect of the pollutants, the monetary burden is likely to increase substantially. However, most of this monetary burden of air pollution gets unnoticed and the impacted individuals continue to bear it.

Maji et al. (2017) reported that the particulate air pollution is a grave concern in urban cities in India. The study reported losses increasing from 0.34 million to 0.75 million due to PM_{10} triggered total DALYs from year 1995 to 2015. Due to PM_{10} , the estimated total costs for 2005 increased from 2714.10 million to 6394.74 million, almost by 135%, over Delhi, while a nearly 59% increase, from 2680.87 million to 4269.60 million, was observed for Mumbai, from 1995 to 2015 (Maji et al. 2017). For the non-OECD (Organisation for Economic Co-operation and Development) economies, including India, the welfare costs from premature deaths are projected to increase, from USD 1.7 trillion in 2015 to reach USD 15–22 trillion by 2060 (OECD 2016).

Ghatak et al. (2016) described the challenges of economic losses due to air pollution. They discussed that urbanization has not been carried out in a structured manner, which triggered different problems like air pollution, and the resulting diseases have caused more income loss than any other disease. Loss of wages and loss of labor productivity caused the consequential loss of household income, especially in an economy where labor is majorly manual and informal. It is this

informality that adds to the vulnerability of those affected, as casual employment signifies that all income losses will be borne by them.

9.4.10 Agricultural Loss

Cities are expanding, as a result of which the agricultural areas within the cities are interspersed with built-up areas and industry. They are therefore more exposed to air pollution also. The smog levels and air pollution are severe problems that eventually lead to crop yields being damaged. Burney and Ramanathan (2014) reported that significant impact on the crop yields is being observed due to the short-lived climate pollutants (SLCPs). States like Uttar Pradesh and Punjab in the IGP region are hit especially hard. In 2010, a 18.9% yield loss for wheat was particularly due to the SLCPs, translating roughly to 24 million tons of wheat loss worth almost 5 billion dollars. Burney and Ramanathan (2014) also asserted that increasing temperatures due to global warming also has significant negative impacts on crop yields. SLCPs like black carbon and ozone trigger climate change, influencing temperature, precipitation, radiation, etc., which then directly affects the plant yield and agricultural productivity (Sharma et al. 2016a).

Similarly, negative impacts of air pollution on the crop yield in Varanasi were corroborated when air pollutants like SO_2 , NO_2 , and O_3 were monitored alongside measurements of respective plant responses—their physiological characteristics, pigment, biomass and yield, and parameter reductions like biomass accumulation, yield, etc. (Agrawal et al. 2003; Agrawal et al. 2006).

Debaje (2014) reported that the highest winter wheat relative yield loss (RYL) was estimated in the North Indian states, where normally high O_3 concentrations prevail due to high NOx emissions and favorable atmospheric conditions for photochemical O_3 formation. O_3 damages the plants as it gets absorbed via stomatal pores on the surface of the leaf. It leads to formation of free radicals, which then attack the cell membranes, causing yellowing, cell injuries, bronzing, reddening, etc., grossly affecting the flowering, crop growth, and crop yield. The exposure of ozone concentration above 40 ppb can cause yield loss and many other agricultural losses. Nearly 90% of the vegetation is affected due to the ozone. Gupta et al. (2017) also reported reduced wheat yields in India by about 5.2%.

9.5 Conclusion

Smog has existed since the thirteenth century. Only the advancement in science during the mid-twentieth century revealed the pollutant species and their hidden chemistry behind it. The awareness coupled with stringent legislations helped many countries combat it. Presently, India is facing the same situation from last decade and the concern of scientists, government, and the public shifted toward the rising extreme events of smog over the Northern India. The reasons for the rising trend of smog events across the North India are numerous. The unique, almost ill-fated, geography of Northern India creating a valley effect, the rise in the emission of pollutants due to the boom in population, industries and energy sector (thermal power plants), etc., turned it into the smog center of our country. The stubble burning during the post-monsoon season is a well-recognized contributor of air pollution, is one of the reasons for formation of smog over northern India. The change in meteorological conditions after the monsoon season and during winters reverses the wind flow. The northwesterly winds transport pollutants emanating from the burning croplands to other IGP states.

The rising and intensifying smog events create a great number of problems not only for the people living in the northern region but also for the whole country. Even though the pollution events are being incessantly and largely observed, only some information is available on the social, economic, and health impacts of smog.

A vast number of studies have been conducted to identify the causes of these smog events, and presently, we do understand the key sources, which roots the foundation of smog over Northern India, but still every year, the region continues to get trapped under these perilous conditions. Immense challenges such as rising population and urbanization, increase in power and goods demand, rising waste production in urban areas, non-compliance of air pollution norms by industries and the general public, and the lack of adequate public transport facility are reasons that make a smog-free winter in North India a distant dream.

There is no doubt to the fact that the frequency of what used to be unusual events has risen in the past few years. There is a strong need to reassess the current situation and adopt measures that are sustainable, innovative, and in sync with the present technologically progressive setup. Policies about air pollution mitigation should be more pragmatic, involving sustainability not merely as an adjunct addition. One of the best ways to spur action in this regard is through education and awareness campaigns. Small initiatives by individuals who care to bring change can go a long way to tackle this problem at the very base.

References

- Agrawal M, Singh B, Rajput M, Marshall F, Bell JNB (2003) Effect of air pollution on peri-urban agriculture: a case study. Environ Pollut 126(3):323–329
- Agrawal M, Singh B, Agrawal SB, Bell JNB, Marshall F (2006) The effect of air pollution on yield and quality of mung bean grown in peri-urban areas of Varanasi. Water Air Soil Pollut 169(1): 239–254
- Balakrishnan K, Dey S, Gupta T, Dhaliwal RS, Brauer M, Cohen AJ, Stanaway JD, Beig G, Joshi TK, Aggarwal AN, Sabde Y (2019) The impact of air pollution on deaths, disease burden, and life expectancy across the states of India: the Global Burden of Disease Study 2017. Lancet Planet Health 3(1):e26–e39

Basu M (2019) The great smog of Delhi. Lung India 36(3):239

Bhaskar BV, Mehta VM (2010) Atmospheric particulate pollutants and their relationship with meteorology in Ahmedabad. Aerosol Air Qual Res 10(4):301–315

Bhome AB (2012) COPD in India: iceberg or volcano? J Thorac Dis 4(3):298-309

- Bray CD, Battye WH, Aneja VP (2019) The role of biomass burning agricultural emissions in the Indo-Gangetic Plains on the air quality in New Delhi, India. Atmos Environ 218:116983
- Brazell JH (1970) Meteorology and the clean air act. Nature 226(5247):694-696
- Burney J, Ramanathan V (2014) Recent climate and air pollution impacts on Indian agriculture. Proc Natl Acad Sci 111(46):16319–16324
- Chauhan A, Singh RP (2017) Poor air quality and dense haze/smog during 2016 in the Indo-Gangetic plains associated with the crop residue burning and Diwali festival. In: IEEE International Geoscience and Remote Sensing Symposium (IGARSS), pp 6048–6051
- Chawala P, Sandhu HAS (2020) Stubble burn area estimation and its impact on ambient air quality of Patiala & Ludhiana district, Punjab, India. Heliyon 6(1):e03095
- Chen TM, Kuschner WG, Shofer S, Gokhale J (2007) Outdoor air pollution: overview and historical perspective. Am J Med Sci 333(4):230–234
- Chetambath R, Jesin Kumar C (2019) Air pollution in Delhi and the health emergency. Indian Pract 72(11):8
- Choudhury S, Rajpal H, Saraf AK, Panda S (2007) Mapping and forecasting of North Indian winter fog: an application of spatial technologies. Int J Remote Sens 28(16):3649–3663
- Davis DL (2002) A look back at the London smog of 1952 and the half century since. Environ Health Perspect 110(12):A734–A735
- Debaje SB (2014) Estimated crop yield losses due to surface ozone exposure and economic damage in India. Environ Sci Pollut Res 21(12):7329–7338
- Evelyn J (1772) Fumifugium, or, the inconvenience of the aer and smoake of London (1661). Humphries, London
- Firket J (1936) Fog along the Meuse valley. Trans Faraday Soc 32:1192-1196
- Gaffney JS, Marley NA (2009) The impacts of combustion emissions on air quality and climate– From coal to biofuels and beyond. Atmos Environ 43(1):23–36
- Garg A, Gupta NC (2020) The great smog month and spatial and monthly variation in air quality in ambient air in Delhi, India. J Health Pollut 10(27):200910
- Gaur A, Tripathi SN, Kanawade VP, Tare V, Shukla SP (2014) Four-year measurements of trace gases (SO₂, NOx, CO, and O₃) at an urban location, Kanpur, in northern India. J Atmos Chem 71(4):283–301
- Ghatak A, Nandy D, Siddhanta S (2016). Burden of Diseases due to Air Pollution in Urban India. Gujarat Institute of Development Research.
- Goyal P, Mishra D, Kumar A (2013) Vehicular emission inventory of criteria pollutants in Delhi. Springerplus 2(1):1–11
- Greenstone M, Fan CQ (2018) Introducing the air quality life index-twelve facts about particulate air pollution, human health, and global policy. Energy Policy Institute at the University of Chicago
- Guo H, Kota SH, Sahu SK, Hu J, Ying Q, Gao A, Zhang H (2017) Source apportionment of PM_{2.5} in North India using source-oriented air quality models. Environ Pollut 231:426–436
- Gupta R, Somanathan E, Dey S (2017) Global warming and local air pollution have reduced wheat yields in India. Clim Chang 140(3–4):593–604
- Guttikunda S (2012) Air pollution in Delhi. Econ Polit Wkly 47(26-27):24-27
- Guttikunda SK, Goel R (2013) Health impacts of particulate pollution in a megacity—Delhi, India. Environ Dev 6:8–20
- Guttikunda SK, Gurjar BR (2012) Role of meteorology in seasonality of air pollution in megacity Delhi, India. Environ Monit Assess 184(5):3199–3211
- Guttikunda SK, Goel R, Pant P (2014) Nature of air pollution, emission sources, and management in the Indian cities. Atmos Environ 95:501–510
- Haagen-Smit AJ (1952) Chemistry and physiology of Los Angeles smog. Ind Eng Chem 44(6): 1342–1346
- Jain N, Bhatia A, Pathak H (2014) Emission of air pollutants from crop residue burning in India. Aerosol Air Qual Res 14(1):422–430

- Jain S, Sharma SK, Choudhary N, Masiwal R, Saxena M, Sharma A, Mandal TK, Gupta A, Gupta NC, Sharma C (2017) Chemical characteristics and source apportionment of PM_{2.5} using PCA/APCS, UNMIX, and PMF at an urban site of Delhi, India. Environ Sci Pollut Res 24(17):14637–14656
- Jain S, Sharma SK, Vijayan N, Mandal TK (2020) Investigating the seasonal variability in source contribution to PM_{2.5} and PM₁₀ using different receptor models during 2013–2016 in Delhi, India. Environ Sci Pollut Res Int 28(4):4660–4675
- Jaswal AK, Kumar N, Prasad AK, Kafatos M (2013) Decline in horizontal surface visibility over India (1961–2008) and its association with meteorological variables. Nat Hazards 68(2): 929–954
- Kandlikar M, Ramachandran G (2000) The causes and consequences of particulate air pollution in urban India: a synthesis of the science. Annu Rev Energy Environ 25(1):629–684
- Khan RK, Strand MA (2018) Road dust and its effect on human health: a literature review. Epidemiol Health 40:e2018013
- Kulkarni R, Jenamani RK, Pithani P, Konwar M, Nigam N, Ghude SD (2019) Loss to aviation economy due to winter fog in New Delhi during the winter of 2011–2016. Atmosphere 10(4): 198
- Kumar S, Goyal P (2014) Impact of a gas based power plant on the air quality of Delhi, India. J Environ Res Dev 8(3):523
- Lohan SK, Jat HS, Yadav AK, Sidhu HS, Jat ML, Choudhary M, Peter JK, Sharma PC (2018) Burning issues of paddy residue management in north-west states of India. Renew Sust Energ Rev 81:693–706
- Maji KJ, Dikshit AK, Deshpande A (2017) Disability-adjusted life years and economic cost assessment of the health effects related to PM_{2.5} and PM₁₀ pollution in Mumbai and Delhi, in India from 1991 to 2015. Environ Sci Pollut Res 24(5):4709–4730
- Mittal SK, Susheel K, Singh N, Agarwal R, Awasthi A, Gupta PK (2009) Ambient air quality during wheat and rice crop stubble burning episodes in Patiala. Atmos Environ 43:238–244
- Murari V, Singh N, Ranjan R, Singh RS, Banerjee T (2020) Source apportionment and health risk assessment of airborne particulates over central Indo-Gangetic Plain. Chemosphere 257:127145
- Nagar PK, Sharma M, Das D (2019) A new method for trend analyses in PM₁₀ and impact of crop residue burning in Delhi, Kanpur and Jaipur, India. Urban Clim 27:193–203
- OECD (2016) Economic consequences of outdoor air pollution. Organisation for Economic Co-operation and Development
- Peshin SK, Sharma A, Sharma SK, Naja M, Mandal TK (2017) Spatio-temporal variation of air pollutants and the impact of anthropogenic effects on the photochemical buildup of ozone across Delhi-NCR. Sustain Cities Soc 35:740–751
- Polivka BJ (2018) The Great London smog of 1952. Am J Nurs 118(4):57-61
- Raatikainen T, Hyvärinen AP, Hatakka J, Panwar TS, Hooda RK, Sharma VP, Lihavainen H (2014) The effect of boundary layer dynamics on aerosol properties at the Indo-Gangetic plains and at the foothills of the Himalayas. Atmos Environ 89:548–555
- Rajaram BS, Suryawanshi PV, Bhanarkar AD, Rao CVC (2014) Heavy metals contamination in road dust in Delhi city, India. Environ Earth Sci 72(10):3929–3938
- Rajput P, Singh DK, Singh AK, Gupta T (2018) Chemical composition and source apportionment of sub-micron particles during wintertime over Northern India: new insights on influence of fog-processing. Environ Pollut 233:81–91
- Rana M, Mittal SK, Beig G (2019) Enhanced ozone production in ambient air at Patiala semi-urban site during crop residue burning events. MAPAN 34(2):273–288
- Ravindra K, Singh T, Mor S, Singh V, Mandal TK, Bhatti MS, Gahlawat SK, Dhankhar R, Mor S, Beig G (2019) Real-time monitoring of air pollutants in seven cities of North India during crop residue burning and their relationship with meteorology and transboundary movement of air. Sci Total Environ 690:717–729

- Ravindra K, Singh T, Sinha V, Sinha B, Paul S, Attri SD, Mor S (2020) Appraisal of regional haze event and its relationship with PM_{2.5} concentration, crop residue burning and meteorology in Chandigarh, India. Chemosphere 273:128562
- Rizwan S, Nongkynrih B, Gupta SK (2013) Air pollution in Delhi: its magnitude and effects on health. Indian J Community Med 38(1):4–8
- Sahu SK, Beig G, Parkhi NS (2011) Emissions inventory of anthropogenic PM2. 5 and PM10 in Delhi during Commonwealth Games 2010. Atmos Environ 45(34):6180–6190
- Sarkar C, Kumar V, Sinha V (2013) Massive emissions of carcinogenic benzenoids from paddy residue burning in North India. Curr Sci 104(12):1703–1709
- Sati AP, Mohan M (2014) Analysis of air pollution during a severe smog episode of November 2012 and the Diwali Festival over Delhi, India. Int J Remote Sens 35(19):6940–6954
- Sawlani R, Agnihotri R, Sharma C, Patra PK, Dimri AP, Ram K, Verma RL (2019) The severe Delhi SMOG of 2016: a case of delayed crop residue burning, coincident firecracker emissions, and atypical meteorology. Atmos Pollut Res 10(3):868–879
- Sawlani R, Agnihotri R, Sharma C (2020) Chemical and isotopic characteristics of PM_{2.5} over New Delhi from September 2014 to May 2015: evidences for synergy between air-pollution and meteorological changes. Sci Total Environ 763:142966
- Saxena P, Naik V (eds) (2018) Air pollution: sources, impacts and controls. Cabi, Wallingford, pp 1–217
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Sembhi H, Wooster MJ, Zhang T, Sharma S, Singh N, Agarwal S, Boesch H, Gupta S, Misra A, Tripathi SN, Mor S (2020) Post-monsoon air quality degradation across Northern India: assessing the impact of policy-related shifts in timing and amount of crop residue burnt. Environ Res Lett 15:104067
- Sharma SK, Mandal TK, Saxena M, Rashmi, Sharma A, Datta A, Saud T (2014) Variation of OC, EC, WSIC and trace metals of PM₁₀ over Delhi. J Atmos Sol Terr Phys 113:10–22
- Sharma A, Sharma SK, Rohtash, Mandal TK (2016a) Influence of ozone precursors and particulate matter on the variation of surface ozone at an urban site of Delhi, India. Sustain Environ Res 26(2):76–83
- Sharma SK, Sharma A, Saxena M, Choudhary N, Masiwal R, Mandal TK, Sharma C (2016b) Chemical characterization and source apportionment of aerosol at an urban area of Central Delhi, India. Atmos Pollut Res 7(1):110–121
- Sharma A, Mandal TK, Sharma SK, Shukla DK, Singh S (2017) Relationships of surface ozone with its precursors, particulate matter and meteorology over Delhi. J Atmos Chem 74(4): 451–474
- Sharma R, Kumar R, Sharma DK, Priyadarshini I, Pham BT, Bui DT, Rai S (2019) Inferring air pollution from air quality index by different geographical areas: case study in India. Air Qual Atmos Health 12(11):1347–1357
- Shyamsundar P, Springer NP, Tallis H, Polasky S, Jat ML, Sidhu HS, Krishnapriya PP, Skiba N, Ginn W, Ahuja V, Cummins J (2019) Fields on fire: alternatives to crop residue burning in India. Science 365(6453):536–538
- Singh A, Rastogi N, Patel A, Singh D (2016) Seasonality in size-segregated ionic composition of ambient particulate pollutants over the Indo-Gangetic Plain: source apportionment using PMF. Environ Pollut 219:906–915
- Singh T, Biswal A, Mor S, Ravindra K, Singh V, Mor S (2020) A high-resolution emission inventory of air pollutants from primary crop residue burning over Northern India based on VIIRS thermal anomalies. Environ Pollut 266:115132
- Slutsky R (2017) Delhi's air pollution and its effects on children's health. Yale Global Health Review

- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sonwani S, Saxena P (eds) (2022) Greenhouse gases: sources, sinks and mitigation. Springer-Nature, Singapore, pp 1–255
- Sonwani S, Saxena P, Shukla A (2021) Carbonaceous aerosol characterization and their relationship with meteorological parameters during summer monsoon and winter monsoon at an industrial region in Delhi, India. Earth Space Sci 8(4):e2020EA001303
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1–2):39–50
- Srivastava A, Gupta S, Jain VK (2008) Source apportionment of total suspended particulate matter in coarse and fine size ranges over Delhi. Aerosol Air Qual Res 8(2):188–200
- Suryawanshi PV, Rajaram BS, Bhanarkar AD, Rao CC (2016) Determining heavy metal contamination of road dust in Delhi, India. Atmósfera 29(3):221–234
- Takigawa M, Patra PK, Matsumi Y, Dhaka SK, Nakayama T, Yamaji K, Kajino M, Hayashida S (2020) Can Delhi's pollution be affected by crop fires in the Punjab region? SOLA 16:86–91
- Terry JP, Jia G, Boldi R, Khan S (2018) The Delhi 'gas chamber': smog, air pollution and the health emergency of November 2017. Weather 73(11):348–352
- Tiwari S, Saxena P (2021) Air pollution and its complications. Springer-Nature, Singapore, pp 1–178
- Vadrevu K, Lasko K (2018) Intercomparison of MODIS AQUA and VIIRS I-Band fires and emissions in an agricultural landscape—implications for air pollution research. Remote Sens 10(7):978
- World Bank, Institute for Health Metrics and Evaluation (2016) The cost of air pollution: strengthening the economic case for action
- Yadav S, Rawal G (2016) The great Delhi smog. Indian J Immunol Respir Med 1(4):78-79
- Zimmer A, Nemery B (2017) The Meuse Valley (1930) Just fog or industrial pollution. Air Pollut Episodes 6:27



Volcanic Emissions: Causes, Impacts, and Its **10** Extremities

Rahul Kant Yadav, Debasish Mahapatra, and Chinmay Mallik

Abstract

Volcanoes are an extreme expression of the mostly invisible movements of the fragments of the earth's crust connecting the interior of the earth to its exterior. Since eons, they have had a major impact on the exterior surface of the earth and its gaseous envelope modulating the earth's surface and its atmosphere. Depending on the type, intensity, and location of an eruption, the impacts can be both short term and long term, mild to extreme. The intensity of a volcanic eruption represents its explosivity in terms of volume of material ejected, ejection height, and distance the volcanic cloud travels away from its origin, which can span a wide range of scales. The requirement of a logarithmic scale to classify an eruption itself indicates how extreme one volcano can be compared to another. Volcanic emissions encompass all the three states of matter with tephra in solid form, lava in liquid form, and several acidic gases. The volume of ejected material for a very large eruption can amount to several cubic miles as in case of Krakatau (1883). Strong eruption can be related to silica-rich magma leading to large viscosity enabling trapping of gases until pressure builds up high enough to cause explosive eruption. The atmospheric impacts of volcanic eruptions range from small particles and ash playing havoc with aviation to ozone-depleting gases to global cooling as a result of emitted sulfur gases. The most explosive volcanoes can easily reach over 20 km and can lead to several feet of ash even 150 km away from the eruption site. Relationships between volcanic events and ENSO are an active field of research. Severe volcanic eruptions like Mt. Pinatubo have had an impact in all the five major spheres of the earth including the lithosphere,

R. K. Yadav · D. Mahapatra · C. Mallik (🖂)

Department of Atmospheric Science, School of Earth Sciences, Central University of Rajasthan, Kishangarh, Rajasthan, India

e-mail: chinmay.mallik@curaj.ac.in

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_10

hydrosphere, atmosphere, biosphere, and the anthroposphere. The potential severe consequences of volcanic eruptions demand greater awareness and investigation of these extreme events with extreme consequences.

Keywords

Volcano · Eruption causes · Extreme events · Atmospheric impacts · Tephra · Volcanic gases

10.1 Introduction

The earth as a planet is dynamic. There are forces on and inside the planet that bring about drastic changes to the planet's interior, surface, and its gaseous envelope, the atmosphere. While the surface forces change the surface of the planet gradually in hundred thousand and millions of years through various exogenic processes like weathering and erosion, there are other changes that may be brought about by endogenic processes. Endogenic processes include diastrophic processes (orogenic or mountain-building processes and epeirogenic processes or continent-building processes) and sudden movements like earthquakes and volcanism.

A volcano is a rupture on the earth's crust through which hot lava, volcanic ash, and gases are spewed out during the course of the eruption from a magma chamber underneath the surface. Most of them are found near the tectonic plate boundaries, where they converge and subduct or diverge. Volcanism can be caused by diapirs 3000 km deep within the earth forcing their way from the core-mantle boundary, with impacts far away from plate boundaries. As a result of this hotspot, volcanism occurs, for example, the Hawaiian hotspot and the Reunion hotspot. As the planet's 71% of total surface area is covered by water, it is only logical that most of these volcanoes are found underwater. Depending upon the type and region of eruption, a volcanic eruption may have a very profound effect on the atmosphere and the environment. They have the potential to change the global climate. As a matter of fact, this has occurred in the past geological age. The evidence of these ancient catastrophic eruptions can be found in the stratigraphic column in many places in the world. Other proxies include evidence of tephra deposits from ice cores and from tree rings, which make valuable paleoclimatic proxies to effectively describe the impacts of volcanic gases and tephra on the atmosphere. While the impacts of volcanic eruptions can be alarming and devastating, their scale and magnanimity provide fascinating and awesome insights. The Vesuvius eruption in 79 AD, despite the damage caused, preserved for us Pompeii and insight into that civilization (Capellas 2007). The Toba eruption in Indonesia was suspected to spew almost 230 cubic miles of erupted material on the planet's surface 75,000 years ago. Indonesia is a region of large eruptions with Mt. Tambora being one of the most explosive ones, killing thousands of people and eclipsing the sun in 1815. The Mt. Pelee eruption notoriously subdued a whole city in its lava killing thirty thousand people in the Caribic in 1932. The Pinatubo eruption of 1991 is known all across the globe for its explosivity and atmospheric impacts including plummeting temperatures and large-scale ozone depletion. The most famous observatory for climate research is located in the cradles of one of largest volcanoes of the world, the Mauna Loa. While eruptions like Krakatoa (1927) have left us with fear due to the devastating effects, eruptions like that of the Etna are impacting the earth and its atmosphere for millions of years now. Though India is a region to many extinct volcanoes, there are also active ones as in Barren and Baratang Islands. However, volcanic eruptions do not recognize any borders and plumes can span transoceanic and transcontinental domains. The Dalaffila eruption in Ethiopia caused SO_2 enhancement as far as Delhi in India (Mallik et al. 2013).

As volcanoes can connect the whole world in terms of their impacts, it is necessary to understand the science of volcanoes in reach of common people. The rationale of this book chapter is to provide a holistic walk pertaining to volcanoes and their atmospheric impacts. It aims to simplify and summarize the information available regarding volcanic eruptions from an atmospheric science perspective. First, we introduce the different types of volcanoes and then discuss the causes of volcanic emissions. This discussion is followed by the impacts of volcanic emissions mainly from an atmospheric and climate perspective and ends with the case study of the most famous volcano the Mt. Pinatubo of 1991.

10.1.1 Types of Volcanoes

To understand the magnanimity of volcanic eruptions, it is essential to understand the type of volcanoes and how they erupt. Depending upon the basis of classification of structure or activity, volcanoes can be grouped into several types as mentioned below.

10.1.1.1 On the Basis of Activity

(a) Active volcanoes: According to the Smithsonian global volcanism program, a volcano is considered active if it has erupted in the Holocene within the last ten thousand years. Most active volcanoes can be traced to the Pacific ring of fire. According to the Smithsonian global volcanism program, a volcano is considered active if it has erupted in the Holocene, i.e., in the last 10,000 years, while the International Association of Volcanology and Chemistry of the Earth's Interior (IAVCEI) considers historical times to define active volcanoes. So, there is actually no consensus among volcanologists on the definition of an active volcanoe is shown in Fig. 10.1. The Smithsonian Global Volcanism Program (SGVP) has documented till date about 560 volcanoes whose historical eruptions have been confirmed (Global Volcanism Program 2013). Despite the fact that the SGVP has documented over 10,000 Holocene eruptions, only six eruptions account for more than half of the total quantified fatalities (Cottrell 2015).



Fig. 10.1 Active volcanoes around the world and the top nine most explosive volcanoes of the past 2000 years superimposed on global aviation route map. (1) Samalas, in 1257. (2) Kuwae of 1452. (3) Tambora in 1815. (4) Mount Churchill in 700 CE. (5) Rabaul Caldera between 531 to 566 CE. (6) The Rabaul caldera in Papua New Guinea is the fifth and sixth from 531 to 566 CE. (7) Quilotoa in 1280. (8) Ilopango in 450 CE. (9) Grimsvotn and Laki eruptions around 1785. The active volcano locations are based on https://volcano.si.edu/list_volcano_holocene.cfm#. The route data are taken from https://openflights.org/data.html and filtered on Codeshare and Airline. The Codeshare filter keeps null. The Airline filter keeps 13 members. The view is filtered on exclusions (Destination airport, Path ID, Source airport), which keeps 75,174 members (*illustration*: Debasish Mahapatra & Madhumay Mallik)

- (b) Dormant volcanoes: Dormant volcanoes are the ones that have not been associated with any eruption in the last ten thousand years but may erupt in the near future. Such volcanoes can erupt suddenly after remaining dormant for a fairly long time and then. The Geological Survey of India has classified the Narcondam Island in the Andaman and Nicobar archipelago as a dormant volcano. Other examples are Mount Kilimanjaro in Tanzania and Mount Fuji in Japan.
- (c) Extinct volcanoes: They are the ones that are thought unlikely to erupt again most likely because it does not have a magma supply. Some examples of extinct volcanoes are Hohentwiel of Germany and Zuidwal in the Netherlands. It is difficult to determine whether the volcano is fully extinct. There are also many active volcanoes in India dating about 750 million years before present.

10.1.1.2 On the Basis of Structure

On the basis of structure, volcanoes may be identified as fissure volcano, shield volcano, composite volcano, supervolcano, submarine volcano, and subglacial volcano. The internal structure of a volcano is shown in Fig. 10.2 along with the eruption cloud.

(a) **Fissure volcano:** Fissure volcanoes are fractures on the earth's crust through which lava flows. Large fissure eruptions are considered as one of the hazardous


Fig. 10.2 Structure of a volcano (adapted from abc.net.au and modified)

volcanic situations because of their huge arrival of gases and aerosols into the troposphere and lower stratosphere (Ilyinskaya et al. 2017). The Holuhraun eruption in Iceland in 2014–2015 is one of the biggest such occasions.

- (b) Shield volcano: It is a large volcano with a very large diameter and very gentle slopes. The name suggests the similarity of its structure to that of a warrior's shield. It is a low-angle volcano formed by the piling up of low-viscosity lava flow, mainly basaltic lava (Paris 2013) Most of these volcanoes are in the mantle plume region and a result of hotspot volcanism. The magma is almost always mafic, basaltic, and basic, and the eruption is generally effusive. They include the largest volcano on earth like Mauna Loa in Hawaii. Giant shield volcanoes are also discovered in Mars and Venus. The largest volcano in the solar system is found on Mars, known as Olympus Mons, and is interestingly a shield volcanoes in the inner planets of the solar system.
- (c) Composite Volcano: They are the most common types of volcanoes. Almost all of these volcanoes are found near subduction zones, i.e., destructive plate boundaries. The lava is acidic, felsic, and ranges from rhyolitic and dacitic to andesitic. Because the lava is highly viscous and has relatively lower temperature than basic lava, it solidifies rapidly. The rapid solidification of lava in the vents sometimes clogs it and thereby increasing the pressure in the vent. This

results in a loud explosion, and the volcano throws a large amount of volcanic bombs and other pyroclastic material. These volcanoes can also be commonly known as stratovolcanoes as they are built by many layers of solidified felsic lava and tephra. These volcanoes have steeper sides and a summit crater that erupts periodically. Some of them have collapsed summit craters called Calderas. Owing to its high viscosity, the lava does not flow extensively. Examples of stratovolcanoes include the Krakatoa in Indonesia, which had a catastrophic eruption in 1883 and the recent Mount Pinatubo eruption. Simkin and Siebert (1994) said that at least 700 stratovolcanoes are active. Pvroclastic density currents, also called pyroclastic flows, which are sort of an avalanche of volcanic tephra, are possibly the most catastrophic among all impacts of volcanic hazards. The density of the current, the gradient of the slope, and the volcanic output rate are deciding factors of its speed. Its kinetic energy is enough to vanguish building, trees, crops, and almost everything in its path. They can incinerate any living organism and that is what makes it particularly so lethal. Example of such an event was the devastation of Pompeii when it was, along with some of its inhabitants, buried by the pyroclastic flows from the eruption of Mount Vesuvius in 79 CE. Stratovolcanoes are on the eighth out of top ten killer volcanoes over twentieth century (Witham 2005), and it has killed more than 0.2 million people since 1783 AD (Tanguy et al. 1998). Mount St. Helens, a stratovolcano in Washington state that erupted on May 18, 1980, caused devastation of 200 sq. miles of forest winds that carried the volcanic ash eastward over the USA during the day, causing total darkness in Spokane, Washington, 236 miles from the volcano (Nania and Bruya 1982). Stratovolcanoes have the most considerable effect on local weather and can also considerably affect the climate of one or both hemispheres and significantly modulate the atmospheric chemistry and the earth's energy balance.

- (d) Supervolcano: According to the USGS, these eruptions can exhibit of Volcanic Explosivity Index (VEI) of 8 and have ejected more than 1000 km³ of materials. VEI is used to measure the explosiveness of an eruption. Supervolcanic eruptions are extremely rare and the most dangerous types. Supervolcanoes can also be associated with formation of widespread lava flows and basaltic traps, and Deccan traps in India and Siberian traps in Russia are such examples of such large igneous provinces (LIPs). These are very large eruptions and can usually be related to hotspot (Yellowstone) and destructive plate boundaries (Toba) (Wotzlaw et al. 2015; Budd et al. 2016). These can cause lasting changes to the climate and can cause extinction events. The Siberian Traps coinciding with the Permian–Triassic extinction event are the largest flood basalt event and were formed around two hundred fifty million years ago and. The Deccan Traps are produced by the Réunion hotspot, currently lying in the Indian Ocean under the Réunion Island, some 66 Mya, and are coincident with the Cretaceous–Paleogene extinction event (Keller 2014).
- (e) **Submarine volcano:** More than 70% of all volcanic eruptions occur underwater. Submarine volcanoes are the volcanoes that are formed underwater. Most submarine volcanoes are found near divergent plate boundaries in the

mid-oceanic ridge. The total number of submarine volcanoes is indeed very high and easily reaches a million, although most of these types of volcanoes are now extinct (Speight and Henderson 2013). Most submarine volcanoes are found in deep waters, but some are present in shallow waters and reveal their presence when they erupt with steam and eject rock high above the surface of the sea. Hydrothermal vents found near submarine volcanoes are rich sites of biological activity. The unlimited supply of water changes the characteristics of these eruptions. The weight of the copious amount of water prevents explosive eruption of the volcano, and when magma comes in contact with the water, it cools rapidly and turns into Obsidian, which is a volcanic glass and advancing lava flows from pillow lava. Seamounts are submarine volcanoes that are extinct and rise abruptly from the seafloor. Despite the fact that no big explosive undersea eruption is yet documented, prevalence of such eruptions is evident from ancient seafloor deposits. Large-scale, caldera-forming explosive submarine eruptions of silicic magmas have occurred on the modern seafloor south of Japan in the geological past, and in 2012, the sea surface expression of such a large-scale explosive eruption was identified for the first time ever (White et al. 2015).

(f) **Subglacial Volcano:** These types of volcanoes are also known by the name of Glaciovolcano, which represents a volcano that is formed beneath the surface of a glacier or ice sheet. Most of these volcanoes are found in Iceland and Antarctica. During the eruption of the Subglacial Volcano, the ice over it melts forming a lake, and the water cools the lava rapidly and thus forming pillow lava shapes similar to underwater eruptions. On a later stage, however the pillow will break off and fall down and form hyaloclastite, tuff breccia, and pillow breccia. If a large amount of meltwater is formed, these volcanoes can produce dangerous floods called a Jökulhlaup. A Jökulhlaup is an Icelandic term and refers to a flood that is produced due to a subglacial volcano melting the surrounding glacier. These volcanoes are generally flat in the early stages due to the weight and pressure of the overlying layer of ice and water. But in later stages, they can erupt subaerially and form the more conventional conical shape. In late evening of September 30, 1996, a Jökulhlaup occurred following the subglacial eruption on the Vatnajökull ice sheet, Iceland. On November 7, the Jökulhlaup came to an end. A new subglacial ridge 6-7 km long was produced by volcanic materials with a volume of 0.7–0.75 km³ and floods affected an area of 750 km², stretching the shoreline by 800 m. The eruption and flood were Iceland's fourth largest natural disasters in the twentieth century (Smellie 1999).

10.2 Causes of Volcanic Emissions

Most of the volcanoes are not continuously in a vigorous state of activity. They spend a lot of their life span at rest, typically for thousands of years before eruption. The landforms and eruptive nature of volcanoes are diversified, reflecting numerous complex interactions, which determine how magma is generated, stored, rises up along dykes, and conduits and finally erupts. Eruptions are **inveigled** by the tectonic setting. Additionally, the different properties of earth's crust can impact the final eruption, and hence, the history of the volcano can also play a major role as past eruptions can also modify these governing properties. However, despite the great uncertainty about how and where different volcanoes can erupt, the causes can be mostly grouped into a set of common chemical and physical processes. Understanding the formation and eruption of volcanoes and the consequences of these eruptions can be obtained from an understanding of geographic and geologic processes, which can cause melting of rocks to melt and alter its composition, leading to the storage of molten magma in the earth's crust and eventually causing into rising to the surface, and interact as well as impact the surroundings.

10.2.1 Plate Tectonics

We have already discussed how dynamic our planet is. Events like earthquake and volcanic eruptions have occurred throughout the history and are constantly changing and reshaping our planet. Yet, it is only a few decades ago that we got a glimpse into why these events occurred that could explain almost every important question in the field of geology. It is like the grand unified theory of earth science. This fascinating new theory is called plate tectonics. According to plate tectonics, the crust is broken into many rigid plates and they move horizontally with respect to each other, which means the crust is not stationary, it is mobile, and the continents move around on earth's surface. As these plates move, they interact with each other and these interactions build the most amazing features like mountain belts and volcanoes and deep ocean trenches. The theory was built on the concept of continental drift and seafloor spreading. In the year 1912, a meteorologist by the name of Alfred Wegener described the idea of continental drift and gave the theory in his book "The Origin of Continents and Oceans." This theory was endorsed by the scientific community after the idea of seafloor spreading was validated by the works of W.J. Morgan and J.T. Wilson in the 1960s and 1970s (Bangar 2008).

10.2.1.1 Crustal Plates

The earth's crust is composed of several rigid but relatively thin plates. The earth's outer layers are categorized into lithosphere and asthenosphere. The lithosphere is not only cooler but also more rigid compared to the asthenosphere, which is much more hotter and plastic. On an average, these plates are 125 km thick. Continental plates (200 km) are thicker than oceanic plates (50–100 km). The plates slide over the partially molten and plastic asthenosphere. The asthenosphere occurs below the lithosphere in the upper mantle at depths of 100–200 km. This zone is highly viscous, ductile, and mechanically weak. In this zone, the velocity of seismic waves decreases, and thus, it has been called the low-velocity zone (Forsyth 1975). The LVZ's lower boundary is located at a depth of 180–220 km (Condie 1997). Due to the high temperature and pressure conditions in the asthenosphere, the

rocks become ductile and plastic, deforming and moving slowly at speeds of cm/y, and eventually moving thousands of kilometers in millions of years. The rigid lithosphere is thought to float on the slowly flowing asthenosphere (Garrison and Ellis 2016). The plates can move with respect to one another format an annual rate of 0–100 mm (Read and Watson 1975). The earth's lithosphere is divided into 20 crustal plates, of which 7 are large and are called major plates, while the rest are called minor plates. The large plates are as follows: (1) the North American plate, (2) the South American plate, (3) the Eurasian plate, (4) the African plate, (5) the Indo-Australian plate, (6) the Pacific Plate, and (7) the Antarctic plate. Some plates are so large that they are constituted from both continental and oceanic crustal portions like the African plate and the South America plate and North American Plate, while the Pacific plate is almost entirely ocean.

10.2.1.2 Plate Boundaries

Plate margins are regions of meeting of two plates. Almost all seismicity volcanicity and tectonic activity happen around plate margins. The majority of the world's active volcanoes occur along these plate boundaries (Fig. 10.1). Depending open the relative motions of the adjoining plates, the *plate boundaries can be classified into three types*.

(a) Divergent plate boundaries

Divergent plate boundaries additionally known as constructive plate boundaries are places where plates move away from each other. This process happens along spreading center where the plates are moving apart and the magma that rises up from the asthenosphere to fill the newly created gap forming the new lithosphere. This phenomenon is called seafloor spreading. Perhaps the best known example of a divergent boundary is the mid-Atlantic ridge. Such boundaries are characterized by high volcanic activity and shallow focus earthquakes. Most of earth's volcanism occurs on the seafloor along spreading zones. These volcanoes are mostly submarine. Iceland being the most famous exception is on the mid-oceanic ridge and has high volcanic activity. The diverging and thinning lithosphere plates allow hot mantle rock to rise and thus causing decompression melting. Ultramafic mantle rock mostly consisting of peridotite partially melts to produce basaltic magma. This is why almost all volcanoes on the ocean floor are basaltic in nature. This basaltic lava erupts underwater and forms pillow-shaped pillow basalt. Deep-sea hydrothermal vents or black smokers are vents that are tall and emit black hot mineral-rich water near mid-oceanic ridges enable entire ecosystems to thrive underwater. Thus, the oceanic ridges are the newest part of earth's lithosphere. The rates of seafloor spreading at ridges vary from 1 cm to 6 cm per year. Some volcanoes also erupt at continental rifts where crustal thinning is caused by diverging continental plates. One such example is the East African Rift Valley.

(b) Convergent Plate Boundaries

A convergent plate boundary, additionally referred to as destructive plate boundary, is the type of plate margin where two or more plates move toward each other. Eventually depending upon the densities, one plate slides beneath the other by a process called subduction. Convergent boundaries occur between oceanic–oceanic plate, oceanic–continental plate, and continental–continental plate. These collisions between the plates have occurred over millions of years and can lead to volcanic activity and sometimes even earthquakes and orogenesis.

Oceanic–continental plate Continental lithosphere is of lower density and thus more buoyant than oceanic lithosphere. By the process of subduction, the denser oceanic plate is pushed beneath the less-dense continental plate when continental and oceanic plates converge. As the subducting plate is forced deeper into the mantle, the temperature reaches the melting point and the plate starts to melt. This partial melting produces magma chambers under the overriding continental plates. And these magma chambers are more buoyant and thus begin its slow ascent often melting and fracturing the overlying layers of rock. If and when a magma chamber rises to the surface, the magma will cause a fissure on the ground and will break through the crust in the form of a volcanic eruption. When multiple eruptions occur in a chain on the overriding plate, it is called a volcanic arc. Some of the well-known examples of continental volcanic are the Cascade Mountains in the western North America and the Andes Mountains in South America. Here, the dense Juan de Fuca oceanic plate dives beneath the North American Plate and the Nazca Plate is subducting under the South American plate, respectively.

Oceanic–oceanic plate The characteristics of a subduction zone in which an oceanic plate is forced to sink under another oceanic plate are the same as a continent– ocean subduction zone. The older, denser plate as mentioned before subducts underneath the less-dense plate. The region where the plate is pushed down into the mantle is marked as an ocean trench. In such a region, there can exist a line of volcanoes that thrives on the upper oceanic plate and is popularly known as an island arc.

Continent–continent plate When two continental plates collide, they can form another type of convergent plate boundary known as continent–continent plate. The continental lithosphere although very think cannot subduct as it is low in density. Thus, during the collision of two continental plates, as subduction of one of the plates is not a possibility, the plates just smash together, which forces the material to up as it cannot go down. The result of the tremendous forces of the collision is the genesis of earthquakes and metamorphic rocks. These continent–continent collision zones are not conducive for volcanoes as the crust is too thick for magma to get through.

(c) Transform Fault Boundaries

These are boundaries along which plates slide past one another. Hence, at transform fault boundaries, there is no production or destruction of lithosphere. The transform faults are characterized by shallow focus earthquakes with horizontal slips.

Hotspot Volcanism Hotspot volcanism is different as it does not occur at the zones of convergence and divergence, rather this type of volcanism occurs at the interior parts of the lithospheric plates. The cause of hotspot volcanism can be traced to abnormally hot centers in the mantle, which are commonly also known as mantle plumes.

10.3 Emissions from Volcanoes

The volcanic eruption can be violently explosive to gently effusive and can persist for few hours to several decades (Siebert et al. 2015). The volcanic eruption may get triggered due to internal magmatic system of the volcano or due to external processes like precipitation, landslides, or earthquakes. The magnitude and intensity of the volcanic eruption are usually described by its total erupted mass and the mass eruption rate and were quantified by Pyle (2015):

Magnitude = \log_{10} (mass, in kg) - 7, and

Intensity = \log_{10} (mass eruption rate, in kg/s) + 3

Newhall and Self (1982) introduced "The Volcano Explosivity Index" (VEI). VEI scale is logarithmic with indices ranging from 0 to 8. The 0 value is given to nonexplosive eruptions, and the largest volcanic eruptions in history are given the value of 8. Events with smaller VEI are usually common, whereas larger VEI events are dramatically less continuous (Siebert et al. 2015).

The style of volcanic eruption includes factors like emission span and relentlessness, extent, gas transition, column height, and association of magma or potentially external source of water. Grouping of eruption style is often qualitative and dependent on historical records of trademark emissions from type volcanoes. Notwithstanding, many sort volcanoes show a scope of emission styles over the long haul, which has led to terms, for example, subplinian or violent Strombolian.

On the basis of the nature of volcanic eruptions, volcanoes can be classified as Plinian, Vulcanian, and Strombolian.

- (a) Plinian eruptions are highly explosive continuously sending out a variety of ejecta with a wide range of sizes. Pyroclastic flows result from gravitational collapse of Plinian eruption columns or lateral explosions and from gravitational collapse of lava domes (Nakada 2000). The explosive eruption generates an enormous amount of tephra, and belches ash, which can exceed 11 km in height into the atmosphere (Auger et al. 2001). These are accompanied by excessive degassing.
- (b) Strombolian eruptions are also violent in nature sending out ejecta tens of meters into the atmosphere. Low-viscosity, silica-poor magmas characterize Strombolian eruptions and include the complex bursting of gas pockets around the volcanic vent (Taddeucci et al. 2015).

(c) Vulcanian eruptions are short-lived, ash-rich, and generate bombs and blocks. In an early earth, gas from volcanic eruptions resulted in the formation of our atmosphere and produced our oceans, without which life would never have existed on this planet. The volcanic emissions majorly contain water vapor (H₂O), carbon dioxide (CO₂), sulfur gases (SO₂, COS, H₂S, CS₂), and hydrogen halides (HCl, HBr, and HF) (Textor et al. 2004). More than 95% by volume of volcanic gases is comprised of H₂O, CO₂, and SO₂ (Symonds et al. 1994). The abundance of these gases differs considerably from volcanoes to volcanoes. The volcanic emissions reveal the composition of their source magma. The characteristics of volcanic emissions depend on the geological setting of the region where it is located; e.g., eruptions near subduction zones are associated with more water vapor and chlorine (Sigurdsson 2000).

Volcanic emissions include violent eruptions and noneruptive degassing. The degassing process occurs continuously during eruptions and through vents and fumaroles. The degassing determines the type of eruption that will take place. Volcanoes are responsible for about 14% to the global sulfur emissions (Graf et al. 1997). It is a no-brainer that gases emitted from subduction zone volcanoes should be commonly very rich in water vapor (95%). However, in hotspot regions the compositions are different and here volcanic gases are much more CO_2 -rich and H_2O -poor (Giggenbach 1996; Wallace 2005). These gas emissions occur not only from the craters, but also from faults, vents, or even through soils or even dissolve in groundwater and springwater.

Although the composition of magmas can be widely variable, in general they are constituted mainly from only eight elements: oxygen, aluminum, silicon, calcium, iron, magnesium, sodium, and potassium. Among these eight elements, oxygen is the most abundant followed by silicon. The magma is formed by melting of rocks, and the magmatic composition depends on the rock it was formed from and the conditions of that melting. Melting induces generation of magma as temperatures rise, pressure is released, and influx of volatiles occurs (Peterfreund 1981).

Most of the magma erupted from volcanoes is of basaltic nature mostly occurring in deep ocean water at the mid-oceanic ridges and forms pillow lava basalt and in hotspot volcanism. When the eruption rate is large, the eruption reaches the atmosphere. The basaltic magma is poor in silicate and rich in magnesium and iron, hence characterized by low viscosity and low gas content. Basaltic magma contributes very less portion of sulfur emissions into the atmosphere and it reaches the stratosphere rarely.

Another kind of magma is called Felsic magma, which is rich in silicate and has high viscosity and is associated with explosive eruptions. Normally, felsic magmas tend to have higher levels of volatile compounds, which are emitted as gases during eruptions. The most abundant volatile in magma is H₂O, followed typically by CO₂, and then by SO₂. Compared to basaltic volcanoes, felsic magma erupts less frequently. Sometimes, they release large amounts of ashes and gases directly into the atmosphere.

10.3.1 Volcanic Material

A volcanic eruption can spew material in form of gases, lava, and tephra as mentioned below:

- (a) Volcanic gases: In an early earth, gases from volcanic eruption gave us not only our atmosphere but also our oceans without which there would be no life on planet earth. The most common gases from volcanoes are water vapor, carbon dioxide, and sulfur dioxide. Other gases that are also emitted during volcanic eruptions are carbon monoxide, H₂S, HCl, CH₄, HF, etc. The abundance of these gases differs considerably from volcanoes to volcanoes. Due to the addition of seawater in the magmas at convergent plate boundaries, volcanoes near subduction zones erupt with more water vapor and chlorine. Similarly, volcanoes at divergent plate boundaries and at hotspots produce significantly less water vapor and chlorine (Sigurdsson 2000).
- (b) Lava: Lava is nothing, but magma erupted to the surface due to volcanic eruption. The temperature of lava usually ranges from 800 to 1200 °C. As we know, the eruption can be either effusive or explosive. Effusive eruption causes lava flow by outpouring of lava. Explosive eruptions on the other hand produce volcanic ashes and tephra. There are two basic types of lava flow 'A'ā and pahoehoe. While 'A'ā flow is rough and rubble-like and composed of broken cinder blokes, the pahoehoe type is smooth and ropy. The surface is formed due to very fluid lava under a solidified crust.
- (c) **Tephra:** Explosive volcanism sometimes ejects materials like rock fragments into the atmosphere. These rock fragments are called tephra regardless of composition or size (Amaral and Rodrigues 2019). Tephra represents airborne pyroclastic material that is ejected during the course of an eruption (Durand and Thorarinsson 1981). Tephra is classified into three size groups: (1) ash, when particles are minuscule to the tune of less than 2 mm in diameter; (2) moderate size particles are called lapilli or volcanic cinders, with diameter between 2 and 64 mm; and (3) large size ejected materials with diameter greater than 64 mm constitute volcanic bombs or blocks. Bombs and blocks are shot from the volcano ballistically. Because these fragments are large, they generally fall near the source. Most particles greater than 1 mm will fall out within hours of eruption. Being smaller in diameter, ash can travel thousands of kilometers from its originating volcano. The wind speed and direction, the air temperature, and the height of the eruption column are the variables that decide the distance that tephra will be transported away from the volcano. Larger particles, lapilli, and volcanic bombs fallout within an hour they are ejected from the volcano. The smallest particles can remain even for a couple of years after the eruption in the atmosphere. Sometimes, they produce amazing sunsets like seen after the 1991 Pinatubo eruption.

10.4 Impacts of Volcanic Emissions

10.4.1 Radiative Forcing

Radiative forcing is a concept to understand change in energy fluxes in the earth atmosphere system. The particles and the gases present in the atmosphere change the balance between incoming solar radiation and outgoing IR radiation (Pulselli and Marchi 2015). The term forcing explains the earth's radiative balance being pushed away from the normal state. The particles present in the atmosphere absorb or scatter solar and terrestrial radiation. Positive forcing is exerted when particles absorb the radiation and it contributes to warming of the earth's surface, whereas negative forcing is exerted when the particles present in the atmosphere scatter the incoming solar radiation and it causes the cooling of the earth's surface (Fig. 10.3).

Volcanism induces short-term climatic variations. The secondary sulfate aerosols (SO_4^{2-}) emitted from volcanoes can change the energy balance of that area (Franklin 1784; Charlson et al. 1991, 1992; Stenchikov et al. 1998). Sulfate aerosols scatter the incoming solar radiation due to their high single scattering albedo. In troposphere, SO₂ gets converted to H₂SO₄ and it increases the aerosol optical depth and the albedo. It causes cooling at the surface (Robock 2000). In the biologically important UV-B region, the average cross section for SO₂ is double of ozone (O₃); thus, for regions of high SO₂, if the changes in UV-B need to be predicted, we must consider the changes in both SO₂ and O₃ columns. Shindell et al. (2003) simulated the volcanic forcing and found that the mean annual cooling for the periodic Pinatubo eruption, Tambora 2P eruption, and Tambora 3P eruption and for the observed 1959–1999 volcanoes were -0.35 °C, -0.77 °C, -1.99 °C, and -0.44 °C, respectively, and the average instantaneous forcing was -0.47, -0.91, -1.39, and -0.44 W/m², respectively.

Crop failure and famine have been found to be associated with large eruptions as an aftermath as a result of the cooling produced, the ashes, and gases emitted. Large historic eruptions are aptly documented in archives such as that of 1815 Tambora eruption, which was associated with large-scale global cooling, which had eventually led not only to famine, but also social unrest, and as if this was not enough, this was followed by epidemic typhus, leading to as described beautifully in Oppenheimer (2003) was what came to be known as the "Year Without a Summer". The Laki eruption of 1783 caused several years of crop failure and cold winters leading to death of ~20% of the Iceland population (Grattan et al. 2003; Thordarson and Self 2003).

Without mention of the little ice age, the discussion would remain incomplete. Regional cooling was very intense during this period roughly between 1300 and 1850 AD with temperature in Europe and North America plummeting by as much as 2 °C. This intense drop in average temperatures can be attributed to several large sulfur-rich explosive eruptions, and may be possibly triggered or enhanced by the massive Samalas volcanic eruption in 1257 (Amos 2013). Geological records suggest that throughout the little ice age there was heightened volcanic activity (Robock 1979). Historical evidence suggests that many rivers and lake in Europe





froze over. Some records even suggest the Baltic Sea freezing (Meacham 2020). The period was marked with advancing glaciers in Scandinavia, with tree line and snowline dropping. Crops failed, and there were widespread droughts across Europe due to the prolonged cold and dry climate. Unemployment and malnutrition and poor living standards resulted in many vector diseases and epidemics (Post 1984). As a result, there was increase in crime rate and caused massive social unrests. The whole period was marked with unpleasant weather, crop failure, large-scale unemployment, increase in intergroup violence, economic hardship, and high mortality rate due to malnutrition and several epidemics. It did not only cause socioeconomic problems but also cause political unrest throughout Europe. Historians also suggest this period of high social unrest may have led to the 30 years of war (1617–1648) (Zhang et al. 2011).

10.4.2 Impact on Ozone

The ozone layer protects the living beings on the planet surface by absorbing 93-99% of the sun's high-frequency ultraviolet light. Oxidation of molecular oxygen leads to O₃ formation, but the mechanisms exist in the troposphere and stratosphere. Stratospheric O₃ is considered beneficial because it prevents the harmful UV from reaching the earth's surface, while tropospheric O₃ has several ramifications to air quality, atmospheric chemistry, crop yield, environment, and human health (Saxena et al. 2019; Saxena and Sonwani 2020). Though most of these ramifications are a problem for the humanity, O₃ being an oxidant may also help to remove pollutants from the atmosphere (Mallik et al. 2018).

Up in the troposphere, radiation is intense enough to destroy O₃ either by itself or through a chain of free radical reactions involving the nitric oxide radical (NO), atomic chlorine (Cl), and atomic bromine (Br). Bromine released during volcanic eruption has the potential to damage the ozone even more compared to chlorine, Br being 40 times more effective than Cl w.r.t. O₃ depletion (Daniel et al. 1999). All of the mentioned radicals can have both natural and anthropogenic origins. Of particular concern is the dramatic increase in the levels of chlorine and bromine attributed to anthropogenic activities, mostly traced to halocarbons, e.g., chlorofluorocarbons (CFCs) (Mohanakumar 2008). The CFCs are stable organic compounds and find their way to the upper atmosphere due to their substantial lifetime. As CFCs reach the stratosphere, UV radiation splits them releasing the Cl and Br atoms from the parent, which were acting as effective reservoirs for these radicals. The below equation shows this mechanism. Here, "h" is Planck's constant and " ν " is frequency of electromagnetic radiation.

$$CFCl_3 + h\nu \to CFCl_2 + Cl^2 \tag{10.1}$$

The free halide easily decomposes O_3 into its the more stable form of O_2 . The halogen oxide can react with another ozone molecule and will give another halogen radical and two oxygen molecules.

$$\mathrm{Cl}^{\circ} + \mathrm{O}_3 \to \mathrm{ClO} + \mathrm{O}_2 \tag{10.2}$$

$$\mathrm{ClO} + \mathrm{O}_3 \to \mathrm{Cl}^\circ + 2\mathrm{O}_2 \tag{10.3}$$

Similar reactions can occur for most halides. As the concern of depletion of O_3 in the stratosphere grew, the scientists started examining the role of volcanoes in ozone depletion. It was observed that the gases emitted by the volcanoes during eruption can quickly reach the upper troposphere and lower stratosphere (UTLS); hence, their lifetime increases as they are saved from the high concentration of oxidants in the lower troposphere/boundary layer. The hydrogen halides particularly HCl emitted during eruption, when it reaches the stratosphere, they lead to destruction of ozone (Solomon et al. 1996; Hofmann and Solomon 1989). In addition, aerosols produced by volcanoes can increase the effectiveness of halides in depletion of O_3 in the stratosphere. While most sulfur eruptions reach the stratosphere, halogens are susceptible to being washed away by water and ice in the rapidly ascending column. The increased lifetime of particles and gases from volcanic emissions in the UTLS region makes them more impactful in their radiative and chemical effects.

Zuev et al. (2015) studied the O_3 depletion caused by Erebus volcanic eruption during which the emitted HCl and SO₂ reached the UTLS (upper troposphere and lower stratosphere) with the aid of high latitude cyclones and polar vortex. The polar stratospheric clouds also significantly assist O_3 depletion by converting chlorine compounds into more destructive forms.

Ozonesonde experiments traced a significant decrease in the total ozone column in lower stratosphere after the eruption of Mt. Pinatubo (Grant et al. 1994; Schoeberl et al. 1993; Tang et al. 2013). The decrease in tropospheric and stratospheric ozone was also noted by Fusco and Logan (2003) and Oltmans et al. (1998). Tang et al. (2013) found that the maximum decrease was 2 ppb on interannual time scale, whereas 5 ppb decrease was observed in monthly averages. Robock (2000) estimated the column depletion, and it ranged from 2% in the tropics to about 7% in the midlatitudes.

10.4.3 Acid Rain

The term acid rain entered the scientific domain in 1872 when Robert Angus Smith used it for the first time to describe the acidic nature of rain around the industrial town of Manchester, UK. Bouwman et al. (2002) identified atmospheric acid deposition in the form of rain, fog, or snow to be a major problem for the environment.

 SO_2 and oxides of nitrogen are the primary cause of acid rain (Singh and Agrawal 2008). Acid rain can occur in the form of H_2SO_4 and HNO_3 , so the emission sources for SO_2 and NO_2 including motor vehicles, power plants, refineries, and industries are precursors for acid rain. These precursor gases are converted into sulfuric acid and nitric acid in wet atmospheric conditions.

Acid rain adversely affects biogeochemical cycles and soil quality, and it also harms flora and fauna. It also causes degradation of buildings and yellowing and weakening of fabrics. Human health is affected by acid rain as it can cause itching, skin burn, respiratory problems, headache, etc.

The sulfur dioxide gas emitted from volcanoes when reacts with oxygen and moisture present in the atmosphere produces acid rain and can also lead to volcanic smog (vog). VOG can not only corrode metals and impact plants but also lead to respiratory and cardiovascular problems in humans. The main vent of Kilauea Volcano on the Hawaii islands emits 63,000-95,000 metric tons of SO₂ every year due to ordinary fuming leading to acid rain, but discrepancies exist if SO₂ is the sole cause of reduced pH (Siegel and Siegel 1984; Gerlach and Graeber 1985; Casadevall et al. 1987; Siegel et al. 1987; Nachbar-Hapai et al. 1989).

10.4.4 Impact on Aviation

Since the beginning of the aviation age in 1960s, with the advancement in technology there have been rapid advances in the sector of air travel. Correspondingly, there has been an increase in probability of an aircraft encountering the volcanic clouds. The ash particles ejected by volcano in the atmosphere impact the fuselage, windscreens, and compressor fan blades and can essentially chafe these forwardfacing surfaces. The navigational and operational instruments might fail critically when contaminated with the ash. Ash particles consist of the glassy silicate material, which can melt at combustion temperatures inside modern jet engines (Chen and Zhao 2015). The melted ash particles rack up as re-solidified deposits in cooler parts of the aircraft leading to deterioration of engine performance even to the point of in-flight compressor stall and loss of thrust power.

Volcanic ash poses a major threat to the aviation sector with immediate and longterm consequences (Casadevall 1993). Over 129 incidents have been reported of aircrafts encountering volcanic ash from 1953 to 2009 (Guffanti et al. 2010). Of all these reported incidents, 79 are related to inflicting some physical damage to the aircraft, and 26 led to severe ramifications among which nine led to in-flight power loss in one or more engines. Miller and Casadevall (2000) documented incident of "all engines out" encountered by a Boeing B747 in 1982 with volcanic ash from Galunggung volcano over Indonesia and Boeing B747 in 1989 from Redoubt volcano over Alaska.

10.4.5 Environment and Health

Of an approximately 500 active volcanoes, 10–40 eruptions occur on an annual basis (Zuskin et al. 2007). Volcanic eruptions not only affect climate, but also affect terrestrial ecosystem and human and animal health (Mather et al. 2003). Volcanic eruptions can pose a threat on the environment even from a distance of thousand kilometers from the eruptive center (Galas 2016). Due to rapid accumulation of the

ashes and the high density of individual volcanic ash particles during ash fallout, the buildings may collapse (Spence et al. 2005). The volcanic ash fall can also contaminate the water supplies (Stewart et al. 2006).

 SO_2 emitted during volcanic eruption can change the precipitation rate (Twomey 1974; Nober et al. 2002). SO_2 acts as cloud condensation nuclei, and it increases the number of cloud droplets resulting in decrease in rain. It increases the albedo of the cloud, which causes cooling at the surface. As discussed above, the sulfate aerosols emitted during a volcanic eruption can react with the oxygen in the troposphere and cause acid rain.

The sulfate aerosols are generally injected in the upper troposphere above the boundary layer. At this height, mixing is weaker and the removal process is slower and the sulfate aerosols remain there for considerable time compared to the anthropogenic sulfur (Graf et al. 1998). Compared to the heating effect of each CO_2 molecule in the atmosphere, a molecule of SO_2 can overcompensate in cooling by at least a factor of 50 (Kaufman et al. 1991). The volcanic aerosols in the stratosphere can lead to a rise in stratospheric temperatures. It also increases the pole to equator temperature gradient (Quiroz 1984; Parker and Brownscombe 1983).

The impact on health can be traced to the type and style of volcanic eruptions, e.g., the effusive emissions, which mainly emit gases, and aerosols may cause disorders of the respiratory system (Horwell et al. 2015; Saxena and Srivastava 2020). We know volcanoes emit hazardous gases. The SO₂ emitted by the volcanoes can be transported globally and potentially trigger acute respiratory diseases like asthma in the exposed people (Hansell and Oppenheimer 2004; Horwell and Baxter 2006).

10.4.6 Volcano and ENSO Relation

El Nino and the Southern Oscillation, also known as ENSO, are a large-scale oceanic warming and the variation in the air pressure of the overlying atmosphere in the tropical Pacific Ocean that occurs every few years. Though ENSO originates and develops in equatorial Pacific region, it affects the climate, ecosystem, and economies around the world (Fang and Xie 2020; Ropelewski and Halpert 1987). ENSO is receptive to external forcing on different time scales, and solar forcing is the cause for millennial ENSO variability due to orbitally induced changes in insolation (Predybaylo et al. 2020; Moy et al. 2002). According to Timmreck (2012), strong volcanic eruptions can rattle ENSO on a short time scale (2–5 years).

As discussed in the previous section, the presence of sulfate aerosols in the atmosphere causes cooling at the surface that can persist for years in the stratosphere with potential to impact the ocean temperature and circulation (Church et al. 2005; Gleckler et al. 2006). Handler and Andsager (1993) proposed that the reduction in incoming shortwave radiation due to the emissions from volcanoes has increased the number of ENSO in the last 150 years. Also, the strong tropical volcanic eruptions may change the atmospheric and oceanic circulation (Mann et al. 2005; McGregor et al. 2010; Shiogama et al. 2010; Zanchettin et al. 2012). At the end of the twentieth

century, the relation between ENSO and strong tropical volcanic eruptions is an active field of research. The three largest eruptions in last 50 years (Agung in 1963, El Chichon in 1982 and Pinatubo in 1991) took place in conjunction with the warm phase of ENSO; therefore, there are possibilities that the strong tropical volcanic eruptions do have a role to play in ENSO system (Ohba et al. 2013). Adams et al. (2003) also found that the ENSO system may become more active due to volcanic eruptions. In contrast, Self et al. (1997) found that the volcano–ENSO relationship does not exhibit a statistically significant value over the last 150 years. Further millennium-long climate and volcanic records analysis points to the fact that even most explosive eruptions are too small to affect ENSO (Emile-Geay et al. 2008).

10.5 Impacts and Extremities

Depending on the type and region of eruption, volcanic events can range from mild to severe resulting in mild to extreme impacts on the different spheres of the Earth. The impacts of volcanoes are already discussed in the above paragraphs. However, depending on the circumstances, the net result is different. The tropospheric aerosols resulting from a quiescent volcano will have a lifetime of weeks, but an explosive eruption injecting the aerosols in the stratosphere can scale up the lifetime from weeks to years. The resulting impact on radiative forcing will hence be much more significant. The extremity of an eruption impact is not only visible in the amount of material that is ejected but also by the distance the impact propagates. Even smaller eruptions can have a long-range impact under favorable meteorological conditions. The Dalaffila eruption of November 2008 in the Afar region of Ethiopia, despite generating only about 0.1–0.2 Tg SO₂ in the eruption cloud, was one of the few recent volcanoes whose effect was observed as far as India (Mallik et al. 2013). SO₂ plume from this small eruption in Ethiopia traversed over 5000 km under favorable winds causing 10–20 times enhancement in background SO₂ columnar levels over the Indo-Gangetic plains of India and injected 1.4 Tg SO₂ over the Delhi region, equivalent to 25% of annual anthropogenic emissions over Delhi (Mallik et al. 2013). Recently in 2018, Manaro Voui Volcano on Ambae Island of South Pacific injected a record amount of 400,000 tons of SO₂ in the UTLS region, which was thrice the amount of SO₂ released from all volcanoes in the previous year (NASA report 2018). The impacts ranged from destruction of homes, burying of crops, blackening of the sky by volcanic ash, acidifying the drinking water, and forcing the entire population of the island to evacuate. No discussion of severity of volcanic impacts is complete without describing the Mount Pinatubo eruption.

10.5.1 Mount Pinatubo: A Case Study

The Mount Pinatubo eruption on June 15, 1991, was a landmark event not only for the Philippines but also for the entire planet as this was the world's largest volcanic eruption to happen in the past 100 years. Mount Pinatubo is an active volcano amid a

chain of volcanoes located on the western fringe of the island of Luzon. They are subduction volcanoes, formed by the Eurasian Plate drifting below the Philippine Plate along the Manila Trench to the west. Smith (1909) provided the first geologic narration about Mount Pinatubo, describing it as destitute of "volcanic ash (and) any of the usual indications of volcanic activity." He surmised that "Mount Pinatubo is not a volcano and we saw no signs of it ever having been one, although the rock constituting it is porphyritic.... The region is quite unique and I have seen nothing in the Philippines quite like it."

Ash production was documented for Mount Pinatubo eruption for the first time at 0851 local time on June 12, 1991, where a series of explosive eruptions with an eruption cloud crossing 19 altitudes (Koyaguchi and Tokuno 1993). Satellite image analysis indicated that the high-level winds (\sim 15–20 m/s) carried the ash clouds from the eruption of June 12 into the airspace west of Manila along the heading of 215° from the volcano (Potts 1993). Pinatubo Volcano Observatory Team also reported the intermittent lapilli fall and fine ash fall mainly in the region of the southwest of Pinatubo. Pinatubo exploded most violently releasing the largest ash cloud on June 15. (Koyaguchi and Tokuno 1993). Because of the passage of Typhoon Yunya, the pattern of dispersion of these ash clouds was complicated (Oswalt et al. 1991). The ash would have moved to west-southwest under normal weather conditions, but the Typhoon Yunya forced the ash clouds to the south (Paladio-Melosantos et al. 1991). The eruption of Mount Pinatubo caused the evacuation of more than 0.2 million people with more than 300 deaths mainly due to heavy ashfall and rain leading to collapsing of buildings. The amount of 6000–10,000 Mt of dacite magma in a Plinian eruption was ejected during the 9-h eruption of Mount Pinatubo volcano on June 15, 1991 (Pallister et al. 1996).

Aerosols from the eruption of Mount Pinatubo caused solar dimming in turn led to a lower troposphere cooling across the globe. This cooling was so strong that it led to a decrease in water vapor concentrations globally. The eruption cloud spanned 1100 km wide and 35 km high ejecting volcanic gases and ash leading to 17 megatons of SO₂ injected into the stratosphere, as documented in "Fire and Mud" by Self et al. (1997). The cooling continued over 2 years up to 1993, leading to 0.4 °C temperature drop over several regions. Bluth et al. (1992) and Hansen et al. (1992, 1996) also reported that around 20 MT of SO₂ was ejected in that eruption. SO₂ from Pinatubo was detected in the stratosphere as late as June 1994, 3 years after the eruption. Parker et al. (1996) observed a global cooling of about 0.5 °C at the surface and 0.6 °C in the troposphere between the eruption of Mount Pinatubo and the northern summer of 1992, interrupted, however, by relative warmth between January and March 1992.

Research on tree rings shows that after the eruptions of large volcanoes including Pinatubo and Tambora, Mongolia, and most parts of southern China continued to receive less rainfall, while an increase in rainfall was observed over mainland Southeast Asia in the reduced precipitation due to Pinatubo led to a record decrease in runoff and river discharge into the ocean from October 1991–September 1992. In April 1993, that is around 2 years after eruption, the ozone column showed a deficit of around 6% compared to the annual mean from 1979 to 1990 (Gleason et al. 1993; Herman and Larko 1994). The cloud spread covered most of the area of the equatorial band from about 10° S to 20° N on June 15.

The eruption had great effect on the aviation sector contaminating some of the world's busiest air traffic corridors. Eruptions between June 12 and 15, 1991, led to as many as 16 damaging encounters between drifting ash clouds from the eruption and jet aircrafts. Out of these 16, three encounters occurred within a 200 km of eruption site with fresh ash clouds less than 3 h old. Further away between 720 and 1740 km west from the volcano, 12 encounters occurred over Southeast Asian region with 0.5- to 1-day-old ash clouds. A total of ten engines were damaged by the ash cloud, with additional two instances of in-flight loss of power. Long-term damage to aircrafts and engines attributed to Pinatubo SO₂ led to resulted in crazing of acrylic airplane windows, premature fading of polyurethane paint on jetliners, and accumulation of sulfate deposits in engines (Pallister et al. 1996).

10.6 Summary

Volcanism is an endogenic process that has the potential to change the earth's surface fairly quickly in contrast to exogenic processes through surface forces whose effects become perceptible in millions of years. A volcano is a rupture on the earth's crust through which hot lava, volcanic ash, and gases are spewed out during the course of the eruption from a magma chamber underneath the surface. Most of the volcanoes on earth are found near the tectonic plate boundaries, where they converge and subduct or diverge. Active volcanoes are those that have erupted in the Holocene, i.e., within the last 10,000 years and mostly concentrated along the Pacific Ring of Fire. Also of concern are dormant volcanoes, which are likely to erupt in near future. Identifying volcanoes on the basis of their structure gives a better insight into the type and intensity of eruption. Plate Tectonics, which is like the grand unified theory of Earth Science, is generally invoked to explain the causes behind volcanic eruptions. Depending on the volume of products and height of eruption, the explosiveness of a volcano is estimated on the basis of VEI scale of 1-8 representing the smallest, nonexplosive to mega-colossal explosions. On the basis of the nature of volcanic eruptions, volcanoes can be classified as Plinian, Vulcanian, and Strombolian. Volcanic emissions include violent eruptions and noneruptive degassing, which occurs continuously during eruptions and through vents and fumaroles. The degassing determines the type of eruption that will take place. The characteristics of volcanic emissions depend on the geological setting of the region where it is located; e.g., eruptions near subduction zones are associated with more water vapor and chlorine.

Eruptions spew out gases, lava, and tephra. Most of the magma erupted from volcanoes is of basaltic nature. The solid tephra includes ash, lapilli, and bombs/ blocks. Most particles greater than 1 mm will fall out within hours of eruption. Ash can travel thousands of kilometers from its originating volcano due to its tiny size of less than 2 mm. The most common gases from volcanoes are water vapor, carbon dioxide, and sulfur dioxide but additionally include carbon monoxide, hydrogen

sulfide, chlorides, and fluorides of hydrogen and methane. Most often, volcanism induces short-term climatic variations. The secondary sulfate aerosols emitted from volcanoes can change the regional energy balance and lead to cooling at the surface. Volcanic cooling can cause crop failure and famine for many years after large eruptions as was observed for Tambora eruption of 1815. The halogen-containing gases from a volcano impact the stratospheric ozone layer. Significant decrease in the total ozone column in lower stratosphere was observed by ozonesondes after the eruption of Mt. Pinatubo. Volcanic material can lead to increased pH of rainwater and can even contaminate drinking water sources of a region as was observed during the recent Manaro Voui eruption of 2018. The severest impacts of volcanoes are borne by the aviation sector leading to engine damage and aviation hazards. Apart from modulating climate due to perturbed radiative balance, volcanic eruptions have also been associated with ENSO, changed rainfall patterns, and modified circulation patterns. The Mt. Pinatubo eruption has acquired a special place in volcanology due to its gigantic scale and extreme impacts. However, on VEI scale, Tambora eruption of 1815 is higher than Pinatubo, while the highest is Yellowstone Caldera, which erupted 0.5 million years ago. But the Pinatubo has given us a lot more information due to the presence of modern technology and satellites.

The effects of volcanic eruptions are already incorporated in global climate models and earth system science models. While on one hand, specialist scientific agencies and vulcanologists are always on lookout for possible new eruptions, a synergistic approach additionally involving other branches of science where eruption effects are anticipated can be an important step to face the challenges posed by volcanic emissions. Scientists working with emission inventories can incorporate the effects of transcontinental and transoceanic plumes in regional inventories for relevant eruptions; e.g., SO₂ enhancement observed over Delhi was traced to an eruption in Afar region of Ethiopia (Mallik et al. 2013). Policymakers will also be benefitted from such inclusions, specifically in regions and localities prone to impacts of volcanic emissions. Such holistic approach will also lead to greater awareness and understanding among nonexpert stakeholders who knowing/unknowingly suffer from these extreme expressions of nature.

References

- Adams J, Mann ME, Ammann CM (2003) Proxy evidence for an El Nino-like response to volcanic forcing. Nature 426:274–278. https://doi.org/10.1038/nature02101
- Amaral AFS, Rodrigues AS (2019) Encyclopedia of environmental health, 2nd edn. ISBN 978-0-444-63952-3
- Amos J (2013) Mystery 13th century eruption traced to Lombok, Indonesia, BBC. https://www.bbc. com/news/science-environment-24332239
- Anatomy of a strato volcano such as Mt Agung in Bali (n.d.) (ABC: Julie Ramsden). abc.net.au: https://www.abc.net.au/news/science/2017-11-22/volcanoes-heres-what-happens-when-theyerupt/8997014?utm_source=abc_news_web&utm_medium =content_ shared &utm_ campaign=abc_news_web

- Auger E, Gasparini P, Virieux J, Zollo A (2001) Seismic evidence of an extended magmatic sill under Mt. Vesuvius Sci (New York, NY) 294:1510–1512. https://doi.org/10.1126/science. 1064893
- Bangar KM (2008) Principles of engineering geology, standard publishers distributers, ISBN 81-8014-115-2
- Bluth GJS, Doiron SD, Schnetzler CC, Krueger AJ, Walter LS (1992) Global tracking of the SO2 clouds from the June, 1991 Mount Pinatubo eruptions. Geophys Res Lett 19:151–154. https:// doi.org/10.1029/91GL02792
- Bouwman AF, Van Vuuren DP, Derwent RG (2002) A global analysis of acidification and eutrophication of terrestrial ecosystems. Water Air Soil Pollut 141:349–382. https://doi.org/ 10.1023/A:1021398008726
- Budd DA, Troll VR, Dahren B, Burchardt S (2016) Persistent multitiered magma plumbing beneath Katla volcano, Iceland. Geochem Geophys Geosyst 17:966–980. https://doi.org/10.1002/ 2015GC006118
- Capellas M (2007) Recovering Pompeii. Sci School 6:14
- Casadevall TJ (1993) Volcanic hazards and aviation safety: lessons of the past decade. Flight safety foundation, flight safety digest, May 1993
- Casadevall TJ, Stokes B, Greenland P, Malinconico L, Casadevail J, Furukawa B (1987) Chapter 29: SOz and CO2 emission rates at Kilauea Volcano 1979-1984. In: Decker R, Wright T, Stauffer P (eds) Volcanism in Hawaii, vol 1. US Geol Surv Prof Paper 1350 US Govt Print Off, Washington, DC, pp 771–780
- Charlson RJ, Langner J, Rodhe H, Leovy CB, Warren SG (1991) Perturbation of the Northern Hemisphere radiative balance by backscattering from anthropogenic sulfate aerosols. Tellus 43: 152–163. https://doi.org/10.3402/tellusa.v43i4.11944
- Charlson RJ, Schwartz SE, Hales JM, Cess RD, Coakley JA, Hansen JE, Hoffman DJ (1992) Climate forcing by anthropogenic aerosols. Science 255:423–430. https://doi.org/10.1126/ science.255.5043.423
- Chen WR, Zhao LR (2015) Review—volcanic ash and its influence on aircraft engine components. Proc Eng 99:795–803. https://doi.org/10.1016/j.proeng.2014.12.604. ISSN 1877-7058
- Church JA, White NJ, Arblaster JM (2005) Significant decadal-scale impact of volcanic eruptions on sea level and ocean heat content. Nature. https://doi.org/10.1038/nature04237
- Condie KC (1997) Plate tectonics and crustal evolution. Butterworth-Heinemann, p 123. ISBN 978-0-7506-3386-4
- Cottrell E (2015) Global distribution of active volcanoes. In: Volcanic hazards, risks and disasters. pp 1–16. https://doi.org/10.1016/b978-0-12-396453-3.00001-0
- Daniel JS, Solomon S, Portmann RW, Garcia RR (1999) Stratospheric ozone destruction: the importance of bromine relative to chlorine. J Geophys Res 104:23871–23880. https://doi.org/ 10.1029/1999JD900381
- Durand M, Thorarinsson S (1981) Greetings from Iceland: ash-falls and volcanic aerosols in Scandinavia. Geogr Ann 63A:109–118. https://doi.org/10.1177/0309133307073887. Progress in Physical Geography: Earth and Environment (2007) 31(1):89–93
- Emile-Geay J, Seager R, Cane MA, Cook ER, Haug GH (2008) Volcanoes and ENSO over the past millennium. J Clim 21(13):3134–3148. https://doi.org/10.1175/2007JCL11884.1
- Fang X, Xie R (2020) A brief review of ENSO theories and prediction. Sci China Earth Sci 63:476– 491. https://doi.org/10.1007/s11430-019-9539-0
- Forsyth DW (1975) The early structural evolution and anisotropy of the oceanic upper mantle. Geophys J Int 43(1):103–162. https://doi.org/10.1111/j.1365-246X.1975.tb00630.x
- Franklin B (1784) Meteorological imaginations and conjectures, Manchester Literary and Philosophical Society Memoirs and Proceedings 2, 122, 1784, reprinted in Weatherwise 35, 262, 1982
- Fusco AC, Logan JA (2003) Analysis of 1970–1995 trends in tropospheric ozone at Northern Hemisphere midlatitudes with the GEOS-CHEM model. J Geophys Res 108(D15):4449. https:// doi.org/10.1029/2002JD002742

- Galas A (2016) Impact of volcanic eruptions on the environment and climatic conditions in the area of Poland (Central Europe). Earth Sci Rev 162:58–64. https://doi.org/10.1016/j.earscirev.2016. 09.014. ISSN 0012-8252
- Garrison TS, Ellis R (2016) Essentials of oceanography. Cengage Learning, p 19. ISBN 978-1-337-51538-2
- Gerlach T, Graeber E (1985) Volatile budget of Kilauea Volcano. Nature 313:273–277. https://doi. org/10.1038/313273a0
- Giggenbach WF (1996) Chemical composition of volcanic gases. In: Monitoring and mitigation of volcano hazards. Springer, Berlin, Heidelberg. https://doi.org/10.1007/978-3-642-80087-0_7
- Gleason J, Bhartia P, Herman J, McPeters R, Newman P, Stolarski R, Planet W (1993) Record low global ozone in 1992. Science 260(5107):523–526. https://doi.org/10.1126/science.260. 5107.523
- Gleckler PJ, AchutaRao K, Gregory JM, Santer BD, Taylor KE, Wigley TML (2006) Krakatoa lives: the effect of volcanic eruptions on ocean heat content and thermal expansion. Geophys Res Lett 33:L17702. https://doi.org/10.1029/2006GL026771
- Global Volcanism Program (2013) Volcanoes of the World, v. 4.9.4 (17 Mar 2021). In: Venzke E (ed) Smithsonian Institution. Downloaded 22 Apr 2021. https://doi.org/10.5479/si.GVP. VOTW4-2013
- Graf H-F, Feichter J, Langmann B (1997) Volcanic sulfur emissions: estimates of source strength and its contribution to the global sulfate distribution. J Geophys Res 102(D9):10727–10738. https://doi.org/10.1029/96JD03265
- Graf H-F, Langmann B, Feichter J (1998) The contribution of Earth degassing to the atmospheric sulfur budget. Chem Geol 147:131–145. https://doi.org/10.1016/S0009-2541(97)00177-0
- Grant WB, Browell EV, Fishman J, Brackett VG, Veiga RE, Nganga D, Minga A, Cros B, Butler CF, Fenn MA, Long CS, Stowe LL (1994) Aerosol-associated changes in tropical stratospheric ozone following the eruption of Mount Pinatubo. J Geophys Res 99(D4):8197–8211. https:// doi.org/10.1029/93JD03314
- Grattan J, Durand M, Taylor S (2003) Illness and elevated human mortality in Europe coincident with the Laki Fissure eruption. In: Volcanic Degassing: Geological Society, Special Publication 213. Geological Society of London, pp 410–414. http://hdl.handle.net/2160/230
- Guffanti M, Casadevall T, Budding K (2010) Encounters of aircraft with volcanic ash clouds: a compilation of known incidents, 1953–2009. USGS data series, p 545. https://pubs.usgs.gov/ds/ 545/
- Handler P, Andsager K (1993) Impact of volcanic aerosols on global climate. Trends Geophys Res 2:581–593
- Hansell AL, Oppenheimer C (2004) Health hazards from volcanic gases: a systematic literature review. Arch Environ Health 59(12):628–639. https://doi.org/10.1080/00039890409602947
- Hansen J, Lacis A, Ruedy R, Sato M (1992) Potential climate impact of Mount Pinatubo eruption. Geophys Res Lett 19:215–218. https://doi.org/10.1029/91GL02788
- Hansen J, Ruedy R, Sato M, Reynolds R (1996) Global surface temperature in 1995: return to pre-Pinatubo level. Geophys Res Lett 23 (pg:1665–1668. https://doi.org/10.1029/96GL01040
- Herman JR, Larko D (1994) Low ozone amounts during 1992–1993 from Nimbus 7 and Meteor 3 total ozone mapping spectrometers. J Geophys Res 99(D2):3483–3496. https://doi.org/10. 1029/93JD02594
- Hofmann DJ, Solomon S (1989) Ozone destruction through heterogeneous chemistry following the eruption of El Chichón. J Geophys Res 94:5029–5041. https://doi.org/10.1029/ JD094iD04p05029
- Horwell CJ, Baxter PJ (2006) The respiratory health hazards of volcanic ash: a review for volcanic risk mitigation. Bull Volcanol 69:1–24. https://doi.org/10.1007/s00445-006-0052-y
- Horwell C, Baxter P, Kamanyire R (2015) Health impacts of volcanic eruptions. In: Loughlin S, Sparks S, Brown S, Jenkins S, Vye-Brown C (eds) Global volcanic hazards and risk. Cambridge University Press, Cambridge, pp 289–294. https://doi.org/10.1017/CBO9781316276273.015

- Ilyinskaya E, Schmidt A, Mather TA, Pope FD, Witham C, Baxter P, Jóhannsson T, Pfeffer M, Barsotti S, Singh A, Sanderson P, Bergsson B, McCormick Kilbride B, Donovan A, Peters N, Oppenheimer C, Edmonds M (2017) Understanding the environmental impacts of large fissure eruptions: aerosol and gas emissions from the 2014–2015 Holuhraun eruption (Iceland). Earth Planet Sci Lett 472:309–322. https://doi.org/10.1016/j.epsl.2017.05.025
- Kaufman YJ, Fraser RS, Mahomey RL (1991) Fossil fuel and biomass burning effect on climate heating or cooling? J Clim 4:578–588
- Keller G (2014) Deccan volcanism, the Chicxulub impact, and the end-Cretaceous mass extinction: coincidence? Cause and effect? Geol Soc Am Spec Pap 505:57–89. https://doi.org/10.1130/ 2014.2505(03)
- Koyaguchi T, Tokuno M (1993) Origin of the giant eruption cloud of Pinatubo, June 15, 1991. J Volcanol Geotherm Res 55(1–2):85–96. https://doi.org/10.1016/0377-0273(93)90091-5
- Mallik C, Lal S, Naja M, Chand D, Venkataramani S, Joshi H, Pant P (2013) Enhanced SO2 concentrations observed over Northern India: role of long-range transport. Int J Remote Sens 34(8):2749–2762. https://doi.org/10.1080/01431161.2012.750773
- Mallik C, Tomsche L, Bourtsoukidis E, Crowley JN, Derstroff B, Fischer H, Hafermann S, Huser I, Javed U, Keßel S, Lelieveld J, Martinez M, Meusel H, Novelli A, Phillips GJ, Pozzer A, Reiffs A, Sander R, Taraborrelli D, Sauvage C, Schuladen J, Su H, Williams J, Harder H (2018) Oxidation processes in the eastern mediterranean atmosphere: evidences from the modelling of HOx measurements over Cyprus. Atmos Chem Phys 18:10825–10847. https://doi.org/10.5194/ acp-18-10825-2018
- Mann ME, Cane MA, Zebiak SE, Clement A (2005) Volcanic and solar forcing of the tropical Pacific over the past 1000 years. J Clim 18(3):447–456. https://doi.org/10.1175/JCLI-3276.1
- Mather T, Pyle DM, Oppenheimer C (2003) Tropospheric volcanic aerosol. In: Robock A, Oppenheimer C (eds) Volcanism and the Earth's atmosphere. American Geophysical Union, pp 189–212. https://doi.org/10.1029/139GM12
- McGregor S, Timmermann A, Timm O (2010) A unified proxy for ENSO and PDO variability since 1650. Clim Past 6:1–17. https://doi.org/10.5194/cp-6-1-2010
- Meacham J (2020, May 7) Pandemics of the past. The New York Times. ISSN 0362-4331. Accessed 8 May 2020
- Miller TP, Casadevall TJ (2000) Volcanic ash hazards to aviation. In: Sigurdsson H (ed) Encyclopedia of volcanoes. Academic, San Diego, pp 915–930
- Mohanakumar K (2008) Stratosphere troposphere interactions. An introduction. Springer. Stratospheric ozone depletion and Antarctic ozone hole. In: Stratosphere troposphere interactions. Springer, Dordrecht. https://doi.org/10.1007/978-1-4020-8217-7_6
- Moy CM, Seltzer GO, Rodbell DT, Anderson DM (2002) Variability of el niño/southern oscillation activity at millennial timescales during the holocene epoch. Nature 420:162–165. https://doi.org/10.1038/nature01194
- Nachbar-Hapai M, Siegel BZ, Russell C (1989) Acid rain in the Kilauea volcano area (Hawaii). Arch Environ Contam Toxicol 18:65–73. https://doi.org/10.1007/BF01056191
- Nakada S (2000) Hazards from pyroclastic flows and surges. In: Sigurdsson H, Houghton B, McNutt SR, Rymer H, Stix J (eds) Encyclopedia of volcanoes. Academic, New York, pp 945–955
- Nania J, Bruya TE (1982) In the wake of Mount St Helens. Ann Emerg Med 11(4):184–191. https:// doi.org/10.1016/s0196-0644(82)80495-2
- NASA Report (2018). https://www.nasa.gov/feature/goddard/2019/2018-s-biggest-volcanic-erup tion-of-sulfur-dioxide. Accessed 8 May 2021
- Newhall CG, Self S (1982) The volcanic explosivity index (VEI): an estimate of explosive magnitude for historical volcanism. J Geophys Res 87(C2):1231–1238. https://doi.org/10. 1029/jc087ic02p01231
- Nober F, Graf H-F, Rosenfeld D (2002) Sensitivity of the global circulation to the suppression of precipitation by anthropogenic aerosols, in print Global Planet. Change. pp 57–80. ISSN 0921-8181. https://doi.org/10.1016/S0921-8181(02)00191-1

- Ohba M, Shiogama H, Yokohata T, Watanabe M (2013) Impact of strong tropical volcanic eruptions on ENSO simulated in a coupled GCM. J Clim 26(14):5169–5182. https://doi.org/ 10.1175/JCLI-D-12-00471.1
- Oltmans SJ, Lefohn AS, Scheel HE, Harris JM, Levy H, Galbally IE, Brunke E-G, Meyer CP, Lathrop JA, Johnson BJ, Shadwick DS, Cuevas E, Schmidlin FJ, Tarasick DW, Claude H, Kerr JB, Uchino O, Mohnen V (1998) Trends of ozone in the troposphere. Geophys Res Lett 25(2): 139–142. https://doi.org/10.1029/97GL03505
- Oppenheimer C (2003) Climatic, environmental and human consequences of the largest known historic eruption: Tambora volcano (Indonesia) 1815. Prog Phys Geogr Earth Environ 27(2): 230–259. https://doi.org/10.1191/0309133303pp379ra
- Oswalt JS, Nichols W, O'Hara JF (1991) Fire and mud, USGS Publication Repository, Meteorological observations of the 1991 Mount Pinatubo eruption
- Paladio-Melosantos ML, Solidum RU, Scott WE, Quiambao RB, Umbal JV, Rodolfo KS, Tubianosa BS, Delos Reyes PJ, Ruelo HR (1991) Fire and mud, USGS Publication Repository, Tephra falls of the 1991 eruptions of Mount Pinatubo
- Pallister JS, Hoblitt RP, Meeker GP, Newhall CG, Knight RJ, Siems DF, Newhall CG, Punongbayan RS (1996) Magma mixing at Mount Pinatubo volcano: petrographic and chemical evidence from the 1991 deposits, fire and mud. Eruptions and Lahars of Mount Pinatubo, Philippines. Seattle University of Washington Press, pp 687–732
- Paris R (2013) Shield volcano. In: Encyclopedia of earth sciences series. pp 910–911. https://doi. org/10.1007/978-1-4020-4399-4_319
- Parker DE, Brownscombe JL (1983) Stratospheric warming following the El Chichón volcanic eruption. Nature 301:406–408. https://doi.org/10.1038/301406a0
- Parker D, Wilson H, Jones P, Christy J, Folland C (1996) The impact of Mount Pinatubo on worldwide temperatures. Int J Climatol 16:487–497. https://doi.org/10.1002/(SICI)1097-0088 (199605)16:5%3C487::AID-JOC39%3E3.0.CO;2-J
- Peterfreund AR (1981) The geology of magma systems background and review
- Post JD (1984) Climatic variability and the European mortality wave of the early 1740s. J Interdiscip Hist 15(1):1–30. https://doi.org/10.2307/203592
- Potts RJ (1993) Satellite observations of Mt. Pinatubo ash clouds. Aust Meteorol Mag 42:59-68
- Predybaylo E, Stenchikov G, Wittenberg AT (2020) El Niño/Southern Oscillation response to low-latitude volcanic eruptions depends on ocean pre-conditions and eruption timing. Commun Earth Environ 1:12. https://doi.org/10.1038/s43247-020-0013-y
- Pulselli FM, Marchi M (2015) Global warming potential and the net carbon balance. In: Reference module in earth systems and environmental sciences. Elsevier. https://doi.org/10.1016/B978-0-12-409548-9.09526-9
- Pyle DM (2015) Sizes of volcanic eruptions. In: The encyclopedia of volcanoes, 2nd edn. Elsevier Inc. https://doi.org/10.1016/b978-0-12-385938-9.00013-4
- Quiroz RS (1984) Compact review of observational knowledge of blocking, with emphasis on the long-wave composition of blocks. In: Annual climate diagnostics workshop, 7th, Boulder, CO., Oct. 18–22, 1982, Proceedings, Wash., D.C., U.S. National Oceanic and Atmospheric Administration, March, 1983, pp 104–107, Climate Analysis Ctr., NMC/NWS/NOAA, Wash., D.C. M.G.A, 35, 8–254
- Read HH, Watson J (1975) Introduction to geology. Halsted, New York, pp 13–15. ISBN 978-0-470-71165-1
- Robock A (1979) The "little ice age": northern hemisphere average observations and model calculations. Science 206:1402–1404
- Robock A (2000) Volcanic eruptions and climate. Rev Geophys 38:191–219. https://doi.org/10. 1029/1998RG000054
- Ropelewski CF, Halpert MS (1987) Global and regional scale precipitation patterns associated with the El Niño/Southern Oscillation. Mon Weather Rev 115:1606–1626. https://doi.org/10.1175/ 1520-0493(1987)115%3C1606:GARSPP%3E2.0.CO;2

- Saxena P, Sonwani S (2020) Remediation of ozone pollution by ornamental plants in indoor environment. Glob J Environ Sci 6(4):497–508
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Srivastava A, Tyagi M, Kaur S (2019) Impact of tropospheric ozone on plant metabolism—a review. Pollut Res 38(1):175–180
- Schoeberl MR, Bhartia PK, Hilsenrath E, Torres O, Corp HSTX (1993) Tropical ozone loss following the eruption of Mt. Pinatubo. Geophys Res Lett 20(1):29–32. https://doi.org/10. 1029/92GL02637
- Self S, Rampino MR, Zhao J, Katz MG (1997) Volcanic aerosol perturbations and strong El Nino events: no general correlation. Geophys Res Lett 24:1247–1250. https://doi.org/10.1029/ 97GL01127
- Shindell DT, Schmidt GA, Miller RL, Mann ME (2003) Volcanic and solar forcing of climate change during the preindustrial era. J Clim 16(24):4094–4107. https://doi.org/10.1175/1520-0442(2003)016%3C4094:VASFOC%3E2.0.CO;2
- Shiogama H, Emori S, Mochizuki T, Yasunaka S, Yokohata T, Ishii M, Nozawa T, Kimoto M (2010) Possible influence of volcanic activity on the decadal potential predictability of the natural variability in near-term climate predictions. Adv Meteorol 2010:657318. https://doi.org/ 10.1155/2010/657318
- Siebert L, Cottrell E, Venzke E, Andrews B (2015) Earth's volcanoes and their eruptions: an overview. In: The encyclopedia of volcanoes, 2nd edn, issue 1. Elsevier. https://doi.org/10. 1016/b978-0-12-385938-9.00012-2
- Siegel S, Siegel B (1984) First estimate of annual mercury flux at the Kilauea main vent. Nature 309:146–147. https://doi.org/10.1038/309146a0
- Siegel B, Siegel S, Nachbar-Hapai M, Russell C (1987) Geotoxicology: are thermal mercury and sulfur emissions hazardous to health? In: El Sahb M, Murtry T (eds) Proceedings of international symposium on natural and man-made catastrophe. Univ of Quebec-Rimouski Reidel, Dordrecht
- Sigurdsson H (2000) Volcanic episodes and rates of volcanism. In: Sigurdsson H (ed) Encyclopedia of volcanoes. Academic, San Diego, pp 271–279. ISBN: 9780080547985
- Simkin T, Siebert L (1994) Volcanoes of the world. Geoscience Press, Tuscon, 368 p
- Singh A, Agrawal M (2008) Acid rain and its ecological consequences. J Environ Biol 29:15–24. Academy of Environmental Biology, India. PMID: 18831326
- Smellie JL (1999) Subglacial eruptions. In: Encyclopedia of volcanoes, 1st edn. ISBN: 9780080547985
- Smith WD (1909) Contributions to the physiography of the Philippine Islands: IV. The country between Subig and Mount Pinatubo. Philippine J Sci 4(A):19–25
- Solomon S, Portmann RW, Garcia RR, Thomason LW, Poole LR, McCormick MP (1996) The role of aerosol variations in anthropogenic ozone depletion at northern midlatitudes. J Geophys Res 101:6713–6727. https://doi.org/10.1029/95JD03353
- Speight MR, Henderson PA (2013) Marine ecology: concepts and applications. Wiley-Blackwell.-ISBN: 978-1-444-33545-3
- Spence RJS, Kelman I, Baxter PJ, Zuccaro G, Petrazzuoli S (2005) Residential building and occupant vulnerability to tephra fall. Nat Hazards Earth Syst Sci 5:477–494. https://doi.org/ 10.5194/nhess-5-477-2005
- Stenchikov GL, Kirchner I, Robock A, Graf H-F, Antuña JC, Grainger RG, Lambert A, Thomason L (1998) Radiative forcing from the 1991 Mount Pinatubo volcanic eruption. J Geophys Res 103:13837–13857. https://doi.org/10.1029/98JD00693
- Stewart C, Johnston DM, Leonard GS, Horwell CJ, Thordarson T, Cronin SJ (2006) Contamination of water supplies by volcanic ashfall: a literature review and simple impact modelling. J Volcanol Geotherm Res 158:296–306. https://doi.org/10.1016/j.jvolgeores.2006.07.002
- Symonds RB, Rose WI, Bluth GJS, Gerlach TM (1994) Volcanic-gas studies: methods, results and applications. Rev Mineral 30:1–66

- Taddeucci J, Edmonds M, Houghton B, James MR, Vergniolle S (2015) Chapter 27: Hawaiian and Strombolian eruptions. In: Sigurdsson H (ed) The encyclopedia of volcanoes, 2nd edn. Academic, pp 485–503. https://doi.org/10.1016/B978-0-12-385938-9.00027-4
- Tang Q, Hess PG, Brown-Steiner B, Kinnison DE (2013) Tropospheric ozone decrease due to the Mount Pinatubo eruption: reduced stratospheric influx. Geophys Res Lett 40:5553–5558. https://doi.org/10.1002/2013GL056563
- Tanguy JC, Ribière C, Scarth A, Tjetjep WS (1998) Victims from volcanic eruptions: a revised database. Bull Volcanol 60(2):137–144. https://doi.org/10.1007/s004450050222
- Textor C, Graf HF, Timmreck C, Robock A (2004) Emissions from volcanoes. In: Granier C, Artaxo P, Reeves CE (eds) Emissions of atmospheric trace compounds. Advances in global change research, vol 18. Springer, Dordrecht. https://doi.org/10.1007/978-1-4020-2167-1_7
- Thordarson T, Self S (2003) Atmospheric and environmental effects of the 1783–1784 Laki eruption: a review and reassessment. J Geophys Res 108(D1):4011. https://doi.org/10.1029/2001JD002042
- Timmreck C (2012) Modeling the climatic effects of large explosive volcanic eruptions. Wiley Interdiscip Rev 3:545–564. https://doi.org/10.1002/wcc.192
- Twomey S (1974) Pollution and planetary albedo. Atmos Environ 8:1251-1265
- Wallace PJ (2005) Volatiles in subduction zone magmas: concentrations and fluxes based on melt inclusion and volcanic gas data. J Volcanol Geotherm Res 140:217–240. https://doi.org/10. 1016/j.jvolgeores.2004.07.023
- White JDL, Schipper CI, Kano K (2015) Submarine explosive eruptions. In: The encyclopedia of volcanoes. pp 553–569. https://doi.org/10.1016/b978-0-12-385938-9.00031-6
- Witham CS (2005) Volcanic disasters and incidents: a new database. J Volcanol Geotherm Res 148(3–4):191–233. https://doi.org/10.1016/j.jvolgeores.2005.04.017
- Wotzlaw JF, Bindeman I, Stern R (2015) Rapid heterogeneous assembly of multiple magma reservoirs prior to Yellowstone super eruptions. Sci Rep 5:14026. https://doi.org/10.1038/ srep14026
- Zanchettin D, Timmreck C, Graf HF (2012) Bi-decadal variability excited in the coupled ocean– atmosphere system by strong tropical volcanic eruptions. Clim Dyn 39:419–444. https://doi.org/ 10.1007/s00382-011-1167-1
- Zhang DD, Lee HF, Wang C, Li B, Pei Q, Zhang J, An Y (2011) The causality analysis of climate change and large-scale human crisis. Proc Natl Acad Sci U S A 108(42):17296–17301. https:// doi.org/10.1073/pnas.1104268108
- Zuev VV, Zueva NE, Savelieva ES, Gerasimov VV (2015) The Antarctic ozone depletion caused by Erebus volcano gas emissions. Atmos Environ 122:393–399. https://doi.org/10.1016/j. atmosenv.2015.10.005
- Zuskin E, Mustajbegovic J, Jelinic J, Pucarin-Cvetkovic J, Milosevic M (2007) Effects of volcanic eruptions on environment and health. Arch Ind Hyg Toxicol 58(4):479–486. https://doi.org/10. 2478/v10004-007-0041-3



Assessment of Extreme Firework Episode in a Coastal City of Southern India: Kannur as a Case Study

C. T. Resmi, T. Nishanth, M. K. Satheesh Kumar, K. T. Valsaraj, and M. Balachandramohan

Abstract

Bursting of firecrackers turns out to be a vital part of festivals and celebrations. The amount of air and noise pollutions caused by such celebrations has been increasing significantly for the past few years till the COVID episode. Particulate matter and toxic gases emitted as part of fire bursting pose a serious threat to human health. A variety of festivals and celebrations taking place in different parts of India pollute the rural and urban areas. Vishu is a regional New Year festival being celebrated with coordinated fireworks all over the south Indian state of Kerala. Intense fireworks usually start on the eve of Vishu from 18:00 to 22: 00 h (IST) and on the day of Vishu from 04:00 to 07:00 h in two spells, which create smoke and dust in the atmosphere that leads to deterioration of air quality. Here, we describe the variations of surface ozone (O_3) and other trace pollutants for pre-, post- and Vishu days for two successive years (2020 and 2021) at Kannur town. No significant variations in air pollutants were observed on the Vishu days in 2020 due to countrywide lockdown to curb the transmission of COVID in India. But higher levels of trace pollutants were found during the intense fireworks during the two spells in the night of Vishu eve and early morning of Vishu day in 2021. Surface O₃ was found to be increased by 51%

C. T. Resmi · M. Balachandramohan

Department of Physics, Erode Arts and Science College, Erode, Tamil Nadu, India

T. Nishanth (\boxtimes) Department of Physics, Sree Krishna College Guruvayur, Guruvayur, Kerala, India

M. K. Satheesh Kumar Department of Atomic and Molecular Physics, MAHE, Manipal, Karnataka, India

K. T. Valsaraj Cain Department of Chemical Engineering, Louisiana State University, Baton Rouge, LA, USA

267

 $^{{\}rm (}^{\rm C}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_11

during the evening spell of fireworks on the eve of Vishu day and 61% on early morning spell on the Vishu day. A sharp increase in NO, NO₂, CO, SO₂, BTEX and NH_3 , PM_{10} and PM_{25} concentration was observed during the evening spell of fireworks on the eve of Vishu day and early morning spell on the Vishu day when compared to the respective time in the pre-Vishu days. Metal concentrations associated with particulate matter were found to be higher during Vishu days than pre- and post-Vishu days. The average concentrations of metals are shown in the order S > Al > Cl > Ca > K > Zn > Na > Mg > Pb > Cu> Ba > Cr > Fe > Ni > Sr > Mn. This shows that air pollution due to trace gases and particulate matter increases exponentially on fireworks days, unlike normal days. Such short duration activities carried out for visual and auditory pleasure have far-reaching effects. Therefore, it is evident that fireworks associated with celebrations and festivals are an event that can influence air quality to a large extent. In a densely populated area like Kerala, it is necessary to assess the consequences of such celebrations using fireworks and come up with a protocol for its use. Further, this work throws light on the absence of air pollution during the festival ban in 2020 in connection with a severe lockdown to curb COVID-19 transmission.

Keywords

Fireworks · COVID-19 · Air pollution · Air quality · Southern India

11.1 Introduction

Celebrations associated with festivals, traditions and cultural events are considered to be an integral part of Indian social life, which is always enriched with spectacular display of fireworks. Thus, fireworks play a vital part of religious celebrations, sporting events, political victories and New Year celebrations in the world. Many cultural and national celebrations such as Bastille Day in France, Guy Fawkes Night in the UK, Lantern festival in China and Taiwan, Diwali festival in India, Anzac Day celebration in Australia, Canada Day in Canada and Independence Day in the USA are celebrated with extensive and colourful fireworks displays. Widespread emissions of air pollutants released by fireworks can reduce visibility and significantly deteriorate air quality (Pope et al. 2016; Kong et al. 2015; Nasir and Brahmaiah 2015) in the fire bursting site and the neighbouring regions, which induces a direct impact on human health (Hamad et al. 2016; Seidel and Birnbaum 2015). The lightening of fireworks starts with an exothermic process engendering intense sound and smoke as a by-product holding gases like CO₂, CO, SO₂, NO, NO₂, VOCs, PM₁₀ and PM_{2.5} along with several metal salts such as aluminium, manganese, magnesium, strontium, barium copper, potassium and cadmium (Garg et al. 2018; Kumar et al. 2016; Singh et al. 2010). Nishanth et al. (2012) have classified a variety of hazardous organic compounds in the atmosphere during the fireworks associated with a south Indian festival of Vishu. Inhalation of trace metals

(Cd, Ni, Cr, Mn, Cu, etc.) released from fireworks results in severe health effects such as neurological, carcinogenic, lung diseases and haematological effects (Greven et al. 2019; Gouder and Montefort 2014; Barman et al. 2008).

Diwali (Deepavali) is a festival of light celebrated across India every year in the end of October or the first week of November with great enthusiasm by all sections of people. Detailed studies by various researchers on the widespread air quality changes caused by Diwali fireworks in north India have already been published (Saxena et al. 2020a; Singh et al. 2019; Peshin et al. 2017; Parkhi et al. 2016; Mandal et al. 2012; Pal et al. 2013; Chatterjee et al. 2013; Tiwari et al. 2012; Kulshrestha et al. 2004; Ravindra et al. 2003).

Firework display is an essential part of all festivals and celebrations, which usually attracts large crowds due to the spectacular display of colours in the sky. Even though it occurs in small duration, it is a potential source of air and sound pollution. Hence, short-term exposure to air pollution results in severe health issues. Since infants are more active in pyrotechnics, they are more prone to mortality from short-term exposure to elevated levels of air pollution (Greven et al. 2019). The oxidative potential of particulate matter, toxic gases and trace metals present in the emission plume of smoke makes fireworks an extreme event that changes the atmosphere in a short duration of time (Sonwani et al. 2021).

Vishu is celebrated as a regional New Year Day in Kerala State, which marks the commencement of agricultural activities. This festival is celebrated not only in Kerala but also in the border areas of the neighbouring states. Being the beginning of a New Year Day, it is often celebrated with extensive fireworks on 13 or 14 April throughout the state. Fireworks usually set off at home on the eve of Vishu from late evening to midnight and on the early morning of Vishu day. Thus, the intense fireworks in a short span create a terrifying smoke and dust in the atmosphere accompanied by extreme sound pollution. This smoke that spread during night time often produces a thick morning fog, which reduces the visibility. Moreover, black carbon dust and toxic gases present in this morning fog severely affect human health to a large extent (Saxena et al. 2020b). The interior of houses is generally contaminated with dust and pollutant gases emitted from the fireworks during night and early morning spells of the fireworks. As a result, inhaling dust and toxic gases in the whole night of Vishu can cause severe health issues, especially to infants and elderly people.

Research activities have been initiated in Kannur University to explore the chemistry of trace gases in a rural environment since 2009 with the support of ISRO and these activities and extended to Kannur town in 2019. This case study is an outcome of our long-term observation in the variations of trace pollutants such as surface O₃, NO, NO₂, CO, SO₂, BTEX and NH₃, PM₁₀ and PM_{2.5} at a residential site of Kannur town during the Vishu festival in 2020 and 2021. This work is a result of the air samples collected from 10 to 17 April in 2020 and 2021. We could not perform the qualitative and quantitative classifications of the organic compounds during fireworks, which is highly relevant.

11.2 Description of Monitoring Site

Kannur is the northern district of Kerala lying along the coastal belt of the Arabian Sea, which makes it one of the most popular tourist destinations in South India. The geographical coordinates are 11.87 N and 75.37 E, 3 m msl, which spans an area of 2961 km² with a population density of 852 per km². The map of Kannur is shown in Fig. 11.1a, and the outlook of Kannur town with the monitoring location is shown in Fig. 11.1b. Kannur town is an urbanized city having few industries including plywood, rub wood, mattress factories and handlooms in its neighbourhood areas. Agriculture and fisheries are the key sectors in the villages of this district. The monitoring station is installed close to the National Highway (NH 17), which has reasonably heavy traffic during the day and minimum at night. Further, it is surrounded by administrative buildings and residential areas and is about 2 km away from the Arabian Sea. Since the industries are quite far, transport-related activities, such as emissions from vehicular traffic, are major sources of various gaseous and particulate pollutants at this site.

Kannur is conferred with a pleasant climate throughout the year, due to the proximity of Arabian Sea in the west and the Western Ghats in the east. Relatively higher temperatures, low rainfall and a faintly humid climate are the main features of the summer season in Kerala. Measurements of trace pollutants were carried out using the respective ground-based gas analyzers from Environment S.A France. Detailed description of the instruments used for the study is reported in our previous publication (Resmi et al. 2020).



Fig. 11.1 (a) Kannur in South India (b) airborne vision of Kannur town and the monitoring station

11.3 Results and Discussion

11.3.1 Variation of Surface O₃

In order to investigate the effect of fireworks on the concentrations of trace pollutants over Kannur town, the study retro is divided into three four, viz. pre-Vishu days (10, 11, 12 April), day before Vishu (13 April), Vishu day (14 April) and post-Vishu days (15, 16 17 April). Figure 11.2 shows the 24-h variation of O_3 over the monitoring station for the study period days during 2020 and 2021. It is found that the diurnal variation of O_3 during these 2 years exhibited a similar profile with different amplitudes. On a diurnal basis, O₃ concentration usually starts to increase in-tune with solar radiation from the morning hours and reaches its maximum level at noontime hours. Then, it exhibits a declining phase in afternoon hours and reaches its low level during the late evening and night hours. The high concentration of O_3 observed in the daytime is primarily due to the enhanced VOC-NOx chemistry in the presence of intense solar radiation. During night-time hours, photochemical production becomes weak and O_3 titration with NO is more pronounced, which leads to a decrease in O_3 concentration. However, no significant change in ozone has been noticed in 2020 as the fire bursting has been banned in the wake of country wide lockdown.

In 2021, O₃ concentrations show a momentous increase on the eve (18:00–22: 00 h) of Vishu day and the early morning (04:00–10:00 h) of Vishu day due to intense fire bursting over the observational site. Thus, just after the fire bursting, O₃ concentration enhanced from ~19.2 ppbv to ~29.0 ppbv (an upsurge of 51%) in the evening spell on the eve of Vishu day and from ~11.4 ppbv to ~18.4 ppbv (an increase of 61%) on the early morning spell on Vishu day. Surface O₃ is usually formed only in daytime by the photodissociation of NO₂ in the presence of sunlight ($\lambda < 420$ nm). Thus, in a closed state, the following set of reactions shows a production and dissociation of O₃ (Seinfeld and Pandis 2006).

$$NO_2 \rightarrow NO + O$$
 (R11.1)

$$O + O_2 + M \rightarrow O_3 + M \tag{R11.2}$$

$$NO + O_3 \rightarrow NO_2 + O_2 \tag{R11.3}$$

The exciting feature of our measurement is the upsurge in O_3 level during the evening and early morning spells of Vishu even in the lack of solar radiation. This growth is connected with fire bursting that occurred in two spells of 2021. Accordingly, the O_3 produced during this episode was mainly at the expense of firecrackers. The additional NO₂ essential for the upsurge of O_3 was produced from the fire bursting event.

$$NO + HO_2 \rightarrow NO_2 + OH$$
 (R11.4)



Fig. 11.2 Variation of surface O₃ during the study period in (a) 2020 and (b) 2021

$$NO + RO_2 \rightarrow NO_2 + RO$$
 (R11.5)

Thus, night-time O_3 enhancement is by the photolysis of NO_2 present in the smoke activated by intense flash of the fireworks.

11.3.2 Variation of Oxides of Nitrogen

Variation of NO and NO₂ in the days of observation in 2020 and 2021 is shown in Fig. 11.3a–d. The diurnal profiles of NO and NO₂ show higher concentrations in the morning and late evening hours and low concentrations during daytime hours. The enhancement in the concentrations of NO and NO₂ during the morning and late evening hours is due to vehicular exhaust and shallow boundary layer height. Soon after sunrise, peroxyacetyl nitrate (PAN) breaks into NO and NO in the presence of sunlight interacts quickly with O₃ or O₂ to form NO₂ due to its short lifetime (Wang et al. 2019). Subsequently, NO_2 undergoes photolysis in the presence of sunlight to form NO and atomic oxygen, which combines with molecular oxygen produces surface O₃ indicated by reactions R1 and R2. After the maximum concentration observed in the morning, NO₂ starts declining until it extends its lowest concentration between 14:00 and 16:00 h, and then reaches a steady upturn until late-night hours. The main reason behind the decline of NO_2 during day time is due to the increased photolysis by intense solar radiation. Thus, this decrease in NO₂ concentration is at the expense of photochemical production of surface O_3 in the presence of intense solar flux. The night-time concentration of NO and NO₂ shows a gradual increase from 19:00 h to midnight on 13 April 2021.



Fig. 11.3 Diurnal variation of NO during the study period in (a) 2020 and (b) 2021 and NO_2 during the study period in (c) 2020 and (d) 2021

In this festival episode, NO concentration also exhibits a small increase from ~6.2 ppbv to ~7.8 ppbv (an upsurge of 26%) observed in the evening spell on the eve of Vishu and from ~7.1 ppbv to ~9.4 ppbv (an upsurge of 32%) on the early morning spell on Vishu day. Surprisingly, the concentration of NO₂ enhances from ~7.5 ppbv to ~11.6 ppbv (an increase of 56%) in the evening spell on the eve of Vishu and from ~8.1 ppbv to ~12.4 ppbv (an upsurge of 53%) on the early morning spell on Vishu day. This shows an increase in NO and NO₂ concentrations in the evening and the early morning spell only in 2021 due to the lockdown imposed in 2020.

11.3.3 Variation of CO and SO₂

Figure 11.4a, b shows the diurnal variations of CO during pre-, post- and Vishu days in 2020 and 2021 at Kannur town. Diurnal variation of CO displayed two peaks in morning and evening due to traffic hours in the town. Usually, the morning and evening peaks are observed at 08:00–10:00 h and 17:00–20:00 h in this location. Various groups in India have reported a similar variation of CO in many urban areas (Beig et al. 2007; Mahapatra et al. 2014; Verma et al. 2017; Resmi et al. 2020). Generally, during pre- and post-Vishu days, the observed concentration of CO varies from 340 to 680 ppbv and an increase (62%) in CO concentration is noticed from the beginning (540 ppbv) of fire bursting on the evening spell on the eve of Vishu day



Fig. 11.4 Diurnal variation of CO during the study period in (**a**) 2020 and (**b**) 2021 and SO₂ during the study period in (**c**) 2020 and (**d**) 2021

and it reaches a maximum value (850–882 ppbv) during 19:00–20:00 h. Likewise, during the early morning on the Vishu day, the concentration of CO shows an upsurge of 72% and attains a maximum value (840–890 ppbv) between 06:00 and 08:00 h with respect to the pre-Vishu concentration of 520–540 ppbv. Further, its concentration is observed to be high during the evening on Vishu day with respect to pre-Vishu days, which account for the bursting of fireworks in the surroundings of the observational site.

The diurnal variations of SO₂ during pre-, post- and Vishu days in 2020 and 2021 at Kannur town are shown in Fig. 11.4c, d. The primary source of SO₂ at the observational site is from vehicles after the combustion of sulphur present in diesel. During daytime, the anthropogenic activities (such as vehicular and industrial actions, and biomass burning) contribute fairly large release of SO₂ in the lower atmosphere and it remains gas-phase SO₂ in night-time hours. The diurnal variation of SO₂ exhibited two distinct peaks during morning (9:00–11:00 h) and evening (19: 00–22:00 h) of the pre- and post-Vishu days, and it varies from 1.7 ± 0.41 to 3.8 ± 0.52 ppbv. Just after the fire bursting, SO₂ concentration enhances from ~3.52 ppbv to ~6.88 ppbv (an increase of 96%) in the evening spell on the eve of Vishu day and from ~2.4 ppbv to ~5.8 ppbv (an increase of 142%) on the early morning spell on Vishu day.

11.3.4 Diurnal Variation of BTEX and NH₃

Benzene, toluene, ethylbenzene and xylene (BTEX) are the main hydrocarbons found in the lower atmosphere in the group of volatile organic compounds (VOCs). Figure 11.5a, b shows the diurnal variation of BTEX and during pre-, post- and Vishu days in 2020 and 2021 at Kannur town. BTEX shows a pronounced enhancement in the morning, and evening hours and low in noontime hours. Daytime and nocturnal concentrations of BTEX exhibit a considerable increase during fire bursting episode in 2021 over Kannur town. BTEX concentration enhances from ~6.3 ppbv to ~12.1 ppbv (an increase of 92%) in the evening spell on the eve of Vishu day and from ~5.5 ppbv to ~11.3 ppbv (an increase of 105%) on the early morning spell on Vishu day.

Figure 11.5c, d describes the 24-h variation of NH₃ during pre-, post- and Vishu days in 2020 and 2021 at Kannur town. NH₃ shows a progressive deviation with higher concentrations during daytime hours due to the anthropogenic activities and a build-up of pollutants by low wind speeds (Kuttippurath et al. 2020; Behera et al. 2013). During pre- and post-Vishu days, the mixing ratios of NH₃ vary from 4.65 ± 0.42 to 5.88 ± 0.48 ppbv on a diurnal basis. The concentration of NH₃ starts increasing from 5.25 ppbv at the beginning of the fire bursting and reaches its maximum level of 5.88 ppbv with an upsurge of 12% in pre-Vishu days. Quite similar to this, on the Vishu day, the concentration of NH₃ increases from 5.02 ppbv to 5.75 ppbv with an increase of 15%. Despite the fire bursting activity is reduced considerably after the morning of 14 April on Vishu day, the concentrations of trace gases except BTEX and NH₃ remain higher level compared to post-Vishu days.



Fig. 11.5 Diurnal variation of BTEX during the study period in (a) 2020 and (b) 2021 and NH_3 during the study period in (c) 2020 and (d) 2021

11.3.5 Variation of PM₁₀ and PM_{2.5}

Particulate matter is formed by a cluster of small and large particles of different chemical configurations, and it consists of both fine $(PM_{2,5})$ and coarse (PM_{10}) modes of airborne particles (Pakbin et al. 2010). The fine mode particles are formed by burning or gas-to-particle conversions and are rich in carbon, sulphates, ammonium and nitrate ions, as well as trace elements (Liu et al. 2015). The anthropogenic emissions of trace gases and particulate matter thus exert a significant influence on the chemistry and radiative balance of the local urban environment. Further, it influences the regional production of surface O_3 and OH radicals affecting the urban air quality. Diurnal variations of PM₁₀ and PM_{2.5} observed in 2020 and 2021 are shown in Fig. 11.6a-d. In 2021, PM₁₀ and PM_{2.5} show higher concentrations during morning (09:00 to 12:00 IST) and evening (16:00 to 20:00) hours due to the crowning traffic and late-night hours due to the accumulation of pollutants by shallow boundary layer. After morning enhancement between 07:00 and 10:00 h, the concentrations of particulate matter gradually decrease until 15:00 h due to the diffusion of pollutants by broadened boundary layer height in the presence of intense solar radiation (Qu et al. 2017).

An increase in concentration of PM_{10} and $PM_{2.5}$ is noticed from the beginning of fire bursting on the eve of Vishu day and the early morning on the Vishu day. Just



Fig. 11.6 Diurnal variation of PM_{10} during the study period in (a) 2020 and (b) 2021 and $PM_{2.5}$ during the study period in (c) 2020 and (d) 2021

after the fire bursting, PM_{10} concentration enhances from ~110 µg/m³ to ~191 µg/m³ (an upsurge of 74%) in the evening spell on the eve of Vishu day and from ~108 µg/m³ to ~195 µg/m³ (an increase of 81%) on the early morning spell on Vishu day. Likewise, an increase in concentration of $PM_{2.5}$ is observed from the beginning of fire bursting on the evening spell on 13 April and morning spell on 14 April 2021. Likewise, mounting levels of $PM_{2.5}$ are observed from ~84 µg/m³ to ~135 µg/m³ (an increase of 61%) in the evening spell on the eve of Vishu day and from ~75 µg/m³ to ~140 µg/m³ (an increase of 87%) on the early morning spell on Vishu day.

11.3.6 Variation of Metal Concentrations Associated with Particulate Matters

The present study extended to monitor the concentration of various inorganic elements present in the ambient air of Kannur town during the observation days in 2021. Beni and Esmaeili (2020); Jan et al. (2015) have reported the adverse effect of metals contained in particulate matter on the respiratory tract of human beings due to their bio-accumulative and poisonous nature in the environment. World Health Organization (WHO 2016) that included PM_{10} containing Pb, Ni, Co, Cd and As is a prominent air pollutant. Exposure to high levels of Pb and Mn can lead to severe neurotoxic and haematological effects in children, and Ni, Cd and Cr cause


Fig. 11.7 Concentration of metals present in ambient air during pre-, post- and Vishu days

oncogenic effects in humans (Shen et al. 2018; Singh et al. 2011). In order to study the metal concentrations bounded with the particulate matter, PM_{10} in the ambient air on the observational days is collected using a high-volume air sampler located on the roof in the monitoring station. The concentration of metals associated with PM_{10} during pre-Vishu, Vishu and post-Vishu days is shown graphically in Fig. 11.7.

It is observed that metal concentrations are found to be higher in Vishu days than pre- and post-Vishu days. The average concentration of metals shows in the order S > Al > Cl > Ca > K > Zn > Na > Mg > Pb > Cu > Ba > Cr > Fe > Ni > Sr > Mn. Based on the concentrations, major upsurge is observed for seven metal species in PM₁₀ samples (S, Al, Cl, Ca, K, Zn and Na) on Vishu day compared to samples from pre- and post-Vishu days. A similar enhancement in metal species has been observed by different groups across the globe (Shivani et al. 2020; Pervez et al. 2016; Tsai et al. 2012; Perrino et al. 2011). Generally, all the firecrackers contain potassium nitrate, potassium chlorates and potassium perchlorates in the form of a black powder, which acts as a combustible material (Moreno et al. 2007). Sulphur (S) is used as a propellant, whereas aluminium (Al) and copper (Cu) are used for colour displays and sparklers, while sodium (Na) was commonly used as metal oxidizer (Pervez et al. 2016; Tandon et al. 2008).

11.4 Conclusion

Colourful fireworks display is one of the main attractions of festival seasons. The south Indian state of Kerala celebrates Vishu, which earmarks the regional New Year, which falls in the middle of April month every year. This festival usually starts

in the evening of Vishu eve with coordinated fireworks set up to late night and resume in the early morning of Vishu day with two distinct spells of fire bursting episodes. Our observations were conducted on trace gases in 2020 and 2021, and it is further observed that the concentrations of all trace gases remain unchanged due to the ban of fireworks in the wake of strict lockdown. The observations in 2021 reveal the increase in O_3 concentration from 19.2 ppbv to 29.0 ppbv on the evening spell on the eve of Vishu day and from 11.4 ppbv to 18.4 ppbv on the morning spell of Vishu day. The concentration of NO increases from its value of ~ 6.2 ppbv to ~ 7.8 ppbv on the evening spell on the eve of Vishu day and from \sim 7.1 ppbv to \sim 9.4 ppbv on the morning spell of Vishu day. The abundance of NO_2 shows a clear upsurge of 4.1 ppbv on the evening spell on the eve of Vishu day and 4.3 ppbv in the morning spell for the year 2021. A sharp increase in CO concentration was observed during the evening spell (62%) and morning spell (72%) of firework event. The observed SO₂ concentration enhanced to 96% on evening spell and 142% to early morning spell for the year 2021. The concentration of BTEX increased from its value of ~6.3 ppbv just after the fire bursting to ~12.1 ppbv on evening spell and from ~5.5 ppbv to ~11.3 ppbv on early morning spell of Vishu day. NH₃ shows an enhancement of 12% in the evening spell and 15% in the morning spell of Vishu day. Mixing ratios of PM₁₀ increase in the evening spell in Vishu eve from 110 μ g/m³ to 191 μ g/m³, and in the morning spell of Vishu day, it is from 108 μ g/m³ to195 μ g/m³. Similarly, the concentration of PM_{2.5} is found to increase from 84 μ g/m³ to 135 μ g/ m^3 in the evening spell in Vishu eve and vary from 75 µg/m³ to 140 µg/m³ in the morning spell of Vishu day. All metal concentrations associated with particulate matters are found to be much higher during Vishu days than pre- and post-Vishu days.

The Indian festivals are the reflections of the vast cultural diversity in the country, and all festivals are becoming colourful, which are being celebrated with much enthusiasm. The common feature of all festivals is an extensive firework display at the end of the festivals. The use of firecrackers in festivals and celebrations in Kerala has been alarmingly increasing in every year. On the day of Vishu, it is a joyful experience to witness the spectacular wonders of the firecrackers in the sky over each house. Such type of celebrations enhances the concentrations of trace gases and particulate matter in the atmosphere, leading to serious health problems. Celebrations employing fireworks in India have not yet been legally restricted except in the lockdown periods. Instead of legally banning such activities, the use of fireworks in celebrations and festivals can be curtailed to some extent once the public is made aware of the harmful effects of fireworks. Thus, a public awareness of the air and sound pollutions of fireworks may be created to discuss on the severity of this man-made environmental threat through the media in order to save our environment.

Acknowledgement The authors are grateful to the Indian Space Research Organization (ISRO), Bangalore, for the financial support provided through the Atmospheric Chemistry, Transport, and Modeling (AT-CTM) project of the Geosphere-Biosphere Programme. Likewise, the authors express their gratitude to the Kerala State Council for Science Technology and Environment (KSCSTE) for granting financial support for the study of air pollutants over the Kannur region. Resmi expresses her gratitude to Dr. R. Sankarasubramanian (Principal) and Dr. D. Manivannan (HOD of Physics) of Erode Arts and Science College, Tamil Nadu, for providing the necessary facilities and constant support. Prof. Valsaraj appreciates the endowment funds support from the Charles and Hilda Roddey Distinguished Professorship at LSU.

References

- Barman SC, Singh R, Negi MPS, Bhargava SK (2008) Ambient air quality of Lucknow City (India) during use of fireworks on Diwali festival. Environ Monit Assess 137:495–504
- Behera NS, Sharma M, Aneja VP, Balasubramanian R (2013) Ammonia in the atmosphere: a review on emission sources, atmospheric chemistry and deposition on terrestrial bodies. Environ Sci Pollut Res 20:8092–8131
- Beig G, Gunthe S, Jadhav DB (2007) Simultaneous measurements of O₃ and its precursors on a diurnal scale at a semi urban site in India. J Atmos Chem 57:239–253
- Beni AA, Esmaeili A (2020) Biosorption, an efficient method for removing heavy metals from industrial effluents: a review. Environ Technol Innov 17:100503
- Chatterjee A, Sarkar C, Adak A, Mukherjee U, Ghosh SK, Raha S (2013) Ambient air quality during Diwali Festival over Kolkata—a mega city in India. Aerosol Air Qual Res 13:1133– 1144. https://doi.org/10.1007/s11356-014-3347-0
- Garg A, Sharma P, Beig G, Ghosh C (2018) Temporal mount in air pollutants allied with religious fiesta: Diwali, festival of Lights. In: Emerging issues in ecology and environmental science. pp 11–25. https://doi.org/10.1007/978-3-319-99398-0_2
- Gouder C, Montefort S (2014) Potential impact of fireworks on respiratory health. Lung India 31(4): 375–379
- Greven FE, Vonk JM, Fischer P, Duijm F, Vink NM, Brunekreef B (2019) Air pollution during New Year's fireworks and daily mortality in the Netherlands. Sci Rep 9:5735. https://doi.org/10. 1038/s41598-019-42080-6
- Hamad S, Green D, Heo J (2016) Evaluation of health risk associated with fireworks activity at Central London. Air Qual Atmos Health 9(7):735–741
- Jan AT, Azam M, Siddiqui K, Ali A, Choi I, Rizwanul Haq QM (2015) Heavy metals and human health: mechanistic insight into toxicity and counter defense system of antioxidants. Int J Mol Sci 16:29592–29630. https://doi.org/10.3390/ijms161226183
- Kong S, Li L, Li X, Yin Y, Chen K, Liu D, Yuan L, Zhang Y, Shan Y, Ji Y (2015) The impacts of firework burning at the Chinese Spring Festival on air quality: insights of tracers, source evolution and aging processes. Atmos Chem Phys 15(4):2167–2184
- Kulshrestha UC, Rao TN, Azhaguvel S, Kulshrestha MJ (2004) Emissions and accumulation of metals in the atmosphere due to crackers and sparkles during Diwali festival in India. Atmos Environ 38(27):4421–4425
- Kumar M, Singh RK, Murari V, Singh AK, Singh RS, Banerjee T (2016) Fireworks induced particle pollution: a spatio-temporal analysis. Atmos Res 180:78–91
- Kuttippurath J, Singh A, Dash SP, Mallick N, Clerbaux C, Van Damme M, Clarisse L, Coheur PF, Raj S, Abbhishek K, Varikoden H (2020) Record high levels of atmospheric ammonia over India: spatial and temporal analyses. Sci Total Environ 740(20):139986
- Liu Z, Hu B, Wang L, Wu F, Gao W, Wang Y (2015) Seasonal and diurnal variation in particulate matter (PM₁₀ and PM_{2.5}) at an urban site of Beijing: analyses from a 9-year study. Environ Sci Pollut Res 22:627–642
- Mahapatra PS, Panda S, Walvekar PP, Kumar R, Das T, Gurjar BR (2014) Seasonal trends meteorological impacts and associated health risks with atmospheric concentrations of gaseous pollutants at an Indian coastal city. Environ Sci Pollut Res 21:11418–11432
- Mandal P, Prakash M, Bassin JK (2012) Impact of Diwali celebrations on urban air and noise quality in Delhi City, India. Environ Monit Assess 184:209–215

- Moreno T, Querol X, Alastuey A, Minguillon M, Pey J, Rodriguez S, Miro JV, Felis C, Gibbons W (2007) Recreational atmospheric pollution episodes: inhalable metalliferous particles from firework displays. Atmos Environ 41(5):913–922
- Nasir U, Brahmaiah D (2015) Impact of fireworks on ambient air quality: a case study. Int J Environ Sci Technol 12(4):1379–1386
- Nishanth T, Praseed KM, Rathnakaran K, Satheesh Kumar MK, Ravi Krishna R, Valsaraj KT (2012) Atmospheric pollution in a semi-urban, coastal region in India following festival seasons. Atmos Environ 47:295–306
- Pakbin P, Hudda N, Cheung KL, Moore KF, Sioutas C (2010) Spatial and temporal variability of coarse (PM₁₀-PM_{2.5}) particulate matter concentrations in the Los Angeles Area. Aerosol Sci Technol 44(7):514–525
- Pal R, Mahima Gupta A, Singh C, Tripathi A, Singh RB (2013) The effects of fireworks on ambient air and possible impact on cardiac health, during Deepawali festival in North India. World Heart J 5(1):21–32
- Parkhi N, Chate D, Ghude SD, Peshin S, Mahajan A, Srinivas R, Surendran D, Ali K, Singh S, Trimbake H, Beig G (2016) Large inter annual variation in air quality during the annual festival 'Diwali' in an Indian megacity. J Environ Sci 43:265–272
- Perrino C, Tiwari S, Catrambone M, Torre SD, Rantica E, Canepari S (2011) Chemical characterization of atmospheric pm in Delhi, India, during different periods of the year including Diwali festival. Atmos Pollut Res 2(4):418–427
- Pervez S, Rajan KC, Shippi D, Watson JG, Chow JC, Matawle JL (2016) Chemical speciation of aerosols and air quality degradation during the festival of lights (Diwali). Atmos Pollut Res 7(1): 92–99
- Peshin SK, Sinha P, Bisht A (2017) Impact of Diwali firework emissions on air quality of New Delhi, India during 2013-2015. Mausam 68:111–118
- Pope R, Marshall A, O'Kane B (2016) Observing UK Bonfire Night pollution from space: analysis of atmospheric aerosol. Weather 71(11):288–291
- Qu YW, Han Y, Wu YH, Gao P, Wang TJ (2017) Study of PBLH and its correlation with particulate matter from one-year observation over Nanjing, Southeast China. Remote Sens 9: 668
- Ravindra K, Mor S, Kaushik CP (2003) Short-term variation in air quality associated with firework events: a case study. J Environ Monit 5:260–264
- Resmi CT, Nishanth T, Satheesh Kumar MK, Balachandramohan M, Valsaraj KT (2020) Annular Solar Eclipse on 26 December 2019 and its Effect on trace pollutant concentrations and meteorological parameters in Kannur, India: a Coastal City. Asian J Atmos Environ 14:289–306
- Saxena P, Srivastava A, Verma S, Singh L, Sonwani S (2020a) Analysis of atmospheric pollutants during fireworks festival 'Diwali' at a residential site Delhi in India. In: Measurement, analysis and remediation of environmental pollutants. Springer, Singapore, pp 91–105
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020b) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29 (8):6639–6656
- Seidel DJ, Birnbaum AN (2015) Effects of Independence Day fireworks on atmospheric concentrations of fine particulate matter in the United States. Atmos Environ 115:192–198
- Seinfeld JH, Pandis SN (2006) Atmospheric chemistry and physics from air pollution to climate change. Wiley, New York
- Shen X, Chi Y, Xiong K (2018) The effect of heavy metal contamination on humans and animals in the vicinity of a zinc smelting facility. PLoS One 14(10):e0207423. https://doi.org/10.1371/ journal.pone.0207423
- Shivani GR, Saxena M, Sharma SK, Mandal TK (2020) Short-term degradation of air quality during major firework events in Delhi, India. Meteorol Atmos Phys 131:753–764. https://doi.org/10. 1007/s00703-018-0602-9
- Singh DP, Gadi R, Mandal TK, Dixit CK, Singh K, Saud T et al (2010) Study of temporal variation in ambient air quality during Diwali festival in India. Environ Monit Assess 169(1):1–13

- Singh R, Gautam N, Mishra A, Gupta R (2011) Heavy metals and living systems: an overview. Indian J Pharmacol 43:246–253
- Singh A, Panth P, Pope D (2019) Air quality during and after festivals: aerosol concentrations, composition and health effects. Atmos Res 227:220–232
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3(690444):1–20
- Tandon A, Yadav S, Attri AK (2008) City-wide sweeping a source for respirable particulate matter in the atmosphere. Atmos Environ 42:1064–1069
- Tiwari S, Chate DM, Srivastava MK, Safai PD, Srivastava AK, Bisht DS, Padmanabhamurty B (2012) Statistical evaluation of PM_{10} and distribution of PM_1 , $PM_{2.5}$, and PM_{10} in ambient air due to extreme fireworks episodes (Deepawali festivals) in megacity Delhi. Nat Hazards 61: 521-531
- Tsai HH, Chien LH, Yuan CS et al (2012) Influences of fireworks on chemical characteristics of atmospheric fine and coarse particles during Taiwan's lantern festival. Atmos Environ 62:256–264
- Verma N, Satsangi A, Lakhani A, Kumari KM, Lal S (2017) Diurnal seasonal and vertical variability in carbon monoxide levels at a semi-urban site in India. Clean Soil Air Water 45: 1600432
- Wang W, Li X, Shao M, Hu M, Zeng L, Wu Y, Tan T (2019) The impact of aerosols on photolysis frequencies and ozone production in Beijing during the 4-year period 2012–2015. Atmos Chem Phys 19:9413–9429
- WHO, World Health Organization (2016) Ambient air pollution: a global assessment of exposure and burden of disease



Air Pollution Episodes: Brief History, Mechanisms and Outlook

12

Narendra Ojha, Meghna Soni, Manish Kumar, Imran Girach, Som Kumar Sharma, and Sachin S. Gunthe

Abstract

The occurrences of pollution episodes have brought out the strong linkages of pollution with health subsequently catalysing numerous regulations worldwide. Pollution episodes have been driven by a variety of mechanisms such as accidental large emission, trapping of pollution by stagnant meteorology and intense photochemistry involving strong anthropogenic emissions. Few regions of the world, e.g. the Indo-Gangetic Plain (IGP) and Northern China Plain (NCP), experience recurring pollution episodes with enhanced levels of fine particulate matter ($PM_{2.5}$), ozone (O₃) and other pollutants. Reductions in industrial and vehicular emissions should be planned by taking unfavourable meteorology also into account, besides control measures on burning of the crop residues and forest biomass to reduce the severity of pollution episodes. The development of greenbelts in residential and commercial areas and green roofing over buildings could reduce dust and other pollutants. Air quality monitoring over dense networks and high-resolution modelling with assimilation of observations

N. Ojha (🖂) · S. K. Sharma

Physical Research Laboratory, Ahmedabad, India

M. Soni Physical Research Laboratory, Ahmedabad, India

Indian Institute of Technology Gandhinagar, Gandhinagar, India

M. Kumar Stockholm University, Stockholm, Sweden

I. Girach SPL, Vikram Sarabhai Space Centre, Thiruvananthapuram, India

S. S. Gunthe Civil Engineering Department, Indian Institute of Technology Madras, Chennai, India

© The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022 P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon:*

Assessment, Impacts and Mitigation, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_12 would be of paramount importance for more accurate forecasts and better preparedness to minimize potential impacts.

Keywords

Air quality · Trace gases · Aerosols · PM2.5 · Ozone · Smog · Health effects

12.1 Introduction

The periods of very high pollution levels lasting for a few hours to several days have commonly been referred to as the air pollution episodes. The impacts of air pollution episodes on the health of living beings depend on the type (toxicity) of pollutants, concentrations and the duration of the episodes (Saxena and Srivastava 2020; Sonwani and Saxena 2022). Some common examples of episodic air pollution events include the accidental releases of toxic pollutants from point sources, largescale forest fires, dust outbreaks in dry regions and season-specific smog in stable atmospheric conditions. Historically, the occurrence of air pollution episodes has established the links between pollution and health of living beings. In addition, numerous studies following up each major episode unravelled the diverse physical, chemical and dynamical processes, which govern pollution over a given region and the factors that could lead to severe pollution episodes. Pollutant concentrations over a particular region depend on the strength of various emission sources, effects of transport from upwind regions, photochemistry and the local atmospheric (meteorological) conditions such as boundary layer height, wind speed and humidity (Saxena and Ghosh 2011). With the dominance of one or more factors, the concentration of pollutants can be strongly enhanced resulting in extreme pollution episodes/events. Pollution episodes have pronounced impacts on the health of living beings and are often linked with premature mortalities.

Air pollution episodes have been caused by a wide variety of different mechanisms. Classic episodes, such as smog of Meuse Valley (1930) and Donora (1948), were linked with the accumulation of industrial emissions by terrain and local meteorological conditions. Strong coal burning and limited atmospheric ventilation caused the Great London smog (1952) leading to death of thousands of people. Summertime smog of Los Angles (1940s) was due to photochemical ozone production involving local emissions of ozone precursors. In contrast, the Bhopal gas tragedy (1984) leading to a large number of deaths within a short span of time was an industrial accident. Natural processes such as volcanoes and forest fires have also been found to dramatically impact the air quality and visibility conditions (Jain et al. 2021). Several regions of the world, such as the Northern China Plain (NCP) and the Indo-Gangetic Plain (IGP), experience recurring air pollution episodes. Higher PM_{2.5} in winters and strong ozone pollution in the summers impose a significant threat to the health of large population living in these regions (Sonwani et al. 2022).

Besides providing scientific understanding, the unprecedented impacts of pollution episodes have led to numerous legislations and regulations worldwide. Donora smog episode is suggested to have catalysed initiatives towards the first clean air legislation in the USA. Air Pollution Control Act of 1955 in the USA was aimed to facilitate research and technical assistance to different states to reduce air pollution. This was followed by the Clean Air Act of 1963 defining emission standards for industries and power plants in the USA. Several additional acts and amendments were also made subsequently (Jacobson 2002), for instance defining the National Ambient Air Quality Standards (NAAQS). In the UK also, Clean Air Act 1956 was passed in response to the London smog episodes leading to shifting of power plants to the rural areas. Industries were required to build chimneys to avoid near-surface accumulation of pollution under the act of 1968. Since then, numerous efforts have been ongoing towards new environmental legislations and implementations in different countries of the world. A detailed history of all major air pollution episodes and their impacts have been covered extensively in the literature (e.g. Jacobson 2002; Brimblecombe 2017; Fowler et al. 2020 and references therein). Here, in this chapter, we present an overview of the history of air pollution episodes and their impacts, key driving mechanisms, followed by an outlook including some perspectives on the control measures.

12.2 Global Distribution of Air Pollution

Air pollution episodes cause adverse health impacts including premature mortalities, and therefore, monitoring and forecasting of air quality are being carried out globally on a regular basis. While pollution has been naturally present since the onset of the atmosphere on the earth, nevertheless, since industrialization (the 1750s) it has increased dramatically over several developed and developing regions across the globe (IPCC 2007; IPCC 2013; Fowler et al. 2020). The twenty-first century has been experiencing a global spread of air pollution with nearly all remote locations, such as the mountain peaks, oceanic regions and free troposphere being affected by increasing amount of several pollutants (e.g. Lelieveld et al. 2001; Seinfeld and Pandis 2006; Sarangi et al. 2014; Monks et al. 2015; Girach et al. 2017, 2020; Ojha et al. 2019). In this scenario, air pollution issues have no longer been confined to highly urbanized and industrialized regions and profound impacts are being experienced in the rural and remote environments (e.g. Ravishankara et al. 2020).

The greatest health risks are predominantly linked with the fine particulate matter ($PM_{2.5}$; particulate matter with aerodynamic diameter $\leq 2.5 \ \mu\text{m}$) affecting the health of millions of people across the globe (Chuang et al. 2007; Zanobetti and Schwartz 2009; Kim et al. 2015). Additionally, ozone (O₃), carbon monoxide (CO), nitrogen oxides (NO_x), sulphur dioxide (SO₂), volatile organic compounds (VOCs), etc., deteriorate the air quality (Forster et al. 2007; Monks et al. 2009; Jacobson 2002; Sahu and Saxena 2015). United Nations' World Health Organization (WHO) and the World Meteorological Organization (WMO), in particular, provide valuable data sets on global air quality, health impacts and meteorological variations. Regional agencies, such as the United States Environmental Protection Agency (US EPA), European Union (EU) and India's Central Pollution Control Board (CPCB) also

Pollutant	Major sources	Standard level
PM _{2.5}	Industrial, vehicular and urban emissions; combustion of fossil and biomass fuels; soil/ mineral dust; sea salt from oceans; living organisms	10 μg·m ⁻³ (annual mean); 25 μg·m ⁻³ (24 h mean)
O ₃	Photochemistry involving NO _x , CO and non-methane hydrocarbons (NMHCs)	$100 \ \mu g \cdot m^{-3}$ (8 h mean)
СО	Incomplete combustion of fossil fuels; biomass burning; oxidation of hydrocarbons	10 mg·m ⁻³ (8 h mean); 7 mg·m ⁻³ (24 h mean)
NO _x	Fossil fuel combustion; biomass burning; lightning; denitrification in soils and plants	40 μ g·m ⁻³ (annual mean); 200 μ g·m ⁻³ (1 h mean)
SO ₂	Fossil fuel combustion; volcanic emissions; mineral ore processing; oxidation of DMS; power generation, motor vehicles	20 μ g·m ⁻³ (24 h mean); 500 μ g·m ⁻³ (10 min mean)
VOCs	Biogenic; gasoline combustion; solvents; building materials; biomass burning; fossil fuel combustion	Benzene: $1 \ \mu g \cdot m^{-3}$; formaldehyde: 0.1 mg·m ⁻³ (30 min mean)

Table 12.1 List of key air pollutants, major sources and standard levels

support and carry out observations from network of monitoring stations, which are eventually disseminated to the general public in terms of Air Quality Index (AQI). Additionally, various research institutions and universities also perform measurements and modelling of the local, regional and global air quality. The objectives of air quality monitoring are mainly to observe and report whether the concentrations of pollutants are below (or above) the standards recommended by the WHO, EPA and local governmental bodies. A list of major air pollutants, emission sources and standard levels, generally based on WHO guidelines - global update 2005, is summarized in Table 12.1.

The concentrations of air pollutants are highly variable in space and time due to the effects of varying emissions and the meteorological conditions (see Sect. 12.4 on details of various mechanisms). Air pollutants are emitted from diverse natural and anthropogenic sources. Major natural sources include forest fires, volcanic eruptions, dust storms and biogenic emissions (Luterbacher et al. 2004; Jacobson 2002; Mather et al. 2012), whereas the major anthropogenic sources include the emissions from the industries, power plants, vehicular exhausts, open burning of crop residues, etc. (Seinfeld 1986; Rhind 2009; Lippmann et al. 2015; Ivanova 2020).

Figure 12.1 shows the global distribution of $PM_{2.5}$ (µgm⁻³) during May and January months representative of contrasting seasons of the year 2015, based on the Copernicus Atmosphere Monitoring Service (CAMS) reanalysis (Inness et al. 2019). PM_{2.5} levels show hotspots over South Asia, East Asia, the Middle-East, Southeast Asia and South Africa regions. PM_{2.5} levels are seen to be significantly higher than the prescribed WHO standards suggesting severe health implications over these hotspot regions. As discussed in Sect. 12.3 also, the IGP and NCP regions are global hotspots of PM_{2.5} in particular during the winter conditions (Fig. 12.1).

Global distributions of ground-level NO_2 and O_3 derived from the CAMS reanalysis are shown in Fig. 12.2 for 2 months, May and January, representing



Fig. 12.1 Global distribution of $PM_{2.5} (\mu gm^{-3})$ during May and January months of the year 2015, based on the CAMS reanalysis



Fig. 12.2 Global distribution of ground-level nitrogen dioxide (NO_2) and ozone (O_3) during May and January months of 2015, based on the CAMS reanalysis

contrasting seasons of the year 2015. NO_2 shows higher levels (16 ppbv and higher) generally near the source locations most prominently over the IGP region, eastern China and North America. Due to shorter lifetime of NO_2 , elevated levels seem to be more near emission source regions. Additionally, the NO_2 build-up is particularly strong during winter conditions due to lesser ventilation. In contrast with NO_2 , elevated levels of O_3 are not confined only to the source regions of precursor

gases. Interestingly, widespread O_3 enhancements are also seen over the remote mountains and oceanic regions due to its transport and chemical production in the urban outflows (e.g. Lawrence and Lelieveld 2010; Girach et al. 2017, 2020). O_3 pollution over continental regions is seen to be most severe during dry spring– summer conditions (Fig. 12.2), whereas PM_{2.5} and primary pollutants are strongly enhanced by stagnant conditions during winter (Fig. 12.1).

12.3 Major Historical Episodes and Impacts

Table 12.2 lists some of the most severe air pollution episodes experienced in different parts of the world through diverse mechanisms. In the late nineteenth and early twentieth century, major air pollution episodes were associated with the combustion in coal and chemical factories. The mixture of smoke and fog has been termed as the smog (smoke + fog). The severity of air pollution episodes depended very strongly on the prevailing meteorological conditions. Here, we describe several major air pollution episodes from history and their impacts.

A severe air pollution episode occurred in the Meuse Valley, Belgium, during December 1930. The episode, which caused more than 60 deaths within 2 days, was a result of strong industrial emissions being trapped in the valley through stagnant winter conditions. Thousands of illnesses related to respiratory problems were reported among people, besides impact on cattle. The episode provided scientific evidences strongly linking the air pollution with severe illness and premature deaths (Nemery et al. 2001). Donora, a mill town in Pennsylvania, experienced a severe pollution episode during October 1948. The episode resulted in the deaths of 20 people immediately and about 50 people in the following days, besides causing

Episode	Period	Remarks
Meuse valley episode	December 1930	Caused by stagnant conditions and industrial emissions; more than 60 people died within 2 days
Donora episode	October 1948	Caused by industrial emissions and temperature inversion; 20 people died and thousands suffered respiratory problems
London smog or the Great smog of 1952	1950s	Caused by industrial emissions in prevalence of cold and low wind conditions; linked with deaths of up to 120,000 people
Los Angeles smog	1940s	Summer smog associated with photochemical ozone formation involving vehicular and industrial emissions
Bhopal gas tragedy	1984	Leakage of industrial methyl isocyanate; considered among worst industrial disasters; thousands of lives lost and about half a million suffered health problems
World Trade Center	2001	Large amount of dust emissions due to collapse of the twin towers
Icelandic Volcano Eyjafjallajökull	April– May 2010	Caused by volcanic eruption; severely impacted the civil aviation

Table 12.2 A list of few major air pollution episodes developed through diverse mechanisms

respiratory problems to about 7000 residents. The episode was driven by strong emissions of sulphur dioxide and hydrogen fluoride in Donora, which were trapped by meteorological inversion into a shallow volume near the surface.

The smog episodes prevalent in London were described as the London-type smog. The episode was most severe on 5–9 December 1952 termed as the Great Smog of London with peak levels of SO_2 and particulate matter recorded as 1.4 ppmv and 4460 μ gm⁻³, respectively. This episode was a result of strong coal burning emissions and prevalence of very low atmospheric ventilation. Visibility was drastically reduced to a few metres, and about 4000 people died within first 4 days due to health issues caused by this episode. Several additional premature mortalities have also been reported subsequently and the great London smog strongly pushed for detailed studies and control measures (Bell and Davis 2001; Brimblecombe 2017).

In the early twentieth century, there was a tremendous growth in the vehicular emissions and industries, which led to the development of severe photochemical smog (O_3 pollution) over Los Angeles, California, USA. In 1943, a thick fog severely impacted the health of people with their eyes sting and running nose. This episode made the "smog" word more common with Los Angeles being referred to as the "smogtown" (Jacobs and Kelly 2008). To differentiate, the photochemical smog is also called as the summer smog as these are most severe in the atmospheric conditions of higher temperature and intense solar radiation. The detailed mechanism producing ozone was first unravelled by Haagen-Smit (1952). Ozone pollution has been shown to have considerable health effects such as eye irritation, nose/throat irritation and respiratory problems, and headache.

The Bhopal gas tragedy in India (2–3 December 1984) is widely considered to be the worst chemical disaster in history (Sriramachari 2004). The disaster was an accidental leakage of industrial methyl isocyanate (CH₃NCO) during winter 1984 causing deaths of more than 2000 within the first few days. In addition, about half a million people suffered severe health issues ranging from eye and throat irritation, cough, conjunctivitis, drowsiness and cardiorespiratory problems. Destruction of the World Trade Center (2001) towers added an unprecedented amount of highly alkaline coarse dust in the air over Manhattan causing severe acute and chronic health effects (Lippmann et al. 2015). Ash from the Icelandic Volcano Eyjafjallajökull severely disrupted civil aviation during April–May 2010 stranding millions of passengers (Alexander 2013; Langmann 2012). During Amazonian deforestation of 2019, 4966 premature deaths were attributable to fire emissions making up 10% of all PM_{2.5}-related premature deaths in Brazil (Nawaz and Henze 2020).

Besides the aforementioned events of severe air pollution, there have been recurring haze episodes in some parts of the East and South Asian regions. The persistent and widespread haze has been observed over the densely populated IGP and over the Himalayan foothills, typically every post-monsoon and winter (e.g. Ramanathan and Ramana 2005; Ojha et al. 2020). Severe haze episodes have also been experienced in China, in particular over the NCP during stagnant meteorological conditions and high relative humidity (An et al. 2019; Fowler et al. 2020).

The combined effect of elevated emissions (biomass burning over north-west India and local emissions) and stagnant weather conditions led to a severe air pollution episode (1–7 November 2016) over New Delhi, India (Kanawade et al. 2020). $PM_{2.5}$, CO and Air Quality Index (AQI) were higher than 500 µg m⁻³, up to 10 mg m⁻³ (~8700 ppbv), and 999, respectively, during 11–14 November 2017 over IGP, including the cities of Delhi, Agra, Kanpur, Lucknow, Patna and Kolkata (Dekker et al. 2019). Unfavourable meteorology such as low wind speeds and shallow atmospheric boundary layers have been found to be a key driver of the extreme pollution episode than the biomass burning over north-west India (Dekker et al. 2019). As per the estimation by David et al. (2019), about 60% of premature mortalities due to $PM_{2.5}$ pollution in India are linked with emissions from the same region, whereas transport from other regions within India and from other countries accounts for 19% and 16%, respectively.

Thick smog blanketed Beijing and northern China affecting more than 600 million people (Sheehan et al. 2014) during January 2013. The widespread extreme haze pollution in northern China caused ~3 km thick haze layers, high UV Aerosol Index >2.5, aerosol optical depth (AOD) > 2.0 and peak value of PM_{10} around 1000 mg·m⁻³ (Tao et al. 2014). Unusual heavy pollution due to intense regional transport became extremely severe under anomalous weather conditions (Tao et al. 2014). Average $PM_{2.5}$ levels for 2013 were more than five times the WHO annual maximum level in 58 Chinese cities (Sheehan et al. 2014).

Besides the high background $PM_{2.5}$ and other air pollutants over annual cycle, sudden spikes during haze events can cause irreversible damage to organ systems, in particular to the cardiovascular system (Awasthi et al. 2010; Gupta 2019). In addition to the winter haze in India and China, the exposures to mineral dust during episodic dust outbreaks have been reported to cause acute silicosis and other respiratory ailments (Derbyshire 2007). The annual premature mortality in India attributed to exposure to $PM_{2.5}$ exceeds one million (Cohen et al. 2017). The projected $PM_{2.5}$ -induced premature mortality burden over India using various RCP (representative concentration pathway) scenarios suggests that the health impacts would remain substantial in the future and that these can be aggravated by the climate change (Chowdhury et al. 2018). While the exact mechanisms of the health burden of dust outbreaks are not very clear, nevertheless, the co-emission of other pollutants, coating and transport of other pollutants with dust particles and the associated biological load are suggested to be important factors.

12.4 Important Mechanisms and Challenges

As discussed in the previous sections, diverse mechanisms have been involved in causing severe air pollution episodes around the world, making it very challenging to deal with at the time of occurrence and to design and implement control measures for future. Many episodes were results of sporadic large emissions of one or more



Fig. 12.3 A summary of various factors that could be important in the development and evolution of air pollution episodes

pollutants enhancing their concentrations to levels much higher than the standards prescribed in Table 12.1. Besides strong emissions, prevailing atmospheric conditions also played very crucial roles in favouring photochemistry to form secondary pollutants (O₃, secondary aerosols) and/or in controlling the dispersion and ventilation of the pollution. A summary of various factors that could be important in the development and evolution of air pollution episodes is compiled in Fig. 12.3. Key driver of individual major episodes are described briefly in Sect. 12.3 and in the literature (Brimblecombe 2017 and references therein).

It has been discussed in cases of individual major air pollution episodes in Sect. 12.3 that extremely high pollution concentrations have often evolved through unusual combination of strong emissions and stagnant meteorological conditions. Several of these episodes took place over different regions of the world over different periods of time in the early twentieth century. In contrast, recurring seasonal pollution episodes are experienced over polluted regions such as in the East Asia and over the IGP region in South Asia. Stronger anthropogenic emissions can get trapped into shallower boundary layers during winters leading to enhanced pollution, which can further lead to widespread haze over these regions (Kumar et al. 2015, 2017; Ojha et al. 2020). The transport of pollution from upwind regions can further exacerbate these episodes. The post-monsoon time high PM_{2.5} is attributed to the transport of crop residue burning over upwind agricultural regions (Sen et al. 2017).



Fig. 12.4 Boundary layer height (m) over South and East Asian regions during summer (May) and winter (January) conditions for the year 2015, derived from the ERA5 reanalysis

The stagnant atmospheric conditions play a key role in confining anthropogenic emissions near the surface during winter over the IGP (Ojha et al. 2020). Variations in the meteorological conditions played pivotal role in more atmospheric ventilation reducing near-surface pollution during the Asia-Pacific Economic Cooperation (APEC) (Ansari et al. 2019). Figure 12.4 shows the variations in the boundary layer height over South and East Asian regions during summer (May) and winter (January) conditions as derived from the ERA5 meteorological reanalysis for the year 2015. Boundary layer height over northern and central Indian regions decreases drastically from summer (800–1600 m) to winter (200–400 m) reducing the vertical mixing and therefore the dilution of pollution. Similar changes are also evident over the eastern Asian region. Recent studies have reported an additional role of complex chemical processes in the persistence of haze episodes. Increased humidity can substantially change the chemical properties of aerosols through these complex processes, which can drastically reduce the visibility in high aerosol mass loading scenarios (Cheng et al. 2016; Gunthe et al. 2020). An example of early morning haze versus clear sky conditions over a tropical megacity, Delhi, India, is shown in Fig. 12.5 (adapted from Dhaka et al. 2020).

In complete contrast to winter haze, dry springs and summers experience intense photochemistry involving NOx, CO and VOCs to produce O_3 in urban and rural environments, called as the photochemical smog. Depending upon precursor concentrations (NO_x and VOCs) and solar radiation, O_3 production takes place through complex and non-linear chemistry (Seinfeld and Pandis 2006). O_3 pollution is especially stronger during dry summer conditions, as shown in Fig. 12.2 due to intense photochemistry. NO participates in O_3 chemical destruction also, and therefore, net O_3 concentrations can be higher in the downwind regions than those over the emission sources of fossil fuel and biomass burning. Higher fossil fuel burning



Fig. 12.5 An example of (**a**) early morning haze and (**b**) clear sky condition over a tropical megacity, Delhi, India (adapted from Dhaka et al. 2020 available at https://www.nature.com/articles/s41598-020-70179-8 as per https://creativecommons.org/licenses/by/4.0/)

over China causes high O_3 events exceeding 200 µg m⁻³ more frequently mainly during the warm season as compared to several other industrialized regions in the world (Lu et al. 2018). Higher temperature during summer months fuels the chemistry and can also enhance biogenic emissions of VOCs resulting in higher O_3 concentrations (Coates et al. 2016). Biomass burning was shown to enhance O₃ levels above the air quality standards during spring and autumn seasons over northern Indian region (Kumar et al. 2016; Ojha et al. 2019). In addition to contributions from fossil fuel and biomass burning, fertilizer-induced NO_x emissions from soil also contribute to O_3 chemistry, especially over the agricultural regions (Chameides et al. 1994). Nitrous acid (HONO), a reservoir of NOx and HOx mainly formed during night, injects a pulse with sunrise (Seinfeld and Pandis 2006). Recently, new sources of daytime HONO are revealed (Kleffmann 2007), which could influence O₃. Similarly, the gas-phase reactions between Cl and VOCs can also increase the production of O_3 (Riedel et al. 2014). In a nutshell, photochemistry intertwined with the meteorology and concentration of precursor gases drives the O₃ pollution. Most of the studies dealt with gas-phase chemistry, but the role of heterogeneous chemistry (reactions on the surface of aerosols, for instance, during thick fog events) also needs to be investigated more in future.

During the El Niño conditions, sea surface temperature is warmer than usual in the Eastern Pacific. This causes changes in the precipitation pattern over the tropical belt. Precipitation reduces and Southeast Asia (especially, maritime continent) and Australia experience drought conditions. In effect, prolonged drier and hotter conditions prevail over Indonesia and Australia. These conditions are conducive for triggering fires over forest regions. Extreme forest fire events have been observed over Indonesia and Australia during El Niño years (e.g. 1997, 2006). Millions of people suffered the effects of air pollution resulting from the forest fires of 1997–1998 over Indonesia. Neighbouring countries (Malaysia and Singapore) experienced thick clouds of smoke and haze. Thailand, Vietnam and the Philippines also felt haze. This was due to extremely dry conditions over forest under the drought influences linked with El Niño (Glover and David 2006).

It is clear that air pollution episodes are results from a variety of man-made and natural factors. Besides regulations on the industrial emissions and efforts to avoid accidental leakages, significant efforts are needed to monitor the air quality over global and regional scales. Contributions of anthropogenic and biomass burning emissions have still been uncertain due to uncertainties in the emission inventories and limited abilities of models to reproduce the measurements. Ozone pollution is anticipated to be stronger in the future because climate warming can offset the potential benefits of emission reductions (Coates et al. 2016). Occurrences of heatwaves and concurrent high ozone pollution can lead to more severe health effects (Stedman 2004). Controlling the biomass burning emissions from both natural forest fires and man-made such as crop residue burning would involve consideration also of socio-economic aspects of a particular region. In view of these, it remains a major challenge to formulate the policies to minimize occurrences and impacts of pollution episodes by taking unfavourable meteorology and climate change also into account for more effective control measures.

12.5 Some Perspectives on Control Measures and Outlook

Numerous acts and regulations have been brought in worldwide to minimize the occurrences and impacts of the air pollution episodes (e.g. Jacobson 2002; Brimblecombe 2017). Here, we discuss some current perspectives and outlook on control measures of pollution episodes. Measures to control air pollution involve a diversified approach requiring the source and process-oriented steps and considering the socio-economic, scientific and cost-benefit aspects. The conventional control systems rely on source-based pollution control strategies, e.g. control of emissions from point and mobile sources or the efficiency improvement of processes (e.g. combustion efficiency) to reduce the emissions (Saxena and Sonwani 2019). Air pollution from large-scale industries, vehicular exhausts and indoor air pollution sources have been a prime focus of air pollution control and management, whereas relatively less emphasis has been on the impacts of natural phenomena (forest fires, volcanoes, secondary formation of aerosol particles, etc.). The important air pollution episodes, which require special attention, are season-specific haze and smog, dust outbreaks and occasional leakage of toxic gases from industrial units. Localized observation and high-resolution modelling studies have suggested a strong interplay of both source strengths and meteorological phenomena in the development of regional haze (e.g. over northern India, Indonesia and East China). Sudden spikes in the concentration of PM2.5, CO, SO2, NO2, NH3, etc., attributed to the burning of crop residue or forest biomass, lead to the formation of haze and smog. The control of these types of pollution events requires important policy interventions and effective management via alternate use of residual biomass or temporal adjustments in crop harvesting periods. The provincial governments in the northern Indian states of Punjab and Haryana have enacted laws against the open burning of crop residue and also provided subsidies for not burning as a preventive measure to reduce haze during post-monsoon. A similar inter-governmental framework has been designed among the countries in the Southeast Asian region. Dust outbreak is another such air pollution episode that requires more preventive measures rather than its direct control due to its natural origin. Although it is difficult to control the long-range transport of dust, nevertheless, its resuspension can be significantly checked with suitable landscape management policies. The development of green belts in residential and commercial areas would help in reducing ambient dust through different processes such as filtration, interception and absorption on plants. The green belts are reported to be effective in gaseous pollution tolerance (Sikarwar et al. 1999; Saxena et al. 2010, 2011). The mandatory developments of green belts for every commercial, residential project of larger areas and industrial infrastructural construction can help in controlling the resuspension of dust transported from distant sources. Chaulya (2004) and Garg et al. (2015) provides recommendation on several plant species that can be used for green belt development to reduce ambient air pollution.

It may, however, be challenging to develop green belts at locations with a large impervious area like cities and other urban settings. In such conditions, green roofing over buildings has significant potentials of intercepting dust and other pollutants. Deutsch et al. (2005) estimated the removal of 58 metric tons of pollutants including aerosols with 100% green roofing. Similar results were observed by Currie and Bass (2005) in Toronto with the removal of 7.9 metric tons of air pollutants annually with 109 ha of green roofs in the city. The implementation of such practices should be emphasized in regions like India and China experiencing rapid urban sprawl and land-use changes. The implementation of the green roofing concepts should be stressed upon through acts and legislations to reduce ambient air pollution. Occasional leakage of toxic gases and other pollutants from industries can also be more controlled through firm legislations and protocols. The key protocols that ensure the control of occasional accidents from leakage include periodic leak detection inspections, maintenance of records of previous leaks and setting up future stringent safety goals. These measures would ensure the prevention of unanticipated pollution episodes. The remote detection of leakage through portable measurement systems can help in preparation for any such accident.

It is of vital significance to depend more on renewable energies and make all possible efforts to minimize fossil fuel burning in the future. Lelieveld et al. (2019) argued that this would have a remarkable effect on the global air quality and save millions of lives, besides helping to limit the global warming. In addition to various control measures, it is also crucial to improve our understanding of the chances of occurrence of air pollution episodes in the future. More efforts with monitoring and modelling of air quality would help in availing better forecast (Kumar et al. 2018)

and hence better preparedness. Satellite-based measurements are being used to successfully retrieve the distribution of $PM_{2.5}$ over large geographical regions (Dey et al. 2020), which would especially be useful over regions with sparse observations and strong spatial heterogeneity. Assimilation of satellite-based measurements is shown to significantly improve the skills of model forecast systems in predicting the short-term air quality (Kumar et al. 2020). Further improvements in the emission estimates, improved meteorology, and chemistry schemes would be valuable in better understanding the air pollution drivers over local to regional scales and designing the control measures.

12.6 Summary

Historically, the air pollution episodes have been driven by unusual combinations of emission with meteorological condition. For example, the Great Smog of London (1952) was driven by coal burning emission and limited atmospheric ventilation in the winter, whereas the photochemical smog over Los Angeles (1943) was due to intense photochemistry involving vehicular and industrial emissions during summer. Additional major episodes were caused by sudden emissions from accidental leak in industry, volcanic eruptions and forest fires. The health impacts of pollution during such episodes were unprecedented and therefore initiated the development and implementation of policies for prevention and control.

The observational and modelling studies unravelled a variety of mechanisms causing pollution episodes, besides providing the recommendations for formulation of effective control strategies. In this line, numerous national and international efforts have been initiated for providing data sets on air pollution and deriving health impacts over local, regional and global scales. Nevertheless, pollution episodes are being experienced both over major cities and in the form of widespread haze across larger geographical areas (e.g. NCP, IGP). Both the NCP and IGP regions have been identified as the global hotspots of elevated PM2.5 during winter conditions. In contrast, the photochemical smog is experienced to be most severe during the dry spring-summer conditions associated with high ozone levels, which also spreads over rural and remote regions. Besides the meteorological conditions and anthropogenic emissions, air pollution is also found to be exacerbated by intense forest fires and crop residue burning. Recently, regulations have been placed to curb emissions from anthropogenic sources and crop residue burning, besides stringent safety measures to prevent accidental release of toxic pollutants from industries and chemical plants. In addition, monitoring of air quality using ground-based network is also expanded to cover the larger geographical areas complimented by the satellitebased instruments.

Despite significant progress on monitoring, modelling and implementation of policies, the intensity and frequency of extreme pollution are on the rise. More studies are therefore needed to fill the data gaps and to validate the satellite retrievals and model simulations, especially over developing regions. Minimizing the usage of fossil fuel is suggested to help both in reducing pollution and in mitigating climate warming. In future, natural emissions may also rise due to climate warming, which could offset some benefits of reduction in the anthropogenic emissions. Hence, further studies on pollution impacts are needed, which also account for the role of climate variabilities leading to frequent pollution episodes. In the end, assimilation of satellite and ground data with models for accurate forecasting of pollution episodes would help towards better preparedness and to minimize health impacts.

Acknowledgements The use of ERA5 and CAMS reanalysis data sets from the ECMWF, UK, and Ferret software (http://ferret.pmel.noaa.gov/Ferret/) from NOAA, USA, are highly acknowledged.

References

- Alexander D (2013) Volcanic ash in the atmosphere and risks for civil aviation: a study in European crisis management. Int J Disaster Risk Sci 4:9–19. https://doi.org/10.1007/s13753-013-0003-0
- An Z, Huang RJ, Zhang R, Tie X, Li G, Cao J, Zhou W, Shi Z, Han Y, Gu Z, Ji Y (2019) Severe haze in northern China: A synergy of anthropogenic emissions and atmospheric processes. Proc Natl Acad Sci 116(18):8657–8666. https://doi.org/10.1073/pnas.1900125116
- Ansari TU, Wild O, Li J, Yang T, Xu W, Sun Y, Wang Z (2019) Effectiveness of short-term air quality emission controls: a high-resolution model study of Beijing during the Asia-Pacific Economic Cooperation (APEC) summit period. Atmos Chem Phys 19:8651–8668. https://doi. org/10.5194/acp-19-8651-2019
- Awasthi A, Singh N, Mittal S, Gupta PK, Agarwal R (2010) Effects of agriculture crop residue burning on children and young on PFTs in North West India. Sci Total Environ 408(20): 4440–4445
- Bell ML, Davis DL (2001) Reassessment of the lethal London fog of 1952: novel indicators of acute and chronic consequences of acute exposure to air pollution. Environ Health Perspect 109:389– 394
- Brimblecombe P (2017) Air pollution episodes. World Scientific (Europe). https://doi.org/10.1142/ q0098. https://www.worldscientific.com/doi/pdf/10.1142/q0098
- Chameides WL, Kasibhatla PS, Yienger J, Levy H (1994) Growth of continental-scale metro-agroplexes, regional ozone pollution, and world food production. Science 264(5155):74–77. https:// doi.org/10.1126/science.264.5155.74
- Chaulya SK (2004) Assessment and management of air quality for an opencast coal mining area. J Environ Manag 70(1):1–14
- Cheng Y et al (2016) Reactive nitrogen chemistry in aerosol water as a source of sulfate during haze events in China. Sci Adv 2(12):e1601530. https://doi.org/10.1126/sciadv.1601530
- Chowdhury S, Dey S, Smith KR (2018) Ambient PM 2.5 exposure and expected premature mortality to 2100 in India under climate change scenarios. Nat Commun 9(1):1–10
- Chuang K, Chan C, Su T, Lee C, Tang C (2007) The effect of urban air pollution on inflammation, oxidative stress, coagulation, and autonomic dysfunction in young adults. Am J Respir Crit Care Med 176(2):370–376. https://doi.org/10.1164/rccm.200611-1627OC
- Coates J, Mar KA, Ojha N, Butler TM (2016) The influence of temperature on ozone production under varying NOx conditions—a modelling study. Atmos Chem Phys 16(18):11601–11615. https://doi.org/10.5194/acp-16-11601-2016
- Cohen AJ, Brauer M, Burnett R, Anderson HR, Frostad J, Estep K, Balakrishnan K, Brunekreef B, Dandona L, Dandona R, Feigin V, Freedman G, Hubbell B, Jobling A, Kan H, Knibbs L, Liu Y, Martin R, Morawska L et al (2017) Estimates and 25-year trends of the global burden of disease attributable to ambient air pollution: an analysis of data from the Global Burden of Diseases Study 2015. Lancet 389(10082):1907–1918. https://doi.org/10.1016/S0140-6736(17)30505-6

- Currie BA, Bass B (2005) Estimates of air pollution mitigation with green plants and green roofs using the UFORE model. In: Proceedings of the Third North American Green Roof Conference: Greening Rooftops for Sustainable Communities, Washington, DC, The Cardinal Group, Toronto, 4–6 May, pp 495–511
- David LM, Ravishankara AR, Kodros JK, Pierce JR, Venkataraman C, Sadavarte P (2019) Premature mortality due to PM2.5 over India: effect of atmospheric transport and anthropogenic emissions. GeoHealth 3(1):2–10. https://doi.org/10.1029/2018GH000169
- Dekker IN, Houweling S, Pandey S, Krol M, Röckmann T, Borsdorff T, Landgraf J, Aben I (2019) What caused the extreme CO concentrations during the 2017 high-pollution episode in India? Atmos Chem Phys 19:3433–3445. https://doi.org/10.5194/acp-19-3433-2019
- Derbyshire E (2007) Natural minerogenic dust and human health. Ambio 36(1):73-77
- Deutsch B, Whitlow H, Sullivan M, Savineau A (2005) Re-greening Washington, DC: a green roof vision based on quantifying storm water and air quality benefits. Accessed Sept 2020
- Dey S, Purohit B, Balyan P, Dixit K, Bali K, Kumar A, Imam F, Chowdhury S, Ganguly D, Gargava P, Shukla VK (2020) A satellite–based high-resolution (1-km) ambient PM_{2.5} database for India over two decades (2000–2019): applications for air quality management. Remote Sens 12:3872
- Dhaka SK, Chetna, Kumar V et al (2020) PM2.5 diminution and haze events over Delhi during the COVID-19 lockdown period: an interplay between the baseline pollution and meteorology. Sci Rep 10:13442. https://doi.org/10.1038/s41598-020-70179-8
- Forster P, Ramaswamy V, Artaxo P, Berntsen T, Betts R, Fahey DW, Haywood J, Lean J, Lowe DC, Myhre G, Nganga J, Prinn R, Raga G, Schulz M, Van Dorland R, Miller HL (eds) (2007) Changes in atmospheric constituents and in radiative forcing, Chapter 2. Cambridge University Press, Cambridge
- Fowler et al (2020) A chronology of global air quality. https://doi.org/10.1098/rsta.2019.0314
- Garg A, Saxena P, Ghosh C (2015) Evaluation of tolerance and sensitivity of selected plant species with special reference to gasoline exhaust pollution. Int J Sci Technol Res 4(2):199–207
- Girach IA, Ojha N, Nair PR, Pozzer A, Tiwari YK, Kumar KR, Lelieveld J (2017) Variations in O3, CO, and CH4 over the Bay of Bengal during the summer monsoon season: shipborne measurements and model simulations. Atmos Chem Phys 17:257–275. https://doi.org/10. 5194/acp-17-257-2017
- Girach IA, Tripathi N, Nair PR, Sahu LK, Ojha N (2020) O3 and CO in the South Asian outflow over the Bay of Bengal: impact of monsoonal dynamics and chemistry. Atmos Environ 233: 117610. https://doi.org/10.1016/j.atmosenv.2020.117610
- Glover, David (2006) Indonesia's fires and haze: the cost of catastrophe. IDRC. ISBN 9781552503324
- Gunthe SS et al (2020) Enhanced aerosol particle growth sustained by high continental chlorine emission in India. Nat Geosci 14:77–84
- Gupta S (2019) Exposure assessment of pulmonary functions of school going children due to agriculture crop residue burning practice in Indo-Gangetic plains of India. Hum Ecol Risk Assess Int J 26(3):843–858
- Haagen-Smit AJ (1952) Chemistry and physiology of Los Angeles smog. Ind Eng Chem 44(6): 1342–1346. https://doi.org/10.1021/ie50510a045
- Inness A, Ades M, Agustí-Panareda A, Barré J, Benedictow A, Blechschmidt A-M, Dominguez JJ, Engelen R, Eskes H, Flemming J, Huijnen V, Jones L, Kipling Z, Massart S, Parrington M, Peuch V-H, Razinger M, Remy S, Schulz M, Suttie M (2019) The CAMS reanalysis of atmospheric composition. Atmos Chem Phys 19:3515–3556. https://doi.org/10.5194/acp-19-3515-2019
- Intergovernmental Panel on Climate Change (IPCC) (2007) Climate change 2007: the physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge and New York

- Intergovernmental Panel on Climate Change (IPCC) (2013) Climate change 2013: the physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge New York, 1535 p. https://doi.org/10.1017/CBO9781107415324
- Ivanova VR (2020) The anthropogenic air pollution and human health. J IMAB 26(2):3057–3062. https://doi.org/10.5272/jimab.2020262.3057
- Jacobs C, Kelly WJ (2008) Smogtown: the lung-burning history of pollution in Los Angeles. Overlook Press, Woodstock
- Jacobson M (2002) Atmospheric pollution: history, science, and regulation. Cambridge University Press, Cambridge. https://doi.org/10.1017/CBO9780511802287
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Kanawade VP, Srivastava AK, Ram K, Asmi E, Vakkari V, Soni VK, Varaprasad V, Sarangi C (2020) What caused severe air pollution episode of November 2016 in New Delhi? Atmos Environ 222. https://doi.org/10.1016/j.atmosenv.2019.117125
- Kim K, Kabir E, Kabir S (2015) A review on the human health impact of airborne particulate matter. Environ Int 74:136–143. https://doi.org/10.1016/j.envint.2014.10.005
- Kleffmann J (2007) Daytime sources of nitrous acid (HONO) in the atmospheric boundary layer. Chem Phys Chem 8(8):1137–1144. https://doi.org/10.1002/cphc.200700016
- Kumar M, Tiwari S, Murari V, Singh AK, Banerjee T (2015) Wintertime characteristics of aerosols at middle Indo-Gangetic Plain: impacts of regional meteorology and long range transport. Atmos Environ 104:162–175
- Kumar V, Sarkar C, Sinha V (2016) Influence of post-harvest crop residue fires on surface ozone mixing ratios in the N.W. IGP analyzed using 2 years of continuous in situ trace gas measurements. J Geophys Res Atmos 121(7):3619–3633. https://doi.org/10.1002/ 2015JD024308
- Kumar M, Raju MP, Singh RK, Singh AK, Singh RS, Banerjee T (2017) Wintertime characteristics of aerosols over middle Indo-Gangetic Plain: vertical profile, transport and radiative forcing. Atmos Res 183:268–282
- Kumar R, Peuch V-H, Crawford JH, Brasseur G (2018) Five steps to improve air-quality forecasts. Nature 561:27–29. https://doi.org/10.1038/d41586-018-06150-5
- Kumar R, Ghude SD, Biswas M, Jena C, Alessandrini S, Debnath S et al (2020) Enhancing accuracy of air quality and temperature forecasts during paddy crop residue burning season in Delhi via chemical data assimilation. J Geophys Res Atmos 125:e2020JD033019. https://doi. org/10.1029/2020JD033019
- Langmann B, Folch A, Hensch M, Matthias V (2012) Volcanic ash over Europe during the eruption of Eyjafjallajökull on Iceland, April–May 2010. Atmos Environ 48:1–8. https://doi.org/10. 1016/j.atmosenv.2011.03.054
- Lawrence MG, Lelieveld J (2010) Atmospheric pollutant outflow from southern Asia: a review. Atmos Chem Phys 10:11017–11096. https://doi.org/10.5194/acp-10-11017-2010
- Lelieveld J et al (2001) The Indian Ocean experiment: widespread air pollution from South and Southeast Asia. Science 291(5506):1031–1036. https://doi.org/10.1126/science.1057103
- Lelieveld J, Klingmüller K, Pozzer A, Burnett RT, Haines A, Ramanathan V (2019) Effects of fossil fuel and total anthropogenic emission removal on public health and climate. Proc Natl Acad Sci 116(15):7192–7197. https://doi.org/10.1073/pnas.1819989116
- Lippmann M, Cohen MD, Chen LC (2015) Health effects of World Trade Center (WTC) Dust: an unprecedented disaster's inadequate risk management. Crit Rev Toxicol 45(6):492–530. https:// doi.org/10.3109/10408444.2015.1044601. PMID: 26058443; PMCID: PMC4686342
- Lu X, Hong J, Zhang L, Cooper OR, Schultz MG, Xu X, Wang T, Gao M, Zhao Y, Zhang Y (2018) Severe surface ozone pollution in China: a global perspective. Environ Sci Technol Lett 5(8): 487–494. https://doi.org/10.1021/acs.estlett.8b0036

- Luterbacher J, Dietrich D, Xoplaki E, Grosjean M, Wanner H (2004) European seasonal and annual temperature variability, trends, and extremes since 1500. Science 303:1499–1503. https://doi. org/10.1126/science.1093877
- Mather TA, Witt MLI, Pyle DM, Quayle BM, Aiuppa A, Bagnato E, Martin RS, Sims KWW, Edmonds M, Sutton AJ, Ilyinskaya E (2012) Halogens and trace metal emissions from the ongoing 2008 summit eruption of Kilauea volcano, Hawai'i. Geochim Cosmochim Acta 83: 292–323. https://doi.org/10.1016/j.gca.2011.11.029. ISSN 0016-7037
- Monks PS, Granier C, Fuzzi S et al (2009) Atmospheric composition change—global and regional air quality. Atmos Environ 43(33):5268–5350. https://doi.org/10.1016/j.atmosenv.2009.08.021. ISSN 1352-2310
- Monks PS, Archibald AT, Colette A, Cooper O, Coyle M, Derwent R, Fowler D, Granier C, Law KS, Mills GE, Stevenson DS, Tarasova O, Thouret V, von Schneidemesser E, Sommariva R, Wild O, Williams ML (2015) Tropospheric ozone and its precursors from the urban to the global scale from air quality to short-lived climate forcer. Atmos Chem Phys 15:8889–8973. https://doi.org/10.5194/acp-15-8889-2015
- Nawaz MO, Henze DK (2020) Premature deaths in Brazil associated with long-term exposure to PM2.5 from Amazon fires between 2016 and 2019. GeoHealth 4:e2020GH000268. https://doi.org/10.1029/2020GH000268
- Nemery B, Hoet PH, Nemmar A (2001) The Meuse Valley fog of 1930: an air pollution disaster. Lancet 357(9257):704–708. https://doi.org/10.1016/S0140-6736(00)04135-0
- Ojha N, Girach I, Sharma K et al (2019) Surface ozone in the Doon Valley of the Himalayan foothills during spring. Environ Sci Pollut Res 26:19155–19170. https://doi.org/10.1007/s11356-019-05085-2
- Ojha N, Sharma A, Kumar M et al (2020) On the widespread enhancement in fine particulate matter across the Indo-Gangetic Plain towards winter. Sci Rep 10:5862. https://doi.org/10.1038/ s41598-020-62710-8
- Ramanathan V, Ramana MV (2005) Persistent, widespread, and strongly absorbing haze over the Himalayan foothills and the Indo-Gangetic Plains. Pure Appl Geophys 162:1609–1626. https:// doi.org/10.1007/s00024-005-2685-8
- Ravishankara AR, David LM, Pierce JR, Venkataraman C (2020) Outdoor air pollution in India is not only an urban problem. PNAS 117(46):28640–28644. https://doi.org/10.1073/pnas. 2007236117
- Rhind SM (2009) Anthropogenic pollutants: a threat to ecosystem sustainability? Philos Trans R Soc Lond Ser B Biol Sci 364(1534):3391–3401. https://doi.org/10.1098/rstb.2009.0122
- Riedel TP, Wolfe GM, Danas KT, Gilman JB, Kuster WC, Bon DM, Vlasenko A, Williams EJ, Lerner BM, Veres PR, Roberts JM, Holloway JS, Lefer B, Brown SS, Thornton JA (2014) An mcm modeling study of nitryl chloride (CINO2) impacts on oxidation, ozone production and nitrogen oxide partitioning in polluted continental outflow. Atmos Chem Phys 14(8): 3789–3800. https://doi.org/10.5194/acp-14-3789-2014
- Sahu LK, Saxena P (2015) High time and mass resolved PTR-TOF-MS measurements of VOCs at an urban site of India during winter: role of anthropogenic, biomass burning, biogenic and photochemical sources. Atmos Res 164:84–94
- Sarangi T, Naja M, Ojha N, Kumar R, Lal S, Venkataramani S, Kumar A, Sagar R, Chandola HC (2014) First simultaneous measurements of ozone, CO, and NOy at a high-altitude regional representative site in the central Himalayas. J Geophys Res Atmos 119:1592–1611. https://doi. org/10.1002/2013JD020631
- Saxena P, Ghosh C (2011) Variation in the concentration of ground level ozone at selected sites in Delhi. Int J Environ Sci 1(7):1899–1911
- Saxena P, Sonwani S (2019) Policy regulations and future recommendations. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 127–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253

- Saxena P, Bhardwaj R, Ghosh C (2010) Plant species recommended for greenbelt development at selected areas in Delhi. In 4th International Conference on Plants & Environmental Pollution organized by NBRI, Lucknow, 8th–11th Dec
- Saxena P, Bhardwaj R, Ghosh C (2011) A sustainable way to control air pollutants with the use of plant species at selected areas in Delhi. Fresenius Environ Bull 20(10):2662–2669
- Seinfeld JH (1986) Atmospheric chemistry and physics of air pollution. Wiley-Interscience, New York
- Seinfeld JH, Pandis SN (2006) Atmospheric chemistry and physics: from air pollution to climate change, 2nd edn. Wiley-Interscience Publication, Hoboken
- Sen A et al (2017) Variations in particulate matter over Indo-Gangetic Plains and Indo-Himalayan Range during four field campaigns in winter monsoon and summer monsoon: role of pollution pathways. Atmos Environ 154:200–224. https://doi.org/10.1016/j.atmosenv.2016.12.054. ISSN 1352-2310
- Sheehan P, Cheng E, English A et al (2014) China's response to the air pollution shock. Nat Clim Chang 4:306–309. https://doi.org/10.1038/nclimate2197
- Sikarwar SS, Guha S, Singh KN (1999) Environmental pollution control through geo-botanical means-a case study of Manikpur block, Korba coalfield, MP, India. Environ Geol 38(3): 229–232
- Sonwani S, Saxena P (eds) (2022) Greenhouse gases: sources, sinks and mitigation. Springer-Nature, Singapore, pp 1–255
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1–2):39–50
- Sriramachari S (2004) The Bhopal Gas Tragedy: an environmental disaster. Curr Sci 86(7):905-920
- Stedman J (2004) The predicted number of air pollution related deaths in the UK during the August 2003 heatwave. Atmos Environ 38:1087–1090. https://doi.org/10.1016/j.atmosenv.2003. 11.011
- Tao M, Chen L, Xiong X, Zhang M, Ma P, Tao J, Wang Z (2014) Formation process of the widespread extreme haze pollution over northern China in January 2013: implications for regional air quality and climate. Atmos Environ 98:417–425. https://doi.org/10.1016/j. atmosenv.2014.09.026. ISSN 1352-2310
- Zanobetti A, Schwartz J (2009) The effect of fine and coarse particulate air pollution on mortality: a national analysis. Environ Health Perspect 117:898–903. https://doi.org/10.1289/ehp.0800108



13

Increasing Atmospheric Extreme Events and Role of Disaster Risk Management: Dimensions and Approaches

Madhavi Jain

Abstract

Under most future emission projection scenarios, global temperature rise is imminent. Rising temperature is closely interlinked with balance in energy and water budget. As per the Clausius–Clapeyron equation, atmospheric moisture rises as 7% per °C rise. This means that both temperature and precipitation extremes such as heatwaves, floods and droughts will be more frequent and intense. Multiple atmospheric extremes events such as heatwaves with low moisture availability could have severe compounding impacts, e.g. increased risk of forest fires, droughts and dust storms. Moreover, global population is increasingly becoming more urban. Most future urban population growth is projected to be in the densely populated cities in Asia and Africa. Temperature and precipitation extremes are often exacerbated due to the presence of dense urban areas and are presented as urban heat islands, CO_2 domes, urban flooding, smog, etc. Cities, in general, show more vulnerability towards atmospheric extremes.

In order to mitigate such disasters, it is crucial to identify vulnerable hotspots and demographic groups most likely to be affected. The chapter outlines various disaster mitigation and adaptation strategies. Importance of science and decision support systems, data availability and sharing, early warning systems, timely information dissemination and post-disaster management are highlighted.

Keywords

Atmospheric extreme events · Temperature · Precipitation · Urbanization · Disaster vulnerability · Mitigation strategies

M. Jain (🖂)

School of Environmental Sciences, Jawaharlal Nehru University, New Delhi, India

 $^{{\}rm \mathbb{C}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_13

13.1 Background

From 1999 to 2018, more than 12,000 extreme events were recorded worldwide (Eckstein et al. 2019). Adapting to and mitigating extreme atmospheric events fuelled by climate change is quite possibly the greatest challenge presented to mankind in the twenty-first century. Not only do such disasters cause tens of thousands of deaths each year, but they also incur huge monetary and gross domestic product (GDP) losses. Nearly, half a million people died as a direct result of these events between 1999 and 2018, and they further coasted countries ~USD 3.5 trillion (Eckstein et al. 2019). Looking closely, the global number of climate-related disasters in the recent times increased from 428 per year (1994–1998) to 707 per year (1999-2003) (Prasad et al. 2008). Moreover, atmospheric extremes pose a danger to all sectors of the society-agriculture and biodiversity, food and water security, health and livelihood, small and large businesses, and culture and tourism (Wernberg et al. 2013; Thornton et al. 2014; Thomas et al. 2019). Such disasters affect both the rich and the poor, although the poor are much more severely impacted (IPCC 2012; Fankhauser and McDermott 2014). Therefore, it becomes imperative to study (1) how 'normal' atmospheric conditions are changing and what it means for the future, (2) what different types of atmospheric extremes exist, (3) which regions and socio-economic groups are more vulnerable to their impacts and finally, and (4) what strategies can be implemented to manage disaster risks and increase resilience. This chapter briefly introduces the reader to all these key concepts. It also focuses on the importance of identifying key vulnerable hotspots and demographic groups that are highly likely to face severe adverse effects of such events.

13.2 Atmospheric Dynamics and Feedbacks

Under normal atmospheric conditions, a balance in the amount and distribution of energy is maintained. Almost all aspects of planetary weather and climate systems are governed by radiation budget balance and the normal functioning of feedback systems, e.g. hydrological cycle (Trenberth et al. 2009). However, since the industrialization era, the earth's air chemistry and atmospheric composition have recorded major changes—most notably in the concentration of carbon dioxide (CO₂). CO₂ atmospheric concentration increased from 280 parts per million (ppm) in 1750 to 410 ppm in 2019 and is projected to reach ~600 ppm by 2050, given current emission trends (Soba et al. 2020). Various anthropogenic practices such as fuel combustion, industrialization, biomass burning, deforestation and land-use changes have inarguably led to significant increase in levels of CO₂, carbon monoxide (CO), methane (CH₄), nitrous oxides (NO_x), tropospheric ozone (O₃), volatile organic compounds (VOCs), chlorofluorocarbons (CFCs) and other trace gases such as sulphur hexafluoride (SF₆) (IPCC 2014; Uprety and Saxena 2021a). Apart from greenhouse gas (GHG) emissions, aerosol column burden is yet another concern. Fine $(PM_{2.5})$ and coarse (PM_{10}) particulate matters (PM_s) are added to the troposphere by natural and anthropogenic sources and get transported via winds to longrange distances (Saxena et al. 2021; Sonwani et al. 2021). Properties such as radiative heating or cooling depend upon aerosol physical and chemical properties, particle residence time, amount and distribution in the atmospheric layers (Quaas et al. 2004; Andrews et al. 2011; IPCC 2014).

The earth's radiation budget and as a consequence the global water budget (hydrological cycle) are greatly affected by such changes in the atmosphere (Trenberth 2011; Brown and Caldeira 2017). Trenberth et al. (2009) provide a detailed schematic of the incoming shortwave radiation from the sun, its distribution, outgoing long-wave radiation emitted from the earth and the net energy stored. The presence of various types of clouds, amount of water vapour, gases and aerosols in the atmosphere, along with albedo and thermal properties of various land covers types, e.g. forest, water bodies and urban areas, determine the amount of radiation that is absorbed, transmitted or reflected (Trenberth et al. 2009). In a globally polluted environment, the atmospheric burden of GHGs such as CO_2 , CH_4 and N_2O increases tremendously. These gases have a high global warming potential (GWP) and global temperature change potential (GTP) (Myhre et al. 2013). Readers are encouraged to go through the extensive data assembled by Myhre et al. (2013) on GWP, GTP and atmospheric lifetimes of major and trace gases found in our environment.

Meanwhile, on the land surface, evapotranspiration process accelerates when the surrounding air has a high sensible heat. In an effort to lower down the Bowen ratio, moisture from plant systems and from the soil beneath is rather converted to latent heat (Jain et al. 2017). Thus, a higher fraction of moisture content enters into the atmospheric system, creating a moisture deficit over many land regions. Coupled with continued rise in ambient temperature and lack of moisture for further surface evaporation (and evapotranspiration), many regions face increased risk of droughts, forest fires and sometimes dust storms (IPCC 2012). Apart from this, these areas are also likely to experience extreme heat stress, severe heatwave events and slowly a new normal of elevated temperature. As per the Clausius-Clapeyron equation, low-level moisture shows a $\sim 7\%$ increase for every degree Celsius (°C) rise in temperature (Allan and Soden 2008). In order to complete the hydrological cycle, this excess water vapour (temporarily stored in the atmosphere) requires a sink. The sink is often presented in the form of heavy and extreme precipitation events, e.g. severe thunderstorms, tropical cyclones and stronger hurricanes where all the excess stored water vapour is suddenly released (Trenberth et al. 2003). Heavy and extreme precipitation events are therefore significantly intensified and more likely under a warmer climate, leading to more floods and large-scale inundation (IPCC 2012). The presence of cities and urban agglomerations often exacerbates the atmospheric extreme events. For example, most cities already face higher night and daytime temperature than their rural surroundings. They are also major sources of air pollution. Thus, the effects of any heatwave event are often amplified for urban residents due to the urban heat island (UHI) effect. Similarly, lack of porous soil surfaces or water reservoirs due to rampant concretization overwhelms the city drainage systems. In this case, even few minutes of a heavy precipitation event are capable of creating flood-like situations, which would be rare in a rural setting.

13.3 Increase in Frequency and Intensity of Atmospheric Extremes

As discussed, anthropogenic activities are significantly causing changes in the global radiation and water budget. Better computational abilities, deeper knowledge of atmospheric dynamics and better data availability have continuously helped scientists simulate atmospheric extremes with higher accuracy (Tippett 2018). Most numerical weather predictions and climate model outputs unanimously agree that atmospheric extremes will be more common under almost all future climate scenarios (IPCC 2014). Temperature and precipitation are the two most important variables that largely govern short-term weather and long-term climate patterns on local, regional and global scales (Trenberth et al. 2009; IPCC 2014; Brown and Caldeira 2017; Tippett 2018). In fact, shifts from 'normal' temperature, precipitation and associated land–atmosphere feedback systems cascade into a multitude of extremes that have far-reaching, unprecedented and devastative effects (IPCC 2012). Various large- and local-scale atmospheric extremes are discussed in detail in the following sub-sections.

13.3.1 Large-Scale Atmospheric Extremes

13.3.1.1 Heat and Cold Waves

An atmospheric heatwave is a deadly temperature stress event that lasts from a few days to many consecutive weeks. To name a few, the Chicago (1995), European (2003 and 2019), Indian subcontinent (2007, 2015 and 2019), Australian (2008 and 2019), California (2016) and Siberian (2020) events are well-known examples of severe heatwaves in the recent decades. Unlike other extremes, e.g. hurricanes, tornadoes, floods and forest fires, the death toll from heatwaves is not always immediately obvious, but it is still one of the leading causes of weather-related mortality (Robinson 2001; Peterson et al. 2013). According to the World Health Organization, in 2015 alone, an additional 175 million people were exposed to heatwave events compared to previous year averages (WHO 2020). Compared to 1971–2000, the global heatwave exposure (95th percentile) under a combined representative concentration pathway and shared socio-economic pathway (RCP8.5-SSP3) is likely to increase ~18-37 times by 2071-2100 (Liu et al. 2017). In other words, with increasing emissions, less sustainable development and a disparate population growth between developed and developing countries, the global heat exposure will quantifiably amount to ~ 1200 to 2300 billion person days. The mortality risk also depends on factors such as individual and community acclimatization to prolonged temperature extremes, means of thermal comfort, i.e. heating and cooling mechanisms and access to affordable health care (Robinson 2001; Kovats and Hajat 2008; Kent et al. 2014). Heatwaves are in part exacerbated due to UHI formations (see Sect. 13.3.2.1).

Simply speaking, a heat (cold) wave is an extended period of above (below) normal climatological temperature that causes unusually high stress to the human



Fig. 13.1 MODIS (**a**) day and (**b**) night land surface temperature (LST; °C) anomaly over Europe for August 2003, compared to 2000–2010 August mean. Map is prepared via NASA earth observation online platform (https://neo.sci.gsfc.nasa.gov/) and ArcGIS ver. 10.6 mapping software. Only land surface is represented, and sea is masked (in black)

body (Robinson 2001). No singular definition of heat and cold waves exists in literature as most regions fine tune this concept based on geographical and climatological set-up. The degree of deviation from climatological normal, heat thresholds (75th or 95th percentile) and minimum days needed to classify the event as a heat (or severe heat) wave is therefore highly subjective (Robinson 2001; Kent et al. 2014; Liu et al. 2017). Steadman (1984) combined temperature and relative humidity to estimate the apparent temperature (temperature felt by the human body as opposed to actual temperature), which was later modified and termed as the 'heat index' by the National Weather Service. Many other such heat indices have been developed since then.

The famous 2003 summer European heatwave blasted through most European countries with extreme sweltering and record-breaking temperatures. Night-time temperatures at some European locations were noted to be several degrees higher than their climatological daily mean (Black et al. 2004). By the end of the event, almost 70,000 deaths were officially reported (WHO 2020). Figure 13.1 shows MODIS day and night land surface temperature (LST) anomaly over Europe during August 2003, compared to its decadal (2000–2010) mean for the same month. A clear and extreme deviation (>5 $^{\circ}$ C and reaching ~12 $^{\circ}$ C over some European regions) from the decadal average temperature (for both day and night) can be noted for August 2003. If some climate model predictions are to be believed, then the atmospheric conditions presented in the 2003 European heatwave would become the 'new normal' by the year 2100 (Black et al. 2004). Similar deadly heatwaves struck the Indian subcontinent in the summer of 2015, with temperatures crossing 45 °C at many places and killing more than 2500 people in India and ~700 in Karachi (Pakistan) (Wehner et al. 2016). According to the authors, both events had clear anthropogenic signatures in the heat indices. Apart from human-induced factors, strong dry winds called 'loo' blowing from north-western parts of Indian subcontinent, high humidity and significantly less pre-monsoon precipitation contributed to increased severity of the event (Pattanaik et al. 2017). In all likelihood,

temperature extremes across the world, in particular heatwaves, are expected to be more frequent, intense, longer in duration and wider in spatial expanse in the future (Robinson 2001). Mishra et al. (2017) in their study over India highlight that severe heatwave events (such as 2015 heatwave) are likely to increase 15 times by 2050 and 92 times by 2100 if global temperatures increase by 2 °C. Cold extremes at a global level are likely to decrease in the future, but nonetheless are an important atmospheric extreme event (IPCC 2012). In the USA, average cold wave events were recorded to be the highest during the 1980s and in the 2000s there was almost a lack of cold waves (Peterson et al. 2013). However, increasing instances of weak stratospheric polar vortex since the 1990s means that the cold polar winds are unable to remain confined at high latitudes and creep into the mid-latitudes, causing many cold extreme events in Northern America and Eurasia (Kretschmer et al. 2018). Apart from these, it should be mentioned that marine heatwaves have also become unequivocally more frequent, longer and intense and are expected to follow the trend with changing atmospheric conditions and global warming scenarios (Tippett 2018).

13.3.1.2 Precipitation Modification

During the twentieth century, the global temperatures at the surface rose by 0.74 ± 0.18 °C (IPCC 2014). Moreover, barring the year 1998, all the remaining top ten warmest years on NOAA records have occurred since 2005, and projections suggest that 2020s will outrank (high confidence) previous records (Arguez et al. 2020). Global temperature rise directly leads to glacier and continental ice shelf melting, which in turn causes sea level rise and changes in ocean salinity. As mentioned earlier, a warmer climate has a higher moisture-holding capacity, a phenomenon governed by the Clausius-Clapeyron equation (see Sect. 13.2). Tropics and subtropics show a strong association between increased sea surface temperature (SST), positive El Niño Southern Oscillation (ENSO) and the total column water vapour, which influences precipitation patterns such as monsoons and Walker circulation (Trenberth 2011). Stronger ENSO events are also closely linked with recurrent floods in some African nations (Ward et al. 2010). In effect, all direct and indirect temperature feedbacks alter the 'normal' frequency, type, amount, timing, distribution and intensity of various large precipitating systems and extreme precipitation events (Trenberth et al. 2003; Goswami et al. 2006; Kishtawal et al. 2010; IPCC 2014; Tippett 2018). Precipitation commonly occurs as rain, snow or hail, and has embedded inter-seasonal, inter-annual and inter-decadal variabilities (IPCC 2012). Snow accumulates in the winter months and replenishes glacier volume and its slow melting in summer months provide freshwater via rivers, streams and channels. In a warmer climate, precipitation is more likely to be in the form of rain (rather than snow), which will severely disrupt freshwater availability (Groisman et al. 2001). Such a scenario is especially concerning for countries such as India, where the majority of freshwater demand for a billion plus population is fulfilled by glacier-fed rivers.

Global observations suggest that the zonally averaged annual precipitation between the years 1900 and 1995 has increased by 7–12% over Northern Hemisphere (30° –85°N) and by ~2% over Southern Hemisphere (0° –55°S) (Dore 2005).

This asymmetry is mainly due to the distribution of earth's landmass, non-linear ocean-atmosphere coupling and, most importantly, anthropogenic influences. Further, recent decadal observations note that over time moist (dry) regions of the world have become wetter (drier) and that many climate models underestimate the precipitation response to the globally warming atmosphere (Dore 2005; Allan and Soden 2008). There has been a significant increase in both heavy and extreme precipitation events, even when no change or fall in total annual precipitation is noted (Allan and Soden 2008). Similarly, for many increasingly wetter regions in the world too, the total annual precipitation is skewed towards increasing extreme events and such a trend is expected to continue into the future (Donat et al. 2016). Observational data sets for 1951-2003 show the Indian summer monsoon (ISM) over central India region has remained stable, with no significant trends in the number of active and break days (Rajeevan et al. 2006). However, upon closer look over the same region (from 1951 to 2000), Goswami et al. (2006) find a statistically significant trend of increasing extreme events and simultaneously decreasing trend of moderate precipitation events during ISM. Cook et al. (2014) observe that the North American monsoon (NAM) shows small and statistically insignificant changes in total season rainfall under RCP8.5 emission scenario. However, the authors in their Coupled Model Intercomparison Project version 5 (CMIP5)-based simulations find a delayed onset and late season retreat of the monsoon as a response of increased GHGs in the atmosphere.

Apart from modifications in large monsoon systems, e.g. ISM and NAM, the twenty-first century is witnessing a higher proportion of severe thunderstorms, tropical and extratropical cyclones, cloud bursts and flash floods, etc. (IPCC 2012; Hawcroft et al. 2018). For example, large-scale moisture incursion in the US west coast, brought on by atmospheric rivers, is expected to have more frequent landfalls with longer spells causing higher precipitation extremes by 2100 (Tippett 2018). The 2017 hurricane season in the Atlantic Ocean was particularly devastating. High ocean heat content (OHC), increased SST and overabundant moisture content provide the perfect recipe to fuel many long and intense tropical storms and hurricanes (Trenberth et al. 2018). The authors find concrete footprints of anthropogenic-driven climate change in the realization of 2017 hurricane Harvey. Substantially high OHC was converted to massive quantity of atmospheric moisture over the Gulf of Mexico, which later on caused record-breaking rainfall and flooding (Risser and Wehner 2017; Trenberth et al. 2018). Urbanization was found to have further exacerbated the rainfall amount and associated flooding in Houston, USA, during this hurricane (Zhang et al. 2018). Behaviour of heavy and extreme precipitation over urban areas is discussed in detail in Sect. 13.3.2.2. Poor land use and land management practices, e.g. urbanization along riverbeds, watershed areas and low-lying coastal areas, deforestation, removal of grasslands, wetlands and coastal mangroves, and creation of dams and other water diversions, add on to precipitation and flood disaster incidences, severity and losses (Peterson et al. 2013). The 2013 Kedarnath flash flood disaster that killed over 6000 people in India is an excellent example in this respect (Allen et al. 2016).

13.3.1.3 Droughts

M. Jain

Drought is a complex and recurring extreme event caused over land that can stretch from few months to many years (Dai 2011). Multiple factors such as sustained high temperature, extended severe heatwave event, high evaporation rate, very less or no precipitation for long periods, overexploitation of groundwater, deforestation and overgrazing can lead to a drought (van Lanen and Peters 2000; Peterson et al. 2013). A drought, be it meteorological, agricultural or hydrological in nature, is essentially the result of a prolonged state of imbalance between net moisture (Dai 2011). When soil moisture falls below a certain threshold, it fails to support plant systems, leaving large areas not suitable for agriculture, and therefore crucial from global food security perspective. Li et al. (2009) used a revised Palmer Drought Severity Index (PDSI) combines effects of drought frequency and severity, crop yield and irrigation extent in order to model future meteorological drought frequencies under various special report on emissions scenarios (SRESs). Their exhaustive study indicates that under increasing temperature, global drought risk by 2100 will enhance, especially in Africa. The authors add that global area under drought is likely to triple compared to baseline figures, and correspondingly, major crop yields will significantly decline. A recent study by Bisht et al. (2019) over India reveals a strong likelihood that the above moderate drought conditions under a warming climate (RCP 4.5 and RCP 8.5) will significantly worse by 2100 as compared to moderate droughts. Compared to 1976–2005, the average drought length, severity and persistence for each successive drought occurrence is likely to keep increasing from near-future (2010–2039), to mid-future (2040–2069) and then to far-future (2070–2099) (Bisht et al. 2019). This would indeed put an excessive amount of pressure on food security of India, considering its massive population. Besides agricultural productivity and food security, droughts are also responsible for huge biodiversity losses, mortality and economic downfall.

13.3.1.4 Dust Storms

Sustained droughts risk depletion of loose fertile topsoil layer through wind erosion, consequently declining soil quality and even leading to desertification in extreme cases. Assessing the eastern Mediterranean soil, Amit et al. (2021) find evidence of long-range Saharan and Arabian fine dust transport occurring since millions of years, and another geologically recent (Pleistocene) dust source from the nearby Negev desert. The authors emphasize that the agricultural productivity of the region was enhanced only after winds rich in coarse silt quartz were supplied. Apart from Saharan Desert and Arabian Desert, the Taklimakan Desert is another major source of trans-continental dust, all three collectively accounting for over 95% of natural dust emissions in the world (Chen et al. 2018). While dust storms carry nutrients to some regions, they also cause additional burden, especially concerning human health and productivity.

One such event, the 1930s 'dust bowl' that occurred in the USA and lasted nearly a decade is elaborated. The challenges of the great depression and World War I forced Americans to mercilessly plough and harvest the Great Plains region in 1920s, well beyond its ecological limits (Worster 2004). What started as an

ecological blunder slowly turned into the worst drought in its climatological history. The absence of rains in 1930, successive droughts beginning 1931 and recurring intense heatwaves in the following years left the ground devoid of any soil moisture by the mid-1930s (Warn 1953). The long intense drought stretched to almost 20 states, and winds carried away any unanchored soil, which as result of dust aerosol precipitation suppression further exacerbated the drought (Dai 2011; Cook et al. 2014). Worster (2004) details that as many as 179 dust storms recorded in April 1933 were sourced from the drought-stricken areas. Between 9 and 11 May 1934, about 350 million tons of loose brown earth was swept up in the atmosphere by highlevel winds creating a 'black blizzard' and many such blizzards, e.g. Black Sunday followed in the years thereafter (Warn 1953; Worster 2004; Cook et al. 2014). By its end in 1939 (with the onset of good rains), the 'dust bowl' event had caused loss of more than 850 million tons of earth, thousands of deaths from heat, starvation and dust pneumonia, migration of almost 2.5 million people and economic losses that ran in millions of USD each day (Worster 2004). Model simulations suggest that by 2050, extreme drought conditions similar to the dust bowl event will likely occur over south-west USA, Southeast Asia, southern Europe and large regions in African and Australia (Dai 2011). The conditions are expected to keep worsening under climate change and remain largely irreversible for about a millennium even after emissions have been curbed (Romm 2011).

13.3.1.5 Forest Fires and Biomass Burning

Forest fires are yet another example of an extreme event that will significantly get impacted by future changes in climate (Jain et al. 2021). Figure 13.2 shows the global fire burn area map for 2015. It includes both forest and agricultural fires across all biomes (Giglio et al. 2018). Fires start due to natural (e.g. friction between dry trees in summer and lightning strikes) or anthropogenic (e.g. accidental small fire or intentional burning which turned massive) cause. Central African and sub-Saharan



Fig. 13.2 Map showing global area (in thousand ha) burning due to all fires (forest and agricultural) for the year 2015. Map plotted at $0.25^{\circ} \times 0.25^{\circ}$ spatial resolution using ArcGIS ver. 10.6 (*data source:* https://sedac.ciesin.columbia.edu/data/)

region (Fig. 13.2) shows large-scale controlled fires due to extensive biomass burning. While many argue that slash and burn practices, intermittent and other seasonal fires help restore ecosystem productivity and control pests, they have numerous detrimental effects pertaining to air quality and health (Bauer et al. 2019).

In general, frequent and large-scale fires of any type tend to cause loss of biodiversity, elevated local temperatures, poor visibility and increased secondary effects related to pollution and smoke. Besides this, widespread fires also force evacuation and cause property damage. Area under fire, its intensity (amount of energy released), time of the year (summer fires are generally more intense due to low moisture in vegetation and a drier climate), fire type (ground surface, canopy crown or a mix) and fuel type and amount (firewood and biomass properties) determine the severity of any forest fire event (Flannigan et al. 2000). Compared to the twentieth century, twenty-first-century forest fires in California are significantly bigger (>12.000 ha burn area) and start earlier in the season, a shift likely due to climate change (Collins et al. 2019). Combined with a 1 °C temperature rise and a significant decrease (-30%) in autumn rainfall since 1980s, the frequency of extreme fire weather (95th percentile) days in autumn is twice as likely now (Goss et al. 2020). The 2020 autumn California forest fires have burnt an area >1.3 million ha and are yet to be fully contained but have already been recorded as the largest wildfire in its history (Forbes 2020). Prolonged lightning strikes initiated the fires, but an extremely dry weather with persistent heatwave conditions and winds turned it into a major extreme event (Temple 2020). The intense heat in wildfire hotspots further created pyro-cumulonimbus clouds, which creates risk of expansion of existing fires due to downdraft winds or entirely new fires due to more lightning strike activity (NASA 2020). In addition, the smoke from these fires has travelled thousands of miles to cities on the US east coast and even as far as Europe, causing high BC and CO levels along with poor visibility and health issues (Los Angeles Times 2020). Similarly, the frequent intense arctic and boreal fires have been linked with increased atmospheric CO₂ seasonality coupled with increasing land-use land cover disturbances, changes in forest types, albedo and net radiation (Wang et al. 2020).

13.3.2 Regional and Local-Scale Atmospheric Extremes

Atmospheric extremes are without a doubt a cause of concern for the entire world, but its effects are felt more in cities. Our world is now predominantly urban, as per the United Nations by another decade almost 67% population would be living in a city (UN 2018). Also, the morphology, dynamical (energy physics, hydro-meteorology, etc.) processes and the microclimate within cities are very different from the natural landscape (Oke 1997; Zhou and Chen 2018). The next subsection addresses how cities tend to exacerbate temperature and precipitation extremes and air pollution-related events.

13.3.2.1 Urban Heat Island and CO₂ Domes

Urban and suburban areas are typically warmer (2–10 °C) than their adjacent rural surroundings (Oke 1997; Rosenzweig et al. 2006; Kishtawal et al. 2010; Zhou and Chen 2018). This is especially noted on calm and clear nights, with the most intense formations usually occurring during summers (Arnfield 2003). When urban development replaces natural elements with asphalt, glass, bricks and steel, it greatly modifies the thermal and radiative properties such as energy balance, albedo, surface roughness length, wind and water flow patterns and the urban canopy layer characteristics (Arnfield 2003; Huong and Pathirana 2013; Zhou and Chen 2018). After sunset, buildings slowly keep emitting radiation that was absorbed during the day, while natural landscapes tend to remain cooler. Reduction in evapotranspiration also causes elevated sensible heat fraction in dense urban morphology (Vardoulakis et al. 2013). Thus, on top of global temperature rise, warmer urban areas face an additive local climate change effect, generally increasing energy demands. Vardoulakis et al. (2013) estimated a 36% increase in cooling demand during peak summer, for a small city in Greece. Such demands turn out to be exceptionally high in megacities, simply because of their massive population size. To meet the increased energy requirements of warmer cities, more GHGs are inadvertently added into the atmosphere. At the same time, cities have another unique phenomenon called CO₂ domes.

Extensive urbanization processes are directly (e.g. vehicular, commercial and industrial emissions) and indirectly (e.g. cement production) responsible for higher CO_2 concentration (IPCC 2012; Uprety and Saxena 2021b). In their pioneering work over Phoenix (Arizona), Idso et al. (1998) observed CO₂ concentrations over urban core during winter season to be 50% higher as compared to nearby rural counterparts. Further extending their work, Idso et al. (2001) note significantly higher CO2 concentrations on weekdays than weekends supporting the transportation and commercial building emission hypothesis. City-induced average peak CO₂ concentration enhancement at near-surface boundary layer was found to be over 40% during weekdays. Astonishingly, on some weekdays the urban centre CO₂ concentration measurement even crossed 650 ppm, while the rural concentrations were approximated at 350 ppm (Idso et al. 2001). High pollution levels and anthropogenic heat also tend to create an inversion layer over the city, which further intensifies UHI (Rosenzweig et al. 2006). Both UHI and CO₂ domes are diminished or dispersed in high wind conditions. These 'normal' urban microclimatic conditions (higher temperatures and CO_2 concentrations) cause a domino effect when larger atmospheric extreme events such as heatwaves strike the cities. Urban people are likely to face more pronounced repercussions such as reduced thermal comfort, heatrelated health issues, increased fatalities especially in older and socio-economically weaker sections of the society and lowered work productivity (Kovats and Hajat 2008; Vardoulakis et al. 2013).

13.3.2.2 Extreme Precipitation Events and Urban Flooding

Changes in global precipitation patterns and associated increased variability have resulted in frequent, short duration, and large or extreme rainfall events that occur

more over certain regions (Dore 2005). In cities, such quick and large volume spells have the potential to overload drainage systems and lead to urban flooding. As most city surfaces are concretized, water is unable to percolate down and instead flows as surface runoff in the designed storm drains. Conversely, soil surfaces in rural areas readily absorb water and any additional runoff gets channelled into catchments, e.g. ponds, lakes, wetlands and rivers. The absence of infiltration, catchments and natural storage along with reduced resistance of water flow in urban areas puts excessive pressure on the city drains (Huong and Pathirana 2013). Whenever the drainage capacity is unable to keep up with the precipitation influx, it leads to water inundation. Characterizing extreme precipitation events, their intensity, frequency and distribution is therefore critical to managing urban flood risks (Zhou et al. 2017). Recent decades have witnessed growing episodes of urban flooding, even with a couple of hours of intense rainfall (NDMA 2010). Moreover, many studies highlight the role of urban microclimate in increased and stronger local precipitation over cities. Based on long-term observational data sets, Kishtawal et al. (2010) conclude that urban areas of India have shown a significantly increasing frequency (climatological trend) of heavy and extreme precipitation during ISM. They further state that heavier (lighter) precipitation rates are more (less) likely in urban areas compared to rural counterparts. Urban enhancement in severe thunderstorms (precipitation intensity >50-70 mm/h) leads to flash flood situations in urban areas (NDMA 2010). Many concurrent studies suggest that with time, rampantly urbanizing cities across the world have reported a significant increase in their episodes of extreme precipitation events (Mishra and Lettenmaier 2011; Shastri et al. 2015; Zhou et al. 2017; Wu et al. 2019). This is attributed to a higher sensible heat flux, higher moisture-holding capacity, increased availability of cloud condensation nuclei through pollutant aerosols and increased surface roughness in the urban areas (Trenberth et al. 2009; Pathirana et al. 2014).

While urban flooding can occur in any city in the world, developed countries are usually better equipped to handle the situation. In developing countries, it is not uncommon to witness cities flooded for many hours or days due to an extreme precipitation event. Moreover, special attention needs to be focused on the highly populous Asian and African cities, which are currently witnessing rampant urbanization and are projected to continue in the future as well (UN 2018). One of the most notable examples of urban floods caused by an extreme precipitation event is the 2005 Mumbai floods in India. On 26 July, an unprecedented amount of rainfall brought the entire city on a standstill-roads were submerged, public transportation was stopped, and even the Mumbai International Airport had to be shut down. The financial capital of India suffered the worst flood in its entire recorded history, leaving two-thirds city inundated, more than 500 people and 6300 animals dead and an estimated loss of USD ~1.7 billion (MMRDA 2006; Gupta 2007; Ranger et al. 2011). Figure 13.3a shows NASA Tropical Rainfall Measuring Mission (TRMM) satellite-derived image of cumulative precipitation over western India on 26 July 2005. Orange and red areas depict regions that received >200 mm rainfall, which by the India Meteorological Department (IMD) is classified as a very heavy rainfall event (Gupta 2007). The event was heavily centred across Mumbai city.


Fig. 13.3 (a) Cumulative precipitation (mm) over western India on 26 July 2005 derived from NASA 3-hourly TRMM_3B42_7 data set at $0.25^{\circ} \times 0.25^{\circ}$ spatial resolution. Map plotted using ArcGIS ver. 10.6 (data source: https://disc.gsfc.nasa.gov/) and (b) hourly measured precipitation (mm) at Santacruz station in Mumbai

From 0830 (local time) 26 July to 0830 (local time) 27 July, Santacruz IMD station situated in northern Mumbai recorded 944 mm total rainfall, while the Colaba station situated ~25 km away experienced only 73 mm rainfall (Kumar et al. 2008). Figure 13.3b shows the hourly measured precipitation (mm) at Santacruz station. More than 100 mm/h rainfall was received consistently from 15:30 local time to 20: 30 local time with 1-h recording as high as 190 mm rain. Combined with inadequate drainage and high tide event along its coast, the city water overflow was unable to recede for days (Gupta 2007). Indian cities in particular face intense pressure to manage urban floods in the future due to extremely high population densities, huge amounts of dust and sediment influx clogging drains and the compact and densely aggregated city structure. Ranger et al. (2011) further highlight that by the year 2080, chances of a 2005 Mumbai-like event striking will be doubled and associated losses could triple.

13.3.2.3 Crop Residue Burning and Smog

Rising background air pollution levels in cities, especially fine aerosols (PM_{2.5}), are being increasingly linked with higher morbidity and mortality rates (Samet et al. 2000; Greenstone and Fan 2018). In an exhaustive study over China, it was found that 165 cities out of 190 were unable to meet its national ambient air quality standards for PM_{2.5} (Zhang and Cao 2015). Another recently published air quality life index report concludes that sustained PM pollution over the Indo-Gangetic region (one of the most polluted and populated regions in the world) has shortened its residents' life expectancy by ~7 years (Greenstone and Fan 2018). Black carbon (BC) is emitted when traditional fossil fuels, low-grade coal, dung and fuelwood, and agricultural biomass are poorly burnt. These fine aerosol particles readily absorb solar radiation, drive melting of glaciers, cause global dimming and reduce precipitation (Gustafsson and Ramanathan 2016). Seasonal crop residue burning (CRB) and slash and burn agricultural practices emit hazardous quantities of smoke, soot, primary pollutants and other toxicants. In two agriculturally dominant Indian states viz. Punjab and Haryana, ~20.3 Mt and ~9.6 Mt crop residues were burnt, respectively, during the year 2017–2018, resulting in a combined emission of 235.8 Gg PM₁₀ and 194.1 Gg PM_{2.5} from both states (Singh et al. 2020). The authors further highlight that \sim 34.8 Tg and \sim 17.3 Tg of CO₂ equivalent GHGs were emitted from these states, respectively, due to CRB. These pollutants travel long distances depending on wind speed, wind direction and varying particle scavenging rates. Such transboundary pollutants often get trapped in cities and add on to their existing air pollution problems (Saxena et al. 2021). Smog (smoke plus fog) is another atmospheric extreme increasingly faced by many urban areas and big metropolitan cities. During winters, dissipation of emitted and transboundary pollutants is inhibited due to colder temperature and episodes of smog are common (Badarinath et al. 2007). Creation of an inversion layer noted in some highly polluted cities is also responsible for amplifying severe smog events (Saxena et al. 2020). A recent study over India reports that CRB causes economic losses greater than US\$300 million each year, and a tripled risk of acute respiratory illnesses (Chakrabarti et al. 2019). It is further noted that CRB abatement in northern India would prove to be highly cost-effective in the long run, with a 5-year period disability-adjusted life year equivalence estimated at ~US\$1.53 billion. Lastly, anthropogenic point and non-point sources of pollution emit varying quantities of NOx, CO and non-methane hydrocarbons, which under favourable temperature and other meteorological conditions, e.g. clear skies and low wind speed, are able to photochemically form secondary pollutants, e.g. tropospheric O_3 and peroxyacetyl nitrate (PAN) (Rappenglück et al. 2000). Secondary air pollutants also have severe health repercussions (Bernstein et al. 2004). Multi-nodal agencies should therefore evaluate all these risks prevalent in the urban setting while formulating future disaster mitigation plans.

13.4 Vulnerability to Disaster

While the number of atmospheric extremes and their severity is likely to increase by the end of this century, their adverse impacts on human and animal lives, infrastructure, society, economy and environment are expected to not affect nations equally (Black et al. 2004; Li et al. 2009; Tippett 2018; Eckstein et al. 2019). Some regions and countries are likely to have higher vulnerability and exposure to extreme events, and consequences are often multiplied by lack of disaster preparedness, high population density, poverty and low incomes, low accessibility to disaster relief and health services, low adaptive capacity and lack of timely information dissemination (IPCC 2012). Demographics have a key role in deciding disaster vulnerability. Morbidity and mortality rates differ with age, gender, class, race, ethnicity and migrant groups (IPCC 2012). Figure 13.4 shows the global population density map projection for 2020 (CIESIN 2018). Some regions such as India and China are inherently vulnerable to higher mortality risks simply because of their extremely high population density (>1000 persons per km² in some areas, Fig. 13.4). Low



Fig. 13.4 Global gridded population density (persons/km²) estimated for the year 2020. Map plotted at $0.25^{\circ} \times 0.25^{\circ}$ spatial resolution using ArcGIS ver. 10.6 (*data source:* https://sedac.ciesin. columbia.edu/data/)

income and highly populated Asian and African countries are likely to face severe devastative effects needing longer recovery periods (Li et al. 2009; Ward et al. 2010; IPCC 2012; UN 2018). However, developed countries are not immune to disaster losses by any means with 2003 European heatwave, 2005 hurricane Katrina and 2017 hurricane Harvey in the USA, being some well-known recent disasters. Not considering the regions' adaptation capacity, some key factors such as topography, climatic biomes and geographical features govern exposure risk. Many coastal cities are vulnerable to hurricanes, cyclones and typhoons. Some are also increasingly facing submergence risks due to global sea level rise. Impacts could range from temporary displacement to permanent shifting of an entire city to a different location. For example, about 80% of Maldives sits at <1 m sea level height and is extremely vulnerable to submergence if IPCC projected global sea level rise (0.3–1 m by 2100) occurs (Stojanov et al. 2017). At the same time, mountainous terrains are also vulnerable to disasters and, additionally, difficult to access and provide rapid disaster management.

Furthermore, atmospheric extreme events are likely to cause multiple disaster events and could have severe compounding impacts (Trenberth et al. 2003). For example, prolonged heatwaves compounded with low moisture availability significantly increase the risk of forest and bushfires, severe drought and famine conditions and are likely to exacerbate UHIs in cities (Dai 2011; IPCC 2012). From seed sowing to crop harvest, different stages of plant growth are majorly dependent on direct and indirect climate feedbacks (IPCC 2014; Ray et al. 2015; Lesk et al. 2016; Deutsch et al. 2018). Rising heat and cold waves, increasing number of hot degree days, untimely precipitation and monsoon and frequent incidences of intense floods, droughts, thunderstorms, hailstorms, frost, etc., along with certain air pollutants, e.g. tropospheric O_3 , are projected to significantly hinder crop yields (Tai et al. 2014; Ray et al. 2015; Lesk et al. 2016). In another scenario, abundant precipitation might lead to widespread floods, which then compounds into a breeding ground for many

water-borne diseases (NDMA 2010). In order to cope with the increased intensity and frequency of atmospheric extremes likely in the future, disaster risks need to be evaluated based on various emission, population and urbanization scenarios. Risk is defined as multiplicative product of hazard, exposure and disaster vulnerability (NIDM 2019). One such study by Eckstein et al. (2019) estimates the global climate risk index due to atmospheric extreme events. They categorize many south and south-east Asian countries such as Pakistan, India, Nepal, Bangladesh, Myanmar, Vietnam, Thailand and the Philippines as long-term high climate risk countries. High-income countries such as Japan, Germany, Canada and USA (Puerto Rico) have also been highly affected by extreme events in the recent years (Eckstein et al. 2019). The following section discusses various useful strategies in order to prepare, adapt and mitigate atmospheric extreme-related disaster events.

13.5 Disaster Risk Management

13.5.1 Scientific Dimensions and Decision Support Systems

Lack of continuous and high-quality data is often a major hindrance in forecasting any extreme event. This is especially true for developing and low-income countries. In order to manage risks associated with atmospheric extreme events, many countries need to update their knowledge base, infrastructure and resources. This includes rapid and aggressive installation of automatic weather stations, Doppler weather radars and other monitoring systems at national, regional and local levels (NDMA 2010). It should be ensured that such stations have a good network density (especially in climate risk and inherently vulnerable areas), a wide spatial coverage and superior measurement accuracy. In the recent decades, availability of satellite data has proved to be a beneficial tool in continuous global monitoring. However, while some satellite data exist in public domain, there are countries that limit data access causing significant hurdles in conducting scientific research. Free availability of information, ease of data access and inter-agency cooperation should be promoted. Decision support systems are also a key aspect of any disaster preparedness. Various global and regional climate models, and numerical weather prediction models are used by meteorologists to simulate extreme events-their likelihood and other functional characteristics. These models benefit from on availability of robust data, high computational efficiency and immense scientific and technical know-how. Both nowcast and forecast predictions need to improve. Nowcasting provides more imminent warnings (within minutes and hours) of say hurricane landfall or severe thunderstorm, while forecasts are based on projections that are weeks, months and years ahead. Such improvements immensely help authorities to issue timely disaster-related warnings to the public and plan for disaster mitigation. On the other hand, long-range projections and probabilistic distributions (10, 20, 50 or 100 years) greatly assist planners, administrative authorities and concerned nodal agencies to include climate-associated risks (e.g. flood and drought return periods) into environmental impact assessments (EIA), city plans along with finetuning region-specific climate risk management and disaster risk reduction strategies (NIDM 2019). Such projections can identify the probability of frequent and rare atmospheric extreme events in the future.

Geographical information system (GIS) incorporates remotely sensed and ground data sets, e.g. weather, land use, terrain, socio-economic profiles, city maps, traffic and other data, and is a helpful tool for visualization, interpretation and modelling. Usually, global and regional climate models consider land surface is considered a time-invariant input when in reality it is quasi-dynamic. This means that urban and population impacts are not properly accounted for or are severely underestimated. In this regard, Stewart and Oke (2012) proposed local climate zones that include parameters such as building heights, canyon widths and anthropogenic heat flux to better represent cities. Urbanization patterns for the future can be predicted using various GIS-based models. When inherent model land representation is replaced with such schemes, and are also updated for relevant time frames, then more accurate future projections can be made, especially over cities. Lastly, hotspot assessments should be emphasized as it pinpoints certain regions—whether global, national or within a city that are either more vulnerable to disaster risk or is likely to face severe impacts due to demographics. Diverting more resources towards capacity building and mitigation strategies in such hotspots can greatly reduce atmospheric extreme event-related losses.

13.5.2 Disaster Risk Reduction, Mitigation and Adaptation

On a global level, all countries need to aggressively cut down on their emissions, as increased GHGs are inarguably linked to global temperature rise and precipitation changes. The 2016 Paris climate agreement (COP-21) signatories aim to keep global temperature rise well below 2 °C by 2100, with some countries aiming for 0.5 °C further reduction (Markandya et al. 2018). It brings hope that many countries are actively relying higher on renewable sources of energy, are pushing for reduced emission technologies and are aiming to become carbon neutral by the next few decades. Concurrently, in order to reduce the chances of many extremes environmental awareness plays a key role. Most disasters are exacerbated due to poor land management practices, e.g. unbridled construction along water bodies, overgrazing and deforestation, filling of wetlands, and poorly planned urbanization and agricultural practices (IPCC 2012). It is therefore important to identify key natural elements such as wetlands and grasslands and to conserve them. This not only helps to keep soil bound together but also helps to retain soil moisture, effectively managing droughts, dust storms and floods. In addition, EIAs need to be conducted in an honest, unbiased and scientific manner and strictly implemented. Careful modelling of precipitation systems (e.g. monsoons), temperature extreme likelihood and drought indices in advance can aid farmers in many agricultural-based decisions. As discussed in earlier sections, atmospheric extreme events are most acutely experienced at a local level, where majority of the population resides or

infrastructure is involved. The following paragraphs detail some disaster risk reduction, mitigation and adaptation strategies.

Near-surface air temperatures can be sufficiently reduced by urban greening practices. Trees primarily lower Bowen ratio, check moisture imbalance, reduce surface wind speeds and provide shade (Oke 1997). Some noteworthy examples include planting trees in the identified hot clusters, e.g. alongside roads and highways, creating parks inside residential blocks and promoting green (living) roofs and vertical gardens (Rosenzweig et al. 2006). The cooling potential depends on several factors such as tree species, plant size and maturity, surface area (tree canopy and wall or roof coverage) and its location in the urban surroundings (Akbari et al. 1997). The authors find that trees planted on the streets and gardens adjacent to buildings help to save energy by 27-30%. The practice of urban agriculture is on the rise in many cities as it provides many additional benefits apart from reducing overall city temperature (d'Amour et al. 2017; Jain 2021). The second mechanism includes lowering the albedo of urban structures through high albedo roofs and cool pavements (Rosenzweig et al. 2006). Cool roofs are able to reduce indoor temperatures by 2–5 °C and therefore provide increased thermal comfort and energy savings (NRDC 2019). A very recent study by Li et al. (2020) proves efficacy of CaCO₃-acrylic base paint in combating UHI through radiative cooling. The synthesized paint has ~96% solar reflectance, 0.94 sky-window emissivity, cooling power greater than 37 W/m^2 and the ability to reduce noon ambient surface temperature by >1.7 °C. Urban heat island can be also be greatly minimized by increasing canyon space in dense urban areas, using materials that radiate lesser heat during night, promoting energy-efficient architecture and green buildings. Pavement watering cooling strategies have been found to be effective in combating heatwaves (Hendel et al. 2016). After the 2010 Indian heatwave that caused additional ~1350 fatalities (NRDC 2019), the Ahmedabad Heat Action Plan 2013 was introduced as the first extreme heat event early warning system in South Asia (Masood et al. 2015). It involves a multi-pronged approach including public awareness programmes, community and social media outreach, early warning systems based on IMD temperature forecasts, healthcare capacity building and adaptation measures such as city-wide mandatory and voluntary cool roof implementations (NRDC 2019). The report also updates that ~2380 additional fatalities (heat stress-related) were avoided post the successful implementation of this action plan.

In case of precipitation extremes, the first challenge is to redesign, expand and timely desilt city drains so that they can support added pressures due to future climate change. This should take into account up-to-date city (or ground station)-specific intensity distribution frequency (IDF) curves, extreme precipitation event climatology, flood return year periods and future extreme event model estimates (NDMA 2010). Mitigation strategies such as increasing green space, rainwater harvesting, check dams and non-concretized pavements reduce burden on drainage system by slowing surface runoff, and instead recharging groundwater. Drought forecasts both short term and long term can aid crop management practices, speedy allocation of drought relief funds to farmers, timely water resource reallocations, improved water resource management policies in future and development of water

recharge and harvesting projects (Bisht et al. 2019). On the other hand, forest fire management requires strong decision support systems to locate high ignition risk zones through forest characteristics and anthropogenic drivers as well as redistribution of higher fire risk forest types and strictly limiting burning of forest debris (Sturtevant et al. 2009). Alternatives such as composting of crop residue, mechanization and improved technology use and production of biochar from agricultural waster present effective sustainable management practices to curtail CRB (Bhuvaneshwari et al. 2019). A technology developed by a premier agricultural institute in India makes use of economically cheap biomass decomposition tablets that when sprayed over crop residue stubble, speeds up decomposition and forms manure (Deccan Herald 2020). Thus, along with air pollution reduction it co-benefits farmers in increasing agricultural productivity. Other than the above-mentioned strategies, emergency shelters that can withstand extreme events such as hurricanes and tropical cyclones need to be set up in vulnerable areas. Such centres should be sufficient in number and have ample capacity to hold displaced populations and vulnerable groups. In case of cooling centres for heatwave events, the locations need to be carefully planned, taking in the fact that the poor and homeless are the most vulnerable. The importance of indigenous knowledge of local populations and stakeholder involvement can be immensely helpful in assessing climate-related changes experienced over generations, capacity-building and fine-tuning disaster mitigation strategies to suit local needs (IPCC 2012). Moreover, NIDM (2019) calls to categorically segregate present and future atmospheric extreme event-related disasters (sudden and slow onset) into 'avoided, unavoided or unavoidable'. For some extremely high disaster risk regions, this means that migration may be the only adaptation strategy left.

In summary, each country needs to identify and update data gaps, recognize information requirements in context of short- and long-term climate risk management, define status quo from national to sub-community levels, examine disaster risks and carry out context-specific risk tolerance assessments for each sub-region and, lastly, implement feasible risk reduction, mitigation and adaptation strategies (NIDM 2019). As a recent example, India adopted its first national-level disaster management plan in 2016, which recognizes and incorporates three landmark climate and sustainable development agreements—the Sendai Framework for DRR 2015–2030, sustainable development goals 2030 and the recently signed 2016 Paris climate agreement (COP-21) (NDMA 2016). The plan specifically aims to make India a disaster-resilient country through its comprehensive disaster prevention, mitigation, quick response and fast recovery approach. It also seeks to improve disaster management, while assigning specific roles and duties of government bodies at all levels (NDMA 2016).

13.5.3 Early Warning Systems and Post-disaster Management

Many countries, including India, already have detailed disaster management and mitigation plans available. At ground level, however, implementation of these policies is lacking or is not stringent enough. Moreover, most developing countries lack proper early warning systems and timely dissemination of information. Due to lack of timely warning, people can neither prepare themselves for the imminent disaster nor are they able to reach designated emergency shelters. Such circumstances increase the death toll multi-fold. Warning systems require robust information based on scientific assessments (discussed in Sect. 13.5.1 and speedy information dissemination through various platforms, e.g. print media, television news updates and weather warnings on mobile phones). The power of social media continues to be an untapped resource, more so in developing nations, in terms of disaster preparedness, early warning system and post-disaster relief. Post-disaster handling requires joint efforts from multiple sectors and government bodies. This means provision of adequate emergency medical services, distribution of food rations and basic necessities, deployment of military to assist with evacuations and providing financial aid to rebuild livelihoods. With the inevitable increase in atmospheric extreme events in the future, we must aim to better adapt, increase resilience and quickly mitigate climate disasters.

13.6 Summary

Mounting scientific evidence strongly suggests that atmospheric extreme events, e.g. heat and cold waves, droughts, floods and forest fires, are significantly increasing in number, frequency and intensity throughout parts of the world and will continue to do so in the future. The chapter introduces the reader to how 'normal' atmospheric conditions are rapidly changing, causing a significant imbalance in energy and water budget. In a globally polluted environment, the atmospheric burden of GHGs such as CO_2 , CH_4 and N_2O increases tremendously, causing warmer temperatures. The role of temperature-precipitation feedbacks and interlinked land-air processes in exacerbating atmospheric extreme events is discussed in detail in this chapter. In future, atmospheric extreme events are likely to cause multiple disaster events one after the other, e.g. heatwave-associated drought and forest fire, and they could have severe compounding impacts. For each atmospheric extreme event, selected case studies from around the world and over India are discussed in order to highlight the dangers of such disasters and the importance of disaster mitigation. Not only do these disasters cause tens of thousands of deaths each year, but they also incur huge monetary and GDP losses. The chapter further discusses the importance of identifying key vulnerable (1) hot spots, e.g. cities, coastal areas and mountainous regions, (2) demographic groups, e.g. low-income, elderly and migrants, and (3) nations, e.g. India, China, Canada and USA that are highly likely to face severe adverse effects of atmospheric extreme events. Lastly, the chapter outlines various disaster risk management strategies by

highlighting the role of scientific dimensions, e.g. data, models and forecasting along with various decision support systems. Disaster risk reduction, mitigation and adaptation involve aggressive GHG reduction, reversing poor land management practices, implementing strategies such as cool roofs, city drainage redesigning and most importantly preparing short- and long-term climate risk assessments at national to city level. Comprehensive disaster prevention plans, mitigation strategies, early warning systems, quick response and fast recovery approaches are needed to become resilient to climate disasters and atmospheric extreme events in the future.

Acknowledgement The author acknowledges the National Aeronautics and Space Administration (NASA) and Center for International Earth Science Information Network (CIESIN), Columbia University, for the data sets retrieved from their respective online platforms.

Conflict of Interest The author declares no conflict of interest.

References

- Akbari H, Kurn DM, Bretz SE et al (1997) Peak power and cooling energy savings of shade trees. Energ Buildings 25(2):139–148
- Allan RP, Soden BJ (2008) Atmospheric warming and the amplification of precipitation extremes. Science 321(5895):1481–1484
- Allen SK, Rastner P, Arora M et al (2016) Lake outburst and debris flow disaster at Kedarnath, June 2013: hydrometeorological triggering and topographic predisposition. Landslides 13(6): 1479–1491
- Amit R, Enzel Y, Crouvi O (2021) Quaternary influx of proximal coarse-grained dust altered circum-Mediterranean soil productivity and impacted early human culture. Geology 49(1): 61–65
- Andrews E, Ogren JA, Bonasoni P et al (2011) Climatology of aerosol radiative properties in the free troposphere. Atmos Res 102(4):365–393
- Arguez A, Hurley S, Inamdar A et al (2020) Should we expect each year in the next decade (2019–28) to be ranked among the top 10 warmest years globally? Bull Am Meteorol Soc 101(5):655–663
- Arnfield AJ (2003) Two decades of urban climate research: a review of turbulence, exchanges of energy and water, and the urban heat island. Int J Climatol 23(1):1–26
- Badarinath KVS, Latha KM, Chand TK et al (2007) Black carbon aerosols and gaseous pollutants in an urban area in North India during a fog period. Atmos Res 85(2):209–216
- Bauer SE, Im U, Mezuman K et al (2019) Desert dust, industrialization, and agricultural fires: health impacts of outdoor air pollution in Africa. J Geophys Res 124(7):4104–4120
- Bernstein JA, Alexis N, Barnes C et al (2004) Health effects of air pollution. J Allergy Clin Immunol 114(5):1116–1123
- Bhuvaneshwari S, Hettiarachchi H, Meegoda JN (2019) Crop residue burning in India: policy challenges and potential solutions. Int J Environ Res Public Health 16(5):832
- Bisht DS, Sridhar V, Mishra A et al (2019) Drought characterization over India under projected climate scenario. Int J Climatol 39(4):1889–1911
- Black E, Blackburn M, Harrison G et al (2004) Factors contributing to the summer 2003 European heatwave. Weather 59(8):217–223
- Brown PT, Caldeira K (2017) Greater future global warming inferred from Earth's recent energy budget. Nature 552(7683):45–50

- Chakrabarti S, Khan MT, Kishore A et al (2019) Risk of acute respiratory infection from crop burning in India: estimating disease burden and economic welfare from satellite and national health survey data for 250 000 persons. Int J Epidemiol 48(4):1113–1124
- Chen S, Jiang N, Huang J et al (2018) Quantifying contributions of natural and anthropogenic dust emission from different climatic regions. Atmos Environ 191:94–104
- CIESIN (2018) Center for International Earth Science Information Network Gridded Population of the World, Version 4 (GPWv4): population density, revision 11. Available via NASA Socioeconomic Data and Applications Center (SEDAC). https://doi.org/10.7927/H49C6VHW. Accessed 17 Aug 2020
- Collins BM, Miller JD, Knapp EE et al (2019) A quantitative comparison of forest fires in central and northern California under early (1911–1924) and contemporary (2002–2015) fire suppression. Int J Wildland Fire 28(2):138–148
- Cook BI, Seager R, Smerdon JE (2014) The worst North American drought year of the last millennium: 1934. Geophys Res Lett 41(20):7298–7305
- d'Amour CB, Reitsma F, Baiocchi G et al (2017) Future urban land expansion and implications for global croplands. Proc Natl Acad Sci 114(34):8939–8944
- Dai A (2011) Drought under global warming: a review. Wiley Interdiscip Rev Clim Chang 2(1): 45–65
- Deccan Herald (2020, Sept 24) Stubble burning: Gopal Rai says Pusa institute's bio-decomposer can help reduce pollution. https://www.deccanherald.com/national/north-and-central/stubble-burning-gopal-rai-says-pusa-institutes-bio-decomposer-can-help-reduce-pollution-892279. html. Accessed 25 Sept 2020
- Deutsch CA, Tewksbury JJ, Tigchelaar M et al (2018) Increase in crop losses to insect pests in a warming climate. Science 361(6405):916–919
- Donat MG, Lowry AL, Alexander LV et al (2016) More extreme precipitation in the world's dry and wet regions. Nat Clim Chang 6(5):508–513
- Dore MH (2005) Climate change and changes in global precipitation patterns: what do we know? Environ Int 31(8):1167–1181
- Eckstein D, Künzel V, Schäfer L et al (2019) Global climate risk index 2020: who suffers most from extreme weather events? Germanwatch Nord-Süd Initiative eV. https://www.germanwatch.org/en/cri. Accessed 26 Aug 2020
- Fankhauser S, McDermott TK (2014) Understanding the adaptation deficit: why are poor countries more vulnerable to climate events than rich countries? Glob Environ Chang 27:9–18
- Flannigan MD, Stocks BJ, Wotton BM (2000) Climate change and forest fires. Sci Total Environ 262(3):221–229
- Forbes (2020, Sept 13) Largest wildfire in California history now up to 877,477 acres burned, 28% contained. https://www.forbes.com/sites/mattperez/2020/09/13/largest-wildfire-in-california-history-now-up-to-877477-acres-burned-28-contained/#3be34c829199. Accessed 15 Sept 2020
- Giglio L, Randerson JT, van der Werf GR (2018) Global fire emissions indicators, grids: 1997-2015. Available via NASA Socioeconomic Data and Applications Center (SEDAC). https://doi.org/10.7927/H400002V. Accessed 17 Aug 2020
- Goss M, Swain DL, Abatzoglou JT et al (2020) Climate change is increasing the likelihood of extreme autumn wildfire conditions across California. Environ Res Lett 15(9):094016
- Goswami BN, Venugopal V, Sengupta D et al (2006) Increasing trend of extreme rain events over India in a warming environment. Science 314(5804):1442–1445
- Greenstone M, Fan CQ (2018) Introducing the air quality life index. AQLI Annual Report. https:// aqli.epic.uchicago.edu/wp-content/uploads/2018/11/AQLI-Report.111918-2.pdf. Accessed 13 Aug 2020
- Groisman PY, Knight RW, Karl TR (2001) Heavy precipitation and high streamflow in the contiguous United States: trends in the twentieth century. Bull Am Meteorol Soc 82(2):219–246
- Gupta K (2007) Urban flood resilience planning and management and lessons for the future: a case study of Mumbai, India. Urban Water J 4(3):83–194

- Gustafsson Ö, Ramanathan V (2016) Convergence on climate warming by black carbon aerosols. Proc Natl Acad Sci 113(16):4243–4245
- Hawcroft M, Walsh E, Hodges K et al (2018) Significantly increased extreme precipitation expected in Europe and North America from extratropical cyclones. Environ Res Lett 13(12):124006
- Hendel M, Gutierrez P, Colombert M et al (2016) Measuring the effects of urban heat island mitigation techniques in the field: application to the case of pavement-watering in Paris. Urban Clim 16:43–58
- Huong HTL, Pathirana A (2013) Urbanization and climate change impacts on future urban flooding in Can Tho city, Vietnam. Hydrol Earth Syst Sci 17(1):379
- Idso CD, Idso SB, Balling RC Jr (1998) The urban CO2 dome of Phoenix, Arizona. Phys Geogr 19(2):95–108
- Idso CD, Idso SB, Balling RC Jr (2001) An intensive two-week study of an urban CO2 dome in Phoenix, Arizona, USA. Atmos Environ 35(6):995–1000
- IPCC (2012) Managing the risks of extreme events and disasters to advance climate change adaptation. In: Field CB, Barros V, Stocker TF et al (eds) A special report of working groups I and II of the intergovernmental panel on climate change. Cambridge University Press, UK and New York, pp 1–583
- IPCC (2014) Climate change 2014: synthesis report. In: Pachauri RK, Meyer LA (eds) Contribution of working groups I, II and III to the fifth assessment report of the intergovernmental panel on climate change. IPCC, Geneva, pp 1–151
- Jain M (2021) Mitigation of urbanization ill-effects through urban agriculture inclusion in cities. In: Diehl J, Kaur H (eds) New forms of urban agriculture: an urban ecology perspective. Springer, Singapore (In print)
- Jain M, Dimri AP, Niyogi D (2017) Land-air interactions over urban-rural transects using satellite observations: analysis over Delhi, India from 1991–2016. Remote Sens 9(12):1283
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Kent ST, McClure LA, Zaitchik BF et al (2014) Heat waves and health outcomes in Alabama (USA): the importance of heat wave definition. Environ Health Perspect 122(2):151–158
- Kishtawal CM, Niyogi D, Tewari M et al (2010) Urbanization signature in the observed heavy rainfall climatology over India. Int J Climatol 30(13):1908–1916
- Kovats RS, Hajat S (2008) Heat stress and public health: a critical review. Annu Rev Public Health 29:41–55
- Kretschmer M, Coumou D, Agel L et al (2018) More-persistent weak stratospheric polar vortex states linked to cold extremes. Bull Am Meteorol Soc 99(1):49–60
- Kumar A, Dudhia J, Rotunno R et al (2008) Analysis of the 26 July 2005 heavy rain event over Mumbai, India using the Weather Research and Forecasting (WRF) model. Q J R Meteorol Soc 134(636):1897–1910
- Lesk C, Rowhani P, Ramankutty N (2016) Influence of extreme weather disasters on global crop production. Nature 529(7584):84
- Li Y, Ye W, Wang M et al (2009) Climate change and drought: a risk assessment of crop-yield impacts. Clim Res 39(1):31–46
- Li X, Peoples J, Huang Z et al (2020) Full daytime sub-ambient radiative cooling with high figure of merit in commercial-like paints. arXiv preprint arXiv:2008.03372
- Liu Z, Anderson B, Yan K et al (2017) Global and regional changes in exposure to extreme heat and the relative contributions of climate and population change. Sci Rep 7:43909
- Los Angeles Times (2020, Sept 15) Smoke from California wildfires reaches the East Coast and Europe. https://www.latimes.com/california/story/2020-09-15/smoke-california-wildfires-reaches-east-coast-europe. Accessed 17 Sept 2020
- Markandya A, Sampedro J, Smith SJ et al (2018) Health co-benefits from air pollution and mitigation costs of the Paris Agreement: a modelling study. Lancet Planet Health 2(3):126–133

- Masood I, Majid Z, Sohail S et al (2015) The deadly heat wave of Pakistan, June 2015. Int J Occup Environ Med 6(4):247
- Mishra V, Lettenmaier DP (2011) Climatic trends in major US urban areas, 1950–2009. Geophys Res Lett 38(16)
- Mishra V, Mukherjee S, Kumar R et al (2017) Heat wave exposure in India in current, 1.5 C, and 2.0 C worlds. Environ Res Lett 12(12):124012
- MMRDA (2006) Mumbai Metropolitan Regional Development Authority. https://mmrda. maharashtra.gov.in/documents/10180/11271261/1/7026649b-8587-43f7-b04a-caf4aa2678b7. Accessed 28 Aug 2020
- Myhre G, Shindell D, Bréon FM et al (2013) Anthropogenic and natural radiative forcing. In: Jacob D, Ravishankara AR, Shine K (eds) Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change. Cambridge University Press, Cambridge, pp 664–721
- NASA (2020, Sept 8) California's creek fire creates its own pyrocumulonimbus cloud. https://www. nasa.gov/feature/goddard/2020/californias-creek-fire-creates-its-own-pyrocumulonimbuscloud. Accessed 15 Sept 2020
- NDMA (2010) National disaster management guidelines: management of urban flooding. National Disaster Management Authority, Government of India. ISBN: 978-93-80440-09-5 https://ndma.gov.in/images/guidelines/management_urban_flooding.pdf. Accessed 28 Aug 2020
- NDMA (2016) National Disaster Management Plan. National Disaster Management Authority, Government of India. https://ndma.gov.in/images/policyplan/dmplan/National%20Disaster%20 Management%20Plan%20May%202016.pdf. Accessed 28 Oct 2020
- NIDM (2019) Climate risk management (CRM) framework for India: addressing loss and damage (L&D). National Institute of Disaster Management, Government of India. https://nidm.gov.in/ PDF/pubs/GIZ_NIDM_Climate%20RiskManagementFramework.pdf. Accessed 25 Aug 2020
- NRDC (2019) Natural Resources Defense Council-Ahmedabad Heat Action Plan. https://www. nrdc.org/sites/default/files/ahmedabad-heat-action-plan-2018.pdf. Accessed 21 Oct 2020
- Oke TR (1997) Urban climates and global environmental change. In: Thompson RD, Perry A (eds) Applied climatology: principles & practices. Routledge, New York, pp 273–287
- Pathirana A, Denekew HB, Veerbeek W et al (2014) Impact of urban growth-driven land use change on microclimate and extreme precipitation: a sensitivity study. Atmos Res 138:59–72
- Pattanaik DR, Mohapatra M, Srivastava AK et al (2017) Heat wave over India during summer 2015: an assessment of real time extended range forecast. Meteorog Atmos Phys 129(4):375–393
- Peterson TC, Heim RR Jr, Hirsch R et al (2013) Monitoring and understanding changes in heat waves, cold waves, floods, and droughts in the United States: state of knowledge. Bull Am Meteorol Soc 94(6):821–834
- Prasad N, Ranghieri F, Shah F et al (2008) Climate resilient cities: a primer on reducing vulnerabilities to disasters. The World Bank, Washington, DC. http://documents1.worldbank. org/curated/en/164671468026943315/pdf/466800PUB0Clim1010FFICIAL0USE00NLY1. pdf. Accessed 14 Aug 2020
- Quaas J, Boucher O, Dufresne JL et al (2004) Impacts of greenhouse gases and aerosol direct and indirect effects on clouds and radiation in atmospheric GCM simulations of the 1930–1989 period. Clim Dyn 23(7–8):779–789
- Rajeevan M, Bhate J, Kale JD et al (2006) High resolution daily gridded rainfall data for the Indian region: analysis of break and active. Curr Sci 91(3):296–306
- Ranger N, Hallegatte S, Bhattacharya S et al (2011) An assessment of the potential impact of climate change on flood risk in Mumbai. Clim Chang 104(1):139–167
- Rappenglück B, Oyola P, Olaeta I et al (2000) The evolution of photochemical smog in the Metropolitan Area of Santiago de Chile. J Appl Meteorol 39(3):275–290
- Ray DK, Gerber JS, MacDonald GK et al (2015) Climate variation explains a third of global crop yield variability. Nat Commun 6:5989

- Risser MD, Wehner MF (2017) Attributable human-induced changes in the likelihood and magnitude of the observed extreme precipitation during Hurricane Harvey. Geophys Res Lett 44(24): 12–457
- Robinson PJ (2001) On the definition of a heat wave. J Appl Meteorol 40(4):762-775

Romm J (2011) The next dust bowl. Nature 478(7370):450-451

- Rosenzweig C, Solecki W, Slosberg R (2006) Mitigating New York City's heat island with urban forestry, living roofs, and light surfaces. A report to the New York State Energy Research and Development Authority. https://www.giss.nasa.gov/research/news/20060130/103341.pdf. Accessed 5 Sept 2020
- Samet JM, Dominici F, Curriero FC et al (2000) Fine particulate air pollution and mortality in 20 US cities, 1987–1994. N Engl J Med 343(24):1742–1749
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8):6639– 6656
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Shastri H, Paul S, Ghosh S et al (2015) Impacts of urbanization on Indian summer monsoon rainfall extremes. J Geophys Res Atmos 120(2):496–516
- Singh T, Biswal A, Mor S et al (2020) A high-resolution emission inventory of air pollutants from primary crop residue burning over Northern India based on VIIRS thermal anomalies. Environ Pollut 266:115132
- Soba D, Shu T, Runion GB et al (2020) Effects of elevated [CO2] on photosynthesis and seed yield parameters in two soybean genotypes with contrasting water use efficiency. Environ Exp Bot 178:104154
- Sonwani S, Saxena P, Shukla A (2021) Carbonaceous aerosol characterization and their relationship with meteorological parameters during summer monsoon and winter monsoon at an industrial region in Delhi, India. Earth Space Sci 8(4):e2020EA001303
- Steadman RG (1984) A universal scale of apparent temperature. J Clim Appl Meteorol 23(12): 1674–1687
- Stewart ID, Oke TR (2012) Local climate zones for urban temperature studies. Bull Am Meteorol Soc 93(12):1879–1900
- Stojanov R, Duží B, Kelman I et al (2017) Local perceptions of climate change impacts and migration patterns in Malé, Maldives. Geogr J 183(4):370–385
- Sturtevant BR, Miranda BR, Yang J et al (2009) Studying fire mitigation strategies in multiownership landscapes: balancing the management of fire-dependent ecosystems and fire risk. Ecosystems 12(3):445
- Tai AP, Martin MV, Heald CL (2014) Threat to future global food security from climate change and ozone air pollution. Nat Clim Chang 4(9):817
- Temple J (2020, Aug 20) Yes, climate change is almost certainly fueling California's massive fires. MIT Technology Review. https://www.technologyreview.com/2020/08/20/1007478/californiawildfires-climate-change-heatwaves/. Accessed 15 Sept 2020
- Thomas K, Hardy RD, Lazrus H et al (2019) Explaining differential vulnerability to climate change: a social science review. Wiley Interdiscip Rev Clim Chang 10(2):565
- Thornton PK, Ericksen PJ, Herrero M et al (2014) Climate variability and vulnerability to climate change: a review. Glob Chang Biol 20(11):3313–3328
- Tippett MK (2018) Extreme weather and climate. NPJ Clim Atmos Sci 1(45):1-2
- Trenberth KE (2011) Changes in precipitation with climate change. Clim Res 47(1-2):123-138
- Trenberth KE, Dai A, Rasmussen RM et al (2003) The changing character of precipitation. Bull Am Meteorol Soc 84(9):1205–1218
- Trenberth KE, Fasullo JT, Kiehl J (2009) Earth's global energy budget. Bull Am Meteorol Soc 90(3):311–324

- Trenberth KE, Cheng L, Jacobs P et al (2018) Hurricane Harvey links to ocean heat content and climate change adaptation. Earth's Future 6(5):730–744
- UN (2018) Revision of world urbanization prospects. United Nations, New York. https:// population.un.org/wup/Publications/Files/WUP2018-Report.pdf. Accessed 8 Aug 2020
- Uprety DC, Saxena P (2021a) Technologies for green house gas assessment in crop studies. Springer, Singapore, pp 1–134
- Uprety DC, Saxena P (2021b) Carbon dioxide. In: Technologies for green house gas assessment in crop studies. Springer, Singapore, pp 7–57
- van Lanen HAJ, Peters E (2000) Definition, effects and assessment of groundwater droughts. In: Vogt JV, Somma F (eds) Drought and drought mitigation in Europe. Advances in natural and technological hazards research. Springer, Dordrecht, pp 49–61
- Vardoulakis E, Karamanis D, Fotiadi A et al (2013) The urban heat island effect in a small Mediterranean city of high summer temperatures and cooling energy demands. Sol Energy 94:128–144
- Wang JA, Sulla-Menashe D, Woodcock CE et al (2020) Extensive land cover change across Arctic– Boreal Northwestern North America from disturbance and climate forcing. Glob Chang Biol 26(2):807–822
- Ward PJ, Beets W, Bouwer LM et al (2010) Sensitivity of river discharge to ENSO. Geophys Res Lett 37(12):1–6
- Warn GF (1953) Drought and dust on the plains. Weatherwise 6(3):67-71
- Wehner M, Stone D, Krishnan H et al (2016) The deadly combination of heat and humidity in India and Pakistan in summer 2015. Bull Am Meteorol Soc 97(12):S81–S86
- Wernberg T, Smale DA, Tuya F et al (2013) An extreme climatic event alters marine ecosystem structure in a global biodiversity hotspot. Nat Clim Chang 3(1):78–82
- WHO (2020) Information and public health advice: heat and health. https://www.who.int/ globalchange/publications/heat-and-health/en/. Accessed 16 Sept 2020
- Worster D (2004) Dust bowl: the southern plains in the 1930s. Oxford University Press, pp 1-304
- Wu M, Luo Y, Chen F et al (2019) Observed link of extreme hourly precipitation changes to urbanization over coastal South China. J Appl Meteorol Climatol 58(8):1799–1819
- Zhang YL, Cao F (2015) Fine particulate matter (PM 2.5) in China at a city level. Sci Rep 5:14884 Zhang W, Villarini G, Vecchi GA et al (2018) Urbanization exacerbated the rainfall and flooding caused by hurricane Harvey in Houston. Nature 563(7731):384–388
- Zhou X, Chen H (2018) Impact of urbanization-related land use land cover changes and urban
- morphology changes on the urban heat island phenomenon. Sci Total Environ 635:1467–1476 Zhou X, Bai Z, Yang Y (2017) Linking trends in urban extreme rainfall to urban flooding in China. Int J Climatol 37(13):4586–4593



Disaster Preparedness and Emergency Response for Air Pollution and Related Health Extremes

14

Anjali Barwal, Sonal Bindal, and Anil Kumar Gupta

Abstract

Over the past few years, air pollution has become a growing problem with an increasing number of acute incidents of air pollution in many cities across the world. As a direct consequence, air quality is contributing extreme risk to public health, accounting for about one in every nine deaths per year. Sudden and extreme air pollution episodes are creating disaster like situations causing an immediate public health emergency. Every year, as per the World Health Organization (WHO), about seven million people die tragically from air pollutioninduced diseases around the world. Approximately four million such deaths occur in the Asia-Pacific region only. Now, it is not possible to perceive air pollution as an isolated environmental problem. It is high time that sudden air pollution management should be implemented into state and city planning processes. There is an urgent need for a concerted approach consisting of the generation and compilation of well-structured and exhaustive databases with geospatial locations on air pollution. There is also the requirement of effective representation and well-timely dissemination of air quality information, use of accurate and complex inferences to make significant implementation in air quality-related policy decisions.

Keywords

Air pollution \cdot Public health emergency \cdot Disaster management \cdot Anthropogenic disasters \cdot Air pollution episodes

329

A. Barwal $(\boxtimes) \cdot S$. Bindal $\cdot A$. K. Gupta

National Institute of Disaster Management, Ministry of Home Affairs, Government of India, New Delhi, India

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_14

14.1 Introduction

Air is mainly composed of dynamic fusion of gases (nitrogen, oxygen, etc.), water vapour and particulate matter. There has been tremendous variation in the air quality composition due to the increased industrialization and development in civilized human society (Mackenzie 2016). The composition of various gaseous air pollutants including dust particles, volatile organic compounds, and heavy metals is constantly increasing because of unmanageable solid and fossil fuel combustion. The chemical composition, reaction sequence, time and mechanism of all these pollutants differ considerably (Singh 2020; Saxena and Naik 2018). These pollutants have different toxicological impacts on human health, depending on exposure period and concentration, as illustrated in Fig. 14.1.

Globally, air pollution contributes substantially to various disease burdens, premature mortality, morbidity, reduced life expectancy, etc., with a greater impact in developing countries than in developed countries (Balakrishnan et al. 2019). The World Health Organization (WHO) report estimates that worldwide around seven million people die every year from various diseases such as stroke, heart disease, lung cancer, pulmonary disorders and respiratory infections, caused by polluted air quality (Schwela and Haq 2020; Sonwani et al. 2021), which means every hour around 800 humans (or say, 13 every minute) died because of the polluted air they breathe. Out of which, around four million such deaths occur in the Asia-Pacific region only (UNEP 2019).

WHO scientists also believe that global warming temperatures could lead to malaria, yellow fever, bone loss, respiratory diseases and other diseases spreading throughout the world. The proliferation of such infections is growing further in warm



Fig. 14.1 Regional estimate of air pollution and its health consequences (source: WHO PHE Infographics on air pollution, 2020)

and humid environments as pathogens often live under such climatic conditions. Understanding of the geographical distribution of diseases for the health planning of each country is required to notify the regulators of the situation of each disease in the world and to include the necessary health initiatives for local residents in their health policies as well (Javadinejad et al. 2020).

In this chapter, an attempt has been made to examine and assess the various components of air pollution, their effects on human health and way forward to deal with air pollution as a disaster and public health emergency with existing policies and programmes. The identified framework will be useful to all stakeholders and decision-makers as a roadmap while designing the future air pollution mitigation policies from disaster management point of view.

14.2 Air Pollution Risks and Episodes

Asia is home to many extremely polluted cities in the world. As per the United Nations Department of Economic and Social Affairs (UNDESA) World Urbanization prospect report, astounding two-third of the world's population would reside in urban areas by 2050. And in this regard, India would lead the world with over 400 million people living in these urban regions. Evidently, sustainable solutions would be required to impart clean air to breathe and sustain life on the planet (Rodríguez-Urrego and Rodríguez-Urrego 2020). As shown in Fig. 14.2, air pollution is among the top five environmental risks leading to premature deaths globally for 2010 to 2050.

Air pollution episodes in the twentieth and twenty-first centuries have had significant social, economic and political effects. In extreme conditions, episodes of air pollution cause global public concern that is correlated with quantifiable health risks (Sonwani et al. 2022). The major notable episodes of air pollution are mentioned below in Table 14.1.

14.3 Air Pollution as Disaster and Its Preparedness

Disaster is a threat that results in substantial physical injury, loss of life or environmental impacts. Disasters are now known to be the result of insufficient risk management. Life exposure to high concentrations of pollutants over a short period of time can result in unexpected impacts as seen in the events of chemical, petrochemicals and other industrial accidents. Disasters involving extreme air pollution are causing an immediate public health crisis leading to an increased pressure on stakeholders and agencies to provide solutions to protect impacted communities during any such emergencies (Chandrappa and Kulshrestha 2016). During any disaster, the air quality management is dependent on the local community preparedness and external aid. Air pollution can also lead to disasters and disasters can trigger the air pollution as well (Harlan and Ruddell 2011). In both cases, human health and the environment would be affected. Moreover, ambient air pollution is considered as



Global premature deaths from selected environmental risks: Baseline. 2010 to 2050

Fig. 14.2 Global premature deaths attributed by the environmental risks (source: Marchal et al. 2011)

S. no.	Major air pollution episodes	Year	Cause
1.	Great Smog of London	1952	Natural
2.	Mount St. Helens Eruption	1980	Natural
3.	Bhopal Gas Tragedy	1984	Anthropogenic
4.	Lake Nyos Disaster	1986	Natural
5.	Kuwaiti Oil Fires	1991	Anthropogenic
6.	Southeast Asian Haze	1997/1998	Natural
7.	Indonesian Forest Fires	1997	Natural
8.	World Trade Center	2001	Anthropogenic
9.	Malaysian Haze	2005	Natural
10.	Southeast Asian Haze	2006	Anthropogenic
11.	North Atlantic Volcanic Ash	2010	Natural
12.	Beijing, China	2013	Natural
13.	Southeast Asian Haze	2015	Anthropogenic
14.	Bushfires in Australia	2019/2020	Natural
15.	Assam's Baghjan Oil Well Fire, India	2020	Anthropogenic

Table 14.1 Major global air pollution episodes

the fifth highest ranking risk factor for mortality, worldwide. There are basically three major disaster components of air pollution as mentioned in Fig. 14.3. These all are interrelated with each other, one drives or subsides the proportion leading to air

Employmental	Disaster	
- Fog/ Smog Dust & sand storms - Forest fires - Industrial/chemical accidents - City/urban pollution - Biomass burning	 Respiratory disorders Cardivascular disorders Neurological disorders Accidental injuries & deaths Reduced fertility Birth defects Cancer 	Socio-Economic impact - Supply chain disruption - Bussiness disruptions - Livlihood loss - Effect on infrastructure/ property assets

Fig. 14.3 Disaster components and its associated risks

pollution disasters. The environmental impacts such as fog, smog, dust storms, industrial or chemical accidents and biomass burning ultimately lead to various health risks such as respiratory, cardiovascular, neurological disorders, accidental injuries or even deaths (Tigala et al. 2018). Further, the health burden related to air pollution is associated with socio-economic impacts such as supply chain disruptions and livelihood losses. Therefore, there is an immediate need to globally minimize air pollution emissions, as WHO has also cautioned air pollution as a "public health emergency".

Disaster preparedness with respect to air pollution is crucial for individuals, households, businesses and communities, but many of them are often not prepared. The recent air pollution episodes and its related series of health emergencies serve to highlight the importance of individual and community responsibility, local government coordination and continuity strategies to ensure the capacity to respond to and recover from any major incidents. The concept of disaster preparedness encompasses measures designed to improve the safety and security of human lives and livelihood during any disaster damage and disruption, as well as the capacity to interact in post-disaster restoration and early recovery actions.

14.4 Air Pollution Emergency Causes

Major air pollution emergencies can have the following dimensions:

14.4.1 Dense Fog and Smog

Due to dense fog conditions, dispersion of particulate matter becomes difficult as air turns very dense due to the high relative humidity (i.e. more than 90%). Smog is the

type of air pollution that affects most cities. Dense fog curbs the visibility and thus affects many human activities including all forms of traffic, aircraft operations and sailing. Moreover, when the fog contains toxic air pollutants, it becomes more harmful to humans and the environment (Saxena et al. 2020a).

Case study In Pakistan's Lahore city, smog had developed on 2–5 November 2016, which shows a very poor Air Quality Index (AQI) when compared to the baseline condition measured during the same time the previous year, i.e. 2015. During smog, the highest pollutant concentration measured was for NOx, which was nearly 17 times higher than the baseline value. The other contaminants (SO₂ and O₃) were four times higher than the baseline value, whereas CO, VOC and PM_{2.5} were two times higher, and PM₁₀ was six times higher, respectively. Cool temperatures, calm winds and high humidity were the weather conditions during the smog incident, which eventually triggered the photochemical reaction that caused the smog. The information was gathered from Lahore's tertiary care hospitals, which showed a rise in the number of cases of ocular surface diseases such as dry eyes, irritation, lid erosion, uveitis, conjunctival diseases, corneal diseases and lacrimation. According to the results obtained, only 200 (30%) out of 700 patients were reported in the year 2015, and 500 (70%) were reported in the year 2016 only (Ashraf et al. 2019). The increased air pollution compelled the provincial government to declare the closure of all of the province's schools.

14.4.2 Chemical/Industrial Accidents

Releases of hazardous or toxic substances may affect a broader range of people and locations. Hazardous or toxic substance releases could have an impact on more geographic areas and people. Not all chemical accidents pose the same threat with respect to air pollution. The sudden toxic gaseous discharges can damage the human health and environment or cause corrosivity to the ancient monuments and modern buildings, etc.

Case study Recently, the Vizag accident at the LG Polymers chemical plant in Visakhapatnam, India, that happened on 7 May 2020, due to malfunction in the cooling system of styrene storage tanks, resulted in the release of 3 tonnes of toxic styrene gas into the atmosphere that spread over the entire region. On that accident day, the styrene concentrations in the air in the region were approximately 500 times higher than the prescribed WHO limit of less than 5 parts per billion (ppb). The facility in Visakhapatnam is spread over 240 hectares (ha), including the adjacent residential areas. There is also a revenue village nearby, which has resulted in a higher exposure rate (Hailwood 2020).

Central Pollution Control Board (CPCB) monitors the real-time ambient air quality monitoring stations network of the monitored three volatile organic compounds (VOCs)—xylene (C_8H_{10}), toluene (C_7H_8) and benzene (C_6H_6)—in Visakhapatnam District, which is 14 km downstream of accidental site of gas

leakage. The data dashboard helped in compilation of xylene levels up to 18 ppb, toluene levels up to 35 ppb and benzene levels up to 12 ppb, against the CPCB standards, i.e. 5 ppb (annual average) (CSE 2020). This helped in creating awareness around the nearby areas for air quality issues.

14.4.3 Dust and Sand Storms

Dust or sand storms are observed throughout most of the continents. Every year, millions of people are affected by droughts and dust storms (UNEP 2019). Sand storms originate from the semi-arid and arid regions of the world (Sarkar et al. 2019). These are also common events in the Central Asia, Middle-East and south-west Asia's arid and semi-arid regions (Indoitu et al. 2012; Barlow et al. 2016). Sandstorms are caused by radiant heat, high wind speed, scarce or dearth of vegetation cover, soil erosion, low humidity and a high-pressure gradient. Dust enhances aerosol, which affects the quality and visibility of air (Csavina et al. 2014). Longer periods of contact with fine dust particles are one of the important factors in the incidence of lung cancer and heart diseases.

Drought also has an indirect impact on air quality and health, which results in impacting crop failures, habitat loss, significant population migration, soil degradation, widespread livestock deaths, increased diseases such as polio, cholera, diphtheria, typhoid and tuberculosis (Barlow et al. 2016). Therefore, there is a need to consider sandstorms as a disaster and develop policies and guidelines accordingly.

Case study In March 2018, a thick sandstorm ravaged Khartoum in Sudan, pressuring the regulators to cancel flights and shut down schools in the city and other nearby towns. In the same way, cities situated in the south-west of Iran are among the most dusty cities in the world. Drought effects are non-structural and spread over a greater geographical region unlike damage from many other natural hazards (Shinoda et al. 2014).

14.4.4 Urban Air Pollution

Globally, increased transportation vehicles have been recognized as one of the major sources of air pollution in developing countries (Tiwari and Saxena 2021). It leads to a 40–80% range of overall air pollution. Hazardous substances in fossil fuels such as lead, nickel, iron, copper and other catalytically active compounds are taken into account on the surface of incomplete fuel combustion. Between tires of cars with road surfaces and between car brakes and the bowl of the wheel also add a lot of toxic substances to the air (Javadinejad et al. 2020). This is due to the lack of management to reduce adverse environmental impacts and other negative transport effects without reducing the benefits of mobility. This problem is now more urgent due to rapid urban development, especially with smart cities plans in case of India, which in turn is expected to dramatically increase the demand for travel.

The backbone of economic development is infrastructure. Various constructions, repair and demolition activities like clearing of land surface, operation of diesel engines, heavy equipment, burning of fuels and working with toxic materials also contribute to the sudden air pollution.

Case study Beijing, a typical representative of rapidly developing cities, declared war on air pollution in 1998. The obstacle was to find sustainable ways to boost air quality in fastest growing cities in the developing world. For 20 years, Beijing has been maintaining it. Its air quality has enhanced dramatically, and the lesson learned provides a blueprint for other growing urban cities to combat air pollution (Schleicher et al. 2012). Beijing's case offers hope that the problems associated with improving air quality can be tackled during the time of explosive growth and motorization (Hao and Wang 2005).

Beijing introduced more rigorous and strict steps to control air pollution in the year 2013. The fine particulate emissions ($PM_{2.5}$) had reduced dramatically by 35% and 25% in the neighbouring Beijing–Tianjin–Hebei area by the end of 2017. Over this period, Beijing's annual emissions were decreased by 83%, 43%, 55% and 42% for sulphur dioxide—SO₂, nitrogen oxides—NO_x, particulate matter—PM₁₀ and volatile organic compounds (VOC), respectively.

In response to extremely serious $PM_{2.5}$ emissions, the Beijing Government has introduced a range of initiatives and a number of policies and steps on the prevention and regulation of air pollution and has taken some concrete action to enhance air quality and to meet the national air quality requirements (Fig. 14.4).

Another significant factor impacting air quality is the weather conditions that restrict the density, dilution, diffusion, transport and transformation of ambient atmospheric contaminants. This further affects the distribution, composition and concentration of atmospheric pollutants.

14.4.5 Biomass Burning

Biomass burning appears to have a high potential to trigger local and regional air quality with higher levels of fine particulate matter (PM 2.5), which could worsen ambient air quality and endanger human health. This massive biomass burning has been increasing over recent years due to human activities. Stubble (crop/agricultural residue) burning is another source of air pollution (Gurjar et al. 2016; Abdurrahman et al. 2020; Saxena et al. 2021). It has been reckoned among the major contributors of air pollution, especially in South Asia. In India, the ambient particulate matter air pollution leads to staggering 670,000 deaths every year (Balakrishnan et al. 2019).

Case study According to the World Air Quality Report (2019), New Delhi (capital city of India) is among the worst polluted cities in the world—ranking 5th. The air quality spikes each year due to farmers burning millions of tons of crop wastes, which affects the ambient air quality of the capital city and NCR (National Capital Region) cities (World Air Quality Report 2019).



Fig. 14.4 Air pollution control measures adopted by Beijing, China

Regions of Delhi NCR are increasingly struggling with air pollution problems, especially during the late fall and early winter. Weather conditions seem to play a very important role in controlling air pollutants. Northern states of India experience a wintertime inversion layer of cool air that leads to the accumulation of pollutants near the ground level. Delhi experiences a decrease in temperatures and wind speeds from mid-September onwards, favouring the accumulation of air pollutants and rising outdoor pollutant levels. In the northern states like Punjab, Haryana, Uttar Pradesh, Uttarakhand and neighbouring areas, an estimated 20 million tonnes of paddy stubble is burned soon after the monsoon crop is harvested (Jaiswal et al. 2019). During festivals and events such as Diwali (despite some restrictions), the widespread use of crackers has helped to ooze poisonous air into residential areas, workplaces and playgrounds (Saxena et al. 2020b). Both of these variables have contributed significantly to the growing cases of non-communicable diseases (NCDs)—chronic respiratory and cardiovascular problems indicating that NCDs



Fig. 14.5 Seasonal PM2.5 air pollution in Delhi, India during the year 2016–2020 (in μ g/m³) (source: Berkeley Earth; IQAIR)

such as cancers and diabetes would lead to over 61% of deaths in India (Tigala et al. 2019).

Figure 14.5 shows the seasonal PM2.5 air pollution in Delhi during the year 2016–2020. Air pollution has led to 24,000 life losses in Delhi in the first half of 2020 including 3 months of the COVID-19 pandemic lockdown (Bhowmick 2020).

The EPCA, or Environment Pollution (Prevention and Control) Authority for Delhi and NCR, is a Supreme Court-appointed committee that has been in charge of all air pollution regulations in the city since 1998. The EPCA, in partnership with the federal and state governments, began implementing the Graded Response Action Plan (GRAP) in 2016. The GRAP has allowed expanded air pollution control so far in 2019. The Delhi government released a health alert/advisory for the first time to protect its residents from health-harming levels of air pollution. In addition to the actions under GRAP, the various governments in the city had declared the following actions: the odd–even rule. Under this scheme, private cars with odd-numbered registration plates were allowed to drive on odd days, and those with even-numbered plates were allowed to drive on even days from 8 a.m. to 8 p.m., except on Sundays. The city government increased the number of buses and metro-coaches and placed a temporary moratorium on surge pricing on app-based taxis like Ola and Uber (Jaiswal et al. 2019). Five key lessons Delhi (or any city facing air pollution) needs to learn to combat air pollution are shown in Fig. 14.6.



Fig. 14.6 Five key lessons Delhi needs to learn to combat air pollution

14.4.6 Forest Fire

Heatwaves combined with drought conditions can result in large wildfires that pollute the air around us (Jain et al. 2021). Forest fires are increasingly becoming a major contributor to air pollution over the past two decades as an indirect impact due to climate change (Vardoulakis et al. 2020). These wildfires may cause sudden rise in outdoor gaseous particles such as carbon monoxide, formaldehyde, nitrogen dioxide and acetaldehyde (Walter et al. 2020).

Case study As per the New South Wales (NSW) Department of Planning, Development, and Environment reports, the recent Australian wildfires have increased ambient air pollution (more than 14 times) during 2019–2020. Sydney has been blanketed in smoke as a result of destructive bushfires, resulting in excessive levels of air pollution. According to an overview of all Australian monitoring stations reported on the World Air Quality Index website, recent events have caused the Sydney area monitoring stations into the top ten of Australia's most polluted places in 2019. Since the beginning of November 2019, the suburbs of Liverpool and Rouse Hill, as well as Sydney area Richmond, have experienced between 9 and 15 days of unhealthy air quality, some of which were rated as "very unhealthy" by the index as shown in Fig. 14.7 (Walter et al. 2020).

In addition to bushfire smoke's deadly consequences, it is associated with more risks of illness and increased visits to hospitals due to respiratory diseases. Increased evidence also indicates that bushfire smoke may increase cardiovascular morbidity, psychiatric problems, adverse birth and eye outcomes.



Fig. 14.7 Sub-regional distribution of air pollution due to Australian bushfires as of December 2019

An extensive study conducted in western North America for understanding the impacts of climate and air quality using field and laboratory analysis during 2016–2019 revealed that the combination of dry and warm climate has produced more frequent fires and with larger magnitude in the U.S. West and Canada, irrespective of fire control practices (Fig. 14.8). The fire suppression practices have caused an increase in build-up of fuels, a breakdown in forest ecology and rise in risk at urban wild and bush boundaries. New regulations have been implemented to allow burning. Climate change along with deforestation will only increase this problem. The area burned is more prone to double for every degree rise in temperature due to global warming.

Unfortunately, there is no reliable way to minimize the impact of bushfire smoke on human health, except wearing facial masks and remaining indoors are widely advised (Vardoulakis et al. 2020). However, face mask's performance and effectiveness depend on their filtration ability, person wearing behaviour and characteristics, e.g. facial hair, or length and frequency of wearing a face mask. Sometimes, wearing a face mask can be painful when the bushfires happen in hot weather.

14.5 Air Pollution as Public Health Emergency

Approximately 91% of the world's population resides in areas where air quality exceeds WHO standards (WHO Ambient Air Pollution Report 2016). By reducing air pollution levels, countries and states can reduce the burden of various infectious diseases. The respiratory and cardiovascular effects on human health would become



Fig. 14.8 Relative risk of fires in the USA classified from low-risk to high-risk regions

better with the lower levels of air pollution, in both long run and short run (WHO Fact Sheet 2018). Air pollution is a major risk factor for preventable non-communicable diseases (NCDs). Air pollution has a wide variety of health consequences, from irritation of the skin and eyes to extreme neurological, cardio-vascular and respiratory disorders such as pneumonia, asthma, bronchitis, tracheitis, chronic obstructive pulmonary disease (COPD), allergic rhinitis, silicosis, emphysema and cancer. Because of the prolonged exposure to high emission levels, it also contributes to a rise in mortality rates (Abdurrahman et al. 2020).

Air pollution is responsible for 29% of all lung cancer deaths and illness, 17% of all acute lower respiratory infections, 25% of ischaemic heart disease, 43% of chronic obstructive pulmonary disease deaths and 24% of all stroke deaths worldwide (McHardy et al. 2020). Many airborne infectious diseases can be transmitted by dust. Researchers assume that inhaling dust particles in humid, dry weather destroys the mucosa of the nose and throat, providing optimal conditions for bacterial infection. Furthermore, dust particles containing iron oxides can increase the risk of infection (WMO 2020).

In addition to these direct effects on individual health outcomes, air pollution has indirect effects on other important health determinants. People cannot go outside to exercise or work during business hours because of dangerously high levels of air pollutants. Every aspect of a person's well-being is impacted. Children under 14 are particularly susceptible to air pollution. Growing children ingest more air per kg of body weight as opposed to adults. As a result, air pollution has the potential to harm any developing organ in a child's body, as well as trigger stunting and mental health issues. But, children are not the only ones who are at risk from air pollution; senior citizens and pregnant women are also at risk. So, air pollution cannot be considered an isolated environmental concern, as it is now becoming a major public health emergency. It is high time that air pollution control and management should become a part of state and city planning processes.

14.6 Existing Policies and Recommendations

There are numerous policies and programmes to resolve the air pollution problem that has been introduced across the world. The performance and effectiveness of these programmes, like any other policy indicator, have been dependent on cooperation and coordination among different stakeholders. Some of the major initiatives launched to combat sudden increase in air pollution in Indian cities have been mentioned below in Table 14.2.

However, the above-mentioned policies state that non-compliance is mostly due to a lack of appropriate monitoring and inspection elements and it also indicates that

S. no.	Government programmes	Tasks
1	National Clean Air Programme (NCAP), January 2019	National-level strategy to address the country's air pollution problem in a holistic manner with an aim of achieving 20–30% reduction in PM_{10} and $PM_{2.5}$ concentrations by 2024 keeping 2017 as the baseline year for the comparison
2	Comprehensive Action Plan (CAP), 2018	Identifying timelines and implementing agencies for specified actions for air pollution prevention, control and mitigation in Delhi and the NCR
3	Graded Response Action Plan (GRAP), January 2017	For the prevention, regulation and reduction in air pollution in Delhi and the National Capital Region. It distinguishes graded responses and implementing agencies for four AQI categories: moderate to poor, very poor, severe and severe + or emergency
4	SAMEER App, 2016	Where public access to air quality information is provided, as well as the ability to file complaints about air polluting activities
5	Environment Education, Awareness and Training Scheme	To raise environmental consciousness among all segments of society and mobilize public interest in environmental protection
6	Pollution under control (PUC)	Certificate of petrol/diesel-driven vehicles which test for carbon monoxide and hydrocarbon
7	Other policies	 Permission to use only pure diesel as fuel for vehicles with a limit of 500 ppm sulphur The use of non-polluting compressed natural gas CNG by buses and trucks as fuel Mandatory blending of 20% ethyl alcohol with petrol and 20% biodiesel with diesel

Table 14.2 Existing policies and recommendations for prevention, regulation and abatement of air pollution in India



Fig. 14.9 Seven-pronged strategic approach to develop a climate change resilient health system

training should be provided to the workers and daily inspection drives should be carried out.

Moreover, the National Health Adaptation Plan (NHAP) has been designed and developed by the National Institute of Disaster Management, Ministry of Home Affairs, Government of India, to create enabling environment, build the capacity of climate-resilient healthcare system, help institutions strengthen their preparedness and enhance the health system resilience in terms of universal health coverage: early warning, surveillance and community mobilization in the context of any health emergency risk management. The NHAP also proposes a seven-pronged strategic approach to achieve its goal of developing a climate change resilient health system (Fig. 14.9).

Indeed, while the air pollution problem has gained considerable political momentum, India has not been able to make any substantial progress in achieving the objectives set out in the NCAP and other policies and programmes. The momentum was further hampered by the onset of the coronavirus pandemic. The economic fallout of the national lockdown has delayed the objectives of lowering pollution from thermal power plants, transport and other sectors, and transitioning to cleaner energy processes.

14.7 Integrated Approach to Deal with Air Pollution as Disaster

There is an urgent need for a concerted approach consisting of the generation and compilation of well-designed and continuous data on air pollution: its effective representation and timely dissemination to allow a wide range of air quality policy decisions. Big data analytics provides a number of approaches to deal with data sources of monitoring, uncertain sensor data, real-time and predictive analytics, and spatiotemporal mapping and forecasting. Various statistical models have become useful for pooling threats across several locations and measuring spatial heterogeneity, resulting in better control of air quality and its effect on public health.

Despite the popularity of air pollution disasters and the lack of options for reducing contamination and exposure, citizens should seek out ways to protect themselves from pollutants inhalation and expect agencies to include guidelines or even provide preventive measures (Han and Naeher 2006). Face masks are one possible particulate emission intervention, but there is little evidence of their efficacy for community use.

There is an immense need for an integrated framework for dealing with air pollution disasters during emergency situations. Even during the times of emergency, various response and recovery actions were initiated in many countries. However, in India, there is an utmost requirement of an integrated framework where all governing and regulatory bodies should work together to curb air pollution emergencies as shown in Fig. 14.10.

14.8 Conclusion

Many cities around the world are rapidly leading the charge to solve the issue of air pollution and protect public health. It is high time that air pollution should be considered as a disaster and public health emergency. During any such disaster, the air quality management is dependent on the local community's preparedness and external aid. A concerted national initiative would be needed to achieve long-term changes in air quality. The collaboration and coordination at all levels of government are very much important for the effective implementation of various air pollutionrelated judicial orders, policies, guidelines and crisis management plans. There is an immense need for an integrated framework for dealing with air pollution as a disaster during emergency situations. There is an utmost requirement of an integrated framework on preparedness guidance and planning where all governing and regulatory bodies, stakeholders, businesses, households and communities should work together to curb air pollution emergencies.



Fig. 14.10 Framework for air pollution as disaster and its response mechanism

References

- Abdurrahman MI, Chaki S, Saini G (2020) Stubble burning: effects on health & environment, regulations and management practices. Environ Adv 2:100011
- Ashraf A, Butt A, Khalid I, Alam RU, Ahmad SR (2019) Smog analysis and its effect on reported ocular surface diseases: a case study of 2016 smog event of Lahore. Atmos Environ 198:257– 264
- Balakrishnan K, Dey S, Gupta T, Dhaliwal RS, Brauer M, Cohen AJ, Sabde Y (2019) The impact of air pollution on deaths, disease burden, and life expectancy across the states of India: the Global Burden of Disease Study 2017. Lancet Planet Health 3(1):e26–e39
- Barlow M, Zaitchik B, Paz S, Black E, Evans J, Hoell A (2016) A review of drought in the Middle East and southwest Asia. J Clim 29(23):8547–8574
- Bhowmick N (2020) In New Delhi, burning season makes the air even more dangerous. Can anything be done? https://www.nationalgeographic.com/environment/2020/11/new-delhi-burn ing-season-makes-air-even-more-dangerous-can-anything-be-done/
- Chandrappa R, Kulshrestha UC (2016) Air pollution and disasters. In: Sustainable air pollution management. Springer, Cham, pp 325–343
- Csavina J, Field J, Felix O, Corral-Avitia AY, Saez AE, Betterton EA (2014) Effect of wind speed and relative humidity on atmospheric dust concentrations in semi-arid climates. Sci Total Environ 487:82–90
- CSE (2020) Gas leak in Vizag. Factsheet. https://www.cseindia.org/gas-leak-in-vizag-10107
- Gurjar BR, Ravindra K, Nagpure AS (2016) Air pollution trends over Indian megacities and their local-to-global implications. Atmos Environ 142:475–495
- Hailwood M (2020) Release of hazardous vapours at LG Polymers chemical plant in Visakhapatnam, Andhra Pradesh, India. Loss Prev Bull (274):9–12
- Han X, Naeher LP (2006) A review of traffic-related air pollution exposure assessment studies in the developing world. Environ Int 32(1):106–120

- Hao J, Wang L (2005) Improving urban air quality in China: Beijing case study. J Air Waste Manage Assoc 55(9):1298–1305
- Harlan SL, Ruddell DM (2011) Climate change and health in cities: impacts of heat and air pollution and potential co-benefits from mitigation and adaptation. Curr Opin Environ Sustain 3(3):126–134
- Indoitu R, Orlovsky L, Orlovsky N (2012) Dust storms in Central Asia: spatial and temporal variations. J Arid Environ 85:62–70
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Jaiswal A, Knowlton K, Limaye V (2019) Air emergency in Delhi: a health crisis hits home. NRDC. https://www.nrdc.org/experts/vijay-limaye/air-emergency-delhi-health-crisis-hits-home
- Javadinejad S, Dara R, Jafary F (2020) Health impacts of extreme events. Saf Extrem Environ 2(2): 171–181
- Mackenzie J (2016) Air pollution: everything you need to know. NRDC. https://www.nrdc.org/ stories/air-pollution-everything-you-need-know
- Marchal V, Dellink R, Van Vuuren D, Clapp C, Chateau J, Magné B, Van Vliet J (2011) OECD environmental outlook to 2050, vol 8. Organization for Economic Co-operation and Development, pp 397–413
- McHardy JS, Aneja K, Cinà MM, Gostin LO (2020) Delhi in a chokehold: air pollution as a public health emergency. Health Affairs, Global Health Policy. https://www.healthaffairs.org/do/10.13 77/hblog20200130.710866/full/
- Rodríguez-Urrego D, Rodríguez-Urrego L (2020) Air quality during the COVID-19: PM2. 5 analysis in the 50 most polluted capital cities in the world. Environ Pollut 266:115042
- Sarkar S, Chauhan A, Kumar R, Singh RP (2019) Impact of deadly dust storms (May 2018) on air quality, meteorological, and atmospheric parameters over the northern parts of India. GeoHealth 3(3):67–80
- Saxena P, Naik V (eds) (2018) Air pollution: sources, impacts and controls. Cabi, Wallingford, pp 1–217
- Saxena P, Sonwani S, Sharma SK, Kumar P, Chandra N (2020a) Carbonaceous aerosol variations in foggy days: a critical analysis during the fireworks festival. Fresenius Environ Bull 29(8):6639– 6656
- Saxena P, Srivastava A, Verma S, Singh L, Sonwani S (2020b) Analysis of atmospheric pollutants during fireworks festival 'Diwali' at a residential site Delhi in India. In: Measurement, analysis and remediation of environmental pollutants. Springer, Singapore, pp 91–105
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Schleicher N, Norra S, Chen Y, Chai F, Wang S (2012) Efficiency of mitigation measures to reduce particulate air pollution—a case study during the Olympic Summer Games 2008 in Beijing, China. Sci Total Environ 427:146–158
- Schwela DH, Haq G (2020) Strengths and weaknesses of the WHO global ambient air quality database. Aerosol Air Qual Res 20(5):1026–1037
- Shinoda M, Nandintsetseg B, Nachinshonhor UG, Komiyama H (2014) Hotspots of recent drought in Asian steppes. Reg Environ Chang 14(1):103–117
- Singh N (2020) Air pollution exposure studies related to human health. In: Saxena P, Srivastava A (eds) Air pollution and environmental health. Environmental chemistry for a sustainable world, vol 20. Springer, Singapore. https://doi.org/10.1007/978-981-15-3481-2_7
- Sonwani S, Madaan S, Arora J, Suryanarayan S, Rangra D, Mongia N, Vats T, Saxena P (2021) Inhalation exposure to atmospheric nanoparticles and its associated impacts on human health: a review. Front Sustain Cities 3(690444):1–20
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1–2):39–50

- Tigala S, Sharma AR, Sachdeva K (2018) Health risk assessment due to biomass smoke exposure in Indian indoor environment: an empirical approach using lung deposition model. Sci Total Environ 640:935–942
- Tigala S, Sharma AR, Sachdeva K (2019) A system dynamics approach to understand respiratory health risk in rural population. Indian J Public Health Res Dev 10(4)
- Tiwari S, Saxena P (2021) Air pollution and its complications. Springer-Nature, Singapore, pp 1– 178
- UNEP (2019) Five ways in which disasters worsen air pollution. United Nations Environment Programme, News and Stories. https://www.unenvironment.org/news-and-stories/story/fiveways-which-disasters-worsen-air-pollution
- Vardoulakis S, Marks G, Abramson MJ (2020) Lessons learned from the Australian bushfires: climate change, air pollution, and public health. JAMA Intern Med 180(5):635–636
- Walter CM, Schneider-Futschik EK, Knibbs LD, Irving LB (2020) Health impacts of bushfire smoke exposure in Australia. Respirology 25(5):495–501
- WHO Ambient Air Pollution Report (2016) Ambient air pollution: a global assessment of exposure and burden of disease. https://apps.who.int/iris/bitstream/handle/10665/250141/97892415113 53-eng.pdf?sequence=1&isAllowed=y
- WHO Fact Sheet (2018) Ambient (outdoor) air pollution. Key Facts. https://www.who.int/en/newsroom/fact-sheets/detail/ambient-(outdoor)-air-quality-and-health
- WMO (2020) Sand and dust storms. World Meteorological Organization. https://public.wmo.int/ en/our-mandate/focus-areas/environment/SDS
- World Air Quality Report (2019) World's most polluted cities 2019 (PM2.5). https://www.iqair. com/world-most-polluted-cities



Cost-Effective Technologies for Control of Air Pollution and Atmospheric-Related Extremes

15

Amara Aziz, Hania Maqsood, and Sukriti Kapur

Abstract

Air pollution is a phenomenon harmful to the ecological system and the standard conditions of human existence. In the face of increasingly serious environmental pollution problems, various studies have been conducted on improving air quality better management of air pollution. Air pollution and higher rate of burning of fossil fuels consequently result into adverse weather patterns and atmospheric extremes like sand and dust storms (SDS) and wildfires. This chapter reviews some air pollution control technologies and their cost-effectiveness. We also provide an overview of atmospheric extreme events.

Keywords

Air pollution control · SDS · Atmospheric extremes · Cost-effective technologies

15.1 Introduction

Air quality has severely degraded over the years, causing profound implications for human beings, materials, and ecosystems (Vlachokostas et al. 2011). According to World Health Organization (WHO), more than 80% of people living in metropolitan cities are exposed to air quality levels above the permissible limits. Rapid urbanization and industrialization have lead to an increase in primary atmospheric pollutants,

A. Aziz · H. Maqsood

Institute of Environmental Engineering and Sciences (IESE), National University of Science and Technology (NUST), Islamabad, Pakistan

S. Kapur (⊠) University of California, Irvine, Irvine, CA, USA

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_15

such as carbon monoxide (CO), particulate matter (PM), nitrogen oxides (NOx), volatile organic compounds (VOCs), polycyclic aromatic hydrocarbons (PAHs), ozone (O₃), and sulfur dioxide (SO₂), especially in developing countries (Manisalidis et al. 2020; Saxena and Sonwani 2019a; Sonwani and Saxena 2022). Several studies have been conducted to study the rise in atmospheric pollutants over the years. Air pollution is known to cause over 0.5 million deaths in developing Asian countries. In India and China, an increase in PM2.5 by 10 μ g/m³ has been associated with an increase in mortality by 0.6% (Chung et al. 2011). Lelieveld et al. (2001) studied air pollution from south and Southeast Asia. They concluded that emissions during winter monsoon were considerable over this region, with increase in NOx, ozone, BC, and sulfates predicted for the future, owing to an increase in fossil fuel emissions. Biomass burning also is severely known to contribute to NOx emissions in the tropics and subtropics of South Asia (Saxena et al. 2012).

A wide range of air pollution control devices can be used to reduce air pollutants for the sake of prevention of their adverse effects to human health and the environment. Several technological applications aid this purpose. Gravitational settling chambers, incinerators, cyclone separators, selective catalytic reduction systems, electrostatic precipitators, bio filters, scrubbers, and fabric filters are the major technologies used to prevent air pollution (Saxena and Sonwani 2019b). The usage decision of the appropriate control devices depends on the air pollutants type and condition at the sources. If the amount of pollutants emitted to the atmosphere is decreased, the atmospheric concentrations of the pollutants will also decrease. Consequently, exposure to air pollution drops and adverse effects of air pollution on human health and the environment are taken under control.

This chapter aims to discuss common technologies for air pollution control and measurement. We discuss commonly used technologies for air pollution management and their cost-effectiveness.

15.2 Air Pollution Management

Air pollution management is necessary for the elimination or reduction of pollutants to acceptable levels. The presence of airborne gaseous pollutants, suspended particulate matter, and many more in the atmosphere are capable of causing adverse effects on human health, animals or plants, and damage to the environment (e.g., climatic modifications). It needs adequate control because the growing pollution sources in the modern world may lead to irreversible destruction to the environment and humanity.

Most air pollution control systems involve a combination of numerous control techniques. Typically, a blend of both technology and administrative controls is used. However, when there are complex pollution sources, more than one type of technological controls are used.

The selection of the appropriate controls depends upon the problem at hand. Some typical questions that need to be kept in mind before selecting the control methods are:

- What is emitted, and in what concentration?
- What is the most susceptible target?
- What are acceptable short-term exposure levels?
- What are acceptable long-term exposure levels?
- What combination of controls must be selected to ensure that the short-term and long-term exposure levels are not exceeded?

There are number of steps to follow for the pollution control selection methods:

15.2.1 Step 1: Emissions Definition

The first step is to determine the possible pollutants from the source. The next step involves estimating the concentration of each pollutant. These steps are keys to designing a control program.

15.2.2 Step 2: Define the Target Groups

All vulnerable targets must be identified. This includes all flora, fauna, humans, and property. In each case, the most vulnerable member of each group must be identified.

15.2.3 Step 3: Determination of Acceptable Exposure Levels

Establish an acceptable exposure level for the sensitive target group. Set up short- or long-term exposure levels depending upon the pollutant.

15.2.4 Step 4: Select Controls

Every pollutant is monitored to guarantee that it does not surpass the acceptable exposure level. Additional controls must be added, if it exceeds the acceptable level.

Rather than considering emissions from just mobile or fixed sources, management of air pollution also needs to involve reflection of supplementary factors (i.e., topography and meteorology) that must integrate into a complete program. For example, ground-level concentrations resulting from similar pollutant emission can greatly affected by the meteorological conditions. The sources of air pollution may disperse over a community or a region and their effects may be felt by, or their control may include, other than one administration. Moreover, air pollution is an issue beyond boundaries. Emissions may be generated at a different source and induce effects in another region by means of long-range transport.
15.3 Technological Control Measures for Air Pollutants

The basic equipment types used for the control of air pollution are:

- Cyclones (inertial separators), fabric filters (baghouses), wet collectors (scrubbers), and electrostatic precipitators for particulate matter
- Incinerators, adsorption units (adsorption beds), and catalytic (catalytic combustion) for gaseous pollutants

15.3.1 Particulate Matter Control

15.3.1.1 Cyclone

The cyclone is renowned technology that is primarily used for coarse and mediumsized particles collection. It works by downward forcing of a gasified suspension. By centrifugal force, the particles move in the outward direction by colliding with the external wall and then slide toward the bottom of the cyclone. There, the gas converges its downward spiral and then moves upwards in an inner smaller spiral. Clean gas is released from top of the cyclone and the collected particles are ejected from bottom of the cyclone by a pipe closed by a rotary valve. The cyclone collector is shown in Fig. 15.1.

Effective recovery of the products can be done using cyclones, finding a wide variety of applications in vaporous cleaning fields. These devices are low-maintenance, inexpensive, and work at high temperature and pressure. Cyclones show an efficiency of 90% for particles bigger than 10 µm, but have lower efficiency



Fig. 15.1 The cyclone collector (Source: J. He et al. 2017)



Fig. 15.2 The wet scrubber chamber (Source: EPA Fact Sheet 2016)

for smaller sized particles (Leith and Jones 1997). Additionally, removing sticky particles having higher moisture content cannot be achieved with the cyclone method.

15.3.1.2 Wet Scrubber

Wet scrubber systems are used for the control of mists, fumes, heavy metals, acid gasses, suspended dust, and trace organics. The device contains multiple scrubbers, attached together, which allow for an effective removal efficiency as compared to a single unit (Cooper and Alley 2011). The wet scrubber is shown in Fig. 15.2.

A direct contact is established between the adsorbent liquid (mostly water) and the particles. Based on the contact between the two, the particles leave the gas phase to be adsorbed onto the liquid phase. The adsorption can either be liquid or gaseous phase, depending on the properties of the particle. The removal efficiency depends on a multitude of factors, such as:

- · Pollutant solubility in the selected scrubbing liquor
- Pollutant concentration in gas phase
- · Gas and liquid phases flow rates
- Contact surfaces of gas-liquid phase
- · Recycling of the solvent and liquor shedding efficiency

15.3.1.3 Electrostatic Precipitators

One of the most efficient and famous means of air pollution control is an electrostatic precipitator. It is a particle control method in which electrical forces are used for the movement of particles out of the gas flow stream on to the collector plates (Treybal 1980). Its processes include:

- The particles ionize like liquid droplets or dry dusts, in air that is contaminated.
- The charged plates attract the oppositely charged particles.
- The deposited particles removal from the plates (Fig. 15.3).

The major difference between other scrubbing methods and ESPs is that in electrostatic precipitators, the parting forces are electrical and are directly applied to the droplets or particles themselves; whereas in other methods, the separation forces are typically applied indirectly with the help of contaminated air system. Hence, ESP is capable of small particles or liquid droplets removal at a high efficiency with low energy consumption or low-cost and low-pressure drop (White 1975).



Fig. 15.3 Schematic representation of an electrostatic precipitator (Source: Becker et al. 2005)

15.3.1.4 Fabric Filtration

Fabric filtration is a physical technique in which a gas stream comprising of solids passes through a permeable fabric medium holding the solids. This process might function in a group or semicontinuous mode in order to remove the solid particles, which are retained, from the filter medium. These systems are also designed to function in a continuous way.

The polluted stream of gas is passed through filter bags that are parallelly positioned, and retain the solid particles onto them. The fabric filter is categorized in two simple groups, depending on the properties of fabric, i.e., felt and woven. Felt fabrics are commonly used in high cleaning systems, whereas woven media are used in low-energy systems. These devices operate at an efficiency close to 100% and thus are effective in pollutant removal.

The advantages and disadvantages of particulate matter control techniques are summarized in Table 15.1.

15.3.2 Gaseous Pollutants Control

15.3.2.1 Adsorption

In adsorption, the molecules of a fluid stick to the surface of a solid material called adsorbent. This method is commonly used to recover volatile solvents, i.e., benzene, toluene, and chlorofluorocarbon, process gas streams drying, and control the odor. The molecules in the airstream are captured and removed, even if they are present at low concentrations. This process is exothermic. Chemical adsorption or chemisorption encompasses an actual chemical bonding by the reaction of adsorbate with the adsorbent. This leads to generation of new chemical bonds such as at the surface of the adsorbent.

15.3.2.2 Absorption

Absorption is a chemical or physical process in which ions, molecules or atoms go in the bulk phase. It involves the transmission of a gaseous pollutant from the air in the water. The water must be capable of either serving as a solvent for the pollutant; moreover, to capture it with chemical reaction.

15.3.2.3 Condensation

Condensation is a separation process for the conversion of one or more volatile components of a mixture of vapor to a liquid with the help of a saturation process. By adequate lowering and increasing the pressure and temperature respectively, any volatile components can be converted to liquids. It requires less installation cost, little equipment, and less maintenance, but the problem of water disposal, lower efficiency, and further treatment is still needed.

15.3.2.4 Incineration

Incineration (thermal oxidation) is a largely used technique for the control of air pollutants such as VOCs. It is the process of converting the VOCs and other gaseous

Advantages	Disadvantages			
Cyclone				
Low capital cost	Comparatively low efficiencies for particles			
Minor maintenance problems	collection with smaller size than 10 mm			
 Ability to operate at high 	 Not used for sticky materials 			
temperature				
 Small spaces required 				
Wet scrubber				
Low capital cost	Improper water/effluent disposal			
Require small space	Problems of corrosion			
 Collect sticky and particles gases 	 High pressure drop difficulties 			
 Process at high-temperature and 	• Solid accumulation problems at the wet-dry			
high-humidity gas streams	interface			
Highly efficient for collecting fine	 Comparatively high maintenance costs 			
particles				
Electrostatic precipitation (ESP)				
Low maintenance and operation	High capital cost			
costs	 Highly sensitive to variations in gas stream 			
Higher collection effectiveness of	• Difficulties with particles, with extremely high or			
coarse and fine particulates	low resistivity			
 Low consumption of energy 	 Ozone production as by-product 			
 Dry dust collection 	 Require large space for installation 			
 Pressure drop is lower 	Highly trained maintenance personnel required			
Collect capacity is higher				
Fabric filtration				
Simple operation	Explosion problems			
Simple maintenance	 Require high maintenance costs 			
 No corrosion issues 	 Difficult operation at high temperature 			
Highly efficient in collecting coarse	Fabric treatment needed after removal process			
and fine particulates	 Respiratory protection required for fabric 			
Insensitivity to gas stream variations	replacement			
Highly efficient in collecting smoke	 Medium pressure-drop required 			
and gaseous contaminants				

Table 15.1 Advantages and disadvantages of particulate control methods (Source: Buonicore and
Davis 1992)

hydrocarbon pollutants into carbon dioxide and water. VOCs and hydrocarbon fume incineration is typically completed in a special incinerator called an afterburner; and a final disposal procedure converts the VOCs to water, carbon dioxide, and inorganic gases. There are two widespread methods of incineration, i.e., catalytic incineration and thermal incineration.

15.3.2.5 Carbon Sequestration

Lower levels of carbon dioxide in the atmosphere can be achieved by effective energy use and using alternate sources of energy (e.g., nuclear, wind, tidal, and solar power). Furthermore, making use of carbon sequestration for storing carbon dioxide underground, in forests and oceans is of utmost importance. Sequestration in forests and oceans depends on natural procedures such as forest growth (Uprety and Saxena 2021). Storage of carbon dioxide underground; a technique which is also called geo sequestration or carbon capture and storage includes driving the gas directly into underground geologic "reservoir" strata. This would involve the removal of carbon dioxide from power plant flue gases, which is an expensive process.

15.3.2.6 Biological System

For controlling air pollutants, mainly volatile organic compounds (VOCs), the biological phenomenon includes the utilization of microbes. It involves the immobilization of air pollutants on a biologically active solid surface. The key role of the process is that the microorganisms used the gaseous contaminants as a source of nutrients and energy, which consequently destroyed and transformed the pollutants into harmless metabolic by-products, i.e., carbon dioxide and water. The procedure through which microorganisms transform or destroy pollutants includes:

- Initially the microbial liquid film on which microbes grow, the gaseous pollutants must be absorbed into the cells of microbes and then it will be metabolized.
- The by-products, i.e., CO₂ must be released from the cells and disperse outside the liquid film.

The key elements required for the design and implementation of biological processes in order to control air pollutants include (Crocker 2000):

- 1. Identification of the pollutant type and its amount in air stream
- 2. Utilization of the appropriate microbial culture along with its suitable medium
- 3. Moisture control
- 4. Suitable residence time for air stream flow rate
- 5. Maintaining pH, nutrient levels, and temperature

Biological mechanisms include Bio filter and Bio scrubber.

15.3.2.7 Bio Filter

It is a biological process which utilizes an organic or synthetic media to culture and nourish the microbes without the utilization of an aqueous flush system.

15.3.2.8 Bio Scrubber

This process uses an inorganic or synthetic media to provide a structural base for culturing the microorganisms requiring a continuous water level or an irregular amount of carbon nutrient that supports the microbes. This method offers numerous advantages such as effective elimination of contaminants, little or no end product pollutants, simple installations, and cost-effectiveness. However, there is lesser suitability at higher concentration levels, a need for maintaining moisture, and the possibility of becoming blocked by particulate matter or biomass growth. A larger area for plant installation is also required.

15.3.3 Control of Air Pollution by Cost-Effective Technologies

Air pollution is becoming inevitable with greater development and urbanization around the globe, which consequently causes many human health issues, such as respiratory and cardiovascular disorders. Controlling of air pollution issues is hard due to its strong flow ability.

Among all types of control technologies of pollutants, comprising of microbial bioremediation, **phytoremediation** stands out for its advantages resulted from self-maintaining, soil stabilization, and other advantages to meet higher public support (Doty et al. 2007).

15.3.3.1 Phytoremediation of Air Pollutants

Particulate Matter (PM)

Vegetation has been used to protect dust in several areas of the world, by a technique known as phytofiltration. This is a cost-effective technique that involves use of plants to reduce toxic pollutants in air. Leaf area, orientation, and shape are some of the structural features that help in effectively capturing dust from the surroundings.

15.3.3.2 Inorganic Pollutants

NO_x

Oxides of nitrogen are considered to be one of the precursors of photochemical reactions taking place in the atmosphere. Deposition, either wet or dry, on the roots or foliage can influence plant systems. Nitrogen Dioxide, on entering plants, is metabolized to organic components such as amino acids. Nitrate, nitrate reductase, or other enzymes in plants can potentially play an important role in this process (Saxena and Srivastava 2020).

SO₂

Approximately 70% of SO₂ in the air comes from fossil fuel burning. SO₂ reaches into plants primarily by stoma, and can be employed in a reductive sulfur cycle in plants. They are transformed into lower oxidation state of sulfur oxides in cell walls. Adenyl-5-phosphosulfate, a protein carrier carries sulfite and sulfide as the intermediates in the reduction pathway. The end products are cysteine or other organic components.

15.3.3.3 Organic Pollutants

Formaldehyde

Formaldehyde, a universal air contaminant, is highly detrimental and is classified as a mutagen and carcinogen constituent. In the 1980s, a NASA's study in the USA discovered that lower concentrations of formaldehyde in atmosphere can be removed by plant leaves, while higher concentrations can be first filtered by activated carbon and then the plant roots and related microbes degrade and integrate remained toxins (Wolverton 1988). At sufficiently low concentrations of formaldehyde (8.5 mg m⁻³), shoots of spider plants can absorb free sugars, lipids, cell-wall components, organic acids, and amino acids. Formaldehyde can be oxidized and then degraded in C1 metabolism in phytodegradation. In soybean cell cultures, serine and phosphatidylcholine are known to be the main metabolic products for formaldehyde (Saxena and Ghosh 2015).

15.4 Air Pollution, Climate Change, and Atmospheric Extremes

There is a natural connection between climate change and air pollution since air pollutants and greenhouse gases originate from the identical source, i.e., fossil fuel combustion (Jacob and Winner 2009; Dennekamp and Carey 2010). In reality, combustion processes release greenhouse gases, like methane (CH₄), carbon dioxide (CO₂), nitrous oxide (N₂O), and air pollutants, like sulfur dioxide (SO₂), particulate matter (PM), carbon monoxide (CO), and nitrogen dioxide (NO₂).

When it comes to defining air quality, pollution and climate change trends are commonly seen to counterbalance each other. While a lot of countries are adopting pollution control strategies to reduce their emissions, climate change often leads to an increase in pollutant levels (Dennekamp and Carey 2010).

The correlation between air quality and climate is notable for ozone. Ozone levels are directly driven by weather, since ozone-generating photochemical reactions of air pollutants (nitrogen oxides; methane; volatile organic compounds, VOCs) need high temperatures and bright sunshine, conditions typical of summer months (Engardt et al. 2009). High temperatures are often associated with dry weather conditions, which significantly contribute to high ozone levels during heatwaves through a drought stress on vegetation that inhibits stomatal uptake of ozone (Jacob and Winner 2009; Ordóñez et al. 2005).

Anthropogenic emissions of ozone precursors are expected to decrease in developed countries, but not in the developing ones. A large increase in ozone concentration is expected to take place in Southern Europe, owing to climate change. This increase is likely to be higher than the reduction in ozone precursors, causing an overall increase. Background ozone concentrations are also predicted to increase due to a higher frequency of forest fires (Engardt et al. 2009).

Particulate matter, based on its chemical composition, is affected differently by climatic factors. Higher global temperatures result in a reduction on nitrate PM concentrations, but also increase sulfate aerosols, owing to faster SO₂ oxidation (Jacob and Winner 2009; Sonwani et al. 2021; Sonwani and Saxena 2021). However, under climate change, the nitrate burden is predicted to increase along with all other aerosol species, except sulfates. Over Europe, the predicted increase in overall aerosol concentrations will be by 20–40 μ g·m⁻³ relative to the present-day values (Liao et al. 2006). Changes are driven by precipitation changes since wet deposition represents the primary PM sink, but also by increments in water vapor that increase oxidation of SO₂ and sulfate concentrations. Air stagnation conditions represent another important factor that will increase PM concentrations in polluted regions

(Chidthaisong 2005; Jacob and Winner 2009). Climate change-related increase in wildfires will also influence particulate levels (Krawchuk et al. 2009; Spickett et al. 2011); an increased fire risk in Mediterranean countries is expected, especially in areas with high forest cover (Chidthaisong 2005). The 2003 heatwave in Europe is emblematic since it was associated with both record wildfires and high PM levels (Jacob and Winner 2009). Particulate matter with a diameter less than 2.5 μ m, commonly known as PM_{2.5} is a major concern during wildfires. These particles have a long lifetime (in the order of days) and can travel larger distances (Naeher et al. 2007). Owing to climate change, wind patterns are predicted to change in the future, which will impact transport of particulate matter (Chidthaisong 2005).

Several primary and secondary pollutants have the potential to increase global warming, either directly or indirectly. Pollutants that are directly released as a result of combustion processes are known as primary pollutants, which include carbon monoxide, nonmethane VOCs, nitrogen oxides, sulfur dioxide, black carbon, and organic carbon aerosols. Secondary pollutants (e.g., ozone) on the other hand, are not emitted from a source, but formed in the atmosphere. Due to large quantities of greenhouse gases emitted, wildfires are known to directly contribute to global warming (Viswanathan et al. 2006). CO, NMVOCs, and NO_x cause an increase in warming, whereas nitrates, organic carbon, and sulfates have a cooling effect (Smith et al. 2009).

As Earth's climate has warmed, a new pattern of more frequent and more intense weather events has extended around the globe. These extreme events are identified based on the past record of weather in a specific region. They consider extreme weather events to be those that produce unusually high or low levels of rain or snow, temperature, wind, or other effects. Usually, these events are deliberated extreme if they are unlike 90 or 95% of comparable weather events that happened before in that same area.

The extreme events give rise to global warming that can further contribute to the intensity of heat waves by enhancing the chances of very warm days and nights. Warming of the air also increases evaporation that can exacerbate drought. More droughts result in dry fields and forests that are inclined to catching fire, and increasing temperatures mean a lengthier wildfire season. Global warming also increases water vapor in the atmosphere, which can lead to more repeated heavy snowstorms and rain.

A moister and warmer atmosphere above the oceans makes it probable that the toughest hurricanes will be stronger, probably be larger and produce more rainfall. Additionally, global warming causes sea level to rise, which increases the amount of seawater, along with more rainfall, that is pushed on to shore during coastal storms.

Understanding the impacts of global warming on extreme weather is significant because it can help notify choices about risk management. For example, if a community well aware of the fact that the increased rainfall due to global warming has converted a 500-year-old flood to a 100-year-old flood, different choices will then be made regarding land management, construction, or floodwall buildup.

15.5 Control Technologies for Atmosphere Extreme Events

15.5.1 Wildfire Control Technologies

Wildfires are a complex mixture of particulates, liquids, and gaseous components. The impact of fires is generally calculated regarding the PM2.5 concentrations in the atmosphere as it is mainly present in fire smoke (Finlay et al. 2012; Dennekamp and Abramson 2011), and is considered as the root cause for health-related issues. Fine particles are generated directly during the burning process and later resulted from the released gases by condensation and atmospheric chemical reactions. Throughout the wildfires, the concentration levels are mostly exceeding the air quality standards, i.e., 24-h mean PM10: 50 μ gm⁻³ and 1-year average mean PM2.5: 25 μ gm⁻³ (Naeher et al. 2007; Finlay et al. 2012; Dennekamp and Abramson 2011). The atmospheric extreme event is recognized in regard to a certain threshold, i.e., the 99th percentile of particulate distribution. This allows extreme events to be differentiated from conditions in which already present pollution sources are dominant.

Researches on health impacts of wildfire usually change for certain atmospheric conditions. Meanwhile the ambient temperature is typically higher during wildfires; there is also a possible opportunity for the co-occurrence of heatwave and wildfire situations which has huge effect on public health. This interaction can be described by fluctuations in temperature determined by smoke constituents, sunlight, emissions, water vapor, and related with other air pollutants (Naeher et al. 2007). Moreover, atmospheric conditions effect smoke dispersal and transportation (Viswanathan et al. 2006). Other wildfire impacts comprise visibility damage and ozone production (Azevedo et al. 2011); these aspects may also interact with PM concentrations in detecting health impacts (Ovadnevaite et al. 2006).

Generally, the present indication proposes impacts of wildfires mainly on public health. The greatest operational challenge in these researches regards the evaluation of exposure; in most cases, PM concentrations from air quality monitoring units are utilized, but these are restricted to specific areas and are, thus, poorly representative of actual exposure. The recent method of using burnt areas from satellite data appears to be more effective in detecting high-risk parts. Furthermore, wildfires are occasional events and their impact depends on regional settings; hence, results are not simplified till date no proper consensus is developed to evaluate the exposure of dispersal process of smoke (Henderson et al. 2011).

Wildfire causes chaos and severe destruction of forestland and wildlife. It also has severe impacts to humans and buildings. Traditional control technology utilizes fire-retardant chemicals which not only pollute the atmosphere but also harmful to the marine wildlife. But the new control technology is better in terms of environmental protection but it is also cost-effective in comparison to the old technology. The use of frozen carbon dioxide which is commonly known as dry ice and it is abundantly available in millions of tons in many areas of the world. This control technology is far cheaper and environmentally friendly as compared to the use of fire-retardant chemicals (De et al. 2019).

15.5.2 Unmanned Aerial Vehicles (UAVs)

In recent years, many Unmanned Ariel Vehicles (UAVs) have been employed to combat forest fires. This has proved to be an efficient substitute for piloted aircrafts in times of such emergencies (Yuan et al. 2015).

15.5.3 Wireless Sensor Network (WSN) Technique

Global attention is being paid to wireless sensor network (WSN) for quick and easy indications of wildfires. WSN is capable of distinguishing between smoke of fake and real fires, indicating alarms and communicating distress signals. The sensor can indicate the presence of CO₂, NO₂, temperature, pressure, and relative humidity in the atmosphere, and is a useful device for wildfire detection (Okokpujie et al. 2019).

15.5.4 Spacecraft Technique

Advanced high resolution radiometers and imaging spectroradiometers are often used for timely response on wildfires (Nakau et al. 2006).

15.5.5 High-Tech Sensor and Camera Devices

High-tech sensors are often able to identify the nature of the smoke, temperature of fires, which makes them effective for early identification of wildfires. They are able to contact the control room on the amount of smoke in the area by means of different algorithms, and help in controlling the fire (Alkhatib 2014).

15.6 Sustainable Control Technology for Atmospheric Extreme Events

15.6.1 Carbon Dioxide Technology

Dry ice (solid CO_2) and liquefied CO_2 can help greatly control wildfires by cutting off oxygen supplies. Immediate clearing of areas around the fire and covering them with 2–3" thick chunks of dry ice and liquefied CO_2 are recommended. Liquefied CO_2 from helicopters, when sprayed on such fires, creates a thick layer where oxygen is displaced. To protect the household communities within 200 m of the fire lines, dry ice chunks need to be spread on the inside, outside as well as the roof of the vacant house. Liquefied or CO_2 gas can be spread around the house, and dry inflammable stuff removed. However, a disadvantage of this method is that dry ice is not produced cost-effectively and we may need to look into other technologies to combat such large-scale disasters (De et al. 2019). Desert Dust Storms Control Technology: Deserts are a known cause of dust storms globally, with North Africa (the Sahara) being the main contributing area (over 50%). Dust particles derive from the earth's crust and affect air quality, in particular PM10 and coarse particulate (PM2.5–10) levels (De Longueville et al. 2013; Karanasiou et al. 2012). Dust storms, similar to wildfires, are transported over large distances, following specific trajectories and changing chemical composition on the way. These episodes can cause PM concentrations to peak in areas having lower emissions (Niemi et al. 2009). For example, during Saharan dust events, PM_{10} concentrations surpass 100–150 µg·m⁻³ in rural areas, much higher than the EU daily limit (Middleton et al. 2008).

Dust storms from North Africa have a significant impact on air quality and human health, especially in Europe and the Mediterranean (De Longueville et al. 2013; Karanasiou et al. 2012). PM on dust days is proven to be higher than nondust days, as confirmed by mortality numbers or hospital admissions for respiratory causes (Jiménez et al. 2010; Samoli et al. 2011; Perez et al. 2012). On the contrary, Athens saw a greater effect on nondust days, while Cyprus observed no effect (Middleton et al. 2008). Studies from Asia and Australia evaluated impacts of dust storms on respiratory diseases such as COPD, asthma, pneumonia, etc. However, evidence relating hospital admissions to particulate matter levels is largely contrasting and unknown (De Longueville et al. 2013; Karanasiou et al. 2012).

It is also important to take into account meteorology when assessing the impacts of wildfires. Cyclones and anticyclones, along with rainfall, may cause long-range dispersion of air masses (Niemi et al. 2009). Dust storms tend to occur during high ambient temperatures and high ozone levels; however, not many studies have quantified the impacts of dust storms on human health, except one (Thalib and Al-Taiar 2012).

Analyzing the effect of dust storms on the environment has several challenges, as specific mortality or morbidity cannot be analyzed due to the low frequency of dust days. Additionally, in urban regions, dust storms are underestimated many times due to the presence of anthropogenic sources in the area (Karanasiou et al. 2012).

15.7 Climate Implications of Sand and Dust Storms (SDS)

Sand and dust storms have drastically damaged our environment by suspending tiny sand and dust particles, called as dust aerosols, in the atmosphere. Sand and dust storms have local and global effects on climate changes, while climatic changes in turn can alter the location and intensity of dust sources in the world. The threats caused by SDS are of great importance for environment as well as for public health because wind erosion can occur in most environments and desert dust events often involve long-range transportation of air particles ranging from 10⁶m of distance (Akhlaq et al. 2012). So, environmental pollution control technologies are designed to alleviate SDS hazards for disaster risk reduction and for SDS impact mitigation. Although many SDS hazards are well-known, the processes involved and their impacts are not all equally understood. Development and implementation of

mitigation policies, in order to address larger impacts of SDS mainly include transboundary pollution, are needed to be designed (Ginoux et al. 2001).

15.8 SDS Hazards

Sand and dust storms generate various hazards for humans and these are categorized according to the three main procedures of the wind erosion mechanism based on entrainment of particulates, their transportation, and their deposition (Middleton 2017). The SDS hazards are soil loss, nutrient loss, crop roots exposure, and undermining structures.

15.9 Control of Wind Erosion on Cropland

When the top most layer is displaced, it signifies as soil erosion mainly caused by wind and it is highly harmful to crop yield and productivity (Larney et al. 1998). Thus, various methods are designed to control wind erosion and these generally classified into three categories:

- Soil management techniques—Several soil management practices help provide a sustainable seed bed for controlling weeds and facilitating plant growth (Triplett and Dick 2008).
- **Protective barriers**—Walls or fences placed at right angles to the wind direction help break the erosive force of winds and prevent wind erosion (Wang et al. 2010).
- Agronomic measures (crop management practices)—An effective rule for controlling wind erosion is to maintain sufficient vegetative cover. Using residues from harvested crops can also help protect soil from wind (Skidmore 1986).

15.10 Control of Wind Erosion on Rangeland

Methods for controlling wind erosion on rangelands are largely comprised of preventive measures designed to reduce the pressure of grazing. Livestock may be excluded from pastures, a ban often enforced with fencing (Vetter 2005).

15.11 Controlling Blowing Sand and Mobile Desert Dunes

Mechanical removal, dissipation, reduce sand supply, and deposition promotion are the control measures for blowing sand and mobile dunes (Watson 1985).

15.12 Control of Wind Erosion at Mining Operations

Wetting mine trails, covering or capping with gravel, topsoil or synthetic materials, and erection of protective barriers as windbreaks are some of the techniques employed in this process. Renewable biopolymers and chemical suppressants developed from petroleum are widely used for mitigative purposes, in regions where water is at a premium (Chen et al. 2014).

15.13 Impact Mitigation of SDS and Its Policies

The encounters related to SDS have resulted in many policies in many areas of the world, specifically in those areas that have experienced issues related to wind erosion. The importance of these policies has strong focus on SDS at all levels. These include UNCCD Regional and National Action Plans and international air pollution policies (e.g., The Convention on Long-Range Transboundary Air Pollution).

15.14 Conclusion

This chapter analyzes air pollution control technologies that meet the sustainable environmental objectives in combination with other control technologies. Air pollution poses serious problems to public health and environment and their combined impacts resulted into extreme atmospheric conditions. Due to rapid climate change, the implications for air quality need to be better understood both for the purpose of air pollution control and societal consequences of climate change. Measuring the associated impact of climate change and air pollution should be an important part in designing, developing, and implementing technologies for effective air pollution control. This chapter briefly touched on the potential effect of air pollutants originated from the burning of fossil fuels which resulted in atmospheric extreme conditions, i.e., wildfires, dust storms, and how these impacts can be mitigated by control techniques. Further research is needed to establish greater outcomes and examine important factors that are associated to the developmental actions to combat air pollution in the long-term.

References

- Akhlaq M, Sheltami TR, Mouftah HT (2012) A review of techniques and technologies for sand and dust storm detection. Rev Environ Sci Biotechnol 11(3):305–322
- Alkhatib AAA (2014) A review on forest fire detection techniques. Int J Distrib Sens Netw 10: 597368

Azevedo JM, Gonçalves FL, de Fátima Andrade M (2011) Long-range ozone transport and its impact on respiratory and cardiovascular health in the north of Portugal. Int J Biometeorol 55(2): 187–202

- Becker K, Zhu W, Lopez J (2005) Microplasmas: environmental and biological applications. Plasma Phys Controlled Fusion 47:B513–B523
- Buonicore AJ, Davis WT (1992) Air pollution control engineering manual. Air and Waste Management Association, Van Nostrand Reinhold, New York
- Chen R, Lee I, Zhang L (2014) Biopolymer stabilization of mine tailings for dust control. J Geotech Geoenviron Eng 141:04014100
- Chidthaisong A (2005) Couplings between changes in the climate system and biogeochemistry. Notes 19
- Chung KF, Zhang J, Zhong N (2011) Outdoor air pollution and respiratory health in Asia. Respirology 16(7):1023–1026
- Cooper CD, Alley FC (2011) Air pollution control: a design approach. Waveland Press Inc., Long Grove
- Crocker BB (2000). Air pollution control methods. Kirk-Othmer Encyclopedia of Chemical Technology.
- De Longueville F, Ozer P, Doumbia S, Henry S (2013) Desert dust impacts on human health: an alarming worldwide reality and a need for studies in West Africa. Int J Biometeorol 57(1):1–19
- De DK, Olawole OC, Joel ES, Ikono UI, Oyedepo SO, Olawole OF et al (2019, Sept) Twenty-first century technology of combating wildfire. In: IOP conference series: earth and environmental science, vol 331, no 1, IOP Publishing, p 012015
- Dennekamp M, Abramson MJ (2011) The effects of bushfire smoke on respiratory health. Respirology 16(2):198–209
- Dennekamp M, Carey M (2010) Air quality and chronic disease: why action on climate change is also good for health. N S W Public Health Bull 21(6):115–121
- Doty SL, James CA, Moore AL, Vajzovic A, Singleton GL, Ma C, Strand SE (2007) Enhanced phytoremediation of volatile environmental pollutants with transgenic trees. Proc Natl Acad Sci 104(43):16816–16821
- Engardt M, Bergström R, Andersson C (2009) Climate and emission changes contributing to changes in near-surface ozone in Europe over the coming decades: results from model studies. Ambio 38(8):452–458
- EPA Fact Sheet (2016) Spray chamber/spray tower wet scrubber, EPA-452/F-03-016
- Finlay SE, Moffat A, Gazzard R, Baker D, Murray V (2012) Health impacts of wildfires. PLoS Curr 4:e4f959951cce2c
- Ginoux P, Chin M, Tegen I, Prospero J, Holben B, Dubovik O, Lin S-J (2001) Sources and global distributions of dust aerosols simulated with the GOCART model. J Geophys Res 106:20255– 20273
- He J, Chen K, Xu J (2017) Urban air pollution and control. In: Encyclopedia of sustainable technologies. pp 243–257
- Henderson SB, Brauer M, MacNab YC, Kennedy SM (2011) Three measures of forest fire smoke exposure and their associations with respiratory and cardiovascular health outcomes in a population-based cohort. Environ Health Perspect 119(9):1266–1271
- Jacob DJ, Winner DA (2009) Effect of climate change on air quality. Atmos Environ 43(1):51-63
- Jiménez E, Linares C, Martínez D, Díaz J (2010) Role of Saharan dust in the relationship between particulate matter and short-term daily mortality among the elderly in Madrid (Spain). Sci Total Environ 408(23):5729–5736
- Karanasiou A, Moreno N, Moreno T, Viana M, De Leeuw F, Querol X (2012) Health effects from Sahara dust episodes in Europe: literature review and research gaps. Environ Int 47:107–114
- Krawchuk MA, Moritz MA, Parisien MA, Van Dorn J, Hayhoe K (2009) Global pyrogeography: the current and future distribution of wildfire. PLoS One 4(4):e5102
- Larney FJ, Bullock MS, Janzen HH, Ellert BH, Olson EC (1998) Wind erosion effects on nutrient redistribution and soil productivity. J Soil Water Conserv 53:133–140
- Leith D, Jones DL (1997) Cyclones. In: Fayed ME, Otten L (eds) Handbook of powder science and technology. Chapman and Hall, New York

- Lelieveld JO, Crutzen PJ, Ramanathan V, Andreae MO, Brenninkmeijer CAM, Campos T et al (2001) The Indian Ocean experiment: widespread air pollution from South and Southeast Asia. Science 291(5506):1031–1036
- Liao H, Chen WT, Seinfeld JH (2006) Role of climate change in global predictions of future tropospheric ozone and aerosols. J Geophys Res Atmos 111(D12)
- Manisalidis I, Stavropoulou E, Stavropoulos A, Bezirtzoglou E (2020) Environmental and health impacts of air pollution: a review. Front Public Health 8:14."

Middleton NJ (2017) Desert dust hazards: a global review. Aeolian Res 24:53-63

- Middleton N, Yiallouros P, Kleanthous S, Kolokotroni O, Schwartz J, Dockery DW et al (2008) A 10-year time-series analysis of respiratory and cardiovascular morbidity in Nicosia, Cyprus: the effect of short-term changes in air pollution and dust storms. Environ Health 7(1):39
- Naeher LP, Brauer M, Lipsett M, Zelikoff JT, Simpson CD, Koenig JQ, Smith KR (2007) Woodsmoke health effects: a review. Inhal Toxicol 19(1):67–106
- Nakau K, Fukuda M, Kushida K, Hayasaka H, Kimura K, Tani H (2006) Forest fire detection based on MODIS satellite imagery, and Comparison of NOAA satellite imagery with fire fighters' information. pp 18–23
- Niemi JV, Saarikoski S, Aurela M, Tervahattu H, Hillamo R, Westphal DL et al (2009) Long-range transport episodes of fine particles in southern Finland during 1999–2007. Atmos Environ 43(6):1255–1264
- Okokpujie K, John SN, Osaghae EN, Okokpujie IP, Okonige R (2019) A wireless sensor networkbased fire protection system with SMS alerts. Int J Mech Eng Technol 10(2):1–10
- Ordóñez C, Mathis H, Furger M, Henne S, Hüglin C, Staehelin J, Prévôt AS (2005) Changes of daily surface ozone maxima in Switzerland in all seasons from 1992 to 2002 and discussion of summer 2003. Atmos Chem Phys 5(5):1187–1203
- Ovadnevaitė J, Kvietkus K, Maršalka A (2006) 2002 summer fires in Lithuania: impact on the Vilnius city air quality and the inhabitants health. Sci Total Environ 356(1–3):11–21
- Perez L, Tobías A, Querol X, Pey J, Alastuey A, Díaz J, Sunyer J (2012) Saharan dust, particulate matter and cause-specific mortality: a case–crossover study in Barcelona (Spain). Environ Int 48:150–155
- Samoli E, Nastos PT, Paliatsos AG, Katsouyanni K, Priftis KN (2011) Acute effects of air pollution on pediatric asthma exacerbation: evidence of association and effect modification. Environ Res 111(3):418–424
- Saxena P, Ghosh C (2015) A sustainable approach towards minimizing atmospheric benzene by the use of plants. Int J Agric Crop Sci 8(3):495
- Saxena P, Sonwani S (2019a) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Sonwani S (2019b) Policy regulations and future recommendations. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 127–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Bhardwaj R, Ghosh C (2012) Status of air pollutants after implementation of CNG in Delhi. Curr World Environ 7(1):109
- Skidmore EL (1986) Wind erosion control. Clim Chang 9:209–218
- Smith KR, Jerrett M, Anderson HR, Burnett RT, Stone V, Derwent R et al (2009) Public health benefits of strategies to reduce greenhouse-gas emissions: health implications of short-lived greenhouse pollutants. Lancet 374(9707):2091–2103
- Sonwani S, Saxena P (2021) Water-insoluble carbonaceous components in rainwater over an urban background location in Northern India during pre-monsoon and monsoon seasons. Environ Sci Pollut Res 28(38):53058–53073
- Sonwani S, Saxena P (eds) (2022) Greenhouse gases: sources, sinks and mitigation. Springer-Nature, Singapore, pp 1–255

- Sonwani S, Saxena P, Shukla A (2021) Carbonaceous aerosol characterization and their relationship with meteorological parameters during summer monsoon and winter monsoon at an industrial region in Delhi, India. Earth Space Sci 8(4):e2020EA001303
- Spickett JT, Brown HL, Rumchev K (2011) Climate change and air quality: the potential impact on health. Asia Pacific. J Public Health 23(2 Suppl):37S–45S
- Thalib L, Al-Taiar A (2012) Dust storms and the risk of asthma admissions to hospitals in Kuwait. Sci Total Environ 433:347–351
- Treybal RE (1980) Mass transfer operations. New York, p 466
- Triplett GB, Dick WA (2008) No-tillage crop production: a revolution in agriculture! Agron J 100: S-153–S-156
- Uprety DC, Saxena P (2021) Carbon dioxide. In: Technologies for green house gas assessment in crop studies. Springer, Singapore, pp 7–57
- Vetter S (2005) Rangelands at equilibrium and non-equilibrium: recent developments in the debate. J Arid Environ 62:321–341
- Viswanathan S, Eria L, Diunugala N, Johnson J, McClean C (2006) An analysis of effects of San Diego wildfire on ambient air quality. J Air Waste Manage Assoc 56(1):56–67
- Vlachokostas C, Achillas C, Slini T, Moussiopoulos N, Banias G, Dimitrakis I (2011) Willingness to pay for reducing the risk of premature mortality attributed to air pollution: a contingent valuation study for Greece. Atmos Pollut Res 2(3):275–282
- Wang X, Zhang C, Hasi E, Dong Z (2010) Has the Three Norths Forest Shelterbelt Program solved the desertification and dust storm problems in arid and semiarid China? J Arid Environ 74:13–22
- Watson A (1985) The control of blowing sand and mobile desert dunes. The control of windblown sand and moving dunes: a review of the methods of sand control in deserts, with observations from Saudi Arabia. Q J Eng Geol Hydrogeol 18:237–252
- White HJ (1975) Role of electrostatic precipitators in particulate control: a retrospective and prospective view. J Air Pollut Control Assoc 25(2):102–107
- Wolverton BC (1988) Foliage plants for improving indoor air quality. In National foliage foundation interiorscape seminar (No. NAS 1.15: 108055).
- Yuan C, Zhang Y, Liu Z (2015) A survey on technologies for automatic forest fire monitoring, detection, and fighting using unmanned aerial vehicles and remote sensing techniques Can. J For Res 45:783–792



16

Ecological and Natural-Based Solutions as Green Growth Strategies for Disaster and Emergency Management of Air Pollution Extremes

Sharfaa Hussain and Raza R. Hoque

Abstract

Air Pollution, across the globe is worsening despite efforts. It is an international concern due to its transboundary nature and its impacts on regional and global scale. Pollutants that are emitted to the atmosphere become an active participant in atmospheric reactions thereby altering the atmospheric processes. These alterations facilitate threatening air pollution episodes, what we call as "Air Pollution Extremes". These extremes harmfully effect human health and the environment. There are mooted proposals from the researchers to consider the air pollution extremes as disasters. Disaster management institutes and agencies are to be given with responsibilities on this regard. Air pollution extremes namely, fog, smog, and dust storms have been responsible for multiple deaths per event and serious environmental havoc. World Health Organization (WHO) regards air pollution as a prime factor for human mortality and is to be blamed for seven million deaths in the world every year; therefore, control measures are imminent. Though countries already have in place the legislations and regulating authorities for the control and mitigation of air pollution, many Asian countries like India are still struggling. In this chapter, a two-way "green approach" has been discussed—lowering the source strength, i.e., keeping a check on emissions right at the source and mitigating the pollution that has already been released to the atmosphere. Adopting "green concepts" in economic development and better city planning, and strategically built green covers could be a promising alternative to avert extreme air pollution.

S. Hussain \cdot R. R. Hoque (\boxtimes)

Department of Environmental Science, Tezpur University, Tezpur, India e-mail: rrh@tezu.ernet.in

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_16

Keywords

Green growth strategies · Air pollution extremes · Natural-based solution

16.1 Introduction

For the past several decades air pollution has been an important issue for atmospheric scientists and policy makers. Air pollution has been in the political agenda in many countries including India and there has been international concern due to its transboundary nature and impacts to the dimension of regional to global scale. There are three major concerns—(1) public health, (2) effect on biota and agriculture, and (3) changing composition of trace constituents of air thereby changing the atmospheric processes.

The world nations have recognized air pollution as a threat to human well-being (Shaddick et al. 2020; Mannucci and Franchini 2017; Saxena and Srivastava 2020) and, therefore, some of the pollutants, e.g., particulate matter (PM), sulfur oxides (SOx), nitrogen oxides (NOx), surface ozone (O₃), and some hydrocarbons like benzene and benzo(a)pyrene (B(a)P) have been identified as criteria pollutants (Saxena and Sonwani 2019a). The criteria air pollutants are regulated by the state agencies irrespective of whether the countries are developed or developing, rich or poor (Saxena and Sonwani 2019b).

Air pollution does affect plant growth and it interferes with the processes of photosynthesis, which would mean that air pollution could have direct impacts on the national and global crop productivity. There are reports of air pollution and associated productivity loss (Burney and Ramanathan 2014; Tai et al. 2014; Avnery et al. 2011), and large populous countries cannot ignore this challenge (Saxena et al. 2019; Kulshrestha and Saxena 2016).

As the pollutants are emitted into the atmosphere, they take part in the atmospheric processes and consequently bring about changes in the earth's radiative budget (climate impacts), chemistry of precipitation (acid deposition), and oxidative pathways leading to building up of photochemical oxidants (photochemical smog) in the air, which is a vicious chemical cycle that lasts for several days with severe visibility impairment and deteriorating air quality. Severe episodic air pollution events, historically known as smog episodes, claimed many lives due to recurrent episodes in London (Read and Parton 2019; Luckin 2003; Kelly and Kelly 2009), Glasgow (Bower et al. 1994), and other cities.

Today, it is understood with fair bit of uncertainty that air pollutants, more precisely the carbonaceous aerosol, have implications on the cloud seeding and the weather modifications (Garvey 1975; Guo and Zheng 2009; Guo et al. 2015; Rosenfeld et al. 2008). Extreme rainfall and long spell of dry days are often regarded to have been fallout of carbonaceous aerosol emissions. Cloud bursts and flood have been reported from varied geographical regions of the world (Gupta et al. 2013; Dimri et al. 2017; Kumar et al. 2018; Minea 2013). It is, thus, with little doubt that human action has led to tremendous pressure on the atmosphere and other

components of the environment leading to visible global impacts, which is why Paul Crutzen had proposed that the present era be named as Anthropocene (Crutzen 2006; Steffen et al. 2007; Monastersky 2015).

Air pollution in many countries is taking the form of a disaster and hence there has to be a ready-to-use disaster risk management plan. A country's government body should formulate laws and acts that check and regulate air pollution. Disaster management authority that can strategize predisaster, immediate postdisaster, and recovery steps taken during a disaster is a necessity. With evident degrading air pollution green concepts came into play. These concepts refer to socio-economic development keeping sustainability and "easy on environment" nature in mind. Today, the term "green" is often prefixed to any process that does minimum damage to the environment, e.g., green industry, green technology, green fuel, green energy, green methods, etc. Green concepts provide an ecological and natural-based solution to tackle and minimize air pollution. In 2008, World Bank first proposed the idea of nature-based approaches (Tzoulas et al. 2020) and it received tremendous support from the International Union for Conservation of Nature (IUCN 2012) and the European Commission (EC 2015). Since then, natural-based solution has gained importance since it gives prime attention to how much damage it does to the environment. This chapter talks about air pollution and its extremes, how anthropogenic actions have lead air pollution to a disaster and how we can mitigate air pollution and its extremes through ecological and natural-based solution as green growth strategies. The aim of the chapter is to use the approach of nature-based solutions that use ecological and environment-friendly methods in urban planning and management to mitigating air pollution.

16.1.1 Air Pollution and Its Threats

From smog enveloping the cities to smoke hanging indoors, air pollution persists everywhere posing a threat to human health and climate. According to the World Health Organization (WHO), air pollution is responsible for about seven million deaths across the globe, every year and for the record, every nine individuals out of ten breathe in polluted air. The WHO also predicts that there will be an additional 250,000 death cases from 2030 to 2050 owing to air pollution and climate change (WHO 2014). The deaths caused by air pollution can be attributed to serious health issues like chronic obstructive pulmonary disease, acute respiratory infections, etc. Studies show that more than 80% of people inhabiting in urban areas breathe in air that exceeds the WHO air quality guidelines (Sonwani et al. 2022). People from parts of developing country and underdeveloped countries suffer from a combined exposure of both indoor and outdoor pollution. It is expected that by 2030, poor air quality will become one of the eminent environmental cause for premature death (WHO 2014). Long-term exposure to polluted air increases the risk of respiratory illness, cardiovascular disease, and cancer (e.g., Dockery et al. 1993; Jerrett et al. 2009; Krewski et al. 2009; Pope III et al. 2009), thereby increasing the annual acute mortality rate. Additionally, their resultant extremes have the potential to obstruct

daily activities, and henceforth calls for strict control measures like Paris temporarily banned vehicles having even-numbered license plates (Porter et al. 2015). Similar policy was also adopted in Delhi with a regime of odd or even number vehicles to run on odd or even number day of the week (Tiwari et al. 2018; Kumar et al. 2017; Sehgal and Gautam 2016; Rao et al. 2017). Air pollution, though a global issue, has large differences in severity. The low-income countries topped the list of air pollution severity. The year 2017, (Fig. 16.1) witnessed an estimated five million deaths globally (Ritchie and Roser 2017). Figure 16.2 shows how air pollution driven deaths vary spatially. It is clear that death rates are 100 times higher in developing countries than Europe and North America.

Air pollution also imposes extensive threats to global food security. With global spike for crop demand due to increasing population, future pollution is expected to leave the global food supply in an extremely vulnerable condition (Alexandratos and Bruinsma 2012). Among various air pollutants, surface ozone (O_3) is explicitly harmful to crop yields (Ainsworth et al. 2012).

16.1.2 Air Pollution: A Brief on Global and National Scenario

Air quality in megacities impacts a significant global population and the affected population will only increase in urban areas over the coming decades. By 2030, the population count in 41 megacities is predicted to rise up to 730 million (UNDP 2015), pointing toward the need to have proper understanding of the influence of dense population on their environment and health of inhabitants. According to the Global Burden of Disease (Lim et al. 2012), ambient air pollution is a major risk factor and global population as large as 96% in the metro cities breathe air with high levels of particulate matter $(PM_{2,5})$ that are above World Health Organization (WHO) air quality standards (Krzyzanowski et al. 2014). Total of 7% of global burden from ambient air pollution comes from the 30 most populated urban areas, while several cities in Eastern and Southern part of Asia have higher per capita effects. Two megacities from India, viz., Mumbai and Delhi are ranked among top ten of populated megacities. It is expected that populations in Delhi and Mumbai will spike from 25.7 to 36.1 million and from 21.0 to 27.8 million, respectively, between 2015 and 2030 (UNDP 2014), thus triggering severity of air pollution. A stationbased PM25 analysis carried out in Delhi found that 69% of samples exceeded the 24-h standards (Tiwari et al. 2013). The average PM2.5 and PM10 recorded in Delhi during 2008–2011 were 123 and 208 μ g/m³ respectively and the levels are above standard (Guttikunda and Goel 2013). Mean $PM_{2.5}$ and PM_{10} concentrations in October 2010 were 111 and 238 μ g/m³, respectively, and hourly O₃ was measured to be above 100 ppb. It was also found that in Delhi, besides multiple emission sources pollutant from outside the capital also enormously contributed to PM (25%) and daytime O_3 (60%) (Marrapu et al. 2014). Major share of India's population ~76.8% is subjected to annual population-weighted mean $PM_{2.5}$ which is higher than the country's National Ambient Air Quality Standards. In 2017, Delhi recorded the highest annual population-weighted mean PM_{2.5} followed by Uttar Pradesh, Bihar,







Fig. 16.2 Death rates across the globe due to air pollution from the year 1990 to 2017. (Source: IHME, Global Burden of Diseases, Ritchie and Roser 2017)

and Haryana with an average higher than 125 μ g/m³ (Balakrishnan et al. 2019). Figure 16.3 illustrates the annual and hourly PM_{2.5} levels of some important cities worldwide. Out of eight cities examined, via satellite observations, CO, an important O₃ precursor, showed an increase over 2000–2008 only in Delhi (Pommier et al. 2013). A study found that if global greenhouse gas can be mitigated, we can clear of 0.5 ± 0.2 million demises triggered by PM_{2.5} and O₃ by the year 2030 (West et al. 2013).

16.2 Factors That Influence Air Pollution

The concentration of a pollutants species is a complex function of emissions (source strength), meteorological conditions (allow dispersion, dilution, and transport), atmospheric reactions (lead to degradation of pollutant species), and deposition processes.

Emissions from industries, vehicles, fuel combustion, and biomass burning are some of the common sources of air pollutants which are quite understood well (Bhuyan et al. 2018; Masiol et al. 2017). Emissions from biomass burning and crustal or street dust are major contributors of air pollution in African and Asian region (Bhuyan et al. 2018; Deka and Hoque 2015; Hsu et al. 2014; Chen et al. 2017; Jain et al. 2021). In the Asian context, there are unique types of episodic biomass burning like the *meji* burning of the Brahmaputra Valley (Deka and Hoque 2014; Hoque and Deka 2010), *holika* burning of northern India (Singh et al. 2017), and crop residue burning in Indian states of Punjab and Haryana (Sawlani et al. 2019; Saxena et al. 2021). The crop residue burning adds to the already polluted air of



Fig. 16.3 (a) Maximum 24-h PM_{2.5} concentrations $(\mu g/m^3)$ at USA Embassy locations of Mumbai, Delhi, and USA E.P.A. sites in Los Angeles and New York (2014); (b) Annual average PM_{2.5} concentrations $(\mu g/m^3)$ retrieved from stations (2014) and satellite data (2011) (Source: Marlier et al. 2016)

Delhi. It is estimated that 39% of the worldwide population uses biomass fuels for cooking and heating, typically in developing countries (IEA 2011). Household biomass burning emissions add to the ambient air pollution too and the varied biomass types that are generally used have characteristic emission profiles (Deka and Hoque 2015).

Meteorological conditions of an area play an eminent role in the building up of pollutants in the atmosphere. Factors like the prevailing wind, turbulence, stability, and temperature inversions affect the transport, dilution, and dispersion of the pollutants (Shenfeld 1970). Inversions lead to very stable atmospheric condition which does not allow vertical mixing and thus traps pollutants. There is a permanent inversion layer called the "boundary layer" at a height (mixing height), which keeps changing. It would also mean that the pollutant concentrations will be greater when the mixing height experienced is low.

Atmospheric deposition is the permanent loss of pollutant from atmosphere through wet or dry deposition and reduces some part of the total atmospheric loading. Pollutants can be deposited onto vegetation, soil systems, and water bodies. Wet deposition refers to precipitation processes and deposits at a similar speed to every surface (Sehmel 1980; Fowler et al. 2004). Although in dry deposition, the rate depends on the macroscopic characteristics of the surface (Padro 1996; Fowler et al. 2004; Grote et al. 2016). Fowler et al. measured the particle deposition velocity and found that urban trees account for 11 mm/s and grass accounts for 3 mm/s. Of total dry deposition, 70% settles on urban trees and 25% on grasses (Fowler et al. 2004). Attributes like larger surface area, higher transpiration rates, and longer in-leaf periods enhance the process of dry deposition (Padro 1996; Branford et al. 2004; Nowak et al. 2006; Cabaraban et al. 2013; Grote et al. 2016). Lichens and mosses serve as an excellent accumulator for atmospheric dry depositions and, therefore, they are often applied in the biomonitoring studies of air pollutants (Daimari et al. 2013, 2020). Lichens also show the early signs of air pollution effects (Hazarika et al. 2011).

However, we do not have control over the meteorology and other atmospheric processes which influence the extent of air pollution. Though, there are some ways like artificial rain induction; however, this has not been of much convenience because of lots of other uncertainties and introduction of additional chemicals into the environment. It is "only" the source strength of pollution which is generally the focus of policy makers and while making strategies for air pollution reduction, lowering of the source strength has been the priority. Also, it is a wise move to understand the meteorology of a place while locating industries in and around a city which can provide with some relief considering the prevailing winds and other information.

16.3 Air Pollution Extremes

The three air pollution extremes discussed here are fog, smog, and sand/dust storms as per the following:

16.3.1 Fog

Fog is the cloud at the surface level atmosphere. Fog is considered as a hazard because it reduces the horizontal visibility to less than 1 km (George 1951). Increased aerosols in the cities due to rapid urbanization and increased number of automobiles catalyze fog events. Increase in the relative humidity and aerosols of particulate matter are expected to be responsible for poor visibility (Oldenborgh et al. 2018). When visibility is less than 400 m, fog is considered dense or heavy. The northern part of India is conveniently fog prone in winters (December–January). Satellite images detected a sizeable fog layer over Pakistan to Bangladesh during the winter (Gautam et al. 2007; Jenamani 2012; Ghude et al. 2017). In Northern part of India, formation of fog is supported by moisture availability, atmospheric stability, and strong low-level inversions, mostly influenced by a western disturbance

(WD) (Pasricha et al. 2003; Dimri et al. 2015). Atmospheric aerosols also serve the purpose of cloud condensation nuclei (CCN) and alter the life cycle of fog (Ghude et al. 2017; Safai et al. 2019).

Some of the active roads, railways, and airports (Delhi, Jaipur, Lucknow, and Varanasi) of India are situated in this fog prone northern region of India and are severely influenced by heavy fog during the winter months (December–January). Figure 16.4 shows the trends of percentage frequency of visibility in a few cities of India. The threat posed calls for mitigation measures to bring the risk under control. A study showed that, the poor visibility trends during winter in most of the airports in North India showed notable increasing trends amounting to 90%, i.e., almost everyday. On the other hand, the Southern India airports had just 20–30% days with poor visibility (De and Dandekar 2001).

Ecological-based solution to control fog: There is a section of plants that are termed as "fog collectors" because they collect fog from the atmosphere as a source of moisture. There is a detailed discussion in the later part of the chapter about criteria on how to select plants that can collect fog efficiently. It would be smart approach to construct a green belt or vertical live walls with fog-collecting plants in areas that tend to get dense fog.

16.3.2 Smog

The term "Smog" was coined in the early twentieth century and is a fusion of two words "smoke" and "fog". However, today we know that smog episodes are periods of very high concentrations of air pollutants rendering the air quality poor. The Great Smog of London is one of the earliest recorded and deadly smog events that killed 12,000 people. London then was extensively using coal as the prime energy source and hence emissions from coal combustion made it lethal. The condition worsened due to the subsidence inversion over London region. Subsidence inversion persisted for longer duration and it made a deadly combination with high emission due to strong source strength of air pollutants (coal burning during that time).

There are two types of smog (1) the London type (classical, reducing type) and (2) the Los Angeles type (photochemical, oxidizing type). India witnessed a severe smog event during November 1–7, 2016. Delhi, the capital of India was caught in a severe air pollution episode (SAPE) or the "Great Delhi Smog". The air quality index (AQI) transcended 500 and the smog layer enveloped the entire capital. Researchers from the University of Hyderabad (UoH), Indian Meteorological Department (IMD), Indian Institute of Tropical Meteorology (New Delhi), Banaras Hindu University, Finnish Meteorological Institute, Finland, Pacific Northwest National Laboratory, Richland in the US, undertook a study that revealed that "air stagnation" catalyzed the event and made it worse. The study also stated that degraded air quality due to high levels of $PM_{2.5}$ has been experienced by Southern Asia. More recently, the winter time intense air pollution haze over Delhi has drawn much focus, which has pushed governments and regulating agencies to the crossroads. The government of Delhi had tried out policies from vehicular



Fig. 16.4 Percentage frequencies of poor visibility and its trends in the Indian cities between 1969 and 1997 (De et al. 2005)

movements to agricultural activities in the rural areas. As an emergency move, the schools were closed and people were advised to stay indoors during the high pollution days.

Smog has been a serious issue in Delhi. The city has been gasping for air every year (Kedia 2017) and root causes are many. Smoke from rural kitchens, vehicular emission, industries, more numbers of vehicles, spike in Compressed Natural Gas (CNG) prices leading to commoners unaffordability, dieselization of cars, jeopardization of nonpolluting public transports, dense population, low investment in public transport and insufficiency of public infrastructure, extensive construction activities, burning of left over crop in adjacent states such as Punjab, Haryana, and Uttar Pradesh contribute together. During the months of autumn and winter, approximately 500 million tons of crop residues are burned in Indo-Gangetic plains leading to mass emissions from biomass burning (Sharma et al. 2010). According to another study, 72% of the total air pollution load in Delhi is from vehicular emissions as calculated using emission factor and activity-based approach suggested by Intergovernmental Panel on Climate Change (IPCC) (Sindhwani and Goyal 2014). Furthermore, during winters, calm winds and low temperatures entrap dust particles and pollutants near the ground setting up smog layer. And then the festival of Diwali, due to fireworks, gives an incremental effect to the air pollution of Delhi.

Natural-based solution to control Smog: Urban air pollution exerts pressure on plants to release stress-induced biogenic emissions (bVOCs) that take part in atmospheric reactions to form secondary organic aerosols (SOAs) and phytochemical smog. Hence, emission profile of plants should be considered while urban plantation. Air Pollution Tolerance Index (APTI) calculates the pollution-tolerant levels of plants and can be used to select plants based on pollution of an area and the plant's tolerance level. There are studies that confirm how some plants can capture particulate matters. So the natural-based approach to mitigate smog would be construction of green belts and extensive plantation of varied plants to tackle all components of smog, considering it is a mixture of fog, pollutants, and particulate matters. Thick lush moss plantation will also be an intelligent approach since moss can entrap access moisture and pollutants from the atmosphere. Details are discussed in the later part of the chapter.

16.3.3 Sand and Dust Storms

Sand and dust storms (SDS) are the occurrence of crystal particles being carried by a strong wind gust. A SDS event takes place when vigorous and turbulent winds gust over parched, loose, fine-grained soil surface that lacks vegetation cover (Middleton and Kang 2017). Strong and speedy wind has the ability to translocate dust for several kilometres and hence dust storms are not just confined to deserts. India was hit by many episodes of acute dust storms, thunderstorms, and lightning and there are reports of deaths due to storms (Illiyas et al. 2014; De et al. 2005). In India such events are common in the western Rajasthan Province, which is covered by the Thar Desert. Dey et al. (2004) reported the effects of a dust storm event in aerosol

Entrainment	Transport	Deposition
Soil loss	Sand blasting of crops	Salt deposition and groundwater
Nutrient, seed, and	Radio communication	salinization
fertilizer loss	problems	Reduction of reservoir storage
Crop root exposure	Microwave attenuation	capacity
Undermining structures	Transport disruption	Drinking water contamination
	Local climate effects	Burial of structures
	Air pollution	Crop growth problems
	Respiratory problems and eye	Machinery problems
	infections	Reduction of solar power
	Disease transmission (humans)	potential
	Disease transmission (plants	Electrical insulator failure
	and animals)	Disruption to power supplies

 Table 16.1 Environmental Hazards to human population caused by wind and dust storms (Middleton and Kang 2017)

 Table 16.2
 Measures for controlling windblown dust (Watson 1985)

Control approach	Examples
Promote deposition	Ditches; fences; tree belts
Enhance transport	Streamlining techniques; creating a smooth texture over the land surface; erecting panels to detect air flow
Reduce land supply	Surface stabilizing techniques; fences; vegetation
Deflect moving sand	Fences; tree belts

parameters and observed a rise in AOD by more than 50% and PM_{10} after a dust storm event. Table 16.1 shows wind erosion and dust storms imposed environmental hazards to human populations.

Natural-based solution to control sand and dust storm: Green fences and tree belts can slow down wind speed and control its devastating nature to a great extent. Sterberg et al., reported that English Ivy can trap suspended particle and has a dust absorption rate of 2.9×10^{10} particles per m² for upper side of the leaf (Sternberg et al. 2010). Such plants can be used in areas that are prone to dust and sand storm. Other control approaches are given in Table 16.2.

16.4 Mitigating Air Pollution

A strategic approach to mitigate air pollution is "Reduce–Extend–Protect" (Hewitt et al. 2020). Minimizing emissions right at the source is the prime step of mitigation and can be achieved by lowering the source strength. That is where government regulations and introduction of green concepts come into play. The other mitigating ways are extending the distance between the source and the sink and protecting

receptor sites from pollutants. This can be achieved by better planning of urban areas which will target the existing pollution and help reduce it.

16.4.1 Air Pollution Risk Reduction and Management

It is quite obvious that with the over exploitation of natural resources, we are on the losing side and slowly the idea of sustainable development took shape, which was clearly defined by the Brundtland Commission in 1987 (Redclift 2005). The "green concepts" are nothing but sustainable socio-economic development with concern toward the environment. As accepted at the fifth Ministerial Conference on Environment and Development in Asia and the Pacific, green growth is a strategy for achieving sustainable development (Sarkar 2013). With the development in the field of greener technologies that emit less and the adoption of greener fuels, much lowering of the source strengths of air pollution can be achieved. With the introduction of green technologies like electric vehicles, significant lowering of air pollution in our cities could be envisaged, provided the needed electricity comes from zero emission technologies like solar or wind power. If we have to rely on coal-fired thermal power to fuel the electric vehicles, then the purpose would be lost. Also, the most coveted hydrogen fuel could change the air pollution scenario across the world. However, due to high cost of production and technology, the developing countries may not get an easy access to it.

For the air pollution control, countries have put in legislations in place and there are institutions that work under the provisions of the laws. For instance in India, the Air (Prevention and Control of Pollution) Act, 1981 was enacted for the control and abatement of air pollution. At the center, Central Pollution Control Board (CPCB) and at state level, the State Pollution Control Boards (SPCBs) were created under the provisions of the Water Act of 1974, which were entrusted with the responsibilities of control and mitigation of air pollution in India. National ambient air quality standards (NAAQS) have been set for the control. Also the pollution control boards can declare air pollution control zones to regulate polluting units, and a permit regime has been put in place. However, most cities in India are struggling to meet the NAAQS.

From the historical understanding of air pollution, it can be realized with definite terms that extreme air pollution can take a form of a disaster. And, in the case of India, air pollution is going to be the biggest disaster (Bhatt 2018). Therefore, it is good to have a disaster management approach in place for reduction of the associated risks.

Disaster risk reduction is about a systematic array of five key zones of decision building: Risk Identification, Risk Reduction, Preparedness, Financial protection, and Resilient reconstruction. Earlier dealing with disaster only meant coming up with emergency response postdisaster, even so to the climax of twentieth century, researchers were convinced that disasters are not natural and that, the main way of minimizing risk lies in minimizing vulnerability and exposure (GFDRR 2014).

Pre-disaster	Immediate post-disaster	Recovery
•	•	•
Public education	Search and rescue	Transitional shelter in form of temporary housing or long-
Awareness raising	Emergency medical treatment	term shelter
Warning and evacuation plans	Damage and needs assessment Provision of services—water, food,	Demolition of critically damaged structures
Pre-positioning of resources and supplies	health, shelter, sanitation, social services, security	Repair of less seriously damaged structures
• Last minute	Resumption of critical infrastructure	Clearance, removal and disposal of debris
preparedness measures	Coordination/management of	Rehabilitation of infrastructure
	development partner support	• New construction
		• Social rehabilitation
		'Building back better' to reduce future risk
		• Employment schemes
		• Reimbursement for losses
		Reassessment of risks

Table 16.3 Activities associated with managing the impacts of disasters (Lal et al. 2012)

Table 16.3 illustrates the basic predisaster, immediate postdisaster, and recovery steps taken during a disaster.

With rising need of disaster risk management in India, the National Disaster Management Act (NDMA) 2005 was enacted. The NDMA holds the responsibility of issuing guidelines and providing with minimum standards of relief to the affected. The Act extends to the whole of India and comes into force when Central Government instructs. The National Institute of Disaster Management (NIDM) has been assigned the pivotal responsibilities to capacity building and policy advocacy, besides other responsibilities, in the issues of disaster management in India under the provisions of the Disaster Management Act 2005. NIDM has in its priority to encourage and create awareness amidst stakeholders, students and teachers of schools and colleges, technical human resources, and others associated with multihazard mitigation, readiness, and response measures (NIDM 2020). NIDM established a network called India Universities and Institutions Network for Disaster Risk Reduction (IUIN-DRR) to work together to tackle disasters. As air pollution is taking a shape of a disaster in the Indian cities and the rural areas around them, it is envisaged that institutions like the Disaster Management Authority (DMA) take connivance and explore possibilities of declaring poor air quality as a disaster. As the NIDM has all the right conditions to handle disasters, extreme air pollution may be a priority for NIDM and take up necessary action (Bhatt 2018).

16.4.2 Green Strategies to Reduce Air Pollution

16.4.2.1 Greening Cities to Mitigate Extreme Air Pollution

Rapid urbanization and population growth have led to increased construction of buildings and skyscrapers with very little space left for greening. A large deal of the deteriorated air quality can be attributed to urban city planning which results into a faulty urban canopy (Ratti et al. 2006; Abhijith et al. 2017). The urban canopy takes up the near-surface volume (Henderson et al. 2016) and interacts with the air flow (Oke 1988). *Green cities, green buildings,* and *green roads* are some concepts that can be strategized together to mitigate the air pollution in urban atmosphere. More and more compact and tall buildings, without enough green cover impair wind flow or deposition of pollutants, thus favor the conditions of heat island formation.

Besides planting trees and grass cover in just streets and parks, other control means like roof greening and vertical plantation have proved useful. A detailed study done on city designs of China, Boya (2012) showed that Beijing's urban planning strategy is in the form of what the author calls as "baking pie" (basically in shape of concentric rings) with green covers surrounding the ring traffic areas. Such green covers are isolated and lack contact with green environment failing to create an independent green system or a healthy microclimate of their own. For a green cover to properly function, they should be connected to form a green belt that acts as the lung of the city, creates a wind channel, and acts as satisfactory sink. This has been mentioned as one major issue for serious atmospheric pollution of Beijing in recent years. Also Sanghai has seen multiple constructions of tall buildings in the recent years. This increase in urban building density restricts natural ventilation and traps the indoor air pollutants and CO₂ thus deteriorating the air quality. Owing to this issue, the author proposed decentralization of the urban spots by breaking down single centers to multicenters, and its functions and populations surrounding them and leaving enough areas for green space.

An ideal green city is not just planning the urban design of the city but also calls for green/sustainable roads and green buildings. Road transport is evidently one of the most economic and widely chosen modes of transport for both good transport and passengers. India has the world's second largest road network of 4.24 million km (MoRTH 2008). The construction of roads also gives out harmful emissions to the atmosphere. Green roads use microsurfacing technique instead of using hot mix where fossil fuel is used both as a raw material and energy source. On an average, hot mix requires approximately 10 m³ of the natural gas and has about 0.525 kg of carbon. Microsurfacing is in use since 1976 in countries like Germany, Spain, and France and was later introduced to India in 1999–2000 (Ryntathiang et al. 2013). Microsurfacing uses no heat, cold in-place recycling thus minimizing noxious fumes and other harmful emissions, making it safer for the environment and the workers.



Fig. 16.5 Outcomes of a permeable obstruction (like hedge) in diluting pollutants. Here the child receptor is exposed to pollutant concentrations c_1 and c_2 via paths d_1 and d_2 . Through d_1 , deposition happens diluting c_1 through d_2 mixing takes place, causing c_2 to decrease. In the absence of barrier live permeable obstruction, the receptor is exposed to c_0 , barely diluted over a tiny distance d_0 (Hewitt et al. 2020)

Microsurfacing uses a mixture of polymer-modified asphalt emulsion, mineral aggregate, mineral filler, water, and other additives, properly proportioned, mixed, and spread on a paved surface in variable thick cross-section which resists compaction (ISSA 2005). Sustainable building/green building is gaining importance due to high demand of constructing buildings to accommodate the rising population. Green buildings are healthy, environmental friendly substitutes that provide benefits in terms of energy and CO_2 reduction and improved health situation for building users (Balaban and de Oliveira 2018).

"Landscaping" is an important aspect that helps us achieve a green city. Planting trees have far more benefits than one truly acknowledges. They regulate the precipitation cycle, have profound role in land management and fix carbon. When green coverage exceeds 30%, the heat island effect is reduced to a great extent; when higher than 60% heat island effect is totally eliminated (Boya 2012).

Trees and hedges act like a semipermeable barrier to air flow (Bradley and Mulhearn 1983; Raine and Stevenson 1977; Tiwary et al. 2006; Gromke et al. 2016) and create turbulence thereby increasing dilution (Fig. 16.5).

16.4.2.2 Afforestation

Afforestation/reforestation is simply planting trees in barren areas or areas which had a green cover in the past. Planting trees will not only provide sink for atmospheric pollutants but also regulate wind speed and soil erosion. Planting trees can successfully reduce 25% of atmospheric CO_2 by 2050 via reduction of emissions, by acting as a low-cost sink for atmospheric CO_2 and can also contribute to sustainable development (Niles et al. 2002; Barker et al. 2007). These new plantations, can entrap 205 billion tonnes of carbon, i.e., about two-thirds of the 300 billion tonnes of carbon released to the atmosphere due to anthropogenic activities since the beginning of Industrial Revolution (ETH 2019).

16.4.2.3 Green Belt Development Planning

Green belts in urban cities are considered as the "lung of the city" as they serve as sink for harmful vehicular and industrial emissions, check dust flow, and also bring down noise pollution. All these years, overpopulation has led to insufficient open spaces and green belts in the cities, e.g., the ridge forests in Delhi and the Borivali National Park in Mumbai, served as the lungs of the cities.

In designing a green belt, certain aspects such as wind velocity, assimilation capacity of the ecosystem, tree height, canopy, topography, amount land availability, interspace between pollution source, soil and water quality, nature, and extend of pollutant are considered. Green belts are known to moderate microclimate by reducing wind speed thus reducing building-energy consumption and reducing heat conduction into buildings by reducing solar radiation (Liu and Harris 2008). Plants best suited for green belt construction are: *Acacia farnesiana, Melaleuca* sp., *Pine, Cedar, Junipers, Eucalyptus, Nerium, Acacia auriculiformis, Bamboo, Pongamia pininata, Azaridicta indica, Casuarina,* etc. (GCPC). Roof planting is another approach to achieve green cover in urban areas. It can also collect rainwater and significantly reduce the heat island effect by lowering temperature of the roof.

16.4.2.4 Vertical Gardening

Vertical greening is a constructive and effective plantation method to reduce temperature in urban areas. In case of land unavailability, plant vines are grown through walls, bars, poles, and other buildings. Creepers, air purifying shrubs, etc. are planted with drip watering system installed. It also helps in achieving shading and ambient temperature reduction in summers, isolates noise, absorbs dust, and as a whole dilutes pollution in the area. A study by Pacheco-Torgal et al., demonstrated that a vertical green-living wall filters direct sunlight since 5-30% of the sunlight energy is reflected, 5–20% is consumed in photosynthesis, 10–50% is converted to heat, 20-40% is utilized for evaporation, and finally approximately only 5-30% of the total light passes through the leaves (Pacheco-Torgal et al. 2014), thus blocking the direct sunlight and providing a cooling effect in summer. In winters, the same vertical green layer also acts as an insulation and prevents escaping of heat from inside the building (Hadba et al. 2017). M. Köhler in his study suggested that green facades can capture about 4% of the annual dust-fall in an inner-city area (Köhler 2008). For example, Boston Ivy entraps heavy metals like Al, Cd, Co, Cr, Cu, Fe, Ni, and Pb that are deposited in suspended particulate matters (Köhler 2008; Thönnessen 2002). The lush growth of English Ivy can also trap suspended particle and has a dust absorption rate of 2.9×10^{10} particles per m² for upper side of the leaf (Sternberg et al. 2010).

Vertical forests were set up in China, to tackle rocketing urban growth and air pollution crisis. The skyscrapers held 1100 trees and 2500 varieties of plants and shrubs that can facilitate 25 tons of CO_2 absorption per year and production of about 60 kg of oxygen a day (Hutt 2017).

16.5 Selection of Plants

Trees can notably influence the chemistry of urban air by the intake and emission of reactive biogenic volatile organic compounds (BVOCs). There are two potential VOC emissions, viz. "constitutive" (cBVOCs) and "stress-induced" BVOCs (sBVOCs). Owing to their highly reactive nature, BVOCs take part in atmospheric processes and facilitate formation of secondary organic aerosol (SOA). Stress is one core factor that effects BVOC emissions. Biogenic VOCs are released to the atmosphere following biotic or abiotic stress exerted on them (Behnke et al. 2009; Faldt et al. 2003). Urban trees suffer harsh conditions (Calfapietra et al. 2015), like high temperature, high pollutant levels, lack of root space, aeration, nutrient deficiency, and more frequent drought/flood episodes (Calfapietra et al. 2013). These factors together negatively affect plants and enhance emissions of BVOCs. Enhanced sBVOC emissions in turn effect the local air quality (Calfapietra et al. 2013; Churkina et al. 2015; Helln et al. 2012; Papiez et al. 2009). The VOCs react with oxides of nitrogen (NOx) in the presence of sunlight and produce photochemical smog. Ozone and PAN are major constituents of photochemical smog. This brings about a vicious cycle in the cities-the photooxidants injure more plants and the plants emit more VOCs. Thus, when taking up a tree plantation program, the following points may be considered: intended shading climate of area and emission profile of the tree. In the absence of any assessments, it is always good to go for local plants.

The differences in emission rates of different plant species vary significantly. There are plants that emit up to 10,000 times more biogenic VOCs than the atmospherically friendly "low emitters". The APTI (Air Pollution Tolerance Index) is a scientific tool to help select plant species for urban plantation depending on their pollution tolerance level. The calculation uses four biochemical parameters, viz. total chlorophyll, ascorbic acid, pH, and relative water content. According to a study that evaluated the APTI of four ornamental plant species, viz. Dracaena deremensis, **Tagetes** erecta, Rosa indica and Dianthus carvophyllus, D. deremensis, and T. erecta were found to be tolerant while R. indica and D. caryophyllus were sensitive to pollution. Based on which, it is suggested that D. deremensis and T. erecta may be used as sinks for pollutants at highly polluted sites whereas R. indica and D. caryophyllus were labeled as bioindicators (Saxena and Ghosh 2013).

A research from Delhi suggested a few trees/plants that can absorb high concentration of Suspended Particulate Matter (SPM). The research evaluated APTI values of 20 flora species and suggested that *Azadirachta indica (neem)*, *Ricinus communis* (castor bean), *Prosopis juliflora (kabulikikar, vilayati babul)*, *Dalbergia sisoo* (*sheesham*), and *Delonix regia* (*gulmohar*) are tolerant to pollution and can be used in setting up green belts (Singh 2017).

Biomonitoring studies can also tell us a lot about a plant's accumulation capacities. Biomonitoring studies have been done in higher plants, shrubs but more commonly in mosses. Mosses are effective biomonitors, belonging to the lower group of plants called bryophytes. Due to unavailability of true roots, mosses are designed to take up nutrients from the atmosphere, thus evaluating them is an indirect evaluation of atmospheric constituents of that area. Since their cell wall lacks cuticle and possesses cation exchangeable sites, it facilitates metal accumulation above their physiological needs (Tessier and Boisvert 1999). Mosses are resistant to a variety of substances that are otherwise critically toxic for other plants species, they can be found in hot-humid tropical regions and also in chilly cold tundra region and have the capability of undergoing complete desiccation and extreme temperatures as high as 110 °C (Cenci et al. 2003; Fernandez et al. 2006; Dragović and Mihailović 2009). Mosses serve as an eminent bioindicator for high sulfur dioxide (SO_2) concentration, heavy metal accumulation, and various other atmospheric contaminants (Giordano et al. 2004). The popular mosses researched about in articles are: Hypnum cupressiforme, Hylocomium splendens, and Pleurozium schreberi (Blagnyte and Paliulis 2010). Thus, owing to their easy intake of pollutants and heavy metals, growing moss in areas of extreme air pollution can be a great idea to create a vast sink and at the same time pollutant profiling of moss can be adopted as low-cost air pollution monitoring technique.

There are various studies on plants being able to serve as a fog collector. Fog is an important source of water since it allows plants to stay hydrated even during times without rain. The efficiency of plants as fog collectors depend on their leaf shapes, pinnate and perforate shaped leaves have showed higher fog yielding capacity and permeability for air carrying the fog than simple shaped leaf. In case of simple leaf, if air strikes its surface it passes around it because of its impermeability. Thus plants with pinnate and perforate leaves, which also have a lower flow resistance for the air carrying fog, are much better for fog collection (Azad 2016). More and detailed research on efficient plants that can collect fog will help us in selection of plants to construct green belts in fog prone areas. There are also specific studies on plants that can better accumulate particulate matters.

Weerakkody et al. (2018) reported that leaf with small size, complex shape, and hairy, unruly or waxy surface facilitates better PM capture. A study on suitable characteristics of leaf for entrapping traffic-emitted PM (PM_1 , $PM_{2.5}$ and PM_{10}) revealed that there was an inverse correlation between leaf size and PM accumulation. In terms of pollutant deposition, evergreen species does a better job than deciduous species, although deciduous species exhibiting longer in-leaf seasons are preferred over deciduous species with brief in-leaf seasons (Abhijith et al. 2017; Sæbø et al. 2012; Smith 2012). The reason being, evergreen species retain functional leaves all round the year, and deciduous species persist without functional leaves typically in winter or dry season. Researches have revealed that gaseous pollutants are managed by stomatal pores through their positive uptake as a consequent of plant photosynthesis and water management processes (Lawson and Blatt
2014). In response to environmental conditions, some plants have shorter stomatal opening time and some have longer opening time. And hence removal of pollutant can be enhanced by selecting plant species with long opening intervals (anisohydric species like Populus and Quercus species) (Barwise and Kumar 2020).

16.6 Conclusion

Despite formulations of environmental policies and risk management strategies, the world is still unable to make a remarkable drop down in the pollutant levels. The serious threats to human health, increasing mortality, and increase in air pollution extremes point to the fact that air pollution is taking the shape of a disaster and therefore disaster management authorities must first consider it as such and strategically formulate full proof policies. Green growth strategies are smart and a convenient approach to mitigate air pollution and hence should be copiously introduced wherever necessary. Natural- and ecological-based solutions are best approaches to mitigate air pollution and its extremes because they are natural and do not cause more pollution. Because if we are to deal with pollution in a way that again has the potential to degrade the quality of environment, the approach will be of no use. Also, we have reached the threshold point of the planet's pollution tolerance due to man-made activities. Hence all we can afford now are natural-based, zero emission, conventional techniques to get back at the existing pollution. Our approach has to spread to all the sources that are responsible for degrading the air quality. Hence strategically built green belts, vertical gardening, and any such urban plantation that has thoroughly checked a plant's pollution tolerance and emission profiles are our go-to natural-based approach. In addition to that what we need is a governing body and disaster management authority that carefully lays out strict regulations in favor of the environment and has a disaster management protocol ready in case of any emergency.

References

- Abhijith KV, Kumar P, Gallagher J, McNabola A, Baldauf R, Pilla F, Broderick B, Di Sabatino S, Pulvirenti B (2017) Air pollution abatement performances of green infrastructure in open road and built-up street canyon environments—a review. Atmos Environ 162:71–86
- Ainsworth EA, Yendrek CR, Sitch S, Collins WJ, Emberson LD (2012) The effects of tropospheric ozone on net primary productivity and implications for climate change. Annu Rev Plant Biol 63: 637–661
- Alexandratos N, Bruinsma J (2012) World agriculture towards 2030/2050: the 2012 revision. ESA Working paper. No. 12–03. FAO, Rome
- Avnery S, Mauzerall DL, Liu J, Horowitz LW (2011) Global crop yield reductions due to surface ozone exposure: 2. Year 2030 potential crop production losses and economic damage under two scenarios of O3 pollution. Atmos Environ 45(13):2297–2309
- Azad MAK (2016) Fog collection on plant surfaces and biomimetic applications. Doctoral dissertation, Universitäts-und Landesbibliothek Bonn

- Balaban O, de Oliveira JA (2018) Sustainable buildings for healthier cities: assessing the co-benefits of green buildings in Japan. J Clean Prod 163:S68–S78
- Balakrishnan K, Dey S, Gupta T, Dhaliwal RS, Brauer M, Cohen AJ, Stanaway JD, Beig G, Joshi TK, Aggarwal AN, Sabde Y (2019) The impact of air pollution on deaths, disease burden, and life expectancy across the states of India: the Global Burden of Disease Study 2017. Lancet Planet Health 3(1):e26–e39
- Barker T, Bashmakov I, Bernstein L, Bogner JE, Bosch PR, Dave R, Metz B, Nabuurs GJ (2007) Contribution of working group III to the fourth assessment report of the IPCC: technical summary. In: Climate change 2007: mitigation contribution of working group III to the fourth assessment report of the intergovernmental panel on climate change, intergovernmental panel on climate change 2009. Cambridge University Press, pp 27–93
- Barwise Y, Kumar P (2020) Designing vegetation barriers for urban air pollution abatement: a practical review for appropriate plant species selection. npj Clim Atmos Sci 3(1):1–19
- Behnke K, Kleist E, Uerlings R, Wildt J, Rennenberg H, Schnitzler JP (2009) RNAi-mediated suppression of isoprene biosynthesis in hybrid poplar impacts ozone tolerance. Tree Physiol 29(5):725–736
- Bhatt MR (2018) Is air pollution a disaster in Indian cities? South Asia Disasters Issue 172:2-3
- Bhuyan P, Deka P, Prakash A, Balachandran S, Hoque RR (2018) Chemical characterization and source apportionment of aerosol over mid Brahmaputra Valley, India. Environ Pollut 234:997– 1010
- Blagnyte R, Paliulis D (2010) Research into heavy metals pollution of atmosphere applying moss as bioindicator: a literature review. Environ Res Eng Manag 54(4):26–33
- Bower JS, Broughton GF, Stedman JR, Williams ML (1994) A winter NO2 smog episode in the UK. Atmos Environ 28(3):461–475
- Boya J (2012) Ecological strategy of green urban design in urban renewal. The specific ecological measures for the existing urban ecological problems in China. In: 48th ISOCARP congress
- Bradley EF, Mulhearn PJ (1983) Development of velocity and shear stress distribution in the wake of a porous shelter fence. J Wind Eng Ind Aerodyn 15(1–3):145–156
- Branford D, Fowler D, Moghaddam MV (2004) Study of aerosol deposition at a wind exposed forest edge using 210 Pb and 137 Cs soil inventories. Water Air Soil Pollut 157(1–4):107–116
- Burney J, Ramanathan V (2014) Recent climate and air pollution impacts on Indian agriculture. Proc Natl Acad Sci 111(46):16319–16324
- Cabaraban MT, Kroll CN, Hirabayashi S, Nowak DJ (2013) Modeling of air pollutant removal by dry deposition to urban trees using a WRF/CMAQ/i-Tree Eco coupled system. Environ Pollut 176:123–133
- Calfapietra C, Fares S, Manes F, Morani A, Sgrigna G, Loreto F (2013) Role of Biogenic Volatile Organic Compounds (BVOC) emitted by urban trees on ozone concentration in cities: a review. Environ Pollut 183:71–80
- Calfapietra C, Peñuelas J, Niinemets Ü (2015) Urban plant physiology: adaptation-mitigation strategies under permanent stress. Trends Plant Sci 20(2):72–75
- Cenci RM, Sena F, Bergonzoni M, Simonazzi N, Meglioli E, Canovi L, Locoro G, Trincherini P (2003) Use of mosses and soils for the monitoring of trace elements in three landfills, used as urban waste disposal sites. In: [Sardinia proceedings 2003]: from the ninth international waste management and landfill symposium, occurred on 6–10 October
- Chen J, Li C, Ristovski Z, Milic A, Gu Y, Islam MS, Wang S, Hao J, Zhang H, He C, Guo H (2017) A review of biomass burning: emissions and impacts on air quality, health and climate in China. Sci Total Environ 579:1000–1034
- Churkina G, Grote R, Butler TM, Lawrence M (2015) Natural selection? Picking the right trees for urban greening. Environ Sci Pol 47:12–17
- Crutzen PJ (2006) The "anthropocene". In: Earth system science in the anthropocene. Springer, Berlin, Heidelberg, pp 13–18

- Daimari R, Hoque RR, Nayaka S, Upreti DK (2013) Atmospheric heavy metal accumulation in epiphytic lichens and their phorophytes in the Brahmaputra Valley. Asian J Water Environ Pollut 10(4):1–2
- Daimari R, Bhuyan P, Hussain S, Nayaka S, Mazumder MJ, Hoque RR (2020) Biomonitoring by epiphytic lichen species—Pyxine cocoes (Sw.) Nyl.: understanding characteristics of trace metal in ambient air of different landuses in mid-Brahmaputra Valley. Environ Monit Assess 192(1):37
- De US, Dandekar MM (2001) Natural disasters in urban areas. Deccan Geogr 39(2):1-12
- De US, Dube RK, Rao GP (2005) Extreme weather events over India in the last 100 years. J Ind Geophys Union 3:173–187
- Deka P, Hoque RR (2014) Incremental effect of festive biomass burning on wintertime PM10 in Brahmaputra Valley of Northeast India. Atmos Res 143:380–391
- Deka P, Hoque RR (2015) Chemical characterization of biomass fuel smoke particles of rural kitchens of South Asia. Atmos Environ 108:125–132
- Dey S, Tripathi SN, Singh RP, Holben BN (2004) Influence of dust storms on the aerosol optical properties over the Indo-Gangetic basin. J Geophys Res Atmos 109(D20)
- Dimri AP, Niyogi D, Barros AP, Ridley J, Mohanty UC, Yasunari T, Sikka DR (2015) Western disturbances: a review. Rev Geophys 53(2):225–246
- Dimri AP, Chevuturi A, Niyogi D, Thayyen RJ, Ray K, Tripathi SN, Pandey AK, Mohanty UC (2017) Cloudbursts in Indian Himalayas: a review. Earth Sci Rev 168:1–23
- Dockery DW, Pope CA, Xu X, Spengler JD, Ware JH, Fay ME, Speizer FE (1993) An association between air pollution and mortality in six US cities. N Engl J Med 329(24):1753–1759
- Dragović S, Mihailović N (2009) Analysis of mosses and topsoils for detecting sources of heavy metal pollution: multivariate and enrichment factor analysis. Environ Monit Assess 157(1–4): 383–390
- EC (2015) Towards an EU research and innovation policy agenda for nature-based solutions & re-naturing cities. European Commission, Brussels
- ETH Press Release (2019) How trees could help to save the climate. https://ethz.ch/en/news-andevents/eth-news/news/2019/07/how-trees-could-save-the-climate.html. Accessed 17 Sept 2020
- Faldt J, Arimura GI, Gershenzon J, Takabayashi J, Bohlmann J (2003) Functional identification of AtTPS03 as (E)-β-ocimene synthase: a monoterpene synthase catalyzing jasmonate-and woundinduced volatile formation in Arabidopsis thaliana. Planta 216(5):745–751
- Fernandez CC, Shevock JR, Glazer AN, Thompson JN (2006) Cryptic species within the cosmopolitan desiccation-tolerant moss Grimmia laevigata. Proc Natl Acad Sci 103(3):637–642
- Fowler D, Skiba U, Nemitz E, Choubedar F, Branford DE, Donovan R, Rowland P (2004) Measuring aerosol and heavy metal deposition on urban woodland and grass using inventories of 210 Pb and metal concentrations in soil. In: Biogeochemical investigations of terrestrial, freshwater, and wetland ecosystems across the globe. Springer, Dordrecht, pp 483–499
- Garvey DM (1975) Testing of cloud seeding materials at the cloud simulation and aerosol laboratory, 1971–1973. J Appl Meteorol 14(5):883–890
- Gautam R, Hsu NC, Kafatos M, Tsay SC (2007) Influences of winter haze on fog/low cloud over the Indo-Gangetic plains. J Geophys Res Atmos 112(D5)
- George JJ (1951) Fog. In: Malone TF (ed) Compendium of meteorology. American Meteorological Society, Boston, pp 1179–1189
- GFDRR U. EU (2014) Planning and implementation of post-Sidr housing recovery: practice, lessons and future implications: recovery framework case study
- Ghude SD, Bhat GS, Prabhakaran T, Jenamani RK, Chate DM, Safai PD, Karipot AK, Konwar M, Pithani P, Sinha V, Rao PS (2017) Winter fog experiment over the Indo-Gangetic plains of India. Curr Sci 112(4):00113891
- Giordano S, Sorbo S, Adamo P, Basile A, Spagnuolo V, Cobianchi RC (2004) Biodiversity and trace element content of epiphytic bryophytes in urban and extraurban sites of southern Italy. Plant Ecol 170(1):1–14

- Gromke C, Jamarkattel N, Ruck B (2016) Influence of roadside hedgerows on air quality in urban street canyons. Atmos Environ 139:75–86
- Grote R, Samson R, Alonso R, Amorim JH, Cariñanos P, Churkina G, Fares S, Thiec DL, Niinemets Ü, Mikkelsen TN, Paoletti E (2016) Functional traits of urban trees: air pollution mitigation potential. Front Ecol Environ 14(10):543–550
- Guo X, Zheng G (2009) Advances in weather modification from 1997 to 2007 in China. Adv Atmos Sci 26(2):240–252
- Guo X, Fu D, Li X, Hu Z, Lei H, Xiao H, Hong Y (2015) Advances in cloud physics and weather modification in China. Adv Atmos Sci 32(2):230–249
- Gupta V, Dobhal DP, Vaideswaran SC (2013) August 2012 cloudburst and subsequent flash flood in the Asi Ganga, a tributary of the Bhagirathi river, Garhwal Himalaya, India. Curr Sci 25:249– 253
- Guttikunda SK, Goel R (2013) Health impacts of particulate pollution in a megacity—Delhi, India. Environ Dev 6:8–20
- Hadba L, Mendonça P, Silva LT (2017) Green walls an efficient solution for hygrothermal, noise and air pollution control in the buildings. In Architecture media politics and society conferenceliving and sustainability: an environmental critique of design and building practices, local and globally, vol 1
- Hazarika N, Daimari R, Nayaka S, Hoque RR (2011) What do epiphytic lichens of Guwahati city indicate? Curr Sci 101(7):824
- Helln H, Tykk T, Hakola H (2012) Importance of monoterpenes and isoprene in urban air in northern Europe. Atmos Environ 59:5966
- Henderson JV, Venables AJ, Regan T, Samsonov I (2016) Building functional cities. Science 352(6288):946–947
- Hewitt CN, Ashworth K, MacKenzie AR (2020) Using green infrastructure to improve urban air quality (GI4AQ). Ambio 1:62–73
- Hoque RR, Deka P (2010) Aerosol and CO emissions during meji burning. Curr Sci (Bangalore) 10: 1270
- Hsu SC, Gong GC, Shiah FK, Hung CC, Kao SJ, Zhang R, Chen WN, Chen CC, Chou CK, Lin YC, Lin FJ (2014) Sources, solubility, and acid processing of aerosol iron and phosphorous over the South China Sea: East Asian dust and pollution outflows vs. Southeast Asian biomass burning. Atmos Chem Phys Discuss 14(15):21433–21472
- Hutt R (2017) China is about to get its first vertical forest. World Economic Forum. https://www. weforum.org/agenda/2017/05/china-is-about-to-get-its-first-vertical-forest/. Accessed 26 Sept 2020
- Illiyas FT, Mohan K, Mani SK, Pradeepkumar AP (2014) Lightning risk in India: challenges in disaster compensation. Econ Polit Wkly 49(23):23–27
- International Energy Agency (IEA) (2011) World Energy Outlook, Energy for all, financing access for the poor. Paris Int. Energy Agency
- ISSA Design Technical Bulletins (2005) International Slurry Surfacing Association
- IUCN (2012) The IUCN Programme 2013–2016, adopted by the IUCN World Conservation Congress, September 2012. International Union for Conservation of Nature, Gland
- Jain M, Saxena P, Sharma S, Sonwani S (2021) Investigation of forest fire activity changes over the central India domain using satellite observations during 2001–2020. GeoHealth 5: e2021GH000528
- Jenamani RK (2012) Development of intensity based fog climatological information system (daily and hourly) at IGI airport, New Delhi for use in fog forecasting and aviation. Mausam 63(1): 89–112
- Jerrett M, Burnett RT, Pope CA III, Ito K, Thurston G, Krewski D, Thun M (2009) Long-term ozone exposure and mortality. N Engl J Med 360(11):1085–1095
- Kedia S (2017) Delhi is gasping for air—six reasons why the national capital is becoming a 'gas chamber'. https://www.yourstory.com/2017/11/delhi-air-pollution-six-reasons. Accessed 12 Sept 2020

- Kelly FJ, Kelly J (2009) London air quality: a real world experiment in progress. Biomarkers 14 (Suppl 1):5–11
- Köhler, M. (2008). Green facades-a view back and some visions. Urban Ecosystems, 11(4), 423-436
- Krewski D, Jerrett M, Burnett RT, Ma R, Hughes E, Shi Y, Turner MC, Pope CA III, Thurston G, Calle EE, Thun MJ (2009) Extended follow-up and spatial analysis of the American Cancer Society study linking particulate air pollution and mortality. Health Effects Institute, Boston
- Krzyzanowski M, Apte JS, Bonjour SP, Brauer M, Cohen AJ, Prüss-Ustun AM (2014) Air pollution in the mega-cities. Curr Environ Health Rep 1(3):185–191
- Kulshrestha U, Saxena P (eds) (2016) Plant responses to air pollution. Springer, Singapore, pp 1–195
- Kumar P, Gulia S, Harrison RM, Khare M (2017) The influence of odd–even car trial on fine and coarse particles in Delhi. Environ Pollut 225:20–30
- Kumar A, Gupta AK, Bhambri R, Verma A, Tiwari SK, Asthana AK (2018) Assessment and review of hydrometeorological aspects for cloudburst and flash flood events in the third pole region (Indian Himalaya). Pol Sci 18:5–20
- Lal PN, Mitchell T, Aldunce P, Auld H, Mechler R, Miyan A, Romano LE, Zakaria S (2012) National systems for managing the risks from climate extremes and disasters. In: Field CB, Barros V, Stocker TF, Qin D, Dokken DJ, Ebi KL, Mastrandrea MD, Mach KJ, Plattner G-K, Allen SK, Tignor M, Midgley PM (eds) Managing the risks of extreme events and disasters to advance climate change adaptation. A special report of working groups I and II of the intergovernmental panel on climate change (IPCC). Cambridge University Press, Cambridge and New York, pp 339–392
- Lawson T, Blatt MR (2014) Stomatal size, speed, and responsiveness impact on photosynthesis and water use efficiency. Plant Physiol 164(4):1556–1570
- Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, Adair-Rohani H, AlMazroa MA, Amann M, Anderson HR, Andrews KG, Aryee MA (2012) A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet 380(9859):2224–2260
- Liu Y, Harris DJ (2008) Effects of shelterbelt trees on reducing heating-energy consumption of office buildings in Scotland. Appl Energy 85(2–3):115–127
- Luckin B (2003) 'The heart and home of horror': the great London fogs of the late nineteenth century. Soc Hist 28(1):31–48
- Mannucci PM, Franchini M (2017) Health effects of ambient air pollution in developing countries. Int J Environ Res Public Health 9:1048
- Marlier ME, Jina AS, Kinney PL, DeFries RS (2016) Extreme air pollution in global megacities. Curr Clim Chang Rep 2(1):15–27
- Marrapu P, Cheng Y, Beig G, Sahu S, Srinivas R, Carmichael GR (2014) Air quality in Delhi during the Commonwealth Games. Atmos Chem Phys 14(19):10619–10630
- Masiol M, Hopke PK, Felton HD, Frank BP, Rattigan OV, Wurth MJ, LaDuke GH (2017) Source apportionment of PM2. 5 chemically speciated mass and particle number concentrations in New York City. Atmos Environ 148:215–229
- Middleton N, Kang U (2017) Sand and dust storms: impact mitigation. Sustainability 9(6):1053
- Minea G (2013) Assessment of the flash flood potential of Bâsca River Catchment (Romania) based on physiographic factors. Open Geosci 5(3):344–353
- Monastersky R (2015) Anthropocene: the human age. Nat News 519(7542):144
- MoRTH (2008) Annual report 2008–2009. Ministry of road transport & highways. New Delhi NIDM (2020). https://nidm.gov.in/about.asp. Accessed 4 Oct 2020
- Niles JO, Brown S, Pretty J, Ball AS, Fay J (2002) Potential carbon mitigation and income in developing countries from changes in use and management of agricultural and forest lands. Philos Trans A Math Phys Eng Sci 360(1797):1621–1639

- Nowak DJ, Crane DE, Stevens JC (2006) Air pollution removal by urban trees and shrubs in the United States. Urban For Urban Green 4(3–4):115–123
- Oke TR (1988) Street design and urban canopy layer climate. Energy Buildings 11(1-3):103-113
- Oldenborgh GJ, Philip S, Kew S, Weele MV, Uhe P, Otto F, Singh R, Pai I, Cullen H, AchutaRao K (2018) Extreme heat in India and anthropogenic climate change. Nat Hazards Earth Syst Sci 18(1):365–381
- Pacheco-Torgal F, Cabeza LF, Labrincha J, De Magalhaes AG (2014) Eco-efficient construction and building materials: life cycle assessment (LCA), eco-labelling and case studies. Woodhead Publishing
- Padro J (1996) Summary of ozone dry deposition velocity measurements and model estimates over vineyard, cotton, grass and deciduous forest in summer. Atmos Environ 30(13):2363–2369
- Papiez MR, Potosnak MJ, Goliff WS, Guenther AB, Matsunaga SN, Stockwell WR (2009) The impacts of reactive terpene emissions from plants on air quality in Las Vegas, Nevada. Atmos Environ 43:4109–4123
- Pasricha PK, Gera BS, Shastri S, Maini HK, John T, Ghosh AB, Tiwari MK, Garg SC (2003) Role of the water vapour greenhouse effect in the forecasting of fog occurrence. Bound-Layer Meteorol 107(2):469–482
- Pommier M, McLinden CA, Deeter M (2013) Relative changes in CO emissions over megacities based on observations from space. Geophys Res Lett 40(14):3766–3771
- Pope CA III, Ezzati M, Dockery DW (2009) Fine-particulate air pollution and life expectancy in the United States. N Engl J Med 360(4):376–386
- Porter WC, Heald CL, Cooley D, Russell B (2015) Investigating the observed sensitivities of air-quality extremes to meteorological drivers via quantile regression. Atmos Chem Phys 15(18):10349–10366
- Raine JK, Stevenson DC (1977) Wind protection by model fences in a simulated atmospheric boundary layer. J Wind Eng Ind Aerodyn 2(2):159–180
- Rao AM, Madhu E, Gupta K (2017) Impact of odd even scheme on transportation systems in Delhi. Transport Dev Econ 3(1):4
- Ratti C, Di Sabatino S, Britter R (2006) Urban texture analysis with image processing techniques: winds and dispersion. Theor Appl Climatol 84(1–3):77–90
- Read C, Parton KA (2019) The impact of the 1952 London smog event and its relevance for current wood-smoke abatement strategies in Australia. J Air Waste Manag Assoc 69(9):1049–1058
- Redclift M (2005) Sustainable development (1987–2005): an oxymoron comes of age. Sustain Dev 13(4):212–227
- Ritchie H, Roser M (2017) Air pollution. Our World in Data. https://ourworldindata.org/airpollution. Accessed 23 Sept 2020
- Rosenfeld D, Lohmann U, Raga GB, O'Dowd CD, Kulmala M, Fuzzi S, Reissell A, Andreae MO (2008) Flood or drought: how do aerosols affect precipitation? Science 321(5894):1309–1313
- Ryntathiang TL, Amar DD, Kumar R, AppaRoa G (2013) Green road approach for the sustainable development in India. Eur J Sustain Dev 2(2):165–165
- Sæbø A, Popek R, Nawrot B, Hanslin HM, Gawronska H, Gawronski SW (2012) Plant species differences in particulate matter accumulation on leaf surfaces. Sci Total Environ 427:347–354
- Safai PD, Ghude S, Pithani P, Varpe S, Kulkarni R, Todekar K, Tiwari S, Chate DM, Prabhakaran T, Jenamani RK, Rajeevan MN (2019) Two-way Relationship between aerosols and fog: a case study at IGI airport, New Delhi. Aerosol Air Qual Res 19(1):71–79
- Sarkar AN (2013) Promoting eco-innovations to leverage sustainable development of eco-industry and green growth. Eur J Sustain Dev 2(1):171–171
- Sawlani R, Agnihotri R, Sharma C, Patra PK, Dimri AP, Ram K, Verma RL (2019) The severe Delhi SMOG of 2016: a case of delayed crop residue burning, coincident firecracker emissions, and atypical meteorology. Atmos Pollut Res 10(3):868–879
- Saxena P, Ghosh C (2013) Ornamental plants as sinks and bioindicators. Environ Technol 34(23): 3059–3067

- Saxena P, Sonwani S (2019a) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Sonwani S (2019b) Policy regulations and future recommendations. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 127–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Srivastava A, Tyagi M, Kaur S (2019) Impact of tropospheric ozone on plant metabolism—a review. Pollut Res 38(1):175–180
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Sehgal M, Gautam SK (2016) Odd even story of Delhi traffic and air pollution. Int J Environ Stud 73(2):170–172
- Sehmel GA (1980) Particle and gas dry deposition: a review. Atmos Environ 14(9):983-1011
- Shaddick G, Thomas ML, Mudu P, Ruggeri G, Gumy S (2020) Half the world's population are exposed to increasing air pollution. npj Clim Atmos Sci 3(1):1–5
- Sharma AR, Kharol SK, Badarinath KV, Singh D (2010) Impact of agriculture crop residue burning on atmospheric aerosol loading—a study over Punjab State, India. Ann Geophys 28(2): 09927689
- Shenfeld L (1970) Meteorological aspects of air pollution control. Atmosphere 8(1):3-13
- Sindhwani R, Goyal P (2014) Assessment of traffic-generated gaseous and particulate matter emissions and trends over Delhi (2000–2010). Atmos Pollut Res 5(3):438–446
- Singh S (2017) These 20 plants can guzzle pollutants, finds DU study Thursday New Delhi. The Pioneer. https://www.dailypioneer.com/2017/page1/these-20-plants-can-guzzle-pollutants-finds-du-study.html. Accessed 19 Sept 2020
- Singh RK, Srivastava MK, Mehrotra BJ, Murari V, Sharma SK, Mandal TK, Singh AK, Singh RS, Banerjee T, Tiwari S (2017) Biomass-burning during Holika at Varanasi (25.3 N, 83.0 E): optical observations and aerosol radiative forcing. In: 2017 AGU fall meeting AGU
- Smith J (2012) Urban air quality. The Woodland Trust, Grantham
- Sonwani S, Saxena P, Khillare PS (2022) Profile of atmospheric particulate PAHs near busy roadway in tropical megacity, India. Inhal Toxicol 34(1–2):39–50
- Steffen W, Crutzen PJ, McNeill JR (2007) The Anthropocene: are humans now overwhelming the great forces of nature. Ambio 36(8):614–621
- Sternberg T, Viles H, Cathersides A, Edwards M (2010) Dust particulate absorption by ivy (Hedera helix L) on historic walls in urban environments. Sci Total Environ 409(1):162–168
- Tai AP, Martin MV, Heald CL (2014) Threat to future global food security from climate change and ozone air pollution. Nat Clim Chang 4(9):817–821
- Tessier L, Boisvert J (1999) Performance of terrestrial bryophytes as biomonitors of atmospheric pollution. A review. Toxicol Environ Chem 68(1–2):179–220
- Thönnessen M (2002) Elementdynamik in fassadenbegrünendem Wilden Wein (Parthenocissus tricuspidata). Kölner Geographische Arbeiten, p 78
- Tiwari S, Bisht DS, Srivastava AK, Shivashankara GP, Kumar R (2013) Interannual and intraseasonal variability in fine mode particles over Delhi: influence of meteorology. Adv Meteorol 2013:740453
- Tiwari S, Thomas A, Rao P, Chate DM, Soni VK, Singh S, Ghude SD, Singh D, Hopke PK (2018) Pollution concentrations in Delhi India during winter 2015–16: a case study of an odd-even vehicle strategy. Atmos Pollut Res 9(6):1137–1145
- Tiwary A, Morvan HP, Colls JJ (2006) Modelling the size-dependent collection efficiency of hedgerows for ambient aerosols. J Aerosol Sci 37(8):990–1015
- Tzoulas K, Galan J, Venn S, Dennis M, Pedroli B, Mishra H, Haase D, Pauleit S, Niemelä J, James P (2020) A conceptual model of the social–ecological system of nature-based solutions in urban environments. Ambio 50(2):335–345

- UNDP (United Nations, Department of Economic and Social Affairs, Population Division) (2014) World urbanization prospects: the 2014 revision. CD-ROM edition
- UNDP (United Nations, Department of Economic and Social Affairs, Population Division) (2015) World urbanization prospects: the 2014 revision,(ST/ESA/SER. A/366). United Nations Population Division
- Watson A (1985) The control of blowing sand and mobile desert dunes. The control of wind blown sand and moving dunes: a review of the methods of sand control in deserts, with observations from Saudi Arabia. Q J Eng Geol Hydrogeol 18:237–252
- Weerakkody U, Dover JW, Mitchell P, Reiling K (2018) Evaluating the impact of individual leaf traits on atmospheric particulate matter accumulation using natural and synthetic leaves. Urban For Urban Green 30:98–107
- West JJ, Smith SJ, Silva RA, Naik V, Zhang Y, Adelman Z, Fry MM, Anenberg S, Horowitz LW, Lamarque JF (2013) Co-benefits of mitigating global greenhouse gas emissions for future air quality and human health. Nat Clim Chang 3(10):885–889
- World Health Organization (WHO) (2014) Ambient (outdoor) air pollution in cities database. http:// www.who.int/phe/health_topics/



17

Policy Implications and Mitigation Strategies for Air Pollution and Atmospheric Extreme Events

Disha Sharma

Abstract

Air pollution is a public health concern worldwide. In India, deteriorating air quality is a grave issue causing nearly 1.1 million deaths in 2015 alone (Health Effects Institute, Burden of disease attributable to major air pollution sources in India, 2018). Globally, a significant proportion of human population is breathing poor air. While some megacities experience only episodic increases in air pollution, several Asian cities suffer from the problem chronically. Increases in ambient concentrations of criteria air pollutants above the prescribed air quality standards lead to a deterioration in air quality on different temporal scales. Direct emissions from natural and anthropogenic sources inject primary pollutants and their chemical transformations in the troposphere lead to the formation of secondary air pollutants. The life span of air pollutants can last from hours to decades depending on their chemical characteristics and concentrations at emissions. The need to control and manage the emissions of air pollutants and their ambient concentrations has been addressed through several international and national policy mechanisms that have paved a way for devising control strategies based on scientific solutions. However, these policy actions have not always provided a win-win solution owing to the complex nature of the problem involving multiple sectors and stakeholders. Learning from successful examples of coordinated implementation and timely achievement of objectives (as in the case of Montreal Protocol), it is imperative to build institutions that mature in their approach as the science on the issue develops. This is particularly applicable in the case of air pollution extremes where it is crucial to look for technological innovation while ensuring coordination between multiple agencies. This chapter reviews all the

D. Sharma (🖂)

School of Public and International Affairs, Princeton University, Princeton, NJ, USA e-mail: dishas@princeton.edu

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_17

existing international and national policies that outline measures to curtail and abate ambient air pollution in chronic and episodic events as well as targeting abatement from natural and anthropogenic sources. The first half of the chapter covers international policies that have been implemented to address the transboundary pollution with global implications. In the second part, policies adopted in India in response to abatement of deteriorating air quality at the state and national level are reviewed.

Keywords

International policies on air pollution \cdot Air quality extremes in India \cdot Air pollution mitigation measures in India \cdot Clean air regulations and laws \cdot India

17.1 Introduction

Air pollution poses a grave danger to human health (Brauer et al. 2013), environment, economy and climate (Von Schneidemesser et al. 2015; Saxena and Srivastava 2020). It is a bigger problem lately, mostly in Asia, causing nearly 1.1 million deaths in 2015 in India (Health Effects Institute 2018). More than 660 million people, equal to almost half of country's population, live in areas that exceed the national air quality standards for PM_{2.5} (40 μ g/m³) and only 1% of total population lives in areas that meet the World Health Organization (WHO) standard (10 μ g/m³) (WHO 2016). While breathing toxic air is a public health emergency, direct emissions of aerosols along with secondary aerosol formation from sulfates and ozone can alter the earth's heat balance and atmospheric chemistry (Li et al. 2019). Poor air quality has a direct impact on the wellbeing of humans along with the potential of altering atmospheric balance and hence is increasingly addressed through policy measures at national and international level.

Air pollution is a policy challenge owing to various factors (World Bank 2008). Multiple sources of emissions from different sectors make it difficult to implement proper sectoral controls which further increase complexity in proposing a policy solution (Dominici et al. 2010). Inter and intra state transport of air pollution makes it have a greater impact on regions downwind of pollution sources. This decreases the willingness of the emitter regions to invest in mitigation, both at national and subnational levels. The transboundary nature of air pollution makes it imperative to have cooperation and collaborations on different spatio-temporal scales at international, national level, and sub national levels to find effective strategies for mitigation. Multiple policy solutions have been proposed over the years in response to gradually and sporadically deteriorating air quality in different parts of the world (Saxena and Sonwani 2019a). The two air pollution categories, gradual increase over the years and extreme events, require different framework for mitigation. Additionally, extremities in air pollution can result from events with regional (volcanic eruptions, wide scale fires, dust storms, etc.) and local (local forest fires, dust storms, agricultural burning, industrial emissions, etc.) implications. Sporadic

events require prior preparedness to avert the predicted disaster. Policies to that end need deliberations that ensure consensus building ahead of time in order to avert the predicted air quality extremes. However, the sporadic nature of some of those events, especially from natural sources, makes it a bigger challenge to address extremities in air quality deterioration through policies. This chapter provides an overview of the international and regional cooperation that has resulted from the collective need to address air pollution, from time to time. Specifically, it covers policies framed around criteria and noncriteria pollutants, including SO_2 , NO_x , O_3 , dust, etc. It also includes the reasons for the success or failure of policy solutions adopted internationally. Institutional framework for guiding mitigation is contextual to consensus building on decision ready science in global and regional mitigation. For India, the role of measures taken to address growing air pollution and response to extreme events, and their implementation within a federal framework are discussed. Although, there exist multiple policy mechanisms for addressing different air quality and related extreme events, their effectiveness is determined by coordination between different stakeholders including government, scientists, policy experts, and the public toward ensuring compliance and measures to curb emissions (Saxena and Sonwani 2019b).

17.2 Global Actions Toward Air Pollution Control

17.2.1 United Nations Economic Commission for Europe (UNECE) Convention on Long-Range Transboundary Air Pollution (CLRTAP)

Early in 1960s, "acid rain" was identified to be harmful for the forest and water ecosystem that affected trees and fish in the northern hemisphere. Air pollution from emissions thousands of kilometers away from the affected regions was responsible for the acid rain events. Scientific evidence indicated that emissions in continental Europe were causing acidification of Scandinavian lakes (UNECE 1979). The 1972 United Nations conference on Human Environment in Stockholm called for international collaborations to address the issue. In the following years, scientific evidence provided a basis to support the hypothesis that local air pollution could have global environmental implications. As a result, first multilateral agreement to address transboundary air pollution materialized in the form of Convention on Long-Range Transboundary Air Pollution (LRTAP) in 1979 (UNCLRTAP 1979). Initially, 32 countries came together to form a regional framework to address transboundary air pollution and build scientific knowledge about the phenomenon. It was established with active engagement of Europe, North America, and Russia and former East Bloc countries. This was the first international accord that was legally binding wherein air pollution abatement through scientific research and policy solutions was planned on a regional scale.

Over the years, the objectives of the convention have broadened and include more pollutants that negatively affect human health and natural ecosystems (Wettestad

2001). It is a successful example of international and intergovernmental cooperation on issues of local emissions that have a far-reaching impact. At present, there are eight protocols that identify specific measures that need to be taken to reduce emissions. Under the convention, regional cooperation to pool resources for tracking emissions and modeling and measurements exercise is facilitated. Institutional framework under LRTAP has contributed to the decline in emissions of criteria air pollutants in the region. Efforts under the convention have resulted in decrease in sulfur emissions in the region and progressively decoupled economic growth from increasing pollution trend. Collective effort under the convention has been significant in achieving 40-80% reduction in emissions of harmful pollutants (US Department of State). As a result, atmospheric acidic deposition has also reduced and resulted in lower critical loads of acidity in Europe. There has been a drop in nitrogen emissions but not to the same extent as the drop in sulfur emissions. This is attributed as the cause of existing problem of acidity in the forests in Europe (Michel and Seidling 2016). Another accomplishment of the convention has been in controlling lead emissions by nearly 80% between 1990 and 2012 (Lorenz and Becher 2014). The highest reduction rates were reported in the initial period of compliance with the framework, in countries like Finland, Denmark, Spain, Germany, etc. (UNECE 2015).

Following is a brief account of protocols within UN CLRTAP, which provide controls for sulfur, NOx, and ozone emissions:

17.2.1.1 Helsinki Protocol for Reduction of Sulfur Emissions

In 1985, Helsinki Protocol for reduction of sulfur emissions was adopted by the Executive Body of CLRTAP, in Helsinki, Finland (Oslo Protocol 2021). It aimed at a 30% reduction in emissions of sulfur oxides in the atmosphere and their transboundary fluxes (Ringquist and Kostadinova 2005). The base year relative to which the reduction was planned was 1980. Under the protocol, all the signatory parties were obliged to develop strategies at national level, including action plan and programs to target reductions as pledged under the international accord. Additionally, the parties were also required to assess the need for reductions further than what the target laid out under the protocol (Aakvik and Tjøtta 2011). Other protocols addressing reduction in sulfur emissions include the 1994 Protocol on Further Reduction of Sulfur Emissions and 1999 Gothenburg Protocol to Abate Acidification, Eutrophication, and Ground-level Ozone.

The 1994 Oslo Protocol for Further Reduction of Sulfur Emissions came into force in August 1998. It adopted an effects-based approach for planning emissions reduction by Parties to the Protocol. The additional feature of the protocol includes an Implementation Committee consisting of eight parties that is responsible for conducting a period review of compliance and proposal of constructive solutions in cases of noncompliance.

17.2.1.2 Sofia Protocol for the Control of Emissions of Nitrogen Oxides or Their Transboundary Fluxes

The Sofia Protocol for the Control of Emissions of Nitrogen Oxides or their Transboundary Fluxes was adopted in 1988 in Sofia (Bulgaria). The protocol established emissions reduction targets for nitrogen oxides with 1987 as the reference year. United States was the only country that chose emission target reduction with respect to 1978. The protocol additionally laid out an effects-based approach wherein the assessment of impacts on affected regions requires scientific analysis and research. It also targets reduction in ammonia and VOCs for their role in eutrophication, photochemical smog, and impacts on human health (Mosier et al. 2002).

17.2.1.3 The 1999 Gothenburg Protocol

The 1999 Gothenburg Protocol to Abate Acidification, Eutrophication, and Groundlevel Ozone in Gothenburg (Sweden) was adopted by the Executive Body in November 1999 (Gothenburg Protocol 2021). Based on scientific research, this is an important protocol that sets limits on emissions on four pollutants—sulfur, NOx, VOCs, and ammonia. The limits sets were negotiated based on severity of emissions and their impacts and the resources available for mitigation (Vestreng et al. 2007). Strict limits on emission pertaining to specific sectors, like industries, electricity, dry cleaning, and vehicles are set. Best available technologies have to be deployed to meet emission limits with the help of collective resources and economic instruments available through the protocol. Upon its full implementation, the protocol would achieve 63% reduction in sulfur emissions, 41% in NOx emissions, 40% in VOC emissions, and 17% in ammonia emissions compared to 1990, in Europe (UNECE).

17.2.2 Montreal Protocol

Montreal Protocol is regarded as one of the most successful international environmental treaties which achieved its objective of prevention of ozone depletion by banning the production and use of ozone-depleting substances that include CFCs, HCFCs, and halons (Montreal Protocol 1987). It is an example of unprecedented level of coordination and commitment shown by the international community. There are many reasons accorded to its success with effective leadership that facilitated collaboration between developed and developing nations being at the helm. Initially, the science on the ozone-depleting phenomenon was not definite and initial estimates of depletion were highly underestimated. Thus, to begin with the framework was modestly designed and was highly flexible to increase or decrease controls as the science developed. With this flexibility, additions were made to the list of chemicals which were to be completely banned rather than being partially phased out (Green 2009). This arrangement built greater confidence in the process. The protocol could achieve a wide scale ratification by putting trade restrictions on countries in the supply chain of banned substances. So, once the producer countries ratified, the consumer countries also followed suit fearing losing trade of the limited supply.

Another reason for the success of the protocol was taking "precautionary principles" even when the science was not fully developed. The approach of common but differentiated responsibilities was also adopted with developing countries given more time to phase out the substances. Another reason for its success was identification of the chemicals and sectors which enabled governments to prioritize actions under the accord. It also provided a framework for long-term planning and innovation in the industry (Velders et al. 2007). It proved to be doubly beneficial as technological innovation led to the production of nonozone-depleting alternatives which had zero global warming potential unlike their former counterparts. Timely assessment and support for compliance also encouraged countries toward early transition using cleaner alternatives.

In 2016, Kigali amendment to the Montreal Protocol was introduced in Kigali, Rwanda. According to this amendment, timelines were proposed for countries to freeze and phase out the production and usage of hydrofluorocarbons (HCFCs) (Kigali amendment, UNIDO 2021). Although, HCFCs do not cause ozone depletion but have a very high greenhouse potential and contribute to climate change. If fully implemented, the Kigali Amendment can prevent 80 gigatonnes of carbon dioxide equivalent (GtCO2eq) being added to the atmosphere by 2050, equated to a reduction of 0.5 °C by the end of the century compared to business as usual (Heath 2017; Polonara et al. 2017; Montreal Protocol 2021).

17.2.3 UN Coalition on Combating Sand and Dust Storms

Sand and dust storms (SDS) are a natural phenomenon linked with land, soil and water management, and climate change. They have the potential to negatively impact the environment by reducing visibility and damaging infrastructure, agriculture, industry, and human health (Sterk et al. 2012). Parts of Africa and the Middle East experience very intense dust storms known by different names such as sirocco, haboob, yellow dust, white storms, and the harmattan. In India, they are referred to as "aandhi". Most active dust sources are found in dry and arid climate zone but can develop anywhere as a result of human influences (Wiesinger et al. 2016). The fluctuation in their intensity and magnitude makes them highly unpredictable and challenging for accurate forecasts.

In 1994, the UN Convention to Combat Desertification (UNCCD 1994) came into being as a binding international legal agreement that linked environmental sustainability directly with developmental pathways adopted in different countries (UNCCD). The Convention focuses on sustainable practices with respect to land use, soil and water management in the arid and semiarid regions, also known as drylands, which are the source regions for dust storms. They also host the most vulnerable ecosystems and peoples accustomed to living in a special habitat. The convention has been ratified by 197 countries. Out of these, some 151 are directly affected by SDS and 45 are recognized as source regions. Poor ecosystem management can create dust sources in any environment. Land degradation and climate change-induced droughts can impact severity and duration of SDS. Thus, the key to mitigate SDS is sustainable developmental planning. Under SDS agreement, parties have to (1) integrate mitigation measures into the subnational, national, and regional implementation of the convention; (2) enhance preparedness and resilience of vulnerable ecosystems and populations to the adverse and negative impacts of SDS; and (3) strengthen multistakeholder platforms and regional initiatives that contribute to addressing SDS. The Global SDS Source Base map developed by the UNCCD secretariat in collaboration with UNEP and WMO provides information for policy development and implementation including risk and vulnerability assessment, modeling, forecasting and early warning, impact, and source mitigation.

17.2.3.1 World Meteorological Organization (WMO) Sand and Dust Storm Warning Advisory and Assessment System (SDS-WAS)

The WMO sand and dust storm project was initiated in 2004 and its Sand and Dust Storm Warning Advisory and Assessment System (SDS-WAS) was launched by the Fifteenth World Meteorological Congress in 2007 (UNEP 2016). The main objective of the advisory is to help increase the country's ability in timely and accurate forecasts of dust storms. It also provides support for building observation networks and dust storm modeling. International partnerships involving research communities help in knowledge exchange in the affected regions (Nickovic et al. 2015).

17.3 India-Specific Policy Response

India is one of the first countries to include an amendment in the constitution that holds the state accountable for improving and protecting the environment to safeguard public health, wildlife, and forests (Bhave and Kulkarni 2015). The 42nd constitutional amendment adopted in 1976 was in response to the commitments made for managing air quality and natural resources in the country at the Stockholm conference on Human Environment in 1972 (1). Along with the national commitments for protection of the environment, improvement in air quality was also ensured under the Directive Principles of State Policy (Article 48-A) 38 and Fundamental Duties (Article 51-Ag) 39 under the Constitution of India (1). Right to clean air has also been recognized as a part of the right to life under Article 21 of the Indian. The legal provisions guaranteed by the constitution provide a template for policy solutions to different temporal and spatial scales of air pollution that the country engages to address.

Addressing various scales of pollution, the Air (Prevention and Control of Pollution) Act was formulated in 1981. This was the first act that explicitly addressed prevention, control, and abatement of deteriorating air quality in the country. This act came into being in response to the commitment to ensure better air quality made at the international platform, the Stockholm Conference 1972. Under this act, the Central Pollution Control Board (CPCB) was set up with the objectives of monitoring air pollution as India adopted a developmental path in the following decades.

What followed was the sister act, called the Environmental Protection Act 1986. This act was the umbrella act that included protection of environment in general and leads to the formulation of other related rules and laws. The notification for lead free petrol and use of catalytic converters in vehicles to reduce tailpipe emissions was ensured through this act (6). These two provisions are setting stage for present day and future regulations for air quality mitigation in the country. Following is a discussion on the existing policy measures that ensure management and control of air pollution in the country.

17.3.1 The Climate and Clean Air Coalition (CCAC)

This is the first global arrangement aimed at air pollution mitigation involving collaborations between governments, civil society, and the private sector. It targets reduction of short-lived climate pollutants emitted from multiple sectors. It aims at mitigation of near term climate change by providing a platform for devising collective solution through knowledge and resource sharing. The coalition was initially focused on reducing emissions of methane, black carbon, and HFCs, but soon realized the importance of short-lived climate pollutants to support action on greenhouse gas reduction. The objectives under the coalition will be achieved by building national and regional actions and building capacity for mitigation; by raising awareness about the short-lived climate pollutant and their impacts; by promoting best practices and sharing successful examples of control and mitigation; building scientific knowledge for effective actions (CCAC).

17.3.2 National Clean Air Program (NCAP)

The National Clean Air Program (NCAP) was launched by the Ministry of Environment, Forests, and Climate Change in 2019 as an urgent response to curtail and control growing air pollution levels in the country (NCAP 2019). Under NCAP, reduction in air pollution levels in cities is proposed by enabling better monitoring and cross-sectoral collaboration of agencies that can help mitigate the problem. The plan seeks to achieve a 20–30% reduction in concentrations of particulate matter, PM_{10} and $PM_{2.5}$, by 2024, compared to 2017 levels. Initially launched as a 5 year plan, it can be extended if the required outcomes are not achieved in the stipulated period.

17.3.2.1 Structure and Functioning

NCAP seeks intersectoral collaborations and coordination between different ministries and government bodies that are responsible for managing different air pollution emission sources. These include the Ministry of Road Transport and Highway, Ministry of Petroleum and Natural Gas, Ministry of New and Renewable Energy, Ministry of Heavy Industry, Ministry of Housing and Urban Affairs, Ministry of Agriculture, Ministry of Health, NITI Aayog, Central Pollution Control Board (CPCB), and the State Pollution Control Boards (SPCBs), and local bodies. Such cross-sectoral collaboration is also facilitated under the National Action Plan

on Climate Change, initiatives on electric vehicles, the Smart Cities Mission among others. The program also seeks to enable partnerships between international organizations, leading technical institutions, philanthropic organizations, and experts from industry and civil society to accomplish its targets. Additionally, the plan incorporates sectoral interventions for control of pollution from major sources including resuspended road dust, vehicular sector and industrial emissions, waste burning, and agricultural residue burning.

17.3.2.2 Challenges

While NCAP is mandated by the central government, the program is very city focused. The 102 nonattainment cities that fail to meet air quality standards for criteria pollutants are required to develop city-specific action plans for air quality control. The plan also entails a budgetary allocation of Rs 300 crores for helping cities build required action plans in the first 2 years. Analysis of the plan by the Center for Energy, Environment, and Water (Ganguly et al. 2020) sheds light on the inherent weaknesses in its implementation. The report shows that the city level plans, prepared in short spans of only a few months, define measures to combat pollution, however lack timelines and priorities in achieving them. Delhi is the only city with a clear legal mandate for implementation of the program, notified by the Supreme Court. Target areas listed for mitigation of pollution in cities fall under several agencies which could result in fragmented accountability. Implementation and timely completion of the program could be jeopardized as a result of poor financial standing of urban local bodies and municipal corporations overlooking the mitigation activities. In the absence of national emission inventory and standard protocol for reporting air pollution emissions, uniform emission targets cannot be set for the country, which would hinder city level source reduction.

17.3.2.3 Key Takeaways

The NCAP is the first effort toward fighting air pollution at the national level, adopting a bottom-up approach starting with city scale mitigation plans involving sector-specific interventions. It also brings together multiple agencies for the first time to collaborate on the issue and coordinate the required measures. However, the city-specific plans need to be more defined in their target setting and timelines for achieving the goals of NCAP. The role of SPCBs in mapping out hotspots within the city by augmenting ground monitoring needs to be supported (Sharma and Singh 2017). NCAP currently does not include a mechanism for quantification of the impact of interventions. Rigorous scientific assessment will have to be carried out to test the impact of the proposed mitigation under the policy. The program sets an example for furthering evidence-based policy solutions for sustained actions to target air quality improvement in country.

17.3.3 Graded Response Action Plan (GRAP) in the National Capital Region (NCR)

National Capital Region (NCR) experiences pollution spikes during winter season owing to increased emissions from periodically active source, namely paddy residue burning, in the neighboring states of Punjab and Haryana. Unfavorable meteorological factors like low boundary layer height and absence of precipitation trap the toxic emissions closer to surface making air quality sore along the Indo-Gangetic Plains (IGP) (Bhatta et al. 2016; Sharma and Kulshrestha 2014). This episode coincides with the festive season of Diwali which also records a spike in criteria pollutants from a few days of firecracker use marking the festivities. Combined effect of agricultural fires, use of firecrackers, and already high emissions, from industrial, vehicular, and residential heating sector, in the region make air pollution surge in this period. Marked at the onset of winters, the increased emissions have led to lingering smog experienced in the NCR in the past couple of years (Tiwari et al. 2018).

Responding to the situation, the Ministry of Environment and Forest, under the directive of the Supreme Court of India in 2016, notified the Environment Protection Control Agency (EPCA) to implement the Graded Response Action Plan (GRAP). The measures under GRAP were notified in 2016 and formulated in 2017; it is the first plan that institutionalized emergency response in case of grave air quality in the NCR. It was an emergency measure proposed to curtail the wintertime air quality deterioration with extremely dangerous air quality recognized under a new category of Air Quality Index (AQI) as "Severe +" or "Emergency" (REF). This emergency response has been prepared by the active engagement of the government officials, scientists, health experts, civil society group, and the judiciary. It is the first plan that has been successful in bringing together multiple agencies responsible for providing a solution to the problem including—pollution control boards, industrial area authorities, municipal corporations, regional officials of the India Meteorological Department, and others. Coordination between the state governments in Delhi, Haryana, Rajasthan, and Uttar Pradesh, states encompassing the NCR, is also provisioned in the plan.

Under GRAP, strict measures to immediately control harmful sources of criteria pollutants are implemented with the provision of penalizing the violators. The national and Delhi government appointed vigilance teams to ensure the plan's implementation and compliance. Some of the measures provisioned under GRAP when AQI is in "Severe+" include barring trucks, other than those carrying essential commodities, from entering Delhi; a complete halt in construction activities, implementing odd even policy for private vehicles, taking additional measures like shutting schools for the duration of high pollution on expert advice (GRAP, CPCB 2016). Applicable since 2017, in 2018, ban on diesel generators was implemented only in Delhi, but in 2019 it was extended to other NCR towns (TOI ref). Rural areas, however, were not included in this ban because of the unreliable power supplies in those areas. Air pollution control measures under other AQI categories described in the GRAP document include closing brick kilns, hot mix plats, banning use of Pet coke as fuel, shutting of Badarpur Power Plant, intensification of public transport

and introduction of BS-VI fuel standards in Delhi, mechanized cleaning, and use of water sprinklers on road to control curbside dust resuspension, etc.

GRAP is the first action plan proposed for Delhi that guides action to curtail predicted extreme levels of air pollutants. As an emergency plan, it has been successful in fixing accountability through interagency cooperation and ensuring time bound action. Although, GRAP is a successful example showcasing feasibility of coordination between multiple stake holders on the issue, severity of pollution episodes is still a matter of concern (REF) pointing at a similar provision for long-term pollution mitigation (Gulia et al. 2021).

17.3.3.1 National Policy for Management of Crop Residues

Crop residue burning in India is recognized as one of the major factors responsible for high air pollution levels along the IGP. Unfavorable meteorological conditions exacerbated by calm surface winds, low precipitation, high surface pressure, and low boundary layer height trap the pollutants close to the surface at the onset of winters, leading to smog-like conditions affecting major cities, including Delhi, along the IGP (Guttikunda and Gurjar 2012; Saxena et al. 2021). Over the years, stubble burning in the rice fields of Haryana and Punjab majorly, has become a cause of concern. Use of new machinery for harvesting rice crop and the short window of two available for preparing the field for sowing wheat, generated standing stubble on fields. In the absence of cost-effective alternatives, farmers resort to burning the stubble and the smoke plume generated travels all the way along the IGP. This plume leads to green house gas emissions along with the emissions of other trace gases and aerosols including CH_4 , CO, N₂O, NO₂, and other hydrocarbons and large amounts of particulate matter.

The policy (NPMCR 2014) to manage crop residue seeks technical and policy interventions to alleviate the issue in the respective states. It states its objectives as: (A) controlling crop residue burning to prevent loss of soil nutrients and minerals by in situ management of crop residue. (B) It also aims at diversified use of crop residue for various purposes like charcoal gasification, power generation, as industrial raw material for production of bio-ethanol, packing material, paper/board/panel industry, composting and mushroom cultivation, etc. (C) Outreach and awareness about the ill effects of the process. (D) Finding technical and policy solution to address the issue in the long run. The policy seeks to achieve this objective through—(1) Promotion of technologies that would guarantee optimum utilization and in situ management of stubble. (2) Use of residue for other purposes like power generation, as industrial raw material for production of bio-ethanol, packaging material for fruit, in paper pulp manufacturing, biogas generation, etc. (3) Capacity building of stakeholders to opt for alternatives. (4) Suitable laws and incentives enabling farmers to opt for better machinery and participate in public-private partnerships. The policy also proposes a monitoring mechanism at the national level to ensure that alternative measures are adopted and burning incidences reduced.

Institutional challenges for effective mitigation of air pollution through policies: Despite the provisions to control and abate air quality deterioration in India, there exist institutional challenges that slow the pace of mitigation actions.

- (a) In India, it is imperative that the policies adopt a bottom-up approach to effectively target the root cause in each sector. Currently, most of the policies operate in a top-down model which ignores the specificities of the problem and root cause of emissions at the point of emission. For example, in designing crop residue burning policies, farmers should be recognized as a major stakeholder and their participation in the process be ensured.
- (b) Public engagement in the issue is missing. The democratic demand for clean air has to be ensured through participation of the common man. Policies should be inclusive of the outreach programs to ensure that the common man's representation while addressing air pollution as a public health issue.
- (c) Stringent implementation of the policies, laws, and rules needs to be ensured in order to meet clean air targets at the subnational and national level. Regular monitoring and review of the existing structure should be carried out for efficient policy actions. Also, the laws need to be more composite and adopt a multipronged approach to ensure maximum cobenefits.

17.3.3.2 Way Forward

Air pollution remains a grave environmental problem having larger adverse impacts on the developing countries with limited resources to mitigate the problem. There exist several international and national policy mechanisms that have paved a way for devising control strategies based on scientific solutions. However, these policy actions have not always been a win-win owing to the complex nature of the problem involving multiple sectors and stakeholders, including governments. Learning from successful examples of coordinated implementation and timely achievement of objectives, as in the case of Montreal Protocol, it is imperative to build institutions that mature in their approach as the science on the issue develops. This is particularly applicable in the case of air pollution extremes experienced in India in the winter season, where it is crucial to look for technological innovation while enduring coordination between multiple agencies. Policy solutions can be more effective if they are informed by decision ready science. The institutional and governance challenges to policy solutions are unique to the sectoral emissions and specific sources in each region locally. Even within India, provincial policies adopt a different approach than national policies to target specific sources within a given timeline. Similarly, chronic and extreme air pollution events call for adopting different frameworks. Also, existing policies, laws, and regulations point are the needs of review and increasing interdisciplinary nature of the mitigation measures while adopting a cobenefits approach. However, policy mitigation can be efficient only if the policies are all inclusive, built on a strong science foundation, and engage multiple stakeholders in the process to control and abate air pollution, globally and locally.

References

- Aakvik A, Tjøtta S (2011) Do collective actions clear common air? The effect of international environmental protocols on sulphur emissions. Eur J Polit Econ 27(2):343–351
- Bhatta GD, Aggarwal PK, Shrivastava AK, Sproule L (2016) Is rainfall gradient a factor of livelihood diversification? Empirical evidence from around climatic hotspots in Indo-Gangetic Plains. Environ Dev Sustain 18(6):1657–1678
- Bhave PP, Kulkarni N (2015) Air pollution and control legislation in India. J Inst Eng 96:259-265
- Brauer M, Freedman G, Frostad J et al (2013) Ambient air pollution exposure estimation for the global burden of disease. Environ Sci Technol 50(1):79–88
- Central Pollution Control Board, Graded Response Action Plan (2016) Graded Response Action Plan for Delhi and NCR. Retrieved from https://cpcb.nic.in/uploads/final_graded_table.pdf
- Dominici F, Peng RD, Barr CD, Bell ML (2010) Protecting human health from air pollution: shifting from a single-pollutant to a multipollutant approach. Epidemiology 21(2):187–194
- Ganguly T, Selvaraj KL, Guttikunda SK (2020) National Clean Air Programme (NCAP) for Indian cities: review and outlook of clean air action plans. Atmos Environ X 8. https://doi.org/10.1016/ j.aeaoa.2020.100096

Gothenburg Protocol (2021). https://unece.org/gothenburg-protocol. Last accessed 29 Jan 2021

- Green BA (2009) Lessons from the Montreal Protocol: guidance for the next international climate change agreement. Environ Law 39:253
- Gulia S, Goyal SK, Kumar R (2021) Air pollution episode analysis and qualitative evaluation of proposed control measures in Delhi City. In: Urban air quality monitoring, modelling and human exposure assessment. Springer, Singapore, pp 225–237
- Guttikunda SK, Gurjar BR (2012) Role of meteorology in seasonality of air pollution in megacity Delhi, India. Environ Monit Assess 184(5):3199–3211
- Health Effects Institute (2018) Burden of disease attributable to major air pollution sources in India
- Heath EA (2017) Introductory note to the Montreal Protocol on substances that deplete the ozone layer (Kigali Amendment). Int Leg Mater 56(1):193–205
- Kigali Amendment, UNIDO (2021) https://www.unido.org/our-focus/safeguarding-environment. Accessed 29 Jan 2021
- Li Z, Wang Y, Guo J, Zhao C, Cribb MC, Dong X et al (2019) East Asian study of tropospheric aerosols and their impact on regional clouds, precipitation, and climate (EAST-AIRCPC). J Geophys Res Atmos 124:13026–13054
- Lorenz M, Becher G (2014) In: Michel A, Seidling W (eds) Forest condition in Europe: 2013 technical report of ICP forests; report under the UNECE convention on long-range transboundary air pollution (CLRTAP). Thünen-Institut, Bundesforschungsinstitut für Ländliche Räume, Wald und Fischerei
- Michel A, Seidling W (2016) Forest condition in Europe: 2016 technical report of ICP forests: report under the UNECE convention on long-range transboundary air pollution (CLRTAP)
- MoEF (2019) National Clean Air Program (NCAP). http://moef.gov.in/wp-content/uploads/2019/0 5/NCAP_Report.pdf
- Montreal Protocol (1987) Montreal protocol on substances that deplete the ozone layer, vol 26. US Government Printing Office, Washington, DC, pp 128–136
- Montreal Protocol (2021). https://www.unenvironment.org/ozonaction/who-we-are/aboutmontreal-protocol. Last accessed 29 Jan 2021
- Mosier AR, Bleken MA, Chaiwanakupt P, Ellis EC, Freney JR, Howarth RB et al (2002) Policy implications of human-accelerated nitrogen cycling. In: The nitrogen cycle at regional to global scales. Springer, Dordrecht, pp 477–516
- National Policy for Management of Crop Residue (2014). http://agricoop.nic.in/sites/default/files/ NPMCR_1.pdf. Last accessed 29 Jan 2021
- Nickovic S, Cuevas Agulló E, Baldasano JM, Terradellas E, Nakazawa T, Baklanov A (2015) Sand and dust storm warning advisory and assessment system (SDS-WAS) science and implementation plan: 2015–2020
- Oslo Protocol (2021) Protocol on Further Reduction of Sulphur Emissions. https://unece.org/ environment-policyair/protocol-further-reduction-sulphur-emissions. Last accessed 29 Jan 2021

- Polonara F, Kuijpers L, Peixoto R (2017) Potential impacts of the Montreal Protocol Kigali Amendment to the choice of refrigerant alternatives. Int J Heat Technol 35(Special Issue 1): S1–S8
- Ringquist EJ, Kostadinova T (2005) Assessing the effectiveness of international environmental agreements: the case of the 1985 Helsinki Protocol. Am J Polit Sci 49(1):86–102
- Saxena P, Sonwani S (2019a) Policy regulations and future recommendations. In: Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 127–157
- Saxena P, Sonwani S (2019b) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157
- Saxena P, Srivastava A (eds) (2020) Air pollution and environmental health. Springer-Nature, Singapore, pp 1–253
- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- Sharma D, Kulshrestha UC (2014) Spatial and temporal patterns of air pollutants in rural and urban areas of India. Environ Pollut 195:276–281
- Sharma VK, Singh S (2017) Strengthening and restructuring-Central Pollution Control Board. Full text-IIPA, New Delhi
- Sterk W, Arens C, Kreibich N, Mersmann F, Wehnert T (2012) Sands are running out for climate protection: the Doha climate conference once again saves the UN climate process while real climate action is shelved for later
- Tiwari S, Thomas A, Rao P, Chate DM, Soni VK, Singh S et al (2018) Pollution concentrations in Delhi India during winter 2015–16: a case study of an odd-even vehicle strategy. Atmos Pollut Res 9(6):1137–1145
- UNCCD (1994) United Nations convention to combat desertification in those countries experiencing serious drought and/or desertification, especially in Africa. UNCCD Secretariat, Bonn. http://www.unccd.int/convention/text/pdf/conv-eng.pdf; http://www.unccd.int/convention/text/pdf/conv-e
- UNCLRTAP (1979) UNA common framework for transboundary cooperation on air pollution. UN Convention on Long-range transport transboundary air pollution. https://www.unece.org/fileadmin/DAM/env/lrtap/full%20text/1979.CLRTAP.e.pdf
- UNECE (1979). https://unece.org/fileadmin//DAM/env/lrtap/lrtap_h1.htm. Last accessed 29 Jan 2021
- United Nations Economic Commission for Europe (UNECE) (2015) Technical report No 8/2015. Long-range Transboundary Air Pollution (LRTAP) report 1990-2013.pdf. Retrieved from https://www.eea.europa.eu/publications/lrtap-emission-inventory-report
- UNEP (2016) Dust forecasts produced by the WMO SDS-WAS Regional Center for Northern Africa, Middle East and Europe and the Barcelona Dust Forecast Center are available on UNEP Live. http://uneplive.unep.org/region/index/AF#data_tab
- Velders GJ, Andersen SO, Daniel JS, Fahey DW, McFarland M (2007) The importance of the Montreal Protocol in protecting climate. Proc Natl Acad Sci 104(12):4814–4819
- Vestreng V, Myhre G, Fagerli H, Reis S, Tarrasón L (2007) Twenty-five years of continuous sulphur dioxide emission reduction in Europe. Atmos Chem Phys 7(13):3663–3681
- Von Schneidemesser E, Monks PS, Allan JD, Bruhwiler L, Forster P, Fowler D, Lauer A, Morgan WT, Paasonen P, Righi M, Sindelarova K (2015) Chemistry and the linkages between air quality and climate change. Chem Rev 115(10):3856–3897
- Wettestad J (2001) The convention on long-range transboundary air pollution (CLRTAP). See Ref 42:197–221
- WHO (2016) Ambient air pollution: a global assessment of exposure and burden of disease. World Health Organization, Geneva
- Wiesinger F, Sutter F, Fernández-García A, Reinhold J, Pitz-Paal R (2016) Sand erosion on solar reflectors: accelerated simulation and comparison with field data. Sol Energy Mater Sol Cells 145:303–313
- World Bank (2008) Sustainable land management sourcebook. The World Bank, Washington, DC



18

Participatory and Collaborative Governance Approach for Management of Atmospheric Processes-Related Extremities

Kamna Sachdeva

Abstract

Management of atmospheric processes-related extremities is a challenging task for all kinds of regulatory authorities. It is a need of an hour to devise a methodology for which collaborative and participatory approach is important, so that micro limits of such extremities can be defined and identified. Unidirectional carrying capacity-based approach has been used by many researchers however multi-disciplinary, multidirectional approach is required where accountability should be in the core. Formulation of policies and its effective implementation with limited empowered work force is near to impossible; hence, bottomup approach where decentralised and self-governance model is the only hope. The active and passive involvement of stakeholders is part of effective decisionmaking process, government agencies need people's participation to manage public programs, its implementation and also for the mass awareness. The chapter talk through different approaches with national and international examples to understand effectiveness of participatory model. Participation of different stakeholders in decision-making process and managing the extremities is not straightforward process, hence role of training for state and non-state actors in this respect is important, for better governance and management.

Further to this, role of participatory approach as potential solution to atmospheric extremities and its evolution has also been discussed. Chapter deliberates on different steps of participatory approach and postulates of effective management and response networks for the atmospheric processes-related extremities (APE).

K. Sachdeva (🖂)

Department of Energy and Environment, TERI School of Advanced Studies, Delhi, India e-mail: kamna.sachdeva@terisas.ac.in

 $^{{\}rm \textcircled{O}}$ The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4_18

Keywords

Atmospheric processes · Extremities · Governance · Stakeholders

18.1 Introduction

Atmospheric extremities are unexpected events which have an ability to cause great damage to people, resources and property. As per the Environment report 2020, almost 1300 deaths have been reported in India due to such events in year 2019 and out of which more than 50% of causalities reported to occur due to extreme rain events. Over the decade, these atmospheric extremities have increased many folds, cloud bursts and hail storms frequencies are important indicators of such extremities; hence, planning and coordination are important to minimise the damage (IPCC 2018). The chapter will provide insight of different aspects of governance of such events. The typical governance tools such as meteorological predictions, contingency funds and graded action plans are at place, but for managing the extremities may also need coordinated people's participation, which feeds to the system for better governance and management. The management of such events are not straightforward, because scientific uncertainties and conflicting interest can influence decisions. Hence, active participation of all the stakeholders and citizens in the policy cycle is must (Melica et al. 2018).

Participatory approach where people of the region are part of decision making is gaining acceptance to improve accountability, but in atmospheric processes people participation with self-governance is required (Richardson and Razzaque 2006). Here it is required that people participate in implementation of governance. The main pillars of participatory planning are communication, knowledge distribution, public learning, integration with different sectors and decentralisation of power. The management of APE demands the change of mind set of social and political systems to minimise the risk, people-centric growth model instead of GDP-based, should be promoted. All these pillars need to work together for managing atmospheric extremities such as smog events of Delhi. The smog event management is a success story that has brought all the stakeholders together for policy deliberation and to lead enforcement and implementation of the response plan. Although Delhi being capital, has advantage of having institutional and legal mechanisms at place which is utmost necessary for better governance. The figure describes the plausible steps which can fit into the governance model for managing Atmospheric Processes-Related Extremities (Fig. 18.1).

The integration of soft approach with other scientific approaches of risk assessment and management of atmospheric extremities needs planning and training (Palermo and Hernandez 2020). The public participation approaches change the governance model and make it more meaningful (Hernandez et al. 2018). When people start participating in decision making, then improvement in governance will takes place, although it may delay the process due to varied perception of stakeholders (Brink and Wamsler 2019).



Fig. 18.1 Steps involved in effective governance of APE

Arnstein (1969) has given ladder of people participation, which goes from simple manipulation level to full citizen control step. Based on Arnstein model, training on deciding optimum level of people participation is also important so that decision making will not be delayed. André et al. (2006) describe passive and active methods of participation, both are important for managing atmospheric extremities; however, both cannot be implemented at the same time. In passive approach of participation, stakeholders receive information and consulted. On the other side, active participation requires mediation, negotiation and invitation of collaboration. The case of Urban heat Island (UHI) management in Kuala Lumpur (Malaysia) is a good example where involvement of different stakeholders has provided best governance model of urban planning to manage this extreme event (Ramakreshnan et al. 2019). Here focus group discussions have been used as tool to acquire information of awareness level of policy makers on UHI. The outcome of the study indicates the relevance of awareness and knowledge creation among non-government and government actors for effective policy making and urban management practices.

18.2 Rationale and Objective

The first and foremost objective of any kind of regulatory decision is to protect public interest and empowered public with adequate knowledge related to the impacts and risk. In planning and management of atmospheric extremities, the range of solutions viz, public participations, formation of review boards, aim to provide theoretical perspective of participatory and collaborative governance for such events has been adopted. Over the period this has been realised that risk aversion and damage reduction can only be possible by public awareness and adequate knowledge of the process. The enabling environment, organisational development and social participations to promote such instruments are required for effective management of APE. The motivation of this work is to advocate public participation and public empowerment as participatory tool for management of atmospheric processes-related extremities.

18.3 Atmospheric Extremities

In the changing climate regime, atmospheric extremities and its governance has been subject of discussions. Atmosphere itself is dynamic in nature and its processes are interdependent on various feedback mechanisms operating in the system. Synergistic effect of ozone and PM2.5 concentrations is a major concern for various countries; many researchers are working on downscaled models to understand atmospheric extremities. Outputs of GCMs have been used to prepare regional level future emission inventories along with energy pathways to fully characterise the range of possible scenarios for future atmospheric extremities. Based on law of physics, we can predict state of atmosphere but management of its extreme behaviours needs integration of various disciplines such social sciences', human behaviour, community development and management and governance. Although before managing the extremities of atmosphere, we need to have effective predictions of deterministic nature of the atmosphere (Saxena and Sonwani 2019).

However, due to interconnectivity of varied and diverse sub-systems viz; environmental, health, society, etc., atmospheric extremities can become one of the systemic risks. In this kind of risk, inappropriate interactions and institutional setting may lead to disasters in anthropogenic systems. There are many theories such as catastrophe, theory of critical phenomenon and self-governance that have explained how the local failure of links may result in consequential failure of other links. Hence, strengthening of conscious decision making with the adequate public participation is important. Risks posed by air pollution and related disasters can be tackled in various ways, requirement is to provide a clear approach and controls at place to protect our society; however, such approaches generally fail, if community does not participate. First is to prioritise the awareness and provide education to the people about the associated risk and their consequences. The common man should be aware of the hyper-risk (interconnected risk) they are dealing with and the precautionary measures that could be taken in such a condition. Any kind of negligence or lack of governance and poor planning could result to worsen the condition of a hyper-risk. Hence, proper communication and well implementation of emergency response plan and graded action plans are quite essential to minimise the impacts. Information dissemination is a key for Participatory and Collaborative Governance.

Interconnected world has given us the good life as well as the depraved and dangers of the post-modern societies. To tackle the hyper-risks, a reflective response is much needed, including, an individual's knowledge about the process, creating an

Public participation	
spectrum	Goal of participation in management of APRE
Inform	Lower participation is expected, only unidirectional flow of information. Output and outcomes are not very effective (example AQI information for the public)
Consult and collaborate	This comes in middle level where government consults stakeholders and collaborates with them before implementation of the intervention. This is been seen in the case of stubble burning and associated emissions in Indo-gangetic plain. Farmers have been consulted by the local governments and try to collaborate with them for the solutions. One of the successful interventions is the use of agriculture residue in the biomass-based power plants developed in the vicinity of the problematic regions
Empowerment	This governance effort is at the highest level of the ladder, the self- governance and control of the situation by the active participation of the community is the example of such empowerment SPARE THE AIR the campaign. The program encourages participants to cut polluting activities, specifically in transport sector, in favour of less polluting alternatives. The self-governance model of participation falls in this category: a three-stage model where first stage is to capture perceptions and expectation of the people about any negative externality (viz, smog, bad air quality and storms, etc.), second stage is integration of participatory tool such as training workshop—this step is very crucial and involves two types of participatory approaches; knowledge sharing and willing to learn. Finally, in stage three, where effects and benefits of self-governance will be explained

Table 18.1 Management participation indexed with public participation spectrum or governance levels in context of air pollution management

organised and collective efforts with organised learning and promoting the research in the arena of interconnected risks like atmospheric extremities.

The table given below gives public participation spectrum in the context of atmospheric extremities, this explains at different levels what kind of involvement and participation from local public is required for the better management and risk aversion of extremities (Table 18.1).

18.4 Participatory and Collaborative Governance Model

Assessment of participatory tools is not an easy task but work of Van der Stroom (2017) has provided toolkit which makes it easy to assess, the broad participatory and governance framework is given below which is applicable in different extreme events, the effectiveness of the framework is directly proportional to the efforts of regulatory authorities to seek participation of stakeholders:

1. Communication and sharing of knowledge: It is important to exchange ideas and thoughts with other participants and the response of participants to each other.

This way information is processed and become usable for policy making (Patel et al. 2007; Pelzer et al. 2013).

- 2. Social Learning: Knowledge of local problems and strategies for the area, this process helps in reframing the perceptions (Pelzer et al. 2014; Reed et al. 2010; Volk et al. 2010).
- 3. Transparency: There should not be power difference this approach should empower stakeholders to raise voice (Goosen et al. 2007).
- 4. Integration: Stimulating integration between different contributing factors; this integration will provide better understanding of problem of particular area (Van der Stroom 2017; Volk et al. 2010).
- Level of Agreement: Conflict resolution and giving consent on appropriate policy decision. This consent can, for example, be on to-be implemented strategies or projects (Hassenforder et al. 2015; Patel et al. 2007).

18.5 Models of Different Countries/Regions

18.5.1 European Union (EU) Model of Air Quality Management

EU's work on long-term benefit approach where main objective levels of air quality which do not harm or pose risk to human health and the environment. The EU acts at different levels to minimise impacts: through legislation; cooperation with sectors responsible for air pollution, as well as international, national and regional authorities and non-governmental organisations and research. The European Commission in 2013 adopted a Clean Air Quality Package which includes innovative measures to curb air pollution. The focused areas are:

- · Public display of air pollution data and exceedances
- Trend analysis of air pollution and policy actions on curbing it in Europe
- Investigating the trade-offs and synergies between air pollution and policies in different areas, including climate change, energy, transport and industry

Clean air policy package of Europe

- An updated National Emission Ceilings (NEC) Directive, with emission ceilings for the years from 2010 onward, and new national emission reduction commitments from 2020 and 2030.
- A new Directive on Medium Combustion Plants, to limit emissions of nitrogen oxides, sulphur dioxide and particulate matter from medium-sized combustion installations. The directive proposes emission limit values for new and existing installations.
- Additional actions focusing on air quality in cities, national and local actions supported by EU funds, as well as a reinforced research and innovation agenda.
- It is people-centric package where transparency and accountability are the keys of success although legal binding in the periphery provides support to this approach.

18.5.2 China Model of Air Pollution Control

The control of Beijing's air pollution achievement can be attributed to its governmental structure in some extent and local sustainable development approach viz, strong willingness, clear goal, supportive legislation, plan and policies, implementation and enforcement arrangement United Nations Environment Programme (2019). Engaging the public in these objectives has strengthened environmental protection even further and increased social harmony. The former Beijing Municipal Environmental Protection Bureau recently reformed into Beijing Municipal Ecology and Environment Bureau (BEE) has integrated with more missions on promoting ecological civilisation. They are committed for blue sky, better environment and happy life for the people of the city. They have agenda to practice at local level with full efforts—to connect local goals with vision of 2030 sustainable goals. To establish effective air pollution management system, many policies and reforms have been implemented simultaneously such as subsidies, fees, pricing and other financial practices, to provide economic incentives for the effective implementation of various measures (Figs. 18.2 and 18.3).

Although great air quality improvement has been seen in Beijing but still it faces challenges to control future air pollution and related problems. The way out is continuous improvement of the atmospheric environment and more vigilant approach.

18.5.3 Indian Scenario of Participatory Governance

Air pollution is a big challenge in India; most of the cities violate regulatory standards. It is reported that IGP (Indo-Gangetic Plains) has higher concentrations of surface ozone and particulate pollution (Saxena et al. 2021). Owing to its geographical and agricultural significance, IGP is an important hotspot for analysing the impact atmospheric-induced extremities. The synergetic impact of changing climate and air pollution can make this region very vulnerable for meteorological-related extremities and disasters.

The traditional knowledge and local perspective are important for governments to take responsive decisions for the region. Lack of people participation is impediment for the government; hence, watchdogs through panchayats have been made to ensure participation. Over the period, centralised planning process has moved to participatory governance process in IGP specifically in framing communities of the region. Participatory planning and governance process are providing promising solutions to the climate- and atmospheric-induced disasters in the regions. The hopeful example which can be replicated in disaster management is Community Food and Water Security System which has been formulated by the local farming communities. The model is comprising of field Gene banks and seed bank: where focus is on lifesaving crops, which restrain extreme temperatures and stress. Water bank: skewed rainfall is witnessed in IGP region, community driven self-help groups have been created for the promotion of water harvesting and equitable use of water in crisis





1998	1999 200	0 2001 20	02 20	03 2004	2005	2006 2007	2008	2009 2010	2011	2012	2013	2014 20	15 2016	5 2017
Emission standard	<u>ا</u>													
Light-duty gasoline vehicle	> 199	9/1-2002/12 China 1	>	2003/1-2005/12 China 2	>	2005/12-2008/2 China 3	\rightarrow	2008	3/3-2013/ Dhina 4	1	>		2013/2- China 5	
Heavy-duty diesel vehicle	Σ	2000/1-2002/12 China I	$\boldsymbol{\Sigma}$	2003/1-2005/12 China II	>	2005/12-2008/0 China III		2008/7-2013/1 China IV ¹	> 2	013/7-2015 China IV ²	/5	2013/2- China V ¹	\rightarrow	2015/6- China V ²
Improve gasoline quality	Unleaded	Reducing	Σ	2003/1-2005/6 <500ppm	>	2005/6-2007/12 <150ppm	X	2008/1- <50p	2012/5 pm		>	2012/6- <10ppm	>	2017/1
Improve diesel quality	gasoline sulfur since 1998 content	Σ	2003/1-2005/6 <500ppm	>	2005/6-2007/12 <350ppm	>	2008/1- <50p	2012/5 pm		>	2012/6- <10ppm	Ch qu	ina 6/VI fuel lity standard	
I/M programs and in- use inspections		200 Trat	1-2003 isition	\gg	R	Il implementation o	2003- of ASM an	d LugDown metho	ods		On	2013- -road RS ³	2015- P	EMS ⁴ test dic buses
Traffic restrictions on specific in-use fleets				\geq		2003- Ye 2004- Freig	How-lab	elled vehicles and motorcycles			\rightarrow	2017/2- I passenge	ight-duty ga r cars of Chi	asoline na 1/2
Clean energy and new energy vehicles		>	C	1999- IG public buses		\rightarrow	> 200	 New energy veh Ten City & Thousa 	nicles pror and Units*	noted by th program	\supset	2014-6	lectric publi and taxies	ics buses
Traffic management	and					2008/10- Dr light-duty	iving rest passeng	riction on er cars	2014-	Tighten the spec	2011- Li quota of al quota f	cense control new light-duty for new energy	passenger (cars, a
economic measure	15					2009- Subs yellow-l	idized scr abelled v	appage of hicles	2011- Ex to older	panded vehicles		yellow-lab sci	elled vehi apped	icles were
Temporary measurements					\geq	The 2008 Oly restrictions, ban reducing use of g	mpic Gan on yellow overnme	es: Odd-even labelled vehicles, nt-owned vehicles	>	The 2014 AP the 2017 fo even restric	FEC Summ rum on th tions, redi	it, the 2015 m the Belt and Roa ucing use of go vehicles	litary parad d Initiative: vernment-c	e, and Odd- owned

Fig. 18.3 Vehicular control measures review (1998–2017) (Source: Former Beijing Municipal Environmental Protection Bureau, Tsinghua University)

Table 18.2 Important initial steps required for the management-oriented outcomes for the governance of atmospheric extremities

Emission monitoring and advisories	Informed decision making
Timely and proactive	Convincing of governments for action and release of
scenario planning	funds for resilience planning

situation. Local level grain banks are another such example where community selfhelp groups are operated by revolving funds, this model provides sustainable food security in distress time.

18.6 Conclusion and Recommendations

As we know half of the world is facing problem of managing extreme atmospheric events, development drivers such as poverty, malnutrition and poor adaptive capacity pose additional threat to the governance. Even in the regions where economic growth and technological and innovation reforms are at place, finds it most challenging governance issue. Climate change is another driver which makes its management more complicated. Need is to mainstream all the atmospheric extremities in the discussion of climate change governance, else it will increase the cost of mitigation. We are at the stage where best efforts are unlikely to manage the threats of atmospheric extremities, as expansion of energy and urban systems is imperative. Hence, dedicated participatory and inclusive approach is required to protect the vulnerable population. The table given below is suggestive of such model along with steps given in table, citizen awareness and actions can build pressure on institutional mechanisms to act and respond. Non state actors' mobilization is important for better governance and management of atmospheric extremities. All

these steps provide empowerment to the good governance system for action (Table 18.2).

In the nutshell, management of atmospheric extremities is a complex process, educating people and ensuring their participation is important. People need to understand how they are affected and what are available solutions for their problems. Government should take responsibility for ensuring the participation to make process inclusive. There is no dearth of access to information, the appropriate evidence based information can be used by the people for influencing the decision of government; hence, it is more effective for governance process to involve people at early stages of management.

References

- André P, Enserink B, Connor D, Croal P (2006) Public participation: international best practice principles. Spec Publ Ser 4:1–3
- Arnstein SR (1969) A ladder of citizen participation. J Am Inst Plann 35(4):216-224
- Brink E, Wamsler C (2019) Citizen engagement in climate adaptation surveyed: the role of values, worldviews, gender and place. J Clean Prod 209:1342–1353
- Goosen H, Janssen R, Vermaat JE (2007) Decision support for participatory wetland decisionmaking. Ecol Eng 30(2):187–199. https://doi.org/10.1016/j.ecoleng.2006.11.004
- Hassenforder E, Smajgl A, Ward J (2015) Towards understanding participatory processes: framework, application and results. J Environ Manag 157:84–95. https://doi.org/10.1016/j.jenvman. 2015.04.012
- Hernandez Y, Pereira ÂG, Barbosa P (2018) Resilient futures of a small island: a participatory approach in Tenerife (Canary Islands) to address climate change. Environ Sci Pol 80:28–37
- IPCC (2018) Global warming of 1.5°C, world meteorological organization, Geneva, Switzerland
- Melica G, Bertoldi P, Kona A, Iancu A, Rivas S, Zancanella P (2018) Multilevel governance of sustainable energy policies: the role of regions and provinces to support the participation of small local authorities in the covenant of mayors. Sustain Cities Soc 39:729–739
- Palermo V, Hernandez Y (2020) Group discussions on how to implement a participatory process in climate adaptation planning: a case study in Malaysia. Ecol Econ 177:106791
- Patel M, Kok K, Rothman DS (2007) Participatory scenario construction in land use analysis: an insight into the experiences created by stakeholder involvement in the Northern Mediterranean. Land Use Policy 24(3):546–561. https://doi.org/10.1016/j.landusepol.2006.02.005
- Pelzer P, Arciniegas G, Geertman S, de Kroes J (2013) Using map TableVR to learn about sustainable urban development. In: Planning support systems for sustainable urban development. Springer, Berlin, Heidelberg, pp 167–186
- Pelzer P, Geertman S, van der Heijden R, Rouwette E (2014) The added value of planning support systems: a practitioner's perspective. Comput Environ Urban Syst 48:16–27. https://doi.org/10. 1016/j.compenvurbsys.2014.05.002
- Ramakreshnan L, Aghamohammadi N, Fong CS, Ghaffarianhoseini A, Wong LP, Noor RM, Hanif NR, Azriyati Wan Abd Aziz WN, Sulaiman NM, Hassan N (2019) A qualitative exploration on the awareness and knowledge of stakeholders towards urban Heat Island phenomenon in greater Kuala Lumpur: critical insights for urban policy implications. Habitat Int 86:28–37
- Reed M, Evely A, Cundill G, Fazey I, Glass J, Laing A, Newig J et al (2010) What is social learning? Ecol Soc 15(4):r1. http://www.ecologyandsociety.org/vol15/iss4/resp1/
- Richardson B, Razzaque J (2006) Public participation in environmental decision making
- Saxena P, Sonwani S (2019) Criteria air pollutants and their impact on environmental health. Springer, Singapore, pp 1–157

- Saxena P, Sonwani S, Srivastava A, Jain M, Srivastava A, Bharti A, Rangra D, Mongia N, Tejan S, Bhardwaj S (2021) Impact of crop residue burning in Haryana on the air quality of Delhi, India. Heliyon 7(5):e06973
- United Nations Environment Programme (2019) A review of 20 year's air pollution control in Beijing. ISBN 978-92-807-3743-1
- Van der Stroom J (2017) Evaluating participatory planning tools. Internship Report. Wageningen University, Deltares, IHE Delft, Wageningen/Delft
- Volk M, Lautenbach S, van Delden H, Newham LT, Seppelt R (2010) How can we make progress with decision support systems in landscape and river basin management? Lessons learned from a comparative analysis of four different decision support systems. Environ Manag 46(6): 834–849. https://doi.org/10.1007/s00267-009-9417-2

Websites

http://envfor.nic.in/division/air-pollution https://ec.europa.eu/environment/air/clean_air/index.htm

Glossary

- Absorption Absorption is a chemical or physical process in which ions, molecules or atoms go in the bulk phase. It involves the transmission of a gaseous pollutant from the air in the water.
- Adsorption In adsorption, the molecules of a fluid stick to the surface of a solid material called adsorbent. This method is commonly used to recover volatile solvents, i.e. benzene, toluene and chlorofluorocarbon, process gas streams drying and control the odour.
- Aerosol Optical Depth (AOD) Aerosol Optical Depth (AOD) is the measure of aerosols (e.g. urban haze, smoke particles, desert dust, sea salt) distributed within a column of air from the instrument (Earth's surface) to the top of the atmosphere.
- Aerosol It consists of very finely subdivided liquid or solid particles dispersed in and surrounded by a gas.
- Air Pollution Tolerance Index (APTI) It is one of the important parameters that could be considered for the selection of traffic barrier plant species. It is an important tool to categorize plants based on their tolerances or sensitivity levels into air pollutants. APTI of plants has been described with four biochemical parameters: total chlorophyll, relative water content (RWC), ascorbic acid and leaf extract pH.
- Anthropogenic Emissions The emissions which have their origin from man-made activities like vehicular exhaust, industrial activities, domestic activities, etc.
- Antioxidant An organic compound that accepts free radicals and thus prevents the auto-oxidation of fats and oil.
- Arid region A land region of low rainfall (less than 250 mm/year).
- Atmosphere A gaseous envelope surrounding the earth. Atmosphere also contains clouds and aerosol.
- Atmospheric Phenomena As commonly used in weather observing practice, an observable occurrence of particular physical (as opposed to dynamic or synoptic) significance within the atmosphere.
- Atmospheric Processes Atmospheric processes play the lead role in determining such fundamental properties of climate as the disposition of incoming solar radiation, temperatures at the Earth's surface, the spatial distribution of water in

P. Saxena et al. (eds.), *Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation*, Disaster Resilience and Green Growth, https://doi.org/10.1007/978-981-16-7727-4

the terrestrial biosphere and the distribution of nutrients in the euphotic zone of the ocean.

- Bioaerosols Airborne particles produced from microbial, viral, fungal and actinomycete, as well as microbes from organic materials, humidifiers, vaporizers, heating, ventilating and air conditioning systems (HVAC) lead to allergies, infections and poisoning.
- Biodiversity Number and relative abundances of different genes (genetic diversity), species and ecosystems (communities) in a particular area.
- Bioindicators Bioindicators are living organisms such as plants, planktons, animals and microbes, which are utilized to screen the health of the natural ecosystem in the environment. They are used for assessing environmental health and biogeographic changes taking place in the environment.
- Biomass Total mass of living organisms in a given area or volume; recently dead plant material often included as dead biomass.
- Biomonitoring Biomonitoring involves the use of organisms to assess environmental contamination, such as of surrounding air or water. It can be done qualitatively by observing and noting changes in organisms or quantitatively by measuring accumulation of chemicals in organism tissues.
- Biosphere The part of earth system comprising all ecosystems and living organisms in the atmosphere.
- Carbon cycle The term used to describe the flow of carbon through the atmosphere, ocean, terrestrial biosphere and lithosphere.
- Carbon sequestration The process of storing carbon dioxide. This can happen naturally, as growing trees and plants turn CO_2 in to biomass (wood, leaves and so on). It can also refer to the capture and storage of CO_2 produced by industry.
- Chronic Obstructive Pulmonary Disease (COPD) Chronic obstructive pulmonary disease, commonly referred to as COPD, is a group of progressive lung diseases. The most common of these diseases are emphysema and chronic bronchitis. Many people with COPD have both of these conditions.
- Cloud A visible mass of particles of condensed vapour (such as water or ice) suspended in the atmosphere of earth.
- Coarse Particulate Matter They are also known as $PM_{10-2.5}$; particles with diameters generally larger than 2.5 µm and smaller than, or equal to, 10 µm in diameter. Note that the term **large coarse particles** in this course refers to particles greater than 10 µm in diameter.
- Criteria Air Pollutants The criteria air pollutants include particle pollution, groundlevel ozone, carbon monoxide, sulphur dioxide, nitrogen dioxide and lead. These pollutants can harm your health and the environment and cause property damage. Of the six pollutants, particle pollution and ground-level ozone are the most widespread health threats. EPA calls these pollutants "criteria" air pollutants because it regulates them by developing human health-based and/or environmentally based criteria (science-based guidelines) for setting permissible levels.

- Cyclone Separator It is a renowned technology that is primarily used for coarse and medium-sized particles collection. It works by downward forcing of a gasified suspension.
- Dew Moisture condensed upon the surfaces of cool bodies especially at night.
- Disaster Risk Management Disaster risk management is the application of disaster risk reduction policies and strategies to prevent new disaster risk, reduces existing disaster risk and manage residual risk, contributing to the strengthening of resilience and reduction of disaster losses.
- Dissolved Organic Carbon Dissolved organic carbon is the predominant material that absorbs UV, and rapid attenuation occurs with moderate dissolved organic carbon concentrations (i.e. more than several mg C/L).
- Dispersion Modelling Dispersion modelling uses mathematical formulations to characterize the atmospheric processes that disperse a pollutant emitted by a source.
- Dobson Unit One Dobson Unit is defined as the number of molecules of O_3 required to create a pure O_3 layer of 0.01 mm thickness at a temperature of 0 °C and sea level (1 atm pressure).
- Dry Deposition Primary and secondary particulates are removed by the gravitational force from the atmosphere over time without precipitation, this process is termed as dry deposition.
- Dynamic meteorology Dynamic meteorology is usually referred to a meteorology study that focuses on fluid mechanism and thermodynamics of atmosphere since the atmosphere is considered as a continuous fluid medium.
- El Nino El Nino is a climate pattern that describes the unusual warming of surface waters in the eastern Pacific Ocean. El Nino is the "warm phase" of a larger phenomenon called the El Nino-Southern Oscillation (ENSO).
- Emission Inventory Emission inventory is an important tool for identifying the source of pollutants and quantitative expression of pollution load in a defined area at a particular time.
- Environmental Hazard Environmental hazards are generally measured as exposure to pollution such as the amount of dust or sulphur dioxide in the air or the presence/amount of soil or groundwater contamination on the property.
- Epidemiology Epidemiology is the branch of medical science that investigates all the factors that determine the presence or absence of diseases and disorders. Epidemiological research helps us to understand how many people have a disease or disorder, if those numbers are changing, and how the disorder affects our society and our economy.
- Fine Particulate Matter It is also known as $PM_{2.5}$; particles generally 2.5 µm in diameter or smaller. This group of particles also encompasses **ultrafine** and **nanoparticles** which are generally classified as having diameters less than 0.1 µm.
- Fog Fog is a visible aerosol comprising tiny water droplets or ice crystals suspended in the air at or near the Earth's surface.
- Glaze Ice coating that forms when supercooled rain, drizzle or fog drops strike surfaces that have temperatures at or below the freezing point; the accumulated water covers the surface and freezes relatively slowly.
- Hail Precipitation in the form of small balls or lumps usually consisting of concentric layers of clear ice and compact snow.
- Hydrometeors Hydrometeors consist of liquid or solid water particles. They may be suspended in the atmosphere, fall through the atmosphere, be blown by the wind from the Earth's surface or be deposited on other objects. Snow or water on the ground is, by convention, not considered a hydrometeor.
- Indo-Gangetic Plain (IGP) It is also called **North Indian Plain**, extensive northcentral section of the Indian subcontinent, stretching westward from (and including) the combined delta of the Brahmaputra Valley and the Ganges (Ganga) River to the Indus River Valley.
- Indoor Air Pollution (IAP) Indoor air pollution is dust, dirt or gases in the air inside a building such as your home or workplace that harms us if we breathe it in.
- La Nina La Nina, the "cool phase" of ENSO, is a pattern that describes the unusual cooling of the region's surface waters.
- Lithometeor A conglomeration of small solid particles (as of dust or sand) that is suspended in the atmosphere and often produces a dry haze.
- Mitigation An anthropogenic intervention for lessening, reducing and diminishing the climatic stress and harm caused by adverse effects of climate change. It includes strategies to reduce air pollutants sources and emissions.
- Montreal Protocol Montreal Protocol is regarded as one of the most successful international environmental treaties which achieved its objective of prevention of ozone depletion by banning the production and use of ozone depleting substances that include CFCs, HCFCs and halons.
- National Clean Air Programme (NCAP) It was launched by the Ministry of Environment, Forests and Climate Change, Government of India, in 2019 as an urgent response to curtail and control growing air pollution levels in the country.
- National Health Adaptation Plan (NHAP) It has been designed and developed by the National Institute of Disaster Management, Ministry of Home Affairs, Government of India, to create enabling environment, build the capacity of climate resilient healthcare system; help institutions strengthen their preparedness, enhance the health system resilience in terms of universal health coverage; early warning, surveillance and community mobilization in the context of any health emergency risk management.
- Particulate Matter (PM) PM stands for particulate matter and it is a mixture of solid particles and liquid droplets found in the air.
- Quiescent Volcano A volcano, which is not active, but is still registering seismic activity. When there is no more seismicity, the volcano is dormant, but still capable of erupting.
- Radiative Forcing Radiative forcing is the change in the net, downward minus upward, radiative flux (expressed in Wm^{-2}) at the tropopause or top of

atmosphere due to a change in an external driver of climate change such as a change in the concentration of carbon dioxide or the output of the Sun.

- Rain Water falling in drops condensed from vapour in the atmosphere.
- Rime An accumulation of granular ice tufts on the windward sides of exposed objects that is formed from supercooled fog or cloud and built out directly against the wind.
- Seamounts They are submarine volcanoes that are extinct and rise abruptly from the sea floor. Despite the fact that no big explosive undersea eruption is yet documented, prevalence of such eruptions is evident from ancient seafloor deposits.
- Secondary Organic Aerosol (SOA) Secondary organic aerosol (SOA), formed in the atmosphere by oxidation of organic gases, represents a major fraction of global submicron-sized atmospheric organic aerosol.
- Sink Any process, activity or mechanism which involves a pollutant gas or aerosol from the atmosphere.
- Snow Precipitation in the form of small white ice crystals formed directly from the water vapour of the air at a temperature less than 32 °F (0 °C).
- Source Any process, activity or mechanism which releases a pollutant gas, an aerosol or a precursor of a pollutant gas or aerosol into the atmosphere.
- South Asian Monsoon It is an important segment of the tropical monsoon system that mainly lies in the range of the seasonal oscillations of the Intertropical Convergence Zone (ITCZ).
- Thermal Inversion Thermal inversion is a natural phenomenon that involves a change in the normal tendency of the air to cool down with altitude.
- Tracheobronchial region The tracheobronchial tree is composed of the trachea, the bronchi and the bronchioles that transport air from the environment to the lungs for gas exchange.
- Ultrafine Particulate Matter Ultrafine particles (UFPs) are aerosols with an aerodynamic diameter of 0.1 μ m (100 nm) or less. There is a growing concern in the public health community about the contribution of UFPs to human health.
- Volcano A volcano is a rupture on the Earth's crust through which hot lava, volcanic ash and gases are spewed out during the course of the eruption from a magma chamber underneath the surface.
- Wet Deposition Wet deposition is the process whereby atmospheric gases mix with suspended water in the atmosphere and are then washed out through rain, snow or fog.
- Wet Scrubber They are used for the control of mists, fumes, heavy metals, acid gases, suspended dust and trace organics.